As filed with the Securities and Exchange Commission on January 16, 2015

Registration No. 333-200695

UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

Amendment No. 1

to

FORM S-4/A

REGISTRATION STATEMENT UNDER THE SECURITIES ACT OF 1933

MANDALAY DIGITAL GROUP, INC.

(Exact name of registrant as specified in its charter)

Delaware(State or other jurisdiction of incorporation or organization)

(Primary Standard Industrial Classification Code Number) 22-2267658 (I.R.S. Employer Identification Number)

2811 Cahuenga Boulevard West Los Angeles, California 90068

(323) 472-5461

(Address, Including Zip Code, and Telephone Number, Including Area Code, of Registrant's Principal Executive Offices)

> William Stone Chief Executive Officer Mandalay Digital Group, Inc. 2811 Cahuenga Boulevard West Los Angeles, California 90068 (323) 472-5461

(Name, Address, Including Zip Code, and Telephone Number, Including Area Code, of Agent for Service)

With copies to:

Ben D. Orlanski, Esq. Katherine J. Blair, Esq. Manatt, Phelps & Phillips, LLP 11355 W. Olympic Blvd. Los Angeles, California 90064 (310) 312-4000

Approximate date of commencement of proposed sale of the securities to the public: As soon as practicable after this registration statement becomes effective and upon completion of the merger.

If the securities being re with General Instruction G, check	0	red in connection with the formation of a h	olding company and there is compliance
	C	fering pursuant to Rule 462(b) under the Se effective registration statement for the san	, .
1	1	Rule 462(d) under the Securities Act, checegistration statement for the same offering.	e
3	2 2	relerated filer, an accelerated filer, a non-a l filer" and "smaller reporting company" in	, , ,
Large accelerated filer	Accelerated filer □	Non-accelerated filer □	Smaller reporting company ⊠
If applicable, place an in the bo	x to designate the appropriate rule p	provision relied upon in conducting this trans	saction:
ε	4(i) (Cross-Border Issuer Tender C 1(d) (Cross-Border Third-Party Te		
	CALCIULATION	LOE DECICED A TION FEE	

CALCULATION OF REGISTRATION FEE

Title of each class of	Amount to be	Proposed maximum offering	Proposed maximum	Amount of registration
securities to be registered	registered (1)	price per unit	aggregate offering price	fee (2)
Common Stock, \$0.0001 par value per share	19,500,000 (3)	N/A	\$56,890 (4)	\$6.61*

- * Previously paid
- (1) Pursuant to Rule 416 under the Securities Act of 1933 (the "Securities Act"), this registration statement also covers an indeterminate number of additional shares of common stock, \$0.0001 par value per share ("common stock"), of Mandalay Digital Group, Inc. ("Mandalay Digital") as may be issuable as a result of stock splits, stock dividends or similar transactions.
- (2) Calculated by multiplying the proposed maximum aggregate offering price of securities to be registered by 0.0001162.
- (3) Represents the estimated maximum number of shares of Mandalay Digital common stock issuable to security holders of Appia, Inc. ("Appia") in the merger of DTM Merger Sub, Inc., a wholly-owned subsidiary of Mandalay Digital, with and into Appia, with Appia surviving as the wholly-owned subsidiary of Mandalay Digital.
- (4) Appia has an accumulated capital deficit, therefore, pursuant to Rule 457(f)(2) of the Securities Act, the proposed maximum aggregate offering price is one-third of the aggregate par value of Appia capital stock being acquired in the proposed merger, and the aggregate exercise (offering) price of employee stock options to be assumed in the transaction described herein, in accordance with Rule 457(h)(1) of the Securities Act. Appia is a private company and no market exists for its equity securities.

The Registrant hereby amends this Registration Statement on such date or dates as may be necessary to delay its effective date until the Registrant shall file a further amendment which specifically states that this Registration Statement shall thereafter become effective in accordance with Section 8(a) of the Securities Act of 1933 or until the Registration Statement shall become effective on such date as the Securities and Exchange Commission, acting pursuant to said Section 8(a), may determine.

The information in this proxy statement/prospectus is not complete and may be changed. A registration statement relating to the shares of Mandalay Digital Group, Inc. common stock to be issued in the merger has been filed with the Securities and Exchange Commission. These securities may not be sold nor may offers to buy be accepted prior to the time the registration statement becomes effective. This proxy statement/prospectus shall not constitute an offer to sell nor the solicitation of an offer to buy nor shall there be any sale of these securities in any jurisdiction in which such offer, solicitation or sale is not permitted or would be unlawful prior to registration or qualification under the securities laws of any such jurisdiction.

PRELIMINARY PROXY STATEMENT/PROSPECTUS SUBJECT TO COMPLETION, DATED JANUARY 16, 2015



To the Stockholders of Mandalay Digital Group, Inc. and the Stockholders of Appia, Inc.:

On behalf of the board of directors of Mandalay Digital Group, Inc., we are pleased to deliver to you this proxy statement/prospectus relating to the proposed issuance and sale by us of shares of our common stock in connection with the merger transaction described below.

The accompanying document is being delivered to (1) Mandalay Digital stockholders to solicit proxies to vote at our annual meeting, and (2) Appia, Inc. security holders as Mandalay Digital's prospectus for its offering of Mandalay Digital common stock in connection with the merger.

On November 13, 2014, we entered into an agreement and plan of merger, which we refer to as the merger agreement, to acquire all the outstanding capital stock of Appia, Inc., which we refer to as Appia, in exchange for the issuance of shares of our common stock to the Appia security holders. DTM Merger Sub, Inc., a wholly-owned subsidiary of Mandalay Digital, will merge with and into Appia, so that Appia as the surviving entity, will become a wholly-owned subsidiary of Mandalay Digital, which transaction we refer to as the merger. In the merger, each security, other than Appia stock options held by employees of Appia, held by an Appia security holder will be converted into the right to receive shares of Mandalay Digital common stock and outstanding Appia stock options held by employees of Appia will be assumed by Mandalay Digital and converted into Mandalay Digital stock options.

Consummation of the merger is subject to a number of conditions described in this proxy statement/prospectus, including the approval by our stockholders of the issuance and sale to the Appia security holders of up to 19,500,000 shares of our common stock, including shares issuable upon exercise of stock options that we will assume pursuant to the merger. For a description of the conditions for the merger, see the section entitled "Summary of the Merger Agreement – Conditions to the Consummation of the Merger" in this proxy statement/prospectus. We are also seeking approval of the issuance of 600,000 shares of our common stock (including a warrant to purchase 400,000 shares) to be issued to a lender in connection with a separate loan transaction. The warrant is not exercisable until the one year anniversary of the closing date of the merger and will terminate if the Company repays the debt prior to such one year anniversary.

Our stockholders are invited to attend the annual meeting of Mandalay Digital on [I], 2015. This letter is accompanied by the attached document, which the Mandalay Digital board of directors is providing to solicit from our stockholders proxies to vote for the issuance of Mandalay Digital common stock in connection with the merger. Our board of directors unanimously recommends that our stockholders vote "FOR" approval of the proposed issuance and sale by us of shares of our common stock in connection with the merger.

In addition to voting to authorize the issuance of Mandalay Digital common stock in the merger, at the annual meeting our stockholders also will be asked to elect directors, approve, on an advisory basis, the executive compensation of our executive officers and the frequency at which approval of executive compensation will be proposed and ratify the appointment of our auditors. Our board of directors also unanimously recommends that our stockholders vote "FOR" these proposals listed in the proxy statement/prospectus.

The date, time and place of the meeting are as follows: $[\mbox{\ensuremath{\square}}]$

Only Mandalay Digital stockholders of record as of [I], 2015 are entitled to attend and vote at the annual meeting.

If you are a holder of Mandalay Digital shares of common stock or preferred stock, whether or not you plan to attend the annual meeting, your vote is very important. Please sign and submit your proxy as soon as possible so that your shares can be voted at the annual meeting in accordance with your instructions. Whether or not you expect to attend the annual meeting, please submit your proxy promptly. Record holders of our common stock and preferred stock can submit a proxy via the Internet, by telephone, or by mailing the enclosed voting instructions). Instructions for using these convenient services appear on the instructions on the enclosed proxy card or voting instructions. Submitting a proxy now will not prevent you from being able to vote in person at the Mandalay Digital annual meeting.

This proxy statement/prospectus provides information about us, Appia and our proposed merger that our stockholders should know when they vote. In particular, the section entitled "Risk Factors" beginning on page 23 contains a description of risks that you should consider in evaluating the proposed merger. We urge you to read this entire proxy statement/prospectus carefully.

On behalf of our company, we look forward to seeing our stockholders at the annual meeting and we thank you for your support.

Sincerely, William Stone Chief Executive Officer

Neither the Securities and Exchange Commission nor any state securities commission has approved or disapproved of the merger, the issuance of the Mandalay Digital common stock in connection with the merger or the other transactions described in this document, or passed upon the adequacy or accuracy of the disclosure in this document. Any representation to the contrary is a criminal offense.

Proxy statement/prospectus dated [], 2015 and first mailed to our stockholders and Appia security holders on or about [], 2015

MANDALAY DIGITAL GROUP, INC.

2811 CAHUENGA BOULEVARD WEST LOS ANGELES, CALIFORNIA 90068

NOTICE OF ANNUAL MEETING OF STOCKHOLDERS TO BE HELD ON [I], 2015

NOTICE IS HEREBY GIVEN that the annual meeting of stockholders (the "annual meeting") of Mandalay Digital Group, Inc., a Delaware corporation ("Mandalay Digital" or the "Company"), will be held at [I] a.m. Pacific time, on [I], 2015, at the offices of Manatt, Phelps & Phillips, LLP, 11355 West Olympic Blvd., Los Angeles, California 90064. You are cordially invited to attend the annual meeting, at which stockholders will be asked to consider and vote upon the following proposals, which are more fully described in the enclosed proxy statement/prospectus:

- (1) To consider and vote upon the issuance and sale of (a) up to 19,500,000 shares of our common stock pursuant to the merger agreement with Appia, Inc. (including shares of common stock issuable upon exercise of stock options to be assumed by Mandalay Digital pursuant to the merger agreement) and, (b) an additional 600,000 shares of our common stock (including a warrant to purchase 400,000 shares) to be issued to a lender in connection with a separate loan transaction, all of which aggregates to more than 20% of Mandalay Digital's common stock outstanding under applicable listing rules of The NASDAQ Stock Market LLC.
- (2) To elect six directors to serve on our board of directors for a one-year term ending as of our annual meeting of stockholders in 2016.
- (3) To approve, in a non-binding advisory vote, the compensation of our named executive officers, commonly referred to as "Say-on-pay".
- (4) To hold an advisory, non-binding vote on the frequency of future advisory votes on the compensation of our named executive officers, commonly referred to as "Say-on-frequency".
- (5) To ratify the appointment of SingerLewak LLP as our independent registered public accounting firm for the fiscal year ending March 31, 2015.
- (6) Transact such other business as may properly come before the meeting and/or any adjournment or postponement thereof.

Only our stockholders of record of our common stock and our preferred stock at the close of business on [I], 2015, (the "Record Date"), are entitled to notice of and vote at the annual meeting and at any adjournments or postponements thereof. A list of such holders as of the Record Date will be available during normal business hours for examination by any such holder for a period of ten days prior to the day of the annual meeting at our principal executive offices.

Our board of directors has approved the issuance of our shares of common stock in connection with the merger with Appia and loan transaction and unanimously recommends that you vote FOR that proposal and FOR the other proposals. A copy of the merger agreement is attached as Appendix A to the accompanying proxy statement/prospectus. The proposals are described in more detail in the accompanying proxy statement/prospectus, which we encourage you to read in its entirety before voting.

All of our stockholders are urged to attend the meeting in person or by proxy. Your vote is important. Whether or not you expect to attend the meeting in person, please sign and submit your proxy as soon as possible so that your shares can be voted at the annual meeting in accordance with the instructions on the enclosed proxy card. The proxy is revocable and will not affect your right to vote in person in the event you attend the annual meeting. You may revoke your proxy at any time before it is voted. If you receive more than one proxy card because your shares are registered in different names or at different addresses, please sign and return each proxy card so that all of your shares will be represented at the annual meeting.

Thank you for your participation. We look forward to your continued support.

By Order of the Board of Directors,

William Stone
Chief Executive Officer

Los Angeles, California [0], 2015

Important Notice Regarding Availability of Proxy Materials for the Fiscal 2015 Annual Meeting of Stockholders to be Held on $[\mbox{\sc d}]$, 2015

 $Our\ Proxy\ Statement,\ Annual\ Report\ on\ Form\ 10-K,\ and\ form\ of\ proxy\ card\ are\ available\ on\ the\ Internet\ at\ www.proxyvote.com.$

ADDITIONAL INFORMATION

This proxy statement/prospectus incorporates by reference important business and financial information about us that is not included in or delivered with this proxy statement/prospectus. You may request this information, which includes copies of our annual, quarterly, and special reports, proxy statements, and other information, from us, without charge, excluding all exhibits, unless we have specifically incorporated by reference an exhibit in this proxy statement/prospectus. Our stockholders and Appia stockholders may obtain documents incorporated by reference in this proxy statement/prospectus by requesting them from us in writing or by telephone at the following address or telephone number:

Mandalay Digital Group, Inc. 2811 Cahuenga Boulevard West Los Angeles, California 90068 (323) 472-5461

To obtain timely delivery, any request for information should be made no later than [II], 2015.

In addition, we provide copies of our Forms 8-K, 10-K, 10-Q, Proxy Statement, and Annual Report at no charge to investors upon request and we make electronic copies of our most recently filed reports available through our website at http://ir.mandalaydigital.com/all-sec-filings as soon as reasonably practicable after filing such material with the SEC.

For a more detailed description of the information incorporated by reference into this proxy statement/prospectus and how you may obtain it, see the sections entitled "Where You Can Find More Information" and "Incorporation of Certain Information by Reference."

ABOUT THIS PROXY STATEMENT/PROSPECTUS

This document forms a part of a Registration Statement on Form S-4 filed by Mandalay Digital Group, Inc. with the Securities and Exchange Commission (the "SEC"). This document constitutes a prospectus of Mandalay Digital under Section 5 of the Securities Act of 1933, as amended (the "Securities Act"), with respect to the shares of Mandalay Digital common stock to be issued in connection with the proposed merger. In addition, this document constitutes a notice of meeting with respect to the Mandalay Digital annual meeting and a proxy statement under Section 14(a) of the Securities Exchange Act of 1934, as amended (the "Exchange Act").

Mandalay Digital has not authorized anyone to give any information or make any statement about the merger that is different from, or in addition to, that contained in this proxy statement/prospectus or in any of the materials that Mandalay Digital has incorporated by reference into this proxy statement/prospectus. Therefore, if anyone does give you information of this sort, you should not rely on it. The information contained in this proxy statement/prospectus speaks only as of the date of this proxy statement/prospectus unless the information specifically indicates that another date applies.

Holders of our common stock who have questions about the annual meeting or how to vote or revote their proxy please contact:

MACKENZIE PARTNERS, INC. 105 Madison Avenue New York, New York 10016 Call Toll-Free (800) 322-2885 Call Collect (212) 929-5500

Email: proxy@mackenziepartners.com

If you are in a jurisdiction where offers to exchange or sell, or solicitations of offers to exchange or purchase, the securities offered by this proxy statement/prospectus or the solicitation of proxies is unlawful, or if you are a person to whom it is unlawful to direct these types of activities, then the offer presented in this proxy statement/prospectus does not extend to you.

TABLE OF CONTENTS

QUESTIONS AND ANSWERS ABOUT THE ANNUAL MEETING AND THE MERGER	1
SUMMARY	11
Why You Are Receiving this Proxy Statement/Prospectus	11
Annual Meeting of Mandalay Digital Stockholders to be held on [.], 2015	11
Who can vote	12
Voting Matters	12
The Companies	13
The Merger	13
Reasons for the Proposed Merger	13
Recommendation of Mandalay Digital's Board of Directors	14
What Appia Stockholders will Receive in the Merger	14
Principal Agreements	14
Approval by Holders of Mandalay Digital Common and Preferred Stock	16
Opinion of Mandalay Digital's Financial Advisor	16
Risk Factors	16
Conditions to the Merger	16
Termination of the Merger Agreement	17
Termination Fees and Expenses	17
Interests of Certain Persons in the Merger	17
Material U.S. Federal Income Tax Consequences of the Merger	17
Anticipated Accounting Treatment of the Merger	18
Regulatory Matters	18
Directors and Management of Mandalay Digital Following Completion of the Merger	18
Appraisal Rights	18
Comparison of Stockholder Rights	18
SELECTED UNAUDITED PRO FORMA CONDENSED COMBINED FINANCIAL DATA	19
STOCK PRICE AND DIVIDEND INFORMATION	21
Mandalay Digital	21
<u>Appia</u>	22
RISK FACTORS	23
CAUTIONARY STATEMENT REGARDING FORWARD-LOOKING INFORMATION	40
ANNUAL MEETING OF MANDALAY DIGITAL STOCKHOLDERS	42
General General	42
Date, Time, and Place	42
Purpose of the Mandalay Digital Annual Meeting	42
Board Recommendation	43
Record Date, Outstanding Shares, and Quorum	43
How to Vote	43
Revocation of Proxies	44
Voting Your Shares	44
Votes Required	44
Withholding Your Vote, Abstentions and Broker non-votes	44
Proxy Solicitation	45
Other Business; Adjournments	45
SECURITY OWNERSHIP OF CERTAIN BENEFICIAL OWNERS AND MANAGEMENT	46
PROPOSAL 1	
APPROVAL OF THE ISSUANCE OF SHARES OF MANDALAY DIGITAL COMMON STOCK	49
Nasdaq Requirements	49
Impact of Issuance on Existing Stockholders	50
Stockholder Vote Requirement	50

<u>THE MERGER</u>	51
<u>Overview</u>	51
Background of the Merger	52
Reasons for the Merger and Recommendation of Mandalay Digital's Board of Directors	58
Opinion of Needham & Company, LLC	60
Conditions to Completion of the Merger	68
Expected Timing of the Merger	69
Consideration Payable in the Merger	69
Debt obligations and Guarantees	70
Material U.S. Federal Income Tax Consequences of the Merger	71
Regulatory Approvals Required For the Merger	71
Fees and Expenses of the Merger	71
Anticipated Accounting Treatment of the Merger	72
Appraisal Rights of Holders of Mandalay Digital Common Stock	72
Securities Law Restrictions on Shares of Mandalay Common Stock Received in the Merger	72
Lock-up Restrictions	72
Rights Pertaining to the Shares of Common Stock Received in the Merger	72
Dividends on Mandalay Digital Common Stock Received in the Merger	72
Interests of Certain Appia Directors, Officers, and Affiliates in the Merger	72
Approval of the Merger by Appia Stockholders	74
Directors and Management of Mandalay Digital Following the Merger	74
Indemnification	74
NASDAQ Listing and Trading	75
SEC Reports and Reporting Requirements	75
PRINCIPAL STOCKHOLDERS OF COMBINED COMPANY	75
MATERIAL U.S. FEDERAL INCOME TAX CONSEQUENCES	78
Tax Opinions	79
U.S. Federal Income Tax Consequences of the Merger	79
Certain Tax Reporting Rules	80
SUMMARY OF THE MERGER AGREEMENT	81
Cautionary Statement Concerning Representations and Warranties Contained in the Merger Agreement	81
General	81
Conversion of Securities in the Merger	81
Closing Consideration Schedule	85
Fractional Shares	85
Escrow of Consideration; Escrow Account	85
Working Capital/Net Debt Adjustment	86
Appia Designees to Mandalay Digital Board of Directors	86
Stockholder Representative	86
Stockholder Representative Expense Fund	86
Closing and Effectiveness of the merger	87
Exchange of Appia Stock Certificates for the Merger Consideration	87
Transfer Restrictions on Mandalay Digital Common Stock	87
Appraisal Rights	87
Representations and Warranties	88
Survival; Indemnification	88
Covenants of Mandalay Digital and Appia	90
Conditions to the Consummation of the merger	93
Definition of "Material Adverse Effect"	95
Termination of the Merger Agreement	96
Amendment	97

ANCILLARY AGREEMENTS	98
Support Agreement	98
Financing Agreements	99
UNAUDITED PRO FORMA CONDENSED COMBINED FINANCIAL INFORMATION	101
NOTES TO UNAUDITED PRO FORMA CONDENSED COMBINED FINANCIAL STATEMENTS	106
BUSINESS OF MANDALAY DIGITAL	110
BUSINESS OF APPIA, INC.	110
<u>Overview</u>	110
Appia's Products	110
<u>Customers</u>	112
<u>Partners</u>	112
Sales and Marketing	112
Employees and Facilities	112
MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS OF	
THE BUSINESS OF APPIA	113
Components of Revenue and Expenses	113
Results of Operations	114
Year Ended December 31, 2013 Compared to Year Ended December 31, 2012	115
<u>Seasonality</u>	116
Financial Condition	116
<u>Assets</u>	116
Liabilities and Working Capital	117
Liquidity and Capital Resources	117
<u>Indebtedness</u>	118
Contractual Obligations	119
Summary of Critical Accounting Policies and Estimates	119
Off-Balance Sheet Arrangements	121
DESCRIPTION OF MANDALAY DIGITAL COMMON STOCK	122
Authorized Capitalization	122
Common Stock	122
Preferred Stock	122
Anti-Takeover Effects of Certain Provisions of Delaware Law	123
COMPARISON OF RIGHTS OF STOCKHOLDERS OF MANDALAY DIGITAL AND APPIA	124
PROPOSAL 2	
ELECTION OF DIRECTORS	132
Information Concerning Current Directors and Nominees	132
Required Vote	134
Board Recommendation	134
BOARD MEETINGS AND COMMITTEES	135
Committees of the Board of Directors	135
Audit Committee	135
<u>Compensation Committee</u>	135
Governance and Nominating Committee	135
<u>CORPORATE GOVERNANCE</u>	136
Board Leadership Structure	136
Role of Board of Directors in Risk Oversight	136
Code of Business and Ethical Conduct	136
<u>Director Independence</u>	137
Related Party Transactions	137
Stockholder Nominations and Bylaw Procedures	138
Executive Sessions of Independent Directors	138
Stockholder Communication with Board Members	138
Board Member Attendance at Annual Meetings	138
EXECUTIVE OFFICERS	139

EXECUTIVE COMPENSATION	140
Summary Compensation Table	140
NARRATIVE DISCLOSURE TO SUMMARY COMPENSATION TABLE	141
OUTSTANDING EQUITY AWARDS AT MARCH 31, 2014	144
DIRECTOR COMPENSATION	145
NARRATIVE TO DIRECTOR COMPENSATION TABLE	145
REPORT OF AUDIT COMMITTEE	146
PROPOSAL 3	
ADVISORY (NON-BINDING) VOTE ON EXECUTIVE COMPENSATION	148
Board Recommendation	148
PROPOSAL 4	
ADVISORY (NON-BINDING) VOTE ON THE FREQUENCY OF THE ADVISORY VOTE ON EXECUTIVE	
COMPENSATION	149
Board Recommendation	149
PROPOSAL 5	
RATIFICATION OF APPOINTMENT OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM FOR FISCAL	
YEAR 2015	150
Board Recommendation	151
OTHER MATTERS	152
HOUSEHOLDING OF PROXY MATERIALS	152
COMPLIANCE WITH SECTION 16(a) OF SECURITIES EXCHANGE ACT	152
STOCKHOLDER PROPOSALS	152
ACCOMPANYING INFORMATION	153
EXPERTS	153
LEGAL MATTERS	153
WHERE YOU CAN FIND MORE INFORMATION	153
INCORPORATION OF CERTAIN INFORMATION BY REFERENCE	154
INDEX TO FINANCIALS	155
Appendix A Agreement and Plan of Merger	
Appendix B Form of Support Agreement	
Appendix C Opinion of Needham & Company, LLC	

QUESTIONS AND ANSWERS ABOUT THE ANNUAL MEETING AND THE MERGER

The following are answers to certain questions that you may have regarding the merger and the annual meeting. We urge you to read carefully the remainder of this document because the information in this section may not provide all the information that might be important to you in determining how to vote. Additional important information is also contained in the appendices to, and the documents incorporated by reference in, this document.

GENERAL

Q: WHY AM I RECEIVING THIS PROXY STATEMENT/PROSPECTUS?

A: This document serves as both a proxy statement and prospectus of Mandalay Digital. As a proxy statement, it is being provided to Mandalay Digital's stockholders in connection with the solicitation of proxies for the Mandalay Digital annual meeting at which Mandalay Digital stockholders will be asked to vote on the proposals set forth in the Notice of Annual Meeting of Stockholders.

As a prospectus, this document is being provided to Appia's security holders because Mandalay Digital is offering to issue shares of its common stock for shares of Appia securities upon completion of the merger.

THE ANNUAL MEETING

O: WHEN AND WHERE WILL THE ANNUAL MEETING BE HELD?

A: The annual meeting will be held at [I], Pacific time, on [I], 2015, at the offices of Manatt, Phelps & Phillips, LLP, 11355 West Olympic Blvd., Los Angeles, California 90064.

Q: WHAT BUSINESS WILL OCCUR AT THE MEETING?

- **A:** Mandalay Digital stockholders are being asked to vote on the following proposals:
 - to consider and vote upon the issuance and sale of (a) up to 19,500,000 shares of our common stock pursuant to the merger agreement with Appia, Inc. (including shares of common stock issuable upon exercise of stock options to be assumed by Mandalay Digital pursuant to the merger agreement and 1,000,000 shares to be deposited into escrow) and, (b) an additional 600,000 shares of our common stock (including a warrant to purchase 400,000 shares, which vests 12 months after issuance under certain circumstances) to be issued to a lender in connection with a separate loan transaction (*Proposal 1*);
 - to elect six directors to serve on our board of directors for a one-year term ending as of our annual meeting of stockholders in 2016 (*Proposal* 2);
 - to approve, in a non-binding advisory vote, the compensation of our named executive officers, commonly referred to as "Sayon-pay" (*Proposal 3*);
 - to hold an advisory, non-binding vote on the frequency of future advisory votes on the compensation of our named executive officers, commonly referred to as "Say-on-frequency" (*Proposal 4*);
 - to ratify the appointment of SingerLewak LLP as our independent registered public accounting firm for the fiscal year ending March 31, 2015 (*Proposal 5*); and

• to transact such other business as may properly come before the meeting and/or any adjournment or postponement thereof.

Mandalay Digital will transact no other business at the annual meeting except for business properly brought before the annual meeting or any adjournment or postponement thereof.

Q: WHAT DOES THE MANDALAY DIGITAL BOARD OF DIRECTORS RECOMMEND?

A: The Mandalay Digital board of directors has approved the merger and the authorization of the shares of common stock issuable pursuant to the merger agreement and in connection with the merger and believes that such action is in the best interests of Mandalay Digital and its stockholders. Accordingly, the board of directors unanimously recommends that the Mandalay Digital stockholders vote **FOR** approval of the authorization of the issuance of the shares of Mandalay Digital common stock in connection with the merger. The board also recommends that the Mandalay Digital stockholders vote **FOR** approval of the other proposals to be considered at the annual meeting, as described in this proxy statement/ prospectus.

Q: WHAT VOTE IS REQUIRED TO APPROVE EACH PROPOSAL AT THE ANNUAL MEETING?

Assuming a quorum is present, the proposal to issue shares of common stock in connection with the merger and loan transaction requires the affirmative vote of not less than a majority of the votes cast by the holders of our common stock and our preferred stock, which we refer to as our preferred stock, voting together with the common stock as a single class on an as-converted to common stock basis, either in person or represented by proxy at the annual meeting and entitled to vote thereon (*Proposal 1*).

The six nominees for director who receive the most votes from common and preferred stockholders, voting together as a single class, will be elected (*Proposal 2*).

Assuming a quorum is present, the affirmative vote of a majority of the shares of common stock and preferred stock, voting together as a single class on an as-converted to common stock basis, either in person or represented by proxy at the annual meeting and entitled to vote on the proposal is required to approve the advisory Say-on-pay proposal (*Proposal 3*), the annual (one-year) alternative in the Say-on-frequency proposal (*Proposal 4*), and the ratification of SingerLewak LLP as our independent registered public accounting firm for the fiscal year ending March 31, 2015 (*Proposal 5*).

Q: WHY ARE HOLDERS OF MANDALAY DIGITAL'S COMMON STOCK BEING ASKED TO APPROVE THE ISSUANCE OF SHARES OF ITS COMMON STOCK?

A: Pursuant to the terms of the merger agreement, upon consummation of the proposed merger, we will issue up to 19,500,000 shares of our common stock to Appia security holders pursuant to the merger agreement, including shares of common stock issuable upon exercise of stock options that we will assume pursuant to the merger agreement and 1,000,000 shares to be deposited into escrow. See "The Merger – Consideration Payable in the Merger" located on page 69 of this proxy statement/prospectus for further information and a discussion of the assumptions in calculating the potential amount of shares to be issued by Mandalay Digital in the merger. We will also issue to a lender 200,000 shares of our common stock and a common stock purchase warrant exercisable for 400,000 shares of common stock in a separate loan transaction. See "Ancillary Agreements – The Financing Agreements" located on page 98 of this proxy statement/prospectus.

The rules of the NASDAQ Stock Market, where our shares of common stock are listed for trading, require the approval of our stockholders prior to the issuance of additional shares of our common stock in connection with the acquisition of the stock or assets of another company if:

- the common stock issued by us will have voting power equal to or in excess of 20% of the voting power outstanding before the issuance of such stock; or
- the number of shares of common stock issued by us will be equal to or in excess of 20% of the number of shares of common stock outstanding before the issuance of such stock.

Therefore, your approval is required for the issuance of shares of our common stock (including a warrant to purchase shares of our common stock) in connection with the merger and a separate loan transaction. Our stockholders are not being asked to vote on the proposed merger itself, but instead solely on whether we may issue the shares of common stock (including a warrant to purchase shares of our common stock) in connection with the merger and a separate loan transaction. If the requisite number of our shares are not voted in favor of the issuance, we will be unable to effect the merger under the terms set forth in the merger agreement.

Q: WHO IS ENTITLED TO VOTE?

A: Holders of record of Mandalay Digital common stock and preferred stock at the close of business on [I], 2015, which is the date that the board of directors has fixed as the Record Date for the annual meeting, are entitled to notice of, and to vote at, the Mandalay Digital annual meeting and any adjournment or postponement thereof. The preferred stock votes together with the common stock as a single class and on an as-converted basis. As of the Record Date, there were outstanding [I] shares of our common stock and 100,000 shares of our preferred stock, which is currently convertible into 20,000 shares of common stock.

Q: WHAT CONSTITUTES A QUORUM?

A: The presence at the Mandalay Digital annual meeting, in person or by proxy, of holders of a majority of the outstanding shares of Mandalay Digital common stock and preferred stock (which votes together with the common stock as a single class and on an asconverted basis) entitled to vote at the annual meeting will constitute a quorum for the transaction of business. Abstentions and broker non-votes, if any, will be included in determining the number of shares present at the meeting for the purpose of determining the presence of a quorum.

Q: HOW MANY VOTES MAY A HOLDER OF COMMON STOCK OR PREFERRED STOCK CAST?

A: Holders of our common stock are entitled to one vote at the annual meeting for each share of common stock held that was issued and outstanding as of the Record Date. The 100,000 shares of our preferred stock outstanding are currently convertible into 20,000 shares of common stock and are entitled to vote together with the common stock as a single class (on an as-converted to common stock basis) on any matters submitted to the holders of the Company's common stock, together with any other voting rights provided to the preferred stock under law or the General Corporation Law of the State of Delaware.

Q: ARE THERE DIFFERENT VOTING PROCEDURES DEPENDING ON HOW I HOLD MY MANDALAY DIGITAL STOCK?

A: Yes. Many holders of our stock hold their shares through a stockbroker, bank, or other nominee rather than directly in their own name. As summarized below, there are some distinctions between shares of our stock held of record and those owned beneficially.

Holder of Record: If your shares of our stock are registered directly in your name with our transfer agent, American Stock Transfer & Trust Company, you are considered, with respect to those shares of

our stock, the holder of record, and these proxy materials are being sent directly to you by us. As the holder of record, you have the right to grant your voting proxy directly to our company or to vote in person at the annual meeting. We have enclosed a proxy card for you to use.

Beneficial Owner: If your shares of our stock are held in a stock brokerage account or by a bank or other nominee, you are considered the beneficial owner of securities held in street name, and these proxy materials are being forwarded to you by your broker or nominee who is considered, with respect to those securities, the holder of record. As the beneficial owner, you have the right to direct your broker or nominee on how to vote and you are also invited to attend the annual meeting. Your broker or nominee has enclosed a voting instruction card for you to use in directing the broker or nominee regarding how to vote your securities. The voting instruction card provides various alternative voting methods, such as via the Internet, by telephone, or by mail.

Q HOW CAN I VOTE MY STOCK IN PERSON AT THE ANNUAL MEETING?

A: Shares of our common stock or preferred stock (which votes on an as-converted basis) held directly in your name as the holder of record may be voted in person at the annual meeting. If you choose to do so, please bring the enclosed proxy card and proof of identification. Even if you plan to attend the annual meeting, we recommend that you also submit your proxy as described below so that your vote will be counted if you later decide not to attend the annual meeting. Securities held in street name may be voted in person by you only if you obtain a signed proxy from the record holder giving you the right to vote the securities in person.

Q: HOW CAN I VOTE MY MANDALAY DIGITAL STOCK WITHOUT ATTENDING THE ANNUAL MEETING?

A: Whether you hold securities directly as the holder of record or beneficially in street name, you may direct your vote without attending the annual meeting. You may vote your directly held securities by granting a proxy or, for securities held in street name, by submitting voting instructions to your broker, bank, or nominee following the instructions on the form included with this proxy statement/prospectus by the deadline indicated on that form.

Q: HOW DO I VOTE?

A: If you are the stockholder of record (that is, the shares are held in your name), you may submit your proxy in one of three convenient ways:

By the Internet: Go to www.voteproxy.com and follow the instructions. You will need the control number that appears on your proxy card included in this proxy statement/prospectus. This method of submitting your proxy will be available until $[\[D\]]$ p.m., Pacific time, on $[\[D\]]$, 2015.

By telephone: On a touch-tone telephone, call toll-free $[\]$ and follow the instructions. You will need the control number that appears in the box on the front of your proxy card included in this proxy statement/prospectus. This method of submitting your proxy will be available until $[\]$ p.m., Pacific time, on $[\]$, 2015.

By mail: If you submit a by traditional proxy card, mark your selections on the proxy card, date the card, and sign your name exactly as it appears on the card, then mail it in the postage-paid envelope enclosed with the materials. You should mail the proxy card in plenty of time to allow delivery prior to the meeting.

If you are a stockholder of record and attend the meeting, you may deliver your completed proxy card in person. If you are not the stockholder of record (that is, your shares are held in the name of a bank, broker, or other holder of record, which is often referred to as held in "street name") then you will

receive instructions from the holder of record that you must follow to ensure that your shares are voted as you wish. You will not be able to vote those shares at the meeting unless you have received, in advance, a proxy card from the record holder (that is, the bank, broker, or other holder of record).

If you complete and properly sign and timely return a proxy card to us or complete your proxy by telephone or online, your shares will be voted as you direct.

Q. WHAT WILL HAPPEN IF I SIGN AND RETURN MY PROXY CARD WITHOUT INDICATING HOW I WISH TO VOTE?

A. Signed and dated proxies received by Mandalay Digital without an indication of how the stockholder intends to vote on a proposal will be voted in favor of each proposal presented to the stockholders.

O: MAY I CHANGE MY VOTE AFTER I HAVE GIVEN IT?

A: You may change your proxy instructions and your vote at any time prior to the vote at the annual meeting. For shares held directly in your name, you may accomplish this by granting a new proxy bearing a later date, which automatically revokes the earlier proxy, and delivering such new proxy to our Secretary either by mail or by calling the phone number, or accessing the Internet address, listed on the proxy card or by attending the annual meeting and voting in person. Attendance at the annual meeting will not cause your previously granted proxy to be revoked unless you specifically request to do so. For securities held beneficially by you, you may accomplish this by submitting new voting instructions to your broker, bank, or nominee by the deadline indicated in the instructions sent to you by your broker, bank, or nominee.

Q: WHO BEARS THE COST OF SOLICITING PROXIES?

A: We will pay the entire cost of preparing, assembling, printing, mailing, and distributing these proxy materials. In addition to the mailing of these proxy materials, the solicitation of proxies or votes may be made in person, by telephone, or by electronic communication by our directors, officers, and employees, who will not receive any additional compensation for such solicitation activities. We have retained the services of MacKenzie Partners to aid in the solicitation of proxies from banks, brokers, nominees, and intermediaries. We will pay MacKenzie Partners a fee of \$15,000 for its services, plus we will reimburse MacKenzie for various out-of-pocket and other expenses. We may also, upon request, reimburse brokerage firms and other persons representing beneficial owners of our stock for their expenses in forwarding solicitation materials to such beneficial owners.

Q: WHAT DO I NEED TO DO NOW?

A: After you have carefully read this proxy statement/prospectus, you should indicate on your proxy card how you want your shares to be voted, then sign, date and mail the proxy card in the enclosed postage-paid envelope as soon as possible so that your shares may be represented and voted at the annual meeting. You should also indicate on your proxy card how you want your shares voted on the other proposals to be considered at the annual meeting.

Q: WHERE CAN I FIND THE VOTING RESULTS OF THE ANNUAL MEETING?

A: We may be able to announce preliminary voting results at the annual meeting and we intend to issue a press release with the final results after the annual meeting is completed. In addition, we intend to publish the final voting results in a report on Form 8-K that we will file with the SEC.

Q: WHAT WILL HAPPEN IF THE PROPOSAL TO ISSUE THE SHARES IN CONNECTION WITH THE MERGER AND LOAN TRANSACTION IS NOT APPROVED?

A: If the proposal to issue up to 19,500,000 shares and the additional 600,000 shares of our common stock (including a warrant to purchase 400,000 shares, which vests 12 months after issuance under certain circumstances) in connection with the merger is not approved, the merger and loan transaction cannot proceed.

Q. ARE PROXY MATERIALS AVAILABLE ON THE INTERNET?

A. Yes. Mandalay Digital has posted materials related to the annual meeting at www.proxyvote.com.

Q: WHO SHOULD I CONTACT IF I HAVE QUESTIONS ABOUT THE ANNUAL MEETING?

A: If you have questions about the annual meeting, please contact MacKenzie Partners by telephone at (800) 322-2885 (toll-free) or via email at proxy@mackenziepartners.com.

THE MERGER

O: WHAT IS THE MERGER?

A: On November 13, 2014, Mandalay Digital entered into a merger agreement with DTM Merger Sub, Inc., our newly-formed wholly-owned subsidiary, which we refer to as Merger Sub, Appia, and Shareholder Representative Services LLC, as the stockholder representative, pursuant to which Merger Sub will merge with and into Appia on the terms and subject to the conditions set forth in the merger agreement, which transaction we refer to as the merger, so that Appia, as the surviving corporation, will become a wholly-owned subsidiary of Mandalay Digital. The surviving corporation will be renamed "Digital Turbine Media, Inc." Furthermore, Mandalay Digital will change its name to "Digital Turbine, Inc." effective on January 20, 2015.

Q: WHY ARE MANDALAY DIGITAL AND APPIA PROPOSING TO MERGE?

A: The boards of directors of Mandalay Digital and Appia are proposing to merge because they believe that the combination strengthens and is in the best interests of both companies and their respective stockholders. Please see "The Merger - Reasons for the Merger and Recommendation of Mandalay Digital's Board of Directors" at page 58 for the various factors considered by the Mandalay Digital board of directors in recommending that Mandalay Digital stockholders vote **FOR** the proposal to issue the shares to the Appia security holders.

Q. WHAT WILL HAPPEN IN THE MERGER?

A. At the closing of the merger, Mandalay Digital's wholly-owned subsidiary will merge with and into Appia, so that Appia, as the surviving corporation, will become a wholly-owned subsidiary of Mandalay Digital. In exchange for their shares of Appia's common stock, preferred stock, and/or options, the Appia stockholders will receive shares of Mandalay Digital common stock and/or options of Mandalay Digital common stock. As a result of the merger, Mandalay Digital will own all of the issued and outstanding securities of Appia and former Appia stockholders will not have any equity ownership interest in the surviving corporation.

Q: WHAT WILL APPIA SECURITY HOLDERS RECEIVE IN THE MERGER?

A: At the effective time of the merger, which we refer to as the effective time, on the terms and subject to the conditions set forth in the merger agreement, all of the equity securities of Appia outstanding immediately prior to the effective time (including outstanding stock options, which will be assumed by Mandalay Digital), other than equity securities held by stockholders that exercise dissenters' rights

under applicable law, will be converted into the right to receive a number of shares (or, in the case of stock options being assumed, stock options) of Mandalay Digital common stock, equal to \$100 million less Appia's net debt and transaction expenses at the time of closing (subject to a working capital adjustment), divided by \$4.50.

Mandalay Digital expects to issue, pursuant to the merger agreement, up to 19,500,000 shares of common stock (including shares underlying assumed stock options). Accordingly, we are asking our stockholders to consider and vote upon the issuance and sale of up to 19,500,000 shares of our common stock in connection with the proposed merger with Appia, Inc. (including shares of common stock issuable upon exercise of stock options to be assumed by Mandalay Digital pursuant to the merger agreement and 1,000,000 shares to be deposited into escrow).

The aggregate number shares of Mandalay Digital common stock to be issued pursuant to the merger agreement will vary depending on a number of factors, including the amount of the estimated net debt at closing. The merger agreement does not provide for an adjustment in the aggregate number of shares of Mandalay Digital common stock or options to be issued by Mandalay Digital to Appia's security holders in the acquisition in the event of fluctuations in the market value of Mandalay Digital's common stock up through the closing date. However, the number of shares to be issued by Mandalay Digital pursuant to the merger agreement is subject to adjustment based on Appia's working capital and net debt as of the closing date of the merger. We currently estimate that, if the merger closes on January 31, 2015, we would issue 19,056,392 shares; additionally, we would assume options for 277,376 shares of our common stock, for an aggregate of 19,333,768 shares, which would represent approximately 33% of the Company's outstanding shares of common stock as of November 13, 2014 (after giving effect to such issuance). Appia's actual results of operations between the date of the financial information included in this proxy statement/prospectus and the closing date may differ from the forecasted results used in this estimate, resulting in differences between the forecast and actual amounts of Appia's working capital and net debt, including transaction expenses, as of the closing date of the merger. See "The Merger – Consideration Payable in the Merger" located on page 69 of this proxy statement/prospectus for further information and a discussion of the assumptions in calculating the potential amount of shares to be issued by Mandalay Digital in the merger.

Q: WHAT IS THE ESCROW?

A: The Company will deposit one million of the shares to be issued to Appia's security holders in connection with the merger into an escrow account to cover indemnification claims by the Company pursuant to the terms of the merger agreement, which shares will be released one year after the closing, subject to any then pending claims.

Q: ARE THE SHARES OF COMMON STOCK TO BE RECEIVED BY THE APPIA SECURITY HOLDERS FREELY TRADEABLE UPON ISSUANCE?

A: Yes, the issuance of the shares of Mandalay Digital common stock to the security holders of Appia pursuant to the terms of the merger agreement will be registered under the Securities Act. However, as to each recipient of Mandalay Digital shares that is considered an "affiliate" (as such term is defined in the Securities Act) of Mandalay Digital, such affiliates will be permitted to sell the shares of our common stock received in the merger only pursuant to an effective registration statement or an exemption from the registration requirements of the Securities Act. This proxy statement/prospectus does not register the resale of shares of our stock held by affiliates. Furthermore, all of the Appia stockholders that executed the support agreements (as described in "Ancillary Agreements – Support Agreements") agreed that, and pursuant to the terms of the merger agreement, the shares of Mandalay Digital common stock to be received by Appia stockholders in the merger are all subject to a lock up restrictions, which restrictions will lapse with respect to one-third of such shares of Mandalay Digital common stock at each of the six, nine and 12 month anniversaries of the closing date of the merger.

Q: WHAT CONDITIONS MUST BE SATISFIED TO COMPLETE THE MERGER?

A: There are a number of closing conditions in the merger agreement, including that Mandalay Digital's stockholders have approved the issuance by Mandalay Digital of shares of common stock pursuant to the merger agreement. In addition, Appia has not yet taken all steps required by Delaware law and its organizational documents to obtain consents from the holders of its voting stock for the approval of the consummation of the merger. For a summary of the conditions that must be satisfied or waived prior to completion of the merger, see the section entitled "Summary of the Merger Agreement" beginning on page 80.

Q: WHAT IS THE SUPPORT AGREEMENT?

A: In connection with the execution of the merger agreement, certain Appia stockholders, who we refer to as the supporting stockholders, entered into a support agreement with Mandalay Digital, which we refer to as the support agreement, pursuant to which they have agreed, among other things, to vote (or deliver written consents with respect to) all of their shares of Appia in favor of the approval of the merger pursuant to the merger agreement and the other transactions contemplated thereby, to vote against any action that would reasonably likely result in a material breach of Appia under the merger agreement, and to vote against any alternative transactions. The support agreement also generally prohibits the supporting stockholders from transferring their shares prior to the consummation of the merger. The support agreement will terminate upon the termination of the merger agreement in accordance with its terms. The support agreement provides for the vote or written consent of a sufficient number of shares of Appia to adopt the merger agreement and approve and the transactions contemplated thereby under Delaware law and Appia's organizational documents. For a further description of the support agreement, see the section entitled "Ancillary Agreements - The Support Agreement" beginning on page 98.

Q: WILL MANDALAY DIGITAL INCUR DEBT OBLIGATIONS AS A RESULT OF THE MERGER?

A: Upon consummation of the merger, Mandalay Digital will guarantee Appia's existing indebtedness outstanding as of closing, which amount is expected to be approximately \$9.5 million (excluding the transaction fees and expenses) assuming the merger closes on January 31, 2015. See "The Merger – Consideration Payable in the Merger" located on page 68 of this proxy statement/prospectus for further information and assumptions about Appia's forecasted indebtedness based on certain closing dates. Upon consummation, and as a condition to the closing, of the merger, Appia will enter into an amended loan agreement with its current senior lender, Silicon Valley Bank (the "SVB Debt"), and a securities purchase agreement with Mandalay Digital and North Atlantic SBIC IV, L.P. ("North Atlantic") pursuant to which Appia will issue to North Atlantic a subordinated debenture in the aggregate principal amount of \$8.0 million, which amount is included in the \$9.5 million referenced above ("North Atlantic Debt"), to retire a like amount of outstanding debt to North Atlantic.

Upon consummation of the merger, Mandalay Digital will issue a secured guaranty to each of the SVB Debt and North Atlantic Debt whereby we will guarantee all of Appia's obligations in connection with such indebtedness and pledge substantially all of our assets, including our intellectual property. The secured guarantees will contain cross-default provisions and covenants, among others, limiting our ability to undergo a change of control, incur indebtedness, grant liens, make dividends in cash and other customary covenants. Mandalay will also issue to North Atlantic 200,000 shares of its common stock and a common stock warrant to purchase 400,000 shares of its common stock, all of which will have registration rights requiring Mandalay Digital to file a registration statement 15 business days after the closing of the merger with Appia. The warrant will have an exercise price of \$0.01 per share exercisable for 10 years, but it will not be exercisable until the one year anniversary of the closing date of the merger and will terminate if the Company repays the North Atlantic Debt prior to such one year

anniversary. We intend to refinance the Appia debt as soon as practicable following the closing of the merger. See "Ancillary Agreements – The Financing Agreements" on page 98 for further information.

Q: WHEN WILL THE MERGER BE COMPLETED?

A: Mandalay Digital and Appia are working to complete the merger as soon as practicable. If the stockholders of Appia approve the merger and the stockholders of Mandalay Digital approve the issuance of shares of common stock in connection with the merger, the parties currently expect that the merger will be completed in the first calendar quarter of 2015. Mandalay Digital cannot predict, however, the actual date on which the merger will be completed because it is subject to factors beyond its control, including whether or when the required regulatory approvals will be received. For further information, please see the section entitled "Summary of the Merger Agreement—Conditions to the Consummation of the Merger" beginning on page 93.

Q: WILL MANDALAY DIGITAL'S BOARD OF DIRECTORS CHANGE AFTER THE MERGER?

A: Upon the closing of the merger, we will appoint Jud Bowman, founder and Chief Executive Officer of Appia, and Craig Forman, Appia's Executive Chairman, to become members of our board of directors.

Q: ARE THERE RISKS ASSOCIATED WITH THE MERGER OF WHICH HOLDERS OF MANDALAY DIGITAL COMMON STOCK SHOULD BE AWARE?

A: Yes. There are risks to us and Appia in connection with the merger, including that the combined company may not achieve the expected financial and operating benefits because of, among other reasons, the risks and uncertainties discussed in the section entitled "Risk Factors." In deciding whether to approve the issuance of shares of our common stock, we urge you to carefully read and consider the risk factors contained in the section entitled "Risk Factors."

Q: DO MANDALAY DIGITAL'S STOCKHOLDERS HAVE APPRAISAL RIGHTS IN CONNECTION WITH THE MERGER?

A: Under Delaware law and our Certificate of Incorporation, holders of our common stock are not entitled to any rights to seek appraisal of their securities or to exercise any preemptive rights in connection with the proposal to issue up to 19,500,000 shares of our common stock pursuant to the merger agreement.

Q: WHAT ARE THE MATERIAL U.S. FEDERAL INCOME TAX CONSEQUENCES OF THE MERGER TO APPIA STOCKHOLDERS?

The merger is intended to qualify, as a reorganization within the meaning of Section 368(a) of the Internal Revenue Code of 1986, as amended, which we refer to as the Code. Assuming the merger qualifies as a reorganization, Appia stockholders generally will not recognize gain or loss for U.S. federal income tax purposes upon the exchange of their shares of Appia stock for shares of Mandalay Digital common stock in connection with the merger. For a more detailed discussion of the material U.S. federal income tax consequences of the transaction, please see the section entitled "Material U.S. Federal Income Tax Consequences" beginning on page 78.

Tax matters are very complicated, and the tax consequences of the merger to any particular stockholder will depend on that stockholder's particular facts and circumstances. Accordingly, you are urged to consult your tax advisor for a full understanding of the tax consequences from the merger to you, including the applicability and effect of federal, state, local and foreign income and other tax laws.

Q: WHAT HAPPENS IF THE MERGER IS NOT COMPLETED?

A: If the merger is not completed, Appia stockholders will not receive any consideration for their shares of Appia capital stock in connection with the merger. Instead, Appia will remain an independent company. Under specified circumstances, Mandalay Digital may be required to pay Appia a fee of \$1.0 million and reimburse Appia for expenses of up to \$500,000 (unless Appia elects to instead pursue other remedies in the case of a termination by Appia based upon an intentional and willful material breach by Mandalay Digital) with respect to the termination of the merger agreement, as described under the section entitled "Summary of the Merger Agreement—Termination of the Merger Agreement; Fees and Expenses" beginning on page 96.

SUMMARY

This summary highlights selected information from this proxy statement/prospectus. Because this is a summary, it does not contain all of the information that may be important to you. You should carefully read this entire document and the other documents we refer to in this document before you decide how to vote. These references will give you a more complete description of the merger agreement and the merger and the other matters to be considered at the annual meeting. We have included page references in this summary to direct you to more complete descriptions of the topics provided elsewhere in this proxy statement/ prospectus.

In addition, we incorporate by reference important business and financial information about Mandalay Digital into this document. For a description of this information, please see the section entitled "Incorporation of Certain Documents by Reference" beginning on page 154. You may obtain the information incorporated by reference into this document without charge by following the instructions in the section entitled "Where You Can Find More Information" on page 153. References to "Mandalay Digital," "the Company," "we," "our" or "us" in this proxy statement/prospectus refer to Mandalay Digital Group, Inc. and its subsidiaries, unless otherwise indicated by the context.

Why You Are Receiving this Proxy Statement/Prospectus

This proxy statement/prospectus is:

- a proxy statement of Mandalay Digital for use in the solicitation of proxies from Mandalay Digital stockholders for its annual meeting; and
- a prospectus of Mandalay Digital relating to the issuance of shares of Mandalay Digital's common stock to the current Appia security holders in exchange for securities of Appia.

Annual Meeting of Mandalay Digital Stockholders to be held on [I], 2015 (Page 42)

The annual meeting of stockholders of Mandalay Digital Group, Inc. will be held at $[\ \ \]$ a.m. Pacific time, on $[\ \ \ \]$, 2015, at the offices of Manatt, Phelps & Phillips, LLP, 11355 West Olympic Blvd., Los Angeles, California 90064 for the following purposes:

- (1) To consider and vote upon the issuance and sale of (a) up to 19,500,000 shares of our common stock pursuant to the merger agreement with Appia, Inc. (including shares of common stock issuable upon exercise of stock options to be assumed by Mandalay Digital pursuant to the merger agreement and 1,000,000 shares to be deposited into escrow) and, (b) an additional 600,000 shares of our common stock (including a warrant to purchase 400,000 shares and vesting 12 months after issuance under certain circumstances) to be issued to a lender in connection with a separate loan transaction, all of which aggregates more than 20% of Mandalay Digital's common stock outstanding under applicable listing rules of The NASDAQ Stock Market LLC.
- (2) To elect six directors to serve on our board of directors for a one-year term ending as of our annual meeting of stockholders in 2016.
- (3) To approve, in a non-binding advisory vote, the compensation of our named executive officers, commonly referred to as "Say-on-pay".
- (4) To hold an advisory, non-binding vote on the frequency of future advisory votes on the compensation of our named executive officers, commonly referred to as "Say-on-frequency".
- (5) To ratify the appointment of SingerLewak LLP as our independent registered public accounting firm for the fiscal year ending March 31, 2015.

(6) Transact such other business as may properly come before the meeting and/or any adjournment or postponement thereof.

Who can vote (Page 43)

You are entitled to vote at the Mandalay Digital annual meeting if you owned shares of common stock or preferred stock at the close of business on the Record Date. Holders of our common stock are entitled to one vote at the annual meeting for each share of common stock held that was issued and outstanding as of the Record Date. The preferred stock is entitled to vote together with the common stock as a single class (on an as if converted basis) on any matters submitted to the holders of the Company's common stock, together with any other voting rights provided to the preferred stock under law or the General Corporation Law of the State of Delaware. On the Record Date, there were [I] shares of common stock and 100,000 shares of preferred stock, which are convertible into 20,000 shares of common stock, outstanding.

You may vote either by attending the annual meeting and voting your shares, or by completing the enclosed proxy card or voting instruction card and mailing it to Mandalay Digital or you may also vote by telephone or via the Internet, as described in this proxy statement/prospectus.

The board of directors of Mandalay Digital is seeking your proxy to use at the annual meeting. We have prepared this proxy statement/prospectus to assist you in deciding how to vote and whether or not to grant your proxy. Please indicate on your proxy card or voting instruction card how you want to vote. Then sign, date and mail the proxy card or voting instruction card as soon as possible so that your shares will be represented at the annual meeting.

If you sign, date and mail your proxy card or voting instruction card without indicating how you wish to vote, your proxy will be counted as a vote **FOR** authorization of the issuance of the shares of Mandalay Digital common stock issuable pursuant to the merger (*Proposal 1*), **FOR** the six director nominees (*Proposal 2*), **FOR** the advisory Say-on-pay proposal (*Proposal 3*), **FOR** the annual (one-year) alternative in the Say-on-frequency proposal (*Proposal 4*), and **FOR** the ratification of SingerLewak LLP as our independent registered public accounting firm for the fiscal year ending March 31, 2015 (*Proposal 5*).

If you sign a proxy, you may revoke it by written notice to the Secretary of Mandalay Digital at any time before it is voted at the annual meeting.

You cannot vote shares held by your broker in "street name." Only your broker can vote those shares, with your instructions. If you do not provide your broker with instructions on how to vote your shares, your broker will not be permitted to vote them other than on the ratification of the appointment of our independent registered public accounting firm (*Proposal 5*).

Voting Matters (Page 44)

The approval of the proposal to authorize the issuance of the shares of Mandalay Digital common stock issuable pursuant to the merger agreement will require the affirmative vote, in person or by proxy, of the holders of a majority of the votes cast by the holders of common stock and the holders of preferred stock (which votes together with the common stock as a single class on an as-converted to common stock basis), at the annual meeting, assuming that a quorum is present in person or by proxy. Brokers who do not receive voting instructions from beneficial owners but who deliver a proxy or attend our annual meeting are considered present at the meeting for purposes of determining whether a quorum exists but are not counted as a vote cast on any non-routine matter presented at the Annual Meeting (all *Proposals* other than *Proposal 5* are non-routine matters).

As for the other matters to be voted on at the annual meeting, assuming a quorum is present, directors will be elected by a plurality vote (*Proposal 2*) (there is no right to vote stock cumulatively), and approval of the other proposals to be considered at the annual meeting (ratification of SingerLewak LLP as Mandalay Digital's auditors for fiscal 2015 (*Proposal 5*), the advisory votes on executive compensation (*Proposal 3*) and say-on-frequency (*Proposal 4*)) will each require the affirmative vote of a majority of the votes present represented by holders of common stock and holders of preferred stock (which votes together with the common stock as a single class and on an as-converted to common stock basis), either in person or by proxy and entitled to vote with respect to each proposal. Each holder of shares of common stock outstanding on the record date will be entitled to one vote for each share held of record. Each holder of shares of preferred stock outstanding on the record date will be entitled to one vote based on an as-converted to common stock basis. Abstentions and broker non-votes will be counted for purposes of determining whether a quorum is present. Abstentions will have the effect of a vote against the matters presented at the annual meeting and broker non-votes will have no effect on any of the matters presented at the annual meeting other than on the ratification of the appointment of our independent registered public accounting firm (*Proposal 5*).

The Companies

Mandalay Digital Group, Inc. (Page 110)

Mandalay Digital, through its wholly owned subsidiary, Digital Turbine, provides end to end mobile content solutions for wireless carriers globally to enable them to better monetize their subscribers. The Company's products include mobile application management through DT Ignite, user experience and discovery through DT IQ, application stores and content through DT Marketplace, and content management and mobile payments through DT Pay. With global headquarters in Los Angeles, and offices in the U.S., Asia Pacific and EMEA, Mandalay Digital's solutions are available worldwide.

Mandalay Digital intends to change its name to "Digital Turbine, Inc." the earlier of the end of January 2015 or at the close of the merger. Our current contact information is as follows: 2811 Caheunga Boulevard West, Los Angeles, California 90068. Our telephone number is (323) 312-4000.

Appia, Inc. (Page 110)

Appia is a leading independent mobile user acquisition network. Appia has delivered more than 88 million app installs for hundreds of advertisers, including 60 of the top 100 grossing apps on the App Store and Google Play. Appia provides the technology, infrastructure, and scale to provide higher lifetime value users for advertisers while driving positive return on ad spend. In addition, Appia partners with a diverse set of publishers including app developers, mobile websites, and carriers to maximize their advertising revenue. Appia's current contact information is as follows: 320 Blackwell Street, Durham, North Carolina, 27701. Its telephone number is (919) 617-1656.

The Merger (Page 51)

On November 13, 2014, Mandalay Digital entered into the merger agreement with Merger Sub, Appia, and Shareholder Representative Services LLC, as the stockholder representative, pursuant to which Merger Sub will merge with and into Appia on the terms and subject to the conditions set forth in the merger agreement, so that Appia, as the surviving corporation, will become a whollyowned subsidiary of Mandalay Digital. The surviving corporation will be renamed to "Digital Turbine Media, Inc." The merger is expected to close in the first calendar quarter of 2015.

Reasons for the Proposed Merger (Page 58)

We believe that that the merger will be transformational for Mandalay Digital's business as it will provide strategic and financial benefits, including the opportunity to achieve revenue and cost synergies, and

enhance strategic and competitive positioning. We expect that combining Appia's app install platform and advertising relationships with Digital Turbine's carrier distribution network will expand our global reach, will accelerate our growth, and will further enhance our scalable business model.

Recommendation of Mandalay Digital's Board of Directors (Page 58)

Based on the reasons for the merger described in this proxy statement/prospectus, our board of directors has unanimously recommended that holders of our common stock and preferred stock (which votes together with the common stock as a single class and on an as-converted basis), vote "FOR" the issuance of up to 19,500,000 shares pursuant to the merger agreement and an additional 600,000 shares of our common stock (including a warrant to purchase 400,000 shares, which vests 12 months after issuance under certain circumstances) in connection with a separate loan transaction.

What Appia Stockholders will Receive in the Merger (Page 69)

At the effective time of the merger, on the terms and subject to the conditions set forth in the merger agreement, all of the equity securities of Appia outstanding immediately prior to the effective time (including outstanding stock options, which will be assumed by Mandalay Digital), other than equity securities held by stockholders that exercise dissenters' rights under applicable law, will be converted into the right to receive a number of shares (or, in the case of stock options being assumed, stock options) of Mandalay Digital common stock, equal to \$100 million less Appia's net debt and transaction expenses at the time of closing (subject to a working capital adjustment), divided by \$4.50. Mandalay Digital expects to issue, pursuant to the merger agreement, up to 19,500,000 shares of common stock (including shares underlying assumed stock options and 1,000,000 shares to be deposited into escrow).

The aggregate number shares of Mandalay Digital common stock to be issued pursuant to the merger agreement will vary depending on a number of factors, including the amount of the estimated net debt at closing. The merger agreement does not provide for an adjustment in the aggregate number of shares of Mandalay Digital common stock or options to be issued by Mandalay Digital to Appia's security holders in the acquisition in the event of fluctuations in the market value of Mandalay Digital's common stock up through the closing date. However, the number of shares to be issued by Mandalay Digital pursuant to the merger agreement is subject to adjustment based on Appia's working capital and net debt as of the closing date of the merger. We currently estimate that, if the merger closes on January 31, 2015, we would issue 19,056,392 shares; additionally, we would assume options for 277,376 shares of our common stock, for an aggregate of 19,333,768 shares, which would represent approximately 33% of the Company's outstanding shares of common stock as of November 13, 2014 (after giving effect to such issuance). Appia's actual results of operations between the date of the financial information included in this proxy statement/prospectus and the closing date may differ from the forecasted results used in this estimate, resulting in differences between the forecast and actual amounts of Appia's working capital and net debt, including transaction expenses, as of the closing date of the merger. See "The Merger – Consideration Payable in the Merger" located on page 68 of this proxy statement/prospectus for further information and a discussion of the assumptions in calculating the potential amount of shares to be issued by Mandalay Digital in the merger.

Principal Agreements (Page 81)

Merger Agreement

The merger agreement, which was executed on November 13, 2014, sets forth the terms and conditions of the merger, including terms for the conversion of securities in the merger, treatment of Series C warrants and Appia options and restricted common stock, escrow of consideration, working capital/net debt adjustment, representations and warranties, indemnification, covenants of each party, closing conditions and termination provisions.

Support Agreement

Concurrently with the execution of the merger agreement, certain Appia stockholders, who collectively hold approximately 77% of the outstanding shares of Appia common stock and approximately 94% of the outstanding shares of Appia preferred stock (which votes together as a single class on an as-converted to common stock basis), each have entered into a support agreement with Mandalay Digital, pursuant to which certain Appia stockholders have agreed, among other things, to vote (or deliver written consents with respect to) all of the shares of Appia capital stock held by such stockholder in favor of the approval of the merger and the merger agreement and the other transactions contemplated thereby, to vote against any action that would reasonably likely result in a material breach of a representation, warranty, covenant or agreement of Appia under the merger agreement, and to vote against any alternative transactions. Each support agreement also generally prohibits the Appia stockholder signing it from transferring such stockholder's shares of Appia capital stock prior to the consummation of the merger. Each support agreement will terminate upon the termination of the merger agreement in accordance with its terms. Collectively, the support agreements provide for the vote or written consent of a sufficient number of shares of Appia to adopt the merger agreement and approve and the transactions contemplated thereby under Delaware law and Appia's organizational documents. See "Ancillary Agreements—Support Agreement" on page 98.

Financing Agreements

Upon consummation of the merger, Mandalay Digital will guarantee Appia's existing indebtedness outstanding as of closing, which amount is expected to be approximately \$9.5 million (excluding the transaction fees and expenses) assuming the merger closes on January 31, 2015. See "The Merger – Consideration Payable in the Merger" located on page 69 of this proxy statement/prospectus for further information and assumptions about Appia's forecasted indebtedness based on certain closing dates. Upon consummation, and as a condition to the closing, of the merger, Appia will enter into an amended loan agreement with its current senior lender, Silicon Valley Bank, and a securities purchase agreement with Mandalay Digital and North Atlantic pursuant to which Appia will issue to North Atlantic a subordinated debenture in the aggregate principal amount of \$8.0 million, which amount is included in the \$9.5 million referenced above, to retire a like amount of outstanding debt to North Atlantic.

Upon consummation of the merger, Mandalay Digital will issue a secured guaranty of each of the SVB Debt and North Atlantic Debt whereby we will guarantee all of Appia's obligations in connection with such indebtedness and pledge substantially all of its assets, including its intellectual property. The secured guarantees will contain cross-default provisions and covenants, among others, limiting our ability to undergo a change of control, incur indebtedness, grant liens, make dividends in cash and other customary covenants. Mandalay will also issue to North Atlantic 200,000 shares of its common stock and a common stock warrant to purchase 400,000 shares of its common stock all of which will have registration rights requiring Mandalay Digital to file a registration statement 15 business days after the closing of the merger with Appia. The warrant will have an exercise price of \$0.01 per share exercisable for 10 years, but it will not be exercisable until the one year anniversary of the closing date of the merger and will terminate if the Company repays the North Atlantic Debt prior to such one year anniversary. We intend to refinance the Appia debt as soon as practicable following the closing of the merger.

Escrow Agreement

Mandalay Digital will deposit one million shares into an escrow account to cover indemnification claims by the Company pursuant to the terms of the merger agreement, which shares will be released one year after the closing, subject to any then pending claims.

Approval by Holders of Mandalay Digital Common and Preferred Stock (Page 49)

Approval of the issuance of shares of our common stock in connection with the merger requires the affirmative vote of holders of shares representing not less than a majority of the votes cast by holders of our common stock and our preferred stock, voting together as a single class on an as-converted to common stock basis, in person or represented by proxy at the annual meeting and entitled to vote thereon, provided that a quorum consisting of the holders of not less than a majority of the votes entitled to be cast by our common and preferred stockholders is present at the annual meeting in person or by proxy.

At the close of business on the Record Date, directors and executive officers of our company beneficially owned and were entitled to vote approximately [\square]% of the [\square] shares of our common stock outstanding on that date. Directors and executive officers of Appia and their affiliates did not beneficially own any of our shares of common stock outstanding on that date.

Opinion of Mandalay Digital's Financial Advisor (Page 60)

At a meeting of our board of directors on November 12, 2014, Needham & Company, LLC ("Needham & Company") rendered its oral opinion to our board of directors, which was confirmed by delivery of a written opinion of the same date, that, as of that date and based upon and subject to the assumptions and other matters described in the written opinion, the consideration to be paid by us in the merger pursuant to the merger agreement was fair to us from a financial point of view. The full text of the written opinion of Needham & Company is attached to this proxy statement/prospectus as Appendix C and is incorporated in this proxy statement/prospectus by reference. The discussion under the section entitled "The Merger— Opinion of Mandalay Digital's Financial Advisor," together with Needham & Company's written opinion, set forth, among other things, the assumptions made, procedures followed, matters considered, and limitations on and scope of the review undertaken by Needham & Company in connection with its opinion. Holders of our common stock and preferred stock should read this opinion and the aforementioned section carefully and in its entirety. Needham & Company provided its opinion to our board of directors for the information and assistance of our board of directors (in its capacity as such) in connection with and for purposes of the board's evaluation of the transaction contemplated by the merger agreement. Needham & Company's opinion relates only to the fairness, from a financial point of view, of the consideration to be paid by us in the merger, which was determined through arm's length negotiations between Mandalay Digital and Appia and not by Needham & Company. The Needham & Company opinion does not address the underlying decision by us to enter into the merger transaction or any other aspect of the merger and is not a recommendation as to how any holder of our common stock or preferred stock should vote or act with respect to the merger or any other matter.

Risk Factors (Page 23)

In deciding how to vote your shares of stock (if you are a Mandalay Digital stockholder) on the matters described in this proxy statement/prospectus, you should carefully consider the risks related to the merger and the combined company. The merger may not achieve the expected benefits because of, among other things, the risks and uncertainties discussed in the sections entitled "Risk Factors" and "Cautionary Statement Regarding Forward-Looking Information." Such risks include, among other things, risks relating to the uncertainty that we and Appia will be able to integrate our businesses successfully, uncertainties as to whether the merger will achieve expected synergies, and uncertainties relating to the performance of the combined company following the merger.

Conditions to the Merger (Page 93)

We and Appia expect to complete the merger after all of the conditions to the merger in the merger agreement are satisfied or waived, including after the receipt of approval of our stockholders of the issuance of

our common stock at our annual meeting of stockholders and of the approval by the requisite vote or consent of Appia's stockholders, and the receipt of all required regulatory approvals. We and Appia currently expect to complete the merger during the first calendar quarter of 2015. However, it is possible that factors outside of either our or Appia's control could cause the merger to be completed at a later time or not at all.

Each party's obligation to complete the merger is subject to the satisfaction or waiver (to the extent waiver is permitted under the merger agreement) of various conditions, including conditions relating to the following matters:

The merger agreement contains customary representations, warranties, covenants, and agreements of the parties thereto, and completion of the transaction is subject to customary closing conditions, including, among others, (i) approval of the issuance of the shares by the Company stockholders pursuant to rules of The NASDAQ Stock Market, LLC, (ii) approval of the merger by the Appia stockholders, (iii) not more than 5% of the Appia stockholders having exercised dissenters' rights, (iv) the execution of certain ancillary agreements, (v) all required consents and approvals necessary to consummate the merger having been obtained, (vi) the effectiveness of the registration statement on Form S-4 (of which this proxy statement/prospectus is a part) for purposes of registering the shares, (v) no injunction or proceeding by a government entity seeking to restrain or prohibit consummation of the merger, (vi) no material adverse changes of Appia, (vii) the other party's compliance with its covenants and agreements contained in the merger agreement (subject to customary materiality qualifiers), (viii) each party's receipt of an opinion from its respective outside legal counsel that is reasonably acceptable and dated as of the closing date of the merger to the effect that, on the basis of the facts, representations and assumptions set forth or referred to in such opinion, the merger will qualify for U.S. federal income tax purposes as a reorganization within the meaning of Section 368(a) of the Code; and (ix) Appia having entered into restated financing arrangements with its lenders. Furthermore, Appia's net debt (including transaction expenses) may not exceed \$13.75 million, \$14.5 and \$15 million if the closing is on or before January 31, 2015, after January 31, 2015 and before February 28, 2015 or after February 28, 2015, respectively.

Termination of the Merger Agreement (Page 96)

The merger agreement contains certain provisions giving Mandalay Digital and Appia the right to terminate the merger agreement under certain circumstances.

Termination Fees and Expenses (Page 97)

If Appia terminates the merger agreement because of an intentional and willful material breach by Mandalay Digital of a representation, warranty, condition or covenant or the failure of Mandalay Digital stockholders to approve the merger agreement (and if all other conditions to Mandalay Digital's obligation to consummate the merger are satisfied), then Mandalay Digital will be required to pay Appia a termination fee of \$1.0 million and reimburse Appia for expenses of up to \$500,000. Alternatively, Appia may elect to instead pursue other remedies in the case of a termination by Appia based upon an intentional and willful material breach by Mandalay Digital.

Interests of Certain Persons in the Merger (Page 72)

Certain directors, executive officers, and significant stockholders of Appia have interests in the merger that may be in addition to or different from those of other Appia stockholders as described in more detail in this proxy statement/prospectus.

Material U.S. Federal Income Tax Consequences of the Merger (Page 78)

The merger is intended to qualify as a "reorganization" within the meaning of Section 368(a) of the Code, and it is a condition to the completion of the merger that Mandalay Digital and Appia each receive written

opinions from their respective outside legal counsel, dated as of the closing date of the merger, to the effect that the merger will qualify as a reorganization within the meaning of Section 368(a) of the Code. Such opinions will be based on certain assumptions, representations and covenants, including representations and covenants contained in representation letters provided by Mandalay Digital, Merger Sub and Appia. Assuming the merger qualifies as a reorganization, Appia stockholders generally will not recognize gain or loss for U.S. federal income tax purposes upon the exchange of their shares of Appia stock for shares of Mandalay Digital common stock in connection with the merger. The issuance of shares of Mandalay common stock in connection with the merger will not be a taxable transaction to Mandalay Digital stockholders for U.S. federal income tax purposes.

Tax matters are very complicated, and the tax consequences of the merger to a particular stockholder will depend on such stockholder's circumstances. Accordingly, you are urged to consult your tax advisor for a full understanding of the tax consequences of the merger to you, including the applicability and effect of U.S. federal, state, local and foreign income and other tax laws.

Anticipated Accounting Treatment of the Merger (Page 72)

Our acquisition of Appia in the merger will be accounted for under the acquisition method of accounting in accordance with U.S. GAAP. From the date of completion of the merger, our results of operations will include Appia's operating results and assets and liabilities.

Regulatory Matters (Page 71)

We and Appia have each agreed to use our reasonable best efforts to take all actions proper or advisable under the merger agreement and applicable laws, rules, and regulations to complete the merger, as well as take other actions specified in the merger agreement, as promptly as practicable.

Directors of Mandalay Digital Following Completion of the Merger (Page 74)

Upon the closing of the merger, we will appoint Jud Bowman, founder and Chief Executive Officer of Appia, and Craig Forman, Appia's Executive Chairman, to become members of our board of directors.

Appraisal Rights (Page 87)

Under Delaware law and our Certificate of Incorporation, holders of our common stock are not entitled to any rights to seek appraisal of their shares or to exercise any dissenter's or preemptive rights in connection with the proposal to issue up to 19,500,000 shares of our common stock pursuant to the merger agreement.

Comparison of Stockholder Rights (Page 124)

We are incorporated in the state of Delaware, and the rights of our stockholders are governed by the Delaware General Corporation Law and by our Certificate of Incorporation, as amended, and by our Bylaws, as amended, which we refer to as our Bylaws. Appia is also incorporated in the state of Delaware and the rights of Appia stockholders are currently governed by the Delaware General Corporation Law and Appia's Certificate of Incorporation and Bylaws. After the completion of the merger, stockholders of Appia who receive shares of our common stock in the merger will become stockholders of Mandalay Digital and will become subject to our Certificate of Incorporation and Bylaws and the applicable provisions of Delaware law.

SELECTED UNAUDITED PRO FORMA CONDENSED COMBINED FINANCIAL DATA

The following Unaudited Pro Forma Combined Financial Statements have been developed from the historical consolidated financial statements of Mandalay Digital and Appia. Mandalay Digital's financials are incorporated by reference into this proxy statement/prospectus and Appia's financials are included elsewhere in this proxy statement/prospectus. The table below should be read together with, and is qualified in its entirety by the historical financial statements, including the notes thereto, of Mandalay Digital and Appia, respectively, as well as the information provided about Appia's revenue included in "Management's Discussion and Analysis of Financial Condition and Result of Operations of the Business of Appia" and the more detailed Unaudited Pro Forma Condensed Combined Financial Statements, including the notes thereto, appearing under "Unaudited Pro Forma Condensed Combined Financial Statements" beginning on page 101 of this proxy statement/prospectus.

Mandalay Digital and Appia have different fiscal year ends, with the most recent annual period of Mandalay Digital ended on March 31, 2014, and the most recent annual period of Appia ended on December 31, 2013. As such, amounts related to the historical operations of Appia have been adjusted to align the periods over which those operations occurred with the periods presented by adding the necessary quarterly results together to match Mandalay Digital's fiscal reporting periods. The Unaudited Pro Forma Combined Statements of Operations for the six months ended September 30, 2014 and for the year ended March 31, 2014 combine the historical consolidated statements of operations of Mandalay Digital and Appia, giving effect to the merger as if it had been consummated on April 1, 2013, the beginning of the earliest period presented. The Unaudited Pro Forma Combined Balance Sheet combines the consolidated balance sheets of Mandalay Digital and Appia, giving effect to the merger as if it had been consummated on September 30, 2014.

The Unaudited Pro Forma Combined Financial Statements are provided for illustrative purposes only and do not purport to represent what the actual consolidated results of operations or the consolidated financial position of Mandalay Digital would have been if the merger had occurred on the dates assumed, nor are they necessarily indicative of future consolidated results of operations or consolidated financial position. The Unaudited Pro Forma Combined Financial Statements do not reflect the costs of any integration activities including planning costs or any benefits that may result from realization of future cost savings from operating efficiencies or revenue synergies expected to result from the merger, except to the extent that such integration costs have been incurred during the periods presented. In addition, the Unaudited Pro Forma Combined Financial Statements (excluding the balance sheet) do not include one-time costs directly attributable to the transaction or professional fees incurred by Mandalay Digital or Appia pursuant to provisions contained in the merger agreement as those costs are not considered part of the purchase price.

For the Six Months Ended September 30, 2014 and the Year Ended March 31, 2014 (In thousands, except per-share amounts)

	x months ended nber 30, 2014	Year ended <u>March 31, 2014</u>
Revenues	\$ 25,486	\$ 73,533
Loss from operations	(14,496)	(19,694)
Loss from operations before income taxes	(15,766)	(22,882)
Income tax provision / (benefit)	355	(272)
Net loss from continuing operations	(16,121)	(22,610)
Basic and diluted net loss per common share from continuing operations	\$ (0.28)	\$ (0.48)

	As of September 30, 2014
Total assets	\$ 111,396
Long term debt, net of discount	8,143

STOCK PRICE AND DIVIDEND INFORMATION

Mandalay Digital

Stock Price Information

The table below sets forth the high and low sales prices per share of Mandalay Digital common stock, as reported on the NASDAQ Stock Market. Our common stock is listed on the NASDAQ Stock Market under the symbol "MNDL." Effective January 20, 2015, our name will change to "Digital Turbine, Inc" and our symbol will change to "APPS." Share prices have been adjusted to reflect a 1-for-5 reverse stock split effective on April 12, 2013.

	High	Low
Fiscal Year Ended March 31, 2015 (to date)		
First quarter	\$4.22	\$3.11
Second quarter	\$6.47	\$3.09
Third quarter	\$4.60	\$2.91
Fourth quarter (as of January 15, 2015)	\$3.55	\$2.93
Fiscal Year Ended March 31, 2014		
First quarter	\$6.00	\$3.80
Second quarter	\$4.79	\$2.31
Third quarter	\$3.29	\$2.28
Fourth quarter	\$4.89	\$2.57
Fiscal Year Ended March 31, 2013		
First quarter	\$5.05	\$3.60
Second quarter	\$4.25	\$3.65
Third quarter	\$4.10	\$3.40
Fourth quarter	\$4.75	\$3.35

On November 13, 2014, the date of execution and announcement of the merger agreement, Mandalay Digital's common stock closed at \$3.44.

The market price of our common stock could change significantly. Because the equity consideration to be paid in the merger will not be adjusted for changes in the market price of our common stock through the closing date of the merger, the value of the consideration that holders of Appia shares will receive in the merger may vary significantly from the market value of our common stock that holders of Appia shares would have received if the merger was completed on November 13, 2014, the trading day on which the announcement of the merger was made, or on the date of this proxy statement/prospectus.

Holders

As of [I], 2015, the record date for the annual meeting, there were [I] holders of record of our common stock. There were also an undetermined number of holders who hold their stock in nominee or "street" name.

Dividend Policies

We have not declared cash dividends on our common stock since our inception and we do not anticipate paying any cash dividends in the foreseeable future. Following completion of the merger, the holders of our common stock will be entitled to receive any dividends as may be declared by our board of directors from funds legally available therefor; provided, however, that we will be subject to covenants restricting our ability to pay dividends pursuant to the terms of the secured guarantees that we will enter into in connection with the closing of the merger. The dividend policy of the combined company will be determined by our board of directors.

Appia

Historical market price information regarding Appia capital stock is not provided because there is no public market for Appia's common stock. As of November 13, 2014, there were 40 holders of Appia capital stock. Appia has not declared cash dividends on its common stock since its inception and it does not anticipate paying any cash dividends in the foreseeable future.

RISK FACTORS

Our stockholders should carefully consider the following factors in evaluating whether to approve the issuance of our common stock in connection with the merger, and Appia's stockholders should carefully consider the following factors in connection with their potential receipt of shares of our common stock in the merger. These factors should be considered in conjunction with the other information included or incorporated by reference in this proxy statement/prospectus. Additional risks and uncertainties not presently known to us, or that are not currently believed to be important to you, also may adversely affect the stockholders of Appia and us following the merger.

Risks Related to the Merger

Even if the proposed merger is completed, we may not be able to fully realize all of the projected synergies and anticipated benefits of the merger.

The success of the merger will depend on, among other things, the ability to realize anticipated revenue and cost savings, enhance the combined company's global reach, enhance our scalable, low capex business model that drives EBITDA and to combine our business with Appia in a manner that potentially creates opportunities through strategic relationships, such as leveraging advertising opportunities and increasing revenue streams for carriers, that does not materially disrupt existing relationships or otherwise result in decreased productivity and that allows us to capitalize on the capabilities of the combined company. If these objectives are not achieved, the anticipated benefits of the merger may not be realized fully or at all or may take longer to realize than expected.

We may not realize projected revenue gains, cost savings and synergies in connection with the proposed merger on the timetable contemplated, if at all. In addition, the actual integration may result in additional and unforeseen expenses, and the anticipated benefits of the integration plan may not be realized. Actual revenue and cost synergies, if achieved at all, may not be achieved at the levels expected and may take longer than anticipated. If these challenges are not adequately addressed, Mandalay Digital and Appia may be unable to realize the anticipated benefits of the integration of the two companies. An inability to realize the full extent of, or any of, the anticipated benefits of the merger, as well as any delays encountered in the integration process, could have a material adverse effect on the combined company's business and results of operations, which may affect the value of the shares of the combined company's common stock.

We may not realize the operating and financial benefits we expect from the merger.

The post-acquisition integration of our company and Appia would be complex, time-consuming, and expensive, and may disrupt the day-to-day management and operation of our respective businesses. Both Mandalay Digital and Appia have operated and, until the completion of the merger, will continue to operate independently. It is possible that the integration process could result in the disruption of our or Appia's ongoing businesses or inconsistencies in standards, controls, procedures or policies that could adversely affect the ability of the combined company to maintain relationships with third parties and employees or to achieve the anticipated benefits of the merger. After the merger, the combined company would need to overcome significant challenges in order to realize any benefits or synergies from the merger. These challenges include the timely, efficient, and successful completion of a number of post-acquisition events, including the following:

- integrating the operations and technologies of the companies, including the rapid integration of Appia's app-install infrastructure with our platform;
- implementing of disclosure controls, internal controls, and financial reporting systems to comply with the requirements of U.S. GAAP and U.S. securities laws and regulations required as a result of integration of Appia as part of a consolidated reporting company under the Exchange Act;
- retaining and assimilating the key personnel of each company;

- resolving possible inconsistencies in operating and product standards, internal controls, procedures and policies, business cultures, corporate governance and reporting practices, and compensation methodologies between the companies;
- retaining existing vendors and customers of the companies and attracting additional customers;
- · retaining strategic partners of each company and attracting new strategic partners; and
- creating uniform business standards, procedures, policies, and information systems.

The execution of these post-acquisition integration events would involve considerable risks and may not be successfully implemented, or if implemented, on a timely basis. These risks include the following:

- potential disruption of ongoing business operations and distraction of the management of the combined company;
- potential strain on financial and managerial controls and reporting systems and procedures of the combined company;
- unanticipated expenses and potential delays related to integration of the operations, technology, and other resources of the companies;
- potential impairment of relationships with employees, suppliers, and customers as a result of the inclusion and integration of management personnel;
- greater than anticipated costs and expenses related to the merger or the integration of the respective businesses of us and Appia following the merger; and
- potential unknown liabilities associated with the merger and the combined operations.

The combined company may not succeed in mitigating these risks or any other problems encountered in connection with the merger. The inability to successfully integrate the operations, technology, and personnel of our company and Appia, or any significant delay in achieving integration of the companies, could have a material adverse effect on the combined company after the merger and, as a result, on the market price of our common stock following the merger.

Failure to complete, or a delay in completing, the proposed merger could adversely affect our and Appia's future business and operations.

The proposed merger is subject to the satisfaction of various closing conditions, including the approval by both our and Appia's stockholders and other conditions described in the merger agreement that are outside the control of us and Appia that may prevent, delay or otherwise materially adversely affect its completion. Our and Appia's respective obligations to complete the merger are also subject to the other conditions listed under "Summary of the Merger Agreement — Conditions to Consummation of the Merger" on page 93. We cannot assure you whether or when these conditions will be satisfied, or whether the merger will be successfully completed. Any delay in completing the merger could cause the combined company not to realize some or all of the synergies and other benefits that we and Appia expect to achieve if the merger is successfully completed within its expected time frame. For instance, a delay in completing the merger may materially adversely affect Appia's cash flow and Appia may require additional financing or take steps to reduce costs prior to the completion of the merger, and as a result would reduce the financial benefits that the combined company would expect to achieve, once the merger is completed.

In the event that the merger is not completed:

- we and Appia would not realize the potential benefits of the merger, including anticipated revenue and cost synergies;
- our and Appia's management's attention from day-to-day business may be diverted, we and Appia may lose key employees, and our and Appia's relationships with our respective customers and partners may be disrupted as a result of uncertainties with regard to our and Appia's business and prospects;
- the business relationship between us and Appia may be adversely impacted;
- under certain circumstances, we may be required to pay to Appia a termination fee of \$1 million and expense reimbursement of up to \$500,000; and
- we and Appia will each incur and must pay significant costs and expenses related to the merger, such as legal, accounting, and advisory fees.

Any such events could adversely affect our and Appia's business and operating results.

Our and Appia's business could suffer due to the announcement, pendency, and consummation of the proposed merger.

The announcement, pendency, and consummation of the merger may have a negative impact on our, Appia's, and the combined company's ability to sell their respective products and services, attract and retain key management, technical, sales, or other personnel, maintain and attract new customers, and maintain strategic relationships with third parties. For example, we and Appia, and following consummation of the merger, the combined company, may experience the deferral, cancellation, or a decline in the size or rate of orders for our services or a deterioration in customer relationships. If this were to occur, our and Appia's cash flows and revenue, respectively, and the revenues of the combined company, could decline materially or any anticipated increases in revenue could be lower than expected. Also, speculation regarding the likelihood of the closing of the merger could increase the volatility of our share price. Any such events could harm our and Appia's, and following the merger the combined company's, operating results and financial condition.

The price of our common stock may fluctuate, and the purchase price payable in the merger will not be adjusted for any changes in the price of our common stock.

The consideration payable in connection with the merger will be paid through the issuance to the stockholders of Appia of up to 19,500,000 shares of our common stock. Under the merger agreement, other than as a result of reclassifications, stock splits, stock dividends, and similar changes effected by us, the number of shares of our common stock to be issued will not be adjusted even if the market price of our common stock fluctuates between the date of the merger agreement and the closing date of the merger. The market price of our common stock at the closing of the merger will likely vary from the market price at the date of this proxy statement/prospectus and at the date of our stockholders' meeting. Stock price changes may result from a variety of factors that are beyond the control of us or Appia, including:

- market reaction to the announcement and pendency of the merger and market assessment of the merits and risks of the merger and the likelihood of the merger being consummated;
- changes in the respective businesses, operations, or prospects of our or Appia's business;
- governmental or litigation developments or regulatory considerations affecting us or our industry;

- general business, market, industry, or economic conditions; and
- other factors beyond the control of us or Appia, including those described elsewhere in this "Risk Factors" section.

Neither party is permitted to "walk away" from the merger or re-solicit the vote of its stockholders solely because of changes in the market price, and therefore value, of our common stock through the closing date of the merger. Any reduction in our stock price would result in Appia stockholders receiving less value in the merger. Conversely, any increase in our stock price would potentially result in Appia stockholders receiving greater value in the merger. The specific dollar value per share of our common stock that Appia's stockholders would receive upon completion of the merger will depend on, among other things, the market value of our common stock at that time and other factors discussed in this proxy statement/prospectus. Our stockholders will not know the exact value of our common stock to be issued in the merger at the time of the stockholders' meeting.

If the merger does not qualify as a "reorganization" within the meaning of Section 368(a) of the Code, the stockholders of Appia may be required to pay substantial U.S. federal income taxes.

Mandalay Digital and Appia intend, and their respective tax counsel will provide opinions to the effect, that the merger will qualify as a "reorganization" within the meaning of Section 368(a) of the Code. The opinions of their respective tax counsel will be based on certain assumptions, representations and covenants made by Mandalay Digital, Merger Sub and Appia. If any of those representations, covenants and assumptions is inaccurate, the conclusions reached by counsel in such opinions may not apply. Moreover, the opinions of their respective tax counsel do not bind the Internal Revenue Service, or IRS, nor do they prevent the IRS from adopting a contrary position. Neither Mandalay Digital nor Appia has requested, or intends to request, a ruling from the IRS, with respect to the tax consequences of the merger, and there can be no assurance that the companies' position would be sustained by a court if challenged by the IRS. If the merger does not qualify as a reorganization within the meaning of Section 368(a) of the Code, Appia stockholders will recognize gain or loss upon their receipt of Mandalay Digital common stock in connection with the merger. For a more complete discussion of the tax consequences of the merger, see the section entitled "Material U.S. Federal Income Tax Consequences" beginning on page 78.

The merger is expected to result in an ownership change for Appia and may result in an ownership change for Mandalay Digital under Section 382 of the Code, substantially limiting the use of the net operating loss, referred to as "NOL," carryforwards and other tax attributes of Appia and possibly of Mandalay Digital to offset future taxable income of the combined company.

In general, under Section 382 of the Code, a corporation that undergoes an "ownership change" within the meaning of Section 382 of the Code is subject to limitations on the utilization of net operating loss carryforwards, which we refer to as NOL carryforwards, generated prior to such ownership change to offset future taxable income. In general, an ownership change occurs if the aggregate stock ownership of certain stockholders increases by more than 50 percentage points over such stockholders' lowest percentage ownership during the testing period (which is generally three years). If an ownership change occurs, Section 382 imposes an annual limitation on the amount of income against which pre ownership change NOL carryforwards may be offset generally equal to the value of the stock of the corporation immediately prior to the ownership change, multiplied by the adjusted federal tax-exempt rate set by the IRS.

The merger is expected to result in an ownership change for Appia and may result in an ownership change for Mandalay Digital under Section 382 of the Code, potentially limiting the use of the NOL carryforwards of Appia and Mandalay Digital to offset future taxable income of the combined company for both federal and state income tax purposes. Whether the merger results in an ownership change for Mandalay Digital depends on other past and future changes in stock ownership. Moreover, the utilization of the combined

company's NOL carryforwards depends on the timing and amount of taxable income earned in the future, which neither Appia nor Mandalay Digital is able to predict. These tax attributes are subject to expiration at various times in the future to the extent that they have not been applied to offset the taxable income of the combined company. These limitations may affect the timing of when these NOL carryforwards can be used which, in turn, may impact the timing of when cash is used to pay the taxes of the combined company and could limit the benefits of NOL carryforwards for the business, financial condition or results of operations of the combined company.

The merger could cause us or Appia to lose key personnel, which could materially affect the combined company's business and require the combined company to incur substantial costs to recruit replacements for lost personnel.

As a result of the merger, our current and prospective employees and Appia's employees could experience uncertainty about their future roles within the combined company. This uncertainty may adversely affect their ability or willingness to continue with the combined company, and the ability of the combined company to attract and retain key management, sales, marketing, and technical personnel. Any failure to retain and attract key personnel could have a material adverse effect on our and Appia's current business and the business of the combined company after the completion of the merger.

Some of Appia's officers and directors have conflicts of interest that may influence them to support or approve the issuance of shares in connection with the Merger.

Certain officers and directors of Appia participate in arrangements that provide them with material interests in the merger that are different from and in addition to those of our stockholders and Appia's stockholders, including, among others, employment agreements and compensation arrangements for key officers, and indemnification arrangements for Appia's directors and officers. These interests, among others, may influence our directors and officers and the directors and officers of Appia to support or approve the merger. For a more detailed discussion, see the section entitled "The Merger — Interests of Certain Appia Directors, Officers, and Affiliates in the Merger" on page 72.

We incur a variety of costs as a result of being a public company, and those costs may increase as a result of the merger.

As a U.S. public company registered with the SEC under the Exchange Act, we incur significant legal, accounting, and other expenses. In addition, the Sarbanes-Oxley Act of 2002, as well as rules subsequently implemented by the SEC and the NASDAQ Stock Market, frequently require changes in corporate governance policies and practices of companies registered with the SEC under the Exchange Act. These rules and regulations increase legal and financial compliance costs and make some activities more time-consuming and costly. In addition, we incur additional costs associated with our Exchange Act public company reporting requirements. These rules and regulations also may make it more difficult and more expensive for us to obtain and pay for, at commercially reasonable rates, director and officer liability insurance, and the combined company may be required to accept reduced policy limits and reduced scope of coverage or incur substantially higher costs to obtain the same or similar levels of coverage. As a result, it may be more difficult for the combined company to attract and retain qualified persons to serve on its board of directors or as executive officers. As a result, implementation of disclosure controls, internal controls, and financial reporting systems complying with the requirements of U.S. GAAP and U.S. securities laws and regulations required as a result of our continued status as a reporting company under the Exchange Act following effectiveness of the merger may be more difficult and costly than anticipated.

Following the effectiveness of the merger, the current principal owners of Appia are expected to own a substantial percentage of our common stock.

We currently estimate that, if the merger closes on January 31, 2015, we would issue 19,056,392 shares; additionally, we would assume options for 277,376 shares of our common stock, for an aggregate of 19,333,768 shares to Appia's former security holders, which would represent approximately 33% of our outstanding shares of common stock as of November 13, 2014 (after giving effect to such issuance). The former Appia stockholders will have influence over our management, operations, and potential significant corporate actions. The interests of the former Appia stockholders could conflict with the interests of our other stockholders and there can be no assurance that the former Appia stockholders would not take actions that favor their interests and not the interests of our other stockholders.

Current holders of our common stock would suffer substantial dilution if the merger is effected.

The merger would dilute the ownership position of our current stockholders. If the merger is effected, we estimate that we would issue 19,056,392 shares; additionally, we would assume options for 277,376 shares of our common stock, for an aggregate of 19,333,768 shares to Appia's former security holders if the merger closes on January 31, 2015, which would represent approximately 33% of our outstanding shares of common stock as of November 13, 2014 (after giving effect to such issuance). We will also issue to a lender 200,000 shares of common stock and a warrant to purchase 400,000 shares of common stock in connection with a separate loan transaction. Consequently, following the merger our current stockholders, as a general matter, would have less influence over the management and policies of our company than they currently exercise over the management and policies of our company.

The shares of our common stock to be received by Appia stockholders as a result of the merger will have different rights from the shares of Appia stock.

Upon consummation of the merger, Appia stockholders, other than those who properly exercise dissenters' rights, will become our stockholders, and their rights as stockholders will be governed by our Certificate of Incorporation and Bylaws. Certain of the rights associated with our common stock are different from, and may be viewed as less favorable than, the rights associated with Appia's outstanding common stock and preferred stock. See "Comparison of Rights of Stockholders of Mandalay Digital and Appia" beginning on page 124 of this proxy statement/ prospectus for a discussion of the different rights associated with our common stock.

We expect to incur significant costs as a result of the integration of our operations with Appia.

There are inconsistencies in standards, controls, procedures and policies, business cultures, and compensation structures between us and Appia. The integration of our operations and the operations of Appia and reconciling the inconsistencies in the standards, controls, procedures and policies, business cultures, and compensation structures between us and Appia may result in additional costs for the combined company. There are no assurances that such inconsistencies can be reconciled seamlessly or at all. The failure to reconcile such inconsistencies may lessen the anticipated benefits of the merger.

The unaudited pro forma condensed combined financial statements included in this proxy statement/prospectus are preliminary and the actual financial condition and results of operations after the merger may differ materially.

The unaudited pro forma condensed combined financial statements in this proxy statement/prospectus are presented for illustrative purposes only are not necessarily indicative of what Mandalay Digital's actual financial condition or results or operations would have been had the merger been completed on the dates indicated. The unaudited pro forma condensed combined financial statements reflect adjustments, which are

based upon preliminary estimates, to record Appia's identifiable assets acquired and liabilities assumed at fair value and the resulting goodwill recognized. The purchase price allocation reflected in this document is preliminary, and final allocation of the purchase price will be based upon the actual purchase price and the fair value of the assets and liabilities of Appia as of the date of the completion of the merger. Accordingly, the final acquisition accounting adjustments may differ materially from the pro forma adjustments reflected in this document.

Risks Related to the Combined Company

The combined company will operate in an intensely competitive industry, and the combined company may not be able to compete successfully.

The mobile advertising market is highly competitive, with numerous companies providing mobile advertising services. The combined company's mobile advertising platform will compete primarily with Facebook, Twitter, and Google, all of which are significantly larger than the combined company and have far more capital to invest in their mobile advertising businesses. The combined company will also compete with in-house solutions used by companies who choose to coordinate mobile advertising across their own properties, such as Yahoo!, Pandora, and other independent publishers. They, or other companies that offer competing mobile advertising solutions, may establish or strengthen cooperative relationships with their mobile operator partners, app developers or other parties, thereby limiting the combined company's ability to promote its services and generate revenue. Competitors could also seek to gain market share from the combined company by reducing the prices they charge to advertisers or by introducing new technology tools for developers. Moreover, increased competition for mobile advertising space from developers could result in an increase in the portion of advertiser revenue that the combined company must pay to developers to acquire that advertising space.

The combined company's business will suffer to the extent that its developers and advertisers purchase and sell mobile advertising directly from each other or through other companies that are able to become intermediaries between developers and advertisers. For example, companies may have substantial existing platforms for developers who had previously not heavily used those platforms for mobile advertising campaigns. These companies could compete with the combined company to the extent they expand into mobile advertising. Other companies, such as large app developers with a substantial mobile advertising business, may decide to directly monetize some or all of their advertising space without utilizing the combined company's services. Other companies that offer analytics, mediation, exchange or other third party services may also become intermediaries between mobile advertisers and developers and thereby compete with the combined company. Any of these developments would make it more difficult for the combined company to sell its services and could result in increased pricing pressure, reduced profit margins, increased sales and marketing expenses or the loss of market share.

The mobile advertising market may develop more slowly than expected, which could harm the business of the combined company.

Advertisers have historically spent a smaller portion of their advertising budgets on mobile media as compared to traditional advertising methods, such as television, newspapers, radio and billboards, or online advertising over the internet, such as placing banner ads on websites. Future demand and market acceptance for mobile advertising is uncertain. Many advertisers still have limited experience with mobile advertising and may continue to devote larger portions of their advertising budgets to more traditional offline or online personal computer-based advertising, instead of shifting additional advertising resources to mobile advertising. If the market for mobile advertising deteriorates, or develops more slowly than we expect, the combined company may not be able to increase its revenue.

The combined company's mobile advertising business will be dependent on the continued growth in usage of smartphones, tablets and other mobile connected devices.

The combined company's mobile advertising business will depend on the continued proliferation of mobile connected devices, such as smartphones and tablets, that can connect to the internet over a cellular, wireless or other network, as well as the increased consumption of content through those devices. Consumer usage of these mobile connected devices may be inhibited for a number of reasons, such as:

- inadequate network infrastructure to support advanced features beyond just mobile web access;
- users' concerns about the security of these devices;
- inconsistent quality of cellular or wireless connection;
- unavailability of cost-effective, high-speed internet service; and
- changes in network carrier pricing plans that charge device users based on the amount of data consumed.

For any of these reasons, users of mobile connected devices may limit the amount of time they spend on these devices and the number of apps they download on these devices. If user adoption of mobile connected devices and consumer consumption of content on those devices do not continue to grow, the combined company's total addressable market size may be significantly limited, which could compromise the combined company's ability to increase revenue and to become profitable.

The combined company will have indebtedness, which could limit its financial flexibility.

The combined company's indebtedness could have significant negative consequences including:

- increasing the combined company's vulnerability to general adverse economic and industry conditions;
- limiting the combined company's ability to obtain additional financing;
- requiring the use of a substantial portion of any cash flow from operations to service indebtedness, thereby reducing the amount of
 cash flow available for other purposes, including capital expenditures;
- limiting the combined company's flexibility in planning for, or reacting to, changes in the combined company's business and the industry in which it competes, including by virtue of the requirement that the combined company remain in compliance with certain negative operating covenants included in the credit arrangements under which the combined company will be obligated; and
- placing the combined company at a possible competitive disadvantage to less leveraged competitors that are larger and may have better access to capital resources.

Although we currently intend to refinance the indebtedness as soon as practicable following the closing of the merger, we cannot provide any assurance that we will be successful or that we will be able to refinance the debt on acceptable terms.

The combined company would need to invest in its operations to integrate us and Appia and to maintain and grow the combined business, and may need additional funds to do so.

The combined company would depend on the availability of adequate capital to maintain and develop its business. We believe that the combined company can meet its capital requirements from internally generated funds, cash in hand, and available borrowings. If the combined company is unable to fund its capital

requirements as currently planned, however, it would have a material adverse effect on the combined company's business, financial condition, and operating results. If the combined company does not achieve our expected operating results, the combined company would need to reallocate its sources and uses of operating cash flows. This may include borrowing additional funds to service debt payments, which may impair the ability of the combined company to make investments in the business or to integrate us and Appia. There is no assurance that the combined company would be able to borrow any such additional funds when needed on commercially acceptable terms or at all.

Should the combined company need to raise funds through incurring additional debt, the combined company may become subject to covenants even more restrictive than those contained in our or Appia's current debt instruments. Furthermore, if we issue additional equity, our equity holders would suffer dilution. There can be no assurance that additional capital would be available on a timely basis, on favorable terms, or at all.

The combined company will be in a net loss position, may incur substantial net losses in the future, and may not achieve profitability.

The combined company will be in a net loss position because, prior to the merger, both Mandalay Digital and Appia accumulated significant net losses and were not profitable. If the combined company is unable to reduce costs and increase revenues, then the combined company may incur further net losses and may not be able to achieve profitability in the future. If there are delays in the distribution of the combined company's products or if it is unable to successfully negotiate with carriers or mobile operators or if these negotiations cannot occur on a timely basis, the combined company may not be able to generate revenues sufficient to meet the needs of the business in the foreseeable future or at all. If the combined company fails to achieve and maintain profitability, the value of the combined company and its common stock could decline significantly.

The combined company's ability to maintain and grow its business will be impaired if mobile connected devices, their operating systems or content distribution channels, including those controlled by the primary competitors of the combined company, develop in ways that prevent the combined company's advertising from being delivered to their users.

The combined company's business model will depend upon the continued compatibility of its mobile advertising platform with most mobile connected devices, as well as the major operating systems that run on them and the thousands of apps that are downloaded onto them.

The design of mobile devices and operating systems is controlled by third parties. These parties frequently introduce new devices, and from time to time they may introduce new operating systems or modify existing ones. Network carriers, such as AT&T, Sprint, as well as other domestic and global operators, may also affect the ability of users to download apps or access specified content on mobile devices. The combined company will have some relationship with T-Mobile and Verizon but these relationships are specific and subject to contractual performance which may not be achieved.

In some cases, the parties that control the development of mobile connected devices and operating systems include companies that the combined company would regard as its most significant competitors. For example, Apple controls two of the most popular mobile devices, the iPhone® and the iPad®, as well as the iOS operating system that runs on them. Apple also controls the App Store for downloading apps that run on Apple® mobile devices. Similarly, Google controls the Google Play and Android™ platform operating system. If the combined company's mobile advertising platform were unable to work on these devices or operating systems, either because of technological constraints or because a maker of these devices or developer of these operating systems wished to impair the combined company's ability to provide ads on them or its ability to fulfill advertising space, or inventory, from developers whose apps are distributed through their controlled channels, the combined company's ability to maintain and grow its business will be impaired.

The combined company will not control the mobile networks over which it will provide its advertising services.

The combined company's mobile advertising platform will be dependent on the reliability of network operators and carriers who maintain sophisticated and complex mobile networks, as well as its ability to deliver ads on those networks at prices that enable it to realize a profit. Mobile networks have been subject to rapid growth and technological change, particularly in recent years. The combined company will not control these networks.

Mobile networks could fail for a variety of reasons, including new technology incompatibility, the degradation of network performance under the strain of too many mobile consumers using the network, a general failure from natural disaster or a political or regulatory shut-down. Individuals and groups who develop and deploy viruses, worms and other malicious software programs could also attack mobile networks and the devices that run on those networks. Any actual or perceived security threat to mobile devices or any mobile network could lead existing and potential device users to reduce or refrain from mobile usage or reduce or refrain from responding to the services offered by the combined company's advertising clients. If the network of a mobile operator should fail for any reason, the combined company would not be able to effectively provide its services to its clients through that mobile network. This, in turn, could hurt the combined company's reputation and cause it to lose significant revenue.

Mobile carriers may also increase restrictions on the amounts or types of data that can be transmitted over their networks. The combined company anticipates generating different amounts of revenue from its advertiser clients based on the kinds of ads the combined company delivers, such as display ads, rich media ads or video ads. In most cases, the combined company will be paid by advertisers on a cost-per-install basis, when a user downloads an advertised app. In other cases, the combined company will be paid on a cost-per-thousand basis depending on the number of ads shown, or on a cost-per-click, or cost-per-action, basis depending on the actions taken by the mobile device user. Different types of ads consume differing amounts of bandwidth and network capacity. If a network carrier were to restrict the amounts of data that can be delivered on that carrier's network, or otherwise control the kinds of content that may be downloaded to a device that operates on the network, it could negatively affect the combined company's pricing practices and inhibit its ability to deliver targeted advertising to that carrier's users, both of which could impair the combined company's ability to generate revenue.

Mobile connected device users may choose not to allow advertising on their devices.

The success of the combined company's business model will depend on its ability to deliver targeted, highly relevant ads to consumers on their mobile connected devices. Targeted advertising is done primarily through analysis of data, much of which is collected on the basis of user-provided permissions. This data might include a device's location or data collected when device users view an ad or video or when they click on or otherwise engage with an ad. Users may elect not to allow data sharing for targeted advertising for a number of reasons, such as privacy concerns, or pricing mechanisms that may charge the user based upon the amount or types of data consumed on the device. Users may also elect to opt out of receiving targeted advertising from combined company's platform. In addition, the designers of mobile device operating systems are increasingly promoting features that allow device users to disable some of the functionality, which may impair or disable the delivery of ads on their devices, and device manufacturers may include these features as part of their standard device specifications. Although we are not aware of any such products that are widely used in the market today, as has occurred in the online advertising industry, companies may develop products that enable users to prevent ads from appearing on their mobile device screens. If any of these developments were to occur, the combined company's ability to deliver effective advertising campaigns on behalf of its advertiser clients would suffer, which could hurt its ability to generate revenue and become profitable.

The combined company may not be able to enhance its mobile advertising platform to keep pace with technological and market developments.

The market for mobile advertising services is characterized by rapid technological change, evolving industry standards and frequent new service introductions. To keep pace with technological developments, satisfy increasing advertiser and developer requirements, maintain the attractiveness and competitiveness of the combined company's mobile advertising solutions and ensure compatibility with evolving industry standards and protocols, the combined company will need to regularly enhance its current services and to develop and introduce new services on a timely basis. Appia has invested significant resources in building and developing real-time bidding, or RTB, infrastructure to provide access to large amounts of advertising inventory and publishers. If the combined company's RTB platform is not attractive to its customers or is not able to compete with alterative mobile advertising solutions, the combined company will not have access to as much advertising inventory and may experience increased pressure on margins.

In addition, advances in technology that allow developers to generate revenue from their apps without assistance from the combined company could harm its relationships with developers and diminish its available advertising inventory within their apps. Similarly, technological developments that allow third parties to better mediate the delivery of ads between advertisers and developers by introducing an intermediate layer between the combined company and its developers could impair its relationships with those developers. The combined company's inability, for technological, business or other reasons, to enhance, develop, introduce and deliver compelling mobile advertising services in response to changing market conditions and technologies or evolving expectations of advertisers or mobile device users could hurt its ability to grow its business and could result in its mobile advertising platform becoming obsolete.

The combined company will depend on publishers, developers and distribution partners for mobile advertising space to deliver its advertiser clients' advertising campaigns, and any decline in the supply of advertising inventory could hurt its business.

The combined company will depend on publishers, developers and distribution partners to provide it with space within their apps, which we refer to as "advertising inventory," on which the combined company will deliver ads. We anticipate that a significant portion of the combined company's revenue will derive from the advertising inventory provided by a limited number of publishers, developers and distribution partners. The combined company will have minimum or fixed commitments for advertising inventory with some but not all of its publishers, developers and distribution partners, including certain wireless carriers in the United States and internationally. The combined company intends to expand the number of publishers, developers and distribution partners subject to minimum or fixed arrangements. Outside of those relationships however, the publishers, developers and distribution partners that will sell their advertising inventory to the combined company are not required to provide any minimum amounts of advertising space to the combined company, nor are they contractually bound to provide the combined company with a consistent supply of advertising inventory. Such publishers, developers and distribution partners can change the amount of inventory they make available to the combined company at any time. They may also change the price at which they offer inventory to the combined company, or they may elect to make advertising space available to its competitors who offer ads to them on more favorable economic terms. In addition, publishers, developers and distribution partners may place significant restrictions on the combined company's use of their advertising inventory. These restrictions may prohibit ads from specific advertisers or specific industries, or they could restrict the use of specified creative content or format. They may also use a fee-based or subscription-based business model to generate revenue from their content, in lieu of or to reduce their reliance on ads.

If publishers, developers and distribution partners decide not to make advertising inventory available to the combined company for any of these reasons, decide to increase the price of inventory, or place significant restrictions on the combined company's use of their advertising space, the combined company may not be able to

replace this with inventory from others that satisfy the combined company's requirements in a timely and cost-effective manner. If this happens, the combined company's revenue could decline or its cost of acquiring inventory could increase.

The combined company's business will depend on its ability to collect and use data to deliver ads, and any limitation on the collection and use of this data could significantly diminish the value of the combined company's services and cause it to lose clients and revenue.

When the combined company delivers an ad to a mobile device, it will often be able to collect anonymous information about the placement of the ad and the interaction of the mobile device user with the ad, such as whether the user visited a landing page or installed an app. As the combined company collects and aggregates this data provided by billions of ad impressions, it intends to analyze it in order to optimize the placement and scheduling of ads across the advertising inventory provided to it by developers. For example, the combined company may use the collected information to limit the number of times a specific ad is presented to the same mobile device, to provide an ad to only certain types of mobile devices, or to provide a report to an advertiser client on the number of its ads that were clicked.

Although the data the combined company will collect is not personally identifiable information, its clients might decide not to allow it to collect some or all of this data or might limit its use of this data. For example, app developers may not agree to provide the combined company with the data generated by interactions with the content on their apps, or device users may not consent to having information about their device usage provided to the developer. Any limitation on the combined company's ability to collect data about user behavior and interaction with mobile device content could make it more difficult for the combined company to deliver effective mobile advertising programs that meet the demands of its advertiser clients.

Although the combined company's contracts with advertisers will generally permit it to aggregate data from advertising campaigns, these clients might nonetheless request that the combined company discontinue using data obtained from their campaigns that have already been aggregated with other clients' campaign data. It would be difficult, if not impossible, to comply with these requests, and responding to these kinds of requests could also cause the combined company to spend significant amounts of resources. Interruptions, failures or defects in its data collection, mining, analysis and storage systems, as well as privacy concerns and regulatory restrictions regarding the collection of data, could also limit its ability to aggregate and analyze mobile device user data from its clients' advertising campaigns. If that happens, the combined company may not be able to optimize the placement of advertising for the benefit of its advertiser clients, which could make its services less valuable, and, as a result, it may lose clients and its revenue may decline.

The combined company's business may depend in part on its ability to collect and use location-based information about mobile connected device users.

The combined company's business model will depend in part upon its ability to collect data about the location of mobile connected device users when they are interacting with their devices, and then to use that information to provide effective targeted advertising on behalf of its advertising clients. The combined company's ability to either collect or use location-based data could be restricted by a number of factors, including new laws or regulations, technology or consumer choice. Limitations on its ability to either collect or use location data could impact the effectiveness of the combined company's platform and its ability to target ads.

The combined company will not have long-term agreements with its advertiser clients, and it may be unable to retain key clients, attract new clients or replace departing clients with clients that can provide comparable revenue to the combined company.

The combined company's success will depend on its ability to maintain and expand its current advertiser client relationships and to develop new relationships. The combined company's contracts with its advertiser clients will not generally include long-term obligations requiring them to purchase the combined company's

services and are cancelable upon short or no notice and without penalty. As a result, the combined company may have limited visibility as to its future advertising revenue streams. The combined company will not be able to provide assurance that its advertiser clients will continue to use its services or that it will be able to replace, in a timely or effective manner, departing clients with new clients that generate comparable revenue. If a major advertising client representing a significant portion of the combined company's business decides to materially reduce its use of the combined company's platform or to cease using the combined company's platform altogether, it is possible that the combined company may not have a sufficient supply of ads to fill its developers' advertising inventory, in which case the combined company's revenue could be significantly reduced. Revenue derived from performance advertisers in particular is subject to fluctuation and competitive pressures. Such advertisers, which seek to drive app downloads, are less consistent with respect to their spending volume, and may decide to substantially increase or decrease their use of the combined company's platform based on seasonality or popularity of a particular app. Advertisers in general may shift their business to a competitor's platform because of new or more compelling offerings, strategic relationships, technological developments, pricing and other financial considerations, or a variety of other reasons. Any non-renewal, renegotiation, cancellation or deferral of large advertising contracts, or a number of contracts that in the aggregate account for a significant amount of revenue, could cause an immediate and significant decline in the combined company's revenue and harm its business.

The combined company's business practices with respect to data could give rise to liabilities or reputational harm as a result of governmental regulation, legal requirements or industry standards relating to consumer privacy and data protection.

In the course of providing its services, the combined company will transmit and store information related to mobile devices and the ads it places, which may include a device's geographic location for the purpose of delivering targeted location-based ads to the user of the device, with that user's consent. Federal, state and international laws and regulations govern the collection, use, retention, sharing and security of data that the combined company will collect across its mobile advertising platform. The combined company will strive to comply with all applicable laws, regulations, policies and legal obligations relating to privacy and data protection. However, it is possible that these requirements may be interpreted and applied in a manner that is inconsistent from one jurisdiction to another and may conflict with other rules or its practices. Any failure, or perceived failure, by it to comply with U.S. federal, state, or international laws, including laws and regulations regulating privacy, data security, or consumer protection, could result in proceedings or actions against the combined company by governmental entities or others. Any such proceedings could hurt the combined company's reputation, force it to spend significant amounts in defense of these proceedings, distract its management, increase its costs of doing business, adversely affect the demand for its services and ultimately result in the imposition of monetary liability. The combined company may also be contractually liable to indemnify and hold harmless its clients from the costs or consequences of inadvertent or unauthorized disclosure of data that it stores or handles as part of providing its services.

The regulatory framework for privacy issues worldwide is evolving, and various government and consumer agencies and public advocacy groups have called for new regulation and changes in industry practices, including some directed at the mobile industry in particular. For example, in early 2012, the State of California entered into an agreement with several major mobile app platforms under which the platforms have agreed to require mobile apps to meet specified standards to ensure consumer privacy. Subsequently, in January 2013, the State of California released a series of recommendations for privacy best practices for the mobile industry. In January 2014, a California law also became effective amending the required disclosures for online privacy policies. It is possible that new laws and regulations will be adopted in the United States and internationally, or existing laws and regulations may be interpreted in new ways, that would affect the combined company's business, particularly with regard to location-based services, collection or use of data to target ads, and communication with consumers via mobile devices.

The U.S. government, including the Federal Trade Commission, or FTC, and the Department of Commerce, is focused on the need for greater regulation of the collection of consumer information, including regulation aimed at restricting some targeted advertising practices. In December 2012, the FTC adopted revisions to the Children's Online Privacy Protection Act, or COPPA, that went into effect on July 1, 2013. COPPA imposes a number of obligations on operators of websites and online services including mobile apps, such as obtaining parental consent, if the operator collects specified information from users and either the site or service is directed to children under 13 years old or the site or service knows that a specific user is a child under 13 years old. The changes broaden the applicability of COPPA, including the types of information that are subject to these regulations, and may apply to information that the combined company will collect through mobile devices or apps that, prior to the adoption of these new regulations, was not subject to COPPA. These revisions will impose new compliance burdens on the combined company. In February 2013, the FTC issued a staff report containing recommendations for best practices with respect to consumer privacy for the mobile industry. To the extent that the combined company or its clients choose to adopt these recommendations, or other regulatory or industry requirements become applicable to the combined company, it may have greater compliance burdens.

As the combined company expands its operations globally, compliance with regulations that differ from country to country may also impose substantial burdens on its business. In particular, the European Union has traditionally taken a broader view as to what is considered personal information and has imposed greater obligations under data privacy regulations. In addition, individual EU member countries have had discretion with respect to their interpretation and implementation of the regulations, which has resulted in variation of privacy standards from country to country. Complying with any new regulatory requirements could force it to incur substantial costs or require us to change its business practices in a manner that could compromise its ability to effectively pursue its growth strategy.

The combined company's business may involve the use, transmission and storage of confidential information, and the failure to properly safeguard such information could result in significant reputational harm and monetary damages.

The combined company may at times collect, store and transmit information of, or on behalf of, its clients that may include certain types of confidential information that may be considered personal or sensitive, and that are subject to laws that apply to data breaches. The combined company intends to take reasonable steps to protect the security, integrity and confidentiality of the information it collects and stores, but there is no guarantee that inadvertent or unauthorized disclosure will not occur or that third parties will not gain unauthorized access to this information despite the combined company's efforts to protect this information. If such unauthorized disclosure or access does occur, the combined company may be required to notify persons whose information was disclosed or accessed. Most states have enacted data breach notification laws and, in addition to federal laws that apply to certain types of information, such as financial information, federal legislation has been proposed that would establish broader federal obligations with respect to data breaches. The combined company may also be subject to claims of breach of contract for such disclosure, investigation and penalties by regulatory authorities and potential claims by persons whose information was disclosed. The unauthorized disclosure of information may result in the termination of one or more of its commercial relationships or a reduction in client confidence and usage of its services. The combined company may also be subject to litigation alleging the improper use, transmission or storage of confidential information, which could damage its reputation among its current and potential clients, require significant expenditures of capital and other resources and cause it to lose business and revenue.

As a result of the merger, we and Appia, as a combined company, would be a substantially larger and broader organization, with a greater geographic diversity relative to our and Appia's current operations, and if management is unable to sufficiently manage the combined company, operating and financial results would suffer.

As a result of the merger, the combined company would have more employees, greater geographic diversity, and customers in multiple distribution channels. The combined company would face challenges

inherent in efficiently managing an increased number of employees over large geographic distances, including the need to implement appropriate policies, benefits, reporting, management, and compliance programs and systems. The inability to manage successfully the larger and internationally diverse organization, or any significant delay in achieving successful management of the organization, could have a material adverse effect on the combined company and, as a result, on the market price of our common stock.

Changes to current accounting principles could have a significant effect on the combined company's reported financial results or the way in which it conducts its business.

We prepare our financial statements in conformity with U.S. GAAP, which are subject to interpretation by the Financial Accounting Standards Board, the American Institute of Certified Public Accountants, the SEC, and various other authorities formed to interpret, recommend, and announce appropriate accounting principles, policies, and practices. A change in these principles could have a significant effect on our reported financial results and may even retroactively affect the accounting for previously reported transactions. Our accounting policies that recently have been or may in the future be affected by changes in the accounting principles are as follows:

- stock-based compensation;
- accounting for uncertain tax positions;
- accounting for goodwill and other intangible assets; and
- accounting issues related to certain features of contingent convertible debt instruments and their effect on diluted earnings per share.

Changes in these or other rules may have a significant adverse effect on our reported financial results or in the way in which we conduct our business. See the discussion in our Quarterly Report on Form 10-Q for the fiscal quarter ended September 30, 2014, which is incorporated by reference into this proxy statement/prospectus, under "Summary of Significant Accounting Policies" and the Notes to Unaudited Consolidated Financial Statements included therein, for additional information about our accounting policies and estimates and associated risks.

Loss or reduction of business from the combined company's large advertiser clients could have a significant impact on the combined company's revenues, results of operations and overall financial condition.

From time to time, a limited number of the combined company's advertiser clients will be expected to account for a significant share of its advertising revenue. This customer concentration increases the risk of quarterly fluctuations in the combined company's revenues and operating results. The combined company's advertiser clients may reduce or terminate their business with it at any time for any reason, including changes in their financial condition or other business circumstances. If a large advertising client representing a substantial portion of its business decided to materially reduce or discontinue its use of its platform, it could cause an immediate and significant decline in its revenue and negatively affect its results of operations and financial condition.

The combined company's customer concentration also increases the concentration of its accounts receivable and its exposure to payment defaults by key customers. The combined company will generate significant accounts receivable for the services that it provides to its key advertiser clients, which could expose it to substantial and potentially unrecoverable costs if it does not receive payment from them.

The combined company's business will be harmed if it cannot increase the capacity of its mobile advertising technology platform to meet advertiser or device user demand.

The combined company must be able to continue to increase the capacity of its technology platforms in order to support substantial increases in the number of advertisers and device users, to support an increasing variety of advertising formats and to maintain a stable service infrastructure and reliable service delivery for its mobile advertising campaigns. If the combined company is unable to efficiently and effectively increase the scale of its mobile advertising platforms to support and manage a substantial increase in the number of advertisers and mobile device users, while also maintaining a high level of performance, the quality of its services could decline and its reputation and business could be seriously harmed. In addition, if the combined company is not able to support emerging mobile advertising formats or services preferred by advertisers, it may be unable to obtain new advertising clients or may lose existing advertising clients, and in either case its revenue could decline.

If the combined company fails to detect click fraud or other invalid clicks on ads, it could lose the confidence of its advertiser clients, which would cause its business to suffer.

The combined company's business will rely on delivering positive results to its advertiser clients. The combined company will be exposed to the risk of fraudulent and other invalid clicks or conversions that advertisers may perceive as undesirable. Because of their smaller sizes as compared to personal computers, mobile device usage could result in a higher rate of accidental or otherwise inadvertent clicks by a user. Invalid clicks could also result from click fraud, where a mobile device user intentionally clicks on ads for reasons other than to access the underlying content of the ads. If fraudulent or other malicious activity is perpetrated by others, and the combined company is unable to detect and prevent it, the affected advertisers may experience or perceive a reduced return on their investment. High levels of invalid click activity could lead to dissatisfaction with its advertising services, refusals to pay, refund demands or withdrawal of future business. Any of these occurrences could damage the combined company's brand and lead to a loss of advertisers and revenue.

System failures could significantly disrupt the combined company's operations and cause it to lose advertiser clients or advertising inventory.

The combined company's success will depend on the continuing and uninterrupted performance of its own internal systems, which the combined company will utilize to place ads, monitor the performance of advertising campaigns and manage its inventory of advertising space. Its revenue will depend on the technological ability of its platforms to deliver ads. Sustained or repeated system failures that interrupt its ability to provide services to clients, including technological failures affecting its ability to deliver ads quickly and accurately and to process mobile device users' responses to ads, could significantly reduce the attractiveness of its services to advertisers and reduce its revenue. The combined systems are vulnerable to damage from a variety of sources, including telecommunications failures, power outages, malicious human acts and natural disasters. In addition, any steps the combined company takes to increase the reliability and redundancy of its systems may be expensive and may not ultimately be successful in preventing system failures.

Activities of the combined company's advertiser clients could damage the combined company's reputation or give rise to legal claims against it.

The combined company's advertiser clients' promotion of their products and services may not comply with federal, state and local laws, including, but not limited to, laws and regulations relating to mobile communications. Failure of its clients to comply with federal, state or local laws or its policies could damage its reputation and expose it to liability under these laws. The combined company may also be liable to third parties for content in the ads it delivers if the artwork, text or other content involved violates copyrights, trademarks or other intellectual property rights of third parties or if the content is defamatory, unfair and deceptive, or otherwise in violation of applicable laws. Although the combined company will generally receive assurance from its advertisers that their ads are lawful and that they have the right to use any copyrights, trademarks or other

intellectual property included in an ad, and although it will normally be indemnified by the advertisers, a third party or regulatory authority may still file a claim against the combined company. Any such claims could be costly and time-consuming to defend and could also hurt the combined company's reputation. Further, if it is exposed to legal liability as a result of the activities of its advertiser clients, the combined company could be required to pay substantial fines or penalties, redesign its business methods, discontinue some of its services or otherwise expend significant resources.

The combined company's business depends on its ability to maintain the quality of its advertiser and developer content.

The combined company must be able to ensure that its clients' ads are not placed in developer content that is unlawful or inappropriate. Likewise, its developers will rely upon the combined company not to place ads in their apps that are unlawful or inappropriate. If the combined company is unable to ensure that the quality of its advertiser and developer content does not decline as the number of advertisers and developers it works with continues to grow, then the combined company's reputation and business may suffer.

Other Risks Factors of Mandalay

In addition to the foregoing risks, Mandalay is, and will continue to be, subject to the risks described in its Annual Report on Form 10-K for the year ended March 31, 2014, and as updated by subsequent Quarterly Reports on Form 10-Q and any amendments thereto, and Current Reports on Form 8-K, all of which are incorporated by reference into this proxy statement/prospectus. See the sections "Where You Can Find More Information" on page 153 and "Incorporation of Certain Information by Reference" on page 154.

CAUTIONARY STATEMENT REGARDING FORWARD-LOOKING INFORMATION

The SEC encourages companies to disclose forward-looking information so that investors can better understand a company's future prospects and make informed investment decisions. These forward-looking statements include all matters that are not historical facts. This proxy statement/prospectus and documents incorporated by reference contain these types of statements. Words such as "anticipates," "estimates," "expects," "projects," "intends," "plans," "believes," "may," or "should" and words or terms of similar substance used in connection with any discussion of future operating results or financial performance identify forward-looking statements. Also, as examples, forward-looking statements may include statements concerning future results from operations, financial position, economic conditions, product releases, revenue and product synergies, cost savings, product or competitive enhancements and any other statement that may be construed as a prediction of future performance or events, including that Appia's technology will enhance Mandalay Digital's existing products or foster new technology innovation, perceived benefits from the business combination to the combined company, or that the acquisition will result in increased revenue, cost savings and better competitive position, or that Mandalay Digital will successfully integrate Appia's technology. These forward-looking statements involve certain known and unknown risks and uncertainties. Factors that could cause actual results to differ materially from those contemplated by the forward-looking statements include, but are not limited to, the following factors:

- the occurrence of any event, change or other circumstances that could give rise to the termination of the merger agreement;
- the inability to complete the merger within the expected time period or at all, including due to the failure to obtain stockholder approval, or the failure to satisfy other conditions to completion of the merger;
- risks related to disruption of management's attention from the ongoing business operations due to the proposed merger; the effect of the announcement of the proposed merger on the Mandalay's or Appia's relationships with their respective customers, lenders, operating results and businesses generally;
- material adverse changes in Mandalay Digital's or Appia's operations or financial results prior to closing;
- the ability to expand the combined company's global reach, accelerate growth and enhance our scalable, low-capex business model that drives EBITDA;
- failure to realize anticipated operational efficiencies, revenue (including projected revenue) and cost synergies and resulting revenue growth, EBITDA and free cash flow conversion if the merger is consummated;
- the ability to achieve internal strategic forecasts;
- inability to refinance the assumed Appia debt subsequent to the closing or to refinance the debt on favorable terms;
- unforeseen challenges related to relationships with operators, publishers and advertisers and expanding and maintaining those relationships;
- the ability to execute upon, and realize any benefits from, potential value creation opportunities through strategic relationships in the future or at all, including the ability to leverage advertising opportunities effectively and increase revenue streams for carriers;
- unforeseen difficulties preventing rapid integration of Appia's app-install infrastructure into Digital Turbine's existing platform;
- the inherent and deal specific challenges in converting discussions with carriers into actual contractual relationships;
- Mandalay Digital's ability as a smaller company to manage international, and as a result of the proposed merger, larger operations, varying and often unpredictable levels of orders, and the challenges inherent in technology development necessary to maintain a competitive advantage; and
- the potential for unforeseen or underestimated cash requirements necessary to enable the transaction synergies to be realized.

See also the section entitled "Risk Factors" in this proxy statement/prospectus and the risk factors disclosed in our Annual Report on Form 10-K and Quarterly Reports on Form 10-Q. These risks and uncertainties are not exhaustive. Other sections of this proxy statement/prospectus describe additional factors that could adversely impact the combined company's business and financial performance. Moreover, the combined company will operate in a very competitive and rapidly changing environment. New risks and uncertainties emerge from time to time, and it is not possible to predict all risks and uncertainties, nor can we assess the impact that these factors will have on the combined company's business or the extent to which any factor, or combination of factors, may cause actual results to differ materially from those contained in any forward-looking statement.

Most of these factors are difficult to predict accurately and are generally beyond our control. You should consider the uncertainty and any risk resulting from such uncertainty in connection with any forward-looking statement that may be made herein. You should not place undue reliance on these forward-looking statements, which speak only as of the date of this proxy statement/prospectus in the case of forward-looking statements contained in this proxy statement/prospectus, or the dates of the documents incorporated by reference into this proxy statement/prospectus in the case of forward-looking statements made in those incorporated documents. You should carefully review this proxy statement/prospectus in its entirety, including, but not limited to, our "Management's Discussion and Analysis of Financial Condition and Results of Operations" and the financial statements and the accompanying notes thereto, which are included in our Annual Report on Form 10-K for the fiscal year ended March 31, 2014, and incorporated by reference herein, and the risks described in the section entitled "Risk Factors" in this proxy statement/prospectus. You should also review the description of Appia's business in the sections entitled "Information Regarding Appia's Operations" and "Management's Discussion and Analysis of Financial Condition and Results of Operations of the Business of Appia," and the financial statements and the accompanying notes thereto in this proxy statement/prospectus. Except for our ongoing obligation to disclose material information under U.S. federal securities laws, we undertake no obligation to release publicly any revisions to any forward-looking statements, to report events, or to report the occurrence of unanticipated events.

We expressly qualify in their entirety all forward-looking statements attributable to us, Appia or the combined company, or any person acting on their behalf, by the cautionary statements contained or referred to in this section.

ANNUAL MEETING OF MANDALAY DIGITAL STOCKHOLDERS

General

We are furnishing this proxy statement/prospectus to holders of our common and preferred stock in connection with the solicitation of proxies by our board of directors for use at the annual meeting of stockholders to be held on [\square], 2015 and at any adjournment, postponement, or continuation thereof. This proxy statement/prospectus is first being furnished to our stockholders on or about [\square], 2015. In addition, this proxy statement/prospectus is being mailed to the Appia stockholders as a prospectus of Mandalay Digital in connection with the issuance by us of shares of our common stock to Appia in connection with the merger described in this proxy statement/prospectus.

Important Notice Regarding the Availability of Proxy Materials for the Stockholders Meeting To Be Held on [0], 2015

We are providing you access to this proxy statement, the accompanying form of proxy card and our annual report on Form 10-K and Form 10-K/A for the fiscal year ended March 31, 2014 both by sending you these proxy materials and by notifying you of the availability of these proxy materials on the Internet. These proxy materials are available on the Internet at www.proxyvote.com.

Date, Time, and Place

The annual meeting of our stockholders will be held on [\square], 2015 at [\square] a.m., Pacific time, at the offices of Manatt, Phelps & Phillips, LLP, 11355 W. Olympic Boulevard, Los Angeles, CA 90064.

Purpose of the Mandalay Digital Annual Meeting

At the annual meeting of our stockholders, and at any adjournment, postponement, or continuation thereof, our stockholders will be asked to consider and vote upon the following proposals:

- (1) To consider and vote upon the issuance and sale of (a) up to 19,500,000 shares of our common stock pursuant to the merger agreement with Appia, Inc. (including shares of common stock issuable upon exercise of stock options to be assumed by Mandalay Digital pursuant to the merger agreement and 1,000,000 shares to be deposited into escrow) and, (b) an additional 600,000 shares of our common stock (including a warrant to purchase 400,000 shares, which vests 12 months after issuance under certain circumstances) to be issued to a lender in connection with a separate loan transaction, all of which aggregates to more than 20% of Mandalay Digital's common stock outstanding under applicable listing rules of The NASDAQ Stock Market LLC;
- (2) To elect six directors to serve on our board of directors for a one-year term ending as of our annual meeting of stockholders in 2016.
- (3) To approve, in a non-binding advisory vote, the compensation of our named executive officers, commonly referred to as "Say-on-pay".
- (4) To hold an advisory, non-binding vote on the frequency of future advisory votes on the compensation of our named executive officers, commonly referred to as "Say-on-frequency".
- (5) To ratify the appointment of SingerLewak as our independent registered public accounting firm for the fiscal year ending March 31, 2015.
 - (6) Transact such other business as may properly come before the meeting and/or any adjournment or postponement thereof.

Board Recommendation

Our board of directors has approved and we have executed the merger agreement, and our board of directors has approved the issuance of shares of our common stock in the merger and loan transaction, and recommends that our stockholders vote "FOR" approval of the issuance of shares of our common stock in the merger pursuant to the merger agreement and in the separate loan transaction, the election of the six director nominees and "FOR" the other proposal described in this proxy statement/ prospectus.

Record Date, Outstanding Shares, and Quorum

Only holders of records of our common stock and our preferred stock at the close of business on [\square], 2015 (the "Record Date") are entitled to notice of and to vote at our annual meeting and any adjournments thereof. As of the Record Date, [\square] shares of our common stock and 100,000 shares of preferred stock, which are convertible into 20,000 shares of common stock, were issued and outstanding.

Holders of our common stock are entitled to one vote at the annual meeting for each share of common stock held that was issued and outstanding as of the Record Date. The Series A preferred stock is entitled to vote together with the common stock as a single class (on an asconverted to common stock basis) on any matters submitted to the holders of the Company's common stock, together with any other voting rights provided to the Series A Preferred under law or the General Corporation Law of the State of Delaware.

The presence, in person or by proxy, of stockholders holding at least a majority of our outstanding common stock and preferred stock, voting together as a class, will constitute a quorum for the transaction of business at our annual meeting. If a quorum is not present at the annual meeting, we expect that the meeting will be adjourned or postponed to solicit additional proxies. Your shares will be counted towards the quorum only if you submit a valid proxy or vote at the annual meeting.

How to Vote

Voting in Person

All stockholders as of the close of business on [I], 2015 can attend the annual meeting. If you plan to attend the annual meeting and wish to vote in person, you will be given a ballot at the annual meeting. You may vote at the annual meeting if you are a stockholder of record (your shares are directly registered in your name on our books and not held through a broker, bank or other nominee). In order to vote at the annual meeting shares held in "street name," you must obtain a valid proxy from your broker, bank, or other nominee, and bring it to the meeting. Follow the instructions from your broker, bank, or other nominee included with these proxy materials, or contact your broker, bank, or other nominee to request a proxy form.

Voting by Proxy

We request that our stockholders complete, date, and sign the accompanying proxy and promptly return it in the accompanying envelope or otherwise mail it to us. All properly executed proxies that we receive prior to the vote at the annual meeting (that have not been revoked) will be voted in accordance with the instructions indicated on the proxies. All properly executed proxies that we receive prior to the vote at the annual meeting that do not provide any direction as to how to vote will be voted in accordance with the recommendations of our board of directors.

If your shares are held in street name, you will need to obtain a proxy form from the institution that holds your shares and follow the instructions included on that form regarding how to instruct your broker to vote your shares.

Revocation of Proxies

You can revoke your proxy at any time before your shares are voted at the annual meeting by (i) sending a written notice of revocation to Corporate Secretary, Mandalay Digital Group, Inc., 2811 Cahuenga Boulevard West, Los Angeles, CA 90068, (ii) submitting by mail, telephone or the Internet another proxy dated as of a later date, or (iii) voting in person at the annual meeting. Merely attending the annual meeting will not revoke your proxy. All revocations of your proxy must be received prior to the voting of your shares at the annual meeting. Voting in person at the annual meeting will replace any previous votes submitted by proxy. If the annual meeting were to be postponed or adjourned, your proxy would still be valid and will be voted at the postponed or adjourned meeting. You would still be able to revoke your proxy until it was voted.

Voting Your Shares

If you complete and submit your proxy voting instructions, the persons named as proxies will follow your instructions. If you submit your proxy voting instructions but do not direct how to vote on each item, the persons named as proxies will vote your shares

- **FOR** authorization of the issuance of the shares of Mandalay Digital common stock issuable pursuant to the merger and the loan transaction (*Proposal 1*),
- **FOR** the director nominees (*Proposal 2*),
- **FOR** the advisory Say-on-pay proposal (*Proposal 3*),
- **FOR** the annual (one-year) alternative in the Say-on-frequency proposal (*Proposal 4*),
- **FOR** the ratification of SingerLewak as our independent registered public accounting firm for the fiscal year ending March 31, 2015 (*Proposal 5*).

The persons named as proxies will vote on any other matters properly presented at the annual meeting in accordance with their best judgment. We have not received notice of other matters that may be properly presented for voting at the annual meeting.

Votes Required

The affirmative vote of a majority of the votes cast by our common and preferred stock present, in person or by proxy, voting as a single class on an as-converted to common stock basis, at the annual meeting is required to approve the issuance of the shares of Mandalay Digital common stock pursuant to the merger agreement and the separate loan transaction (*Proposal 1*). The directors will be elected by a plurality of the voting power of our common stock and preferred stock, voting together as a single class on an as-converted to common stock basis, at the annual meeting (*Proposal 2*). The affirmative vote of a majority of the voting power of our common and preferred stock present, in person or by proxy, voting as a single class on an as-converted to common stock basis, at the annual meeting is required to approve the advisory Say-on-pay and Say-on-frequency proposals, and to ratify the appointment of SingerLewak as our independent registered public accounting firm for the fiscal year ending March 31, 2015 (*Proposals 3, 4 and 5, respectively*).

Withholding Your Vote, Abstentions and Broker non-votes

In the election of the directors (*Proposal* 2), you can withhold your vote for any or all of the nominees. Withheld votes will be excluded entirely from the vote and will have no effect on the outcome. With regard to all other proposals, you can vote to "abstain." If you vote to "abstain," your shares will be counted as present at the meeting for purposes of determining whether a quorum exists, but such abstention will have the effect of a vote against the proposal.

A "broker non-vote" occurs when a broker submits a proxy without voting on one or more of the non-routine matters. A bank, brokerage firm or other nominee is not permitted by applicable regulatory requirements

to vote on non-routine matters without instructions from the beneficial owner of the shares. The share issuance proposal, the election of directors and the advisory Say-on-pay and Say-on frequency proposals are deemed non-routine matters; consequently, absent instructions from you, your broker may not vote your shares on the share issuance proposal, the election of the directors and the advisory Say-on-pay and Say-on-frequency proposals (*Proposals 1, 2, 3 and 4*). Your broker may vote your shares on the ratification of the appointment of our independent registered public accounting firm (*Proposals 5*). A broker non-vote is considered present at the meeting for purposes of determining whether a quorum exists but is not counted as a vote cast on any non-routine matter presented at the annual meeting.

Proxy Solicitation

We will bear the entire cost of solicitation of proxies from our stockholders, including preparation, assembly, printing, and mailing of this proxy statement/prospectus and the proxy card. Copies of solicitation materials will be furnished to banks, brokerage houses, fiduciaries, and custodians holding in their names shares of our common stock or preferred stock beneficially owned by others to forward to such beneficial owners. We may reimburse persons representing beneficial owners of our common stock for their costs of forwarding solicitation materials to such beneficial owners. In addition to solicitation by use of the mails, proxies may be solicited by directors, officers, employees, or agents of our company in person or by telephone or other means of communication. No additional compensation will be paid to directors, officers, or other regular employees of ours for such services.

We have retained MacKenzie Partners to aid in the solicitation of proxies from banks, brokers, nominees, and intermediaries. We will pay MacKenzie a fee of \$15,000 for its services, plus we will reimburse MacKenzie for various out-of-pocket and other expenses.

Other Business; Adjournments

We do not expect that any matter other than the proposals presented in this proxy statement/prospectus will be brought before our annual meeting. However, if other matters incident to the conduct of the annual meeting are properly presented at the annual meeting or any adjournment, postponement, or continuation of the annual meeting, the persons named as proxies will vote in accordance with their best judgment with respect to those matters.

Adjournments may be made for the purpose of, among other things, soliciting additional proxies.

SECURITY OWNERSHIP OF CERTAIN BENEFICIAL OWNERS AND MANAGEMENT

As of November 13, 2014, there were approximately 37,797,221 shares of our common stock and 100,000 shares of preferred stock outstanding. The following table presents information regarding the beneficial ownership of our common stock, including shares of common stock issuable upon conversion of our preferred stock, as of such date by:

- each person who beneficially owns more than 5% of the outstanding shares of our common stock;
- each director;
- each person who was a named executive officer for fiscal 2014;
- · each additional person who is currently an executive officer; and
- all current directors and executive officers as a group.

Beneficial ownership is determined in accordance with the rules of the SEC. These rules generally provide that a person is the "beneficial owner" of securities if such person has or shares the power to vote or direct the voting of securities, or to dispose or direct the disposition of securities or has the right to acquire such powers within 60 days. In computing the number of shares beneficially owned by a person and the percentage ownership of that person, shares of common stock subject to options, warrants or rights held by that person that are currently exercisable or convertible or exercisable, convertible or issuable within 60 days of November 13, 2014, are deemed outstanding. Such shares, however, are not deemed outstanding for the purpose of computing the percentage ownership of any other person. The percentages in the table have been rounded to the nearest hundredth. Except as specifically indicated in the footnotes to this table, we believe that the persons identified in this table have sole voting and investment power with respect to all shares of common stock shown as beneficially owned by them, subject to community property laws where applicable.

Number of Shares

Doroontogo

Name of Beneficial Owner (1)	Number of Shares Beneficially Owned	Owned (%)
Adage Capital Partners LP (2)	2,913,891	7.7%
Columbus Capital Management, LLC (3)	2,065,314	5.5%
Robert S. Ellin (4)	2,047,010	5.4%
Directors (excluding Executive Officers)		
Peter Guber (5)	4,094,153	10.6%
Paul Schaeffer (6)	478,454	1.3%
Robert Deutschman (7)	174,079	*
Christopher Rogers (8)	105,121	*
Jeffrey Karish (9)	88,454	*
Named Executive Officers for Fiscal 2014		
Peter Adderton (10)	2,092,500	5.5%
William Stone (11)	390,832	*
Lisa Lucero (12)	51,952	*
Jeffrey Klausner (13)	50,000	*
Additional Current Executive Officers		
Andrew Schleimer (14)	260,000	*
Kirstie Brown (15)	97,317	*
All Current Directors and Executive Officers as a Group		
(9 individuals) (16)	7,780,910	19.9%

^{*} Represents less than 1%.

⁽¹⁾ Except as otherwise indicated, the address of each of the persons listed above is c/o Mandalay Digital Group, Inc., 2811 Cahuenga Boulevard West, Los Angeles, CA 90068.

- (2) Consists of (i) 2,413,891 shares of common stock, and (ii) 500,000 shares of common stock issuable upon exercise of warrants at an exercise price of \$3.50 per share. Based on a Schedule 13G/A filed with the SEC on March 14, 2012, Adage Capital Partners GP, L.L.C. is the general partner of Adage Capital Partners, L.P., and as such has discretion over the portfolio securities beneficially owned by the Adage Capital Partners, L.P., Adage Capital Advisors, L.L.C., is the managing member of Adage Capital Partners GP, L.L.C., and Robert Atchinson and Phillip Gross are the managing members of Adage Capital Partners GP, L.L.C. and Adage Capital Advisors, L.L.C. and general partners of Adage Capital Partners, L.P., all of which share voting and dispositive power over the shares. The address for the reporting stockholders is 200 Clarendon Street, 52nd Floor, Boston, MA 02116.
- (3) Based on a Schedule 13G filed with the SEC on November 12, 2014, Columbus Capital Management, LLC ("CCM") and Matthew D. Ockner are the beneficial owners of 2,065,314 shares of common stock, which consist of (i) 1,102,600 shares held for the account of Columbus Capital Partners, L.P. ("CCP"); and (ii) 962,714 shares held for the account of Columbus Capital QP Partners, L.P. ("CCQP"). CCM is the general partner to CCP and CCQP. Mr. Ockner is the managing member of CCM. In such capacities, CCM and Mr. Ockner may be deemed to have voting and dispositive power over the shares held for the account of CCP and CCQP. The address of CCM and Mr. Ockner is 1 Market Street, Spear Tower, Suite 3790, San Francisco, CA 94105.
- Based on a Schedule 13D/A filed with the SEC July 7, 2014 and information known by the Company, (i) Trinad Capital Master Fund, Ltd. ("Trinad Master") is the beneficial owner of 1,664,228 shares of common stock and 35,714 shares of common stock issuable upon exercise of warrants at an exercise price of \$3.50 per share; (ii) Trinad Management LLC ("Trinad LLC") (a) is the beneficial owner of 20,000 shares of common stock issuable upon conversion of 100,000 shares of Series A Preferred Stock, assuming a conversion price \$5.00 per share, and (b) as the investment manager of Trinad Master, may be deemed to be the beneficial owner of the 1,664,228 shares of common stock owned by Trinad Master; (iii) Trinad Capital LP ("Trinad LP"), as the owner of 89.5% of the shares of Trinad Master, and Trinad Advisors II, LLC ("Trinad GP"), as the general partner of Trinad LP, each may be deemed to be the beneficial owner of a pro rata portion of the 1,664,228 shares of common stock held by Trinad Master; (iv) Robert S. Ellin (a) is the beneficial owner of 226,667 shares of common stock, (b) as the managing member of Trinad LLC, may be deemed to be the beneficial owner of the 20,000 shares of common stock issuable upon conversion of 100,000 shares of Series A Preferred Stock held by Trinad LLC, and (c) as portfolio manager for Trinad Master, may be deemed to be the beneficial owner of the 1,664,228 shares of common stock held by Trinad Master. Mr. Ellin also holds options to purchase 100,000 shares of common stock. The address of Mr. Ellin is 620 N. Beverly Drive Beverly Hills, CA 90210. With respect to all of the shares that are directly or indirectly held by Trinad Master, Trinad LLC, Trinad LP and Trinad GP, Mr. Ellin has the sole power to vote and dispose or direct the disposition of the shares. Mr. Ellin disclaims beneficial ownership of the shares of common stock reported above, except to the extent that any such shares are held in his name.
- (5) The Guber Family Trust (the "Trust"), of which Peter Guber serves as a trustee, is the beneficial owner of: (i) 3,091,722 shares of common stock, (ii) 35,714 shares of common stock issuable upon exercise of warrants at an exercise price of \$3.50 per share, and (iii) 666,667 shares of common stock issuable upon exercise of warrants at an exercise price of \$1.25 per share. Mr. Guber disclaims beneficial ownership of the shares of common stock directly and beneficially owned by the Trust, except to the extent of his pecuniary interest therein. Mr. Guber directly beneficially owns 200,000 shares of common stock and options to purchase 100,000 shares of common stock at an exercise price of \$13.75 per share. After November 13, 2014, Mr. Guber purchased 28,571 shares of common stock.
- (6) Consists of (i) 190,000 shares held by Mr. Schaeffer, (ii) 228,454 shares held by the Paul and Judy Schaeffer Living Trust for which Mr. Schaeffer serves as a trustee, and (iii) options to purchase 60,000 shares of common stock at an exercise price of \$13.75 per share. Mr. Schaeffer disclaims beneficial ownership of the securities held by the trust except to the extent of his pecuniary interest therein. After November 13, 2014, Mr. Schaeffer purchased 10,000 shares of common stock.

- (7) Includes 134,079 shares held by the Robert and Ellen Deutschman Family Trust. Mr. Deutschman disclaims beneficial ownership of such shares except to the extent of his pecuniary interest therein. After November 13, 2014, Mr. Deutschman purchased 30,000 shares of common stock.
- (8) After November 13, 2014, Mr. Rogers purchased 7,142 shares of common stock.
- (9) After November 13, 2014, Mr. Karish purchased 5,000 shares of common stock.
- (10) Consists of (i) 1,812,500 shares of common stock, (ii) 30,000 shares of common stock issuable upon exercise of warrants at an exercise price of \$1.95 per share, and (iii) options to purchase 250,000 shares of our common stock exercisable at an exercise price of \$2.65 per share.
- (11) Includes options to purchase 108,333 shares of common stock at an exercise price of \$2.54 per share.
- (12) Includes options to purchase 1,952 shares of common stock at an exercise price of \$4.05 per share. Ms. Lucero is not currently deemed an executive officer.
- (13) Based on information as of July 8, 2014, consists of options to purchase 50,000 shares of common stock at an exercise price of \$2.54. Mr. Klausner resigned on July 8, 2014 and is included in this table pursuant to rules of the SEC.
- (14) Includes options to purchase 60,000 shares of common stock at an exercise price of \$2.75 per share.
- (15) Includes options to purchase 41,666 and 20,834 shares of common stock at an exercise price of \$4.50 and \$2.83 per share, respectively.
- (16) Includes shares underlying options and warrants held by directors and executive officers included in the group.

PROPOSAL 1

APPROVAL OF THE ISSUANCE OF SHARES OF MANDALAY DIGITAL COMMON STOCK

We are seeking the approval of our stockholders, as required by NASDAQ Marketplace Rule 5635(a)(1), for the issuance of up to 19,500,000 shares pursuant to the merger agreement and an additional 600,000 shares of our common stock (including a warrant to purchase 400,000 shares, which vests 12 months after issuance under certain circumstances) in a separate loan transaction as described in this proxy statement/prospectus. Approval of our stockholders of the issuance of those shares is a condition to the obligations of our company and Appia to effecting the merger. If the proposal to issue up to 19,500,000 shares of our common stock is not approved, we cannot effect the merger under the terms set forth in the merger agreement.

Nasdaq Requirements

Our common stock is listed for trading on the NASDAQ Stock Market. The Marketplace Rules of the NASDAQ Stock Market require that we obtain the approval of holders of our common and preferred stock prior to the issuance of additional shares of our common stock in any transaction if:

- 1. the common stock has, or will have upon issuance, voting power equal to or in excess of 20% of the voting power outstanding before the issuance of such stock or of securities convertible into or exercisable for common stock; or
- 2. the number of shares of common stock to be issued is, or will be, equal to or in excess of 20% of the number of shares of common stock outstanding before the issuance of the common stock or securities.

As of the Record Date, there were [I] shares of our common stock outstanding and 100,000 shares of preferred outstanding.

Pursuant to the merger agreement and in connection with the merger, we will issue up to 19,500,000 shares of our common stock. The merger agreement does not provide for an adjustment in the aggregate number of shares of Mandalay Digital common stock or options to be issued by Mandalay Digital to Appia's security holders in the acquisition in the event of fluctuations in the market value of Mandalay Digital's common stock up through the closing date. However, the number of shares to be issued by Mandalay Digital is subject to adjustment based on Appia's working capital and net indebtedness as of the closing date of the merger. We currently estimate that, if the merger closes on January 31, 2015, we would issue 19,056,392 shares; additionally, we would assume options for 277,376 shares of our common stock, for an aggregate of 19,333,768 shares, representing approximately 33% of our common stock outstanding after giving effect to the merger, based on the number of shares of our common stock outstanding on November 13, 2014. Appia's actual results of operations between the date of the financial information included in this proxy statement/prospectus and the closing date may differ from the forecasted results used in this estimate, resulting in differences between the forecast and actual amounts of Appia's working capital and net indebtedness as of the closing date of the merger. See "The Merger – Consideration Payable in the Merger" located on page 68 of this proxy statement/prospectus for further information and a discussion of the assumptions in calculating the potential amount of shares to be issued by Mandalay Digital in the merger.

We will also issue, in a separate private placement, 200,000 shares of common stock plus a common stock purchase warrant exercisable for 400,000 shares of common stock to a lender pursuant to financing arrangements. See "Ancillary Agreements – The Financing Agreements" on page 97 for further information about assumption of debt and issuance of guarantees in connection with the financing arrangements.

Impact of Issuance on Existing Stockholders

Existing holders of our common stock will have rights that are equal to those of the holders of the shares of our common stock that would be issued in the merger, as the shares to be issued in the merger are the same class of common stock as the class of our common stock currently outstanding. In determining whether to vote for this proposal, stockholders should consider that they are subject to the risk of substantial dilution of their interests that will result from the issuance of shares of our common stock, and that as a result of the issuance of such shares of our common stock, the current stockholders will own a smaller percentage of our outstanding shares of common stock. The Appia stockholders would hold approximately 33% of our common stock outstanding after giving effect to the merger and assuming it were to close on January 31, 2015, based on the number of shares of our common stock outstanding on November 13, 2014.

Stockholder Vote Requirement

Approval of the issuance of the Mandalay Digital common stock in connection with the merger requires the affirmative vote of not less than a majority of the votes of holders of our common and preferred stock, voting together as a single class on an as-converted to common stock basis, present in person or represented by proxy at the annual meeting and entitled to vote thereon, provided that a quorum is present at the annual meeting.

THE MERGER

The following is a description of the material aspects of the merger. While Mandalay Digital believes that the following description covers the material terms of the merger, the description may not contain all of the information that is important to Mandalay Digital stockholders and Appia stockholders. Mandalay Digital encourages its stockholders and the Appia security holders to carefully read this entire proxy statement/prospectus, including "Summary of the Merger Agreement" stating on page 81 and the merger agreement attached to this proxy statement/prospectus as Appendix A and incorporated herein by reference, for a more complete understanding of the merger.

Overview

The Mandalay Digital board of directors has approved, and on November 13, 2014 we entered into, the merger agreement, which provides for Merger Sub, a wholly-owned subsidiary of Mandalay Digital, to be merged with and into Appia, with Appia surviving the merger as a wholly-owned subsidiary of Mandalay Digital.

At the effective time of the merger, on the terms and subject to the conditions set forth in the merger agreement, all of the equity securities of Appia outstanding immediately prior to the effective time (including outstanding stock options, which will be assumed by Mandalay Digital), other than equity securities held by stockholders that exercise dissenters' rights under applicable law, will be converted into the right to receive a number of shares (or, in the case of stock options being assumed, stock options) of Mandalay Digital common stock, equal to \$100 million less Appia's net debt and transaction expenses at the time of closing (subject to a working capital adjustment), divided by \$4.50.

Mandalay Digital expects to issue, pursuant to the merger agreement, up to 19,500,000 shares of common stock (including shares underlying assumed stock options and 1,000,000 shares to be deposited into escrow). The aggregate number shares of Mandalay Digital common stock to be issued pursuant to the merger agreement will vary depending on a number of factors, including the amount of the estimated net debt at closing. The merger agreement does not provide for an adjustment in the aggregate number of shares of Mandalay Digital common stock or options to be issued by Mandalay Digital to Appia's security holders in the acquisition in the event of fluctuations in the market value of Mandalay Digital's common stock up through the closing date. However, the number of shares to be issued by Mandalay Digital pursuant to the merger agreement is subject to adjustment based on Appia's working capital and net debt as of the closing date of the merger. We currently estimate that, if the merger closes on January 31, 2015, we would issue 19,056,392 shares; additionally, we would assume options for 277,376 shares of our common stock, for an aggregate of 19,333,768 shares, which would represent approximately 33% of the Company's outstanding shares of common stock as of November 13, 2014 (after giving effect to such issuance). Appia's actual results of operations between the date of the financial information included in this proxy statement/prospectus and the closing date may differ from the forecasted results used in this estimate, resulting in differences between the forecast and actual amounts of Appia's working capital and net debt, including transaction expenses, as of the closing date of the merger. See "The Merger – Consideration Payable in the Merger" located on page 69 of this proxy statement/prospectus for further information and a discussion of the assumptions in calculating the potential amount of shares to be issued by Mandalay Digital in the merger.

We will deposit 1,000,000 of the shares into an escrow account to cover indemnification claims by Mandalay Digital pursuant to the terms of the merger agreement, which shares will be released one year after the closing, subject to any then pending claims. As of November 13, 2014, there were 37,797,221 shares of our common stock outstanding. The shares of our common stock to be issued in the merger will be issued from our authorized but unissued shares of common stock.

Upon consummation of the merger, Mandalay Digital will guarantee Appia's existing indebtedness outstanding as of closing, which amount is expected to be approximately \$9.5 million (excluding the transaction fees and expenses) assuming the merger closes on January 31, 2015. See "The Merger – Consideration Payable

in the Merger" located on page 68 of this proxy statement/prospectus for further information and assumptions about Appia's forecasted indebtedness based on certain closing dates. Upon consummation, and as a condition to the closing, of the merger, Appia will enter into an amended loan agreement with its current senior lender, Silicon Valley Bank, and a securities purchase agreement with Mandalay Digital and North Atlantic pursuant to which Appia will issue to North Atlantic a subordinated debenture in the aggregate principal amount of \$8.0 million, which amount is included in the \$9.5 million referenced above, to retire a like amount of outstanding debt to North Atlantic.

Mandalay Digital will issue a secured guaranty of each of the SVB Debt and North Atlantic Debt whereby we will guarantee all of Appia's obligations in connection with such indebtedness and pledge substantially all of its assets, including its intellectual property. The secured guarantees will contain cross-default provisions and covenants, among others, limiting our ability to undergo a change of control, incur indebtedness, grant liens, make dividends in cash and other customary covenants. The SVB Debt will mature at June 30, 2015, with respect to the revolving line, and April 1, 2016, with respect to the term loan. The North Atlantic Debt will mature two years after the date of issuance. SVB and North Atlantic will also enter into a subordination agreement pursuant to which North Atlantic will agree to, among other things, limit its rights to receive payment, exercise remedies and enforce its rights for a certain period while an event of default under the North Atlantic debt exists.

Mandalay will also issue to North Atlantic 200,000 shares of its common stock and a common stock warrant to purchase 400,000 shares of its common stock at an exercise price of \$0.01 per share exercisable for 10 years; provided that the warrant is not exercisable until the one year anniversary of the closing date of the merger and will terminate if the Company repays the North Atlantic Debt prior to such one year anniversary. We intend to refinance the Appia debt as soon as practicable following the closing of the merger.

Appia stockholders will be subject to lock up restrictions lapsing with respect to one-third of such shares at each of the six, nine and 12 month anniversaries of the closing date. Upon the closing of the merger, Mandalay Digital will appoint Jud Bowman and Craig Forman as members to the board of directors.

Background of the Merger

From time to time, Mandalay Digital considers various potential strategic acquisition transactions to enhance shareholder value. Mandalay Digital has completed four strategic transactions since late 2011 and has reviewed others, including the opportunities to acquire intellectual property and technologies, to acquire distribution for products and services in geographies not currently served and to acquire a complementary advertising tech platform.

In May 2012, William Stone, Mandalay Digital's Chief Executive Officer (and then a consultant to Mandalay Digital) contacted Judson S. Bowman, Appia's Chief Executive Officer, to discuss potential operator business development relationships. Mr. Stone had previously been introduced to Mr. Bowman in 2009 when Appia (then named Pocketgear) acquired Handango, Inc., an independent, third party application store. Mr. Stone served on the board of directors of Handango at the time of the acquisition and had previously been the Chief Executive Officer of Handango (although Mr. Stone did not receive any consideration in connection with the acquisition). While the May 2012 meeting did not conclude with any specific next steps, Mr. Stone and Mr. Bowman agreed to stay in touch and look for opportunities to collaborate. In July and August 2013, Mr. Stone and Mr. Bowman had several conversations about the shared vision for Mandalay Digital and Appia and how both companies could be more valuable together versus standalone. Although Mandalay Digital and Appia shared a common vision, the combination of the parties' valuation expectations and the limited operating history of each company made any potential acquisition discussions premature at the time. Thus, Mr. Bowman and Mr. Stone agreed to stay in touch and look for commercial opportunities to work together. They discussed one specific opportunity in July and August of 2013 where Mandalay Digital would potentially acquire some of Appia's legacy assets and contracts. Although the parties did not enter into a commercial transaction at that time they agreed to continue to look for opportunities to work together in the future.

Mandalay Digital's executive management team believed that given Mandalay Digital's pending United States summer 2014 launch of its advertising business with DT Ignite and DT IQ, Mandalay Digital would benefit significantly from building, acquiring and/or contracting with third parties such as Appia to use a sophisticated mobile app advertising platform. In light of the length of time it would take to build such a platform, the related execution risk (and Appia's status as a leading independent app installation network) and the pending launch of the advertising business, in March 2014, Mr. Stone contacted Mr. Bowman again to discuss potential strategic relationships and transactions.

On March 13, 2014, Mr. Stone and Andrew Schleimer, Mandalay Digital's Executive Vice President and Chief Financial Officer (and then a consultant to Mandalay Digital), held a telephone conference with Mr. Bowman and Timothy Oakley, Appia's Chief Financial Officer, to discuss Mandalay Digital's capitalization, financial condition and operations.

On March 18, 2014, Mr. Stone delivered a non-binding letter of intent outlining a proposed structure and purchase price for an acquisition of Appia by Mandalay Digital. Mandalay Digital's proposal contemplated total aggregate consideration of up to \$110 million to acquire Appia on a cash free and debt free basis, consisting of \$30 million in cash, \$50 million in Mandalay Digital common stock and potential earnout consideration of up to \$30 million (involving the same relative mix of cash and stock).

On March 22, 2014, representatives of America's Growth Capital (AGC), financial advisors to Appia, responded to Mr. Stone's proposal by proposing a \$125 million aggregate purchase price for Appia, consisting of \$30 million in cash, \$65 million in Mandalay Digital common stock and \$30 million of potential earnout consideration.

During the course of the next several months, representatives of Mandalay Digital and Appia, including Mr. Stone, Mr. Schleimer and Jeffrey Klausner, Mandalay Digital's then Chief Financial Officer, engaged in a number of meetings and discussions regarding the terms of the proposed business combination transaction as well as preliminary due diligence matters. In March, 2014, Mandalay Digital engaged legal and accounting advisors to assist in its preliminary evaluation of a potential transaction, retaining Latham & Watkins LLP (Latham) as M&A counsel and Edward Thomas Associates, Inc. as external accounting advisor. Mandalay Digital provided Appia with a preliminary due diligence request on April 5, 2014 and, commencing April 17, 2014, Appia began providing Mandalay Digital and its outside advisors, including Latham and Edward Thomas Associates, with access to certain due diligence materials through an online data room.

Based on its preliminary due diligence investigation of Appia's financial results, on April 23, 2014, Mandalay Digital submitted a revised non-binding letter of intent to Appia, proposing aggregate consideration of up to \$100 million, consisting of \$30 million of cash, \$40 million of Mandalay Digital common stock and potential cash and stock earnout consideration of up to \$30 million.

On April 28, 2014, Mandalay Digital's Board of Directors (the Board) held a meeting and discussed, among other things, the Appia opportunity and the status of the proposed transaction including the proposed terms of the transaction. The Board directed Mandalay Digital's management to continue discussions with Appia with the objective of entering into a non-binding letter of intent outlining the principal terms of a potential transaction.

On April 29, 2014, Mr. Stone, Mr. Schleimer, Mr. Bowman and a representative of AGC participated on a conference call to discuss the potential transaction.

Following this call on April 29, 2014, representatives of AGC submitted a counterproposal to Mandalay Digital which provided for up to \$115 million of total consideration, consisting of \$30 million in cash, \$55 million of stock and potential cash and stock earnout consideration of \$30 million.

On May 6, 2014, Mr. Schleimer, Mr. Bowman and a representative of AGC met in person in New York City to negotiate open issues with respect to Mandalay Digital's proposal. In addition to the amount of the overall purchase price, one of the issues of significant importance to Mandalay Digital in those discussions was the imposition of a minimum "floor" on the deemed value of Mandalay Digital's stock for purposes of determining how many shares of Mandalay Digital stock would be issued in the transaction. Representatives of Appia, in turn, proposed an overall "cap" on the deemed value of Mandalay Digital's stock for purposes of the transaction. The parties exchanged revised drafts of the non-binding letter of intent over the next several days.

On May 14, 2014, Mandalay Digital and Appia entered into non-binding letter of intent with respect to a business combination transaction which contemplated Mandalay Digital's acquisition of Appia for total aggregate consideration of up to \$115 million consisting of \$30 million in cash, \$55 million in Mandalay Digital stock and up to \$30 million of proposed contingent earnout consideration. The May 2014 letter of intent provided that the number of shares of Mandalay Digital common stock to be issued in the transaction would be calculated based on the 25 trading day value of Mandalay Digital's shares prior to the execution of definitive agreements with respect to the transaction, subject to a "floor" of \$4.10 per share and a "cap" of \$5.00 per share. Mandalay Digital's common stock closed at \$3.66 per share on May 14, 2014. The letter of intent also contemplated that the definitive agreements would provide for a termination fee of \$2.0 million in the event Mandalay Digital was unable to consummate the transaction due to its inability to secure financing or stockholder approval.

On June 1, 2014, Mr. Stone and Mr. Schleimer circulated a memorandum to Mandalay Digital's Board providing a status update with respect to the transaction. Mr. Stone and Mr. Schleimer also recommended the engagement of specialty financial advisory firms to assist with an evaluation of potential financing sources to fund the cash portion of the purchase price and to refinance Appia's existing indebtedness.

During the course of the month of June 2014, representatives of Mandalay Digital and Appia engaged in numerous meetings and conference calls to discuss due diligence matters, as well as potential modifications to the structure of the proposed transaction.

On June 12, 2014 Mandalay Digital engaged two specialized financial advisory firms to assist Mandalay Digital in arranging potential debt financing alternatives.

From June 12 through August 1, 2014 Mr. Stone, Mr. Schleimer and Robert Deutschman, an outside director of Mandalay Digital, together with Mandalay Digital's advisors, met with potential lenders on numerous occasions to determine financing alternatives and structures available to Mandalay Digital. Mandalay Digital held separate parallel discussions with Silicon Valley Bank (SVB) regarding a potential financing arrangement with respect to the proposed transaction.

On June 25, 2014 and July 23, 204, Mandalay Digital's Board held meetings and discussed, among other things, the status of the proposed transaction, an update on transaction timing and process, and an update on the financing alternatives that would potentially be available to Mandalay Digital.

During July 2014, representatives of Mandalay Digital and Appia held numerous in person and telephonic meetings to, among other things, further discuss due diligence matters, for Appia to provide Mandalay Digital with an update on Appia's second quarter financial results and to discuss potential alternative transaction structures.

On July 25, representatives of Mandalay Digital updated representatives of Appia on the status of discussions with potential sources of financing. At that time representatives of Mandalay Digital proposed to representatives of Appia that a portion of the consideration to be paid in the transaction should be in the form of subordinated notes to be issued to Appia stockholders so as to reduce the overall cost of funds in the transaction. In addition, Mandalay Digital and Appia's representatives discussed the possibility of simplifying the transaction structure by eliminating the \$30 million of potential earnout consideration in exchange for increasing the value of

the consideration to be paid at closing by \$15 million (to an aggregate of \$100 million). Based on these discussions, it became clear that both parties believed that the elimination of the earnout consideration would simplify the transaction and better align the incentives of all parties (to maximize the value of Mandalay Digital common stock) following the closing of a transaction.

Consistent with the foregoing, on July 31, 2014, Mr. Schleimer delivered a new proposal to representatives of Appia for the structure and financing of the proposed transaction. The revised proposal contemplated payment of an aggregate of \$100 million of consideration to acquire Appia on a cash free, debt free basis with a normalized level of working capital, consisting of \$15.0 million in cash, \$60 million of Mandalay Digital common stock and \$25 million of Mandalay Digital subordinated notes (and no earnout consideration). A revised amended and restated letter of intent reflecting these terms was delivered to representatives of Appia on August 5, 2014.

On August 7, 2014, representatives of Appia advised Mr. Schleimer that Appia would be prepared to move forward under the new structure that had been proposed by Mandalay Digital if Mandalay Digital would agree to eliminate the "floor" and "cap" on the deemed value of Mandalay Digital shares for purposes of the transaction and if a conversion feature were added to the proposed subordinated notes.

On August 11, 2014, Mandalay Digital's Board met via conference call to discuss the key terms and conditions of Mandalay Digital's updated proposal and Appia's response thereto, as well as the various financing alternatives that were potentially available to Mandalay Digital, including the possibility of paying a portion of the consideration in the form of subordinated notes to Appia equity holders.

Following this board meeting, representatives of Mandalay Digital informed representatives of Appia that Mandalay Digital would be willing to eliminate the "floor" and "cap" on the deemed value of Mandalay Digital stock but that Mandalay Digital would not agree to Appia's proposal that the subordinated notes should be convertible into shares of Mandalay Digital common stock.

On August 18, 2014, Appia and Mandalay Digital signed a revised non-binding letter of intent generally reflecting the terms proposed in Mandalay Digital's July 31, 2014 proposal, except that the "floor" and "cap" on the deemed value of Mandalay Digital stock was eliminated. Mandalay Digital's common stock closed at \$4.51 per share on August 18, 2014.

On August 22, 2014, Mr. Bowman, Mr. Oakley, Mr. Schleimer and representatives of AGC, Latham, Manatt, Phelps and Phillips, LLP, Mandalay Digital's outside corporate and securities counsel (Manatt), Morningstar Law Group, Appia's outside corporate and securities counsel (Morningstar), and Goodwin Procter LLP, Appia's outside M&A counsel (Goodwin), participated in a call to discuss certain legal and timing issues related to the proposed transaction.

During the course of the next several weeks, representatives of Mandalay Digital and Appia met in person and by telephone to discuss the financial performance of the respective companies and other due diligence matters, including meetings of the respective senior management teams of Appia and Mandalay Digital in Orange County, California on September 10 and September 11, 2014. During this period, the parties also discussed the possibility of further simplifying the transaction structure by having all of the consideration paid in the form of Mandalay Digital common stock (and eliminating the subordinated notes). Mandalay Digital's executive management team believed that this change in structure, if accomplished at an appropriate valuation of Mandalay Digital common stock, would benefit the combined company by significantly reducing its debt burden. Representatives of Appia indicated to representatives of Mandalay Digital that various Appia stockholders would prefer to pursue an all-stock transaction.

On September 15, 2014 Mr. Schleimer delivered a further amended and restated draft of a letter of intent to Appia's representatives which contemplated that all of the consideration to be paid in the transaction would be

paid in the form of Mandalay Digital common stock, subject to adjustment for the amount of net debt of Appia at closing and subject to a customary working capital adjustment. Although the parties never executed this further amended and restated letter of intent, they agreed in principle to this change in structure and instructed their legal advisors to begin preparing and negotiating drafts of definitive agreements.

On September 29, 2014, Mandalay Digital entered into a non-binding term sheet for a formula-based revolving line of credit with Silicon Valley Bank SVB to finance the acquisition and support working capital needs.

On October 3, 2014, representatives of Latham circulated on behalf of Mandalay Digital a draft of a proposed definitive merger agreement to representatives of Goodwin.

From October 3 until November 13, 2014, representatives of Appia, Mandalay Digital, Latham, Morningstar and Goodwin exchanged drafts of, and discussed terms of, the merger agreement, support agreements and other ancillary documents. These discussions focused on, among other things:

- the representations and warranties in the definitive agreement merger agreement
- the parties' respective indemnification obligations and limitations thereon
- · the amount and triggers for any termination fee and expense reimbursement that might by payable by Mandalay Digital
- the conditions to the parties' obligation to consummate the merger agreement
- the terms and conditions of the support agreements to be entered into by stockholders of Appia that are affiliates of Appia, including
 the length of post-closing lock-up periods

From October 3 until November 13, 2014, Mr. Stone and Mr. Schleimer held numerous meetings with Mr. Deutschman in his capacity as an outside director of Mandalay Digital to update and discuss with him the terms of the merger agreement, support agreements and other ancillary documents.

On October 6, 2014, Mr. Schleimer contacted AGC and informed them that based on conversations to date with potential sources of financing and input from Mandalay Digital's Board, Mandalay Digital would be requesting that Appia's existing lenders (North Atlantic Capital (NAC) and Silicon Valley Bank (SVB)) participate in the transaction by rolling over (or otherwise amending and/or restating) their existing debt facilities (subject to agreed-upon modifications) following the closing, thereby obviating the need for Mandalay Digital to secure outside third party financing to refinance those debt facilities.

During the week of October 6, 2014, representatives of Appia engaged in discussions with NAC and SVB and reported back to Mandalay Digital that both lenders had indicated their willingness to proceed on the basis of Mandalay Digital's proposal, subject to negotiation and execution of definitive agreements and Mandalay Digital's willingness to guarantee Appia's indebtedness following the closing.

Based on the progress in discussions, Mandalay Digital decided that it would be prudent to engage an investment banking firm to render a fairness opinion in the event that Mandalay Digital proceeded with a transaction. On October 17, 2014, Mandalay Digital retained Needham & Company to render an opinion to the Mandalay Digital board as to the fairness, from a financial point of view, of the consideration that might be paid in a transaction with Appia.

On October 7, 2014, Mr. Stone, Mr. Schleimer, Mr. Bowman and Mr. Oakley participated in a telephone conference in which Mr. Bowman and Mr. Oakley presented Appia's preliminary financial and operating results for the third quarter.

During the week of October 14, 2014, representatives of Mandalay Digital, Appia and NAC exchanged drafts of a term sheet regarding the terms of the roll-over of the NAC debt following the closing.

On October 23, 2014, Mr. Bowman, Mr. Schleimer and representatives of AGC, Latham and Goodwin met in New York City to discuss the terms of the Merger Agreement.

On October 28, 2014, NAC and Mandalay Digital signed a revised term sheet with respect to the replacement of the existing NAC debt facilities with a single new facility of equal principal amount following the closing.

From October 28 until November 13, 2014, representatives of Appia, Mandalay Digital, Latham, Morningstar, SVB and NAC exchanged drafts of, and discussed terms of, the SVB and NAC consent agreements to the debt roll-over and related loan documentation.

On November 3, 2014, Mr. Stone, Mr. Schleimer, Mr. Bowman and Mr. Forman held a conference call to negotiate certain terms of the Merger Agreement and ancillary documents. During this conference call Mr. Stone and Mr. Schleimer informed Mr. Bowman and Mr. Forman that Mandalay Digital would be willing to proceed with the transaction only if the Mandalay Digital common stock price were fixed at \$4.50 per share for purposes of determining how many shares of Mandalay Digital common stock would be issued to Appia's stockholders in the transaction. Mandalay Digital's closing stock price on November 3, 2014 was \$3.75 and, during the period from March 3, 2014 through November 3, 2014 had ranged from \$3.16 to \$5.89. On the following day, representatives of Appia informed representatives of Mandalay Digital that Appia's principal stockholders would be willing to proceed with the transaction on the basis of a \$4.50 deemed value for Mandalay Digital's common stock (although the consideration would be allocated internally among Appia preferred and common stockholders according to the Appia charter documents and Appia's internal agreements).

On November 9, 2014, Mandalay Digital's Board of directors held a special meeting to review the transaction. At this meeting:

- Representatives of Latham advised the Board of its fiduciary duties in the context of a potential transaction.
- Representatives of Latham and Mr. Stone and Mr. Schleimer provided the Board with a detailed summary of the most recent drafts of the merger agreement, the support agreement and the NAC and SVB debt arrangements.
- Mr. Stone and Mr. Schleimer updated the Board on Appia's financial results, synergies that could result from the transaction and certain other matters related to the proposed operations of the combined company if the transaction were consummated.

In addition, at this meeting members of the Board discussed the proposed termination fee of \$1.5 million and expense reimbursement obligation of up to \$500,000 that would be payable under certain circumstances by Mandalay Digital if Mandalay Digital's stockholders did not approve the proposed transaction. Based on these discussions, the Board directed management to contact representatives of Appia to request a reduction in the amount of the proposed \$1.5 million termination fee.

On November 9, 2014, Mr. Schleimer contacted representatives of Appia to discuss various outstanding issues, including the amount of the proposed termination fee and the lock up release dates. During that discussion, representatives of Appia indicated that they were prepared to agree to a reduction in the termination fee and separately requested the post-closing lock-up periods with respect to Mandalay Digital common stock issued in the transaction be relaxed (at that time, Mandalay Digital proposed to release shares from the lock-ups on the one year anniversary of the transaction with respect to Mr. Bowman and on the 6, 9 and 12 month anniversaries of the closing date (with respect to 25%, 25% and 50% of the shares respectively) with respect to all other Appia stockholders). Following these discussions, the parties agreed to reduce the amount of the termination fee from \$1.5 million to \$1.0 million and that the shares of Mandalay Digital common stock issued to all Appia stockholders would be released from the lock-up obligations in three equal installments on the 6, 9 and 12 month anniversaries of the closing date.

From November 9, 2014 through November 13, 2014, representatives of Appia, Mandalay Digital, Latham, Goodwin, Morningstar, SVB and NAC exchanged drafts of definitive agreements relating to the merger and the debt financing transactions.

On November 12, 2014, Mandalay Digital's Board of directors met to consider the proposed merger agreement, support agreements and debt financing arrangements with SVB and NAC. At this meeting:

- Representatives of Latham, Mr. Stone and Mr. Schleimer summarized the changes that had been made to the transaction agreements since the Board's last meeting, including the reduction in the termination fee from \$1.5 million to \$1.0 million.
- Representatives of Needham & Company delivered Needham & Company's oral opinion to the Board, which opinion was later
 confirmed in the written opinion of Needham & Company dated November 12, 2014, to the effect that, as of such date and based
 upon and subject to the assumptions and other matters described in the written opinion, the consideration to be paid by Mandalay
 Digital in the merger pursuant to the merger agreement was fair to Mandalay Digital from a financial point of view.

After discussions with executive management, Latham, Manatt and Needham & Company, Mandalay Digital's Board adopted resolutions approving the merger, the merger agreement, the debt financing arrangements and the transactions contemplated thereby.

On November 13, 2014, Appia's board of directors adopted resolutions by written consent unanimously approving the merger, the merger agreement and the transaction contemplated thereby.

On November 13, 2014 after the public markets closed on that date, representatives of Appia and Mandalay Digital executed and delivered the merger agreement, the parties to the support agreements executed and delivered the support agreements and representatives of Appia, Mandalay Digital, NAC and SVB executed consents and waivers under the existing NAC and SVB debt facilities with respect the transaction, which included forms of the definitive financing arrangements to be executed by the parties at the closing.

After the public markets closed on that date, Mandalay Digital and Appia issued a press release announcing the execution of the merger agreement.

Reasons for the Merger and Recommendation of Mandalay Digital's Board of Directors

By vote at a meeting held on November 12, 2014, Mandalay Digital's board of directors determined that the merger agreement and the merger, including the issuance of Mandalay Digital common stock in connection with the merger are advisable and in the best interests of Mandalay Digital and its stockholders and approved the merger agreement, the merger and the other transactions contemplated thereby. Mandalay Digital's board of directors recommends that Mandalay Digital stockholders vote FOR the issuance of up to 19,500,000 shares of our common stock pursuant to the merger agreement.

In evaluating the merger agreement, Mandalay Digital's board of directors consulted with Mandalay Digital's management, Mandalay Digital's legal advisors and reviewed the analysis and opinion of Needham & Company and, in deciding to approve the merger agreement and to recommend that Mandalay Digital stockholders vote to approve the issuance of Mandalay Digital common stock to Appia stockholders pursuant to the merger agreement, Mandalay Digital's board of directors considered several factors that it viewed as supporting its decision, including the following material factors:

- that the merger could enhance Mandalay Digital's competitive position by delivering a superior revenue opportunity for operator and OEM customers;
- that the merger could enhance Mandalay Digital Digital's scalable, low-capex business model, thereby increasing EBITDA and free cash flow;
- that as a result of the merger Mandalay Digital would own the largest independent app install network;

- revenue synergies expected to result for the combined company from the removal of ad partner and aggregator fees;
- the ability for the combined company to achieve cost synergies by eliminating overlapping campaign management and CPI-infrastructure functions as well as duplicative corporate headcount;
- that the merger could result in growth in advertising revenue as a result of rapid integration of Appia's app-install infrastructure into Digital Turbine's existing platform;
- that the merger could increase Mandalay Digital's direct access to leading publishers and advertisers and enable Mandalay Digital to build a marketplace of scale to leverage advertising opportunities and increase revenue streams for carriers;
- the breadth of Appia's employee talent pool with deep expertise, management and capabilities in ad tech;
- that Mandalay Digital's share price is set at \$4.50 per share for purposes of determining the number of shares to be issued by Mandalay Digital in the merger and that, as a result, the number of Mandalay Digital shares to be issued in the merger will not fluctuate based on fluctuations in Mandalay Digital's share price;
- the opinion of Needham & Company dated November 12, 2014 that based upon and subject to the factors and assumptions set forth therein, the consideration to be paid by Mandalay Digital in the merger pursuant to the merger agreement is fair to Mandalay Digital from a financial point of view;
- Latham and Watkins LLP's advice regarding fiduciary duties;
- that the Board of Directors of Mandalay Digital reserves the right to change its recommendation regarding the merger subject to the terms and conditions set forth in the merger agreement:
- that Appia's existing lenders had consented to the transaction and agreed to permit the roll-over (or replacement) of the Appia debt following the closing;
- that Appia's stockholders, which include a number of prominent and respected venture capital firms, will become stockholders of Mandalay Digital and will be subject to the transfer restrictions under the support agreements; and
- the commitment by both parties to complete the merger pursuant to their respective obligations under the terms of the merger
 agreement and the likelihood that requisite stockholder approval from Appia will be obtained in a timely manner as a result of
 the obligations of the parties to the support agreements.

Mandalay Digital's board of directors also considered a variety of risks and other potentially negative factors concerning the merger agreement and the transaction contemplated thereby. These factors included:

- the possibility that the merger may not be completed, or that completion may be unduly delayed, including because Mandalay Digital stockholders may not approve the issuance of the shares by Mandalay Digital to Appia stockholders in connection with the merger or because of reasons beyond the control of Mandalay Digital or Appia;
- the risk that failure to complete the merger could negatively affect the price of Mandalay Digital common stock and future business and financial results of Mandalay Digital;
- the risk of not capturing some or all of the anticipated revenue synergies and cost savings between Mandalay Digital and Appia and the risk that other anticipated benefits might not be realized in the expected timeframe or at all;
- the fact that Appia has historically incurred operating losses and that additional financing may be required following of the transaction:
- the substantial costs to be incurred in connection with the transaction, including integration costs and transaction expenses arising from the merger;

- □ the obligation to pay Appia a \$1 million termination fee and \$500,000 of expense reimbursement if the merger agreement is terminated under certain circumstances;
- the risk that Mandalay Digital may be obligated to issue an additional 400,000 shares of common stock to one of Appia's lenders if that lender's debt has not been repaid or refinanced on or prior to the first anniversary of the closing date; and
- the other matters described under the caption "Risk Factors" on page 23.

This discussion of the information and factors considered by Mandalay Digital's board of directors in reaching its conclusion and recommendations is not intended to be exhaustive and is not provided in any specific order or ranking. In view of the wide variety of factors considered by Mandalay Digital's board of directors in evaluating the merger agreement and the transactions contemplated by it, including the merger, and the complexity of these matters, Mandalay Digital's board of directors did not find it practicable to, and did not attempt to, quantify, rank or otherwise assign relative weight to those factors. In addition, different members of Mandalay Digital's board of directors may have given different weight to different factors. Mandalay Digital's board of directors did not reach any specific conclusion with respect to any of the factors considered and instead conducted an overall review of such factors and determined that, in the aggregate, the potential benefits considered outweighed the potential risks or possible negative consequences of approving the merger agreement and the issuance of shares of Mandalay Digital common stock thereunder.

Opinion of Needham & Company, LLC

We retained Needham & Company, LLC ("Needham & Company") to render an opinion as to the fairness, from a financial point of view, to us of the consideration to be paid by us in the merger pursuant to the merger agreement. Needham & Company was not requested to, and did not, provide independent financial advice to us during the course of the negotiation or structuring of the merger or participate in the negotiation or structuring of the merger.

On November 12, 2014, Needham & Company delivered its oral opinion, which it subsequently confirmed in writing, to our board of directors that, as of that date and based upon and subject to the assumptions and other matters described in the written opinion, the consideration to be paid by us in the merger pursuant to the merger agreement was fair to us from a financial point of view. Needham & Company provided its opinion for the information and assistance of our board of directors (in its capacity as such) in connection with and for the purpose of the board's evaluation of the transaction contemplated by the merger agreement. Needham & Company's opinion relates only to the fairness, from a financial point of view, to us of the consideration to be paid by us in the merger, which was determined through arm's length negotiations between Mandalay Digital and Appia and not by Needham & Company. Needham & Company's opinion does not address any other aspect of the merger, or any related transaction, and does not constitute a recommendation to any stockholder of Mandalay Digital as to how that stockholder should vote or act on any matter relating to the merger.

The complete text of Needham & Company's opinion, which sets forth the assumptions made, procedures followed, matters considered, and qualifications and limitations on and scope of the review undertaken by Needham & Company, is attached to this proxy statement/prospectus as Appendix C and is incorporated by reference in this proxy statement/prospectus. The summary of Needham & Company's opinion set forth below is qualified in its entirety by reference to the full text of the opinion. Mandalay Digital stockholders should read this opinion carefully and in its entirety.

In arriving at its opinion, Needham & Company, among other things:

- reviewed a draft of the merger agreement dated November 10, 2014;
- reviewed certain publicly available information concerning Mandalay Digital and Appia and certain other relevant financial and operating data of Mandalay Digital and Appia furnished to Needham & Company by Mandalay Digital;

- reviewed the historical stock prices and trading volumes of Mandalay Digital common stock;
- held discussions with members of management of Mandalay Digital concerning the current operations of and future business
 prospects for Mandalay Digital and Appia and joint prospects for the combined companies, including the potential cost savings
 and other synergies that may be achieved by the combined companies;
- reviewed certain financial forecasts with respect to Mandalay Digital and Appia prepared by the respective managements of
 those companies and provided to Needham & Company by Mandalay Digital and held discussions with members of Mandalay
 Digital management concerning those forecasts;
- compared certain publicly available financial data of companies whose securities are traded in the public markets and that Needham & Company deemed generally relevant to similar data for Mandalay Digital and Appia;
- reviewed the financial terms of certain business combinations that Needham & Company deemed generally relevant; and
- reviewed such other financial studies and analyses and considered such other matters as Needham & Company deemed appropriate.

In connection with its review and in arriving at its opinion, Needham & Company assumed and relied on the accuracy and completeness of all of the financial, accounting, legal, tax and other information discussed with or reviewed by it for purposes of its opinion and did not independently verify, nor did Needham & Company assume responsibility for independent verification of, any of that information. Needham & Company assumed the accuracy in all material respects of the representations and warranties contained in the merger agreement and all agreements related thereto. In addition, Needham & Company assumed that the merger will qualify as a reorganization within the meaning of Section 368(a) of the Internal Revenue Code of 1986 and will be consummated on the terms and subject to the conditions set forth in the draft merger agreement furnished to Needham & Company without waiver, modification or amendment of any material term, condition or agreement thereof and that, in the course of obtaining the necessary regulatory or third party approvals, consents and releases for the merger, no material delay, limitation, restriction or condition will be imposed that would have an adverse effect on Mandalay Digital, Appia or the contemplated benefits of the merger. Needham & Company also assumed that any adjustments pursuant to the terms of the merger agreement to the aggregate exercise price of all Appia options, as provided to Needham & Company by Mandalay Digital and used by Needham & Company in its analyses, or the Adjustment Amount (as defined in the merger agreement) calculated as of September 30, 2014, would not be material to Needham & Company's analyses. In addition, Needham & Company assumed that the financial forecasts for Mandalay Digital and Appia prepared by the respective managements of Mandalay Digital and Appia (including adjustments thereto expected by Mandalay Digital to result from the merger) were reasonably prepared on bases reflecting the best currently available estimates and judgments of such managements, at the time of preparation, of the future operating and financial performance of Mandalay Digital, Appia and the combined companies, and Needham & Company relied, without independent verification, upon the estimates of Mandalay Digital management of the potential cost savings and other synergies, including the amount and timing thereof, that may be achieved as a result of the merger. Needham & Company expressed no opinion with respect to any of those forecasts, including those costs savings and other synergies, or the assumptions on which they were based.

Needham & Company did not assume any responsibility for or make or obtain any independent evaluation, appraisal or physical inspection of the assets or liabilities of Mandalay Digital, Appia or any of their respective subsidiaries nor did Needham & Company evaluate the solvency or fair value of Mandalay Digital, Appia or any of their respective subsidiaries under any state or federal laws relating to bankruptcy, insolvency or

similar matters. Needham & Company's opinion states that it was based on economic, monetary and market conditions as they existed and could be evaluated as of its date, and Needham & Company assumed no responsibility to update or revise its opinion based upon circumstances and events occurring after its date. Needham & Company's opinion is limited to the fairness, from a financial point of view, to us of the consideration to be paid by us pursuant to the merger agreement and Needham & Company expressed no opinion as to the fairness of the merger to, or any consideration received in connection therewith by, the holders of any class of securities, creditors or other constituencies of Mandalay Digital, or as to our underlying business decision to engage in the merger or the relative merits of the merger as compared to other business strategies that might be available to us. In addition, Needham & Company expressed no opinion with respect to the amount or nature or any other aspect of any compensation payable to or to be received by any officers, directors or employees of any party to the merger, or any class of those persons, relative to the consideration to be paid by us pursuant to the merger agreement or with respect to the fairness of any such compensation. Needham & Company did not express any opinion as to what the value of our common stock will be when issued pursuant to the merger or the prices at which our common stock will actually trade at any time.

We imposed no limitations on Needham & Company with respect to the investigations made or procedures followed by Needham & Company in rendering its opinion.

In preparing its opinion, Needham & Company performed a variety of financial and comparative analyses. The following paragraphs summarize the material financial analyses performed by Needham & Company in arriving at its opinion. The order of analyses described does not represent relative importance or weight given to those analyses by Needham & Company. Some of the summaries of the financial analyses include information presented in tabular format. The tables are not intended to stand alone, and in order to more fully understand the financial analyses used by Needham & Company, the tables must be read together with the full text of each summary. The following quantitative information, to the extent it is based on market data, is, except as otherwise indicated, based on market data as it existed on or prior to November 12, 2014, and is not necessarily indicative of current or future market conditions.

Contribution Analysis, Needham & Company reviewed and analyzed the implied percentage contribution of each of Mandalay Digital and Appia to pro forma combined operating results for the last 12 months ended September 30, 2014, and pro forma projected fiscal year 2015 and fiscal year 2016 combined operating results. In calculating the pro forma projected combined operating results, Needham & Company used Mandalay Digital management's financial forecasts of Mandalay Digital's projected fiscal years 2015 and 2016 operating results and Appia management's financial forecasts of projected operating results for the four quarter periods that correspond to Mandalay Digital's fiscal years 2015 and 2016. Needham & Company assumed that each company's contribution to the applicable financial metric reflected its contribution to the combined company's corresponding pro forma financial metric and did not take into account any of the possible synergies or expenses that may be realized as a result of or following completion of the merger. Needham & Company reviewed, among other things, the implied percentage contributions to pro forma combined revenue, gross profit, and earnings before interest, taxes, depreciation, amortization, and stock compensation expense, or adjusted EBITDA. Needham & Company also reviewed the implied percentage contributions to pro forma combined net cash as of September 30, 2014. The following tables present the results of this analysis and the estimated percentage ownership of the combined company on a pro forma basis by the Mandalay Digital stockholders and Appia stockholders, based on an assumed total of 20.1 million shares of our common stock to be issued pursuant to the merger assuming the Adjustment Amount (as defined in the merger agreement) calculated as of September 30, 2014. In calculating pro forma equity ownership, Needham & Company used the treasury stock method to calculate the number of pro forma fully diluted shares of our common stock outstanding after the merger.

	Implied	Implied Percentage Contribution		
	Percentage Contribution			
	Mandalay Digital	Appia		
Pro forma combined revenues				
Last 12 months	44.5%	55.5%		
Fiscal 2015E	54.5%	45.5%		
Fiscal 2016E	67.2%	32.8%		
Pro forma combined gross profit				
Last 12 months	51.6%	48.4%		
Fiscal 2015E	68.9%	31.1%		
Fiscal 2016E	72.9%	27.1%		
Pro forma combined adjusted EBITDA				
Fiscal 2016E	94.4%	5.6%		
Pro forma combined net cash at September 30, 2014	89.7%	10.3%		
	Estimated Pro Forma Percentage Contribution			
	Mandalay Digital	Appia		
Pro forma equity ownership contribution	66%	34%		

Information as to relative contributions to EBITDA for the last 12 months and projected fiscal year 2015 was not meaningful due to Mandalay Digital's and Appia's actual and projected negative EBITDA for those periods.

The results of the contribution analysis are not necessarily indicative of the contributions that the respective businesses may have in the future.

Selected Companies Analysis — Appia. Using publicly available information, Needham & Company compared selected historical and projected financial and market data ratios for Appia implied by the merger to the corresponding data and ratios of publicly traded companies that Needham & Company deemed relevant because they have lines of businesses that may be considered similar to or have similar characteristics with Appia's lines of business. These companies, referred to as the Appia selected companies, consisted of the following:

Conversant, Inc.

Criteo S.A.

Mandalay Digital

Marin Software Incorporated

Millennial Media, Inc.

Rocket Fuel Inc.

The Rubicon Project, Inc.

Tremor Video, Inc.

TubeMogul, Inc.

YuMe, Inc.

The following table sets forth information concerning the following multiples for the Appia selected companies, to the extent publicly available, and for Appia implied by the merger:

- enterprise value as a multiple of last 12 months, or LTM, revenues;
- enterprise value as a multiple of projected calendar year 2014 revenues;
- enterprise value as a multiple of projected calendar year 2015 revenues;
- enterprise value as a multiple of projected calendar year 2016 revenues; and
- enterprise value as a multiple of projected calendar year 2016 adjusted EBITDA (excluding the impact of selected company multiples deemed not meaningful).

Needham & Company also reviewed, to the extent publicly available, for the Appia selected companies, enterprise value as a multiple of LTM, projected calendar year 2014 and projected calendar year 2015 adjusted EBITDA, but did not compare the results with corresponding data of Appia because of Appia's negative LTM, projected calendar year 2014 and projected calendar year 2015 adjusted EBITDA.

Needham & Company calculated multiples for the Appia selected companies based on the closing stock prices of those companies on November 11, 2014 and for Appia based on an implied enterprise value of \$78.7 million reflecting the assumed merger consideration of 20.1 million shares of our common stock assuming the Adjustment Amount (as defined in the merger agreement) calculated as of September 30, 2014 and our closing stock price on November 11, 2014. All financial information excludes the impact of non-recurring items. Adjusted EBITDA amounts excluded the impact of stock-based compensation expense.

	Appia Selected Companies			Appia Implied by	
	High	Low	Mean	Median	Merger
Enterprise value to LTM revenues	6.2x	0.4x	3.0x	3.0x	2.6x
Enterprise value to projected calendar year 2014 revenues	5.6x	0.4x	2.8x	2.8x	2.6x
Enterprise value to projected calendar year 2015 revenues	4.2x	0.3x	1.9x	1.6x	1.6x
Enterprise value to projected calendar year 2016 revenues	3.2x	0.3x	1.8x	1.9x	0.9x
Enterprise value to projected calendar year 2016 adjusted EBITDA	45.7x	9.0x	22.2x	12.7x	8.6x

Needham & Company also reviewed, for the selected companies and for Appia, and on a calendar year basis, projected 2013 to 2014, 2014 to 2015, and 2015 to 2016 revenue growth percentages, and noted that Appia's projected decrease in revenue from 2013 to 2014 of 34.9% fell below the mean and median 2013 to 2014 revenue growth percentages of the selected companies of 35.3% and 25.6%, respectively, but that Appia's projected 2014 to 2015 and 2015 to 2016 revenue growth percentages of 66.0% and 72.5% exceeded the mean and median projected growth percentages of the selected companies of 55.1% and 24.7% for 2014 to 2015 and 22.0% and 22.9% for 2015 to 2016, respectively.

Selected Transactions Analysis. Needham & Company analyzed publicly available financial information for the following selected merger and acquisition transactions, which represent transactions announced since June 1, 2011 that involved target companies that were advertising technology companies involved in lines of business that may be considered similar to or have similar characteristics with Appia's lines of business and with enterprise values of less than \$650 million:

Target
BrightRoll, Inc.
Nexage
Taptica Ltd.
interCLICK, Inc.
Dotomi, Inc.
Hipcricket, Inc.
AdXpose, Inc.
MediaMind Technologies, Inc.

Needham & Company also analyzed publicly available financial information for the following selected merger and acquisition transactions, which represent additional transactions announced since June 1, 2011 that involved privately-held target companies that were advertising technology companies involved in lines of business that may be considered similar to or have similar characteristics with Appia's lines of business and with enterprise values of less than \$650 million. These selected transactions were differentiated from the transactions described in the preceding paragraph by the fact that, because the target companies were privately-held, disclosure of their underlying financial information was generally limited and/or not necessarily proximate to the announcement date of the transaction. While still being relevant information, this lack of proximity generally resulted in LTM metrics that were not fully comparable to those calculated for the transactions described in the preceding paragraph and, thus, were separately shown.

Acquirer	Target
Amobee, Inc.	Kontera Technologies, Inc.
Amobee, Inc.	Adconion Media Group Ltd.
Rocket Fuel, Inc.	X Plus Two Solutions, Inc.
LinkedIn Corporation	Bizo, Inc.
Opera Mediaworks, Inc.	AdColony
blinkx Plc	Rhythm NewMedia, Inc.
Millennial Media, Inc.	JumpTap, Inc.
Twitter, Inc.	MoPub, Inc.
AOL, Inc.	Adap.tv, Inc.
blinkx Plc	Prime Visibility Media Group, Inc.

In reviewing the selected transactions, Needham & Company calculated for the selected transactions, to the extent publicly available, and for Appia implied by the merger:

- enterprise value as a multiple of LTM revenues; and
- enterprise value as a multiple of projected next 12 months revenues (which, in the case of Appia, was the next 12 months beginning October 2014).

Needham & Company also reviewed, for the selected transactions, enterprise value as a multiple of LTM adjusted EBITDA, but determined that the results were not meaningful because of Appia's negative LTM adjusted EBITDA.

Needham & Company calculated multiples for Appia based on an implied enterprise value of \$78.7 million reflecting the assumed merger consideration of 20.1 million shares of our common stock and our closing stock price on November 11, 2014.

The following table sets forth information concerning the multiples described above for the selected transactions and the same multiples implied by the merger.

	Selected Advertising Technology Transactions			nsactions	Appia Implied by	
	High	Low	Mean	Median	Merger	
Enterprise value to LTM revenues	13.3x	1.0x	5.9x	6.1x	2.6x	
Enterprise value to projected next 12 months revenues	5.0x	1.8x	3.5x	3.6x	2.0x	

		Financial Data as of More Than a Quarter Before the Announcement Date			
	High	Low	Mean	Median	Merger
Enterprise value to LTM revenues	139.2x	1.2x	17.8x	4.5x	2.6x
Enterprise value to projected next 12 months revenues	2.6x	2.6x	2.6x	2.6x	2.0x

eted Advertising Technology Transactions With

Discounted Cash Flow Analysis. Needham & Company performed an illustrative discounted cash flow analysis to determine indicators of illustrative implied equity values for Appia based on Appia management's financial forecasts. Needham & Company calculated a range of indications of the present value of unlevered free cash flows for Appia for projected calendar years 2015 through 2017 using discount rates ranging from 15.0% to 20.0%. The range of discount rates, reflecting an estimated range of weighted average costs of capital of Appia, was selected by Needham & Company utilizing its professional judgment and experience after taking into account the trading price volatility and risk associated with the common stocks of the 10 selected publicly-traded advertising technology companies, including Mandalay Digital, listed under "Selected Companies Analysis—Appia" as a surrogate for Appia common stock, the current market environment, the opportunity cost of equity capital and other relevant factors. Needham & Company then calculated a range of illustrative terminal values at the end of 2017 by applying multiples ranging from 1.7x to 2.9x to Appia management's estimate of its projected calendar year 2017 revenues. The range of multiples was selected by Needham & Company utilizing its professional judgment and experience after taking into account the enterprise value to LTM revenues multiple of 2.6x implied by the merger. These illustrative terminal values were then discounted to calculate ranges of implied indications of present values using discount rates ranging from 15.0% to 20.0%. Needham & Company then added the ranges of the implied present values of Appia's unlevered free cash flows for the projected years to the ranges of implied present values of Appia's terminal values to derive ranges of implied present enterprise values of Appia. Needham & Company then added net cash (debt) as of September 30, 2014, to derive a range of illustrative implied present equity values of \$112.4 million to \$222.4 million. Needham & Company noted that the equity value implied by the assumed consideration of 20.1 million shares of our common stock and our common stock closing stock price on November 11, 2014 of \$3.45 was \$69.2 million.

Selected Companies Analysis — Mandalay Digital. Using publicly available information, Needham & Company compared selected historical and projected financial and market data ratios for Mandalay Digital to the corresponding data and ratios of publicly traded companies that Needham & Company deemed relevant because they have lines of businesses that may be considered similar to or have similar characteristics with our lines of business. These companies, referred to as the Mandalay selected companies, consisted of the following:

Conversant, Inc.
Criteo S.A.
Marin Software Incorporated
Millennial Media, Inc.
Rocket Fuel Inc.
The Rubicon Project, Inc.
Tremor Video, Inc.
TubeMogul, Inc.
YuMe, Inc.

The following table sets forth information concerning the following multiples for the Mandalay selected companies, to the extent publicly available, and for us:

- enterprise value as a multiple of last 12 months, or LTM, revenues;
- enterprise value as a multiple of projected calendar year 2014 revenues;
- enterprise value as a multiple of projected calendar year 2015 revenues; and

 enterprise value as a multiple of projected calendar year 2015 adjusted EBITDA (excluding the impact of selected company multiples deemed to be not meaningful).

Needham & Company also reviewed, for the Mandalay selected companies, enterprise value as a multiple of projected calendar 2016 revenues and enterprise value as a multiple of LTM, projected calendar year 2014 and projected calendar year 2016 adjusted EBITDA, but did not compare the results with our corresponding data because of our negative LTM and projected calendar year 2014 adjusted EBITDA and unavailable projected calendar year 2016 revenues and adjusted EBITDA.

Needham & Company calculated multiples for the Mandalay selected companies and for us based on the closing stock prices of those companies on November 11, 2014. All financial information excludes the impact of non-recurring items. Adjusted EBITDA amounts excluded the impact of stock-based compensation expense.

	Mandalay Selected Companies			Mandalay	
	High	Low	Mean	Median	Digital
Enterprise value to LTM revenues	6.2x	0.4x	2.8x	2.7x	4.9x
Enterprise value to projected calendar year 2014 revenues	5.6x	0.4x	2.6x	2.6x	4.6x
Enterprise value to projected calendar year 2015 revenues	4.2x	0.3x	2.0x	2.2x	1.1x
Enterprise value to projected calendar year 2015 adjusted EBITDA	55.1x	9.8x	25.9x	15.2x	7.6x

Pro Forma Transaction Analysis. Needham & Company prepared pro forma analyses of the financial impact of the merger based on an assumed merger consideration of 20.1 million shares of our common stock and using our management's financial forecasts of our projected fiscal year 2016 operating results, Appia management's financial forecasts of projected operating results for the four quarter periods that correspond to our fiscal year 2016, and estimates of synergies and transaction expenses from our management. Based on these estimates and assumptions, Needham & Company's analysis showed that the merger would result in accretion to our estimated fiscal year 2016 non-GAAP earnings per share. The financial forecasts that underlie this analysis are subject to substantial uncertainty and, therefore, actual results may be substantially different, and these differences may be material.

No company, transaction or business used in the "Selected Companies Analysis—Appia," "Selected Transactions Analysis" or "Selected Companies Analysis—Mandalay" as a comparison is identical to Appia, Mandalay Digital or the merger. Accordingly, an evaluation of the results of these analyses is not entirely mathematical; rather, it involves complex considerations and judgments concerning differences in the financial and operating characteristics and other factors that could affect the acquisition, public trading or other values of the selected companies or selected transactions or the business segment, company or transaction to which they are being compared.

The summary set forth above does not purport to be a complete description of the analyses performed by Needham & Company in connection with the rendering of its opinion. The preparation of a fairness opinion is a complex analytical process involving various determinations as to the most appropriate and relevant quantitative and qualitative methods of financial analyses and the application of those methods to the particular circumstances and, therefore, such an opinion is not readily susceptible to summary description. Accordingly, Needham & Company believes that its analyses must be considered as a whole and that selecting portions of its analyses or the factors it considered, without considering all analyses and factors, could create a misleading or incomplete view of the process underlying its analyses and opinion. Needham & Company did not attribute any specific

weight to any factor or analysis considered by it. The fact that any specific analysis has been referred to in the summary above is not meant to indicate that such analysis was given greater weight than any other analysis.

In performing its analyses, Needham & Company made numerous assumptions with respect to industry performance, general business and economic conditions and other matters, many of which are beyond our or Appia's control. Any estimates contained in or underlying these analyses, including estimates of future performance, are not necessarily indicative of actual values or predictive of future results or values, which may be significantly more or less favorable than those estimates. Additionally, analyses relating to the values of businesses or assets do not purport to be appraisals or necessarily reflect the prices at which businesses or assets may actually be sold or the prices at which any securities have traded or may trade at any time in the future. Accordingly, these analyses and estimates are inherently subject to substantial uncertainty. Needham & Company's opinion and its related analyses were only one of many factors considered by our board of directors in its evaluation of the merger and should not be viewed as determinative of the views of our board of directors or management with respect to the exchange ratio or the merger.

Under the terms of its engagement letter with Needham & Company, we have paid or agreed to pay Needham & Company a nonrefundable fee for rendering the Needham & Company opinion that we and Needham & Company believe is customary for transactions of this nature. Needham & Company's fee is not contingent on the consummation of the merger. In addition, we have agreed to reimburse Needham & Company for certain of its out-of-pocket expenses and to indemnify Needham & Company and related persons against various liabilities, including certain liabilities under the federal securities laws.

Needham & Company is a nationally recognized investment banking firm. As part of its investment banking services, Needham & Company is regularly engaged in the valuation of businesses and their securities in connection with mergers and acquisitions, negotiated underwritings, secondary distributions of securities, private placements and other purposes. Needham & Company was retained by our board of directors to render an opinion in connection with the merger based on Needham & Company's experience as a financial advisor in mergers and acquisitions as well as Needham & Company's familiarity with us and our industry generally. Needham & Company has not in the past two years provided investment banking or financial advisory services to us or Appia for which Needham & Company has have received compensation. Needham & Company may in the future provide investment banking and financial advisory services to Mandalay Digital, Appia or their respective affiliates unrelated to the merger, for which services Needham & Company would expect to receive compensation. In the normal course of its business, Needham & Company may actively trade the equity securities of Mandalay Digital for its own account or for the account of its customers and, therefore, may at any time hold a long or short position in those securities.

Conditions to Completion of the Merger

Our and Appia's obligation to complete the merger is subject to the satisfaction or waiver of various conditions, including the following:

- our registration statement on Form S-4 registering the sale of our shares of common stock to be issued in the merger, of which this proxy statement/prospectus is a part, shall have become and remain effective under the Securities Act and shall not be the subject of any stop order or proceedings seeking a stop order;
- we have each received the requisite approval of our stockholders;
- receipt of the approval of (x) at least a majority of the outstanding shares of Appia common stock and Appia preferred stock (voting on an as-converted to Appia common stock basis), and (y) two-thirds of the outstanding shares of Appia Series A Preferred Stock, Appia Series B Preferred Stock, Appia Series C Preferred Stock and Appia Series D Preferred Stock (voting on an as-converted to Appia common stock basis), in each case entitled to vote on upon the merger.

- there shall not have been threatened or pending any suit, action, or proceeding by any government entity or any other person seeking to restrain or prohibit the consummation of the merger;
- each party's representations and warranties shall be accurate, subject to certain qualifications and exceptions;
- each party's receipt of an opinion from its outside legal counsel that is reasonably acceptable and dated as of the closing date of the merger to the effect that, on the basis of the facts, representations and assumptions set forth or referred to in such opinion, the merger will qualify for U.S. federal income tax purposes as a reorganization within the meaning of Section 368(a) of the Code;
- each party shall be in compliance in all material respects with its undertakings, covenants, and contractual obligations under the merger agreement; and
- there shall not have occurred any Material Adverse Change (as defined in the merger agreement) since September 30, 2014 with respect to Appia.

For further information relating to the various conditions to closing of the merger, see the section entitled "Summary of the Merger Agreement —Conditions to the Consummation of the Merger" on page 92.

Expected Timing of the Merger

We and Appia expect to complete the merger after all the conditions to the merger in the merger agreement are satisfied or waived, including after the receipt of approval of our stockholders of the issuance of shares of our common stock at our annual meeting and receipt of the approval of the requisite vote by Appia's stockholders, and the receipt of all required regulatory approvals. We and Appia currently expect to complete the merger during the first calendar quarter of 2015. However, it is possible that factors outside of either company's control could cause the merger to be completed at a later time or not at all. The merger agreement provides that the merger agreement may be terminated at any time prior to closing of the merger if, among other things, the conditions to the effectiveness of the merger have not been satisfied or waived on or before April 30, 2015 and the party requesting the termination has not willfully breached a covenant in the merger agreement that causes the condition not to occur.

Consideration Payable in the Merger

At the effective time of the merger, on the terms and subject to the conditions set forth in the merger agreement, all of the equity securities of Appia outstanding immediately prior to the effective time (including shares of common stock issuable upon exercise of stock options to be assumed by Mandalay Digital pursuant to the merger agreement), other than stockholders exercising dissenter's rights under applicable law, will be converted into the right to receive a number of shares (or, in the case of stock options being assumed, stock options) of Mandalay Digital common stock, equal to \$100 million less Appia's net debt and transaction fees and expenses at the time of closing (subject to a working capital adjustment), divided by \$4.50. For further information see "Summary of the Merger Agreement – Conversion of Securities in the Merger" starting on page 81.

The following table illustrates the aggregate number of shares that Mandalay Digital would issue to the Appia stockholders assuming a closing date of September 30, 2014, January 31, 2015 and March 31, 2015. The September 30, 2014 information is based on the unaudited pro forma financial statements included in the section "Unaudited Pro Forma Condensed Combined Financial Information" on page 101 of this proxy statement/prospectus. The information included for January 31 and March 31, 2015 is based on information and financial projections provided by Appia and is based on certain assumptions, including the following:

- Appia's operating results will not differ materially from Appia's management forecast.
- Appia's working capital will not differ materially from Appia' management forecast
- Appia will not experience any potential material effects of extraordinary business events, such as loss of any material customers or natural disasters.

The information provided in the table below is for illustrative purposes only. The actual aggregate number of shares that Mandalay Digital will issue at the closing of the merger will be based on the existing indebtedness, aggregate transaction fees and expenses and working capital at that time. There can be no guarantee that the assumptions on which the forecasts in the table below will be correct. In addition, there can be no assurance that Appia's working capital adjustment and net debt will not be significantly higher or lower than projected. This information was not prepared with a view to public disclosure or compliance with the published guidelines of the SEC or guidelines established by the American Institute of Certified Public Accountants regarding forecasts or projections. These forecasts do not purport to present our operations in accordance with U.S. GAAP. Appia's actual results of operations between the date of the financial information included in this proxy statement/prospectus and the closing date may differ materially from the forecasted results used in this estimate, resulting in differences between the forecast and actual amounts of Appia's working capital and net indebtedness as of the closing date of the merger.

No independent accountants have compiled, examined or performed any procedures with respect to this information nor have any independent accountants expressed any opinion or any other form of assurance on such information or its achievability or the assumptions on which they are based.

(in thousands, except per-share amounts):

	Pro Forma 9/30/2014	Forecast <u>1/31/2015</u>	Forecast 3/31/2015
Enterprise Value	\$100,000	\$100,000	\$100,000
Plus: Working Capital Adjustment	(172)	50	(234)
Less: Net Debt	(10,291)	(13,204)	(14,631)
Plus: Aggregate Exercise Price of Appia Options	156	156	156
Aggregate Consideration	\$ 89,692	\$ 87,002	\$ 85,291
Price Per Share	\$ 4.50	\$ 4.50	\$ 4.50
Aggregate MNDL Shares - fully diluted	19,932	19,334	18,953

Based on Appia's estimated indebtedness, transaction fees and expenses and working capital adjustment and assuming a closing date during the first quarter of calendar 2015, Mandalay Digital currently expects to issue, pursuant to the merger agreement, up to 19,500,000 shares of common stock (including shares underlying assumed stock options and 1,000,000 shares to be deposited into escrow). If the merger closes on January 31, 2015, we currently estimate that we would issue 19,056,392 shares; additionally we would assume options for 277,376 shares of our common stock, for an aggregate of 19,333,768 shares, which would represent approximately 33% of the Company's outstanding shares of common stock as of November 13, 2014 (after giving effect to such issuance). The merger agreement does not provide for an adjustment in the aggregate number of shares of Mandalay Digital common stock or options to be issued by Mandalay Digital to Appia's security holders in the acquisition in the event of fluctuations in the market value of Mandalay Digital's common stock up through the closing date. We will deposit 1,000,000 of the shares issued at the closing into an escrow account to cover indemnification claims by Mandalay Digital pursuant to the terms of the merger agreement, which shares will be released one year after the closing, subject to any then pending claims. As of November 13, 2014, there were 37,797,221 shares of our common stock outstanding. The shares of our common stock to be issued in the merger will be issued from our authorized but unissued shares of common stock.

Debt obligations and Guarantees

Upon consummation of the merger, Mandalay Digital will guarantee Appia's existing indebtedness outstanding as of closing, which amount is expected to be approximately \$9.5 million (excluding the transaction

fees and expenses) assuming the merger closes on January 31, 2015. See "The Merger – Consideration Payable in the Merger" located on page 69 of this proxy statement/prospectus for further information and assumptions about Appia's forecasted indebtedness based on certain closing dates. Upon consummation, and as a condition to the closing, of the merger, Appia will enter into an amended loan agreement with its current senior lender, Silicon Valley Bank, and a securities purchase agreement with Mandalay Digital and North Atlantic pursuant to which Appia will issue to North Atlantic a subordinated debenture in the aggregate principal amount of \$8.0 million, which amount is included in the \$9.5 million referenced above, to retire a like amount of outstanding debt to North Atlantic. We will also issue a secured guaranty of each of the SVB Debt and North Atlantic Debt whereby we will guarantee all of Appia's obligations in connection with such indebtedness and pledge substantially all of its assets, including its intellectual property. The secured guarantees will contain cross-default provisions and covenants, among others, limiting our ability to undergo a change of control, incur indebtedness, grant liens, make dividends in cash and other customary covenants. Mandalay will also issue to North Atlantic 200,000 shares of its common stock and a common stock warrant to purchase 400,000 shares of its common stock at an exercise price of \$0.01 per share exercisable for 10 years; provided that the warrant is not exercisable until the one year anniversary of the closing date of the merger and will terminate if the Company repays the North Atlantic Debt prior to such one year anniversary. We intend to refinance the Appia debt as soon as practicable following the closing of the merger.

Material U.S. Federal Income Tax Consequences of the Merger

The merger is intended to qualify as a "reorganization" within the meaning of Section 368(a) of the Code, and it is a condition to the completion of the merger that Mandalay Digital and Appia each receive written opinions from their respective outside legal counsel, dated as of the closing date of the merger, to the effect that the merger will qualify as a reorganization within the meaning of Section 368(a) of the Code. Such opinions will be based on certain assumptions, representations and covenants, including representations and covenants contained in representation letters provided by Mandalay Digital, Merger Sub and Appia. Assuming the merger qualifies as a reorganization, Appia stockholders generally will not recognize gain or loss for U.S. federal income tax purposes upon the exchange of their shares of Appia stock for shares of Mandalay Digital common stock in connection with the merger. The issuance of shares of Mandalay common stock in connection with the merger will not be a taxable transaction to Mandalay Digital stockholders for U.S. federal income tax purposes.

Tax matters are very complicated, and the tax consequences of the merger to a particular stockholder will depend on such stockholder's circumstances. Accordingly, you are urged to consult your tax advisor for a full understanding of the tax consequences of the merger to you, including the applicability and effect of U.S. federal, state, local and foreign income and other tax laws. For more information, see the section entitled "Material U.S. Federal Income Tax Consequences" beginning on page 78.

Regulatory Approvals Required For the Merger

Other than (i) the requirement that this proxy statement/prospectus be filed with the SEC; (ii) the requirement that the SEC declare the registration statement of which this proxy statement/prospectus is a part effective and that we distribute this proxy statement/prospectus as required under the Securities Act and the Exchange Act; and (iii) the requirement that we make certain other filings under the Exchange Act and pursuant to NASDAQ Stock Market rules, we are not aware of any material federal or state regulatory requirements or approvals that must be complied with or obtained in connection with the merger. We cannot predict whether we will obtain all required regulatory approvals to complete the merger, or whether any approvals will include conditions that would be detrimental to us or the combined company.

Fees and Expenses of the Merger

Generally, all fees and expenses incurred in connection with the merger agreement, the merger, or the transactions related thereto will be paid by the party incurring such fees and expenses. See "Summary of Merger Agreement – Termination of the Merger Agreement" on page 96 for a description of fees payable upon termination.

Anticipated Accounting Treatment of the Merger

The acquisition of Appia by us in the merger will be accounted for under the acquisition method of accounting in accordance with U.S. GAAP. From the date of the completion of the merger, the combined company's results of operations will include Appia's operating results and Appia's assets and liabilities, including identifiable intangible assets, at fair value with the excess purchase price allocated to goodwill.

Appraisal Rights of Holders of Mandalay Digital Common Stock

Under Delaware law and our Certificate of Incorporation, our stockholders are not entitled to any rights to seek appraisal of their shares of our common stock or to exercise any "dissenter's rights" or preemptive rights in connection with the proposal to approve the share issuance to be made pursuant to the merger agreement.

Securities Law Restrictions on Shares of Mandalay Common Stock Received in the Merger

We anticipate that all shares of our common stock issued to Appia security holders in connection with the merger will be freely transferable under applicable U.S. securities laws, except as to each recipient of Mandalay Digital shares that is considered an "affiliate" (as such term is defined in the Securities Act) of Mandalay Digital, in which case such affiliates will be permitted to sell the shares of our common stock received in the merger only pursuant to an effective registration statement or an exemption from the registration requirements of the Securities Act. Shares issued pursuant to the merger agreement will also be subject to lock-up restrictions as described below. This proxy statement/prospectus does not register the resale of shares of our stock held by affiliates.

Lock-up Restrictions

All of the Appia stockholders that signed the support agreements agreed that, and pursuant to the terms of the merger agreement, the shares of Mandalay Digital common stock to be received by stockholders in the merger are all subject to lock up restrictions, which restrictions will lapse with respect to one-third of such stockholders share of Mandalay Digital common stock at each of the six, nine and 12 month anniversaries of the closing date. See section entitled "Ancillary Agreements – The Support Agreement" on page 98.

Rights Pertaining to the Shares of Common Stock Received in the Merger

The shares of our common stock to be issued in the merger will be the same class of capital stock as our currently outstanding common stock and thus have rights identical to our currently outstanding shares of common stock. See the section entitled "Comparison of Rights of Stockholders of Mandalay Digital and Appia" on page 124.

Dividends on Mandalay Digital Common Stock Received in the Merger

We have not declared cash dividends on our common stock since our inception and we do not anticipate paying any cash dividends in the foreseeable future. Following completion of the merger, the holders of our common stock will be entitled to receive any dividends as may be declared by our board of directors from funds legally available therefor; provided, however, that we will be subject to covenants restricting our ability to pay dividends pursuant to the terms of the secured guarantees that we will enter into in connection with the closing of the merger. The dividend policy of the combined company will be determined by our board of directors.

Interests of Certain Appia Directors, Officers, and Affiliates in the Merger

Appia's executive officers and members of Appia's board may have interests in the merger that are different from, or are in addition to, the interests of Appia stockholders generally. Appia's board was aware of

these interests and considered them, among other matters, in adopting resolutions approving and declaring advisable the merger, the merger agreement and the transactions contemplated thereby and recommending to the stockholders of Appia that they consent to the merger agreement and merger.

Executive Officer Employment Arrangements

Judson Bowman, Appia's chief executive officer, is party to an employment agreement with Appia. In the event that Mr. Bowman is not employed by the surviving entity after the merger, Mr. Bowman will be entitled to receive salary continuation payments at his current annual base salary of \$200,000 for a period of eighteen (18) months and medical, disability and life insurance coverage at substantially the same coverage levels as Mr. Bowman has on the date of his termination for a period of eighteen (18) months. Mr. Bowman would be required to execute a general release in favor of Appia to receive his salary and benefits continuation.

Timothy Oakley, Appia's chief financial officer, is party to an employment letter with Appia. If Mr. Oakley's employment is terminated by Appia without cause, Mr. Oakley is entitled to receive salary continuation payments at his current annual base salary of \$184,250 for a period of ninety (90) days. Mr. Oakley is required to execute a general release in favor of Appia to receive his salary continuation.

Non-Compete Agreement

Mr. Bowman has entered into a Non-Compete Agreement with Mandalay Digital in connection with the execution of the merger agreement that becomes effective upon the consummation of the merger. The Non-Compete Agreement restricts Mr. Bowman for eighteen (18) months following the closing of the merger (i) from providing services to, or otherwise assisting, any business that is competitive with the business of Appia as it is conducted as of the closing date, (ii) from employing or soliciting any employee, consultant, agent or representative of Appia to cease his or her relationship with Appia, and (iii) from soliciting any advertiser, publisher, vendor or supplier to Appia to cease doing business with Appia or to reduce the amount of business that such person does with Appia. Mr. Bowman has not received any additional consideration in connection with the Non-Compete Agreement. It is a condition to the obligations of Mandalay to consummate the merger that the Non-Compete Agreement has not been challenged as unenforceable or repudiated by Mr. Bowman.

Accelerated Vesting of Restricted Stock and Options

Craig Forman, the chairman of Appia's board, holds a restricted stock grants for 196,500 shares of Appia common stock and an option to purchase 46,968 shares of Appia common stock at an exercise price of \$0.28 per share. As of November 13, 2014, 114,000 shares under the restricted stock grants and 35,226 shares subject to the option were unvested. At the closing of the merger, assuming continued service by Mr. Forman to Appia until that time, 27,500 of the unvested shares under the restricted stock grants shall immediately vest and all unvested shares subject to the option shall immediate vest and become exercisable. Such shares are included in the amount of shares to be issued pursuant to the merger agreement.

Ken Hayes, Appia's chief revenue officer, holds options to purchase 123,250 shares of Appia common stock at exercise prices of \$0.26 to \$0.28 per share. As of November 13, 2014, 61,189 shares subject to the options were unvested. At the closing of the merger, assuming continued service by Mr. Hayes to Appia until that time, 14,845 unvested shares will immediately vest and become exercisable. If Mr. Hayes' employment with Appia is terminated without cause within twelve (12) months following the Closing Date, certain of his unvested shares will immediately vest and become exercisable. Such shares are included in the amount of shares to be issued pursuant to the merger agreement.

Timothy Oakley, Appia's chief financial officer, holds a restricted stock grant for 175,000 shares of Appia common stock. As of November 13, 2014, 98,438 shares under the restricted stock grant were unvested. At the closing of the merger, assuming continued service by Mr. Oakley to Appia until that time, 63,985 unvested shares will immediately vest. Such shares are included in the amount of shares to be issued pursuant to the merger agreement.

Indemnification; Directors' and Officers' Insurance

Following the consummation of the merger for a period of six (6) years, the surviving entity in the merger will indemnify each director or officer of Appia for any and all actions taken by those individuals prior to the Effective Time to the fullest extent and in the same manner as Appia provided indemnification to those directors and officers pursuant to Appia's Certificate of Incorporation and director and officer indemnity agreements in effect at the Effective Time. In addition, if any indemnified person becomes a defendant in any actual action, the surviving entity will advance to that individual his or her legal and other expenses consistent with the terms and conditions for that advancement under Appia's indemnification provisions and agreements. Under the terms of the merger agreement, Appia will purchase an extended reporting period endorsement under Appia's existing directors' and officers' liability insurance coverage for Appia's directors and officers that will provide such directors and officers with coverage for six (6) years following the Effective Time, and the cost of such insurance will be treated as an expense of Appia in connection with the transaction and therefore will be deducted from the consideration available to the Appia stockholders.

Approval of the Merger by Appia Stockholders

The merger and the other transactions contemplated by the merger agreement will require the approval of the holders of (y) at least a majority of the outstanding shares of Appia common stock and Appia preferred stock (voting on an as converted to Appia common stock basis) and (z) two-thirds of the outstanding shares of Appia Series A Preferred Stock, Appia Series B Preferred Stock, Appia Series C Preferred Stock and Appia Series D Preferred Stock (voting on an as converted to Appia common stock basis), in each case entitled to vote upon the merger and the other transactions contemplated by the merger agreement. For further information, see "Ancillary Agreement – The Support Agreement" on page 98.

Directors of Mandalay Digital Following the Merger

Upon the close of the merger, Mandalay Digital will appoint Jud Bowman, Appia's founder and chief executive officer, and Craig Forman, Appia's Executive Chairman, as members to the board of directors. Below is further information about Messrs. Bowman and Forman.

Jud Bowman is the founder of Appia, and has served as its Chief Executive Officer since the company's inception in 2008. Prior to Appia, Mr. Bowman co-founded Motricity (Nasdaq: MOTR) in September 1999 and served as the Chief Technology Officer until 2008. Mr. Bowman is currently on leave from Stanford University, where he was named a President's Scholar, and is a graduate of the North Carolina School of Science and Mathematics.

Craig Forman currently serves as the Executive Chairman of the board of directors of Appia. From March 2006 to May 2009, Mr. Forman previously served as President of Value Added Services and subsequently as President of Access & Audience at EarthLink Inc. where he also led such shared services as Operations, Information Technology and Customer Support. Prior to joining EarthLink, Mr. Forman was a senior executive at Yahoo Inc., where he had served since February 2004 as head of that Internet portal company's Media and Information businesses. Mr. Forman is currently the non-executive chairman of Success Television, a company he founded before joining EarthLink, and a director of McClatchy Co., a leading U.S. newspaper and information company (NYSE: MNI) and Yellow Media Ltd. (TSE:Y), a leading Canadian digital media and marketing solutions company. Mr. Forman has a master's degree in law from Yale and an undergraduate degree from Princeton.

Indemnification

Insofar as indemnification for liabilities arising under the Securities Act may be permitted to directors, officers, or persons controlling our company, we have been informed that in the opinion of the SEC such indemnification is against public policy as expressed in the Securities Act and is therefore unenforceable.

NASDAQ Listing and Trading

We will apply to have the shares of our common stock to be issued in the merger approved for listing on the NASDAQ Stock Market, where our common stock is currently traded. Following the merger, Appia stockholders who receive shares of our common stock and wish to trade their shares on the NASDAQ Stock Market should contact such stockholders' financial or brokerage institution for assistance in making the necessary arrangements. Certain limitations may apply to making such arrangements. Appia stockholders who become holders of our common stock are responsible for and must bear all costs arising in connection with the above arrangements.

SEC Reports and Reporting Requirements

As a U.S. public company registered under the Exchange Act and whose common stock is listed on the NASDAQ Stock Market, we are, and we will continue to be, subject to current quarterly and annual financial reporting requirements and we file, and will continue to file, public reports with the SEC that are accessible (in English) on our website at www.mandalaydigital.com and at the SEC's website at www.sec.gov.

All of our stockholders who, directly or indirectly, own more than 5% of the total outstanding shares of our common stock are required to report such ownership to the SEC pursuant to the Exchange Act. These filings are made public by the SEC. Also, directors, certain officers, and holders of 10% or more of our common stock are subject to the insider trading reporting requirements of Section 16 of the Exchange Act.

PRINCIPAL STOCKHOLDERS OF COMBINED COMPANY

The following table and the related notes present certain information with respect to the beneficial ownership of the combined company upon consummation of the merger, by:

- each person expected to be a director or executive officer of the combined company (Peter Adderton is not standing for reelection as a director and therefore is excluded);
- Each person or group who is expected by the management of the combined company to become the beneficial owner of
 more than 5% of the common stock of the combined company upon the consummation of the merger, and
- all persons expected to be directors or executive officers of the combined company as a group.

The percentage of beneficial ownership calculation assumes that the merger closes on January 31, 2015 (for further information, see "The Merger – Consideration Payable in the Merger" on page 68) and is based on total outstanding shares of 56,853,613 which is the sum of (i) 37,797,221 shares of common stock outstanding as of November 13, 2014 and (ii) 19,056,392 shares that would be issued to the Appia stockholders in the merger based on a closing date of January 31, 2015, and which amount (x) includes 1,000,000 shares placed in the escrow, and (y) excludes 277,376 shares underlying options that Mandalay Digital would assume based on such closing date. The information below assumes: (i) that no options held by Appia stockholders are exercised upon any vesting of such options at the Effective Time of the merger in accordance with the terms of the merger agreement, (iii) that no additional shares of Mandalay Digital common stock or securities convertible into Mandalay Digital common stock are issued by Mandalay Digital subsequent to November 13, 2014 other than in connection with the merger, and (iv) that each director's and executive officer's beneficial ownership of Mandalay Digital common stock does not change prior to consummation of the merger.

Beneficial ownership is determined in accordance with the rules of the SEC. These rules generally provide that a person is the "beneficial owner" of securities if such person has or shares the power to vote or direct the voting of securities, or to dispose or direct the disposition of securities or has the right to acquire such powers within 60 days. In computing the number of shares beneficially owned by a person and the percentage ownership of that person, shares of common stock subject to options, warrants or rights held by that person that are currently exercisable or convertible or exercisable, convertible or issuable within 60 days of January 31, 2015, are deemed outstanding. Such shares, however, are not deemed outstanding for the purpose of computing the percentage ownership of any other person. The percentages in the table have been rounded to the nearest hundredth. Except as specifically indicated in the footnotes to this table, we believe that the persons identified in this table have sole voting and investment power with respect to all shares of common stock shown as beneficially owned by them, subject to community property laws where applicable.

Name of Beneficial Owner (1)	Number of Shares Beneficially Owned	Percentage Owned (%)
5% Stockholders (Excluding Directors and Officers)	Deficiency Owned	Owned (70)
Trident Capital, Inc. (2)(3)	5,619,341	9.9%
Venrock (2)(4)	4,760,964	8.4%
Adage Capital Partners LP (5)	2,913,891	5.1%
Expected Directors and Executive Officers		
Peter Guber (6)	4,122,724	7.2%
Judson S. Bowman (2)	2,520,120	4.4%
Paul Schaeffer (7)	488,454	*
William Stone (8)	418,331	*
Andrew Schleimer (9)	260,000	*
Robert Deutschman (10)	204,079	*
Christopher Rogers	112,263	*
Kirstie Brown (11)	97,317	*
Jeffrey Karish	93,454	*
Craig I. Forman (2)(12)	105,706	*
All Expected Directors and Executive Officers as a Group		
(10 individuals) (13)	8,422,448	14.5%

^{*} Represents less than 1%.

- (1) Except as otherwise indicated, the address of each of the persons listed above is c/o Mandalay Digital Group, Inc., 2811 Cahuenga Boulevard West, Los Angeles, CA 90068.
- (2) For purposes of this table, the calculation of shares that would be received by Appia securityholders at the closing of the merger assumes (i) that the average of the closing price of Mandalay Digital common stock over the 30 day period as of 3 days prior to closing is equal to the closing price on November 13, 2014, which was \$3.44, and (ii) that each holder of Series A Preferred Stock of Appia converts into common stock immediately prior to closing of the merger. For a more detailed description of how the price per share will be calculated, please see the section entitled "Summary of the Merger Agreement" at page 80.
- Upon the consummation of the merger, entities affiliated with Trident Capital, Inc. will be the beneficial owner of 5,619,341 shares of Mandalay Digital common stock. These shares include: (i) 5,463,931 shares of Mandalay Digital common stock to be issued to Trident Capital Fund-VII, L.P. ("Trident VII") in respect of 1,909,976 shares of Appia Series B Preferred Stock currently owned by Trident VII; and (ii) 155,410 shares of Mandalay Digital common stock to be issued to Trident Capital Fund-VII Principals Fund, L.L.C. ("Trident Principals VII") in respect of 54,325 shares of Appia Series B Preferred Stock owned by Trident Principals VII. Trident Capital Management-VII, L.L.C. ("TCM-VII") is the sole general partner of Trident VII and the sole Managing Member of Trident Principals VII. The managing members of TCM-VII are Donald R. Dixon, John H. Moragne and Arneek Multani (collectively, the "Managers"), together in the case of certain such individuals with their respective

- family planning vehicles. The Managers of TCM-VII share voting and investment power with respect to the shares held by each fund. Each Manager disclaims beneficial ownership of the shares held by such funds except to the extent of the Manager's respective economic interest in the funds. The address of the entities affiliated with Trident Capital, Inc. is 505 Hamilton Avenue, Suite 200, Palo Alto, CA 94301.
- Upon the consummation of the merger, entities affiliated with Venrock will be the beneficial owner of 4,760,964 shares of Mandalay Digital common stock which consists of shares to be issued in connection with the following: (i) 125,063 shares of Appia's Series A Preferred Stock held by Venrock Associates VI, L.P., (ii) 1,250,620 shares of Appia's Series C Preferred Stock held by Venrock Associates VI, L.P., (iii) 9,819 shares of Appia's Series A Preferred Stock held by Venrock Partners VI, L.P., and (iv) 98,194 shares of Appia's Series C Preferred Stock held by Venrock Partners VI, L.P. The address of the entities affiliated with Venrock is 3340 Hillview Avenue, Palo Alto, CA 94304.
- Consists of (i) 2,413,891 shares of common stock, and (ii) 500,000 shares of common stock issuable upon exercise of warrants at an exercise price of \$3.50 per share. Based on a Schedule 13G/A filed with the SEC on March 14, 2012, Adage Capital Partners GP, L.L.C. is the general partner of Adage Capital Partners, L.P., and as such has discretion over the portfolio securities beneficially owned by the Adage Capital Partners, L.P., Adage Capital Advisors, L.L.C., is the managing member of Adage Capital Partners GP, L.L.C., and Robert Atchinson and Phillip Gross are the managing members of Adage Capital Partners GP, L.L.C. and Adage Capital Advisors, L.L.C. and general partners of Adage Capital Partners, L.P., all of which share voting and dispositive power over the shares. The address for the reporting stockholders is 200 Clarendon Street, 52nd Floor, Boston, MA 02116.
- The Guber Family Trust (the "Trust"), of which Peter Guber serves as a trustee, is the beneficial owner of: (i) 3,120,343 shares of common stock, (ii) 35,714 shares of common stock issuable upon exercise of warrants at an exercise price of \$3.50 per share, and (iii) 666,667 shares of common stock issuable upon exercise of warrants at an exercise price of \$1.25 per share. Mr. Guber disclaims beneficial ownership of the shares of common stock directly and beneficially owned by the Trust, except to the extent of his pecuniary interest therein. Mr. Guber directly beneficially owns 200,000 shares of common stock and options to purchase 100,000 shares of common stock at an exercise price of \$13.75 per share.
- 200,000 shares are held by Mr. Schaeffer and 228,454 shares are held by the Paul and Judy Schaeffer Living Trust for which Mr. Schaeffer serves as a trustee. Mr. Schaeffer disclaims beneficial ownership of the securities held by the trust except to the extent of his pecuniary interest therein. Mr. Schaeffer directly owns options to purchase 60,000 shares of our common stock exercisable at a price of \$13.75 per share, which options are fully vested. Mr. Schaeffer will have the sole power to vote or to direct the vote and to dispose or to direct the disposition of 60,000 shares of common stock underlying the options granted to Mr. Schaeffer on June 18, 2008 upon their exercise.
- (8) Includes options to purchase 141,666 shares of common stock at an exercise price of \$2.54 per share.
- (9) Includes options to purchase 60,000 shares of common stock at an exercise price of \$2.75 per share.
- (10) Includes 134,079 shares held by the Robert and Ellen Deutschman Family Trust. Mr. Deutschman disclaims beneficial ownership of such shares except to the extent of his pecuniary interest therein.
- (11) Includes options to purchase 41,666 and 20,834 shares of common stock at an exercise price of \$4.50 and \$2.83, respectively.
- (12) Upon the consummation of the merger, Mr. Forman will be the beneficial owner of 105,706 shares of Mandalay Digital common stock which consists of shares to be issued in connection with the following: (i) 196,5000 shares of Appia common stock issued under restricted stock award agreements, of which 110,000 will be fully vested at the Effective Time; and (ii) options to purchase 46,968 shares of Appia common stock at an exercise price of \$0.28 per share which will be fully vested at the Effective Time.
- (13) Includes shares underlying options and warrants held by the expected directors and executive officers of the combined company included in the group.

MATERIAL U.S. FEDERAL INCOME TAX CONSEQUENCES

The following discussion summarizes the material U.S. federal income tax consequences of the merger that may be relevant to U.S. holders (as defined below) of shares of Appia stock who exchange their Appia stock for Mandalay Digital common stock in connection with the merger. This discussion is based upon the Code, Treasury regulations, administrative rulings and judicial decisions currently in effect, all of which are subject to change and to differing interpretations, possibly with retroactive effect. Any such change could affect the accuracy of this discussion. Subject to the qualifications, assumptions and limitations contained herein, the statements of law and legal conclusions contained under this caption "Material U.S. Federal Income Tax Consequences" regarding the material U.S. federal income tax consequences of the merger, constitute the opinions of Latham & Watkins LLP and Goodwin Procter LLP, respectively.

The discussion is limited to Appia stockholders who hold their Appia stock, and will hold their Mandalay Digital common stock, as "capital assets" within the meaning of Section 1221 of the Code (generally, property held for investment). In addition, this discussion does not address any U.S. federal estate or gift tax consequences, nor any state, local or foreign tax consequences, of the merger. This discussion also does not address tax considerations applicable to a holder's particular circumstances or to holders that may be subject to special tax rules, including, without limitation:

- banks, insurance companies or other financial institutions;
- real estate investment trusts;
- regulated investment companies;
- tax-exempt organizations;
- entities or arrangements that are treated as partnerships for U.S. federal income tax purposes and investors in such partnerships;
- S corporations or investors in an S corporation;
- U.S. expatriates;
- persons that are not U.S. holders (as defined below);
- non-U.S. persons;
- broker-dealers;
- traders in securities that elect mark-to-market treatment;
- dealers in securities or foreign currency;
- holders liable for the alternative minimum tax;
- holders that hold Appia stock as part of a straddle, hedge against currency risk, constructive sale or conversion transaction;
- holders that exercise dissenters' rights;
- U.S. holders that have a functional currency other than the U.S. dollar;
- holders that acquired their Appia stock through the exercise of options or otherwise as compensation or through a taxqualified retirement plan; or
- holders of options or restricted shares granted under any Appia benefit plan.

Furthermore, this discussion does not consider the potential effects of any state, local or foreign tax laws or the Medicare contribution tax on net investment income. This summary does not purport to be a complete analysis of the potential tax considerations relating to the merger, and is not tax advice. Therefore, you are urged to consult your tax advisors with respect to the application of the U.S. federal income tax laws to your particular situation, as well as any tax consequences of the merger arising under the federal estate or gift tax rules, the laws of any state, local, foreign or other taxing jurisdiction or any applicable tax treaty.

For purposes of this discussion, a "U.S. holder" is a beneficial owner of Appia stock who is, for U.S. federal income tax purposes: (i) an individual who is a citizen or resident of the United States; (ii) a corporation, or other entity taxable as a corporation for U.S. federal income tax purposes, created or organized under the laws of the United States or any of its political subdivisions; (iii) an estate that is subject to U.S. federal income tax on

its income regardless of its source; or (iv) a trust (A) if a U.S. court is able to exercise primary supervision over its administration and one or more U.S. persons have the authority to control all substantial decisions of the trust or (B) that has made a valid election to be treated as a United States person for U.S. federal income tax purposes.

If a partnership (or other entity treated as a partnership for U.S. federal income tax purposes) holds Appia stock, the tax treatment of a partner in the partnership generally will depend upon the status of the partner and the activities of the partnership. If you are a partnership or a partner of a partnership holding Appia stock, you should consult your tax advisor regarding the tax consequences of the merger to you.

Tax Opinions

Mandalay Digital and Appia intend for the merger to qualify as a reorganization within the meaning of Section 368(a) of the Code. It is a condition to the closing of the merger that Latham & Watkins LLP, counsel to Mandalay Digital, and Goodwin Procter LLP, counsel to Appia, deliver opinions, dated as of the closing date, to the effect that the merger will constitute a reorganization under Section 368(a) of the Code. If either Mandalay Digital or Appia waives this opinion condition after the registration statement of which this proxy statement/prospectus is a part is declared effective by the SEC, and the change in tax consequences is material, Mandalay Digital will recirculate an updated version of this proxy statement/prospectus and resolicit proxies from the Mandalay Digital or Appia stockholders, as applicable. These tax opinions will be based on customary assumptions, including assumptions regarding the absence of changes in existing facts and the consummation of the merger strictly in accordance with the merger agreement and this proxy statement/prospectus, and the truth and accuracy, as of the completion of the merger, of certain representations and covenants made in representation letters by Mandalay Digital, Merger Sub and Appia (which we refer to collectively as the "Representations and Assumptions"). If any of these Representations and Assumptions is inaccurate in any way, or any of the covenants are not complied with, these opinions could be adversely affected. Neither Mandalay Digital nor Appia has requested a ruling from the IRS with respect to any of the U.S. federal income tax consequences of the merger, and these tax opinions will not bind the courts or the IRS, nor will they preclude the IRS from adopting a position contrary to those expressed in the opinions.

U.S. Federal Income Tax Consequences of the Merger

Subject to the Representations and Assumptions, the merger will qualify as a reorganization within the meaning of Section 368(a) of the Code. Assuming that the merger so qualifies, the U.S. federal income tax consequences will be as described below.

Except as discussed below, a U.S. holder of Appia stock generally will recognize no gain or loss upon the receipt of Mandalay Digital common stock for their Appia stock surrendered in the merger. However, each U.S. holder of Appia stock will be treated as receiving in the merger (in addition to Mandalay Digital common stock), an amount of cash equal to the sum of (i) such U.S. holder's share of the cash set aside to serve as the expense fund to pay and reimburse the stockholder representative for any third party expenses incurred in connection with the stockholder representative performing its duties pursuant to the merger agreement, and (ii) any transfer taxes otherwise payable by such U.S. holder in respect of the merger that are instead paid by Mandalay Digital. Each U.S. holder of Appia stock will recognize gain, but not loss, for U.S. federal income tax purposes in an amount equal to the lesser of (a) the amount of cash such U.S. holder is treated as having received in the merger or (b) the amount, if any, by which the sum of (1) the fair market value, as of the effective time of the merger, of any Mandalay Digital common stock such U.S. holder receives in the merger and (2) the amount of cash such U.S. holder is treated as having received in the merger, exceeds such U.S. holder's adjusted tax basis in its shares of Appia common stock surrendered in the merger. Any gain recognized generally will be treated as long-term capital gain if the U.S. holder's holding period for the Appia stock surrendered exceeds one year at the effective time of the merger, except for gain treated as a dividend as discussed below under "—Potential Treatment of Cash as a Dividend."

The U.S. holder's aggregate adjusted tax basis in the Mandalay Digital common stock received in the merger will be equal to the U.S. holder's aggregate adjusted tax basis in its Appia stock surrendered for the Mandalay Digital common stock, reduced by the amount of cash treated as having been received in exchange for shares of Appia stock in the merger and increased by the amount of gain recognized in the exchange or any amount treated as a dividend, as described below. The holding period for the Mandalay Digital common stock will include the period during which the shares of Appia stock were held.

For the purposes of the above discussion of basis and holding periods for shares of Appia stock and Mandalay Digital common stock, stockholders who acquired different blocks of Appia stock at different times for different prices must calculate their basis and holding periods separately for each identifiable block of such stock exchanged or received in the merger.

Potential Treatment of Cash as a Dividend

Gain recognized by a U.S. holder in the merger as described above may be treated as a dividend for U.S. federal income tax purposes to the extent of the U.S. holder's ratable share of Appia's accumulated "earnings and profits" and possibly also Mandalay Digital's accumulated "earnings and profits." In general, the determination of whether such gain recognized will be treated as capital gain or a dividend depends upon whether and to what extent the exchange reduces the U.S. holder's deemed percentage of stock ownership of Mandalay Digital. For purposes of this determination, the U.S. holder generally will be treated as if it first exchanged all of its shares of Appia stock solely for Mandalay Digital common stock and then Mandalay Digital immediately redeemed a portion of the Mandalay Digital common stock in exchange for the cash the U.S. holder was treated as having actually received, which redemption we refer to in this proxy statement/prospectus as the "deemed redemption." Such gain recognized by a U.S. holder pursuant to the deemed redemption will be treated as capital gain if the deemed redemption is (i) "substantially disproportionate" with respect to the U.S. holder (and after the deemed redemption the U.S. holder actually or constructively owns less than 50% of the voting power of the outstanding Mandalay Digital common stock) or (ii) not "essentially equivalent to a dividend."

The deemed redemption generally will be "substantially disproportionate" with respect to a U.S. holder if the percentage of the outstanding Mandalay Digital common stock the U.S. holder actually and constructively owns immediately after the deemed redemption is less than 80% of the percentage of the outstanding Mandalay Digital common stock that the U.S. holder is deemed actually and constructively to have owned immediately before the deemed redemption. The deemed redemption will not be considered to be "essentially equivalent to a dividend" if it results in a "meaningful reduction" in the U.S. holder's deemed percentage of stock ownership of Mandalay Digital. In applying the above tests, the U.S. holder may, under the constructive ownership rules, be deemed to own stock that is owned by other persons or otherwise in addition to the stock the U.S. holder actually owns or owned. The IRS has ruled that a minority shareholder in a publicly held corporation whose relative stock interest is minimal and who exercises no control with respect to corporate affairs is considered to have a "meaningful reduction" if the shareholder has at least a relatively minor reduction in such shareholder's percentage of stock ownership under the above analysis.

As these rules are complex and dependent upon your specific circumstances, you should consult your tax advisor to determine whether you may be subject to these rules.

Certain Tax Reporting Rules

Under applicable Treasury regulations, "significant holders" of Appia stock generally will be required to comply with certain reporting requirements. An Appia stockholder should be viewed as a "significant holder" if, immediately before the merger, such holder held 1% or more, by vote or value, of the total outstanding Appia stock or Appia stock with a tax basis of \$1 million or more. Significant holders generally will be required to file a statement with the holder's U.S. federal income tax return for the taxable year that includes the consummation of the merger. That statement must set forth the holder's tax basis in, and the fair market value of, the shares of

Appia stock surrendered pursuant to the merger (both as determined immediately before the surrender of shares), the date of the merger, and the name and employer identification number of Mandalay Digital, Appia and Merger Sub and the holder will be required to retain permanent records of these facts. You should consult your tax advisor as to whether you may be treated as a "significant holder."

SUMMARY OF THE MERGER AGREEMENT

Below is a summary of the merger agreement, which is attached to this proxy statement/prospectus as *Appendix A* and is incorporated by reference into this document. You should read the merger agreement in addition to this Summary. The merger agreement, not this summary, is the legal document which governs the merger. Capitalized terms used in this section and not otherwise defined in this proxy statement/prospectus have the meanings ascribed to them in the merger agreement.

Cautionary Statement Concerning Representations and Warranties Contained in the Merger Agreement

You should not rely upon the representations and warranties contained in the merger agreement or the descriptions of such representations and warranties in this proxy statement/prospectus, as factual information about Mandalay Digital or Appia. In reviewing the agreements included or incorporated by reference in this proxy statement/prospectus, it is important to note that they are included to provide investors and stockholders with information regarding their terms, and are not intended to provide any other factual or disclosure information about Mandalay Digital or Appia or the other parties to the agreements. These representations and warranties have been made solely for the benefit of the other parties to the merger agreement, and were not intended to be and should not be relied upon by stockholders of Mandalay Digital or Appia; should not be treated as categorical statements of fact, but rather as a way of allocating risk between the parties; have in some cases been qualified by disclosures that were made to the other party in connection with the negotiation of the merger agreement, which disclosure schedules are not reflected in the merger agreement or disclosed in this proxy statement/prospectus; may apply standards of materiality in a way that is different from what may be material to investors; and were made only as of the date of the merger agreement or such other date or dates as may be specified in the merger agreement and are subject to more recent developments.

Accordingly, these representations and warranties may not describe the actual state of affairs as of the date they were made or at any other time. Information about Mandalay Digital can be found elsewhere in this proxy statement/prospectus and in other public filings Mandalay Digital makes with the SEC. Information about Appia can be found elsewhere in this proxy statement/prospectus.

General

The Boards of Directors of both Mandalay Digital and Appia have adopted and approved the merger agreement, which provides for the acquisition by Mandalay Digital of Appia through the merger. The merger will result in DTM Merger Sub, Inc., a Delaware corporation and wholly-owned subsidiary of Mandalay Digital, merging with and into Appia, so that Appia, as the surviving corporation, will become a wholly-owned subsidiary of Mandalay Digital. The surviving corporation will be renamed Digital Turbine Media, Inc. In exchange for their shares of Appia's common stock, preferred stock, and/or options to purchase shares of Appia common stock, the Appia stockholders will receive shares of Mandalay Digital common stock and/or options to purchase shares of Mandalay Digital common stock. Former Appia stockholders will not have any equity ownership interest in the surviving corporation.

Conversion of Securities in the Merger

The Aggregate Stock Consideration is equal to \$100,000,000 minus (i) the deemed value of 1,000,000 Mandalay Digital shares to be deposited in escrow (\$4,500,000) plus (ii) the aggregate exercise price of all options to purchase shares of Appia common stock plus (iii) the Adjustment Amount (which can be a negative

number). For purposes of the merger agreement, the "Adjustment Amount" means an amount, which may be a positive or a negative number, equal to (A) the estimated Working Capital as of the closing minus \$1,350,000 (the working capital target), minus (B) the estimated Net Debt as of the closing. The number of shares of Mandalay Digital common stock to be issued is determined by dividing the Aggregate Stock Consideration by \$4.50 (the "Price Per Share"). As discussed in detail below, the aggregate number shares of Mandalay Digital common stock to be issued in connection with the merger will vary depending on a number of factors, including the amount of the estimated Net Debt at closing.

At the effective time of the merger, each issued and outstanding share of Appia preferred stock (if not otherwise converted into Appia common stock) will be cancelled and extinguished and will be converted automatically into the right to receive the consideration set forth below without interest:

- Each outstanding share of Series A Preferred Stock (other than any Dissenting Shares (as defined in the merger agreement) and Treasury Shares (as defined in the merger agreement)) will be converted automatically into the right to receive the Per Share Common Consideration (as defined in the merger agreement) and its pro rata portion of shares released from escrow;
- Each outstanding share of Series B Preferred Stock (other than any Dissenting Shares and Treasury Shares) will be converted automatically into the right to receive the Per Share Series B Preferred Consideration (as defined in the merger agreement);
- Each outstanding share of Series C Preferred Stock (other than any Dissenting Shares and Treasury Shares) will be converted automatically into the right to receive the Per Share Series C Preferred Consideration (as defined in the merger agreement);
- Each outstanding share of Series D Preferred Stock (other than any Dissenting Shares and Treasury Shares) will be converted automatically into the right to receive the Per Share Series D Preferred Consideration (as defined in the merger agreement); and
- Each outstanding share of Series FF Preferred Stock (other than any Dissenting Shares and Treasury Shares) will be converted automatically into the right to receive the Per Share Common Consideration.

At the effective time of the merger, each issued and outstanding share of Appia common stock (other than any shares of Appia restricted common stock, Dissenting Shares and Treasury Shares) will be cancelled and converted into the right to receive the Per Share Common Consideration and its pro rata portion of shares released from escrow which shall be subject to adjustment and in accordance with the terms of Appia's restated charter. Each issued and outstanding share of Appia capital stock that is held in the treasury of Appia or owned by Appia, Mandalay Digital or DTM Merger Sub, Inc. will be automatically cancelled and will receive no consideration.

Each issued and outstanding share of the common stock of DTM Merger Sub, Inc. will be converted into and become one fully paid and nonassesable share of common stock of Digital Turbine Media, Inc. (the surviving corporation in the merger).

For the purpose of calculating the Per Share Common Consideration, the Price Per Share (\$4.50 per share) and the shares of Mandalay Digital common stock issuable to the holders of Appia common stock, Appia preferred stock, Appia options, Appia restricted common stock and Series C Warrants (as defined below), adjustments will be made to reflect the effect of any stock split, split-up, reverse stock split, dividend or distribution of securities convertible into Appia common stock, Appia preferred stock or Mandalay Digital common stock, reorganization, recapitalization, reclassification or other like change with respect to Appia common stock, Appia preferred stock or Mandalay Digital common stock having a record date occurring on or after the date of the merger agreement and prior to the effective time of the merger. The merger agreement provides that in no event will the number of shares of Mandalay Digital common stock being issued for all

outstanding shares of Appia capital stock, all Appia options and all Series C Warrants be greater than the Parent Stock Closing Consideration except as a result of adjustments.

The aggregate number shares of Mandalay Digital common stock to be issued in connection with the merger will vary depending on a number of factors, including when the closing occurs and the amount of the Net Debt at closing.

(in thousands, except per-share amounts):

	Pro		
	Forma	Forecast	Forecast
	9/30/2014	1/31/2015	3/31/2015
Enterprise Value	\$100,000	\$100,000	\$100,000
Plus: Working Capital Adjustment	(172)	50	(234)
Less: Net Debt	(10,291)	(13,204)	(14,631)
Plus: Aggregate Exercise Price of Appia Options	156	156	156
Aggregate Consideration	\$ 89,692	\$ 87,002	\$ 85,291
Price Per Share	\$ 4.50	\$ 4.50	\$ 4.50
Aggregate MNDL Shares - fully diluted	19,932	19,334	18,953

The allocation of shares of Mandalay Digital common stock to be issued in the merger among stockholders of Appia will vary depending on a number of factors, including when the closing occurs, the amount of the Net Debt at closing and the final calculation of the closing price per share of the Mandalay Digital common stock. The final allocation of shares of Mandalay Digital common stock to be issued in the merger among stockholders of Appia will depend on the final calculation of the closing price per share of the Mandalay Digital Common Stock, which is referred to as the Closing Price Per Share in the merger agreement. The Closing Price Per Share shall be equal to the average of the closing prices of the Mandalay Digital common stock over the 30 day period ending three calendar days prior to the closing of the merger, which we refer to as VWAP, except however if the VWAP is greater than \$4.50, in which case the Closing Price Per Share shall be equal to \$4.50, or if the VWAP is less than \$3.50, in which case the Closing Price Per Share shall be equal to \$3.50.

Treatment of Series C Warrants

Series C Warrants include, under the merger agreement, the Series C Penny Warrants and Series C Other Warrants under the Preferred Stock Purchase Warrants dated April 4, 2013 and October 31, 2013. The treatment of Series C Warrants under the merger agreement is as follows:

- Each Series C Penny Warrant issued and outstanding immediately prior to the effective time will terminate and be cancelled and be converted into the right to receive an amount equal to the Series C Penny Warrants Consideration.
- Each Series C Other Warrant issued and outstanding immediately prior to the effective time will terminate and be cancelled and be converted automatically into the right to receive an amount equal to the Series C Other Warrants Consideration.
- Prior to the effective time, Appia has agreed to take all necessary actions to effect the termination of the Series C Warrants
 at or after the effective time.

Treatment of Appia Options and Restricted Common Stock

Immediately following the effective time of the merger, each share of Appia restricted stock and each Appia option held by a non-employee director of Appia as of November 13, 2014, will vest in full and become

exercisable. At the effective time of the merger, each Appia option that is outstanding at the effective time and that is held by a holder who is (a) an employee of Appia or PocketGear and continues his or her employment with Mandalay Digital or one of its subsidiaries or remains or becomes an employee of the surviving corporation or (b) an independent contractor or consultant of Appia or PocketGear who continues its service with Appia, PocketGear or Mandalay Digital, other than a service provider who is not eligible to be included on the Form S-8 registration statement, will be assumed by Mandalay Digital subject to the same terms, conditions and restrictions. Following the effective time, (a) options to purchase Appia common stock will convert to options to purchase a number of shares of Mandalay Digital stock equal to the product of the number of Mandalay Digital shares subject to the outstanding Mandalay Digital option immediately prior to the effective time multiplied by the Option Exchange Ratio (as defined in the merger agreement), rounded down to the nearest whole number of shares of Mandalay Digital common stock, (b) the per share exercise price for Mandalay Digital common stock issuable upon exercise of each assumed option to purchase stock will be equal to the quotient determined by dividing the per share exercise price outstanding immediately prior to the effective time by the Option Exchange Ratio, rounded up to the nearest whole cent, and (c) all references to "Appia" in the 2008 Stock Incentive Plan of Appia and the stock option agreements will instead be references to Mandalay Digital.

Each outstanding, unexercised option to purchase Appia common stock that is vested and exercisable at the effective time and held by a holder (a) that does not continue his or her employment with Mandalay Digital or one of its subsidiaries or remains or becomes an employee of the Surviving Corporation or (b) an independent contractor or consultant of Appia or PocketGear who does not continues his or her service with Appia, PocketGear or Mandalay Digital, other than a service provider who is not eligible to be included on the Form S-8 registration statement (each, a "Deemed Exercise Option"), will be cancelled at the effective time in exchange for a right to receive, subject to any withholding obligations contemplated in the merger agreement, on or after the effective time, the number of shares of Mandalay Digital common stock equal to the quotient obtained by dividing (i) the product of (x) the excess, if any, of the Per Share Common Consideration Equivalent Amount (as defined in the merger agreement) over the applicable exercise price of such option to purchase Appia common stock, times (y) the number of Appia shares subject to the outstanding, unexercised vested option to purchase common stock immediately prior to the effective time, divided by (ii) \$4.50, rounded down to the nearest whole share. If the applicable exercise price of such option exceeds the Per Share Common Consideration, no payment will be made and the holder will have no further rights or interests from and after effective time. Each outstanding, unexercised Appia option that is unvested and unexercisable at the effective time and that is held by a holder that does not fall under (a) or (b) above will be cancelled at the effective time for no consideration and in no event will Mandalay Digital assume the Appia options.

At the effective time of the merger, each award of shares of Appia restricted common stock that is outstanding and unvested at the effective time of the merger (each, a "Restricted Stock Award") will be assumed by Mandalay Digital subject to the same terms, conditions and restrictions and (i) will cover a number of shares of Mandalay Digital common stock equal to the product of the number of Appia shares subject to the outstanding restricted stock immediately prior to the effective time, multiplied by the Per Share Common Consideration, rounded down to the nearest whole number of shares of Mandalay Digital common stock, (ii) a number of the Mandalay Digital common stock subject to the assumed Restricted Stock Award equal to the applicable pro rata allocation of escrowed shares will be held in the escrow account and (iii) all references to "Appia" in the 2008 Stock Incentive Plan of Appia and the stock option agreements will instead be references to Mandalay Digital. Mandalay Digital has agreed to prepare and file with the SEC a registration statement on Form S-8 and use its reasonable best efforts to have it declared effective as soon as reasonably practicable. Appia has agreed to take all necessary actions to effect the termination of the 2008 Stock Incentive Plan and the Appia options and restricted common stock.

Closing Consideration Schedule

Appia has agreed to deliver to Mandalay Digital, no later than five business days prior to the closing date, a "Closing Consideration Schedule" setting forth, as of closing:

- all Appia stockholders (except for those holding no shares other than Restricted Stock Awards) and their respective addresses, the number and type of shares of Appia capital stock held (including the respective certificate numbers and the class or series), the pro rata portion and the pro rata allocation of the escrowed shares applicable to each Appia stockholder, the number of shares of Mandalay Digital common stock to be issued to each Appia stockholder;
- all Series C Warrantholders (as defined in the merger agreement) entitled to receive consideration, the number of shares of Mandalay Digital common stock to be issued to each Series C Warrantholder;
- all Appia optionholders entitled to receive consideration, the number of Mandalay Digital options to be issued for the Appia options, the vesting schedule of the Appia options, the number of shares subject to the Appia options that are vested and outstanding, the number of shares subject to the Appia options that are unvested and outstanding, and the consideration to be paid, as applicable, to each Appia optionholder;
- all holders of Deemed Exercise Options (as defined in the merger agreement) entitled to receive consideration, and their respective
 addresses and social security numbers or tax identification numbers, if applicable, the number of shares of Appia common stock
 subject to each Deemed Exercise Option, the number of shares of Mandalay Digital common stock to be issued; and
- all holders of Restricted Stock Awards (as defined in the merger agreement) entitled to receive consideration, and their respective
 addresses and social security numbers or tax identification numbers, if applicable, the number of shares of Appia common stock
 subject to each Restricted Stock Award, the vesting schedule of the Restricted Stock Awards, the pro rata portion and pro rata
 allocation of the escrowed shares applicable in respect of each Restricted Stock Award, the number of shares of Mandalay Digital
 common stock to be issued in respect of each Restricted Stock Award.

Fractional Shares

No fractional shares of Mandalay Digital common stock will be issued in connection with the merger and no dividends or other distributions with respect to the Mandalay Digital common stock will be payable on or with respect to any fractional share.

Escrow of Consideration; Escrow Account

At the closing, Mandalay Digital, the stockholder representative and American Stock Transfer Company, the Escrow Agent, will enter into the escrow agreement pursuant to which Mandalay Digital will cause to be deposited 1,000,000 shares of Mandalay Digital common stock in an escrow account to be held for the satisfaction of any post-closing purchase price adjustments and any indemnification claims by Mandalay Digital. For purposes of satisfying claims for indemnification or any post-closing adjustment to the purchase price, all shares of Mandalay Digital common stock in the escrow account will be deemed to have a value equal to \$4.50 (the "Price Per Share"), as may be adjusted pursuant to the terms of the merger agreement. The Appia stockholders who contribute a portion of the aggregate consideration that they are entitled to receive under the terms of the merger agreement to the escrow will be the owners of the shares deposited in the escrow account and will be entitled to vote all escrowed shares on their own behalf.

In the event of any claim for indemnification under the merger agreement, the Parent Indemnitees (as defined below) will be first to recover any damages from the escrow account then, subject to certain limitations,

if the remaining shares are insufficient to cover damages, the Parent Indemnitees may seek recovery directly from each Appia stockholder on a pro rata basis. Pursuant to the merger agreement, within five days following the one-year anniversary of the closing date, Mandalay Digital and stockholder representative will instruct the Escrow Agent to deliver to each the applicable Appia stockholders his, her or its pro rata allocation of (i) Mandalay Digital shares of common stock remaining in the escrow account minus (ii) (a) pending claims determined as of the one-year anniversary divided by (b) the Price Per Share. See the section entitled "Summary of the Merger Agreement—Survival; Indemnification" below at page 88.

Working Capital/Net Debt Adjustment

No later than five (5) business days prior to the closing date, Appia will prepare and deliver to Mandalay Digital an unaudited statement of estimated Working Capital as of the Closing Date (the "Pre-Closing Statement"), which shall fairly present in all material respects the estimated Working Capital (the "Estimated Working Capital") and the estimated Net Debt (the "Estimated Net Debt") of Appia as of the closing date and of the Adjustment Amount, which Adjustment Amount will be used to calculate the Aggregate Stock Consideration as described above (see "Summary of the Merger Agreement—Conversion of Securities in the Merger" on page 81).

No later than 135 calendar days following the closing, Mandalay Digital has agreed to prepare and deliver to the Stockholder Representative an unaudited statement of Working Capital as of the Closing Date (the "Closing Statement") presenting Appia's Working Capital and Net Debt as of the Closing Date and setting forth the final Adjustment Amount.

If the final Adjustment Amount is a positive number, Mandalay Digital agrees to issue to each Appia Equityholder, its pro rata portion of such amount by check or wire transfer of immediately available funds to the Exchange Agent for distribution. If the final Adjustment Amount is a negative number, then Mandalay Digital and Stockholder Representative will execute and deliver joint written instructions to the Escrow Agent instructing distribution to Mandalay Digital from the Escrow Account of a number of shares of Mandalay Digital Common Stock equal to the final Adjustment Amount divided by the Price Per Share.

Appia Designees to Mandalay Digital Board of Directors

Mandalay Digital and Appia agree to cooperate to take all actions reasonably necessary so that immediately following the effective date: (a) the number of directors that will comprise all of the members of the board of directors of Mandalay Digital is increased as required and (b) two individuals currently serving on the Appia's Board of Directors, and designated by Appia within five calendar days following the date of execution of the merger agreement, are appointed to Mandalay Digital's Board of Directors. For more information about the designees, see "The Merger – Directors and Management of Mandalay Digital Following the Merger" on page 74.

Stockholder Representative

By executing the merger agreement and upon executing the written consent, the stockholders of Appia, Shareholder Representative Services LLC is appointed as the stockholder representative for purposes of taking certain actions and giving certain consents on behalf of any Appia stockholder relating to the merger agreement and the escrow agreement.

Stockholder Representative Expense Fund

Upon the closing, Appia will wire to the stockholder representative US\$25,000 to serve as the expense fund to pay and reimburse the stockholder representative for any third party expenses incurred in connection with the stockholder representative performing its duties pursuant to the merger agreement or the escrow agreement. Any losses incurred in connection with the stockholder representative's execution and performance of his duties

(including legal fees and related expenses), may be received from (i) the funds in the expense fund and (ii) to the extent any amounts remain at the end of the escrow period, amounts in the escrow account.

Closing and Effectiveness of the merger

The closing of the merger is to take place on the fifth business day following the satisfaction or, to the extent permitted under the merger agreement and by applicable law, waiver of all conditions to the obligations of the parties set forth in the merger agreement and described below (other than such conditions as may, by their terms, only be satisfied at the closing or on the Closing Date, but subject to such satisfaction or waiver thereof) (see "Summary of the Merger Agreement—Conditions to the Consummation of the Merger" on page 93).

Exchange of Appia Stock Certificates for the Merger Consideration

The merger agreement provides that Digital Turbine Media, Inc. will, cause American Stock Transfer & Trust Company, the exchange agent appointed by Mandalay Digital in connection with the merger, to mail within three business days following the effective time, to each record holder of Appia capital stock, a letter of transmittal and instructions for surrendering and exchanging the record holder's Appia share certificates for Mandalay Digital common stock. Upon surrender of an Appia share certificate for exchange to the exchange agent, together with a duly completed and validly executed letter of transmittal, and such other documents as the exchange agent may reasonably require, the record holder of the Appia certificate will be entitled to receive the number of uncertificated shares of Mandalay Digital common stock entitled to receive under the merger agreement and as set forth on the Closing Consideration Schedule. No interest will be paid on the merger consideration payable in connection with the merger agreement.

If any Appia share certificate has been lost, stolen or destroyed, upon the making of an affidavit of that fact by the person claiming such certificate to be lost, stolen or destroyed and, if required by Mandalay Digital or the exchange agent, the delivery by such person of an indemnification agreement acceptable to Mandalay Digital and the exchange agent, or if required by the exchange agent, posting by such person of a bond in such amount as Mandalay Digital or the exchange agent may determine is reasonably necessary as indemnity against any claim that may be made against it with respect to such certificate, the exchange agent will deliver in exchange for such lost, stolen or destroyed certificate the consideration payable in respect thereof.

Mandalay Digital, Digital Turbine Media, Inc., the Exchange Agent and the Escrow Agent are entitled to deduct and withhold from consideration otherwise payable under the merger agreement to any Appia Equityholder or Person, amounts needed to be withheld with respect to making any such payment under the Code (as defined in the merger agreement) or any provision of state, local or foreign tax law.

Transfer Restrictions on Mandalay Digital Common Stock

All shares of Mandalay Digital common stock to be issued in connection with the merger shall be subject to restrictions on the transfer of such shares during the one year period following the closing date. The transfer restrictions expire (i) 180 days following the closing date with respect to 33 and 1/3% of each Appia Equityholder's Mandalay Digital common stock, (ii) 270 days following the closing date with respect to 33 and 1/3% of each Appia Equityholder's shares of Mandalay Digital common stock and (iii) 365 days following the Commencement Date with respect to 33 and 1/3% of each Appia Equityholder's Mandalay Digital common stock.

Appraisal Rights

Any shares of Appia company capital stock that are issued and outstanding immediately prior to the effective time of the merger and that have not approved the merger via an affirmative vote or written consent (or which the holder has not effectively withdrawn or lost such holder's appraisal rights), and with respect to which a

demand for appraisal has been properly made in accordance with the DGCL, will not be converted into or represent a right to receive the merger consideration otherwise payable with respect to such shares of Appia common stock and preferred stock. If any holder of dissenting shares has effectively withdrawn or loses its appraisal rights, such shares will receive the consideration set forth in the merger agreement. Appia has agreed to give Mandalay Digital (i) prompt notice of any written demand for appraisal received and (ii) the opportunity to participate in all negotiations and proceedings with respect to any demands made prior to closing. Communications between Appia and any Appia stockholder will require the consent of Mandalay Digital. Any excess dissenting share payments will be recoverable by Mandalay Digital. For purposes of distributing any portion of escrowed shares, no dissenting stockholder will participate in the escrow and any payment by Mandalay Digital that would have been required to be made to such holder of dissenting shares will be retained by Mandalay Digital.

Representations and Warranties

The merger agreement contains customary representations and warranties of the parties. These include representations and warranties of Appia with respect to, among other things: organization; the equity interests and liabilities of its subsidiary; capitalization; authorization; consents and approvals; absence of certain changes; facilities; condition and sufficiency of assets; contracts and commitments; permits; no conflict or violation; financial statements and other financial information; undisclosed liabilities; books and records; litigation; labor matters; compliance with law; intellectual property; employee benefit plans; transactions with certain persons; certain payments; tax matters; insurance; publishers; advertisers; accounts receivable; compliance with environmental laws; banking relationships; no brokers; disclosure documents; no other agreements to sell the assets or capital stock of Appia and vote required. The merger agreement also contains customary representations and warranties of Mandalay Digital and DTM Merger Sub, Inc., including among other things: organization; authorization; no conflict or violation; Mandalay Digital common stock; no brokers; SEC reports and financial statements; disclosure documents; Mandalay Digital capitalization; litigation;, absence of certain changes; compliance with law and intellectual property.

The representations, warranties and covenants made by Appia in the merger agreement are qualified by information contained in a disclosure memorandum delivered to Mandalay Digital and DTM Merger Sub, Inc. in connection with the execution of the merger agreement. The representations, warranties and covenants made by Mandalay Digital and DTM Merger Sub, Inc. in the merger agreement are also qualified by information contained in a disclosure memorandum delivered to Appia in connection with the execution of the merger agreement. Certain representations and warranties were made as of a specific date.

Survival; Indemnification

Survival of Representations and Warranties and Covenants

The representations and warranties of Appia, Mandalay Digital and DTM Merger Sub, Inc. contained in the merger agreement survive for twelve months following the closing date, other than representations and warranties (i) of Appia relating to organization, subsidiaries, capitalization, authorization and no brokers and of Mandalay Digital relating to organization, authorization and no brokers, each of which survive for six years following the closing date, and of Appia relating to tax matters, which shall survive until thirty days following the expiration of the applicable statute of limitations under applicable tax law (collectively, the "Fundamental Representations and Warranties"), and (ii) of Appia with respect to intellectual property, which shall survive for thirty months following the closing date (the "IP Representations").

Indemnification of Mandalay Digital

Pursuant to the terms of the merger agreement, the Appia Equityholders severally, and not jointly, on a pro rata basis, will indemnify, save and hold harmless Mandalay Digital, DTM Merger Sub, Inc., Appia and each

of Mandalay Digital's direct and indirect stockholders, affiliates and subsidiaries (including after the closing, the surviving corporation), and each of their respective representatives (the "Parent Indemnitees") against any and all costs, losses, taxes, liabilities, obligations, damages, lawsuits, deficiencies, claims, demands, expenses (whether or not arising out of third-party claims), reasonable attorneys' fees and all amounts paid in investigation, defense or settlement of any of the foregoing (collectively, "Damages") as a result of:

- 1. any breach of any representation or warranty or the inaccuracy of any representation made by Appia in or pursuant to the merger agreement (in each case without giving effect to materiality qualifications or Material Adverse Effect qualifications with respect to the calculation of Damages incurred by a Parent Indemnitee);
- 2. any breach of any covenant or agreement made by Appia in or pursuant to the merger agreement;
- 3. any claim by any person for brokerage or finder's fees or commissions or similar payments based upon any agreement or understanding alleged to have been made by any such person with Appia or any stockholder of Appia (or any person acting on their behalf) in connection with any transactions contemplated by the merger agreement;
- 4. any "Long Term Liabilities" that are unpaid as of the closing and not taken into account in the final calculation of "Closing Net Debt" on the Closing Statement;
- 5. any Company Transaction Expenses that are unpaid as of the Closing and not taken into account in the final calculation of Net Debt on the Closing Statement;
- 6. any Excess Dissenting Share Payments;
- 7. any claim or allegation that any Appia Equityholder or other Person is entitled to any amount in connection with the merger (other than as set forth in the merger agreement and in the Closing Consideration Schedule);
- 8. various taxes, including all taxes of Appia or PocketGear related to any Pre-Closing Tax Period; or
- 9. certain other matters set forth in the disclosure schedules.

Indemnification of the Appia Equityholders

Pursuant to the terms of the merger agreement, Mandalay Digital will indemnify, save and hold harmless the equityholders of Appia (the "Appia Equityholders"), from and against any and all Damages incurred in connection with, arising out of, resulting from or incident to (i) any breach of any representation or warranty or the inaccuracy of any representation made by Mandalay Digital or DTM Merger Sub, Inc. in or pursuant to the merger agreement (in each case without giving effect to materiality qualifications or Material Adverse Effect qualifications with respect to the calculation of Damages incurred by an Appia Equityholder Indemnitee); or (ii) any breach of any covenant or agreement made by Mandalay Digital or DTM Merger Sub, Inc. in or pursuant to the merger agreement.

Other Limitations

Neither Mandalay Digital, on the one hand, nor any of the Appia Equityholders, on the other hand, shall be liable for any Damages until the aggregate amount otherwise due to the Appia Equityholders or Parent Indemnitees, as the case may be, exceeds an accumulated total of \$350,000 (the "Basket"); provided, that this limitation shall apply only to the extent a party is entitled to indemnification exclusively with respect to any breach of any representation or warranty or the inaccuracy of any representation other than the Fundamental

Representations or in the case of fraud (which is to be interpreted to require the element of scienter). Once the aggregate amount of Damages exceeds the Basket, then the Appia Equityholders or Parent Indemnitees, as the case may be, shall have the right to recover all Damages from and including the first dollar of damages without regard to the Basket, subject to the other limitations set forth in the merger agreement.

Neither Mandalay Digital, on the one hand, nor any of the Appia Equityholders, on the other hand, shall be liable for any Damages due to the Appia Equityholders or Parent Indemnitees, as the case may be, in excess of \$4,500,000; provided, that this limitation shall apply only to the extent a party is entitled to indemnification exclusively with respect to any breach of any representation or warranty or the inaccuracy of any representation other than the Fundamental Representations, the IP Representation or in the case of fraud (which is to be interpreted to require the element of scienter). In addition, the aggregate liability of the Appia Equityholders in connection with indemnification claims for breaches of the IP Representations shall not exceed \$20,000,000 in the aggregate (including any amounts recovered from the escrow).

The aggregate liability of the Appia Equityholders in connection with indemnification claims pursuant to the merger agreement shall not exceed the total value of the Aggregate Consideration (valued at the Price Per Share) payable to the Appia Equityholders hereunder (including the amount in escrow); no Appia Equityholder shall be liable to any Parent Indemnitee for any Damages (as defined in the merger agreement) in excess, in the aggregate, of the total value of the Aggregate Consideration (valued at the Price Per Share) payable to such Appia Equityholder under the merger agreement (including any such Appia Equityholder's Pro Rata Allocation (as defined in the merger agreement) of the amount deposited in escrow), except in the case of fraud (which shall be interpreted to require the element of scienter) committed by an Appia Equityholder (and in the event of such fraud only with respect to the Appia Equityholder who committed such fraud). Mandalay Digital shall not be liable to the Appia Equityholder Indemnitees pursuant to the merger agreement for any Damages in excess, in the aggregate, of the total value of the Aggregate Consideration (valued at the Price Per Share) payable to the Appia Equityholders under the merger agreement (including the amount deposited in escrow).

Covenants of Mandalay Digital and Appia

Covenants Relating to the Conduct of Appia's Business

Appia has agreed that prior to the closing date, Appia will conduct its business in the ordinary course of business consistent with past practice; and Appia will not take any action inconsistent with the merger agreement or the consummation of the merger. Appia has also agreed that neither it nor PocketGear will take certain other actions during the period between the execution of the merger agreement and the closing date, subject to certain limited exceptions as set forth in the merger agreement, without the prior written consent of Mandalay Digital, including the following:

- issue, or commit to issue, any capital stock of Appia or PocketGear;
- amend any of the organizational documents of Appia or PocketGear;
- adopt or change any material tax or other accounting methods, principles or practices or change any annual tax accounting
 period; make or change any material tax election; settle or compromise any material claim, notice, audit report or assessment in
 respect of taxes; file any amended tax return; enter into any tax allocation agreement, tax sharing agreement, tax indemnity
 agreement or closing agreement relating to any tax; knowingly surrender any right to claim a tax refund; or consent to any
 extension or waiver of the statute of limitations period applicable to any material tax claim or assessment;
- revalue any assets of Appia, PocketGear or relating to their business;
- fail to take reasonable actions to prevent any damage, destruction or loss (whether or not covered by insurance) adversely affecting any of the assets of Appia, PocketGear or their business;

- acquire any corporation, partnership, limited liability company, other business organization or division thereof or any material
 amount of assets, or enter into any joint venture, strategic alliance, exclusive dealing, noncompetition or similar contract or
 arrangement;
- except for the merger, adopt a plan of complete or partial liquidation, dissolution, merger, consolidation, restructuring, recapitalization or other reorganization of Appia or of PocketGear, or otherwise alter Appia's or PocketGear's corporate structure:
- cancel any indebtedness or waive or release any material right or claim of Appia or PocketGear or relating to their business, except in the ordinary course of business consistent with past practice;
- (i) hire or terminate any employee, contingent worker or director, (ii) pay, announce, promise or grant, whether orally or in writing, increase or establish (each, as applicable) any wages, base pay, fees, salaries, compensation, bonuses, incentives, deferred compensation, pensions, severance or termination or change in control payments, retirement, profit sharing, fringe benefits, equity or equity-linked awards, employee benefit plans, or any other form of compensation or benefits payable by Appia or PocketGear, including without limitation, any increase or change pursuant to any Appia benefit plan (except as required by law), or (iii) enter into, adopt, materially amend or terminate any Appia benefit plan;
- take any action which could reasonably be expect to cause, or fail to take any reasonable action to prevent, any change in employee relations which has or is reasonably likely to have a material adverse effect on the productivity, the financial condition or, results of operations of Appia or PocketGear;
- enter into any collective bargaining agreement or other contracts with any labor union;
- enter into, amend, cancel or terminate any contract, lease or permit to which Appia or PocketGear is a party or otherwise relating to their business;
- enter into any contract with any related party of Appia or PocketGear;
- mortgage, pledge or otherwise encumber any assets of Appia or PocketGear or relating to their business, except purchase money mortgages arising in the ordinary course of business;
- sell, assign or transfer any assets of Appia or PocketGear or relating to their business;
- incur indebtedness of Appia or PocketGear for borrowed money, commit to borrow money, make or agree to make loans or guarantee indebtedness;
- incur liabilities of Appia or PocketGear, except liabilities incurred in the ordinary course of business, or increase or change any
 assumptions underlying or methods of calculating, any doubtful account contingency or other reserves of Appia or PocketGear;
- pay, discharge or satisfy any liabilities of Appia or PocketGear other than in the ordinary course of business;
- make any single capital expenditure relating to the business in excess of \$10,000 or capital expenditures that are, in the aggregate, in excess of \$50,000 incur any obligations to make any capital expenditures or execute any lease;
- fail to pay or satisfy when due any material liability of Appia, PocketGear or related to their business;

- except in the ordinary course of business, accelerate the collection of or discount any accounts receivable, delay the payment of
 accounts payable or defer expenses, reduce inventories or otherwise increase cash in hand;
- amend or otherwise take any action that would permit or cause any Appia options or restricted stock award to accelerate in contemplation of or as a consequence of the merger or other transactions contemplated by the merger agreement;
- fail to carry on diligently the business in the ordinary course so as to preserve for Mandalay Digital the business and the goodwill of the customers, publishers and advertisers of the business and others having business relations with Appia, PocketGear or their business;
- dispose, sell, transfer, assign, encumber, pledge, license, abandon, fail to maintain, or allow to lapse any intellectual property of Appia, PocketGear or relating to their business or dispose or disclose (or agree to disclose upon the occurrence of any event, non-event or default) to any person any intellectual property of Appia, PocketGear or relating to their business not theretofore a matter of public knowledge; or
- agree, whether orally or in writing, to do any of the things described in the preceding bullets other than as expressly permitted.

In addition, each party has covenanted to use its reasonable best efforts both to cause the merger to qualify as a "reorganization" within the meaning of Section 368(a) of the Code, including by not taking any action that such party knows could prevent such qualification, and to obtain certain tax opinions from their respective counsels that the merger will so qualify, including by providing tax representation letters containing certain statements and representations as requested by such counsels.

No Solicitation

Appia has agreed that until the closing of the merger or the earlier termination of the merger agreement, neither Appia nor any Appia stockholder will, and each of will cause each of their respective representatives (including investment bankers, attorneys and accountants), not to, directly or indirectly, enter into, solicit, initiate or continue any discussions or negotiations with, or encourage or respond to any inquiries or proposals by, or participate in any negotiations with, or provide any information to, or otherwise cooperate in any other way with, any person or group concerning any sale of all or a portion of Appia's assets, or of any capital stock of Appia, or any merger, consolidation, liquidation, dissolution or similar transaction involving Appia. Appia has agreed to advise any prospective purchaser or soliciting party, by written notice (with a copy to Mandalay Digital) of the terms of the no solicitation provisions and promptly notify Mandalay Digital (orally and in writing) if any such offer, or any inquiry or contact with any person is made. Appia has also agreed not to release any third party from, or waive any provision of, any confidentiality or standstill agreement to which Appia is a party.

Director and Officer Indemnification

For a period of six years following the closing of the merger, the surviving corporation will honor in all respects the obligations of Appia with respect to all rights to indemnification (including advancement of expenses) or exculpation existing in favor of, and all limitations on the personal liability of, any person who is now, or has been at any time prior to the signing of the merger agreement, or who becomes prior to the effective time of the merger, a director or officer of Appia or of Pocketgear under Appia's organizational documents or director or officer of Appia in any indemnification agreements in effect as of the date of the merger agreement. Prior to the effective time of the merger, Appia shall purchase (and pay in full all premiums on) an extended reporting period endorsement under Appia's existing directors' and officers' liability insurance coverage for Appia's directors and officers that provide such directors and officers with coverage for six years following the effective time of the merger.

Conditions to the Consummation of the Merger

Conditions to Each Party's Obligations to Effect the Merger

Conditions to Appia's obligation to effect the merger is subject to the satisfaction of several conditions (any of which may be waived in writing by Appia), including:

- all representations and warranties of Mandalay Digital and DTM Merger Sub, Inc. contained in the merger agreement (i) that are qualified by materiality or words of similar import shall be true and correct in all respects, as of the date of the merger agreement and at and as of the closing date, except that such representations and warranties that are made as of a specific date need only be true and correct as of such date, and (ii) that are not qualified by materiality shall be true and correct in all material respects at and as of the date of the merger agreement and at and as of the closing date, except that such representations and warranties that are made as of a specific date need only be true and correct as of such date;
- Mandalay Digital and DTM Merger Sub, Inc. shall have performed and satisfied in all material respects all agreements and covenants required to be performed by Appia and DTM Merger Sub, Inc. prior to or on the closing date;
- No suit, action, investigation, inquiry or other proceeding by any governmental authority or other person or legal or
 administrative proceeding shall have been instituted or threatened which seeks to restrain, enjoin, prevent the consummation of
 or otherwise affect the transactions contemplated by the merger agreement or the ancillary agreements or which questions the
 validity or legality of the transactions contemplated hereby or thereby;
- The Form S-4 shall have become effective in accordance with the provisions of the Securities Act and no stop order suspending the effectiveness of the Form S-4 shall have been issued by the SEC and remain in effect and no proceeding to that effect shall have been commenced or threatened;
- The shares of Mandalay Digital common stock to be issued in the merger shall have been approved for listing on the NASDAQ, subject to official notice of issuance;
- Appia shall have received a written opinion of its counsel, Goodwin Procter LLP dated as of the closing date to the effect that, on the basis of facts, representations and assumptions set forth in such opinion, the merger will qualify as a reorganization within the meaning of Section 368(a) of the Code;
- Any waiting period expirations or terminations, consents, clearances, waivers and approvals under any applicable antitrust, competition or merger control laws shall have been obtained, and there is no order or injunction prohibiting, limiting or preventing the consummation of the transactions contemplated herein pursuant to any antitrust, competition or merger control law:
- Mandalay Digital and DTM Merger Sub, Inc. shall have executed and delivered the ancillary agreements to which Mandalay and DTM Merger Sub, Inc. are parties;
- Appia shall have obtained the approval of its stockholders to the merger;
- Mandalay Digital shall have delivered to Appia evidence of (i) the adoption and approval by Mandalay Digital, as the sole stockholder of DTM Merger Sub, Inc., of the merger agreement, the merger and the other transactions contemplated hereby in accordance with the DGCL and Mandalay Digital's organizational documents and (ii) the approval by the Mandalay Digital stockholders of the issuance of the shares of Mandalay Digital common stock in connection with the merger;

- Mandalay Digital shall have obtained the approval by the Mandalay Digital stockholders of the issuance of the shares of Mandalay Digital common stock in connection with the merger; and
- The consummation of the transactions contemplated by the merger agreement will not be prohibited by any applicable law.

Conditions to Mandalay Digital's and DTM Merger Sub, Inc.'s Obligations to Effect the Merger

The respective obligations of Mandalay Digital and DTM Merger Sub, Inc. to effect the merger are subject to the satisfaction of several additional conditions (any of which may be waived in writing by Mandalay Digital), including:

- All representations and warranties of Appia contained in the merger agreement (i) that are qualified by materiality or Material Adverse Effect or words of similar import shall be true and correct in all respects, as of the date of the merger agreement and at and as of the closing date, except that such representations and warranties that are made as of a specific date need only be true and correct as of such date, and (ii) that are not qualified by materiality or Material Adverse Effect shall be true and correct in all material respects at and as of the date of the merger agreement and at and as of the closing date, except that such representations and warranties that are made as of a specific date need only be true and correct as of such date.
- Appia shall have performed and satisfied in all material respects all agreements and covenants required hereby to be performed by Appia prior to or on the closing date;
- All required third-party consents, including from governmental authorities, necessary for the valid consummation of the transactions contemplated by the merger agreement shall have been obtained;
- No suit, action, investigation, inquiry or other proceeding by any governmental authority or other person or legal or administrative proceeding shall have been instituted or threatened which seeks to restrain, enjoin, prevent the consummation of or otherwise affect the transactions contemplated by the merger agreement or the ancillary agreements or which questions the validity or legality of the transactions contemplated hereby or thereby;
- The Form S-4 shall have become effective in accordance with the provisions of the Securities Act and no stop order suspending the effectiveness of the Form S-4 shall have been issued by the SEC and remain in effect and no proceeding to that effect shall have been commenced or threatened;
- Any waiting period expirations or terminations, consents, clearances, waivers and approvals under any applicable antitrust, competition or merger control laws shall have been obtained, and there is no order or injunction prohibiting, limiting or preventing the consummation of the transactions contemplated herein pursuant to any antitrust, competition or merger control law;
- The consummation of the transactions contemplated by the merger agreement will not be prohibited by any applicable law;
- Appia shall have delivered to Mandalay Digital evidence that the Appia stockholders have approved the merger;
- Mandalay Digital shall have obtained the approval by the Mandalay Digital stockholders of the issuance of the shares of Mandalay Digital common stock in connection with the merger;
- Stockholders holding not more than 5% of the Appia capital stock outstanding immediately prior to the effective time (or on an as-converted to Appia common stock basis) shall have exercised, and shall not have any continued right to exercise, appraisal, dissenters' or similar rights under any applicable Law with respect to shares of capital stock of the Company held by them;

- The Non-competition Agreement signed by Jud Bowman shall be in full force and effect and shall not have been repudiated;
- Appia shall have provided Mandalay Digital with executed resolutions of its board of directors authorizing the termination of Appia's 401(k) plan;
- If a 280G Vote is required (i) Appia shall have received and delivered to Mandalay Digital a parachute payment waiver from each person that is eligible to receive a payment that may constitute a "parachute payment" under Section 280G of the internal revenue code prior to soliciting the Section 280G Approval and (ii) Appia's stockholders shall have (A) approved, pursuant to the method provided for in the regulations promulgated under Section 280G of the internal revenue code, any such "parachute payments" or (B) shall have voted upon and disapproved such "parachute payments," and, as a consequence, such "parachute payments" shall not be paid or provided for in any manner and Mandalay Digital and its affiliates shall not have any liabilities with respect to such "parachute payments;"
- The shares of Mandalay Digital common stock to be issued in the merger shall have been approved for listing on the NASDAQ, subject to official notice of issuance;
- Mandalay Digital shall have received a written opinion of its counsel, Latham & Watkins LLP, dated as of the closing date, to the effect that, on the basis of facts, representations and assumptions set forth in such opinion, the merger will qualify as a reorganization within the meaning of Section 368(a) of the Code;
- Since September 30, 2014, there shall not have been any Material Adverse Change;
- Each Appia stockholder shall have executed and delivered each ancillary agreement to which such stockholder is to be a party;
- Each of the secured guarantees and financing agreements required to rollover the Existing Credit Arrangements of Appia shall have been executed and delivered by the applicable lenders and other parties thereto (other than Mandalay Digital) and such agreements, instruments and other deliverables shall be in full force and effect, and no event of default or similar event shall exist under the Existing Credit Arrangements or will exist upon entry into the financing agreements that will amend and restate the Existing Credit Arrangements; and
- If the Closing Date is on or before January 31, 2015, Net Debt as of the Closing shall not be greater than \$13,750,000. If the Closing Date is after January 31, 2015 but prior to the February 28, 2015 then Net Debt as of the Closing shall not be greater than \$14,500,000. If the Closing Date is after February 28, 2015, but prior to the Outside Date then Net Debt as of the Closing shall not be greater than \$15,000,000.

For purposes of the merger agreement, "Net Debt" means (i) all Long Term Liabilities minus (ii) all cash and cash equivalents of Appia determined in accordance with GAAP plus (iii) all unpaid Company Transactions Expenses incurred by Appia prior to the effective time (including Appia transaction expenses that have not been billed prior to the effective time but that will be billed or become payable in the future). "Long Term Liabilities" means all Indebtedness (including the current portion thereof) and other long term liabilities of Appia or any of its subsidiaries on a consolidated basis determined in accordance with GAAP (including any such liabilities arising as a result of the consummation of the merger and the transactions contemplated hereby).

Definition of "Material Adverse Effect"

For the purposes of this summary and pursuant to the terms of the merger agreement, "Material Adverse Effect" or "Material Adverse Change" shall mean with respect to Appia, PocketGear or Appia's business any

material adverse effect or change in the condition (financial or other), business, results of operations, assets, Liabilities or operations of Appia, PocketGear or Appia's business taken as a whole or on the ability of Appia to consummate the transactions contemplated hereby, or any event or condition which would or would reasonably be expected to, with the passage of time, constitute a "Material Adverse Effect" or "Material Adverse Change"; provided, however, that none of the following are taken into account in determining whether there has been or would reasonably be expected to be a "Material Adverse Effect" or "Material Adverse Change": (a) any change in general economic, credit, capital or financial markets or general political conditions; (b) any outbreak of hostilities, acts of war or terrorism; (c) any hurricane, tornado, flood, earthquake or other natural disaster occurring after the date hereof; (d) any change in applicable law or GAAP (as defined in the merger agreement) enacted after the date hereof; (e) any adverse change in general conditions in the industries in which such Person primarily operates; (f) except with respect to the representations and warranties related to "consents and approvals," the negotiation, execution or announcement of the transactions contemplated by, or the performance of obligations under, the merger agreement or the ancillary agreements, including effects related to the identity of Mandalay Digital; (g) any failure to meet internal or published projections, estimates or forecasts of revenues, earnings, or other measures of financial or operating performance for any period (provided that the underlying causes of such failure may be considered in determining whether there was or is a Material Adverse Effect); or (h) actions specifically required to be taken by Appia pursuant to this Agreement or that are taken following receipt of Mandalay Digital's written consent to such action; except in the cases of clauses (a) – (e), to the extent that such person is disproportionately adversely affected thereby as compared with other participants in the industries in which such person primarily operates.

Termination of the Merger Agreement

The merger agreement may be terminated at any time before the effective time of the merger:

- by mutual written consent of Mandalay Digital and Appia;
- by either Mandalay Digital or Appia if (i) the closing shall not have occurred on or before April 30, 2015, or (ii) if, at the Mandalay Digital Stockholder Meeting (including any adjournment or postponement thereof), the approval by the Mandalay Digital stockholders of the issuance of the shares of Mandalay Digital common stock in connection with the merger shall not have been obtained; provided, however, that this provision shall not be available to Mandalay Digital if Appia has the right to terminate this Agreement under fourth bullet below and this provision shall not be available to Appia if Mandalay Digital has the right to terminate this Agreement under the third bullet below;
- By Mandalay Digital if (i) there is a material breach of any representation or warranty of Appia in the merger agreement or in the Support Agreement or any covenant or agreement to be complied with or performed by Appia or Appia stockholders pursuant to the terms of the merger agreement or the failure of a condition to Mandalay Digital's obligations to effect the merger to be satisfied (and such condition is not waived in writing by Mandalay Digital) on or prior to April 30, 2015, or the occurrence of any event which results or would result in the failure of a condition to Mandalay Digital's obligations to effect the merger to be satisfied on or prior to April 30, 2015 or (ii) Appia shall have failed to obtain the approval of the merger from its stockholders;
- By Appia if there is a material breach of any representation or warranty of Parent or DTM Merger Sub, Inc. in the merger agreement or of any covenant or agreement to be complied with or performed by Parent or DTM Merger Sub, Inc. pursuant to the terms of the merger agreement or the failure of a condition to Appia's obligations to effect the merger to be satisfied (and such condition is not waived in writing by Appia) on or prior to April 30, 2015, or the occurrence of any event which results or would result in the failure of a condition to Appia's obligations to effect the merger to be satisfied on or prior to April 30, 2015; or

• By Appia, if (i) all of the conditions to the parties' obligations to effect the merger have been satisfied or waived in accordance with the terms of the merger agreement as of the date the closing should have been consummated pursuant to the terms of the merger agreement (other than (A) those conditions that by their nature are to be satisfied (including the delivery of documents) at the closing, which conditions would have been satisfied assuming the closing had occurred at such time and (B) the receipt of the approval of the merger by the stockholders of Appia), (ii) the approval by the Mandalay Digital stockholders of the issuance of the shares of Mandalay Digital common stock in connection with the merger shall not have been obtained at Mandalay Digital's stockholder meeting (or any adjournment or postponement thereof), (iii) Appia has certified in writing to Mandalay Digital on such date that all of the conditions to the parties' obligations to effect the merger have been satisfied or waived in accordance with the terms of the merger agreement (or that conditions that by their nature are to be satisfied (including by delivery of documents) at the closing, which conditions would have been satisfied assuming the closing had occurred at such time) and that it is ready, willing and able to consummate the transactions contemplated hereby on such date and throughout the three (3) business days following delivery of such notice, and (iv) Mandalay Digital fails to consummate the closing within such three (3) business day period.

Effect of Termination

If the merger is terminated as described in the section entitled "Summary of the Merger Agreement—Termination of the Merger Agreement" above, the merger agreement will be void, except that for certain designated provisions of the merger agreement, and each party will redeliver all documents, work papers and other material of any other party relating to the transactions contemplated hereby, and no party shall have any liability to any other party to the merger agreement, provided that nothing shall limit or restrict the availability of specific performance or other injunctive relief to the extent that specific performance or such other relief would otherwise be available to a party.

Fees and Expenses

If the merger agreement is terminated by Appia pursuant to either of the last two bullets in the section entitled "Summary of the Merger Agreement—Termination of the Merger Agreement" above, then Mandalay Digital is required to pay Appia an amount equal to the sum of (i) a nonrefundable fee in the amount in cash equal to \$1,000,000 and (ii) expense reimbursement up to \$500,000 of Appia's transaction expenses (collectively, the "termination amount"). Mandalay Digital is required to pay the termination amount to Appia, by wire transfer of immediately available funds to an account designated by Appia, within two (2) Business Days after the date of such termination unless, solely in the case of a termination of the merger agreement pursuant to the second to last bullet in the section entitled "Summary of the Merger Agreement—Termination of the Merger Agreement" above as a result of willful and intentional breach by Mandalay Digital or DTM Merger Sub, Inc., Appia notifies Mandalay Digital in writing at the time of termination that it will pursue a remedy other than the termination amount (in which case Mandalay Digital will have no obligation to pay the termination amount). Except as otherwise expressly provided in the merger agreement, whether or not the transactions contemplated by the merger agreement are consummated, all costs and expenses incurred in connection with the merger agreement and the transactions contemplated thereby shall be paid by the party incurring such expenses.

Amendment

The merger agreement may not be amended or modified (before or after receipt of the receipt of the approval of the Mandalay Digital stockholders of the issuance of the Mandalay Digital common stock in connection with the merger) except in an instrument in writing signed by the stockholder representative and Mandalay Digital. No amendment, supplement, modification or waiver of the merger agreement will be binding unless executed in writing by the party to be bound thereby.

ANCILLARY AGREEMENTS

Support Agreement

The following description of certain material features of the support agreements is intended to be a summary only. This summary is qualified in its entirety by the full text of the form of the support agreements that is attached hereto as *Appendix B*.

Concurrently with the execution of the merger agreement, certain directors, executive officers and their affiliates that were Appia stockholders each executed a support agreement with Mandalay. As of the Appia record date, these directors, executive officers and their affiliates held approximately 77% of the outstanding shares of Appia common stock, approximately 94% of the outstanding shares of Appia preferred stock and approximately 89% of the outstanding shares of Appia common stock and preferred stock (on an as-converted to Appia common stock basis), all of which shares of Appia capital stock are subject to the support agreements. The (i) affirmative vote, at a duly called and held meeting of stockholders of Appia or (ii) approval by written consent in lieu of a meeting (in each case to approve the merger, certain Appia charter amendment and the other transactions contemplated by the merger agreement), of the holders of (y) at least a majority of the outstanding shares of Appia Common Stock and Appia Preferred Stock (voting on an as-converted to Appia common stock basis) and (z) two-thirds of the outstanding shares of Appia Series A Preferred Stock, Appia Series B Preferred Stock, Appia Series C Preferred Stock and Appia Series D Preferred Stock (voting on an as-converted to Appia common stock basis), in each case entitled to vote upon the merger and the other Transactions contemplated by the merger agreement, is the only vote of the holders of any class or series of shares of capital stock of Appia necessary to approve the merger, certain Appia charter amendment and the other transactions contemplated by the merger agreement. Therefore, under the support agreements, Appia expects to receive the requisite votes or number of consents sufficient to satisfy these approval requirements.

All of the Appia stockholders that signed the support agreements agreed not to transfer any of their respective shares of Appia capital stock prior to the closing of the merger and have further agreed that any additional shares of Appia capital stock acquired by such stockholders are also subject to the terms of the support agreements.

In addition, pursuant to the terms of the support agreement each of the signing stockholders of Appia have made certain additional covenants, including the following:

Confidentiality

All of the Appia stockholders that signed the support agreements agreed not to, subject to certain exceptions, disclose the confidential and proprietary information of Appia that any such stockholders may have in their possession.

Public Announcements

All of the Appia stockholders that signed the support agreements agreed not to, subject to certain exceptions, make any press release or other public disclosure with respect to the merger agreement, the support agreements and the transactions contemplated by such agreements.

Lock-Up

All of the Appia stockholders that signed the support agreements agreed that the shares of Mandalay Digital common stock to be received by such stockholders in the merger are all subject to lock up restrictions, which lock up restrictions will lapse with respect to one-third of such stockholders, share of Mandalay Digital common stock at each of the six, nine and 12 month anniversaries of the closing date.

Release

All of the Appia stockholders that signed the support agreements agreed to release Mandalay Digital, Merger Sub, Appia and their respective affiliates, subsidiaries, officers, directors, stockholders, partners, members, agents, successors and permitted assigns from certain claims (as defined in the support agreements) that such stockholders may have against such parties. The release did not extend to certain claims, including those arising out of the merger agreement or with respect to such stockholder's status as a current or former employee of the consultant of Appia for certain matters.

Outstanding Appia Shares and Required Vote

As of the Appia record date, the outstanding voting securities of Appia consist of (A) 3,750,205 shares of Appia common stock (excluding any shares of Appia restricted stock), (B) 177,501 shares of Appia restricted stock, (C) 3,275,000 shares of Appia Series A Preferred Stock, (D) 2,569,169 shares of Appia Series B Preferred Stock, (E) 1,348,814 shares of Appia Series C Preferred Stock, (F) 89,127 shares of Appia Series D Preferred Stock, and (G) 2,991,500 shares of Appia Series FF Preferred Stock. The (i) affirmative vote, at a duly called and held meeting of stockholders of Appia or (ii) approval by written consent in lieu of a meeting (in each case to approve the merger, the certain Appia charter amendment and the other transactions contemplated by the merger agreement), of the holders of (y) at least a majority of the outstanding shares of Appia common stock and Appia preferred stock (voting on an as-converted to Appia common stock basis) and (z) two-thirds of the outstanding shares of Appia Series A Preferred Stock, Appia Series B Preferred Stock, Appia Series C Preferred Stock and Appia Series D Preferred Stock (voting on an as-converted to Appia common stock basis), in each case entitled to vote upon the merger and the other transactions contemplated by the merger agreement, is the only vote of the holders of any class or series of shares of capital stock of Appia necessary to approve the merger, the certain Appia charter amendment and the other transactions contemplated by the merger agreement. As of the Appia record date, certain directors, executive officers and their affiliates held approximately 77% of the outstanding shares of Appia common stock, approximately 94% of the outstanding shares of Appia preferred stock and approximately 89% of the outstanding shares of Appia common stock and preferred stock (on an as-converted to Appia common stock basis). All of these shares of Appia capital stock are subject to the support agreements described above.

Financing Agreements

Upon consummation of the merger, Appia intends to enter into an amended loan agreement with its current senior lender, Silicon Valley Bank (the "SVB Debt") and Mandalay Digital will issue a secured guaranty of the SVB Debt whereby it will guaranty all of Appia's obligations in connection with the SVB Debt and pledge substantially all of its assets, including its intellectual property, to Silicon Valley Bank in support of the SVB Debt.

Additionally, upon consummation of the merger, Mandalay Digital will enter into a securities purchase agreement with Appia and North Atlantic SBIC IV, L.P. ("North Atlantic") whereby Appia will issue to North Atlantic a subordinated debenture in the aggregate principal amount of \$8.0 million ("North Atlantic Debt"), to retire a like amount of outstanding debt to North Atlantic and, Mandalay Digital will issue a secured guaranty of the North Atlantic Debt whereby it will guarantee all of Appia's and Mandalay Digital's obligations in connection with the North Atlantic Debt and pledge substantially all of its assets, including its intellectual property, to North Atlantic in support of the North Atlantic Debt. Mandalay Digital will also issue to North Atlantic 200,000 shares of its common stock and a common stock warrant to purchase 400,000 shares of its common stock, all of which will have registration rights requiring Mandalay Digital to file a registration statement 15 business days after the closing of the merger with Appia. The warrant will have an exercise price of \$0.01 per share exercisable for 10 years, but it is not exercisable until the one year anniversary of the closing date of the merger and will terminate if Mandalay Digital repays the North Atlantic Debt prior to such one year anniversary.

The secured guarantees to be issued upon consummation of the merger to each of Silicon Valley Bank and North Atlantic will contain covenants, among others, limiting the Company's ability to undergo a change of control, incur indebtedness, grant liens, make dividends in cash and other customary covenants. A violation of the provisions of the secured guarantees, as well as an event of default existing under the SVB Debt or North Atlantic Debt, will constitute an event of default under each guaranty. Upon the occurrence and continuation of such event of defaults, Silicon Valley Bank and North Atlantic may exercise certain remedies against the Company and substantially all its assets, including foreclosure of such assets. The SVB Debt will mature at June 30, 2015, with respect to the revolving line, and April 1, 2016, with respect to the term loan. The North Atlantic Debt will mature two years after the date of issuance. SVB and North Atlantic will also enter into a subordination agreement pursuant to which North Atlantic will agree to, among other things, limit its rights to receive payment, exercise remedies and enforce its rights for a certain period while an event of default under the North Atlantic debt exists.

UNAUDITED PRO FORMA CONDENSED COMBINED FINANCIAL INFORMATION

On November 13, 2014, Mandalay Digital Group, Inc., a Delaware corporation ("Mandalay Digital"), entered into an Agreement and Plan of Merger (the "merger agreement") with Appia, Inc., a Delaware corporation ("Appia"), and DTM Merger Sub, Inc., a Delaware corporation and a wholly-owned subsidiary of Mandalay Digital ("Merger Sub"), pursuant to which Mandalay Digital will acquire Appia (the "merger"), on the terms and subject to the conditions set forth in the merger agreement. The transactions contemplated by the merger agreement have not yet been consummated.

The Unaudited Pro Forma Condensed Combined Balance Sheet combines the consolidated balance sheets of Mandalay Digital and Appia, giving effect to the merger as if it had been consummated on September 30, 2014. The Unaudited Pro Forma Combined Statements of Operations for the six months ended September 30, 2014, and for the year ended March 31, 2014, combine the historical consolidated statements of operations of Mandalay Digital and Appia, giving effect to the merger as if it had been consummated on April 1, 2013, the beginning of the earliest period presented. Mandalay Digital and Appia have different fiscal year ends, with the most recent annual period of Mandalay Digital ended on March 31, 2014, and the most recent annual period of Appia ended on December 31, 2013. As such, amounts related to the historical operations of Appia have been adjusted to align the period over which those operations occurred with the periods presented by adding the necessary quarterly results to match Mandalay Digital's fiscal reporting periods. In addition, certain line items of the balance sheet and income statements were combined or reclassified in order to make the information comparable.

The Unaudited Pro Forma Condensed Combined Financial Statements were prepared using the acquisition method of accounting in accordance with FASB ASC Topic 805, Business Combinations, with Mandalay Digital considered as the accounting acquirer and Appia as the accounting acquiree. Accordingly, consideration to be given by Mandalay Digital to complete the merger with Appia will be allocated to assets and liabilities of Appia based on their estimated fair values as of the completion date of the merger. As of the date of this proxy statement/prospectus, Mandalay Digital has not completed the detailed valuation studies necessary to arrive at the required estimates of the fair value of the Appia assets to be acquired and the liabilities to be assumed and the related allocations of purchase price, nor has it identified all of the adjustments necessary to conform Appia's accounting policies to Mandalay Digital's accounting policies. A final determination of the fair value of Appia assets and liabilities will be based on the actual net tangible and intangible assets and liabilities of Appia that exist as of the date of completion of the merger and, therefore, cannot be made prior to the completion of the transaction. Additionally, the value of the consideration to be given by Mandalay Digital to complete the merger will be determined in part based on the trading price of Mandalay Digital's common stock at the time of the completion of the merger. Accordingly, the pro forma purchase price adjustments are preliminary and are subject to further adjustments as additional information becomes available and as additional analyses are performed. The preliminary pro forma purchase price adjustments have been made solely for the purpose of providing the Unaudited Pro Forma Combined Financial Statements presented below. Mandalay Digital estimated the fair value of Appia's assets and liabilities based on discussions with Appia management, preliminary valuation studies, due diligence and information presented in public filings. Until the merger is completed, both companies are limited in their ability to share information. Upon completion of the merger, final valuations will be performed. Increases or decreases in the fair value of relevant balance sheet amounts will result in adjustments to the balance sheet and/or statements of operations. There can be no assurance that such finalization will not result in material changes. There are no pro forma tax adjustments as both Mandalay and Appia have historically incurred losses and have an effective US tax rate of 0%. The actual effective tax rate after the merger may differ from this rate.

These Unaudited Pro Forma Condensed Combined Financial Statements have been developed from, and should be read in conjunction with, (1) the accompanying notes thereto, (2) the unaudited interim consolidated financial statements of Mandalay Digital contained in its Quarterly Report on Form 10-Q for the six months ended September 30, 2014, (3) the audited consolidated financial statements of Mandalay Digital contained in its

Annual Report on Form 10-K for the fiscal year ended March 31, 2014, both of which are incorporated by reference into this proxy statement/prospectus, (4) the unaudited interim consolidated balance sheet of Appia as of September 30, 2014, and the related consolidated statements of comprehensive income (loss) and cash flows for the nine months September 30, 2014, which are included in this registration statement, and (5) the audited consolidated balance sheets of Appia as of December 31, 2013 and 2012, and the related consolidated statements of comprehensive income (loss), shareholders' equity, and cash flows for each of the two fiscal years in the period ended December 31, 2013, which are included in this proxy statement/prospectus. The Unaudited Pro Forma Condensed Combined Financial Statements are provided for illustrative purposes only and do not purport to represent what the actual consolidated results of operations or the consolidated financial position of Mandalay Digital would have been if the merger had occurred on the dates assumed, nor are they necessarily indicative of future consolidated results of operations or consolidated financial position.

The Unaudited Pro Forma Condensed Combined Financial Statements do not reflect the costs of any integration activities including planning costs or any benefits that may result from realization of future cost savings from operating efficiencies or revenue synergies expected to result from the merger, except to the extent that such integration costs have been incurred during the periods presented. In addition, the Unaudited Pro Forma Condensed Combined Financial Statements (excluding the balance sheet) do not include one-time costs directly attributable to the transaction or professional fees incurred by Mandalay Digital or Appia pursuant to provisions contained in the merger agreement as those costs are not considered part of the purchase price.

Unaudited Pro Forma Condensed Combined Balance Sheet As of September 30, 2014 (In thousands)

	Historical Mandalay US\$	Historical Appia US\$	Pro Forma Adjustments US\$	Footnote	Pro Forma Combined US\$
ASSETS					
Current assets					
Cash and cash equivalents	16,715	1,952	(3,343)	(a)	15,324
Restricted cash	200	-			200
Accounts receivable, net	4,337	5,090	(64)	(b)	9,363
Deposits	86	-			86
Prepaid expenses and other current assets	349	245			594
Total current assets	21,687	7,287	(3,407)		25,567
Property and equipment, net	422	2,677	2,605	(c)	5,704
Other long-term assets	-	9	(9)	(f)	-
Deferred tax assets	541	-			541
Intangible assets, net	6,913	-	8,000	(d)	14,913
Goodwill	6,309		58,362	(e)	64,671
TOTAL ASSETS	35,872	9,973	65,551		111,396
LIABILITIES AND STOCKHOLDERS' EQUITY	<u>,</u>		<u> </u>		,
Current liabilities					
Accounts payable	3,817	721			4,538
Accrued license fees and revenue share	2,373	2,477	(64)	(b)	4,786
Accrued compensation	2,122	727			2,849
Current portion of long term debt	-	600			600
Deferred tax liabilities	1,024	-			1,024
Other current liabilities	708	1,291	(1,058)	(a)	941
Total current liabilities	10,044	5,816	(1,122)		14,738
Long term and convertible debt, net of discount	-	7,499	801	(f)	7,543
			(757)	(i)	
Other long-term liabilities	-	3,129	(3,129)	(g)	123
			123	(i)	
Total liabilities	10,044	16,444	(4,084)		22,404
Redeemable convertible preferred stock	-	39,552	(39,552)	(g)	-
Stockholders' equity:					
Preferred stock	100	3	(3)	(g)	100
Common stock	7	3	(3)	(g)	9
			2	(h)	
Additional paid-in capital	196,040	-	62,528	(h)	259,202
			634	(i)	
Treasury Stock	(71)	-			(71)
Accumulated other comprehensive loss	(129)	116	(116)	(g)	(129)
Accumulated deficit	(170,119)	(46,145)	46,145	(g)	(170,119)
Total stockholders' equity	25,828	(46,023)	109,187		88,992
TOTAL LIABILITIES AND STOCKHOLDERS' EQUITY	35,872	9,973	65,551		111,396

See accompanying Notes to Unaudited Pro Forma Condensed Combined Financial Statements

Unaudited Pro Forma Condensed Combined Statement of Operations For the Year Ended March 31, 2014 (In thousands, except per-share amounts)

	Historical Mandalay US\$	Historical Appia US\$	Pro Forma Adjustments US\$	Footnote	Pro Forma Combined US\$
Net revenues	\$ 24,404	\$ 49,129			\$ 73,533
Cost of revenues					
License fees and revenue share	14,789	36,080			50,869
Other direct cost of revenues	1,769	-			1,769
Total cost of revenues	16,558	36,080	-		52,638
Gross profit	7,846	13,049	-		20,895
Operating expenses					
Product development	7,869	4,016	868	(1)	12,753
Sales and marketing	1,915	1,871			3,786
General and administrative	13,432	6,547			19,979
Amortization of intangible assets			3,917	(m)	3,917
Impairment of intangible assets	154				154
Total operating expenses	23,370	12,434	4,785		40,589
Loss from operations	(15,524)	615	(4,785)		(19,694)
Interest and other income / (expense)					
Interest income / (expense)	(1,407)	(1,003)	(235)	(o)	(2,645)
Foreign exchange transaction gain / (loss)	33	-			33
Change in fair value of warrant derivative liabilities gain / (loss)	(811)	(85)	85	(n)	(811)
Loss on extinguishment of debt	(442)	-			(442)
Gain / (loss) on settlement of debt	74	-			74
Gain/ (loss) on disposal of fixed assets	-	-			-
Gain on change on valuation of long term contingent liability	603	-			603
Other income					
Interest and other expense	(1,950)	(1,088)	(150)		(3,188)
Loss from operations before income taxes	(17,474)	(473)	(4,935)		(22,882)
Income tax provision / (benefit)	(272)				(272)
Net loss from continuing operations, net of taxes	\$ (17,202)	\$ (473)	\$ (4,935)		\$ (22,610)
Basic and diluted net loss per common share from continuing operations	\$ (0.63)				\$ 0.48
Weighted average common shares outstanding, basic and diluted	27,478		19,553		47,031

See accompanying Notes to Unaudited Pro Forma Condensed Combined Financial Statements

Unaudited Pro Forma Condensed Combined Statement of Operations For the Six Months Ended September 30, 2014 (In thousands, except per-share amounts)

	Historical Mandalay US\$	Historical Appia US\$	Pro Forma Adjustments US\$	Footnote	Pro Forma Combined US\$
Net revenues	\$ 11,016	\$ 14,596	\$ (126)	(j)	\$ 25,486
Cost of revenues					
License fees and revenue share	7,112	11,567	(126)	(j)	18,553
Other direct cost of revenues	689				689
Total cost of revenues	7,801	11,567	(126)		19,242
Gross profit	3,215	3,029	-		6,244
Operating expenses					
Product development	4,114	1,496	434	(1)	6,044
Sales and marketing	1,504	1,700			3,204
General and administrative	6,922	2,852	(240)	(k)	9,534
Amortization of intangible assets	<u>-</u>		1,958	(m)	1,958
Total operating expenses	12,540	6,048	2,152		20,740
Loss from operations	(9,325)	(3,019)	(2,152)		(14,496)
Interest and other income / (expense)	<u> </u>		<u> </u>		
Interest income / (expense)	(128)	(827)	(152)	(o)	(1,267)
			(160)	(p)	
Foreign exchange transaction gain / (loss)	(7)	-			(7)
Change in fair value of warrant derivative liabilities gain / (loss)	-	(2,019)	2,019	(n)	-
Loss on extinguishment of debt	-	-			-
Gain / (loss) on settlement of debt	(10)	-			(10)
Gain/ (loss) on disposal of fixed assets	2	-			2
Gain on change on valuation of long term contingent liability	-	-			-
Other income	12	<u> </u>			12
Interest and other expense	(131)	(2,846)	1,707		(1,270)
Loss from operations before income taxes	(9,456)	(5,865)	(445)		(15,766)
Income tax provision / (benefit)	355				355
Net loss from continuing operations, net of taxes	\$ (9,811)	\$ (5,865)	\$ (445)		\$ (16,121)
Basic and diluted net loss per common share from continuing operations	\$ (0.26)				\$ (0.28)
Weighted average common shares outstanding, basic and diluted	37,464		19,553		57,017

See accompanying Notes to Unaudited Pro Forma Condensed Combined Financial Statements

NOTES TO UNAUDITED PRO FORMA CONDENSED COMBINED FINANCIAL STATEMENTS

1. Basis of Presentation

On November 13, 2014, Mandalay Digital and Appia announced that they had entered into an Agreement and Plan of Merger pursuant to which Mandalay Digital will issue shares of its common stock in exchange for all of Appia's outstanding common and preferred stock and warrants. The number of shares to be issued by Mandalay Digital is subject to adjustment based on Appia's working capital and net indebtedness as of the closing date of the merger. Based on Appia's working capital and net indebtedness as of September 30, 2014, Mandalay Digital would issue approximately 19,553,000 shares of common stock. Additionally, Appia's equity awards outstanding at the closing date will be assumed by Mandalay Digital and converted into equity awards with respect to Mandalay Digital common stock. Vested equity awards held by Appia's employees and service providers are considered part of the purchase price; accordingly, the estimated purchase price includes an estimated fair value of equity awards to be issued by Mandalay Digital of approximately \$547 thousand.

The preliminary total purchase price of the proposed transaction is as estimated follows (in thousands):

Fair value of Mandalay common shares	\$61,983
Fair value of Mandalay options issued in exchange for Appia options	547
Assumption of Appia net debt	6,948
Total estimated purchase price	\$69,478

The value of Mandalay Digital's common stock used to estimate the purchase price was \$3.17 per share, the closing price on November 21, 2014. An increase or decrease in the market price of Mandalay Digital's common stock of 20% would increase or decrease the value of the Mandalay Digital common stock to be received by Appia stockholders upon completion of the transactions as set forth below, with a corresponding increase or decrease in goodwill that will be recorded in connection with the transaction (in thousands, except per-share amounts):

		Percentage change in stock price				
	-20%			20%		
Market price of Mandalay common stock	\$	2.54	\$	3.80		
Fair value of Mandalay common shares to be issued	\$	49,586	\$	74,379		

Under the acquisition method of accounting, the total preliminary estimated purchase price as shown in the table above is allocated to the Appia net tangible and identifiable intangible assets acquired and liabilities assumed based on their estimated fair values as of the date of the completion of the transaction. The preliminary estimated purchase price has been allocated based on estimates taking into account various factors as described in the introduction to these unaudited pro forma condensed combined financial statements. The allocation of the estimated purchase price discussed below is preliminary because the transaction has not yet been completed. The final allocation of the purchase price will be based on the fair values of Appia's assets and liabilities as of the closing of the Merger. The preliminary allocation of the estimated purchase price is as follows (in thousands):

Current assets	\$ 5,335
Property and equipment, excluding developed technology	282
Developed technology	5,000
Advertiser relationships	5,000
Publisher relationships	2,500
Trade name and trademarks	500
Goodwill	58,362
Current liabilities	(7,502)
Total estimated purchase price	\$69,478

Upon completion of the fair value assessment after the merger, it is anticipated that the final purchase price allocation will differ from the preliminary assessment outlined above. Any changes to the initial estimates of the fair value of assets and liabilities will be recorded as adjustments to those assets and liabilities and residual amounts will be allocated to goodwill.

2. Adjustment of Appia Financial Statements

Mandalay Digital and Appia have different fiscal year ends, with the most recent annual period of Mandalay Digital ended on March 31, 2014, and the most recent annual period of Appia ended on December 31, 2013. As such, amounts related to the historical operations of Appia have been adjusted to align the period over which those operations occurred with the periods presented by (i) adding the results for the quarter ended March 31, 2014 to, and deducting the results for the quarter ended March 31, 2013 from, the results for the fiscal year ended December 31, 2013, and (ii) deducting the results for the quarter ended March 31, 2014 from the results for the nine month period ended September 30, 2014.

3. Pro Forma Adjustments

Certain reclassifications have been made to conform Appia's historical reported balances to Mandalay Digital's financial statement basis of presentation.

The pro forma adjustments included in the unaudited pro forma condensed financial statements are as follows:

Pro Forma Balance Sheet Adjustments:

- (a) <u>Cash.</u> Adjustment to cash reflects payment of certain Appia liabilities due upon closing, including accrued interest and prepayment fees, and transaction costs.
- (b) <u>Intercompany balances.</u> Adjustment to accounts receivable and accrued license fees reflects the elimination of intercompany balances between Mandalay Digital and Appia.

- (c) <u>Property and Equipment.</u> Adjustment reflects the portion of the preliminary purchase price allocation relating to the excess of the fair value of developed technology over the historical balance of capitalized software costs, net of amortization.
- (d) <u>Intangible assets.</u> Adjustment reflects the portion of the preliminary purchase price allocation relating to identified intangible assets, including advertiser and publisher relationships and trade names.
- (e) Goodwill. Adjustment reflects the portion of the preliminary purchase price allocation relating to goodwill.
- (f) <u>Long-term debt.</u> Adjustment eliminates the remaining unamortized discount on Appia's debt, and eliminates historical debt issuance costs. Appia's debt is summarized as follows (in thousands):

	Historical US \$	Pro Forma US \$
Bank debt—current portion	\$ 600	\$ 600
		
Bank debt—noncurrent portion	300	300
Subordinated debt	8,000	8,000
Less discount	(801)	(757)
Subtotal—noncurrent	7,499	7,543
Total debt	\$8,099	\$8,143

- (g) Appia stockholders' equity, redeemable convertible preferred stock and warrant liability. Adjustment eliminates all of Appia's historical equity balances, preferred stock, and warrant liabilities which will be exchanged for Mandalay Digital common stock.
- (h) Equity interests issued to holders of Appia equity interests. Adjustment reflects the issuance of approximately 19,553,000 shares of Mandalay Digital common stock to the holders of Appia's equity, redeemable convertible preferred stock and warrant liability and approximately 203,000 Mandalay Digital options issued in exchange for vested Appia options. Unvested equity awards, consisting of approximately 129,000 Mandalay Digital options and approximately 47,000 restricted shares of Mandalay Digital common stock, are excluded from the calculation of the purchase price.
- (i) <u>Common stock and warrant issued to lender</u>. Adjustment reflects the issuance of 200,000 shares of Mandalay Digital common stock to the holder of Appia's subordinated debt. The holder will also receive a warrant to acquire 400,000 additional shares of Mandalay Digital common stock, exercisable only the subordinated note is not repaid in full within twelve months of the closing of the merger. The estimated fair value of the warrant is \$123 thousand.

Pro Forma Statements of Operations Adjustments:

- (j) <u>Intercompany transactions</u>. Adjustment to revenues and cost of revenues eliminates transactions between Mandalay Digital and Appia.
- (k) <u>Transaction-related costs.</u> Adjustment to eliminate professional fees and other direct costs incurred by Mandalay Digital and Appia related to the merger.
- (1) <u>Amortization of technology.</u> Adjustment to amortize the excess of the fair value of developed technology over the historical balance of capitalized software costs, based on an estimated useful life of 3 years.

- (m) <u>Amortization of intangible assets.</u> Adjustment to record the estimated amortization expense on identified intangible assets recorded as part of the purchase price allocation. The adjustment is based on estimated useful lives of 2 years for advertiser and publisher relationships and 3 years for trade names and trademarks.
- (n) Other expenses. Adjustment to eliminate historical expense relating to changes in the fair value of the warrant liability.
- (o) <u>Interest expense</u>. Adjustment to eliminate historical amortization of debt discount and debt issuance costs, and record amortization of the new discount relating to the common shares and warrant to be issued to the holder of Appia's subordinated debt.
- (p) <u>Interest expense</u>. Adjustment to record additional interest expense on Appia's subordinated debt due to the increase in the contractual interest rate from 10% to 14% following the first anniversary of the closing of the merger.

BUSINESS OF MANDALAY DIGITAL

Financial and other information relating to Mandalay Digital is set forth in Mandalay Digital's 2014 Annual Report on Form 10-K, as amended on Form 10-K/A and Quarterly Reports on Form 10-Q for the period s ended June 30, 2014 and September 30, 2014, which are incorporated by reference in this proxy statement/prospectus, and elsewhere in this proxy statement/ prospectus. Mandalay Digital will furnish you with copies of the documents incorporated by reference in this proxy statement/ prospectus upon request. See "Where You Can Find More Information" and "Documents Incorporated by Reference" on pages 153 and 154, respectively.

BUSINESS OF APPIA, INC.

Overview

Appia is a leading independent mobile user acquisition network. Since its formation, Appia has delivered more than 88 million app installs for hundreds of advertisers, including 60 of the top 100 grossing apps on the App Store and Google Play. Appia provides the technology, infrastructure, and scale to provide higher lifetime value users for advertisers while driving positive return on ad spend. In addition, Appia partners with a diverse set of publishers including direct developer relationships, mobile websites, carriers, and mediated relationships to maximize their advertising revenue.

Appia, Inc. was founded in 2008 and incorporated under the General Corporation Law of the State of Delaware on April 3, 2008. Appia operates, through leases or outsourced contractors, offices in multiple cities including Durham, North Carolina, San Francisco, California, Singapore, Cork, and Mexico City.

Appia's Products

Appia's mobile user acquisition platform is a demand side platform, or DSP. This platform allows mobile advertisers to engage with the right customers for their Apps at the right time to gain them as customers. Appia accesses mobile ad inventory through publishers including direct developer relationships, mobile websites, carriers and mediated relationships; as well as purchasing inventory through exchanges using real-time bidding ("RTB"). The advertising revenue generated by Appia's platform is shared with publishers according to contractual rates in the case of direct or mediated relationships. When inventory is accessed using real-time bidding, Appia buys inventory at a rate determined by the marketplace.

Appia's Platform provides core functionality that is summarized below.

Data Targeting

Appia offers a variety of audience targeting capabilities to its advertisers. This can include country, device, operating system, category, and location targeting.

Optimization

Appia has built tools and technology that allow it to effectively drive the highest value user through a performance-focused cost-per-install (CPI) model. Appia has developed proprietary ad serving algorithms that allow Appia systems to optimize the install to drive the most valuable user while maximizing ad spend, all with minimal human involvement. These algorithms allow Appia to utilize all forms of environmental and device data when making decisions on what ad to show.

Programmatic

Appia has built real-time bidding infrastructure to provide access to large amounts of inventory and publishers. Appia is integrated with many of the major mobile Supply Side Platforms or exchanges. Appia's

RTB, or, platform processes more than one billion ad requests per day and values each ad request in real-time based on predictive algorithms and user profile insights to purchase the impressions that will drive the highest quality customers for our advertisers. Appia uses multiple datacenters across the United States, Tokyo, and Ireland to accomplish this.

Advertiser Quality Tracking

Through an Open Event API, Appia's Advertiser Quality Tracking capabilities provide the ability to utilize virtually any data that an advertiser can provide about the quality of their customers that can help to inform the campaign going forward. Appia utilizes this data to identify the inventory that best meets advertisers' campaign needs and optimizes using post install data to hyper-target the most value users across supply for the campaign. These data points are delivered in an anonymous fashion, ensuring that Appia is compliant with privacy best practices. Appia does not have access to personally identifiable information about a user.

Via

Appia's Via provides a simple, visually-rich and unified mobile advertising reporting and control environment that enables advertisers and publishers to see revenue performance and campaign results. Publishers are able to view mobile monetization performance including revenue, impressions, clicks, and installs; as well as optimize their supply strategy. Advertisers are able to view actionable analysis of their campaigns including spend, performance, and geo coverage. In addition, they can manage and optimize campaigns within Via.

Via also serves as a publisher self-service portal for app developers that have fewer than 100,000 unique users. Within Via, app developers can create their account, manage their app placements, and gain access to integration methods including SDK, mobile web link, and ad tags.

SDK

Appia provides a software development kit, or SDK, for publishers and developers seeking to integrate ad formats while maintaining a small code footprint. The Appia SDK supports iOS and Android, contains a Unity plugin, and supports Appia's App Wall, interstitials, and banner ads. Appia's AppWall is a targeted app discovery tool that can be used in-app or via mobile web. Apps are shown based on rating (4+ stars), country, device, and category to ensure the most relevant apps are displayed to the end user. The AppWall is fully customizable to match the look and feel of individual apps or mobile websites.

Campaign Optimization Services

The Appia Campaign Optimization team assists advertisers in driving the highest value users across Appia's premium supply. The Appia Campaign Optimization team provides:

- the inputs for the system to hyper-target the most valuable user across supply for the campaign;
- guidance on campaign optimization and spend; and
- analytics to help advertisers optimize bidding strategy.

Full Solution

Appia has focused on establishing differentiation in the performance mobile market by creating technology and solutions to drive the most valuable user at the most cost effective rate. In addition, it has developed a wide offering of solutions for its advertisers and publishers, either directly or through partnerships. This includes mobile conversion tracking and measurement including post-install event tracking, mobile video, rich ad types including App Walls, geo-targeting and location based targeting.

Customers

Appia's customer is the mobile advertiser focused on driving higher value user acquisition. Appia currently has more than 150 global advertisers with over 1,000 campaigns. Appia's advertisers include 60 of the top 100 grossing apps across iOS and Android. Appia's advertisers are focused on driving performance mobile user acquisition because of its focus on optimization and quality.

Partners

Appia's partners include direct developers, carriers, mobile websites, and mediated relationships. Appia's partner base spans 200 countries and 1 billion users, and leans towards publishers that are seeking to monetize Android and iOS traffic globally through performance mobile advertising.

Sales and Marketing

Appia's sales force includes Advertising Sales and Business Development. The Advertising Sales team calls on direct mobile app advertisers and agencies representing mobile app advertisers to enable the more than 150 mobile advertisers that span 1,000 campaigns across iOS and Android. The Business Development team calls on carriers, mobile websites, and app developers on a global basis, to increase the supply of publisher and developer inventory. Additionally, Appia has a publisher self-service solution, Via, that is managed through marketing and utilizes marketing automation to support the supply of inventory.

Appia's marketing team drives demand in the market through a variety of channels including, but not limited to, search engine marketing, paid social advertising, targeted lead generation, banner advertising, and email marketing. In addition, it utilizes sophisticated marketing automation solutions to drive relevancy and engagement throughout the customer lifecycle.

Employees and Facilities

As of November 13, 2014, Appia had approximately 65 full-time employees. Appia leases facilities in Durham, North Carolina and San Francisco, California.

MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS OF THE BUSINESS OF APPIA

You should read the following discussion and analysis of Appia's financial condition and results of operations with Appia's audited and unaudited consolidated financial statements and related notes thereto included elsewhere in this proxy statement/prospectus. This discussion and analysis of financial condition and results of operations contains certain forward-looking statements, which are subject to risks, uncertainties and contingencies, which could cause Appia's actual business, results of operations or financial condition to differ materially from those expressed in, or implied by, those statements.

Components of Revenue and Expenses

Revenue

Advertising Revenue— Appia is a mobile advertising network focused on driving installs of mobile applications (both iOS and Android) on behalf of advertisers who want to grow their mobile audience. Mobile users view advertisements to download applications, and these advertisements are displayed in mobile applications and mobile websites. Revenues are recognized when the related advertising services are delivered based on the specific terms of the advertising contract/insertion order, which are primarily based on the number of applications installed. Appia pays distribution partners a percentage of the advertising revenue, which is included in cost of revenue. The advertising revenue derived is reported gross of the payment to the distribution partners as we are the primary obligor to the advertisers, who are the customers of the advertising service.

Cost of Revenue

Cost of revenue consists primarily of the payments Appia makes to its publisher and developer partners for their advertising space on which mobile ads are delivered. These payments are typically determined in advance as a percentage of the advertising revenue earned from mobile ads placed on the publisher's mobile page or developer's app or as a fixed fee for the ad space. Appia recognizes cost of revenue on a partner-by-partner basis at the same time as it recognizes the associated revenue. Costs owed to partners but not yet paid are recorded on the consolidated balance sheets as accrued cost of revenue.

Operating Expenses

Operating expenses consist of sales and marketing, technology, and general and administrative expenses. Salaries and personnel costs are the most significant component of each of these expense categories

Sales and marketing expense. Sales and marketing expense consists primarily of salaries and personnel costs for sales and marketing, sales operations, corporate marketing and business development, including commissions and bonuses. Additional sales and marketing expenses include marketing programs, lead generation, consulting, travel, and other related overhead.

Technology and development expense. Technology and development expense primarily consists of salaries, bonuses, and personnel costs for development and product management. Additional technology expenses include consulting, travel, and other related overhead.

General and administrative expense. General and administrative expense primarily consists of salaries and personnel costs for publisher and advertiser operations, technical operations employees, administration, finance, accounting, and human resources employees, including bonuses. Additional general and administrative expenses include costs related to Appia's ad serving network, including data center costs, stock based compensation, consulting and professional fees, travel, bad debt expense, insurance, and other corporate expenses.

Results of Operations

Nine Months Ended September 30, 2014 Compared to Nine Months Ended September 30, 2013

The following table presents an overview of results of operations for the nine-month periods ended September 30, 2014 and 2013:

	- 1	Nine Months Ended September 30,		d to Period Change	
	2014	2013	Amount	Percentage	
		(dollars in	thousands)		
Revenue	\$ 21,930	\$ 38,470	\$ (16,540)	(43.0)%	
Cost of revenue	17,366	27,912	10,546	37.8	
Gross profit	4,564	10,558	(5,994)	(56.8)	
Operating expense:					
Sales and marketing	2,569	2,655	86	3.2	
Technology	2,246	1,823	(423)	(23.2)	
General and administrative	3,431	4,426	995	22.5	
Total operating expense	8,246	8,904	658	7.4	
Loss from operations	(3,682)	1,654	(5,336)	(322.6)	
Other (expense)					
Interest expense—net	(3,215)	(387)	2,828	(730.7)	
Total other expense	(3,215)	(387)	2,828	(730.7)	
Net income (loss)	\$ (6,897)	\$ 1,267	\$ (8,164)	(644.4)	

Revenue

Total revenue decreased \$16.5 million, or (43.0%), to \$21.9 million for the nine months ended September 30, 2014 from \$38.5 million for the same period ended 2013. This decrease was primarily attributable to a decrease in revenue of \$18.1 million from one advertiser client. The revenue from this leading game developer peaked in calendar Q3 2013 with significantly larger marketing spend to promote a top title, and then declined as the game's sales decreased in the market. This advertiser continues to be a top client; however, at lower revenue levels than in 2013.

Expense

Cost of Revenue. Cost of revenue decreased \$10.5 million, or 37.8%, to \$17.4 million during the nine months ended September 30, 2014 from \$27.9 million during the same period ended 2013. This decrease is primarily the direct result of the decrease in Appia's revenue, as, in most cases, Appia pays a percentage of advertising revenue to developers and publishers for use of their ad space in delivering mobile ads for Appia's advertiser clients. This decrease was offset by an increase in the percentage of revenue share paid to partners due to competitive pressures.

Sales and Marketing. Sales and marketing expenses decreased \$0.1 million, or 3.2%, to \$2.6 million in the nine months ended September 30, 2014 from \$2.7 million in the same period ended 2013. The decrease in sales and marketing expense was primarily attributable to a decrease of \$0.2 million in employee related expenses offset by an increase of \$0.1 million in marketing programs and lead generation.

Technology. Technology expenses increased \$0.4 million, or 23.2%, to \$2.2 million in the nine months ended September 30, 2014 from \$1.8 million in the same period ended 2013. The increase is attributable

to an increase of \$0.8 million in employee related costs, an increase of \$0.2 million in contractor costs, and an increase of \$0.4 million of capitalized software amortization. This increase was offset by a decrease of \$1.0 million in expense related to additions to capitalized software.

General and Administrative. General and administrative expenses decreased \$1.0 million, or 22.5%, to \$3.4 million in the nine months ended September 30, 2014 from \$4.4 million in the same period ended 2013. This decrease was due to a decrease of \$0.5 million in employee related costs, legal costs and consulting costs. In addition, Appia recognized a reduction in expense of \$1.3 million related to the write off of liabilities assumed from a prior acquisition. This decrease was offset by an increase of \$0.8 million hosting and network related costs.

Income (Loss) from Operations. Income from operations decreased \$5.3 million to a loss of \$3.7 million in the nine months ended September 30, 2014, from income of \$1.7 million for the same period ended 2013. This decrease was a result of the decrease of gross profit and the increase in operating expense.

Other Expense. Other expense increased \$2.8 million to \$3.2 million in the nine months ended September 30, 2014, from \$0.4 million for the same period ended 2013. This increase was due to \$0.7 million of additional interest expense for subordinated debt, \$0.1 million of amortization of debt discount, and \$2.0 million of expense resulting from the change in fair value of Appia's preferred warrant liability associated with our subordinated debt.

Net Income (Loss). For the reasons discussed above, net income decreased \$8.2 million to a loss of \$6.9 million in the nine months ended September 30, 2014 from income of \$1.3 million in the same period ended 2013. Adjusted for \$1.3 million related to the write off of liabilities and \$2 million of interest expense related the change in fair value of preferred warrant liability, net loss would have been \$6.2 million in the nine months ended September 30, 2014.

Year Ended December 31, 2013 Compared to Year Ended December 31, 2012

The following table presents an overview of results of operations for the years ended December 31, 2013 and 2012:

	Year Ended	December 31,	Period to Per	riod Change
	2013	2012	Amount	Percentage
		(dollars in	thousands)	
Revenue	\$ 46,860	\$ 10,517	\$ 36,343	345.6%
Cost of revenue	33,863	7,086	(26,777)	(377.9)
Gross profit	12,997	3,431	9,566	278.8
Operating expense:				
Sales and marketing	3,334	2,245	(1,089)	(48.5)
Technology	2,797	2,531	(266)	(10.5)
General and administrative	6,647	3,669	(2,978)	(81.2)
Total operating expense	12,778	8,445	(4,333)	(51.3)
Loss from operations	219	(5,014)	5,233	(104.4)
Other (expense)				
Interest expense—net	(745)	(65)	680	(1,046.2)
Total other expense	(745)	(65)	680	(1,046.2)
Net income (loss)	<u>\$ (526)</u>	\$ (5,079)	\$ 4,553	(89.6)

Revenue

Total revenue increased \$36.3 million, or 345.6% to \$46.9 million in 2013 from \$10.5 million in 2012. This growth was primarily attributable to an increase in the number of advertiser clients using Appia's platform as well as an increase in spending from its existing advertiser clients. One advertiser client drove \$20.1 million, or 55.4%, of the increased revenue. The revenue from this leading game developer peaked in calendar Q3 2013 with significantly larger marketing spend to promote a top title.

Expense

Cost of Revenue. Cost of revenue increased \$26.8 million, or 377.9%, to \$33.9 million in 2013 from \$7.1 million in 2012, primarily the direct result of the increase in Appia's revenue, as, in most cases, Appia pays a percentage of that revenue to developers and publishers for use of their ad space in delivering mobile ads for Appia's advertiser clients.

Sales and Marketing. Sales and marketing expenses increased \$1.1 million, or 48.5%, to \$3.3 million in 2013 from \$2.2 million in 2012. The increase was primarily the result of an increase \$0.9 million in employee related expenses and a \$0.2 million increase in marketing programs.

Technology. Technology expenses increased \$0.3 million, or 10.5%, to \$2.8 million in 2013 from \$2.5 million in 2012. The increase was attributable to an increase of \$1.2 million in employee related costs, \$0.1 million of contractor costs, and \$0.1 million of capitalized software amortization. This increase was offset by a decrease of \$1.1 million in expense related to additions to capitalized software.

General and Administrative. General and administrative expenses increased \$3.0 million, or 81.2%, to \$6.7 million in 2013 from \$3.7 million in 2012. The increase was primarily the result of an increase in employee related expenses. Also contributing to the increase were professional fees, rent, hosting and network related costs, and other related overhead.

Income (*Loss*) *from Operations*. Income from operations increased \$5.2 million to \$0.2 million in 2013, from a loss of \$5.0 million in 2012. This increase in operating income was resultant of the increase in revenue and gross profit exceeding the increase in operating expense.

Other Expense. Other expense increased \$0.7 million or 1,046.2% to \$0.8 million in 2013 from \$0.1 million in 2012 reflecting interest expense associated with \$8.0 million of subordinated debt raised during the year.

Net Loss. For the reasons discussed above, net loss decreased \$4.6 million to \$0.5 million in 2013 from \$5.1 million in 2012.

Seasonality

Appia's revenue tends to be seasonal in nature, with the fourth quarter of each calendar year historically representing the largest percentage of our total revenue for the year. Many advertisers spend the largest portion of their advertising budgets during the fourth quarter, in preparation for the holiday season. During 2013, the second and third quarters were higher than the fourth quarter due to the surge in revenue from a leading game developer for its top game title.

Financial Condition

Assets

Appia's current assets totaled \$7.3 million and \$14.0 million at September 30, 2014 and September 30, 2013, respectively. Total assets were \$10.0 million and \$15.2 million at September 30, 2014 and September 30, 2013, respectively. The decrease in current assets was primarily due to the decrease in cash for operating expenses and a decrease in accounts receivable associated with the decrease in revenue.

Liabilities and Working Capital

At September 30, 2014, Appia's total liabilities were \$16.4 million, compared to \$13.1 million at September 30, 2013. The change in liabilities was mainly due to the decrease of accounts payable of \$1.3 million, offset by an increase in long-term debt of \$2.1 million and \$2.5 million of warrant liability. Appia had positive working capital of \$1.5 million and \$6.9 million at September 30, 2014 and September 30, 2013, respectively, for a decrease of \$5.4 million of working capital. This was mainly comprised of a decrease in cash and accounts receivable associated with a decrease in revenue, offset by a decrease in accounts payable.

Liquidity and Capital Resources

Appia's principal source of liquidity has historically been cash provided by financing activities. Appia's principal uses of cash have been to fund operations, capital expenditures and debt service requirements. Total cash and cash equivalents were \$2.0 million at September 30, 2014, compared to \$5.6 million at September 30, 2013.

The following table presents a summary of Appia's cash flow for the periods indicated (dollars in thousands):

	Nine months ended		Year Ended					
	 September 30,			December 3			31,	
	 2014		2013		2013		2012	
Cash provided by (used in):	_							
Operating activities	\$ (3,479)	\$	(1,459)	\$	(1,799)	\$	(6,385)	
Investing activities	(1,919)		(1,078)		(1,406)		(137)	
Financing activities	(438)		5,045		7,904		1,962	

Operating Activities

Net cash used in operations for the nine months ended September 30, 2014 was \$3.5 million compared to \$1.5 million for the nine months ended September 30, 2013. The \$2.0 million increase in cash used in operations was primarily due to an increase of \$8.2 million in net loss from operations offset by an decrease in accounts receivable and prepaid expenses of \$3.5 million, an increase in accounts payable and accrued expenses of \$0.5 million, and an increase of \$2.0 million in the fair value of preferred warrant liability.

Net cash used in operations for the year ended December 31, 2013 was \$1.8 million compared to \$6.4 million for year ended December 31, 2012. The \$4.6 million decrease in cash used in operations was primarily due to a decrease of \$4.6 million in the net loss from operations.

Investing Activities

Net cash used in investing activities for the nine months ended September 30, 2014 was \$1.9 million compared to \$1.1 million for the nine months ended September 30, 2013. This \$0.8 million increase in net cash used in investing activities was due to an increase of \$1.0 million in capitalization of internally developed software offset by a decrease in fixed asset purchases of \$0.2 million.

Net cash used in investing activities for the year ended December 31, 2013 was \$1.4 million compared to \$0.1 million for the year ended December 31, 2012. This \$1.3 million increase in net cash used in investing activities was principally due to the capitalization of internally developed software of \$1.2 million and an increase in fixed asset purchases of \$0.1 million.

Financing Activities

Net cash used by financing activities was \$0.4 million for the nine months ended September 30, 2014 compared to cash provided of \$5.0 million for the nine months ended September 30, 2013. The decrease of \$5.5

million in net cash used by financing activities is primarily due to a decrease of \$5.2 million related to the issuance of subordinated debt and an increase of \$0.3 million in term loan payments.

Net cash provided by financing activities was \$7.9 million for the year ended December 31, 2013 compared to \$2.0 million in 2012. The increase of \$5.9 million in net cash provided by financing activities is primarily due to an increase of \$6.9 million related to the issuance of subordinated debt and a decrease of \$0.4 million of payments on Appia's term debt, offset by a decrease of \$1.4 million related to the issuance of redeemable preferred stock.

Appia estimates that based on current plans and assumptions, that its available cash will not be sufficient to satisfy its liquidity requirements without further financing for the next 12 months. Its operations may be dependent on its ability to secure additional financing, which may include obtaining credit facilities, or other financing mechanisms. There is no assurance that future financing will be available to Appia on acceptable terms or at all.

Indebtedness

Short-term and long-term portions of debt are as follows as of September 30, 2014 (dollars in thousands):

Debt	2014	2013
	\$	\$
Short-term		
Bank debt	600	600
Subordinated debt	-	-
Total	600	600
Long-term		
Bank debt	300	900
Subordinated debt	8,000	5,000
Less - Debt discount	(801)	(564)
Total	7,499	5,336

In August 2012, Appia entered into a \$1.3 million term loan with a bank that bears interest at the greater of 2.5% + prime or 6.5% and is repayable in 36 months. At the same time Appia obtained a \$0.7 million line of credit based on outstanding accounts receivable that bears interest at the greater of 1.5% + prime or 5.50%. These facilities are secured by substantially all the assets of Appia and include non-financial covenants. Since then Appia has periodically amended the facilities to extend the maturity date and modify certain definitions. As of September 30, 2014, Appia had a line of credit of \$3.5 million with \$0 outstanding and the term loan has \$0.9 million outstanding maturing April 2016.

In April 2013, Appia borrowed \$5.0 million of subordinated debt with a commercial lender. The facility is interest only, repayable on April 2019. The loan bears interest of 10% per annum if paid monthly, otherwise accrues at 15% per annum and includes non-financial covenants. Facility includes a prepayment premium ranging from 1-5% of the unpaid balance depending on date of the prepayment. Appia also issued to the lender series C preferred stock warrants. In October 2013, Appia entered into a second facility with the same lender for \$3.0 million with same terms, including additional series C warrants, and repayment date. At September 30, 2014, \$8.0 million is outstanding. For a further description of the treatment of Appia's debt subsequent to the merger, see "The Merger – Debt Obligations and Guarantees" on page 69.

Contractual Obligations

Appia's contractual obligations and commitments as of September 30, 2014 are summarized in the table below (dollars in thousands):

		Less than		More than
	Total	1 year 1 - 3 years	3 - 5 years	5 years
		(in thousands)		
Long-term debt	\$ 900	\$ 600 \$ 300	\$ —	\$ —
Subordinated debt	8,000		8,000	_
Operating leases	313	<u>313</u> —		
Total contractual obligations	\$ 9,213	\$ 913 \$ 300	\$ 8,000	\$ —

Appia's contractual obligations and commitments as of December 31, 2013 are summarized in the table below:

	Less than		More than	
	Total	1 year 1 - 3 y	ears 3 - 5 years	5 years
		(in thou	sands)	
Long-term debt	\$ 1,350	\$ 600 \$	750 \$ —	\$ —
Subordinated debt	8,000		<u> </u>	8,000
Operating leases	560	386	<u> </u>	
Total contractual obligations	\$ 9,910	\$ 986 \$	924 \$ —	\$ 8,000

Summary of Critical Accounting Policies and Estimates

Appia's management's discussion and analysis of our financial condition and results of operations is based on our consolidated financial statements, which have been prepared in accordance with United States Generally Accepted Accounting Principles ("U.S. GAAP"). The preparation of these consolidated financial statements requires us to make estimates and assumptions that affect the reported amounts of assets and liabilities, disclosure of contingent assets and liabilities at the date of the consolidated financial statements and the reported amounts of revenue and expenses during the reported period. In accordance with U.S. GAAP, Appia bases its estimates on historical experience and on various other assumptions that it believes are reasonable under the circumstances. Actual results may differ from these estimates.

Revenue Recognition

Appia recognizes revenue based on the activity of mobile users viewing advertisements and downloading applications via developer applications and mobile websites. Revenues are recognized when the related advertising services are delivered based on the specific terms of the advertising contract/insertion order, which are primarily based on the number of applications installed or downloaded. Appia recognizes revenue based on these terms, as the services have been provided, the fees Appia charges are fixed or determinable, persuasive evidence of an arrangement exists, and collectability is reasonably assured.

In the normal course of business, Appia acts as an intermediary in executing transactions with third parties. The determination of whether revenue should be reported on a gross or net basis is based on an assessment of whether Appia is acting as the principal or an agent in the transaction. In determining whether Appia acts as the principal or an agent, Appia follows the accounting guidance for principal-agent considerations. While none of the factors identified in this guidance is individually considered presumptive or determinative,

because Appia is the primary obligor and is responsible for (i) identifying and contracting with third-party advertisers, (ii) performing all billing and collection activities including retaining credit risk, and (iii) discretion in selecting suppliers, Appia acts as the principal in these arrangements and therefore reports revenue earned and costs incurred on a gross basis.

Cost of Revenue

Cost of revenue consists primarily of amounts due to publishers for the advertising inventory utilized in running mobile advertisements. These amounts are typically either a percentage of the advertising revenue earned by Appia based on mobile advertisements that are run on each publishers inventory or a fixed fee for the ad space. Appia recognizes the cost of revenue as the associated revenue is recognized, on a publisher by publisher basis during the period the advertisements run on the publisher's properties. Costs owed to publisher but not yet paid are recorded as accrued cost of revenue.

Fair Value Measurements

Appia follows ASC 820 – "Fair Value Measurement", which provides a common definition of fair value as the price that would be received to sell an asset or paid to transfer a liability in a transaction between market participants. ASC 820 does not require any new fair value measurements, but applies to most current accounting pronouncements that require or permit fair value measurements.

The standard provides guidance on the methods used to measure fair value, including the market, income, and cost approach. These approaches require the use of certain assumptions in models that market participants would use in pricing assets and liabilities. The models may use readily observable, market corroborated, or generally unobservable inputs.

The standard also requires assets and liabilities that are measured at fair value on a recurring basis (at least annually) to be classified and disclosed in one of the following three categories:

- Level 1 Quoted market prices in active markets for identical assets or liabilities;
- Level 2 Observable market-based inputs or unobservable inputs that are corroborated by market data; and
- Level 3 Unobservable inputs that are not corroborated by market data.

The fair values of Appia's cash equivalents, accounts receivable, accounts payable and accrued liabilities approximate their carrying values because of their short-term nature. Appia's debt approximates fair value due to the nature of having variable interest rates.

Appia has issued warrants to purchase preferred stock in connection with the issuance of subordinated debt. Appia accounted for these warrants in long-term liabilities at fair value upon the issuance based on the specific terms of the agreement. The fair value of the preferred stock warrants was estimated using the Black-Scholes option-pricing model. The warrants are not actively traded and are valued using observable and unobservable market data for inputs. Significant market data inputs used to determine the fair value of the warrants include the equity value of the transaction, estimated the business enterprise value of Appia using a discounted cash flow analysis, expected life of the warrants, risk-free interest rate and the implied volatility of Appia's stock.

Appia follows ASC 825 – "Financial Instruments" which permits companies to voluntarily choose, at specified election dates, to measure specified financial assets and liabilities and other items at fair value that are not currently required to be measured at fair value. Subsequent changes in fair value would then be required to be reported in earnings each reporting period. Appia currently has not elected the fair value option for any eligible items. Any future effect of this pronouncement will be dependent upon the nature and amount of eligible items that Appia elects to account for using the fair value option.

Redeemable Convertible Preferred Stock

The carrying value of redeemable convertible preferred stock is increased each reporting period such that the carrying amount will equal the redemption amount at the redemption date. These increases, if any, are affected through charges against additional paid-in capital, to the extent it is available, and then against accumulated deficit.

Off-Balance Sheet Arrangements

Appia does not have, and has never had, any relationships with unconsolidated entities or financial partnerships, such as entities often referred to as structured finance or special purpose entities, which would have been established for the purpose of facilitating off-balance sheet arrangements or other contractually narrow or limited purposes.

DESCRIPTION OF MANDALAY DIGITAL COMMON STOCK

The following description summarizes the material terms and provisions of our capital stock. For the complete terms of our capital stock, please refer to our certificate of incorporation, as amended, and our bylaws, as amended, which are exhibits to the registration statement of which this proxy statement/prospectus is a part. The terms of these securities may also be affected by the General Corporation Law of the State of Delaware. The summary below is qualified in its entirety by reference to our certificate of incorporation and our bylaws.

Authorized Capitalization

We have 202,000,000 shares of capital stock authorized under our certificate of incorporation, consisting of 200,000,000 shares of common stock, par value \$0.0001 per share, and 2,000,000 shares of preferred stock, of which 100,000 have been designated as Series A Convertible Preferred Stock, par value \$0.0001 per share, or Series A Preferred Stock. As of November 13, 2014, we had 37,797,221 shares of common stock outstanding and 100,000 shares of our Series A Preferred Stock outstanding which are currently convertible into 20,000 shares of common stock. Our authorized shares of common stock and preferred stock are available for issuance without further action by our stockholders, unless such action is required by applicable law or the rules of any stock exchange or automated quotation system on which our securities may be listed or traded. If the approval of our stockholders is not so required, our board of directors may determine not to seek stockholder approval.

Common Stock

Holders of our common stock are entitled to such dividends as may be declared by our board of directors out of funds legally available for such purpose, subject to any preferential dividend rights of any then outstanding preferred stock. The shares of common stock are neither redeemable nor convertible. Holders of common stock have are not entitled to preemptive or subscription rights to purchase any of our securities under our charter documents.

Each holder of our common stock is entitled to one vote for each such share outstanding in the holder's name. No holder of common stock is entitled to cumulate votes in voting for directors.

In the event of our liquidation, dissolution or winding up, the holders of our common stock are entitled to receive pro rata our assets that are legally available for distribution, after payments of all debts and other liabilities and subject to the prior rights of any holders of preferred stock then outstanding. All of the outstanding shares of our common stock are, and the shares of common stock issued upon the conversion of any securities convertible into our common stock will be, fully paid and non-assessable. The shares of common stock offered by this prospectus will also be, fully paid and non-assessable.

Our common stock is listed on the NASDAQ Stock Market under the symbol "MNDL." American Stock Transfer is the transfer agent and registrar for our common stock. Its address is 6201 15th Avenue Brooklyn, NY 11219, and its telephone number is (800) 937-5449.

Preferred Stock

Our certificate of incorporation permits us to issue up to 2,000,000 shares of preferred stock in one or more series and with rights and preferences that may be fixed or designated by our board of directors without any further action by our stockholders.

Series A Convertible Preferred Stock

We currently have 100,000 shares of our Series A Preferred Stock designated, and as of November 13, 2014, we had 100,000 shares of our Series A Preferred Stock outstanding. While shares of our Series A Preferred Stock are outstanding, holders of the Series A Preferred Stock are entitled to receive any dividends if and when declared by the Company's board of directors on the Company's common stock on an as-converted basis.

The Series A Preferred Stock is convertible at any time at the option of the holder into shares of our common stock based on dividing the original purchase price plus the amount of any accumulated but unpaid dividends, by the conversion price then in effect (as may be adjusted).

The Series A Preferred Stock is entitled to vote together with the common stock as a single class (on an as-converted to common stock basis) on any matters submitted to the holders of the Company's common stock, together with any other voting rights provided to the Series A Preferred under law or the General Corporation Law of the State of Delaware.

The Series A Preferred Stock is entitled to receive, prior and in preference to our common stock or any other class designated as junior to the Series A Preferred Stock, upon liquidation, dissolution or winding up of the Company, whether voluntary or involuntary, or in the event of its insolvency, an amount per share equal to the greater of (i) \$10.00 per share of Series A Preferred Stock (subject to certain adjustments) or (ii) such amount per share as would have been payable had the Series A Preferred Stock been converted into our common stock immediately prior to such liquidation, dissolution or winding up. Each holder of Series A Preferred Stock also has the right to a cash-out election in the event of certain transactions, including a consolidation or merger of the Company (excluding a transaction involving a reincorporation or a merger with a wholly-owned subsidiary), a sale of all or substantially all of the assets of the Company, the issuance by the Company in a single or integrated transaction shares of common stock (or securities convertible into common stock) representing a majority of the shares of common stock outstanding immediately following such issuance, or any other form of acquisition where the Company is the target and a change of control occurs such that the acquirer has the power to elect a majority of the Company's board of directors.

Anti-Takeover Effects of Certain Provisions of Delaware Law

The following is a summary of certain provisions of Delaware law. This summary does not purport to be complete and is qualified in its entirety by reference to the corporate law of Delaware and our certificate of incorporation and bylaws.

Effect of Delaware Anti-Takeover Statute. We may be subject to Section 203 of the Delaware General Corporation Law, an antitakeover law. In general, Section 203 prohibits a Delaware corporation from engaging in any business combination with any interested stockholder for a period of three years following the date that the stockholder became an interested stockholder, unless:

- prior to that date, the board of directors of the corporation approved either the business combination or the transaction that resulted in the stockholder becoming an interested stockholder;
- upon consummation of the transaction that resulted in the stockholder becoming an interested stockholder, the interested stockholder owned at least 85% of the voting stock of the corporation outstanding at the time the transaction commenced, excluding for purposes of determining the number of shares of voting stock outstanding (but not the voting stock owned by the interested stockholder) those shares owned by persons who are directors and officers and by excluding employee stock plans in which employee participants do not have the right to determine whether shares held subject to the plan will be tendered in a tender or exchange offer; or
- on or subsequent to that date, the business combination is approved by the board of directors of the corporation and authorized at an annual or special meeting of stockholders, and not by written consent, by the affirmative vote of at least 66-2/3% of the outstanding voting stock that is not owned by the interested stockholder.

Section 203 defines "business combination" to include the following:

- any merger or consolidation involving the corporation and the interested stockholder;
- any sale, transfer, pledge or other disposition of 10% or more of the assets of the corporation involving the interested stockholder;
- subject to certain exceptions, any transaction that results in the issuance or transfer by the corporation of any stock of the corporation to the interested stockholder;
- any transaction involving the corporation that has the effect of increasing the proportionate share of the stock of any class or series of the corporation beneficially owned by the interested stockholder; or
- the receipt by the interested stockholder of the benefit of any loans, advances, guarantees, pledges or other financial benefits provided by or through the corporation.

In general, Section 203 defines an interested stockholder as any entity or person beneficially owning 15% or more of the outstanding voting stock of the corporation, or who beneficially owns 15% or more of the outstanding voting stock of the corporation at any time within a three-year period immediately prior to the date of determining whether such person is an interested stockholder, and any entity or person affiliated with or controlling or controlled by any of these entities or persons.

COMPARISON OF RIGHTS OF STOCKHOLDERS OF MANDALAY DIGITAL AND APPIA

This section of the registration statement/prospectus describes the material differences between the rights of the holders of Mandalay Digital stock and holders of Appia stock. While Mandalay Digital believes that the description covers the material differences between the two, this summary may not contain all of the information that is important to you. You should carefully read this entire document and the other documents we refer to for a more complete understanding of the differences between being a stockholder of Mandalay Digital and being a stockholder of Appia.

Both Mandalay Digital and Appia are incorporated under the laws of the State of Delaware and, accordingly, the rights of the stockholders of each are currently, and will continue to be, governed by the Delaware General Corporation Law (the "DGCL"). The rights of Mandalay Digital contained in its restated certificate of incorporation, as amended, and bylaws, as amended, of Mandalay Digital differ from the rights of Appia stockholders under the amended and restated certificate of incorporation and bylaws of Appia. If the merger is completed, Appia stockholders will become stockholders of Mandalay Digital, and their rights will be governed by the DGCL, the certificate of incorporation and bylaws of Mandalay Digital. The following is a summary of the material differences as of the date of this proxy statement/prospectus between the rights of Mandalay Digital stockholders under Mandalay Digital's certificate of incorporation and bylaws, which are referred to as the Mandalay Digital charter and Mandalay Digital bylaws, respectively, and the rights of Appia stockholders under Appia's amended and restated certificate of incorporation and bylaws, which are referred to as the Appia charter and Appia bylaws. Copies of the Mandalay Digital charter, the Mandalay Digital bylaws, the Appia charter and the Appia bylaws will be sent to holders of Mandalay Digital common and preferred stock or Appia common and preferred stock upon request. See "Where You Can Find More Information" beginning on page 153.

The following discussion of these differences is only a summary of the material differences and does not purport to be a complete description of all the differences. Mandalay Digital and Appia urge you to consult the DGCL and read each of their charters and bylaws in their entirety, each as amended, restated, supplemented or otherwise modified from time to time, for a more complete understanding of these differences.

Authorized Capital; Outstanding Capital Stock

Mandalay Digital Stockholder Rights

The authorized capital stock of Mandalay Digital consists of 200,000,000 shares of common stock, par value \$0.00001 per share, and 2,000,000 shares of preferred stock, par value \$0.00001 per share

As of November 13, 2014, there were approximately 37,797,221 shares of Mandalay Digital common stock outstanding that were held of record by approximately [\square] persons and there were 100,000 shares of Mandalay Digital preferred stock outstanding.

Blank Check Preferred Stock

The Mandalay Digital charter provides that the board of directors is authorized to establish and designate one or more series of preferred stock, to fix the number of shares constituting such series, and to fix the voting powers, designations, powers, preferences and relative participating, optional or other special rights, and qualifications, limitations or restrictions thereof.

Number of Directors

The Mandalay Digital charter provides that the number of directors shall be fixed by the Mandalay Digital bylaws. The Mandalay Digital bylaws authorize the board of directors to fix the number of directors.

Appia Stockholder Rights

The authorized capital stock of Appia consists of 17,625,600 shares of common stock, par value \$0.001 per share and 11,537,614 shares of preferred stock, par value \$0.001 per share.

The preferred stock is divided into five series: Series A Preferred consists of 3,275,000 shares. Series B Preferred consists of 2,569,169 shares. Series C Preferred consists of 2,612,818 shares. Series D Preferred consists of 89,127 shares. Series FF Preferred consists of 2,991,500 shares.

As of November 13, 2014, there were approximately 3,927,706 shares of Appia common stock outstanding and 10,273,610 shares of Appia preferred stock outstanding that were held of record by approximately 40 persons.

The Appia charter provides that any increase or decrease in the authorized number of shares of preferred stock must be approved by at least two-thirds of the outstanding shares of the Series A Preferred stock, Series B Preferred stock, Series C Preferred stock, Series D Preferred stock (collectively, the "Senior Preferred"). In addition, each series of preferred (excluding Series FF) voting as a separate class must approve the increase or decrease in the authorized number of shares in its own series by a majority vote.

The Appia charter provides that the number of directors shall be fixed by the Appia bylaws. The Appia bylaws authorize the board of directors to determine the number of directors.

Rights of Holders of Preferred Stock to Elect Directors

The Mandalay Digital charter authorizes preferred stock, but no rights to elect directors are attached to preferred stock.

The Appia charter authorizes the holders of (i) Series A Preferred stock to elect one director, (ii) Series B Preferred stock to elect two directors, (iii) Series C Preferred stock to elect one director, (iv) common stock and Series FF Preferred stock, voting together as a single class, to elect two directors, and (v) common stock, Series FF Preferred and Senior Preferred (as defined therein), voting together as one class, to elect all remaining directors.

Classification of Board of Directors

The Mandalay Digital charter and Mandalay Digital bylaws do not provide for different classifications of the board of directors. The Appia charter and Appia bylaws do not provide for different classifications of the board of directors.

Removal of Directors

The Mandalay Digital bylaws provide that a director may be removed, with or without cause, by vote of the holders of a majority of the shares then entitled to vote.

The Appia charter and bylaws provide that a director may be removed, with or without cause, by vote of the holders of a majority of the shares then entitled to vote at an election of such director.

Vacancies on the Board of Directors

The Mandalay Digital bylaws provide that any vacancy occurring on the board of directors may be filled by vote of the stockholders or stockholders' written consent, or by vote of the board or directors' written consent. If the number of directors then in office is less than a quorum, than directors may fill the vacancy by majority vote.

The Appia charter provides that each series of Preferred Stock that has the right to elect a director also has the right to fill any vacancy caused by the resignation, death or removal of the director elected by that series of Preferred stock. The Appia bylaws provide that any vacancy may be filled by a majority vote of the remaining directors. Stockholders may elect a director to fill a vacancy not filled by the directors.

Limitation on Liability

The Mandalay Digital charter provides that no director shall be personally liable to Mandalay Digital or its stockholders for monetary damages for breach of fiduciary duty as a director, except for liability for a director's acts or omissions that:

- The Appia charter provides that the liability of the directors for monetary damages shall be eliminated to the fullest extent under applicable law.
- were in breach of the director's duty of loyalty to Mandalay Digital or its stockholders;
- were not in good faith or involved intentional misconduct or a knowing violation of the law;

- resulted in a violation of section 174 of the DGCL for unlawful payment of a dividend or unlawful stock purchases or redemptions; or
- involved transactions from which the director derived an improper personal benefit.

Indemnification

The Mandalay Digital charter and Mandalay Digital bylaws provide that a director shall not be liable for breach of fiduciary duty to the fullest extent permitted by the DGCL.

The Mandalay Digital bylaws provide indemnification to each person who was or is a party or is threatened to be made a party to any threatened, pending or completed action, suit or proceeding, whether civil criminal, administrative or investigative, by reason of the fact that such person was or is a director, officer, or employee of Mandalay Digital, or is or was serving at the request of Mandalay Digital as a director, officer, manager, employee or agent of another corporation, limited liability company, partnership, joint venture, trust, or other enterprise, against all expenses, liability and loss reasonably incurred provided that such person acted in good faith and in a manner such person reasonably believed to be in or not opposed to the best interests of Mandalay Digital, except that no indemnification shall be made in respect of any claim, issue or matter as to which such person shall have been adjudged to be liable to Mandalay Digital unless and only to the extent that a court having jurisdiction shall determine that despite the adjudication of liability but in view of all the circumstances of the case, such person is fairly and reasonably entitled to indemnification.

The Appia charter provides that Appia is authorized to provide indemnification of, and advancement of expenses to, directors, officers, employees, and other agents to the fullest extent permitted by applicable law.

The Appia bylaws provide that Appia shall provide indemnification to each person who was or is a party or is threatened to be made a party to any threatened, pending or completed action, suit or proceeding, whether civil criminal, administrative or investigative, by reason of the fact that such person was or is a director, officer or employee of Appia, or is or was serving at the request of Appia as a director, partner, officer, manager, employee or agent of another corporation, limited liability company, partnership, joint venture, trust, or other enterprise, against all expenses, judgments, fines and amounts paid in settlement actually and reasonably incurred by such person in connection with such action, suit or proceeding if he acted in good faith and in a manner such person reasonably believed to be in or not opposed to the best interests of Appia.

The Appia bylaws further provide that Appia shall provide indemnification to each person who was or is a party or is threatened to be made a party to any threatened, pending or completed action, suit or proceeding brought by or in the right of the Corporation to procure a judgment in its favor by reason of the fact that such person was or is a director, officer, employee or agent of Appia, or is or was serving at

the request of Appia as a director, partner, officer, manager, employee or agent of another corporation, limited liability company, partnership, joint venture, trust, or other enterprise, against all expenses, judgments, fines and amounts paid in settlement actually and reasonably incurred by such person in connection with such action, suit or proceeding if the person acted in good faith and in a manner such person reasonably believed to be in or not opposed to the best interests of Appia, except that no indemnification shall be made in respect of any claim, issue or matter as to which such person shall have been adjudged to be liable to Appia unless and only to the extent that the Delaware Court of Chancery or the court in which action or suit was brought shall determine upon application that, despite the adjudication of liability but in view of all the circumstances of the case, such person is fairly and reasonably entitled to indemnity for such expenses which the Court of Chancery or such other court shall deem proper.

The Appia bylaws provide that the board of directors will make a determination of the right to indemnification under the standard of conduct detailed in the bylaws.

The Appia bylaws provide that any action required or permitted to be taken by the stockholders at annual or special meetings may be taken by consent in writing, as permitted by the DGCL.

The Appia bylaws provide that special meetings of Appia stockholders may be called by the President or board of directors. The President or board of

written request of any director or holder of at least

10% of the shares entitled to vote at the meeting.

directors shall call a special meeting upon the

Stockholder Action by Written Consent

The Mandalay Digital bylaws provide that any action required or permitted to be taken by the stockholders at annual or special meetings may be taken by consent in writing, as permitted by the DGCL.

Special Meetings of Stockholders

The Mandalay Digital bylaws provide that a special meeting of Mandalay Digital stockholders may be called by the board of directors, the chairman of the board, the CEO, the president, or the holders of at least a majority of the issued and outstanding shares of Mandalay Digital.

Notice of Stockholder Meetings

The Mandalay Digital bylaws provide that notice must be given to each Stockholder of record entitled to vote not less than 10 nor more than 60 days before the meeting.

Stockholder Proposals

The Mandalay Digital bylaws do not include any provisions governing stockholder proposals.

Stockholder Nominations

The Mandalay Digital bylaws do not include any provisions governing stockholder nominations for the board of directors.

Amendment of Certificate of Incorporation

The Mandalay Digital charter provides that the charter may be amended in the manner prescribed by the DGCL. If required by the DGCL, the Mandalay Digital charter may be amended by the adoption of a resolution of the Mandalay Digital board of directors, followed by the vote of a majority of the outstanding voting power entitled to vote thereon and a majority of the outstanding stock of each class entitled to vote thereon as a separate class.

The Appia bylaws provide that notice must be given to each Stockholder of record entitled to vote not less than 10 nor more than 60 days before the meeting.

The Appia bylaws do not include any provisions governing stockholder proposals.

The Appia bylaws do not include any provisions governing stockholder nominations for the board of directors.

Under the DGCL, the Appia charter may be amended by the adoption of a resolution of the Appia board of directors, followed by the vote of a majority of the outstanding voting power entitled to vote thereon and a majority of the outstanding stock of each class entitled to vote thereon as a separate class.

The Appia charter provides that any amendment or repeal of the charter that alters the voting or other powers, preferences, special rights or privileges of the Preferred stock must be approved by at least two-thirds of the outstanding shares of Senior Preferred voting on an as-if converted basis.

Any amendment, waiver or repeal of any provision of the charter that changes the rights, preferences, privileges, powers or restrictions of an individual series of Preferred stock requires the approval of the holders of a majority of the outstanding shares of that series of Preferred stock unless (A) such amendment, waiver or repeal has been approved by the holders of at least two-thirds of the outstanding shares of the Senior Preferred voting together as a single voting group on an asconverted basis, and (B) such amendment, waiver or repeal affects all shares of Senior Preferred in a similar manner.

Amendment of Bylaws

The Mandalay Digital bylaws may be amended, repealed or adopted, by the vote of the holders of a majority of the shares then entitled to vote or by the stockholders' action by written consent, or by vote of the board of directors or by the directors' action by written consent.

The Appia bylaws provide that the bylaws may be amended or repealed by a vote of the majority of all directors or by the affirmative vote of the holders of the shares then issued and entitled to vote that constitute at least a majority of the aggregate voting power of the outstanding capital stock of Appia.

The Appia charter authorizes the board of directors to adopt, amend or repeal the bylaws. The stockholders also have the power to adopt, amend or repeal the bylaws by vote of the holders of at least a majority of the voting power of all the then outstanding shares entitled to vote.

The Appia charter provides that any amendment or repeal of the bylaws that alters the voting or other powers, preferences, special rights or privileges of the preferred stock must be approved by at least two-thirds of the outstanding shares of Senior Preferred voting on an as-if converted basis.

Any amendment, waiver or repeal of any provision of Appia's bylaws that changes the rights, preferences, privileges, powers or restrictions of an individual series of Preferred stock requires the approval of the holders of a majority of the outstanding shares of that series of Preferred stock unless (A) such amendment, waiver or repeal has been approved by the holders of at least two-thirds of the outstanding shares of the Senior Preferred voting together as a single voting group on an asconverted basis, and (B) such amendment, waiver or repeal affects all shares of Senior Preferred in a similar manner.

Quorum

The Mandalay Digital bylaws provide that the holders of a majority of the issued and outstanding shares of common stock entitled to vote shall constitute a quorum at each meeting of stockholders.

The Mandalay Digital bylaws provide that a majority of the total number of directors then in office shall constitute a quorum at each meeting of the board of directors.

The Appia bylaws provide that the holders of a majority of the issued and outstanding shares of common stock and preferred stock entitled to vote shall constitute a quorum at each meeting of stockholders.

The Appia bylaws provide that a majority of the total number of directors then in office shall constitute a quorum at each meeting of the board of directors.

Certain Business Combinations/Anti-Takeover Provisions Under Delaware law, a corporation can elect not to be governed by section 203 of the DGCL, which generally protects publicly held Delaware corporations from unfair transactions and tactics by persons who acquire large blocks of stock without prior board approval. Mandalay Digital's charter does not provide for any election in connection with section 203 and is therefore subject to the restrictions of section 203 of the DGCL.

In general, section 203 prohibits a publicly held Delaware corporation from engaging in a "business combination" with an "interested" stockholder for a period of three years after the date of the transaction in which the person became an interested stockholder, unless the business combination or the transaction by which the person became an interested stockholder is approved in a prescribed manner. A "business combination" includes certain mergers, asset sales and other transactions resulting in a financial benefit to the interested stockholder.

Subject to exceptions, an "interested" stockholder is a person who, alone or together with his affiliates and associates, owns 15% or more of the corporation's voting stock.

These provisions could have the effect of delaying, deferring or preventing a change in control of Mandalay Digital or reducing the price that certain investors might be willing to pay in the future for Mandalay Digital common stock.

Appia is not subject to section 203 of the DGCL because shares of Appia are not publicly traded.

PROPOSAL 2

ELECTION OF DIRECTORS

Information Concerning Current Directors and Nominees

Our Board has six members, all of whose terms expire at our annual meeting. All directors are nominated for reelection or election, as applicable, to a term that will expire at our 2016 annual meeting. Each of our nominees was nominated based on the assessment of our Nominating and Governance Committee ("Governance Committee") and our Board that he has demonstrated: an ability to make meaningful contributions to our Board; relevant business experience; strong communication and analytical skills; a reputation for honesty and ethical conduct; excellent decision-making ability; and good judgment. Our Board consists of, and seeks to continue to include, persons whose diversity of skills, experience and background are complementary to those of our other directors.

While management presently has no knowledge that the nominees will refuse or be unable to serve as directors for the prescribed term, the accompanying proxy card grants the proxy holders the power to vote the proxy for substitute nominees in the event that the nominees become unavailable to serve.

The current board members, each of whom is a nominee, except Mr. Adderton, are as follows:

Name	Age	Position
Robert Deutschman(1)	57	Chairman of the Board & Nominee
Peter Guber	72	Current Director & Nominee
Jeffrey Karish ⁽²⁾	41	Current Director & Nominee
Christopher Rogers(3)	56	Current Director & Nominee
Paul Schaeffer ⁽⁴⁾	67	Current Director & Nominee
William G. Stone III	46	Chief Executive Officer, Current Director & Director Nominee

- (1) Chairman of our Audit Committee, Member of our Governance Committee
- (2) Member of our Compensation Committee and Chairman of our Governance Committee
- (3) Member of our Audit Committee and Chairman of our Compensation Committee
- (4) Member of our Audit Committee

Director Nominees

Robert Deutschman. Mr. Deutschman joined our Board of Directors on May 23, 2013 and was appointed Chairman of the Board in December 2014. Mr. Deutschman serves as Vice Chairman of Cappello Group, Inc., a merchant bank, since 2008. Prior to joining Cappello, Mr. Deutschman was a Managing Director of Saybrook Capital Corp., and a Senior Vice President at Houlihan, Lokey. Mr. Deutschman holds a Bachelor of Arts degree from Haverford College, with honors, and a Juris Doctor from Columbia University School of Law, where he was a Harlan Fiske Stone scholar. Since 2004, Mr. Deutschman serves as the Vice Chairman of the Board of Directors of Enron Creditors Recovery Corp. (formerly Enron Corp.), a position he assumed upon Enron's 2004 emergence from bankruptcy; to date, the creditors of Enron have received in excess of \$25 billion in proceeds from the liquidation of the Enron estate. Mr. Deutschman also serves on the board of the RAND Center for Corporate Ethics and Governance. The board of directors nominated Mr. Deutschman to serve as a director based on the entirety of his experience and skills, although the board noted specifically his extensive investment banking and financial experience and background in strategic advising, mergers and acquisitions and capital raising for institutions and private companies.

Peter Guber. Mr. Guber served as non-executive Chairman or non-executive Co-Chairman of the Board of Directors since August 2007 and until December 2014. Mr. Guber is Chairman and CEO of the multimedia Mandalay Entertainment Group. Prior to Mandalay Digital, Mr. Guber was Chairman and CEO of Sony Pictures Entertainment, Chairman and CEO of Polygram Entertainment, Co-Founder of Casablanca Record & Filmworks and President of Columbia Pictures. Mr. Guber produced or executive produced (personally or through his companies) films that garnered five Best Picture Academy Award nominations (winning for Rain Man) and box office hits that include The Color Purple, Midnight Express, Batman, Flashdance and The Kids Are All Right. Mr. Guber is the Owner and Co-executive Chairman of the NBA franchise, the Golden State Warriors. He is also a weekly entertainment and media analyst for Fox Business News and a full professor at UCLA. Mr. Guber serves on the board of directors of Demand Media, an online media publishing company (NYSE: DMD), and is Co-Founder of Geek Chic Daily, a daily email newsletter with inside information on technology & apps, video games, comics, TV & film. Mr. Guber is a noted author with works including "Inside The Deep" and "Shootout: Surviving Fame and (Mis)Fortune in Hollywood." Mr. Guber wrote the cover article for the Harvard Business Review titled, "The Four Truths of the Storyteller" and has also authored op-ed pieces for the New York Times and the San Francisco Chronicle. Mr. Guber recently released his third book, "Tell To Win — Connect, Persuade, and Triumph with the Hidden Power of Story", which became a #1 New York Times bestseller. The board of directors nominated Mr. Guber to serve as a director based on the entirety of his experience and skills, although the board noted specifically his extensive experience in the media and entertainment industries.

Jeffrey Karish. Mr. Karish has been a member of our Board of Directors since May 23, 2013. Mr. Karish currently serves as President of Windsor Media, Inc. Windsor Media's principal lines of business include a private equity fund, a venture capital fund and fixed income management. Previously, Mr. Karish was Head of Media Strategy and Corporate Development at Yahoo!, Inc. Before Yahoo, he was a management consultant at McKinsey & Company, and a key member of McKinsey's West Coast Media and Technology practice.

Mr. Karish has been a director of the Banc of California, Inc. since 2011. Mr. Karish holds a Juris Doctor from Harvard University, a Masters of Philosophy in International Economics from Cambridge University and a Bachelor of Arts in History from the University of California Berkeley. The board appointed Mr. Karish to serve as a director based on the entirety of his experience and skills, although the Board specifically noted his extensive strategic and managerial expertise, and extensive background in media and corporate development.

Christopher Rogers. Mr. Rogers has been a member of our Board of Directors since May 2012. Mr. Rogers is a partner at Lumia Capital. Previously he has served as Senior Vice President, Corporate Development and Spectrum, of Sprint Nextel Corporation where he evaluated and executed strategic initiatives, including mergers, acquisitions, divestitures, equity investments and joint ventures within the mobile communication and e-commerce sectors. He also was responsible for management and oversight of wireless spectrum licenses and Sprint Nextel's investment portfolio of emerging technology start-ups. Prior to its merger with Sprint in 2005, Rogers was Co-Founder and Senior Vice President of Nextel Communications, Inc. as well as Co-Founder of FleetCall Communications, the predecessor to Nextel Communications, and Founder and Chairman of Dispatch Communications, Inc., which was sold to Fleet Call/Nextel in 1993. Rogers holds a Juris Doctor in Communications Law and has served as a Director on multiple public and private company Boards and as a Director for several Washington, DC-based philanthropic organizations. The board appointed Mr. Rogers to serve as a director based on the entirety of his experience and skills, although the Board specifically noted his extensive communications expertise, particularly in strategy, mergers and acquisitions and licensing, and as well as his deep managerial and corporate development experience.

Paul Schaeffer. Mr. Schaeffer has been a member of our Board of Directors since August 2007. He is the Vice Chairman, Chief Operating Officer and Co-Founder of Mandalay Entertainment Group. Along with Peter Guber, Mr. Schaeffer is responsible for all aspects of Mandalay Entertainment Group's motion picture and television business, focusing primarily on the corporate and business operations. Prior to forming Mandalay Entertainment Group, Mr. Schaeffer was the Executive-Vice President of Sony Pictures Entertainment ("SPE"), overseeing the worldwide corporate operations for SPE including Worldwide Administration, Financial Affairs,

Human Resources, Corporate Affairs, Legal Affairs and Corporate Communications. Mr. Schaeffer is a member of the Academy of Motion Pictures, Arts, & Sciences. A veteran of 20 years of private law practice, Mr. Schaeffer joined SPE from Armstrong, Hirsch and Levine, where he was a senior partner working with corporate entertainment clients. He also spent time as an accountant with Arthur Young & Company in Philadelphia. He graduated from the University of Pennsylvania Law School and received his accounting degree from Pennsylvania State University. The Company considered Mr. Schaeffer to be a valuable resource when it selected him as a director based on his having served for more than 5 years as the Chairman of the Finance Committee, and a member of the Board of Trustees of Children's Hospital Los Angeles, where he also served as a chairman of its Audit Committee, and member of its Compensation Committee and Executive Committee for more than five years.

William Stone III. Mr. Stone became our Chief Executive Officer in October 2014 and was appointed as a Director effective January 16, 2015. Previously, since November 2013, he served as the President and Chief Operating Officer of the Company. From August 2012 to November 2013, Mr. Stone served as the Chief Executive Officer of the Company's wholly owned subsidiary, Digital Turbine USA, Inc. From March 2012 to August 2012 Mr. Stone served as a consultant to the Company. Mr. Stone was previously Senior Vice President of QUALCOMM Inc. and President of its subsidiary FLO TV Inc. from 2009 to 2011. Prior to Qualcomm, Stone was the CEO and President of the smartphone application storefront provider, Handango, (acquired by Appia Inc.) from 2007 to 2009. Mr. Stone has extensive global experience in wireless, technology, mobile content, marketing and distribution, having held executive positions at several operators such as Verizon, Vodafone, and AirTouch. Mr. Stone has a BA and MBA from Rice University.

Required Vote

Directors are elected by a "plurality" of the voting power of our common stock and preferred stock, voting together as a single class on an as-converted to common stock basis). Plurality means that the nominee with the largest number of votes is elected, up to the maximum number of directors to be chosen (in this case, six directors). Stockholders can either vote "FOR" the nominee or withhold authority to vote for the nominee. However, shares that are withheld will have no effect on the outcome of the election of directors. Broker non-votes also will not have any effect on the outcome of the election of the directors.

Board Recommendation

Our Board Recommends that Stockholder Vote "FOR" Each of the Above-Named Director Nominees.

BOARD MEETINGS AND COMMITTEES

During fiscal year 2014, our Board held 19 meetings. Our Board has three separately designated committees: our Audit Committee, our Compensation Committee, and our Nominating and Governance Committee. Each member of these standing committees has been determined to meet the standards for director independence under the rules of the SEC and is an "independent director" as defined under NASDAQ listing standards.

Each committee has the power to engage independent legal, financial or other advisors, as it may deem necessary, without consulting or obtaining the approval of our Board or any of our officers. Each then-director attended, either in person or by electronic means, 75% or more of the aggregate of (i) the total number of meetings of our Board and (ii) the total number of meetings held by all committees of our Board on which such director served during the 2014 fiscal year.

Committees of the Board of Directors

Audit Committee

The current members of our Audit Committee are Robert Deutschman (Chairman), Christopher Rogers and Paul Schaeffer. Our Audit Committee held four meetings during the 2014 fiscal year and is primarily responsible for hiring our independent auditors, approving the services performed by our independent auditors and reviewing their reports regarding our accounting practices and systems of internal accounting controls. Our Board determined that each of our Audit Committee members is "independent" pursuant to the rules of The NASDAQ Stock Market and Securities and Exchange Commission rules and that Mr. Deutschman is an "audit committee financial expert" as defined by applicable Securities and Exchange Commission rules. The Audit Committee acts pursuant to a written charter adopted by our Board, which is available on our website at http://ir.mandalaydigital.com/governance-documents.

Compensation Committee

The current members of our Compensation Committee are Christopher Rogers and Jeffrey Karish. Our Compensation Committee held 13 meetings during the 2014 fiscal year, and is primarily responsible for reviewing and approving our general compensation policies and setting both the cash and non-cash compensation levels for our executive officers. Our Compensation Committee also has the authority to administer our employee stock purchase plan and our stock incentive plan and to make option grants under our stock incentive plan. During the 2014 fiscal year, the Compensation Committee has consulted with independent organizations for background information as to general compensation levels currently being offered in our industry. All members of the Compensation Committee are non-employee directors and are "independent" pursuant to the rules of The NASDAQ Stock Market. Our executive officers do not perform any role in determining or recommending the amount or form of executive or director compensation; however, as a member of our Board, our Chief Executive Officer reviews and participates in compensation decisions of executive officers other than himself. The Compensation Committee acts pursuant to a written charter adopted by our Board, which is available on our website at http://ir.mandalaydigital.com/governance-documents.

Governance and Nominating Committee

The current members of our Governance Committee are Robert Deutschman and Jeffrey Karish. Our Governance Committee did not hold any meetings during the 2014 fiscal year. Our Governance Committee annually screens and recommends to the Board-qualified candidates for election or appointment to our Board, recommends the number of members of our Board, and evaluates and reviews the independence of existing and prospective directors. The members of the Governance Committee are non-employee directors and are "independent" directors under the rules of The NASDAQ Stock Market. The Nominating Committee acts pursuant to a written charter adopted by our Board, which is available on our website at http://ir.mandalaydigital.com/governance-documents.

CORPORATE GOVERNANCE

Board Leadership Structure

The roles of Chairman and Chief Executive Officer are held by different individuals. Peter Guber is our non-executive Chairman of the Board. We believe that having a non-executive Chairman acting in a leadership position to direct the meetings and procedures of the Board is beneficial to our Company and its employees and conveys a clear, cohesive message of Board independence and leadership to our employees and industry partners. Our Board believes that this structure provides an efficient and effective leadership model for our Company, facilitates efficient and open communication between our directors and management team, and helps to involve our other independent Board members in Board activities and decision making.

Role of Board of Directors in Risk Oversight

One of the responsibilities of our Board is to review and evaluate the process used to assess major risks facing our Company and to periodically review assessments prepared by our senior management of such risks, as well as options for their mitigation. Frequent interaction between our directors and members of senior management assist in this effort. Communications between our Board and senior management regarding long-term strategic planning and short-term operational practices include matters of material risk inherent in our business.

Our Board also plays an active role, as a whole and at the committee level, in overseeing management of our risks. Our entire Board is formally apprised at least annually of our enterprise risk management efforts. Our Board regularly reviews information regarding our credit, liquidity and operations, as well as the risks associated with each. Our Audit Committee is responsible for overseeing the management of financial and accounting risks. Our Compensation Committee is responsible for overseeing the management of risk-taking relating to executive compensation plans and arrangements. While each committee is responsible for the evaluation and management of such risks, our entire Board is regularly informed through committee reports.

Code of Business and Ethical Conduct

The Board has established a corporate Code of Conduct which qualifies as a "code of ethics" as defined by Item 406 of Regulation S-K of the Exchange Act and applies to the Company's principal executive officer, principal financial officer, principal accounting officer and all other officers, directors, and employees. Among other matters, the Code of Conduct is designed to deter wrongdoing and to promote:

- honest and ethical conduct, including the ethical handling of actual or apparent conflicts of interest between personal and professional relationships;
- full, fair, accurate, timely and understandable disclosure in our SEC reports and other public communications;
- compliance with applicable governmental laws, rules and regulations;
- prompt internal reporting of violations of the Code of Conduct to appropriate persons identified in the code; and
- accountability for adherence to the Code of Conduct.

A full text of our Code of Conduct is published on our website at www.mandalaydigital.com under the tab "Investor Relations." If we amend our Code of Conduct as it applies to the principal executive officer, principal financial officer, principal accounting officer or controller (or persons performing similar functions) or grant a waiver from any provision of the Code of Conduct to any such person, we shall disclose such amendment or waiver on our website at http://ir.mandalaydigital.com/governance-documents.

Director Independence

Our Board annually reviews and determines the independence of each director and nominee for election as a director in accordance with the independence standards set forth in the NASDAQ Marketplace rules. Based on these standards, our Board determined that each of Robert Deutschman, Peter Guber, Jeffrey Karish, Christopher Rogers and Paul Schaeffer, who collectively constitute a majority of our Board, is independent.

Related Party Transactions

Equity Financings

On April 12, 2013 the Company entered into Equity Financing Binding Term Sheets (the "Equity Agreements"), with Trinad Capital Master Fund, Ltd. ("Trinad") and the Guber Family Trust (the "Guber Trust") pursuant to which the Company agreed to issue a total of 285,714 shares of its common stock and warrants to purchase a total of 71,428 shares of its common stock (subject to adjustment), for a total purchase price of \$1,000,000 in cash for both Equity Agreements. The warrants have an exercise price of \$3.50 per share (subject to adjustment), a five year term, and may be exercised only following the first anniversary of the date of issuance. In addition, the Equity Agreements provided that (a) the Company would use its best efforts to file a registration statement covering the shares of common stock issued and the shares of common stock issuable upon exercise of the warrants within 120 calendar days after the date of the Equity Agreement and use its best efforts to cause the registration statement to become effective as soon as possible thereafter; and (b) each investor would have a right to participate in future financings by the Company (subject to standard exceptions) for a period of two years on a pro rata basis in accordance with his ownership interests in the Company (on a fully diluted basis assuming exercise of the applicable warrant).

Trinad Capital Master Fund is one of our principal stockholders and an affiliate of our former Executive Chairman, Robert Ellin. The Guber Trust is one of our principal stockholders, and Peter Guber, the Company's non-executive Chairman, serves as Trustee of the Guber Trust.

Senior Secured Convertible Notes

On June 21, 2010, we sold and issued \$2,500,000 of Senior Secured Convertible Notes due June 21, 2013 (the "New Senior Secured Notes") to certain significant stockholders, comprised of a \$1,500,000 New Senior Secured Note sold and issued to Trinad Capital Master Fund and a \$1,000,000 New Senior Secured Note sold and issued to the Guber Family Trust (the "Offering"). Trinad Capital Master Fund is one of our principal stockholders and an affiliate of our former Executive Chairman, Robert Ellin. Peter Guber, our non-executive Chairman, serves as trustee of the Guber Family Trust. The New Senior Secured Notes had a three-year term at an interest rate of 10% per annum payable in arrears semi-annually. Notwithstanding the foregoing, at any time on or prior to the 18th month following the original issue date of the New Senior Secured Notes, we could, at our option, in lieu of making any cash payment of interest, elect that the amount of any interest due and payable on any interest payment date on or prior to the 18th month following the original issue date of the New Senior Secured Notes be added to the principal due under the New Senior Secured Notes. The accrued and unpaid principal and interest due on the New Senior Secured Notes were convertible at any time at the election of the holder into shares of our common stock at a conversion price of \$0.75 per share, subject to adjustment. The New Senior Secured Notes were secured by a first priority lien on substantially all of our and our subsidiaries' assets. On June 20, 2013, the holders of the New Senior Secured Notes agreed to amend the New Senior Secured Notes to extend the June 21, 2013 Maturity Date of the notes. The Maturity Date was extended on an interim basis and then to July 9, 2013. On July 8, 2013, the Noteholders entered into an amendment to the Senior Secured Notes, dated as of July 7, 2013 that extended the July 9, 2013

On August 21, 2013, each of the New Senior Convertible Notes was converted and an aggregate of 4,497,664 shares of the Company's common stock were issued to the noteholders.

Each purchaser of a New Senior Secured Note also received a warrant ("Warrant") to purchase shares of our common stock at an exercise price of \$1.25 per share, subject to adjustment. For each \$50,000 of New Senior Secured Notes purchased, the purchaser received a Warrant to purchase 33,333 shares of common stock. Each Warrant has a five-year term. Trinad Capital Master Fund cashless exercised their warrant on February 7, 2014 for a net issuance of 692,874 shares of common stock of the Company. The Guber Family Trust still holds its warrant to purchase 666,667 of common stock of the Company.

Stockholder Nominations and Bylaw Procedures

The Governance Committee does not have a specific written policy or process regarding the nominations of directors, nor does it maintain minimum standards for director nominees. However, the Governance Committee does consider the knowledge, experience, integrity and judgment of potential candidates for nominations to the Board. The Governance Committee will consider persons recommended by stockholders for nomination for election as directors. The Governance Committee will consider and evaluate a director candidate recommended by a stockholder in the same manner as a committee-recommended nominee.

To nominate a person for election to our Board, a stockholder must set forth all information relating to the nominee that is required to be disclosed in solicitations of proxies for election of directors, or is otherwise required, in each case pursuant to Regulation 14A under the Securities Exchange Act of 1934. Such notice must also contain information about the stockholder making the nomination and the beneficial owner, if any, on behalf of whom the nomination is made, including name and address, class and number of shares owned, and representations regarding the intention to make such a nomination and to solicit proxies in support of it. We may require any proposed nominee to furnish information concerning his or her eligibility to serve as an independent director or that could be material to a reasonable stockholder's understanding of the independence of the nominee.

Deadlines to Submit Nominations

Stockholders wishing to recommend director candidates must follow the prior notice requirements as described under "2015 Stockholder Proposals," below.

Stockholder nominations must be addressed to Corporate Secretary, Mandalay Digital Group, Inc., 2811 Cahuenga Boulevard West, Los Angeles, CA 90068, or faxed to (323) 969-1946 with a confirmation copy sent by mail.

Executive Sessions of Independent Directors

The independent members of our Board meet in executive session, without any employee directors or other members of management in attendance, at least annually and additionally, as circumstances warrant.

Stockholder Communication with Board Members

We maintain contact information for stockholders, both telephone and email, on our website (http://www.mandalaydigital.com) under the heading "For Investors" where a stockholder will find our telephone number and mailing address as well as a link for providing email correspondence to Investor/Public Relations. Communications sent to Investor/Public Relations and specifically marked as a communication for our Board will be forwarded to our Board or specific members of our Board as directed in the stockholder communication. In addition, communications for our Board received via telephone or mail are forwarded to our Board by one of our employees.

Board Member Attendance at Annual Meetings

Our Board does not have a formal policy regarding attendance of directors at our annual stockholder meetings.

EXECUTIVE OFFICERS

The following table sets forth certain information regarding our executive officers:

Name	Age	Position
William G. Stone III	46	Chief Executive Officer, Current Director & Director Nominee
Andrew Schleimer	37	Executive Vice President, Chief Financial Officer
Kirstie Brown	38	Principal Accounting Officer

Biographical information regarding Mr. Stone is set forth above under "Election of Directors (Proposal 2) – Information Concerning Current Directors and Nominees."

Andrew Schleimer. Mr. Schleimer has served as the Executive Vice President and Chief Financial Officer of the Company since July 8, 2014. Mr. Schleimer was an advisor to the Company's Board of Directors since late 2011 and became a consultant to the Board in April 2012, advising and consulting on the company's mergers and acquisitions, financing activities, and investor relations. From September 2010 through July 2014, Mr. Schleimer was President of Au Courant Capital Corp., a financial advisory consulting firm specializing in mergers and acquisitions and operational support. Mr. Schleimer's services to the Company as an advisor and consultant were provided through Au Courant Capital Corp. From September 2010 through November 2012, he served as Executive Vice President, Strategic Development, of Dick Clark Productions where he was responsible for identifying new sponsorship revenues as well as creating new revenue streams – primarily via digital applications – and sourcing new business and acquisition opportunities. Prior to September 2010, he was Executive Vice President, Strategic Development & In-Park Service, of Six Flags, Inc. During his tenure, he was responsible for strategic planning and new business development, including domestic and international theme park development, non-traditional brand extensions, and all in-bound and out-bound brand licensing, as well as full profit and loss responsibility for In-Park Services revenue. Mr. Schleimer was an employee of Trinad Capital Master Fund, a large shareholder, and such relationship ended in June 2013. Mr. Schleimer has a background in investment banking with a focus on mergers and acquisitions, and joined Six Flags from UBS Investment Bank, where he served as Vice President in the bank's Mergers and Acquisitions department. At UBS, he advised on over \$150 billion of transactions in the media, entertainment, technology, telecom and consumer products sectors. Mr. Schleimer received a B.S. from Cornell University School of Hotel and Restaurant Management in 1999.

Kirstie Brown. Ms. Brown was appointed as the Company's principal accounting officer in September 2014. Since 2012, she has also served as the Executive Vice President of Global Finance & Operations of Digital Turbine USA, Inc., and the Chief Financial Officer of Digital Turbine Asia Pacific Pty Ltd., which the Company acquired in April 2013. Previously, from 2005 to 2012, Ms. Brown served as Chief Financial Officer and Secretary of Mirror Image Access (Australia) Pty Ltd ("MIA") and of the Mirror Image International Group, a group comprised of five of MIA's subsidiaries in the United States, United Kingdom and Singapore. Ms. Brown presently serves on the Board of five of the Company's subsidiaries comprising the Digital Turbine Asia Pacific Group: Digital Turbine Holdings Pty Ltd, Digital Turbine Asia Pacific Pty Ltd, Digital Turbine Technology (IP) Pty Ltd, Digital Turbine IP Pty Ltd, and Digital Turbine Singapore Pty Ltd.

EXECUTIVE COMPENSATION

Summary Compensation Table

The following table sets forth information concerning the total compensation paid during our fiscal year ended March 31, 2014, and our fiscal year ended March 31, 2013, for: (i) our Chief Executive Officer, (ii) our other most highly compensated executive officer who was serving as an executive officer as of March 31, 2014, and (iii) two additional individuals who served as the former Chief Financial Officer and former Principal Financial Officer of the Company, respectively, during the fiscal year ended March 31, 2014, but who were not serving as executive officers at the end of such year (collectively, our "named executive officers").

	Fiscal				Option		
Position	Year Ended	Salary (\$)	Bonus (\$)	Stock Awards (1) (\$)	Awards (2) (\$)	All Other (\$)	Total (\$)
Peter Adderton (3) Former Chief Innovation Officer and previous Chief Executive Officer	March 31, 2014 March 31, 2013	475,000 465,847	500,000		1,233,283	115,872 109,709	2,324,155 575,556
Bill Stone (4) Chief Executive Officer and previous President and Chief Operating Officer	March 31, 2014 March 31, 2013	350,000 213,333	195,000 58,333	1,200,000	708,055 —	30,870 18,692	1,283,925 1,471,666
Jeffrey Klausner (5) Previous Chief Financial Officer	March 31, 2014 March 31, 2013	96,955 —	_ _	_	712,915	8,086	817,956 —
Lisa Lucero (6) Corporate Controller and previous Principal Financial Officer	March 31, 2014 March 31, 2013	203,077 161,538	18,000	_	_	13,738 12,377	234,815 173,915

- (1) The amounts in the "Stock" column reflect the aggregate grant date fair value of each restricted stock award that was granted during the period ended March 31, 2014, computed in accordance with FASB ASC Topic 718. We estimated the fair value of restricted stock based on the fair value at the time of grant. The fair value for awards that are expected to vest is then amortized on a straight-line basis over the requisite service period of the award, which is generally the vesting term. The amount of expense recognized represents the expense associated with the restricted stock we expect to ultimately vest based upon an estimated rate of forfeitures; this rate of forfeitures is updated as necessary and any adjustments needed to recognize the fair value of restricted stock that actually vest or are forfeited are recorded. Vesting schedules for unvested stock grants for each officer are described below under "Narrative Disclosure to Summary Compensation Table".
- (2) The amounts in the "options" column reflect the fair value of the option award as of the grant date in accordance with FASB ASC Topic 718. The fair value of options granted during the year ended March 31, 2014 is computed using the Black-Scholes option pricing model. The valuation calculation requires the input of subjective assumptions, including the expected volatility of our common stock, interest rates, expected dividend rates and an option's expected life. Expected volatilities are based on the historical volatility of the Company's stock. The Company uses historical data to estimate option exercise and employee termination in determining forfeiture rates. The risk-free interest rate used in this model correlates to a U.S. constant rate Treasury security with a contractual life that approximates the expected term of the option award. The expected term of options granted is estimated based on the vesting term of the award.

- (3) Mr. Adderton was Chief Innovation Officer from October 2, 2014 until January 16, 2015. Prior to that, he was our Chief Executive Officer starting on December 28, 2011 following a period of service as our interim Chief Executive Officer that began on July 15, 2011. Mr. Adderton was reimbursed \$83,325 and \$79,992 for his personal assistant in the years ended March 31, 2014 and 2013, respectively. The reimbursement was paid through payment to Skycrest Ventures, LLC, and such amounts are disclosed under "All Other". Mr. Adderton received Company paid health benefits of \$32,547 and \$29,717 in the years ended March 31, 2014 and 2013, respectively.
- (4) Mr. Stone was appointed as our President and Chief Operating Officer on November 24, 2013. On September 9, 2014, the Company and Mr. Stone entered into an employment agreement pursuant to which Mr. Stone, became the Chief Executive Officer of the Company on October 2, 2014. Pursuant to his previous agreement, Mr. Stone was granted options to purchase 300,000 shares of common stock of the Company. On July 8, 2014, Mr. Stone was granted options to purchase an additional 200,000 of common stock of the Company. On August 1, 2012, Mr. Stone was appointed as the Chief Executive Officer of the Company's wholly owned subsidiary, Digital Turbine USA, Inc. and was granted 300,000 shares of common stock of the Company vesting monthly over three years. Mr. Stone received Company paid health benefits of \$30,870 and \$18,692 in the years ended March 31, 2014 and 2013, respectively.
- (5) Mr. Klausner was appointed as our Chief Financial Officer on November 24, 2013 and resigned on July 8, 2014. Pursuant to his employment agreement, Mr. Klausner was granted options to purchase 300,000 shares of common stock of the Company, 250,000 of which terminated in connection with his resignation and 50,000 of which vested pursuant to his separation agreement dated July 8, 2014. The stock options are exercisable at \$2.54 per share for a period of 180 days following the date of separation.
- (6) Ms. Lucero served as our Principal Financial Officer from December 26, 2012 until November 24, 2013. Ms. Lucero also serves as our Corporate Secretary. On January 3, 2012, Ms. Lucero was granted 100,000 shares of restricted common stock of the Company, of which 50,000 vested immediately and 50,000 vested upon approval of the Company's board of directors on November 4, 2013.

NARRATIVE DISCLOSURE TO SUMMARY COMPENSATION TABLE

Employment and Separation Agreements with Peter Adderton. On October 1, 2013, we entered into an employment agreement with Mr. Adderton to become the Company's Chief Executive Officer. Mr. Adderton's employment agreement provided for a two-year term and an annual salary of \$500,000. Mr. Adderton received a signing bonus of \$500,000 and options to purchase 500,000 shares of common stock under the Company's equity incentive plan at the closing price of the Company's common stock on October 1, 2013. The options vest as follows: 250,000 options vest on the one year anniversary of the effective date, and 250,000 options vest on the two year anniversary of the effective date. On January 15, 2015, in connection with his resignation, the Company and Mr. Adderton entered into a separation agreement that provided for, among other things, payment of an amount equal to one-half unused vacation days, accrued but unpaid salary and reimbursement of expenses, termination of repurchase rights on approximately 600,000 shares (with 227,500 shares remaining subject to repurchase rights in connection with certain limited indemnities by Mr. Adderton), cancellation of 250,000 options, and a lock-up agreement.

Employment Agreement with Andrew Schleimer. On July 8, 2014, we entered into an employment agreement with Andrew Schleimer to become the Company's Executive Vice President and Chief Financial Officer. Mr. Schleimer's employment agreement provides for a two-year term with the potential to be extended for one year at an annual salary of \$300,000 (\$350,000 in the optional third year, if Mr. Schleimer exercises an extension option if certain to-be-determined performance criteria are met, or if the Company exercises its extension option). In addition, Mr. Schleimer received options for 300,000 shares of common stock under the Company's equity incentive plan at the closing price of the Company's common stock on July 8, 2014. The options will vest over a four year term as follows: 25% on the first anniversary of his start date, then 6,250 shares on a monthly basis for the following three years. In the event of a change of control, all unvested options shall vest immediately. Mr. Schleimer will have a bonus opportunity, on an annual basis, to receive a performance bonus of up to 50% of his base salary. Mr. Schleimer's bonus opportunity is based on corporate performance

criteria, and personal performance criteria. The corporate performance criteria are, for the remainder of the current fiscal year ended March 31, 2015, based on previously budgeted Company revenue and Adjusted EBITDA targets and on five factors related to different operational and financial aspects of the Company's finance and accounting functions listed in the Agreement. A similar structure applies for the balance of the term, except that the Compensation Committee may use, instead of Adjusted EBITDA, a different measure that it determines to be the most important earnings measure used publicly by the Company (for example, EBITDA without adjustment), and will also determine the target level of revenue and of such earnings measure. The Compensation Committee will first consult with Mr. Schleimer prior to making such determinations. Mr. Schleimer will receive a relocation payment of \$37,500 if the Company requires him to relocate under certain circumstances. He will also receive expense reimbursement, including for his office use in New York City for a specified period, subject to extension by the CEO in his discretion.

Employment Agreement with Bill Stone.

Previous Employment Agreement. On November 24, 2013, we entered into an employment agreement with Bill Stone to become the Company's President and Chief Operating Officer. The Agreement had a term of two years and Mr. Stone received a salary of \$350,000 per year. Mr. Stone was eligible to receive, on an annual basis, a performance bonus of up to 100% of his base salary based on criteria consistent with those applicable to the Company's CEO, as determined mutually by the CEO, Compensation Committee and Mr. Stone. In addition, Mr. Stone received options to purchase 300,000 shares of common stock under the Company's equity incentive plan at the closing price on November 25, 2013. The options vest over a three year term as follows: 100,000 options on the first anniversary of the option grant date, then the remaining shares vest on a pro rata monthly basis for the following two years. Mr. Stone was also the President of Digital Turbine, Inc., the Company's wholly-owned subsidiary.

<u>CEO Employment Agreement</u>. On September 9, 2014, we entered into an employment agreement pursuant to which Mr. Stone became the Chief Executive Officer of the Company on October 2, 2014. The agreement has a term of two years, and Mr. Stone will receive a salary of \$500,000 per year, a one-time \$100,000 signing bonus and performance bonus opportunities as follows:

- A pro-rated bonus under his prior employment agreement for the current fiscal year through October 2, 2014 (the "Transition Date").
- For each of (1) the period from the Transition Date through the end of the current fiscal year (the "<u>Stub Period</u>"), (2) the next fiscal year and (3) the year following the next fiscal year through the end of the term of the agreement, Mr. Stone shall be eligible to receive an annual incentive bonus of up to 150% of his salary for that period subject to satisfaction of certain performance-related milestones specified in the agreement. The milestones for the Stub Period are based on the Company's achievement of Stub Year Revenue and Adjusted EBITDA targets and the milestones for the full-year 1 and full-year 2 periods are based on the Company's achievement of non-GAAP measures of consolidated Company earnings such as EBITDA or adjusted EBITDA, as further specified in the agreement

In connection with entering into the agreement, the vesting (but not the exercise price) of the 200,000 options granted to Mr. Stone on July 8, 2014 was adjusted so that 50,000 options shall vest on the one-year anniversary of the original grant date (i.e., July 8, 2015), 150,000 options shall vest on a monthly basis over the 3 years following such first anniversary, and all such unvested options granted shall vest immediately upon a change of control of the company. In addition, on the effective date of the agreement, Mr. Stone received a grant of a new stock option to purchase 50,000 shares of common stock of the Company at an exercise price equal to the closing price of the Company's common stock on September 10, 2014 under the Company's equity incentive plan, which shall vest as follows: 12,400 options shall vest on the one-year anniversary of the grant date, 37,500 options shall vest on a monthly basis over the 3 years following the first anniversary of the grant date, and all unvested options shall vest immediately upon a change of control of the Company.

Employment Agreement with Lisa Lucero. Ms. Lucero joined the Company on August 27, 2007. She receives a base salary of \$180,000 per year, payable in accordance with our payroll guidelines. She is eligible for a bonus at the discretion of the Board of Directors. During the fiscal year ended March 31, 2014, we did not grant Ms. Lucero any stock options or other equity compensation. Effective January 3, 2012, we granted Ms. Lucero 100,000 shares of restricted common stock, of which 50,000 vested immediately and 50,000 vested on November 4, 2013.

Employment and Separation Agreements with Jeffrey Klausner. On November 22, 2013, the Company entered into an employment agreement with Mr. Klausner, effective November 25, 2013, pursuant to which Mr. Klausner was employed as the Chief Financial Officer of the Company. The agreement had a two year term with an annual salary of \$275,000. In addition, Mr., Klausner had a bonus opportunity, on an annual basis, to receive a performance bonus of up to 50% of his base salary. Mr. Klausner's bonus opportunity in the first year was based on ten factors, each worth 5% of base salary, related to different operational and financial aspects of the Company's finance and accounting functions listed in the agreement. Criteria for the second year of the term were to be mutually agreed upon between the Company and Mr. Klausner at least 30 days before the end of the first year. In addition, Mr. Klausner received options for 300,000 shares of common stock under the Company's equity incentive plan at the closing price of the Company's common stock on November 25, 2013. The options were to vest over a four year term as follows: 25% on the first anniversary, then 6,250 shares on a monthly basis for the following three years. Mr. Klausner resigned from the Company on July 8, 2014. On that same date, Mr. Klausner and the Company entered into an agreement providing for his separation from the Company and certain related matters (the "Klausner Agreement"). Under the Klausner Agreement, the Company shall pay to Mr. Klausner a separation amount equal \$206,250, less applicable payroll deductions in resolution of any claims he may have. In addition, the Company agreed that, should Mr. Klausner elect to continue his group health benefits through COBRA, the Company will pay the costs of his COBRA premiums during a period of nine (9) months after the separation date. In addition, Mr. Klausner shall receive acceleration of vesting of the options granted under his employment agreement with the Company, on a pro-rata basis, as if the vesting schedule had been monthly rather than annual, advanced to July 31, 2014, (i.e., an aggregate of fifty thousand (50,000) options), exercisable at a price of \$2.54. The vested options shall remain subject to the terms of the Company's Amended and Restated 2011 Equity Incentive Plan and shall expire 180 days following the separation date.

Other than as described above, we have no plans or arrangements with respect to remuneration received or that may be received by our named executive officers to compensate such officers in the event of termination of employment (as a result of resignation, retirement, change of control) or a change of responsibilities following a change of control.

OUTSTANDING EQUITY AWARDS AT MARCH 31, 2014

The following table presents information regarding outstanding options and unvested stock awards held by our named executive officers as of March 31, 2014.

	Option Awards						Stock Awards			
Name	Number of Securities Underlying Unexercised Options (#) Exercisable	Number of Securities Underlying Unexercised Options (#) Unexercisable	Number of Securities Underlying Unexercised Unearned Options (#)	Option Exercise Price (\$)	Option expiration date	Number of shares or units of stock that have not vested (#)	Market value of shares or units of stock that have not vested (\$)	Equity incentive plan awards: Number of unearned shares, units or other rights that have not vested (#)	Equity incentive plan awards: Market or payout value of unearned shares, units or other rights that have not vested (\$)	
Peter Adderton Former Chief Innovation Officer and previous Chief Executive Officer (1)		500,000		\$ 2.65	10/1/2023		<u></u>	602,500	\$ 2,379,875	
Bill Stone Chief Executive Officer (2)	_	300,000	_	\$ 2.54	11/25/2023	191,666	\$757,081		_	
Jeffrey Klausner Previous Chief Financial Officer (3)	_	300,000	_	\$ 2.54	11/25/2023	_	_	_	_	
Lisa Lucero Corporate Controller and previous Principal Financial Officer (4)	1,951	_	_	\$ 4.05	10/9/2017	_	_	_	_	

- (1) Effective December 28, 2011, we granted Mr. Adderton 1,807,500 shares of restricted common stock, one third of which vested as of December 29, 2011, and another third vested on July 3, 2013. Mr. Adderton also received an option to purchase 500,000 shares of our common stock at a per share price of \$2.65 on October 1, 2013.
- (2) Effective August 1, 2012, we granted Mr. Stone 300,000 shares of restricted common stock of which 108,334 have vested. On August 1, 2012, Mr. Stone was granted 300,000 shares of common stock of the Company vesting monthly over three years. A portion of the grant was deferred 6 months. On November 25, 2013, Mr. Stone was granted 300,000 stock options exercisable at the exercise price of \$2.54. The options vest over a three year term as follows: 100,000 options on the first anniversary of the option grant date, then the remaining shares vest on a pro rata monthly basis for the following two years. On July 8, 2014, Mr. Stone was granted 200,000 stock options with an exercise price of \$4.11 per share. The options vest over a four year term as follows: 150,000 options on the third anniversary of the option grant date, then the remaining shares vest on a pro rata monthly basis for the following year, for a total of four years. On September 10, 2014, Mr. Stone was granted 50,000 options with an exercise price of \$5.89 per share and 12,500 of which will vest on the one-year anniversary of the grant date, and 37,500 of which will vest on a monthly basis over the three years following the first anniversary of the grant date to become fully vested on September 10, 2018.
- (3) On November 25, 2013, we granted 300,000 stock options to Mr. Klausner, and on July 8, 2014, pursuant to his separation agreement, the vesting of the options were accelerated, on a pro-rata basis, as if the vesting schedule had been monthly rather than annual, advanced to July 31, 2014, (i.e., an aggregate of fifty thousand (50,000) options). The stock options are exercisable at \$2.54 per share for a period of 180 days following his separation date.
- (4) On January 31, 2008, we granted 1,951 stock options to Ms. Lucero under the 2007 equity incentive plan. The stock options are exercisable at the exercise price of \$4.05.

DIRECTOR COMPENSATION

The following table presents information regarding compensation paid to our directors during the fiscal year ended March 31, 2014.

	Fees Earned or				
	Paid in Cash	Stock Awards	Option Awards	All Other	
Name	(\$)	(\$)(1)	(\$)	Compensation (\$)	Total (\$)
Peter Guber (2)	32,000	48,000	_	_	80,000
Robert S. Ellin (3)	_	_	_	_	_
Paul Schaeffer (4)	32,000	48,000	_	_	80,000
Peter Adderton (5)	_	_	_	_	_
Chris Rogers (6)	32,000	48,000	_	_	80,000
Jeffrey Karish (7)	32,000	48,000	_	_	80,000
Robert Deutschman (8)	32,000	48,000	_	_	80,000

- (1) The amounts in this column reflect the aggregate grant date fair value of each restricted stock award that was granted in the period ended March 31, 2014 and is computed in accordance with FASB ASC Topic 718. The fair value for awards that are expected to vest is then amortized on a straight-line basis over the requisite service period of the award, which is generally the stock vesting term. The amount of expense recognized represents the expense associated with the stock we expect to ultimately vest based upon an estimated rate of forfeitures; this rate of forfeitures is updated as necessary and any adjustments needed to recognize the fair value of stock that actually vest or are forfeited are recorded.
- (2) Mr. Guber is the non-executive Chairman of the Board of Directors. During the period ended March 31, 2014, Mr. Guber received quarterly cash payments and was granted 17,204 shares of common stock of which 8,602 remained unvested as of that date.
- (3) Mr. Ellin was Executive Chairman of the Board of Directors from December 28, 2011 until April 15, 2013, and served as a Director from April 15, 2013 until September 20, 2013. Mr. Ellin resigned from the Board of Directors of the Company on September 20, 2013.
- (4) Mr. Schaeffer is a director of the Company and a member of the Audit Committee. During the period ended March 31, 2014, Mr. Schaeffer received quarterly cash payments and was granted 17,204 shares of common stock of which 8,602 remained unvested as of that date.
- (5) Mr. Adderton was appointed as our interim Chief Executive Officer on July 15, 2011 and was appointed as our Chief Executive Officer on December 28, 2011 and as Chief Innovation Officer on October 1, 2014. His compensation for fiscal year 2014 is fully reflected in the "Summary Compensation Table" above. Mr. Adderton received no additional compensation for his service as a Director. Mr. Adderton resigned as a Director and as Chief Innovation Officer effective January 16, 2015.
- (6) Mr. Rogers is a Director of the Company and a member of the Audit Committee and the Compensation Committee. During the period ended March 31, 2014, Mr. Rogers received quarterly cash payments and was granted 17,204 shares of common stock of which 8,602 remained unvested as of that date.
- (7) Mr. Karish is a Director of the Company and a member of the Compensation Committee. During the period ended March 31, 2014, Mr. Karish received quarterly cash payments and was granted 17,204 shares of common stock of which 8,602 remained unvested as of that date.
- (8) Mr. Deutschman is a Director of the Company and the Chairman of the Audit Committee. During the period ended March 31, 2014, Mr. Deutschman received quarterly cash payments and was granted 17,204 shares of common stock of which 8,602 remained unvested as of that date.

NARRATIVE TO DIRECTOR COMPENSATION TABLE

Non-employee director compensation for a new director is granted under the Board member Equity Ownership and Retention Policy (the "Policy"). The Policy, which is administered by the independent Compensation Committee of the Board and can be amended by such committee, requires each non-management board member to acquire shares of the Company having a value equal to three times his or her annual cash retainer within five years, requires any employee director and the Chief Executive Officer to acquire shares of the

Company having a value equal to three times his or her annual salary within five years and requires the Chief Operating Officer to acquire shares of the Company having a value equal to two times his or her annual salary within five years. Unvested restricted stock or restricted stock units and unvested stock options will not be considered when determining an individual's stock ownership, and vested but unexercised stock options will be treated as equivalent to one-half a share. The Policy does not affect the vesting restrictions on any equity awards but supersedes any post-vesting lock-up that was then currently applicable to any person covered by the Policy. Failure to meet or show sustained progress toward meeting the ownership requirements of the Policy may result in reduction in future long term incentive grants and/or the requirement to retain all stock obtained through the vesting or exercise of equity awards.

The Company's compensation program for the non-employee directors is as follows: each such director receives an annual cash retainer of \$48,000 (payable in equal quarterly installments) plus an annual grant for restricted Company common stock under the Company's Amended and Restated 2011 Equity Incentive Plan having a value of \$48,000 on the grant date (with quarterly vesting).

Effective July 1, 2014, the compensation program also provides for an additional annual cash retainer of \$5,000 (payable in quarterly installments) and annual grant of restricted Company common stock under the 2011 Plan having a value of \$5,000 on the grant date (with quarterly vesting) for non-employee members of the Audit Committee of the Board (other than the Chair) and an additional annual cash retainer of \$7,500 (payable in quarterly installments) and annual grant of restricted Company common stock under the 2011 Plan having a value of \$7,500 on the grant date (with quarterly vesting) for a non-employee Chairman of the Audit Committee. Also, effective July 1, 2014, the Chairman of the Board receives an additional cash retainer of \$24,000 (payable in equal quarterly installments) plus an additional annual grant for restricted Company common stock under the 2011 Plan having a value of \$24,000.

REPORT OF AUDIT COMMITTEE

The information contained in this Audit Committee Report shall not be deemed incorporated by reference in any filing under the Securities Act of 1933 or the Securities Exchange Act of 1934, whether made before or after the date hereof and irrespective of any general incorporation language in any such filing (except to the extent that we specifically incorporate this information by reference) and shall not otherwise be deemed "soliciting material" or "filed" with the SEC or subject to Regulation 14A or 14C, or to the liabilities of Section 18 of the Securities Exchange Act of 1934 (except to the extent that we specifically request that this information be treated as soliciting material or specifically incorporate this information by reference).

The functions of our Audit Committee (references in this section to "we" and "our" mean the Audit Committee) are primarily focused on three areas:

- the adequacy of the internal controls and financial reporting process of Mandalay Digital Group, Inc. (the "Company") and the reliability of its consolidated financial statements;
- the appointment, compensation, retention and oversight of the Company's independent registered public accounting firm; and
- the Company's compliance with legal and regulatory requirements.

We operate under a written charter, which has been approved by the Board of Directors. The Company has made the Audit Committee charter available on its website at http://ir.mandalaydigital.com/governance-documents.

We meet with management periodically to consider the adequacy of the Company's internal controls and the objectivity of the Company's financial reporting. We discuss these matters with the Company's independent registered public accounting firm and with appropriate financial personnel. We periodically (at least quarterly) meet privately with both the independent registered public accounting firm and the Company's

financial personnel, each of whom has unrestricted access to us. We also appoint the independent registered public accounting firm and review their performance and independence from management. In addition, we review the Company's financing plans.

Management is responsible for the financial reporting process, including the system of internal control, and the preparation of consolidated financial statements in accordance with generally accepted accounting principles. The Company's independent registered public accounting firm is responsible for auditing those financial statements. Our responsibility is to monitor and review these processes. However, we are not professionally engaged in the practice of accounting or auditing and are not experts in the fields of accounting or auditing, including with respect to auditor independence. We rely on, without independent verification, the information provided to us and on the representations made by management and the independent registered public accounting firm.

In this context, we held four meetings during fiscal year 2014. The meetings were designed, among other things, to facilitate and encourage communication among us, management, the internal accountants and the Company's independent registered public accounting firm for fiscal year 2014, SingerLewak LLP ("SingerLewak"). We discussed with SingerLewak the overall scope and plans for their audit. We also met with SingerLewak, with and without management present, to discuss the results of their audit and quarterly reviews and the Company's internal controls. We reviewed and discussed the audited consolidated financial statements for the fiscal year ended March 31, 2014 with management and with SingerLewak.

We also discussed with SingerLewak matters required to be discussed with audit committees under generally accepted auditing standards, including, among other things, matters related to the conduct of the audit of the Company's consolidated financial statements and the matters required to be discussed by Statement on Auditing Standards No. 16, *Communications with Audit Committees (AS 16)*, as adopted by the Public Company Accounting Oversight Board.

We have received the written disclosures and the letter from SingerLewak required by the applicable requirements of the Public Company Accounting Oversight Board regarding the independent accountants' communications with us concerning independence, and we discussed with SingerLewak their independence from the Company. When considering SingerLewak's independence, we considered whether their provision of services to us beyond those rendered in connection with their audit and review of the Company's consolidated financial statements was compatible with maintaining their independence. We also reviewed, among other things, the amount of fees paid to SingerLewak for audit and non-audit services (primarily tax services).

Based on our review and these meetings, discussions and reports, and subject to the limitations on our role and responsibilities referred to above and in the Audit Committee charter, we recommended to the Board of Directors that the Company's audited consolidated financial statements for the fiscal year ended March 31, 2014 be included in the Company's annual report on Form 10-K for filing with the Securities and Exchange Commission.

Members of the Audit Committee

Robert Deutschman (Chairman)

Christopher Rogers

Paul Schaeffer

PROPOSAL 3

ADVISORY (NON-BINDING) VOTE ON EXECUTIVE COMPENSATION

We are required to permit a separate non-binding stockholder vote to approve the compensation of the Company's named executive officers, as disclosed pursuant to the compensation disclosure rules of the Securities and Exchange Commission (which disclosure includes the compensation tables and narrative discussion). This vote is not intended to address any specific item of compensation or the compensation of any particular officer, but rather to provide stockholders with an opportunity to make an advisory vote with respect to the overall compensation of our named executive officers and our compensation practices.

This proposal, commonly known as a "Say-on-pay," permits stockholders to endorse or not endorse our executive compensation through the following resolution:

"RESOLVED, that the stockholders approve, on an advisory basis, the compensation paid to the Company's named executive officers, as disclosed pursuant to the compensation disclosure rules of the Securities and Exchange Commission (which disclosure includes the compensation tables and narrative discussion)."

Because the stockholders' vote is advisory, it will not be binding on the Board. However, the Board's Compensation Committee will take into account the outcome of the vote when considering future executive compensation arrangements.

Board Recommendation

Our Board Recommends a Vote "FOR" this Proposal 3.

PROPOSAL 4

ADVISORY (NON-BINDING) VOTE ON THE FREQUENCY OF THE ADVISORY VOTE ON EXECUTIVE COMPENSATION

At the annual meeting, the stockholders will vote on a non-binding, advisory proposal regarding the frequency of future advisory stockholder votes on executive compensation as discussed in Proposal 3 in this proxy statement. Stockholders will have the opportunity to cast an advisory vote on whether the stockholder vote on executive compensation should occur every one, two or three years. Stockholders may also abstain from voting on the matter.

Because your vote is advisory, it will not be binding upon our Company or our Board. However, our Board will take into account the outcome of the vote when considering the frequency of the advisory stockholder vote on executive compensation. Your advisory vote will not be construed (i) as overruling a decision by our Company or our Board, (ii) to create or imply any change to the fiduciary duties of our Company or our Board, (iii) to create or imply any additional fiduciary duties for our Company or our Board, or (iv) to restrict or limit the ability of stockholders to make proposals for inclusion in proxy materials related to executive compensation.

Our Board believes that a non-binding stockholder vote on executive compensation should occur every year to provide the stockholders with an opportunity to annually express their views on our executive compensation. The results of such vote would then be considered by our Compensation Committee in connection with the subsequent review of executive compensation. Accordingly, the Board recommends voting for an advisory stockholder vote on executive compensation to be held annually, and therefore recommends voting for the "one year" alternative.

We emphasize, however, that you are not voting to approve or disapprove our Board's recommendation. Instead, your proxy card provides you with four options regarding this non-binding, advisory proposal. You may cast an advisory vote for the stockholder vote on executive compensation to occur every year, every two years, every three years, or you may abstain from voting on the matter.

Board Recommendation

Our Board Recommends a Vote under this Proposal 4 for an Advisory Vote Every ONE Year.

PROPOSAL 5

RATIFICATION OF APPOINTMENT OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM FOR FISCAL YEAR 2015

Our audit committee has appointed SingerLewak LLP to audit our accounts for the fiscal year ending March 31, 2015. Such firm, which has served as our independent registered public accounting firm since April 2009, has reported to us that none of its members has any direct financial interest or material indirect financial interest in our Company.

A proposal will be presented at the Annual Meeting to ratify the audit committee's appointment of SingerLewak as our independent registered public accounting firm. Although stockholder ratification of the audit committee's action in this respect is not required, our Board considers it desirable for stockholders to pass upon such appointment.

A representative of SingerLewak is expected to attend the annual meeting and will be afforded the opportunity to make a statement and/or respond to appropriate questions from stockholders.

Fees for professional services provided by SingerLewak for the fiscal years ended March 31, 2014 and 2013 were as follows:

	ear Ended rch 31, 2014	 ear Ended rch 31, 2013
Audit fees	\$ 408,504	\$ 292,556
Audit related fees		
Tax fees		
All other fees	132,072	35,886
Total	\$ 540,576	\$ 328,442

- 1. Audit services include audit work performed in the preparation of financial statements, as well as work that generally only the independent auditor can reasonably be expected to provide, including comfort letters, statutory audits, and attest services and consultation regarding financial accounting and/or reporting standards.
- 2. Audit-Related services are for assurance and related services that are traditionally performed by the independent auditor, including due diligence related to mergers and acquisitions, employee benefit plan audits, and special procedures required to meet certain regulatory requirements.
- 3. *Tax* services include all services performed by the independent auditor's tax personnel except those services specifically related to the audit of the financial statements, and includes fees in the areas of tax compliance, tax planning, and tax advice.
 - 4. Other Fees are those associated with services not captured in the other categories.

Prior to engagement, the Audit Committee pre-approves these services by category of service. The fees are budgeted and the Audit Committee requires the independent auditor and management to report actual fees versus the budget periodically throughout the year by category of service. During the year, circumstances may arise when it may become necessary to engage the independent auditor for additional services not contemplated in the original pre-approval. In those instances, the Audit Committee requires specific pre-approval before engaging the independent auditor.

The Audit Committee may delegate pre-approval authority to one or more of its members. The member to whom such authority is delegated must report, for informational purposes only, any pre-approval decisions to the Audit Committee at its next scheduled meeting.

Board Recommendation

Our Board Unanimously Recommends a Vote "FOR" Ratification of the Appointment of SingerLewak as our Independent Registered Public Accounting Firm for the Fiscal Year Ending March 31, 2015.

OTHER MATTERS

Our Board of Directors knows of no other matters to be brought before the annual meeting. However, if other matters should come before the meeting, it is the intention of each person named in the proxy to vote in accordance with his judgment on such matters.

HOUSEHOLDING OF PROXY MATERIALS

"Householding" is a program, approved by the SEC, which allows companies and intermediaries (e.g., brokers) to satisfy the delivery requirements for proxy statements and annual reports by delivering only one package of stockholder proxy materials to any household at which two or more stockholders reside. If you and other residents at your mailing address own shares of our common stock in street name, your broker or bank may have notified you that your household will receive only one copy of our proxy materials. Once you have received notice from your broker that they will be "householding" materials to your address, "householding" will continue until you are notified otherwise or until you revoke your consent. If, at any time, you no longer wish to participate in "householding" and would prefer to receive a separate proxy statement, or if you are receiving multiple copies of the proxy statement and wish to receive only one, please notify your broker if your shares are held in a brokerage account. If you hold shares of our common stock in your own name as a holder of record, "householding" will not apply to your shares.

COMPLIANCE WITH SECTION 16(a) OF SECURITIES EXCHANGE ACT

Section 16(a) of the Exchange Act requires our officers, directors, and persons owning more than ten percent of a registered class of our equity securities ("ten percent stockholders") to file reports of ownership and changes of ownership with the SEC. To the best of our knowledge, based solely on review of the copies of such reports and amendments thereto furnished to us, we believe that during the fiscal year ended March 31, 2014, all Section 16(a) filing requirements applicable to our officers, directors, and ten percent stockholders were met, except that Mr. Schaeffer filed one late report reporting one transaction, Mr. Guber filed one late report reporting four transactions, Mr. Robert Ellin filed one late report reporting four transactions, Mr. Karish filed one late report reporting one transaction, Mr. Deutschman filed one late report reporting one transaction, in July 2014, Mr. Schleimer filed two late reports reporting one transaction.

STOCKHOLDER PROPOSALS

Stockholders are entitled to submit proposals on matters appropriate for stockholder action consistent with regulations of the SEC. In order for stockholder proposals for the fiscal year 2016 annual meeting to be eligible for inclusion in our proxy statement, our Secretary must receive them at our principal offices not later than [I], 2015. Such proposals should be submitted, in writing, to Mandalay Digital Group, Inc., Attn: Corporate Secretary, 2811 Cahuenga Boulevard West, Los Angeles, CA 90068.

Stockholders wishing to submit proposals for the fiscal year 2016 annual meeting of stockholders outside of Rule 14a-8 may do so. Under Rule 14a-4 promulgated under the Exchange Act, if a proponent of a proposal that is not intended to be included in the proxy statement fails to notify us of such proposal at least 45 days prior to the anniversary of the mailing date of the preceding year's proxy statement or a reasonable time before we send our proxy materials for such meeting if the date of the meeting has changed by more than 30 days from the prior year, then we will be allowed to use our discretionary voting authority under proxies solicited by us when the proposal is raised at such annual meeting of stockholders, without any discussion of the matter in the proxy statement. We were not notified of any stockholder proposals to be addressed at our annual meeting, and will therefore be allowed to use our discretionary voting authority if any stockholder proposals are raised at the annual meeting.

ACCOMPANYING INFORMATION

Accompanying this proxy statement is a copy of our annual report to stockholders on Form 10-K and Form 10-K/A, for our fiscal year ended March 31, 2014. Such annual report includes our audited consolidated financial statements for the two fiscal years ended March 31, 2014 and 2013. No part of such annual report shall be regarded as proxy-soliciting material or as a communication by means of which any solicitation is being or is to be made.

EXPERTS

Our consolidated financial statements as of March 31, 2013 and 2014, and for each of the years in the two-year period ended March 31, 2014 have been incorporated by reference herein in reliance upon the reports of SingerLewak LLP, independent registered public accounting firm, incorporated by reference herein, and upon the authority of said firm as experts in accounting and auditing.

The audited financial statements of Appia, Inc. as of December 31, 2012 and 2013, and for each of the years in the two-year period ended December 31, 2013, included in this proxy statement/prospectus have been so included in reliance upon the report of Grant Thornton LLP, independent certified public accountants, upon the authority of said firm as experts in accounting and auditing.

LEGAL MATTERS

The legality under Delaware law of the shares of our common stock offered by this proxy statement/prospectus will be passed upon for us by Manatt, Phelps & Phillips, LLP, Los Angeles, California. Manatt, Phelps & Phillips, LLP owns 402,144 shares of common stock and warrants to purchase an additional 23,214 shares of common stock of Mandalay Digital. Latham & Watkins, LLP, Los Angeles, California will provide the tax opinion to Mandalay Digital referred to under "Material United States Federal Income Tax Considerations," and Goodwin Procter, LLP, Boston, Massachusetts, will provide the tax opinion to Appia referred to under "Material United States Federal Income Tax Considerations."

WHERE YOU CAN FIND MORE INFORMATION

We file annual, quarterly and special reports, proxy statements and other information with the SEC. You may read and copy any document we file with the SEC at the SEC's Public Reference Room at 100 F Street, N.E., Washington D.C. 20549. Please call the SEC at 1-800-SEC-0330 for further information on the operation of the Public Reference Room. The SEC also maintains a web site that contains reports, proxy and information statements, and other information regarding registrants that file electronically with the SEC at http://www.sec.gov. In addition, we maintain a web site that contains information about us at http://www.mandalaydigital.com. The information contained on or that may be obtained from our website is not, and shall not be deemed to be, a part of this prospectus.

We have filed with the SEC a registration statement on Form S-4, of which this prospectus is a part, including exhibits, schedules and amendments filed with, or incorporated by reference in, this registration statement, under the Securities Act, with respect to the securities registered thereby. This prospectus does not contain all of the information set forth in the registration statement and exhibits and schedules to the registration statement. For further information with respect to our company and the securities registered thereby, reference is made to the registration statement, including the exhibits to the registration statement. Statements contained in this prospectus as to the contents of any contract or other document referred to in, or incorporated by reference in, this prospectus are not necessarily complete and, where that contract is an exhibit to the registration statement, each statement is qualified in all respects by the exhibit to which the reference relates. Copies of the registration statement, including the exhibits and schedules to the registration statement, may be examined and copied (upon payment of applicable fees) at the SEC's Public Reference Room at 100 F Street, N.E., Washington D.C. 20549. Please call the SEC at 1-800-SEC-0330 for further information on the operation of the Public Reference Room. The registration statement is also available to you on the SEC's web site, http://www.sec.gov.

INCORPORATION OF CERTAIN INFORMATION BY REFERENCE

This document is part of a registration statement we have filed with the SEC. You should rely only on the information incorporated by reference or provided in this document. No one else is authorized to provide you with different information. You should not rely on any other representations. We are not making an offer of these securities in any state where the offer is not permitted. Our affairs may change after this prospectus is distributed. You should not assume that the information in or incorporated by reference into this prospectus is accurate as of any date other than the date on the front of those documents. You should read all information supplementing or incorporated by reference into this prospectus. The SEC allows us to incorporate by reference the information we file with the SEC, which means that we can disclose important information to you by referring you to those documents. The information that we incorporate by reference is considered to be part of this document.

We incorporate by reference into this proxy statement/prospectus the following documents (other than any portions of such documents that are not deemed "filed" under the Exchange Act in accordance with the Exchange Act and applicable SEC rules):

- Our Annual Report on Form 10-K for the year ended March 31, 2014, filed with the SEC on June 30, 2014, as amended by Amendment No. 1 on Form 10-K/A filed with the SEC on July 29, 2014 and as amended by Amendment No. 2 on Form 10-K/A filed with the SEC on January 8, 2015;
- Our Quarterly Reports on Form 10-Q for the quarters ended June 30, 2014 and September 30, 2014, filed with the SEC on August 14, 2014 and November 14, 2014, respectively;
- Our Current Reports on Form 8-K and Form 8-K/A, as applicable, filed with the SEC on April 15, 2014, June 25, 2014, July 9, 2014, September 15, 2014, October 9, 2014, November 13, 2014, November 18, 2014 and January 16, 2015 (but excluding the portions of such reports expressly noted as being furnished and not filed);
- The description of our common stock contained in our registration statement on Form 8-A filed with the SEC on June 6, 2013; and
- All documents filed by us under Section 13(a), 13(c), 14, or 15(d) of the Exchange Act after the date of this proxy statement/prospectus and before the termination of this offering.

Certain Current Reports on Form 8-K dated both prior to and after the date of this proxy statement/prospectus are or will be furnished to the SEC and shall not be deemed "filed" with the SEC and will not be incorporated by reference into this prospectus. However, all other reports and documents filed by us after the date of this proxy statement/prospectus under Sections 13(a), 13(c), 14 and 15(d) of the Exchange Act prior to the termination of the offering of the securities covered by this proxy statement/prospectus will also be deemed incorporated by reference in this proxy statement/prospectus and considered to be part of this proxy statement/prospectus from the date those documents are filed. Information in such future filings updates and supplements the information provided in this proxy statement/prospectus. Any statements in any such future filings will automatically be deemed to modify and supersede any information in any document we previously filed with the SEC that is incorporated or deemed to be incorporated herein by reference to the extent that statements in the later filed document modify or replace such earlier statements. If you make a request, orally or in writing, for any information that has been incorporated by reference into this proxy statement/prospectus but not delivered with this proxy statement/prospectus, we will provide you, without charge, a copy of any or all of that information. Requests for this information should be submitted in writing to our Secretary, at our principal executive offices at Mandalay Digital Group, Inc., 2811 Cahuenga Blvd West, Los Angeles, California 90068 or by telephone at (323) 472-5461.

INDEX TO FINANCIALS

Appia, Inc.

For the years ended December 2013 and 2012	F-1
Report of Independent Certified Public Accountants	F-3
Consolidated Balance Sheets as of December 31, 2013 and 2012	F-5
Consolidated Statements of Operations and Comprehensive Loss for the Years Ended December 31, 2013 and 2012	F-6
Consolidated Statements of Changes in Stockholders' Deficit for the Years Ended December 31, 2013 and 2012	F-7
Consolidated Statements of Cash Flows for the Years Ended December 31, 2013 and 2012	F-8
Notes to Consolidated Financial Statements	F-9
For the nine months ended September 30, 2014 and 2013	F-26
Balance Sheets as of September 30, 2014 and 2013	F-28
Statements of Operations for the Nine Months Ended September 30, 2014 and 2013	F-29
Statements of Stockholders' Deficit for the Nine Months Ended September 30, 2014	F-30
Statements of Cash Flows for the Nine Months Ended September 30, 2014 and 2013	F-31
Notes to Financial Statements	F-32

Financial Statements and Report of Independent Certified Public Accountants

Appia, Inc.

For the years ended December 2013 and 2012

Appia, Inc.

Table of contents

Report of Independent Certified Public Accountants	F-3 - F-4
Consolidated financial statements:	
Balance sheets	F-5
Statements of operations and comprehensive loss	F-6
Statements of changes in stockholders' deficit	F-7
Statements of cash flows	F-8
Notes to financial statements	F-9 - F-25



REPORT OF INDEPENDENT CERTIFIED PUBLIC ACCOUNTANTS

Audit • Tax • Advisory

Grant Thornton LLP

4140 ParkLake Avenue, Suite 130
Raleigh, NC 27612-3723

T 919.881.2700 F 919.881.2795 www.GrantThornton.com

Board of Directors Appia, Inc.:

We have audited the accompanying consolidated financial statements of **Appia, Inc.** (a Delaware company), and Subsidiary, which comprise the consolidated balance sheets as of December 31, 2013 and 2012, and the related consolidated statements of operations and comprehensive loss, changes in stockholders' deficit, and cash flows for the years then ended, and the related notes to the financial statements.

Management's responsibility for the financial statements

Management is responsible for the preparation and fair presentation of these consolidated financial statements in accordance with accounting principles generally accepted in the United States of America; this includes the design, implementation, and maintenance of internal control relevant to the preparation and fair presentation of consolidated financial statements that are free from material misstatement, whether due to fraud or error.

Auditor's responsibility

Our responsibility is to express an opinion on these consolidated financial statements based on our audits. We conducted our audits in accordance with auditing standards generally accepted in the United States of America. Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the consolidated financial statements are free from material misstatement.

An audit involves performing procedures to obtain audit evidence about the amounts and disclosures in the consolidated financial statements. The procedures selected depend on the auditor's judgment, including the assessment of the risks of material misstatement of the consolidated financial statements, whether due to fraud or error. In making those risk assessments, the auditor considers internal control relevant to the entity's preparation and fair presentation of the consolidated financial statements in order to design audit procedures that are appropriate in the circumstances, but not for the purpose of expressing an opinion on the effectiveness of the entity's internal control. Accordingly, we express no such opinion. An audit also includes evaluating the appropriateness of accounting policies used and the reasonableness of significant accounting estimates made by management, as well as evaluating the overall presentation of the consolidated financial statements.

Grant Thornton LLP U.S. member firm of Grant Thornton International Ltd



We believe that the audit evidence we have obtained is sufficient and appropriate to provide a basis for our audit opinion.

Opinion

In our opinion, the consolidated financial statements referred to above present fairly, in all material respects, the financial position of Appia, Inc. and subsidiary as of December 31, 2013 and 2012, and the results of its operations and its cash flows for the years then ended in accordance with accounting principles generally accepted in the United States of America.

/s/ Grant Thornton LLP

Raleigh, North Carolina June 26, 2014

Grant Thornton LLPU.S. member firm of Grant Thornton International Ltd

Appia, Inc.

Consolidated balance sheets

(in thousands, except share data and per share data)

December 31	2013	2012
	\$	\$
Assets		
Current assets:		
Cash and cash equivalents	7,791	3,091
Accounts receivable, net	5,718	4,073
Prepaid expenses and other current assets	565	300
Total current assets	14,074	7,464
Property and equipment, net	1,432	392
Other assets	19	<u> </u>
Total assets	15,525	7,870
Liabilities, redeemable convertible preferred stock and stockholders' deficit		
Current liabilities:		
Accounts payable	2,732	2,491
Accrued expenses and other current liabilities	2,919	2,754
Short-term capital lease	-	55
Short-term debt	600	501
Deferred revenue	76	<u>35</u> 9
Total current liabilities	6,327	6,160
Warrant liability	1,109	-
Long-term debt	7,816	910
Total liabilities	15,252	7,070
Commitments (Note 9)		
Redeemable convertible preferred stock, Series A, 3,275,000 shares authorized; 3,275,000 issued and outstanding as		
of December 31, 2013 and 2012, respectively	4,737	4,469
Redeemable convertible preferred stock, Series B, 2,569,169 shares authorized; 2,569,169 issued and outstanding as		
of December 31, 2013 and 2012, respectively	19,207	17,986
Redeemable convertible preferred stock, Series C, 2,612,818 shares authorized; 1,348,814 issued and outstanding at		
December 31, 2013 and 2012, respectively	12,184	11,367
Redeemable convertible preferred stock, Series D, 89,127 shares authorized; 89,127 issued and outstanding at	1 001	1 100
December 31, 2013 and 2012, respectively	1,601	1,468
Total redeemable convertible preferred stock	37,729	35,290
Stockholders' deficit:		
Series FF preferred stock, par value \$0.001; 2,991,500 shares authorized; 2,991,500 shares issued and outstanding as of December 31, 2013 and 2012	3	3
Common stock, par value \$0.001; 17,625,600 shares authorized; 3,909,932 and 3,584,808 shares issued and		
outstanding as of December 31, 2013 and 2012, respectively	3	3
Accumulated other comprehensive income	-	59
Accumulated deficit	(37,462)	(34,555)
Total stockholders' deficit	(37,456)	(34,490)
Total liabilities, redeemable convertible preferred stock and stockholders' deficit	15,525	7,870

The accompanying notes are an integral part of these consolidated financial statements.

Appia, Inc.

Consolidated statements of operations and comprehensive loss

(in thousands)

For the years ended December 31	2013	2012
	\$	\$
Revenues	46,860	10,517
Cost of revenues	33,863	7,086
Gross profit	12,997	3,431
Operating expenses:		
Sales and marketing	3,334	2,245
Technology	2,797	2,531
General and administrative	6,647	3,669
Total operating expenses	12,778	<u>8,44</u> 5
Income (loss) from operations	219	(5,014)
Other (expense):		
Interest expense	(745)	(65)
Loss before income taxes	(526)	(5,079)
Income tax (benefit)	-	
Net loss	(526)	(5,079)
Other comprehensive loss:		
Foreign currency translation adjustments	(59)	(30)
Other comprehensive loss	(59)	(30)
Comprehensive loss	(585)	(5,109)

The accompanying notes are an integral part of these consolidated financial statements.

Appia, Inc.

Consolidated statements of changes in stockholders' deficit

(in thousands, except share data)

<u>-</u>	Series F Preferred		Common stock		Additional		Accumulated Other	Total
	Shares	Value	Shares	Value	Paid-in Capital	Accumulated Deficit	Comprehensive Income	Stockholders' Deficit
		\$		\$	\$	\$	\$	\$
Balance as of December 31,								
2011	2,991,500	3	3,618,277	3	-	(27,154)	89	(27,059)
Issuance of common stock	-	-	5,906	-	1	-	-	1
Forfeiture of restricted stock								
shares	-	-	(39,375)	-	-	-	-	-
Stock-based compensation					00			00
expense	-	-	-	-	28	-	-	28
Accretion of Series A preferred					(29)	(240)		(070)
stock Accretion of Series B preferred	-	-	-	-	(29)	(249)	-	(278)
stock	_	_	_	_	_	(1,221)	_	(1,221)
Accretion of Series C preferred						(1,221)		(1,221)
stock	_	_	_	_	_	(817)	_	(817)
Accretion of Series D preferred						(0.7)		(011)
stock	-	-	-	-	-	(35)	-	(35)
Foreign currency translation						,		,
adjustment	-	-	-	-	-	-	(30)	(30)
Net loss for the period	-	-	-	-	-	(5,079)	-	(5,079)
Balance as of December 31,								
2012	2,991,500	3	3,584,808	3	-	(34,555)	59	(34,490)
Issuance of common stock	-	-	75,749	-	20	-	-	20
Issuance of restricted stock	-	-	280,000	-	-	-	-	-
Forfeiture of restricted stock								
shares	-	-	(30,625)	-	-	-	-	-
Stock-based compensation					40			40
expense	-	-	-	-	40	-	-	40
Accretion of Series A preferred stock					(60)	(209)		(260)
Accretion of Series B preferred	-	-	-	-	(60)	(209)	-	(269)
stock	_	_	_	_	_	(1,222)	_	(1,222)
Accretion of Series C preferred						(1,222)		(1,222)
stock	_	_	_	-	_	(817)	_	(817)
Accretion of Series D preferred						(0.1.)		(0)
stock	-	-	-	-	-	(133)	-	(133)
Foreign currency translation						,		,
adjustment	-	-	-	-	-	-	(59)	(59)
Net loss for the period	-	-	-	-	-	(526)	-	(526)
Balance as of December 31,								
2013	2,991,500	3	3,909,932	3		(37,462)	<u>-</u>	(37,456)

The accompanying notes are an integral part of these consolidated financial statements.

Appia, Inc.

Consolidated statements of cash flows

(in thousands

\$	 \$
(526)	(5,079)
366	206
11	17
	28
85	
(, ,	(3,728)
()	(154)
	1,433
(283)	248
192	<u>64</u> 4
(1,799)	(6,385)
(1,162)	
(244)	(137)
(1,406)	(137)
-	1,433
8,200	1,300
20	1
(55)	(105)
(261)	(667)
7,904	1,962
1	(2)
4,700	(4,562)
3,091	7,653
7,791	3,091
·	
752	68
. 0_	30
2.441	2,351
_,	50
	366 11 40 89 85 (1,733) (281) 241 (283) 192 (1,799) (1,162) (244) (1,406) 8,200 20 (55) (261) 7,904 1 4,700 3,091 7,791 752 2,441

The accompanying notes are an integral part of these financial statements.

Appia, Inc. (in thousands except share data and per share data)

Notes to consolidated financial statements

1 Organization and Basis of Presentation

On April 3, 2008, Pinmot, Inc. was incorporated for the purpose of acquiring a portfolio of mobile technology assets. On June 3, 2008, Pinmot, Inc. acquired the smartphone assets of Motricity, Inc. and the outstanding common stock of Motricity Deutschland GmbH. Pinmot, Inc. was subsequently renamed PocketGear, Inc. and its German subsidiary was renamed PocketGear Deutschland GmbH. PocketGear, Inc, subsequently renamed Appia, Inc. (the Company) as of February 2011. The Company is a mobile advertising network focused on driving installs of mobile applications (iOS/Android) on behalf of advertisers who want to grow their mobile audience. The Company provides mobile advertising performance solutions for both advertisers and publishers. Its technology is focused on driving user acquisition and monetization through ad walls and other ad formats.

2 Summary of Significant Accounting Policies Principles of Consolidation

The consolidated financial statements include the accounts of Appia, Inc. and its wholly-owned subsidiary, PocketGear Deutschland GmbH. All significant intercompany balances have been eliminated.

Use of Estimates

The preparation of consolidated financial statements in conformity with accounting principles generally accepted in the United States of America requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities at the date of the consolidated financial statements and the reported amounts of revenues and expenses during the reporting period. Actual results could differ from those estimates.

Revenue Recognition Advertising Revenue Recognition

The Company recognizes revenue based on the activity of mobile users viewing advertisements and downloading applications via developer applications and mobile websites. Revenues are recognized when the related advertising services are delivered based on the specific terms of the advertising contract/insertion order, which are primarily based on the number of applications installed or downloaded. The Company recognizes revenue based on these terms, as the services have been provided, the fees the Company charges are fixed or determinable, persuasive evidence of an arrangement exists, and collectability is reasonably assured.

In the normal course of business, the Company acts as an intermediary in executing transactions with third parties. The determination of whether revenue should be reported on a gross or net basis is based on an assessment of whether the Company is acting as the principal or an agent in the transaction. In determining whether the Company acts as the principal or an

Appia, Inc. (in thousands except share data and per share data)

agent, the Company follows the accounting guidance for principal-agent considerations. While none of the factors identified in this guidance is individually considered presumptive or determinative, because the Company is the primary obligor and is responsible for (i) identifying and contracting with third-party advertisers, (ii) performing all billing and collection activities including retaining credit risk, and (iii) discretion in selecting suppliers, the Company acts as the principal in these arrangements and therefore reports revenue earned and costs incurred on a gross basis.

Mobile Storefront Revenues

The Company's revenues are derived from sales of applications developed by third parties for use on handheld devices using the Company's e-commerce services, features and technologies to operate the sites.

The Company recognizes revenue from sales of third-party content related products in accordance with the Financial Accounting Standards Board (FASB) Accounting Standards Codification (ASC) 605 – "Revenue Recognition." Product sales, net of promotional discounts, rebates, and return allowances, are recorded when the products are shipped and title passes to customers. Retail items are sold to customers pursuant to a sales contract that provides for transfer of both title and risk of loss upon delivery to the customer. Return allowances, which reduce product revenue by an estimate of expected product returns, are estimated using historical experience. The Company periodically provides incentive offers to customers to encourage purchases. Current discount offers, when accepted by customers, are treated as a reduction to the sales price of the related transaction.

The Company utilizes its own proprietary technology platform to accept and verify customer orders, respond to customer support needs and to execute fulfillment on behalf of its own and third-party products. These applications also manage the process of accepting, authorizing, and charging customer credit cards. The Company uses historical results as a basis for providing an allowance for chargebacks and returns.

The Company applies ASC 605 in determining whether it is appropriate to record the gross amount of product sales and related costs or the net amount earned as commissions. Regarding sales of third-party content related products through its own and affiliate websites, the Company believes it is the primary obligor in sales transactions, has latitude in establishing prices, and holds responsibility for processing of cash receipts related to the transactions. As such, the Company records and presents revenue from these transactions on a gross basis.

The Company also enters into arrangements where it builds customized storefronts, supplying tools to ingest content and merchandising as well as providing data hosting and monitoring services. Revenue is generated from sales of applications from the storefronts and split via revenue share basis. In addition, there are instances where customers also pay one-time upfront fees and/or ongoing periodic fees. For the upfront/periodic fees the Company follows a services model. The upfront fees are deferred until the storefront is launched and operational and recognized taken ratably over the estimated life of the customer relationship.

Appia, Inc. (in thousands except share data and per share data)

Cost of Revenue

Cost of revenue consists primarily of amounts due to publishers for the advertising inventory utilized in running mobile advertisements. These amounts are typically either a percentage of the advertising revenue earned by the Company based on mobile advertisements that are run on each publishers inventory or a fixed fee for the ad space. The Company recognizes the cost of revenue as the associated revenue is recognized, on a publisher by publisher basis during the period the advertisements run on the publisher's properties. Costs owed to publisher but not yet paid are recorded as accrued cost of revenue.

Technology Costs

Technology expenses consist primarily of wages and benefits for software application development personnel. The Company has focused its research and development efforts on both improving ease of use and functionality of its existing products as well as developing new offerings. The Company expenses research and development costs that do not meet the capitalization criteria for internal-use software. Research and development expense, which includes the amortization of capitalized internally developed software costs, was \$1,615 and \$1,990, respectively for the years ended December 31, 2013 and 2012.

Advertising Costs

The costs of advertising are expensed as incurred or the first time the advertising takes place in accordance with ASC 720 – "Other Expenses." For 2013 and 2012, the Company incurred advertising costs of \$229 and \$39, respectively.

Sales and Marketing Costs

Sales and Marketing expense primarily consists of salaries and payroll related costs for Sales and Marketing employees. Additional expenses include costs related to lead generation, communication and promotion.

Cash and Cash Equivalents

The Company considers all highly liquid investments with maturity of three months or less at the date of purchase to be cash equivalents. Cash equivalents are stated at cost and consist primarily of money market accounts. Cash and cash equivalents deposited in banks were in excess of federally insured limits as of December 31, 2013 and December 31, 2012.

Accounts Receivable and Allowance for Doubtful Accounts

Accounts receivable are presented at their face amount, less an allowance for doubtful accounts, on the Company's accompanying consolidated balance sheet. Accounts receivable consist of amounts currently due from customers. The Company generally does not require collateral on sales. The Company periodically evaluates the collectability of its accounts receivable based on a combination of factors. In circumstances where the Company is aware of a specific customer's inability to meet its financial obligations, a specific reserve for bad debts against amounts due is recorded to reduce the related accounts receivable to an amount that the Company reasonably believes is collectible. As of December 31, 2013 and 2012 the Company had reserved \$546 and \$252, respectively, for allowance for doubtful accounts.

Appia, Inc. (in thousands except share data and per share data)

Property and Equipment

Property and equipment is primarily comprised of office furniture and computer equipment, which are recorded at cost and depreciated using the straight-line method over their estimated useful lives ranging from three to five years as follows:

	Useful Life
	(years)
Computer equipment - Servers	5
Computer equipment - Office	3
Software	5

Included in property and equipment is the cost of internal-use software and website development, including software used to upgrade and enhance the Company's websites. The Company follows the provisions of ASC 350 – "Intangibles – Goodwill and Other." All costs related to the development of internal-use software other than those incurred during the application development stage are expensed. Costs incurred during the application development stage are capitalized and amortized over the estimated useful life of the software (generally three years).

Upon retirement or sale, the cost of assets disposed of and the related accumulated depreciation and amortization are removed from the accounts and any resulting gain or loss is credited or charged to income. Repairs and maintenance costs are expensed as incurred.

Fair Value Measurements

The Company follows ASC 820 – "Fair Value Measurement", which provides a common definition of fair value as the price that would be received to sell an asset or paid to transfer a liability in a transaction between market participants. ASC 820 does not require any new fair value measurements, but applies to most current accounting pronouncements that require or permit fair value measurements.

The standard provides guidance on the methods used to measure fair value, including the market, income, and cost approach. These approaches require the use of certain assumptions in models that market participants would use in pricing assets and liabilities. The models may use readily observable, market corroborated, or generally unobservable inputs.

The standard also requires assets and liabilities that are measured at fair value on a recurring basis (at least annually) to be classified and disclosed in one of the following three categories:

- **Level 1** Quoted market prices in active markets for identical assets or liabilities;
- Level 2 Observable market-based inputs or unobservable inputs that are corroborated by market data; and
- **Level 3** Unobservable inputs that are not corroborated by market data.

The fair values of the Company's cash equivalents, accounts receivable, accounts payable and accrued liabilities approximate their carrying values because of their short-term nature. The Company's debt approximates fair value due to the nature of having variable interest rates.

Appia, Inc. (in thousands except share data and per share data)

The Company has issued warrants to purchase preferred stock in connection with the issuance of subordinated debt. The Company accounted for these warrants in long-term liabilities at fair value upon the issuance based on the specific terms of the agreement. The fair value of the preferred stock warrants was estimated using the Black-Scholes option-pricing model. The warrants are not actively traded and are valued using observable and unobservable market data for inputs. Significant market data inputs used to determine the fair value of the warrants include the estimated the business enterprise value of the Company using a discounted cash flow analysis, expected life of the warrants, risk-free interest rate and the implied volatility of The Company's stock.

The following table summarizes the Company's financial assets and liabilities measured at fair value on a recurring basis as of the respective balance sheet dates.

Liabilities	Balance Sheet	December 31, 2013			December 31, 2012		
(in thousands)	Classification	Level 1	Level 2	Level 3	Level 1	Level 2	Level 3
		\$	\$	\$	\$	\$	\$
Warrant liability	Noncurrent liabilities	-	-	1,109	-	-	-

The Company follows ASC 825 – "Financial Instruments" which permits companies to voluntarily choose, at specified election dates, to measure specified financial assets and liabilities and other items at fair value that are not currently required to be measured at fair value. Subsequent changes in fair value would then be required to be reported in earnings each reporting period. The Company currently has not elected the fair value option for any eligible items. Any future effect of this pronouncement will be dependent upon the nature and amount of eligible items that the Company elects to account for using the fair value option.

Impairment of Long-lived Assets

The Company evaluates the recoverability of its property and equipment and definite-lived intangible assets in accordance with ASC 360 Property, Plant and Equipment. ASC 360 requires recognition of impairment of long-lived assets in the event the net book value of such assets exceeds the estimated future undiscounted cash flows attributable to such assets. Accordingly, when indicators of impairment are present, the Company evaluates the carrying value of these assets in relation to the operating performance of the business and the future undiscounted cash flows expected to result from the use and disposition of these assets. The Company did not have any impairment for the years ended December 31, 2013 and 2012.

Redeemable Convertible Preferred Stock

The carrying value of redeemable convertible preferred stock is increased each reporting period such that the carrying amount will equal the redemption amount at the redemption date. These increases, if any, are affected through charges against additional paid-in capital, to the extent it is available, and then against accumulated deficit.

Stock-based Compensation

The Company applies ASC 718 – "Compensation – Stock Compensation," which requires the measurement and recognition of compensation expense, using a fair value-based method, for all share-based awards made to the Company's employees and directors, including grants of

Appia, Inc. (in thousands except share data and per share data)

stock options, restricted stock and other stock-based plans. The application of this standard requires significant judgment and the use of estimates, particularly assumptions such as stock price volatility and expected option lives, to value equity-based compensation. The Company recognizes the stock compensation expense over the requisite service period of the individual grants, which generally equals the vesting period.

Foreign Currency Translation

The Euro has been designated as the functional currency of the Company's wholly-owned subsidiary in Germany. Assets and liabilities recorded in foreign currency are translated into U.S. dollars at periodend exchange rates. Revenues and expenses are recorded at average rates of exchange during the year. The resulting gains and losses arising from the translation of net assets located outside the U.S. are recorded as a cumulative translation adjustment on the consolidated balance sheets. Realized and unrealized transaction gains and losses are included in income or loss in the period in which they occur. Foreign currency transaction gains or losses realized by the Company were insignificant for the period ending December 31, 2013 and 2012.

Income Taxes

The Company accounts for income taxes using an asset and liability method which requires the recognition of deferred tax assets or liabilities for the temporary differences between financial reporting and tax bases of the Company's assets and liabilities and for tax carryforward at enacted statutory tax rates in effect for the years in which the differences are expected to reverse. The effect on deferred taxes of a change in tax rates is recognized in income in the period that includes the enactment date. In addition, valuation allowances are established when necessary to reduce deferred tax assets to the amounts expected to be realized. A full valuation allowance has been recorded at December 31, 2013 and 2012. The Company recognizes the tax benefit of an uncertain tax position only after determining that the relevant tax authority would more-likely-than-not sustain the position following an audit. The Company recognizes interest and penalties related to uncertain tax positions in income tax expense.

Concentration of Credit Risk and Significant Customers

The Company partners with various suppliers to acquire mobile users. For the year ended December 31, 2013 and 2012, one supplier made up 37% and 74%, respectively, of the total cost of revenues.

The Company generally works with numerous advertisers for whom it drives installs and downloads for. During 2013 one customer made up 46% of revenues in the consolidated statement of operations.

Contingencies

In accordance with ASC 450 – "Contingencies" amounts associated with contingencies are recorded as charges to earnings when the Company, after taking into consideration the facts and circumstances associated with each matter, including settlement offers, if any, has determined that it is probable that a liability has been incurred and the amount of the loss can reasonably be estimated. The Company is subject to various legal matters in the ordinary

Appia, Inc. (in thousands except share data and per share data)

course of business. In the opinion of management, the ultimate outcome of such matters will not have a material adverse effect on the financial condition or results of operation of the Company.

Accumulated Other Comprehensive Loss

In addition to net losses, comprehensive loss includes changes in foreign currency translation adjustments. Comprehensive loss for the period ending December 31, 2013 and 2012 was \$585 and \$5,109, respectively.

Recent Accounting Pronouncements

In July 2013, the FASB issued a revised accounting standard, which clarifies that an unrecognized tax benefit, or a portion of an unrecognized tax benefit, should be presented in the financial statements as a reduction to a deferred tax asset for a net operating loss carryforwards, a similar tax loss, or a tax credit carryforwards if such settlement is required or expected in the event the uncertain tax position is disallowed. In situations where a net operating loss carryforwards, a similar tax loss, or a tax credit carryforwards is not available at the reporting date under the tax law of the applicable jurisdiction or the tax law of the jurisdiction does not require, and the entity does not intend to use, the deferred tax asset for such purpose, the unrecognized tax benefit should be presented in the financial statements as a liability and should not be combined with deferred tax assets. This standard is effective prospectively for fiscal years, and interim periods within those years, beginning after December 15, 2013. Retrospective application is permitted. The Company will adopt this standard in 2014 and does not expect the adoption to have a material impact on its financial statements.

3 Property and Equipment

Property and equipment consists of the following at December 31, 2013 and 2012:

	2013	2012
	\$	 \$
Computer equipment	890	699
Software	1,370	160
Less - Accumulated depreciation and amortization	(828)	(467)
Property and equipment, net	1,432	392

Depreciation and amortization expense for the year ended December 31, 2013 and 2012 was \$170 and \$177, respectively.

Unamortized software development costs as of December 31, 2013 and 2012 were \$1,001 and \$35, respectively. Software development depreciation expense for the year ended December 31, 2013 and 2012 was \$196 and \$29, respectively.

Appia, Inc. (in thousands except share data and per share data)

4 Debt

Short-term and long-term portions of debt are as follows as of December 31:

Debt	2013	2012
	\$	\$
Short-term Short-term		
Bank debt	600	501
Subordinated debt	-	-
Total	600	501
Long-term		
Bank debt	750	910
Subordinated debt	8,000	-
Less - Debt discount	(934)	
Total	7,816	910

In February 2010, the Company entered into an agreement with Silicon Valley Bank (SVB) to borrow \$2,000. The funds were used to pay off existing debt that Handango had outstanding with SVB. The borrowings are secured by substantially all of the assets of the Company and bears interest at the greater of prime plus 3% or 7%. Both interest and principal are payable monthly over 36 months. The Company paid and capitalized \$34 of loan origination fees.

In June 2010, the Company entered into an Amendment with SVB. The amendment modified the original loan, lowering the interest rate to the greater of prime plus 2% or 6%. The Company paid and capitalized \$9 of loan origination fees. The loan was paid off in February 2013.

In August 2012, the Company borrowed \$1,300 from Silicon Valley Bank in the form of a term loan, repayable over 36 months. In addition the Company obtained a \$700 line of credit based on outstanding accounts receivable with all amounts due in 1 year. The loan bears interest at the greater of 2.5%+prime or 6.5%. The line of credit has interest at the greater of 1.5%+prime or 5.50%. The Company paid and capitalized interest of \$13 of loan origination fees. The bank is also due a success fee in the event of an exit event. This fee can range from \$0 to \$150, depending on the Company's valuation at the time of the event.

In April 2013, the Company amended its loan agreement with Silicon Valley Bank to increase the term loan to \$1,500 and increase the line of credit to \$2,000. The new maturity date for the term loan and line of credit is April 2016 and the interest rates remain unchanged. The Company paid and capitalized interest of \$12 of loan origination fees. The bank is also due a Make-Whole Premium ranging from 0-2% of the unpaid balance depending on the date of the payoff.

In April 2013, the Company borrowed \$5,000 from North Atlantic SBIC IV, LP in the form of an interest only term loan, repayable in April 2019. The loan bears interest of 10% if paid monthly, otherwise accrues at 15% per month. The borrower is due a prepayment premium ranging from 1-5% of the unpaid balance depending on date of the prepayment. The Company also issued 263,200 Series C preferred stock warrants at a \$0.001 exercise price and 112,800 Series C

Appia, Inc. (in thousands except share data and per share data)

preferred stock warrants at \$7.41 exercise price. The Company recorded a debt discount of \$615 related to the issuance of these warrants which is being amortized over the term of the debt. The loan is subordinate to Silicon Valley Bank.

In October 2013, the Company amended its loan agreement with Silicon Valley Bank to increase its line of credit to \$3,500. The maturity date and interest remain unchanged. The Company paid and capitalized interest of \$5 of loan origination fees.

In October 2013, the Company borrowed \$3,000 from North Atlantic SBIC IV, LP in the form of an interest only term loan, repayable in April 2019. The loan bears interest of 10% if paid monthly, otherwise accrues at 15% per month. The borrower is due a prepayment premium ranging from 1-5% of the unpaid balance depending on date of the prepayment. The Company also issued 157,920 Series C preferred stock warrants at a \$0.001 exercise price and 67,680 Series C preferred stock warrants at a \$7.41 exercise price. The Company recorded a debt discount of \$408 related to the issuance of these warrants which is being amortized over the term of the debt. The loan is subordinate to Silicon Valley Bank.

The Company's debt agreement with Silicon Valley Bank contain various restrictive covenants. The Company failed to meet some of these required covenants for the fiscal years being reported on. On April 4, 2013, October 2013 and March 2014 the bank provided a waiver to the Company for these requirements.

The Company's debt agreements with North Atlantic SBIC IV, LP contain various restrictive covenants. The Company failed to meet some of these required covenants for year ended December 31, 2013. The Company received waivers from North Atlantic SBIC IV, LP on March 2014.

Future maturities of debt at December 31, 2013 are as follows:

Year	Amount
	\$
2014	600
2014 2015 2016	600
2016	150
2017	-
Thereafter	8,000
Total	9,350

The Company also enters into agreements to lease hardware. The Company applies ASC 840 Leases to determine if the leases should be treated as operating or capital. During the years ended December 31, 2013 and 2012 the Company entered into leases of \$0 and \$51, respectively. The leases are typically payable over 24 months. Total accumulated depreciation associated with these assets is \$106 and \$64 for December 31, 2013 and 2012, respectively. These leases are were fully paid off in 2013.

Appia, Inc. (in thousands except share data and per share data)

5 Income Taxes

The income tax provision attributable to loss before income taxes for the years ended December 31, 2013 and 2012 was zero due to the valuation allowances recorded against the deferred tax assets associated with the Company's net losses of \$526 and \$5,079 in 2013 and 2012, respectively.

Significant components of the Company's deferred tax assets and liabilities at December 31, 2013 and 2012 are as follows:

	2013	2012
	\$	\$
Current		
Accrued expenses	78	88
Deferred rent	3	10
Allowance for doubtful accounts	192	97
Deferred revenue	-	9
Valuation allowance	(273)	(204)
Total net deferred tax, current	-	-
Noncurrent		
Tax loss carryforwards	8,521	8,273
Intangible assets	2,171	2,595
Warrant liability	30	-
Property and equipment	(431)	(8)
Stock compensation	(22)	(33)
Valuation allowance	(10,269)	(10,827)
Total net deferred tax, noncurrent	-	-
Net deferred tax	-	

At December 31, 2013, the Company has federal and state net operating and economic loss carryforwards of \$22,105 and \$15,010, respectively, available to offset future federal and state taxable income which will begin to expire in 2028 and 2023. The Company also has a charitable contribution carryover of \$1 which begin to expire in 2017. The Company has foreign net operating loss carryforwards of \$1,513 available to offset future foreign taxable income which has no expiration date.

The Tax Reform Act of 1986 contains provisions which limit the ability to utilize the net operating loss carryforwards in the case of certain events including significant changes in ownership interests. If the Company's net operating loss carryforwards are limited, and the Company has taxable income which exceeds the permissible yearly net operating loss carryforwards, the Company would incur a federal income tax liability even though net operating loss carryforwards would be available in future years.

During 2010, the Company recorded goodwill in connection with the Handango acquisition that was subsequently fully impaired for book purposes in the year ended December 31, 2010. The goodwill will be amortized for tax purposes over 15 years under Section 197.

Appia, Inc. (in thousands except share data and per share data)

The Company has established a valuation allowance against its deferred tax assets due to the uncertainty surrounding the realization of such assets. The valuation allowance decreased by \$489 during 2013, as a result of the increase in the tax loss carryforwards.

The Company files income tax returns in the U.S. federal jurisdiction and various states and foreign countries. With few exceptions, the Company is no longer subject to U.S. federal, state and local or non-U.S. income tax examinations by tax authorities for years before 2008. As of December 31, 2013 and 2012, no taxing jurisdictions have proposed any significant adjustments to the Company's tax positions. Management does not anticipate any material changes to its financial position or the requirement of material additional payments.

Taxes computed at the statutory federal income tax rate of 34% are reconciled to the provision for income taxes as follows:

_	2013		2012		
	% of Pretax			% of Pretax	
	Amount	Earnings	Amount	Earnings	
	\$	%	\$	%	
United States federal tax at statutory rate	(179)	(34.0%)	(1,758)	(34.0%)	
State taxes (net of federal benefit)	(5)	(0.9%)	(235)	(4.5%)	
Change in valuation reserve	(487)	(92.8%)	1,942	37.5%	
Change in state rate	235	44.6%	-	0.0%	
Other, including adjustments to true-up prior year tax returns and foreign					
exchange rate differences	375	71.5%	-	0.0%	
Foreign tax rate differential	(2)	(0.4%)	26	0.5%	
Nondeductible expenses	63	12%	25	0.5%	
Provision for income taxes	-	0.0%	-	0.0%	

6 Stockholders' Deficit and Redeemable Convertible Preferred Stock

The Company has authorized 17,625,600 shares of common stock, \$0.001 par value and 11,537,614 shares of preferred stock, par value \$0.001, of which 3,275,000 shares are designated as Series A convertible preferred stock (Series A preferred), 2,569,169 as Series B convertible preferred stock (Series B preferred) and 2,612,818 as Series C convertible preferred stock (Series C preferred), 89,127 as Series D convertible preferred stock (Series D preferred) and 2,991,500 shares are designated as Series FF convertible preferred stock (Series FF preferred).

On April 3, 2008, the Company was founded through the issuance of 1,000,000 shares of common stock at a price of \$0.001 per share.

On April 15, 2008, the Company issued 2,902,250 shares of Series FF preferred stock at a price of \$0.001 per share and 2,424,630 shares of restricted common stock at a price of \$0.001 per share. Shares of restricted common stock vest over a period of four years. On October 13, 2008, the Company repurchased 223,880 shares of restricted common stock at a price of \$0.001 per share.

Appia, Inc. (in thousands except share data and per share data)

On June 3, 2008, the Company issued 3,250,000 shares of Series A preferred stock at a price of \$1.00 per share. As part of this transaction, the Company incurred transaction costs of \$83, which were applied against the proceeds of this sale. The carrying value of the preferred stock will be accreted to its redemption price, including dividends.

On December 10, 2008, the Company issued 25,000 shares of Series A preferred stock at a price of \$1.00 per share.

On June 9, 2010, the Company issued 2,140,974 shares of Series B preferred stock at a price of \$5.84 per share. As part of this transaction, the Company incurred transaction costs of \$75, which were applied against the proceeds of this sale. The carrying value of the preferred stock will be accreted to its redemption price, including dividends.

On August 6, 2010, the Company issued 428,195 shares of Series B preferred stock at a price of \$5.84 per share. As part of this transaction, the Company incurred transaction costs of \$31, which were applied against the proceeds of this sale. The carrying value of the preferred stock will be accreted to its redemption price, including dividends.

On April 27, 2010, the Company allowed the exchange of 89,250 shares of Common Stock for Series FF Preferred Stock, for one of the Company's founders.

On March 21, 2011, the Company issued 1,348,814 shares of Series C preferred stock at a price of \$7.41 per share. As part of this transaction, the Company incurred transaction costs of \$85, which were applied against the proceeds of this sale. The carrying value of the preferred stock will be accreted to its redemption price, including dividends.

On September 26, 2012, the Company issued 89,127 shares of Series D preferred stock at a price of \$16.83 per share. As part of this transaction, the Company incurred transaction costs of \$67, which were applied against the proceeds of this sale. The carrying value of the preferred stock will be accreted to its redemption price, including dividends.

Significant features of the common stock and Series A, Series B, Series C, Series D and FF preferred are as follows:

Voting

The holders of Series A, Series B, Series C, Series D and Series FF preferred stock shall be entitled to the number of votes equal to the number of shares of common stock into which such shares of preferred stock could be converted. The holders of common stock are entitled to vote equally based on their respective shares outstanding.

Dividends

Series A, Series B, Series C and Series D preferred stockholders are entitled to receive annual dividends of \$0.08 per share. Holders of Series FF preferred stock and common stock shall be entitled to dividends when and if declared by the Board of Directors and shall share dividends on a pro rata basis based on the number of shares of common stock that would be held by each upon conversion of all Series A, Series B, Series C, Series D and Series FF preferred stock into common stock.

Appia, Inc. (in thousands except share data and per share data)

Conversion

Each share of Series A, Series B, Series C and Series D preferred stock shall be convertible, at the option of the holder, at any time after the date of issuance and without payment of additional consideration, into such number of common shares as is determined by dividing the original issue price of \$1.00 by \$1.00 for Series A, by \$5.84 for Series B, by \$7.41 for Series C and by \$16.83 for Series D preferred stock holders. Each share of Series FF preferred stock shall be convertible, at the option of the holder, at any time after the date of issuance and without payment of additional consideration, into such number of common shares as is determined by dividing the original issue price of \$0.001 by \$0.001. The conversion price is subject to adjustment for subdivisions, dividends, combinations, reclassifications, merger, sale, and other transactions. If the Company issues or sells additional shares of Common Stock less than the original issue price, that is deemed dilutive (Qualifying Dilutive Issuance), Series B and Series C are entitled to reduced conversion rates equal to the price the additional shares are issued at. The Series B and Series C Qualifying Dilutive issuance dates expired on April 9, 2012.

Automatic Conversion

Each share of Series A and Series B preferred stock shall automatically be converted into common stock at the then effective conversion price upon the completion of an underwritten public offering involving the sale of the Company's common stock at prices per share greater than three times the Original Issue price of Series C Preferred Stock and net cash proceeds of at least \$50 million. Each share of Series A and B preferred stock shall also automatically be converted into common stock at the then effective conversion price upon the written consent or agreement of two-thirds (2/3) of the Series A and Series B preferred voting as a group.

Each share of Series C preferred stock shall automatically be converted into common stock at the then effective conversion price upon the completion of an underwritten public offering involving the sale of the Company's common stock net cash proceeds of at least \$50 million. Each share of Series C preferred stock shall also automatically be converted into common stock at the then effective conversion price upon the written consent or agreement of the majority of the Series C preferred holders voting as a group.

Each share of Series D preferred stock shall automatically be converted into common stock at the then effective conversion price upon the closing of a qualified public offering or upon the written consent or agreement of the majority shares voting together as a single group.

Appia, Inc. (in thousands except share data and per share data)

Liquidation Preference

Upon liquidation, dissolution, or winding up of the Company, holders of the Series A, Series B, Series C and Series D preferred shall be entitled to receive, prior and in preference to any distribution of the assets to holders of Series FF preferred or common stock, an amount equal to the original issue price of \$1.00 per share for Series A, \$5.84 per share for Series B, \$7.41 per share for Series C and \$16.83 for Series D, plus all accrued dividends and all declared and unpaid dividends on the Series A, Series B, Series C and Series D preferred stock. Any assets remaining after such preferential distribution shall be distributed ratably to holders of Series FF preferred stock and common stock on an as-if converted basis. The liquidation value per share, aggregate liquidation value and carrying value for the Series A, Series B, Series C and Series D preferred as of December 31, 2013 and 2012 are as follows:

	2013	2012
Series A Preferred	\$	\$
Liquidation values per share	1.45	1.37
Aggregate liquidation value	4,737	4,476
Carrying value	4,737	4,469
Series B Preferred		
Liquidation values per share	7.49	7.02
Aggregate liquidation value	19,238	18,038
Carrying value	19,207	17,986
Series C Preferred		
Liquidation values per share	9.06	8.47
Aggregate liquidation value	12,222	11,422
Carrying value	12,184	11,367
Series D Preferred		
Liquidation values per share	18.53	17.19
Aggregate liquidation value	1,652	1,532
Carrying value	1,601	1,468

Redemption Preference

The certificate of incorporation also provides for the redemption of the Series A, Series B, Series C and Series D preferred upon agreement by two-thirds (2/3) of the outstanding Series A, Series B, Series C and Series D shares, voting as a single class of stock, beginning on December 31, 2015. The redemption price for the Series A, Series B, Series C and Series D preferred shall be the greater of the original issue price of \$1.00 per share for Series A, \$5.84 per share for Series B, \$7.41 per share for Series C, \$16.83 for Series D plus all accrued, declared, and unpaid dividends or the fair market value of Series A, Series B, Series C or Series D preferred stock on the date of the request.

7 Stock Options and Restricted Stock

During 2008, the Company adopted the 2008 Stock Option Plan (the Plan). A total of 1,942,247 shares of the Company's common stock have been reserved for issuance to employees,

Appia, Inc. (in thousands except share data and per share data)

directors and consultants. The terms of the stock option grants are determined by the Board of Directors. Since inception, all options were granted with an exercise price equal to the deemed fair value of the common stock on the date of grant. Under the Plan, options become vested over various periods, generally ranging up to four years, and expire not more than ten years after the date of grant.

In determining the compensation cost of options granted, as specified by ASC 718 – "Compensation – Stock Compensation," the fair value of each option or restricted stock grant has been estimated on the date of grant using the Black-Scholes option-pricing model. The assumptions used in these calculations are summarized as follows:

	2013	2012
Expected dividend yield		-
Expected stock price volatility	39% - 40%	40% - 45%
Risk-free interest rate	1.71%- 2.96%	1.25% - 2.24%
Expected life of options	10	10
Estimated forfeiture rate for employee option	15%	15%

The Company calculates expected stock volatility using historical volatility for a peer group of companies, as the Company believes the expected volatility will approximate historical volatility of the peer group. The risk-free interest rate for the expected terms of the stock options is based on the rates paid on US Treasury instruments with similar lives at the time of the grant. The estimated forfeiture rate for employee options is based on historic forfeiture rates.

The following table summarizes the Company's stock option grant activity during the period from January 1, 2012 through December 31, 2013:

	Number of Shares	Weighted Average Exercise Price
		\$
Balance at December 31, 2011	191,137	0.22
Granted	501,900	0.26
Vested	(5,906)	0.18
Forfeited	(56,581)	0.25
Balance at December 31, 2012	630,550	0.25
Granted	262,018	0.28
Vested	(75,749)	0.26
Forfeited	(141,353)	0.26
Balance at December 31, 2013	675,466	0.26

Appia, Inc. (in thousands except share data and per share data)

The following table summarizes the stock options outstanding at December 31, 2013 and 2012:

	Exercise Price	Options Outstanding	Weighted Average Remaining Contractual Life of Options Outstanding	Options Exercisable	Weighted Average Grant Date Fair Value of Options Exercisable
					\$
	0.10	44,700	6.50	37,200	0.07
	0.26	585,850	9.00	50,030	0.17
December 31, 2012		630,550		87,230	
•	.10	41,200	5.40	39,844	0.07
	0.26	416,248	7.95	201,983	0.15
	0.28	218,018	9.65	-	-
December 31, 2013		675,466		241,827	

As of December 31, 2013 and 2012, there was \$55 and \$57, respectively of total unrecognized compensation cost related to non-vested options. The Company recorded \$19 and \$16 of compensation cost for December 31, 2013 and 2012, respectively related to stock options recognized over a weighted average remaining life of 8.3 years.

The following table summarizes non-vested stock options during the period from January 1, 2012 through December 31, 2013:

	Number of Shares	Weighted Average Grant Date Fair Value
		\$
Non-vested common stock options outstanding, December 31, 2011	145,753	0.11
Granted	501,900	0.15
Vested	(54,424)	0.13
Forfeited	(49,909)	0.15
Non-vested common stock options outstanding, December 31, 2012	543,320	0.14
Granted	262,018	0.15
Vested	(190,438)	0.14
Forfeited	(136,948)	0.14
Non-vested common stock options outstanding, December 31, 2013	477,952	0.15

The Company at times issues Restricted Common Stock grants to employees, directors and consultants. The grants typically vest over a period of 4 years. The Company includes these costs in stock-based compensation and recognized on a straight-line basis. At December 31, 2013 and 2012 there was \$63 and \$26 of unrecognized compensation expense related to restricted stock. The Company recorded \$22 and \$12 as compensation cost related to the restricted stock for the years ended December 31, 2013 and 2012 respectively.

Appia, Inc. (in thousands except share data and per share data)

The following table summarizes the Company's restricted stock award activity during the period from January 1, 2012 through December 31, 2013:

	Number of
	Shares
Non-vested restricted stock outstanding at December 31, 2011	355,687
Granted	-
Vested	(113,500)
Forfeited	(39,375)
Non-vested restricted stock outstanding at December 31, 2012	202,812
Granted	280,000
Vested	(104,062)
Forfeited	(30,625)
Non-vested restricted stock outstanding at December 31, 2013	348,125

8 Employee Benefit Plan

The Company's eligible employees participate in a 401(k) plan. Participants may contribute a percentage of their salary, limited by IRS regulations. The Plan allows the Company to match a percentage of the amount contributed by the employee. In addition, the Company may make discretionary profit-sharing contributions to the plan. The Company did not elect to make any contributions during the years ended December 31, 2013 or 2012.

9 Commitments

The Company leases its office facilities and certain office equipment under operating lease agreements. Non-cancelable leases may contain escalation clauses which require rent to be recognized on a straight-line basis over the term of the lease agreement. Expense related to operating leases, recognized on the straight-line basis, was approximately \$1,013 and \$715 for the years ended December 31, 2013 and 2012, respectively.

Future minimum lease payments under non-cancelable operating leases at December 31, 2013 are as follows:

	Amount
	\$
2014	386
2015	174
Thereafter	-
	560

10 Subsequent Events

Management has evaluated subsequent events through June 26, 2014, the date of the financial statements were available for issuance. All matters requiring recognition or disclosure have been included in the consolidated financial statements.

Financial Statements

Appia, Inc.

For the nine months ended September 30, 2014 and 2013

Appia, Inc.

Table of contents

Consolidated financial statements:

Balance sneets (unaudited)	F-28
Statements of operations and comprehensive income (loss) (unaudited)	F-29
Statements of changes in stockholders' deficit (unaudited)	F-30
Statements of cash flows (unaudited)	F-31
Notes to financial statements	F-32 - F-48

F-27

Appia, Inc.

Consolidated balance sheets (Unaudited)

(in thousands, except share data and per share data)

September 30	2014	2013
	\$	\$
Assets		
Current assets:		
Cash and cash equivalents	1,952	5,600
Accounts receivable, net	5,090	8,058
Prepaid expenses and other current assets	245	306
Total current assets	7,287	13,964
Property and equipment, net	2,677	1,232
Other assets	9	<u> </u>
Total assets	9,973	15,213
Liabilities, redeemable convertible preferred stock and stockholders' deficit		
Current liabilities:		
Accounts payable	721	2,079
Accrued expenses and other current liabilities	4,494	4,256
Short-term capital lease	-	6
Short-term debt	600	600
Deferred revenue	1	245
Total current liabilities	5,816	7,186
Warrant liability	3,129	615
Long-term debt	7,499	5,336
Total liabilities	16,444	13,137
Commitments (Note 9)		
Redeemable convertible preferred stock, Series A, 3,275,000 shares authorized; 3,275,000 issued and outstanding as of		
September 30, 2014 and 2013, respectively	4,932	4,672
Redeemable convertible preferred stock, Series B, 2,569,169 shares authorized; 2,569,169 issued and outstanding as of		
September 30, 2014 and 2013, respectively	20,123	18,902
Redeemable convertible preferred stock, Series C, 2,612,818 shares authorized; 1,348,814 and outstanding at		
September 30, 2014 and 2013, respectively	12,796	11,981
Redeemable convertible preferred stock, Series D, 89,127 shares authorized; 89,127 issued and outstanding at		
September 30, 2014 and 2013, respectively	1,701	1, <u>56</u> 8
Total redeemable convertible preferred stock	39,552	37,123
Stockholders' deficit:		
Series FF preferred stock, par value \$0.001; 2,991,500 shares authorized; 2,991,500 shares issued and outstanding as		
of September 30, 2014 and 2013	3	3
Common stock, par value \$0.001;17,625,600 shares authorized; 3,927,456 and 3,909,932 shares issued and		
outstanding as of September 30, 2014 and 2013, respectively	3	3
Accumulated other comprehensive income	116	14
Accumulated deficit	(46,145)	(35,067)
Total stockholders' deficit	(46,023)	(35,047)
Total liabilities, redeemable convertible preferred stock and stockholders' deficit	9,973	15,213

The accompanying notes are an integral part of these consolidated financial statements.

Appia, Inc.

Consolidated statements of operations and comprehensive income (loss) (Unaudited)

(in thousands)

For the nine months ended September 30	2014	2013
	\$	\$
Revenues	21,930	38,470
Cost of revenues	17,366	27,912
Gross profit	4,564	10,558
Operating expenses:		
Sales and marketing	2,569	2,655
Technology	2,246	1,823
General and administrative	3,431	4,426
Total operating expenses	8,246	8,904
Income (loss) from operations	(3,682)	1,654
Other (expense):		
Interest expense	(3,215)	(387)
Income (loss) before income taxes	(6,897)	1,267
Income tax (benefit)	-	
Net income (loss)	(6,897)	1,267
Other comprehensive income (loss):		
Foreign currency translation adjustments	116	(45)
Other comprehensive income (loss)	116	(45)
Comprehensive income (loss)	(6,781)	1,222

The accompanying notes are an integral part of these consolidated financial statements.

Appia, Inc.

Consolidated statements of changes in stockholders' deficit (Unaudited)

(in thousands, except share data)

	Series I				Additional		Other	Total
_	Preferred	stock	Common	stock	Paid-in	Accumulated	Comprehensive	Stockholders'
	Shares	Value	Shares	Value	Capital	Deficit	Income	Deficit
		\$		\$	\$	\$	\$	\$
Balance as of December 31,								
2013	2,991,500	3	3,909,932	3	-	(37,462)	-	(37,456)
Issuance of common stock	-	-	47,524	-	12	-	-	12
Forfeiture of restricted stock								
shares	-	-	(30,000)	-	-	-	-	-
Stock-based compensation								
expense	-	-	-	-	27	0	-	27
Accretion of Series A								
preferred stock	-	-	-	-	(39)	(157)	-	(196)
Accretion of Series B								
preferred stock	-	-	-	-	-	(916)	-	(916)
Accretion of Series C								
preferred stock	-	-	-	-	-	(613)	-	(613)
Accretion of Series D								
preferred stock	-	-	-	-	-	(100)	-	(100)
Foreign currency translation								
adjustment	-	-	-	-	-	-	116	116
Net loss for the period	-	-	-	-	-	(6,897)	-	(6,897)
Balance as of September 30,								
2014	2,991,500	3	3,927,456	3	-	(46,145)	116	(46,023)

The accompanying notes are an integral part of these consolidated financial statements.

Appia, Inc.

Consolidated statements of cash flows (Unaudited)

(in thousands)

For the nine months ended September 30	2014	2013
	\$	\$
Cash flows from operating activities:		
Net income (loss)	(6,897)	1,267
Adjustments to reconcile net loss to net cash used in operating activities:		
Depreciation and amortization	637	238
Amortization of deferred loan costs	10	8
Stock-based compensation expense	27	39
Amortization of debt discount	133	51
Loss on sale of assets	37	
Change in fair market value of warrant liability	2,019	-
Changes in operating assets and liabilities:		
Accounts receivable	628	(3,997)
Prepaid expenses and other current assets	321	(17
Accounts payable	(2,013)	(413)
Deferred revenue	(75)	(113
Accrued expenses and other liabilities	1,694	1,478
Net cash flows used in operating activities	(3,479)	(1,459
Cash flows from investing activities:		
Internally developed software	(1,898)	(907)
Proceeds from sale of assets	22	-
Purchases of property and equipment	(43)	(171)
Net cash flows used in investing activities	(1,919)	(1,078
Cash flows from financing activities:		
Loan proceeds	-	5,200
Stock option exercise	12	5
Principal payments on capital leases	-	(49)
Payments on term loan	(450)	(111)
Net cash flows provided (used) by financing activities	(438)	5,045
Effect of exchange rate changes on cash and cash equivalents	(3)	1
Net increase (decrease) in cash and cash equivalents	(5,839)	2,509
Cash and cash equivalents, beginning of year	7,791	3,091
Cash and cash equivalents, end of year	1,952	5,600
Supplemental disclosure of cash flow information	·	
Cash paid during the year for interest	63	286
Supplemental disclosure of non-cash investing and financing activities		
Accretion of redeemable convertible preferred stock	1,825	1,832

The accompanying notes are an integral part of these financial statements.

Appia, Inc. (in thousands except share data and per share data)

Notes to consolidated financial statements

1 Organization and Basis of Presentation

On April 3, 2008, Pinmot, Inc. was incorporated for the purpose of acquiring a portfolio of mobile technology assets. On June 3, 2008, Pinmot, Inc. acquired the smartphone assets of Motricity, Inc. and the outstanding common stock of Motricity Deutschland GmbH. Pinmot, Inc. was subsequently renamed PocketGear, Inc. and its German subsidiary was renamed PocketGear Deutschland GmbH. PocketGear, Inc, subsequently renamed Appia, Inc. (the Company) as of February 2011. The Company is a mobile advertising network focused on driving installs of mobile applications (iOS/Android) on behalf of advertisers who want to grow their mobile audience. The Company provides mobile advertising performance solutions for both advertisers and publishers. Its technology is focused on driving user acquisition and monetization through ad walls and other ad formats.

2 Summary of Significant Accounting Policies Principles of Consolidation

The consolidated financial statements include the accounts of Appia, Inc. and its wholly-owned subsidiary, PocketGear Deutschland GmbH. All significant intercompany balances have been eliminated.

Use of Estimates

The preparation of consolidated financial statements in conformity with accounting principles generally accepted in the United States of America requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities at the date of the consolidated financial statements and the reported amounts of revenues and expenses during the reporting period. Actual results could differ from those estimates.

Revenue Recognition Advertising Revenue Recognition

The Company recognizes revenue based on the activity of mobile users viewing advertisements and downloading applications via developer applications and mobile websites. Revenues are recognized when the related advertising services are delivered based on the specific terms of the advertising contract/insertion order, which are primarily based on the number of applications installed or downloaded. The Company recognizes revenue based on these terms, as the services have been provided, the fees the Company charges are fixed or determinable, persuasive evidence of an arrangement exists, and collectability is reasonably assured.

In the normal course of business, the Company acts as an intermediary in executing transactions with third parties. The determination of whether revenue should be reported on a gross or net basis is based on an assessment of whether the Company is acting as the principal or an agent in the transaction. In determining whether the Company acts as the principal or an agent, the Company follows the accounting guidance for principal-agent considerations. While

Appia, Inc. (in thousands except share data and per share data)

none of the factors identified in this guidance is individually considered presumptive or determinative, because the Company is the primary obligor and is responsible for (i) identifying and contracting with third-party advertisers, (ii) performing all billing and collection activities including retaining credit risk, and (iii) discretion in selecting suppliers, the Company acts as the principal in these arrangements and therefore reports revenue earned and costs incurred on a gross basis.

Mobile Storefront Revenues

The Company's revenues are derived from sales of applications developed by third parties for use on handheld devices using the Company's e-commerce services, features and technologies to operate the sites.

The Company recognizes revenue from sales of third-party content related products in accordance with the Financial Accounting Standards Board (FASB) Accounting Standards Codification (ASC) 605 – "Revenue Recognition." Product sales, net of promotional discounts, rebates, and return allowances, are recorded when the products are shipped and title passes to customers. Retail items are sold to customers pursuant to a sales contract that provides for transfer of both title and risk of loss upon delivery to the customer. Return allowances, which reduce product revenue by an estimate of expected product returns, are estimated using historical experience. The Company periodically provides incentive offers to customers to encourage purchases. Current discount offers, when accepted by customers, are treated as a reduction to the sales price of the related transaction.

The Company utilizes its own proprietary technology platform to accept and verify customer orders, respond to customer support needs and to execute fulfillment on behalf of its own and third-party products. These applications also manage the process of accepting, authorizing, and charging customer credit cards. The Company uses historical results as a basis for providing an allowance for chargebacks and returns.

The Company applies ASC 605 in determining whether it is appropriate to record the gross amount of product sales and related costs or the net amount earned as commissions. Regarding sales of third-party content related products through its own and affiliate websites, the Company believes it is the primary obligor in sales transactions, has latitude in establishing prices, and holds responsibility for processing of cash receipts related to the transactions. As such, the Company records and presents revenue from these transactions on a gross basis.

The Company also enters into arrangements where it builds customized storefronts, supplying tools to ingest content and merchandising as well as providing data hosting and monitoring services. Revenue is generated from sales of applications from the storefronts and split via revenue share basis. In addition, there are instances where customers also pay one-time upfront fees and/or ongoing periodic fees. For the upfront/periodic fees the Company follows a services model. The upfront fees are deferred until the storefront is launched and operational and recognized taken ratably over the estimated life of the customer relationship.

Cost of Revenue

Cost of revenue consists primarily of amounts due to publishers for the advertising inventory utilized in running mobile advertisements. These amounts are typically either a percentage of

Appia, Inc. (in thousands except share data and per share data)

the advertising revenue earned by the Company based on mobile advertisements that are run on each publishers inventory or a fixed fee for the ad space. The Company recognizes the cost of revenue as the associated revenue is recognized, on a publisher by publisher basis during the period the advertisements run on the publisher's properties. Costs owed to publisher but not yet paid are recorded as accrued cost of revenue.

Technology Costs

Technology expenses consist primarily of wages and benefits for software application development personnel. The Company has focused its research and development efforts on both improving ease of use and functionality of its existing products as well as developing new offerings. The Company expenses research and development costs that do not meet the capitalization criteria for internal-use software. Research and development expense, which includes the amortization of capitalized internally developed software costs, was \$2,246 and \$1,823, respectively for the period ended September 30, 2014 and 2013.

Advertising Costs

The costs of advertising are expensed as incurred or the first time the advertising takes place in accordance with ASC 720 – "Other Expenses." For the period ended September 30, 2014 and 2013, the Company incurred advertising costs of \$267 and \$148, respectively.

Sales and Marketing Costs

Sales and Marketing expense primarily consists of salaries and payroll related costs for Sales and Marketing employees. Additional expenses include costs related to lead generation, communication and promotion.

Cash and Cash Equivalents

The Company considers all highly liquid investments with maturity of three months or less at the date of purchase to be cash equivalents. Cash equivalents are stated at cost and consist primarily of money market accounts. Cash and cash equivalents deposited in banks were in excess of federally insured limits as of September 30, 2014 and September 30, 2013.

Accounts Receivable and Allowance for Doubtful Accounts

Accounts receivable are presented at their face amount, less an allowance for doubtful accounts, on the Company's accompanying consolidated balance sheet. Accounts receivable consist of amounts currently due from customers. The Company generally does not require collateral on sales. The Company periodically evaluates the collectability of its accounts receivable based on a combination of factors. In circumstances where the Company is aware of a specific customer's inability to meet its financial obligations, a specific reserve for bad debts against amounts due is recorded to reduce the related accounts receivable to an amount that the Company reasonably believes is collectible. As of September 30, 2014 and 2013 the Company had reserved \$560 and \$438, respectively, for allowance for doubtful accounts.

Appia, Inc. (in thousands except share data and per share data)

Property and Equipment

Property and equipment is primarily comprised of office furniture and computer equipment, which are recorded at cost and depreciated using the straight-line method over their estimated useful lives ranging from three to five years as follows:

	Useful Life (years)
Computer equipment - Servers	5
Computer equipment - Office	3
Software	5

Included in property and equipment is the cost of internal-use software and website development, including software used to upgrade and enhance the Company's websites. The Company follows the provisions of ASC 350 – "Intangibles – Goodwill and Other." All costs related to the development of internal-use software other than those incurred during the application development stage are expensed. Costs incurred during the application development stage are capitalized and amortized over the estimated useful life of the software (generally three years).

Upon retirement or sale, the cost of assets disposed of and the related accumulated depreciation and amortization are removed from the accounts and any resulting gain or loss is credited or charged to income. Repairs and maintenance costs are expensed as incurred.

Fair Value Measurements

The Company follows ASC 820 – "Fair Value Measurement", which provides a common definition of fair value as the price that would be received to sell an asset or paid to transfer a liability in a transaction between market participants. ASC 820 does not require any new fair value measurements, but applies to most current accounting pronouncements that require or permit fair value measurements.

The standard provides guidance on the methods used to measure fair value, including the market, income, and cost approach. These approaches require the use of certain assumptions in models that market participants would use in pricing assets and liabilities. The models may use readily observable, market corroborated, or generally unobservable inputs.

The standard also requires assets and liabilities that are measured at fair value on a recurring basis (at least annually) to be classified and disclosed in one of the following three categories:

- **Level 1** Quoted market prices in active markets for identical assets or liabilities;
- **Level 2** Observable market-based inputs or unobservable inputs that are corroborated by market data; and
- **Level 3** Unobservable inputs that are not corroborated by market data.

The fair values of the Company's cash equivalents, accounts receivable, accounts payable and accrued liabilities approximate their carrying values because of their short-term nature. The Company's debt approximates fair value due to the nature of having variable interest rates.

Appia, Inc. (in thousands except share data and per share data)

The Company has issued warrants to purchase preferred stock in connection with the issuance of subordinated debt. The Company accounted for these warrants in long-term liabilities at fair value upon the issuance based on the specific terms of the agreement. The fair value of the preferred stock warrants was estimated using the Black-Scholes option-pricing model. The warrants are not actively traded and are valued using observable and unobservable market data for inputs. Significant market data inputs used to determine the fair value of the warrants include the equity value of the transaction (see Note 10), estimated the business enterprise value of the Company using a discounted cash flow analysis, expected life of the warrants, risk-free interest rate and the implied volatility of The Company's stock.

The following table summarizes the Company's financial assets and liabilities measured at fair value on a recurring basis as of the respective balance sheet dates.

Liabilities	Balance Sheet	Se	ptember 30, 20	14	Se	eptember 30, 20	13
(in thousands)	Classification	Level 1	Level 2	Level 3	Level 1	Level 2	Level 3
		\$	\$	\$	\$	\$	\$
Warrant liability	Non-current liabilities	-	-	3,129	-	-	61 ₅

The Company follows ASC 825 – "Financial Instruments" which permits companies to voluntarily choose, at specified election dates, to measure specified financial assets and liabilities and other items at fair value that are not currently required to be measured at fair value. Subsequent changes in fair value would then be required to be reported in earnings each reporting period. The Company currently has not elected the fair value option for any eligible items. Any future effect of this pronouncement will be dependent upon the nature and amount of eligible items that the Company elects to account for using the fair value option.

Impairment of Long-lived Assets

The Company evaluates the recoverability of its property and equipment and definite-lived intangible assets in accordance with ASC 360 Property, Plant and Equipment. ASC 360 requires recognition of impairment of long-lived assets in the event the net book value of such assets exceeds the estimated future undiscounted cash flows attributable to such assets. Accordingly, when indicators of impairment are present, the Company evaluates the carrying value of these assets in relation to the operating performance of the business and the future undiscounted cash flows expected to result from the use and disposition of these assets. The Company did not have any impairment for the period ended September 30, 2014 and 2013.

Redeemable Convertible Preferred Stock

The carrying value of redeemable convertible preferred stock is increased each reporting period such that the carrying amount will equal the redemption amount at the redemption date. These increases, if any, are affected through charges against additional paid-in capital, to the extent it is available, and then against accumulated deficit.

Stock-based Compensation

The Company applies ASC 718 – "Compensation – Stock Compensation," which requires the measurement and recognition of compensation expense, using a fair value-based method, for all share-based awards made to the Company's employees and directors, including grants of

Appia, Inc. (in thousands except share data and per share data)

stock options, restricted stock and other stock-based plans. The application of this standard requires significant judgment and the use of estimates, particularly assumptions such as stock price volatility and expected option lives, to value equity-based compensation. The Company recognizes the stock compensation expense over the requisite service period of the individual grants, which generally equals the vesting period.

Foreign Currency Translation

The Euro has been designated as the functional currency of the Company's wholly-owned subsidiary in Germany. Assets and liabilities recorded in foreign currency are translated into U.S. dollars at periodend exchange rates. Revenues and expenses are recorded at average rates of exchange during the year. The resulting gains and losses arising from the translation of net assets located outside the U.S. are recorded as a cumulative translation adjustment on the consolidated balance sheets. Realized and unrealized transaction gains and losses are included in income or loss in the period in which they occur. Foreign currency transaction gains or losses realized by the Company were insignificant for the period ending September 30, 2014 and 2013.

Income Taxes

The Company accounts for income taxes using an asset and liability method which requires the recognition of deferred tax assets or liabilities for the temporary differences between financial reporting and tax bases of the Company's assets and liabilities and for tax carryforwards at enacted statutory tax rates in effect for the years in which the differences are expected to reverse. The effect on deferred taxes of a change in tax rates is recognized in income in the period that includes the enactment date. In addition, valuation allowances are established when necessary to reduce deferred tax assets to the amounts expected to be realized. A full valuation allowance has been recorded at September 30, 2014 and 2013. The Company recognizes the tax benefit of an uncertain tax position only after determining that the relevant tax authority would more-likely-than-not sustain the position following an audit. The Company recognizes interest and penalties related to uncertain tax positions in income tax expense.

Concentration of Credit Risk and Significant Customers

The Company partners with various suppliers to acquire mobile users. For the periods ended September 30, 2014 and 2013, one supplier made up 11% and 41%, respectively, of the total cost of revenues.

The Company generally works with numerous advertisers for whom it drives installs and downloads for. During the period ended September 30, 2013 one customer made up 54% of revenues in the consolidated statement of operations.

Derecognition of Liabilities

The Company recognized a gain of \$1,268 in September 2014 related to the reversal of liabilities assumed in a previous acquisition. The Company has determined that its obligation to pay these liabilities has expired. The gain is recorded as a reduction to general and administrative expenses in the consolidated statement of operations for the nine months ended September 30, 2014.

Appia, Inc. (in thousands except share data and per share data)

Contingencies

In accordance with ASC 450 – "Contingencies" amounts associated with contingencies are recorded as charges to earnings when the Company, after taking into consideration the facts and circumstances associated with each matter, including settlement offers, if any, has determined that it is probable that a liability has been incurred and the amount of the loss can reasonably be estimated. The Company is subject to various legal matters in the ordinary course of business. In the opinion of management, the ultimate outcome of such matters will not have a material adverse effect on the financial condition or results of operation of the Company.

Accumulated Other Comprehensive Income (Loss)

In addition to net losses, comprehensive loss includes changes in foreign currency translation adjustments. Comprehensive income (loss) for the period ending September 30, 2014 and 2013 was (\$6,781) and \$1,222, respectively.

Recent Accounting Pronouncements

In August 2014, the FASB issued new accounting guidance which addresses management's responsibility to evaluate whether there is substantial doubt about the entity's ability to continue as a going concern and to provide related footnote disclosures. Management's evaluation should be based on relevant conditions and events that are known and reasonably knowable at the date that the financial statements are issued. This guidance is effective for the fiscal year ending after December 15, 2016, and for fiscal years and interim periods thereafter. Early adoption is permitted. The Company does not expect to early adopt this guidance and does not believe that the adoption of this guidance will have a material impact on its financial statements.

In May 2014, the FASB issued new accounting guidance on revenue recognition, which provides for a single five-step model to be applied to all revenue contracts with customers. The new standard also requires additional financial statement disclosures that will enable users to understand the nature, amount, timing and uncertainty of revenue and cash flows relating to customer contracts. Companies have an option to use either a retrospective approach or cumulative effect adjustment approach to implement the standard. This guidance is effective for private companies for fiscal years and interim periods within those years beginning after December 15, 2017. Early adoption is not permitted. The Company is currently evaluating the impact that the implementation of this standard will have on its financial statements.

3 Property and Equipment

Property and equipment consists of the following at September 30, 2014 and 2013:

	2014	2013
	\$	 \$
Computer equipment	780	822
Software	3,275	1,11 <u></u> 5
Less - Accumulated depreciation and amortization	(1,378)	(705)
Property and equipment, net	2,677	1,232

Appia, Inc. (in thousands except share data and per share data)

Depreciation and amortization expense for the period ended September 30, 2014 and 2013 was \$637 and \$238, respectively.

Unamortized software development costs as of September 30, 2014 and 2013 were \$2,395 and \$831, respectively. Software development depreciation expense for the period ended September 30, 2014 and 2013 was \$497 and \$101, respectively.

4 Debt

Short-term and long-term portions of debt are as follows as of September 30:

Debt	2014	2013
	\$	\$
Short-term Short-term		
Bank debt	600	600
Subordinated debt	-	-
Total	600	600
Long-term		
Bank debt	300	900
Subordinated debt	8,000	5,000
Less - Debt discount	(801)	(564)
Total	7,499	5,336

In February 2010, the Company entered into an agreement with Silicon Valley Bank (SVB) to borrow \$2,000. The funds were used to pay off existing debt that Handango had outstanding with SVB. The borrowings are secured by substantially all of the assets of the Company and bears interest at the greater of prime plus 3% or 7%. Both interest and principal are payable monthly over 36 months. The Company paid and capitalized \$34 of loan origination fees.

In June 2010, the Company entered into an Amendment with SVB. The amendment modified the original loan, lowering the interest rate to the greater of prime plus 2% or 6%. The Company paid and capitalized \$9 of loan origination fees. The loan was paid off in February 2013.

In August 2012, the Company borrowed \$1,300 from Silicon Valley Bank in the form of a term loan, repayable over 36 months. In addition the Company obtained a \$700 line of credit based on outstanding accounts receivable with all amounts due in 1 year. The loan bears interest at the greater of 2.5%+prime or 6.5%. The line of credit has interest at the greater of 1.5%+prime or 5.50%. The Company paid and capitalized interest of \$13 of loan origination fees. The bank is also due a success fee in the event of an exit event. This fee can range from \$0 to \$150, depending on the Company's valuation at the time of the event.

In April 2013, the Company amended its loan agreement with Silicon Valley Bank to increase the term loan to \$1,500 and increase the line of credit to \$2,000. The new maturity date for the term loan and line of credit is April 2016 and the interest rates remain unchanged. The Company paid and capitalized interest of \$12 of loan origination fees. The bank is also due a Make-Whole Premium ranging from 0-2% of the unpaid balance depending on the date of the payoff.

Appia, Inc. (in thousands except share data and per share data)

In April 2013, the Company borrowed \$5,000 from North Atlantic SBIC IV, LP in the form of an interest only term loan, repayable in April 2019. The loan bears interest of 10% if paid monthly, otherwise accrues at 15% per month. The borrower is due a prepayment premium ranging from 1-5% of the unpaid balance depending on date of the prepayment. The Company also issued 263,200 Series C preferred stock warrants at a \$0.001 exercise price and 112,800 Series C preferred stock warrants at \$7.41 exercise price. The Company recorded a debt discount of \$615 related to the issuance of these warrants which is being amortized over the term of the debt. The loan is subordinate to Silicon Valley Bank. The Company has elected to not pay interest starting with the January 1, 2014 payment and is accruing it at 15%.

In October 2013, the Company amended its loan agreement with Silicon Valley Bank to increase its line of credit to \$3,500. The maturity date and interest remain unchanged. The Company paid and capitalized interest of \$5 of loan origination fees.

In October 2013, the Company borrowed \$3,000 from North Atlantic SBIC IV, LP in the form of an interest only term loan, repayable in April 2019. The loan bears interest of 10% if paid monthly, otherwise accrues at 15% per month. The borrower is due a prepayment premium ranging from 1-5% of the unpaid balance depending on date of the prepayment. The Company also issued 157,920 Series C preferred stock warrants at a \$0.001 exercise price and 67,680 Series C preferred stock warrants at a \$7.41 exercise price. The Company recorded a debt discount of \$408 related to the issuance of these warrants which is being amortized over the term of the debt. The loan is subordinate to Silicon Valley Bank. The Company has elected to not pay interest starting with the January 1, 2014 payment and is accruing it at 15%.

In March 2014, the Company entered into an Amendment with SVB. The amendment modified the terms of the line of credit to expand the amount of foreign accounts receivable to be included in the eligible borrowing base. The maturity date and interest remain unchanged.

The Company's debt agreement with Silicon Valley Bank contain various restrictive covenants. The Company failed to meet some of these required covenants for the periods being reported on. On April 4, 2013, October 2013 and March 2014 and November 2014 the bank provided a waiver to the Company for these requirements.

The Company's debt agreements with North Atlantic SBIC IV, LP contain various restrictive covenants. The Company failed to meet some of these required covenants for the periods being reported on. The Company received waivers from North Atlantic SBIC IV, LP on March 2014 and November 2014.

Future maturities of debt at September 30, 2014 are as follows:

Year	Amount
	\$
2014	150
2015	600
2016 2017	150
2017	-
Thereafter	8,000
Total	8,900

Appia, Inc. (in thousands except share data and per share data)

The Company also enters into agreements to lease hardware. The Company applies ASC 840 Leases to determine if the leases should be treated as operating or capital. Total accumulated depreciation associated with these assets is \$52 and \$96 for the period ended September 30, 2014 and 2013, respectively. These leases are were fully paid off in 2013.

5 Income Taxes

The Company computes its provision for income taxes by applying the estimated annual effective tax rate, adjusted for any material items. For each of the nine months ended September 30, 2014 and 2013, no provision or benefits for income taxes have been recorded and accordingly the Company's effective tax rate for each of the nine months ended September 30, 2014 and 2013 was 0%. The Company has recorded a full valuation allowance against its net deferred tax assets based on the Company's assessment regarding the realizability of these net deferred tax assets in future periods. At September 30, 2014, the Company had no unrecognized tax benefits that would affect the Company's effective tax rate.

Estimated net operating loss carryforwards (NOL's) as of September 30, 2014 are \$26,765 for federal purposes and \$17,810 for state purposes. The NOL's begin to expire in 2028 and 2023 respectively. With few exceptions, the company is no longer subject to US federal, state, and local non-US income tax examinations by tax authorities for years before 2010.

6 Stockholders' Deficit and Redeemable Convertible Preferred Stock

The Company has authorized 17,625,600 shares of common stock, \$0.001 par value and 11,537,614 shares of preferred stock, par value \$0.001, of which 3,275,000 shares are designated as Series A convertible preferred stock (Series A preferred), 2,569,169 as Series B convertible preferred stock (Series B preferred) and 2,612,818 as Series C convertible preferred stock (Series C preferred), 89,127 as Series D convertible preferred stock (Series D preferred) and 2,991,500 shares are designated as Series FF convertible preferred stock (Series FF preferred).

On April 3, 2008, the Company was founded through the issuance of 1,000,000 shares of common stock at a price of \$0.001 per share.

On April 15, 2008, the Company issued 2,902,250 shares of Series FF preferred stock at a price of \$0.001 per share and 2,424,630 shares of restricted common stock at a price of \$0.001 per share. Shares of restricted common stock vest over a period of four years. On October 13, 2008, the Company repurchased 223,880 shares of restricted common stock at a price of \$0.001 per share.

On June 3, 2008, the Company issued 3,250,000 shares of Series A preferred stock at a price of \$1.00 per share. As part of this transaction, the Company incurred transaction costs of \$83, which were applied against the proceeds of this sale. The carrying value of the preferred stock will be accreted to its redemption price, including dividends.

On December 10, 2008, the Company issued 25,000 shares of Series A preferred stock at a price of \$1.00 per share.

Appia, Inc. (in thousands except share data and per share data)

On June 9, 2010, the Company issued 2,140,974 shares of Series B preferred stock at a price of \$5.84 per share. As part of this transaction, the Company incurred transaction costs of \$75, which were applied against the proceeds of this sale. The carrying value of the preferred stock will be accreted to its redemption price, including dividends.

On August 6, 2010, the Company issued 428,195 shares of Series B preferred stock at a price of \$5.84 per share. As part of this transaction, the Company incurred transaction costs of \$31, which were applied against the proceeds of this sale. The carrying value of the preferred stock will be accreted to its redemption price, including dividends.

On April 27, 2010, the Company allowed the exchange of 89,250 shares of Common Stock for Series FF Preferred Stock, for one of the Company's founders.

On March 21, 2011, the Company issued 1,348,814 shares of Series C preferred stock at a price of \$7.41 per share. As part of this transaction, the Company incurred transaction costs of \$85, which were applied against the proceeds of this sale. The carrying value of the preferred stock will be accreted to its redemption price, including dividends.

On September 26, 2012, the Company issued 89,127 shares of Series D preferred stock at a price of \$16.83 per share. As part of this transaction, the Company incurred transaction costs of \$67, which were applied against the proceeds of this sale. The carrying value of the preferred stock will be accreted to its redemption price, including dividends.

Significant features of the common stock and Series A, Series B, Series C, Series D and FF preferred are as follows:

Voting

The holders of Series A, Series B, Series C, Series D and Series FF preferred stock shall be entitled to the number of votes equal to the number of shares of common stock into which such shares of preferred stock could be converted. The holders of common stock are entitled to vote equally based on their respective shares outstanding.

Dividends

Series A, Series B, Series C and Series D preferred stockholders are entitled to receive annual dividends of \$0.08 per share. Holders of Series FF preferred stock and common stock shall be entitled to dividends when and if declared by the Board of Directors and shall share dividends on a pro rata basis based on the number of shares of common stock that would be held by each upon conversion of all Series A, Series B, Series C, Series D and Series FF preferred stock into common stock.

Conversion

Each share of Series A, Series B, Series C and Series D preferred stock shall be convertible, at the option of the holder, at any time after the date of issuance and without payment of additional consideration, into such number of common shares as is determined by dividing the original issue price of \$1.00 by \$1.00 for Series A, by \$5.84 for Series B, by \$7.41 for Series C and by \$16.83 for Series D preferred stock holders. Each share of Series FF preferred stock shall be convertible, at the option of the holder, at any time after the date of issuance and without payment of additional consideration, into such number of common shares as is determined by

Appia, Inc. (in thousands except share data and per share data)

dividing the original issue price of \$0.001 by \$0.001. The conversion price is subject to adjustment for subdivisions, dividends, combinations, reclassifications, merger, sale, and other transactions. If the Company issues or sells additional shares of Common Stock less than the original issue price, that is deemed dilutive (Qualifying Dilutive Issuance), Series B and Series C are entitled to reduced conversion rates equal to the price the additional shares are issued at. The Series B and Series C Qualifying Dilutive issuance dates expired on April 9, 2012.

Automatic Conversion

Each share of Series A and Series B preferred stock shall automatically be converted into common stock at the then effective conversion price upon the completion of an underwritten public offering involving the sale of the Company's common stock at prices per share greater than three times the Original Issue price of Series C Preferred Stock and net cash proceeds of at least \$50 million. Each share of Series A and B preferred stock shall also automatically be converted into common stock at the then effective conversion price upon the written consent or agreement of two-thirds (½3) of the Series A and Series B preferred voting as a group.

Each share of Series C preferred stock shall automatically be converted into common stock at the then effective conversion price upon the completion of an underwritten public offering involving the sale of the Company's common stock net cash proceeds of at least \$50 million. Each share of Series C preferred stock shall also automatically be converted into common stock at the then effective conversion price upon the written consent or agreement of the majority of the Series C preferred holders voting as a group.

Each share of Series D preferred stock shall automatically be converted into common stock at the then effective conversion price upon the closing of a qualified public offering or upon the written consent or agreement of the majority shares voting together as a single group.

Appia, Inc. (in thousands except share data and per share data)

Liquidation Preference

Upon liquidation, dissolution, or winding up of the Company, holders of the Series A, Series B, Series C and Series D preferred shall be entitled to receive, prior and in preference to any distribution of the assets to holders of Series FF preferred or common stock, an amount equal to the original issue price of \$1.00 per share for Series A, \$5.84 per share for Series B, \$7.41 per share for Series C and \$16.83 for Series D, plus all accrued dividends and all declared and unpaid dividends on the Series A, Series B, Series C and Series D preferred stock. Any assets remaining after such preferential distribution shall be distributed ratably to holders of Series FF preferred stock and common stock on an as-if converted basis. The liquidation value per share, aggregate liquidation value and carrying value for the Series A, Series B, Series C and Series D preferred as of September 30, 2014 and 2013 are as follows:

	2014	2013
Series A Preferred	\$	 \$
Liquidation values per share	1.52	1.43
Aggregate liquidation value	4,932	4,672
Carrying value	4,932	4,672
Series B Preferred		
Liquidation values per share	7.84	7.37
Aggregate liquidation value	20,140	18,940
Carrying value	20,123	18,902
Series C Preferred		
Liquidation values per share	9.51	8.91
Aggregate liquidation value	12,824	12,024
Carrying value	12,796	11,981
Series D Preferred		
Liquidation values per share	19.54	18.20
Aggregate liquidation value	1,744	1,624
Carrying value	1,701	1,568

Redemption Preference

The certificate of incorporation also provides for the redemption of the Series A, Series B, Series C and Series D preferred upon agreement by two-thirds (2/3) of the outstanding Series A, Series B, Series C and Series D shares, voting as a single class of stock, beginning on December 31, 2015. The redemption price for the Series A, Series B, Series C and Series D preferred shall be the greater of the original issue price of \$1.00 per share for Series A, \$5.84 per share for Series B, \$7.41 per share for Series C, \$16.83 for Series D plus all accrued, declared, and unpaid dividends or the fair market value of Series A, Series B, Series C or Series D preferred stock on the date of the request.

7 Stock Options and Restricted Stock

During 2008, the Company adopted the 2008 Stock Option Plan (the Plan). A total of 1,942,247 shares of the Company's common stock have been reserved for issuance to employees,

Appia, Inc. (in thousands except share data and per share data)

directors and consultants. The terms of the stock option grants are determined by the Board of Directors. Since inception, all options were granted with an exercise price equal to the deemed fair value of the common stock on the date of grant. Under the Plan, options become vested over various periods, generally ranging up to four years, and expire not more than ten years after the date of grant.

In determining the compensation cost of options granted, as specified by ASC 718 – "Compensation – Stock Compensation," the fair value of each option or restricted stock grant has been estimated on the date of grant using the Black-Scholes option-pricing model. The assumptions used in these calculations are summarized as follows as of September 30, 2014 and December 31, 2013:

	2014	2013
Expected dividend yield		
Expected stock price volatility	39% - 40%	39% - 40%
Risk-free interest rate	2.42% - 2.71%	1.71% - 2.96%
Expected life of options	10	10
Estimated forfeiture rate for employee option	15%	15%

The Company calculates expected stock volatility using historical volatility for a peer group of companies, as the Company believes the expected volatility will approximate historical volatility of the peer group. The risk-free interest rate for the expected terms of the stock options is based on the rates paid on US Treasury instruments with similar lives at the time of the grant. The estimated forfeiture rate for employee options is based on historic forfeiture rates.

The following table summarizes the Company's stock option grant activity during the period from December 31, 2013 through September 30, 2014:

	Number of Shares	Weighted Average Exercise Price
		\$
Balance at December 31, 2013	675,466	0.26
Granted	42,850	0.28
Vested	(47,522)	0.25
Forfeited	(31,050)	0.27
Balance at September 30, 2014	639,744	0.26

Appia, Inc. (in thousands except share data and per share data)

The following table summarizes the stock options outstanding at September 30, 2014 and December 31, 2013:

	Exercise Price	Options Outstanding	Weighted Average Remaining Contractual Life of Options Outstanding	Options Exercisable	Weighted Average Grant Date Fair Value of Options Exercisable
					\$
	.10	41,200	5.40	39,844	0.07
	0.26	416,248	7.95	201,983	0.15
	0.28	218,018	9.65	-	
December 31, 2013		675,466		241,827	
	.10	39,000	4.63	39,000	0.07
	0.26	349,826	7.18	190,016	0.15
	0.28	250,918	8.88	50,180	0.15
September 30, 2014		639,744		279,196	

As of September 30, 2014 and December 31, 2013, there was \$36 and \$55, respectively of total unrecognized compensation cost related to non-vested options. The Company recorded \$13 of compensation cost for September 30, 2014 related to stock options recognized over a weighted average remaining life of 8.2 years.

The following table summarizes non-vested stock options during the period from December 31, 2013 through September 30, 2014:

	Number of Shares	Weighted Average Grant Date Fair Value
		\$
Non-vested common stock options outstanding, December 31, 2013	477,952	0.15
Granted	42,850	0.15
Vested	(135,704)	0.15
Forfeited	(24,550)	0.14
Non-vested common stock options outstanding, September 30, 2014	360,548	0.15

The Company at times issues Restricted Common Stock grants to employees, directors and consultants. The grants typically vest over a period of 4 years. The Company includes these costs in stock-based compensation and recognized on a straight-line basis. At September 30, 2014 and December 31, 2013 there was \$34 and \$63 of unrecognized compensation expense related to restricted stock. The Company recorded \$14 as compensation cost related to the restricted stock for the period ended September 30, 2014.

Appia, Inc. (in thousands except share data and per share data)

The following table summarizes the Company's restricted stock award activity during the period from December 31, 2013 through September 30, 2014:

	Number of
	Shares
Non-vested restricted stock outstanding at December 31, 2013	348,125
Granted	-
Vested	(112,813)
Forfeited	(30,000)
Non-vested restricted stock outstanding at September 30, 2014	205,312

8 Employee Benefit Plan

The Company's eligible employees participate in a 401(k) plan. Participants may contribute a percentage of their salary, limited by IRS regulations. The Plan allows the Company to match a percentage of the amount contributed by the employee. In addition, the Company may make discretionary profit-sharing contributions to the plan. The Company did not elect to make any contributions during the period ended September 30, 2014 or 2013.

9 Commitments

The Company leases its office facilities and certain office equipment under operating lease agreements. Non-cancelable leases may contain escalation clauses which require rent to be recognized on a straight-line basis over the term of the lease agreement. Expense related to operating leases, recognized on the straight-line basis, was approximately \$332 and \$317 for the period ended September 30, 2014 and 2013, respectively.

Future minimum lease payments under non-cancelable operating leases at September 30, 2014 are as follows:

	Amount

2014	78
2014 2015	235
Thereafter	-
	<u>31</u> 3

10 Subsequent Events

Management has evaluated subsequent events through December 2, 2014, the date of the financial statements were available for issuance. All matters requiring recognition or disclosure have been included in the consolidated financial statements.

On November 13, 2014 the Company has signed a definitive agreement to sell all of its outstanding shares to a Mandalay Digital Group Inc. Both companies still must obtain shareholder approval before the transaction can close. Mandalay Digital will issue at closing a number of shares and stock options to purchase its common stock equivalent to \$100 million less Appia's net debt and transaction expenses (subject to working capital adjustments), at a

Appia, Inc. (in thousands except share data and per share data)

agreed-upon value for Mandalay Digital common stock of \$4.50 per common share. Based on an estimate of Appia's net debt at closing, including transaction fees and expenses, Appia stockholders would receive in the aggregate approximately 19 million shares, representing pro forma ownership of the combined company of 33%.

Appendix A

AGREEMENT AND PLAN OF MERGER

by and among

MANDALAY DIGITAL GROUP, INC.,

DTM MERGER SUB, INC.,

APPIA, INC.

and

SHAREHOLDER REPRESENTATIVE SERVICES LLC,

as Stockholder Representative

Dated as of November 13, 2014

TABLE OF CONTENTS

		Page
Article I. T	THE MERGER	A-4
1.1.	The Merger	A-4
1.2.	Effective Time	A-5
1.3.	Effects of the Merger	A-5
	Exchange Agent	A-5
	Subsequent Actions	A-7
	Conversion of Securities	A-7
1.7.	Treatment of Series C Warrants	A-9
	Treatment of Company Options and Company Restricted Stock	A-9
	Appraisal Rights	A-11
	Closing Consideration Schedule	A-12
	Escrow of Consideration	A-12
	Working Capital/Net Debt Adjustment	A-13
	Company Designees	A-15
Article II	CLOSING	A-15
	The Closing Deliveries at Closing	A-15 A-15
	-	
Article III.	REPRESENTATIONS AND WARRANTIES OF THE COMPANY	A-17
	Organization of the Company	A-17
	Subsidiaries	A-17
	Capitalization	A-18
	Authorization	A-20
	Consents and Approvals	A-21
	Absence of Certain Changes or Events	A-21
	Facilities	A-23
	Condition and Sufficiency of Assets	A-24
	Contracts and Commitments	A-25
3.10.		A-27
	No Conflict or Violation	A-27
	Financial Statements and Other Financial Information	A-27
	Undisclosed Liabilities	A-28
	Books and Records	A-28
	Litigation	A-28
	Labor Matters	A-28
	Compliance with Law	A-30
	Intellectual Property	A-30
	Employee Benefit Plans	A-35
	Transactions with Certain Persons	A-37
	Certain Payments	A-37
	Tax Matters	A-37
	Insurance	A-40
	Publishers	A-40
	Advertisers	A-41
	Accounts Receivable	A-41
	Compliance with Environmental Laws	A-41
3.28.	Banking Relationships	A-41

		Page
3.29.	No Brokers	A-41
3.30.	Disclosure Documents	A-41
3.31.	No Other Agreements to Sell the Assets or Capital Stock of the Company	A-42
3.32.	Vote Required	A-42
Article I	V. REPRESENTATIONS AND WARRANTIES OF PARENT AND MERGER SUB	A-42
4.1.	Organization of Parent and Merger Sub	A-42
4.2.	Authorization	A-42
4.3.	No Conflict or Violation	A-43
4.4.	Parent Common Stock	A-43
4.5.	No Brokers	A-44
4.6.	SEC Reports and Financial Statements	A-44
4.7.	Disclosure Documents	A-44
4.8.	Parent Capitalization	A-44
4.9.	Litigation	A-45
	Merger Sub	A-45
	Absence of Certain Changes	A-45
	Compliance with Laws	A-45
4.13.	Intellectual Property	A-45
Article V	V. COVENANTS	A-45
5.1.	Further Assurances	A-45
5.2.	Audits	A-47
5.3.	No Solicitation	A-47
5.4.	Notification of Certain Matters	A-48
5.5.	Investigation	A-48
5.6.	Conduct of Business	A-48
5.7.	Tax Matters	A-50
5.8.	Employment Matters	A-52
5.9.	Preparation of S-4 and Stockholder Meetings	A-53
5.10.	Listing	A-55
5.11.	Public Announcements	A-55
5.12.	Consents; Notices	A-55
5.13.	Termination of Certain Agreements	A-55
5.14.	Director and Officers Indemnification	A-55
5.15.	Financing	A-56
	Tax Representation Letters	A-56
5.17.	Transfer Restrictions with respect to Parent Common Stock	A-57
5.18.	Documents and Information	A-58
5.19.	Other Agreements of Parent	A-58
Article V	I. CONDITIONS TO CLOSING	A-58
6.1.	Conditions to Obligations of the Company	A-58
6.2.	Conditions to Obligations of Parent and Merger Sub	A-59
ARTICL	E VII. INDEMNIFICATION; REMEDIES	A-61
7.1.	Survival of Representations, Etc	A-61
7.2.	Indemnification	A-62
7.3.	Escrow Account	A-69
7.4.	Exclusive Remedy	A-70

	Page
ARTICLE VIII. TERMINATION	A-70
8.1. Termination	A-70
8.2. Effect of Termination	A-71
8.3. Termination Amount	A-71
ARTICLE IX. MISCELLANEOUS	A-72
9.1. Defined Terms	A-72
9.2. Notices	A-89
9.3. Rules of Construction	A-90
9.4. Titles	A-90
9.5. Entire Agreement	A-90
9.6. Assignment	A-91
9.7. Amendment or Modification	A-91
9.8. Waiver	A-91
9.9. Severability	A-91
9.10. References	A-91
9.11. Burden and Benefit	A-92
9.12. Governing Law	A-92
9.13. Consent to Jurisdiction	A-92
9.14. Waiver of Trial by Jury	A-93
9.15. Legal Fees	A-93
9.16. Arbitration	A-93
9.17. Specific Performance	A-94
9.18. Cumulative Remedies	A-94
9.19. Expenses	A-94
9.20. Representation by Counsel	A-94
9.21. Execution and Counterparts	A-94
9.22. Acknowledgement of Representation	A-94
Exhibit A Stockholders Signing the Support Agreement	
Exhibit B Support Agreement	
Exhibit C Form of Certificate of Merger	
Exhibit D Form of Secured Guaranties	
Exhibit E Form of Financing Agreement	
Exhibit F Form of Escrow Agreement	
Exhibit G Form of Latham & Watkins LLP Reorg Opinion	
Exhibit H Form of Goodwin Procter LLP Reorg Opinion	
Exhibit I Non-Competition Agreement	
Exhibit J Charter Amendment	

AGREEMENT AND PLAN OF MERGER

This AGREEMENT AND PLAN OF MERGER (this "<u>Agreement</u>"), dated as of November 13, 2014, is by and among: (i) Mandalay Digital Group, Inc., a Delaware corporation ("<u>Parent</u>"); (ii) DTM Merger Sub, Inc., a Delaware corporation and a wholly-owned subsidiary of Parent ("<u>Merger Sub</u>"); (iii) Appia, Inc., a Delaware corporation (the "<u>Company</u>") and (iv) Shareholder Representative Services LLC, a Colorado limited liability company solely in its capacity as the representative of the Company Equityholders, as Stockholder Representative (the "<u>Stockholder Representative</u>").

RECITALS

WHEREAS, each of Parent, Merger Sub and the Company desire to effect the acquisition of the Company by Parent through the merger of Merger Sub with and into the Company (the "Merger");

WHEREAS, the parties intend that immediately following the Merger, the Company shall be the surviving corporation of the Merger, all pursuant to the terms and subject to the conditions hereinafter set forth and in accordance with the General Corporation Law of the State of Delaware (the "DGCL");

WHEREAS, the boards of directors of Parent, Merger Sub and the Company have determined that the Merger is advisable and in the best interests of their respective companies and stockholders, and have approved this Agreement and the Merger provided for herein upon the terms and subject to the conditions hereinafter set forth;

WHEREAS, for U.S. federal income tax purposes, it is intended that the Merger shall qualify as a "reorganization" within the meaning of Section 368(a) of the Code, and this Agreement shall constitute a "plan of reorganization" within the meaning of United States Treasury Regulations Sections 1.368-2(g) and 1.368-3(a) for purposes of Sections 368, 354 and 361 of the Code;

WHEREAS, contemporaneously with the execution and delivery of this Agreement, and as a condition and inducement to Parent's willingness to enter into this Agreement, each Company Stockholder listed on Exhibit A is entering into a Support Agreement with Parent in the form of Exhibit B (the "Support Agreement"); and

WHEREAS, Parent, Merger Sub and the Company desire to make certain representations, warranties, covenants and agreements in connection with the Merger.

AGREEMENT

NOW THEREFORE, in consideration of the respective covenants and promises contained herein and for other good and valuable consideration, the receipt and adequacy of which are hereby acknowledged, the parties hereto agree as follows:

Article I. THE MERGER

1.1. <u>The Merger</u>. Pursuant to the terms and subject to the conditions of this Agreement, at the Effective Time (as defined below), the Company and Merger Sub shall consummate the Merger in accordance with the DGCL pursuant to which (a) Merger Sub shall be merged with and into the Company and the separate corporate existence of Merger Sub shall thereupon cease; (b) the Company shall be the successor or surviving corporation

in the Merger and shall continue to be governed by the laws of the State of Delaware; (c) the separate corporate existence of the Company with all its rights, privileges, immunities, powers and franchises shall continue unaffected by the Merger; and (d) the Company shall succeed to and assume all the rights and obligations of Merger Sub. The corporation surviving the Merger is sometimes hereinafter referred to as the "Surviving Corporation." The Merger shall have the effects set forth in the applicable provisions of the DGCL.

- 1.2. <u>Effective Time</u>. Concurrently with the Closing on the Closing Date, the parties shall file a Certificate of Merger in the form attached hereto as <u>Exhibit C</u> (the "<u>Certificate of Merger</u>") with the Secretary of State of the State of Delaware in accordance with the relevant provisions of the DGCL. The Merger shall become effective upon the filing of the Certificate of Merger or at such later time as is agreed to by the parties hereto and specified in the Certificate of Merger (the time at which the Merger becomes effective is herein referred to as the "<u>Effective Time</u>").
 - 1.3. Effects of the Merger. At the Effective Time, and without any further action on the part of the Company or Merger Sub:
- (a) the certificate of incorporation of the Company, as in effect immediately prior to the Effective Time, shall be amended and restated in the Merger so as to be identical in all respects to the certificate of incorporation of Merger Sub as in effect immediately prior to the Effective Time, except that the name of the Surviving Corporation shall be Digital Turbine Media, Inc., and, as so amended and restated, such certificate of incorporation shall be the certificate of incorporation until thereafter amended as provided therein or by applicable law;
- (b) the bylaws of the Company, as in effect immediately prior to the Effective Time, shall be amended and restated in the Merger so as to be identical in all respects to the bylaws of Merger Sub as in effect immediately prior to the Effective Time, except that the name of the Surviving Corporation shall be Digital Turbine Media, Inc., and, as so amended and restated, such bylaws shall be the bylaws of the Surviving Corporation until thereafter amended as provided therein or by applicable law;
- (c) the directors and officers of Merger Sub immediately prior to the Effective Time shall, from and after the Effective Time, be the directors and officers of the Surviving Corporation, until their respective successors shall have been duly elected, designated or qualified, or until their earlier death, resignation or removal in accordance with the Surviving Corporation's certificate of incorporation and bylaws; and
 - (d) the Merger shall, from and after the Effective Time, have all of the effects provided by the DGCL and applicable law.

1.4. Exchange Agent.

- (a) As soon as practicable following the date of this Agreement, Parent shall appoint American Stock Transfer & Trust Company, LLC to act as exchange agent (the "Exchange Agent") for the payment and delivery of the payment of the applicable consideration as provided in Section 1.6.
- (b) <u>Share Transfer Books</u>. At the Effective Time, the share transfer books of the Company shall be closed, and thereafter there shall be no further registration of transfers of shares of Company Capital Stock. From and after the Effective Time, Persons who held shares of Company Capital Stock shall cease to have rights with respect to such shares, except as otherwise provided for herein.

(c) Exchange Procedures.

(i) As promptly as practicable following the Effective Time (but in no event later than three (3) Business Days thereafter), the Surviving Corporation shall cause the Exchange Agent to mail) to each holder of record of any shares of Company Capital Stock (A) a letter of transmittal (a "<u>Letter of Transmittal</u>") in the

form reasonably satisfactory to the Company, Parent and the Exchange Agent (which shall be based on the Exchange Agent's standard form and will include additional representations, warranties, releases and covenants consistent with those contained in the Support Agreement), which shall specify that delivery shall be effected, and risk of loss and title to the certificates representing such Capital Stock (each a "Certificate") shall pass, only upon proper delivery of such certificates (or affidavits of loss in lieu thereof as provided in Section 1.4.(f) below) to the Exchange Agent, and (B) instructions for use in effecting the surrender of such Certificates in exchange for the aggregate number of shares of Parent Common Stock issued in book-entry form (rounded down to the nearest whole share of Parent Common Stock after aggregating all shares of Parent Common Stock which such Company Stockholder is entitled to receive hereunder) to be issued in respect of such shares of Company Capital Stock pursuant to Section 1.6 hereof and as set forth on the Closing Consideration Schedule.

- (ii) Upon surrender of a Certificate (or affidavit of loss in lieu thereof) to the Exchange Agent, together with a Letter of Transmittal duly completed and validly executed in accordance with the instructions thereto, and such other documents as may reasonably be required by the Exchange Agent, the Exchange Agent shall issue to the holder of such Certificate that number of uncertificated shares of Parent Common Stock that such Holder is entitled to receive pursuant to Section 1.6 and as set forth on the Closing Consideration Schedule. Prior to accepting any Certificates, the Exchange Agent may require compliance with such reasonable terms and conditions as the Exchange Agent may impose to effect an orderly exchange thereof in accordance with customary exchange practices. Until surrendered as contemplated by this Section 1.4.(c) each Certificate shall be deemed, at any time after the Effective Time, to represent only the right to receive, upon such surrender, the applicable consideration as contemplated by this Article I subject to compliance with the terms of this Agreement. No interest shall be paid or accrued for the benefit of holders of the Certificates on the applicable consideration payable upon the surrender of the Certificates hereunder.
- (iii) In the event of a transfer of ownership of shares of Company Capital Stock that is not registered in the transfer records of the Company, it shall be a condition of payment that any Certificate surrendered in accordance with the procedures set forth in this Section 1.4.(c) shall be properly endorsed or shall be otherwise in proper form for transfer and that the Person requesting such payment shall have paid any transfer Taxes and other Taxes required by reason of the payment of the applicable consideration pursuant to Section 1.6 and as set forth on the Closing Consideration Schedule to a Person other than the registered holder of the Certificate and shall have established to the reasonable satisfaction of Parent that such Tax either has been paid or is not applicable.
- (d) No Further Ownership Rights in Company Capital Stock. At and after the Effective Time, each holder of issued and outstanding Company Capital Stock immediately prior to the Effective Time shall cease to have any rights as a holder of securities of the Company, except for the right of a Company Stockholder to surrender its, his or her Stock Certificate(s) in exchange for the consideration payable in respect of such Company Capital Stock hereunder and such holder's share, if any, of any distributions of amounts held in escrow pursuant to Section 1.11 hereof, or in the case of a Dissenting Stockholder, to perfect its, his or her right to receive payment (if any) for Dissenting Shares pursuant to the DGCL, if applicable. After the Effective Time, there shall be no further registration of transfers on the stock transfer books of the Surviving Corporation of the Company Capital Stock outstanding immediately prior to the Effective Time. If, after the Effective Time, Stock Certificates are presented to the Surviving Corporation for any reason, they shall be cancelled and exchanged as provided in (and subject to compliance with the provisions of) this Article I, except as otherwise provided by Law.
 - (e) No Fractional Shares. No fractional shares of Parent Common Stock shall be issued in connection with the Merger.
- (f) <u>Lost, Stolen or Damaged Stock Certificates</u>. In the event that any Stock Certificate shall have been lost, stolen or destroyed, upon the making of an affidavit of that fact by the Person claiming such Stock Certificate to be lost, stolen or destroyed and, if reasonably required by Parent or the Exchange Agent, the delivery by such Person of an indemnification agreement in such form and containing such substance acceptable

to Parent and the Exchange Agent and, if required by the Exchange Agent, the posting of a bond in such amount as the Exchange Agent may determine reasonably necessary as indemnity against any claim that may be made with respect to such Certificate, the Exchange Agent will pay in exchange for such lost, stolen or destroyed Stock Certificate the consideration to be paid with respect thereto, subject to the terms and conditions in this Article I.

- (g) Withholding Taxes. Parent, the Surviving Corporation, the Exchange Agent and the Escrow Agent shall be entitled to deduct and withhold (or cause to be deducted and withheld) from the consideration otherwise payable pursuant to this Agreement (including consideration held in escrow pursuant to Section 1.11) to any Company Equityholder or other Person such amounts as may be required to be deducted and withheld with respect to the making of such payment under the Code or under any provision of state, local or foreign Tax law. To the extent that amounts are so deducted and withheld and paid over to the appropriate Tax Authority, such deducted and withheld amounts shall be treated for all purposes of this Agreement as having been paid to the Person in respect of which such deduction and withholding was made. For the avoidance of doubt, nothing in this Section 1.4(g) shall limit the Company's ability and obligation to withhold any Taxes in connection with the exercise of vested Company Options by a holder at or prior to the Closing, including by requiring such holder to pay the Company cash in an amount equal to any applicable withholding Taxes.
- 1.5. <u>Subsequent Actions</u>. If at any time after the Effective Time the Surviving Corporation shall determine, in its sole discretion, or shall be advised, that any deeds, bills of sale, instruments of conveyance, assignments, assurances or any other actions or things are necessary or desirable to vest, perfect or confirm of record or otherwise in the Surviving Corporation its right, title or interest in, to or under any of the rights, properties or assets of the Company acquired or to be acquired by the Surviving Corporation as a result of, or in connection with, the Merger or otherwise to carry out this Agreement, then the officers and directors of the Surviving Corporation shall be authorized to execute and deliver, in the name and on behalf of the Company, all such deeds, bills of sale, instruments of conveyance, assignments and assurances and to take and do, in the name and on behalf of the Company or otherwise, all such other actions and things as may be necessary or desirable to vest, perfect or confirm any and all right, title or interest in, to and under such rights, properties or assets in the Surviving Corporation or otherwise to carry out this Agreement.

1.6. Conversion of Securities.

- (a) Effect on Company Preferred Stock. At the Effective Time, by virtue of the Merger and without any action on the part of Merger Sub, the Company or the holders of shares of Company Capital Stock, each share of Series A Preferred Stock, Series B Preferred Stock, Series C Preferred Stock, Series D Preferred Stock, and Series FF Preferred Stock (collectively, the "Company Preferred Stock") issued and outstanding immediately prior to the Effective Time will be cancelled and extinguished and will be converted automatically into the right to receive upon surrender of the certificate representing such shares of Company Preferred Stock in the manner provided in Section 1.4 hereof, upon the terms and subject to the conditions set forth in this Section 1.6 and throughout this Agreement, including the escrow provisions set forth in Section 1.11 and Article VII hereof, the consideration set forth below without interest:
- (i) each outstanding share of Series A Preferred Stock (other than any Dissenting Shares and Treasury Shares) will be converted automatically into the right to receive the Per Share Common Consideration;
- (ii) each outstanding share of Series B Preferred Stock (other than any Dissenting Shares and Treasury Shares) will be converted automatically into the right to receive the Per Share Series B Preferred Consideration;
- (iii) each outstanding share of Series C Preferred Stock (other than any Dissenting Shares and Treasury Shares) will be converted automatically into the right to receive the Per Share Series C Preferred Consideration;

- (iv) each outstanding share of Series D Preferred Stock (other than any Dissenting Shares and Treasury Shares) will be converted automatically into the right to receive the Per Share Series D Preferred Consideration; and
- (v) each outstanding share of Series FF Preferred Stock (other than any Dissenting Shares and Treasury Shares) will be converted automatically into the right to receive the Per Share Common Consideration.

(b) Effect on Company Common Stock.

- (i) At the Effective Time, by virtue of the Merger and without any action on the part of Parent, Merger Sub, the Company or the holders of shares of Company Capital Stock, each share of Company Common Stock issued and outstanding immediately prior to the Effective Time (other than any (A) shares of Company Restricted Stock (which shall be treated in accordance with Section 1.8.(c) below), (B) Dissenting Shares and (C) Treasury Shares) will be cancelled and will be converted automatically into the right to receive upon surrender of the applicable Certificate representing such shares of Company Common Stock in the manner provided in Section 1.4 hereof, upon the terms and subject to the conditions set forth in this Section 1.6 and throughout this Agreement, including the escrow provisions set forth in Section 1.11 and Article VII hereof, the Per Share Common Consideration.
- (c) Notwithstanding the foregoing, the consideration payable to each holder of shares of Company Capital Stock as of the Effective Time pursuant to this <u>Section 1.6</u> shall be subject to adjustment as provided herein.
- (d) The parties acknowledge and agree that the purpose and intent of this <u>Section 1.6</u> is to allocate the consideration payable hereunder in accordance with the terms of the Company's Restated Charter.
- (e) <u>Cancellation of Certain Shares</u>. Each issued and outstanding share of Company Capital Stock that is held in treasury of the Company or owned by the Company, Parent or Merger Sub (collectively, "<u>Treasury Shares</u>") shall automatically be cancelled and retired and shall cease to exist, and no consideration shall be delivered in exchange therefor.
- (f) <u>Merger Sub Common Stock</u>. Each issued and outstanding share of the common stock, par value \$0.001 per share, of Merger Sub shall be converted into and become one fully paid and nonassessable share of common stock of the Surviving Corporation.
- (g) The Per Share Common Consideration, the Price Per Share and the shares of Parent Common Stock issuable to the holders of Common Stock (other than holders of Company Restricted Stock) or Company Preferred Stock pursuant to Section 1.6 and the holders of Company Options and Company Restricted Stock pursuant to Section 1.8 and to the holders of Series C Warrants pursuant to Section 1.7 shall be adjusted to reflect fully the appropriate effect of any stock split, split-up, reverse stock split, dividend (whether stock or cash) or distribution of securities convertible into Company Common Stock, Company Preferred Stock, or Parent Common Stock, reorganization, recapitalization, reclassification or other like change with respect to the Company Common Stock, Company Preferred Stock or Parent Common Stock having a record date occurring on or after the date of this Agreement and prior to the Effective Time.
- (h) In no event will the number of shares of Parent Common Stock (taking into account the number of shares of Parent Common Stock underlying the Company Options being assumed) being issued in exchange for and upon conversion of all outstanding shares of Company Capital Stock (including Company Restricted Stock), all Company Options and all Series C Warrants be greater than the Parent Common Stock Closing Consideration (except, if applicable, as a result of adjustments made pursuant to Section 1.6.(g).

1.7. Treatment of Series C Warrants.

- (a) Each Series C Penny Warrant issued and outstanding immediately prior to the Effective Time will terminate and be cancelled and be converted automatically into the right to receive an amount equal to the Series C Penny Warrants Consideration applicable to such Series C Penny Warrant.
- (b) Each Series C Other Warrant issued and outstanding immediately prior to the Effective Time will terminate and be cancelled and be converted automatically into the right to receive an amount equal to the Series C Other Warrants Consideration applicable to such Series C Other Warrant.
- (c) Prior to the Effective Time, the Company shall take all necessary actions, including obtaining appropriate resolutions of the Company's board of directors, providing all notices, obtaining any necessary consents and taking all other actions necessary or desirable to effect the termination of the Series C Warrants as provided in this Section 1.10, which notices, resolutions, consents and other written materials shall be subject to the review and approval of Parent (which approval shall not be unreasonably withheld).
- (d) At or after the Effective Time, each Series C Warrantholder shall surrender and deliver to Exchange Agent (A) Series C Warrant(s) owned by such Series C Warrantholder, (B) a duly completed and executed Letter of Transmittal and (C) such other documents and information as may reasonably be required by Parent or the Exchange Agent, in exchange for the aggregate number of shares of Parent Common Stock issued in book-entry form (rounded to the nearest whole share of Parent Common Stock after aggregating all shares of Parent Common Stock which such Series C Warrantholder is entitled to receive hereunder) to be issued in respect of such Series C Warrant pursuant to Section 1.6.(a) hereof and as set forth on the Closing Consideration Schedule. No interest will be paid or accrued upon surrender of any Series C Warrants.
- (e) If the consideration payable to Series C Warrantholders is to be paid to a Person other than the Person in whose name the relevant Series C Warrant surrendered in exchange therefor is registered, it shall be a condition to the payment of such amounts that the Series C Warrant so surrendered shall be properly endorsed and otherwise be in proper form for transfer, that such transfer otherwise be proper, that the Person requesting such transfer pay to the Surviving Corporation, any transfer or other Taxes payable by reason of the foregoing or establish to the satisfaction of the Surviving Corporation that such Taxes have been paid or are not required to be paid and that the transferee duly execute and deliver to Parent a Letter of Transmittal.

1.8. Treatment of Company Options and Company Restricted Stock.

- (a) Each share of Company Restricted Stock and each Company Option, in each case, held by a non-employee director of the Company and outstanding as of the date hereof, shall vest in full and, as applicable, become exercisable, immediately following the Effective Time.
- (b) At the Effective Time, by virtue of the Merger and without any action on the part of Parent, the Company or the holders of Company Options, each Company Option (or any portion thereof) that is outstanding at the Effective Time (whether vested or unvested and, after giving effect to any accelerated vesting provision in effect on the date hereof, including pursuant to Section 1.8(a))) and that is held by a holder who is a Continuing Service Provider will be assumed by Parent. Each Company Option assumed by Parent will continue to be subject to the same terms, conditions and restrictions applicable to such option immediately prior to the Effective Time, including without limitation applicable vesting and exercise restrictions (with vesting based, in the case of time-based vesting restrictions, on continued services to the Company, the Surviving Corporation, Parent or Parent's Subsidiaries through the applicable vesting date), except for administrative changes that are not adverse to the holder of the Company Option or to which the holder expressly consents, and except that (i) each Company Option will, following the Effective Time, cover (in lieu of shares of Company Stock) a number of shares of Parent Common Stock equal to the product of the number of Company shares subject to the outstanding Company Option immediately prior to the Effective Time, multiplied by the Option Exchange Ratio, rounded

down to the nearest whole number of shares of Parent Common Stock, (ii) the per share exercise price for the Parent Common Stock issuable upon exercise of such assumed Company Option will be equal to the quotient determined by dividing the per share exercise price for such Company Option outstanding immediately prior to the Effective Time by the Option Exchange Ratio, rounded up to the nearest whole cent, and (iii) all references to the "Company" in the Company Equity Plan and the stock option agreements will instead be references to Parent. It is the intention of the parties that, (i) to the extent any Company Option so assumed by Parent qualified as an incentive stock option as defined in Section 422 of the Code prior to the Effective Time, such Company Option shall qualify following the Effective Time as an incentive stock option to the extent permitted under Section 422 of the Code, and (ii) the assumption of the Company Options by Parent shall comply with Sections 409A and 424 of the Code, and this Section 1.8 shall be construed in a manner consistent with such intent.

(c) By virtue of the Merger and without any action on the part of Parent, the Company or the holders of Company Options, each outstanding, unexercised Company Option (or any portion thereof) that is vested and exercisable (after giving effect to any accelerated vesting provisions in effect on the date hereof, including pursuant to Section 1.8.(a)) at the Effective Time and that is held by a holder who is not a Continuing Service Provider (each, a "Deemed Exercise Option") shall, in each case, be cancelled at the Effective Time in exchange for the right to receive, subject to satisfaction of the withholding obligations contemplated by Section 1.4(g), on or as soon as practicable after the Effective Time, that number of shares of Parent Common Stock equal to the quotient obtained by dividing (i) the product of (x) the excess, if any, of the Per Share Common Consideration Equivalent Amount over the applicable exercise price of such Deemed Exercise Option, times (y) the number of Company shares subject to the outstanding, unexercised Deemed Exercise Option immediately prior to the Effective Time, by (ii) the Closing Price Per Share, rounded down to the nearest whole share. For the avoidance of doubt, if the applicable exercise price of a Deemed Exercise Option exceeds the Per Share Common Consideration Equivalent Amount, no payment shall be made in respect of such Deemed Exercise Option, and the holder thereof shall have no further rights or interests therein from and after the Effective Time. By virtue of the Merger and without any action on the part of Parent, the Company or the holders of Company Options, each outstanding, unexercised Company Option (or any portion thereof) that is unvested and unexercisable (after giving effect to any accelerated vesting provisions in effect on the date hereof, including pursuant to Section 1.8.(a)) at the Effective Time and/or that is held by a holder who is not a Continuing Service Provider shall, in each case, be cancelled at the Effective Time for no consideration. In no event shall the Company Options described in this Section 1.8(c) be assumed by Parent. Notwithstanding anything contained herein to the contrary, with respect to any Tax withholding on the consideration payable to any such holders of a Deemed Exercise Option, Parent may require, as a condition to any payment or share delivery hereunder, that any such holders remit to the Surviving Corporation a check payable to the Surviving Corporation having a fair market value equal to the sums required to be withheld by federal, state, local and/or foreign Tax law.

(d) At the Effective Time, by virtue of the Merger and without any action on the part of Parent, the Company or the holders of Company Restricted Stock, each award of shares of Company Restricted Stock (or portion thereof) that is outstanding and unvested at the Effective Time (after giving effect to any accelerated vesting provisions in effect on the date hereof, including pursuant to Section 1.8.(a)) (each, a "Restricted Stock Award") will be assumed by Parent. Each Restricted Stock Award so assumed by Parent will continue to be subject to the same terms, conditions and restrictions applicable to such Restricted Stock Award immediately prior to the Effective Time, including without limitation applicable transfer and vesting restrictions (with vesting based, in the case of time-based vesting restrictions, on continued services to the Company, the Surviving Corporation, Parent or Parent's Subsidiaries through the applicable vesting date), except for administrative changes that are not materially adverse to the holder of the Restricted Stock Award or to which the holder consents, and except that (i) each Restricted Stock Award so assumed will, following the Effective Time, cover (in lieu of shares of Company stock) a number of shares of Parent Common Stock equal to the product of the number of Company shares subject to the outstanding Restricted Stock Award immediately prior to the Effective Time, multiplied by the Per Share Common Consideration, rounded down to the nearest whole number of shares of Parent Common Stock, (ii) a number of shares of Parent Common Stock subject to the assumed Restricted Stock Award equal to the applicable Pro Rata Allocation of the Escrowed Shares (determined in accordance with

the escrow provisions set forth in <u>Section 1.11</u> and <u>Article VII</u> hereof) will be held in the Escrowed Account, and (iii) all references to the "Company" in the Company Equity Plan and the restricted stock agreements will instead be references to Parent.

- (e) Following the Effective Time, Parent shall prepare and file with the SEC a registration statement on Form S-8 with respect to the Parent Common Stock subject to Company Options assumed by Parent pursuant to Section 1.8(b) hereof and shall use its reasonable best efforts to have such registration statement declared effective as soon as reasonably practicable at the Effective Time and no later than ten (10) days following the Closing Date. Following the Effective Time, no Person shall have any right or interest in any Company Option or Restricted Stock Award except as expressly provided herein.
- (f) Prior to the Effective Time, the Company shall take all necessary actions, including obtaining appropriate resolutions of the Company's board of directors, providing all notices, obtaining any necessary consents and taking all other actions necessary or desirable to effect the termination of the Company Equity Plan and the Company Options and Company Restricted Stock as provided in this Section 1.8 hereof, which notices, resolutions, consents and other written materials shall be subject to the review and approval of Parent. In addition, prior to the Effective Time, if requested by Parent, the Company shall provide notice to each holder of Company Options and/or Company Restricted Stock of the anticipated treatment of such awards in the Merger, which notice shall be subject to the advance review and consent of Parent (which consent shall not be unreasonably withheld).

1.9. Appraisal Rights.

- (a) Notwithstanding anything in this Agreement to the contrary, any shares of Company Capital Stock held by a holder who has properly demanded and not effectively withdrawn or lost such holder's appraisal rights for such shares under the DGCL or other similar rights (if any) under applicable law (collectively, the "<u>Dissenting Shares</u>"), shall not be converted into or represent a right to receive a portion of the consideration as set forth in <u>Section 1.6</u> hereof, but the holder thereof shall only be entitled to such rights as are provided by the DGCL or other applicable law (if any).
- (b) Notwithstanding the provisions of Section 1.9.(a) hereof, if any holder of Dissenting Shares shall effectively withdraw or lose (through failure to perfect or otherwise) such holder's appraisal rights with respect to such shares under the DGCL (and/or any other similar rights under other applicable law (if any such other rights have been purportedly invoked)), then, as of the later of the Effective Time and the occurrence of such event, such shares shall automatically be converted into and represent only the right to receive the consideration set forth in and subject to the provisions of this Agreement, upon surrender of the certificate(s) formerly representing such shares.
- (c) Prior to the Closing Date, the Company shall give Parent (i) prompt notice of any written demand for appraisal received by the Company pursuant to the applicable provisions of the DGCL (and of any similar demand purportedly made under other applicable law) and (ii) the opportunity to participate in all negotiations and proceedings with respect to such demands. Prior to the Closing Date, the Company shall not, except with the prior written consent of Parent, make any payment with respect to any such demands or offer to settle or settle any such demands. Prior to the Closing Date, any communication to be made by the Company to any Company Stockholder with respect to such demands shall be submitted to Parent in advance and shall not be presented to any Company Stockholder prior to the Company receiving Parent's consent. Notwithstanding the foregoing, to the extent that Parent, the Surviving Corporation or the Company (i) makes any payment or payments in respect of any Dissenting Shares in excess of the consideration that otherwise would have been payable in respect of such shares in accordance with this Agreement or (ii) incurs any losses (including reasonable attorneys' and consultants' fees, costs and expenses and including any such reasonable fees, costs and expenses incurred in connection with investigating, defending against or settling any action or proceeding) in respect of any Dissenting Shares ((i) and (ii) together, "Excess Dissenting Share Payments"), Parent shall be entitled to recover the amount of such Excess Dissenting Share Payments in accordance with the terms of Article VII hereof.

- (d) Notwithstanding any provision of <u>Article I</u> or <u>Article VII</u> hereof to the contrary, in the event that there are any Dissenting Shares, then (i) any payment by Parent pursuant to <u>Section 1.11</u> hereof that would have been required to have been paid to such holder of Dissenting Shares (each a "<u>Dissenting Stockholder</u>") (assuming, for this purpose, that such Dissenting Stockholder had not exercised appraisal rights) shall be retained by Parent and (ii) such Dissenting Stockholder shall not be considered a Company Equityholder for purposes of a distribution of any portion of the Escrowed Shares.
- 1.10. <u>Closing Consideration Schedule</u>. No later than five (5) Business Days prior to the Closing Date, the Company shall deliver to Parent a schedule (the "Closing Consideration Schedule") in a form acceptable to Parent, which schedule shall be certified as complete and correct by the Company's chief executive officer and chief financial officer and which shall accurately set forth, as of the Closing:
- (i) all Company Stockholders (except for Company Stockholders holding no shares other than Restricted Stock Awards) and their respective addresses, the number and type of shares of Company Capital Stock held by such Company Stockholders (including the respective certificate numbers and the class or series of Company Capital Stock held by such Company Stockholder), the Pro Rata Portion and the Pro Rata Allocation of the Escrowed Shares applicable to each Company Stockholder, the number of shares of Parent Common Stock to be issued to each Company Stockholder, and such other information relevant thereto which Parent may reasonably request;
- (ii) all Series C Warrantholders entitled to receive consideration pursuant to <u>Section 1.7</u> hereof, the number of shares of Parent Common Stock to be issued to each Series C Warrantholder (the aggregate number of shares of Parent Common Stock to be issued to the Series C Warrantholders in respect of the Series C Warrants pursuant to <u>Section 1.7</u>, the "<u>Warrant Consideration</u>");
- (iii) all Company Optionholders entitled to receive consideration pursuant to <u>Section 1.8(b)</u> hereof, the number of Parent Options to be issued for the Company Options, the vesting schedule of the Company Options, the number of shares subject to the Company Options that are unvested and outstanding, the number of shares subject to the Company Options that are unvested and outstanding, and the consideration to be paid, as applicable, to each Company Optionholder (the aggregate shares of Parent Common Stock subject to Parent Options to be issued to the Company Optionholders in respect of Company Options pursuant to <u>Section 1.8(b)</u>, the "<u>Assumed Option Consideration</u>");
- (iv) all holders of Deemed Exercise Options entitled to receive consideration pursuant to Section 1.8(c) hereof, and their respective addresses and social security numbers or tax identification numbers, if applicable, the number of shares of Company Common Stock subject to each Deemed Exercise Option, the number of shares of Parent Common Stock to be issued in respect of each Deemed Exercise Option, and such other information relevant thereto which Parent may reasonably request (the aggregate shares of Parent Common Stock to be issued in respect of Deemed Exercise Option pursuant to Section 1.8(c), the "Deemed Exercise Option Consideration"); and
- (v) all holders of Restricted Stock Awards entitled to receive consideration pursuant to Section 1.8 hereof, and their respective addresses and social security numbers or tax identification numbers, if applicable, the number of shares of Company Common Stock subject to each Restricted Stock Award, the vesting schedule of the Restricted Stock Awards, the Pro Rata Portion and Pro Rata Allocation of the Escrowed Shares applicable in respect of each Restricted Stock Award, the number of shares of Parent Common Stock to be issued in respect of each Restricted Stock Award, and such other information relevant thereto which Parent may reasonably request (the aggregate shares of Parent Common Stock to be issued in respect of Restricted Stock Awards pursuant to Section 1.8, the "Restricted Stock Consideration").
- 1.11. <u>Escrow of Consideration</u>. In order to at least partially satisfy, and to establish a procedure for the satisfaction of, claims by Parent for payment by the Company Equityholders of any post-Closing purchase price adjustment as set forth in <u>Section 1.12</u> hereof and Claims by the Parent Indemnitees for indemnification pursuant

to <u>Article VII</u> hereof, Parent, the Stockholder Representative and the Escrow Agent shall enter into the Escrow Agreement on the Closing Date, pursuant to which Parent shall deposit with the Escrow Agent One Million (1,000,000) shares of Parent Common Stock (the "<u>Escrowed Shares</u>"). Parent shall be deemed to have contributed (on behalf of each Company Equityholder who holds Participating Escrow Stock) each such Company Equityholder's Pro Rata Allocation of the Escrowed Shares as set forth on the Closing Consideration Schedule (and the consideration payable to each such Company Equityholder pursuant to this <u>Article I</u> shall be reduced by such amounts). Pursuant to the terms and subject to the conditions of the Escrow Agreement, the Escrow Agent shall establish an escrow account into which the Escrow Agent shall deposit the Escrowed Shares (the "<u>Escrow Account</u>").

The timing and methodology for the release of the Escrowed Shares shall be governed by the terms and subject to the conditions set forth in this Agreement and the Escrow Agreement; provided, however, that each of Parent and the Stockholder Representative agrees that it will act in good faith and cooperate with one another to execute and deliver such joint written instructions, including with respect to any distributions of the Escrowed Shares, to the Escrow Agent as are required to implement the intent of this Agreement and the Escrow Agreement. Subject to Section 1.9.(d) hereof and the terms of the Escrow Agreement, in the event of a distribution of any amounts from the Escrow Account to the Company Equityholders (who hold Participating Escrow Stock), each such Company Equityholder shall be entitled to receive a portion of such distribution equal to such Company Equityholder's Pro Rata Allocation of such distribution. For purposes of satisfying any claims for indemnification and/or any adjustment pursuant to Section 1.12 under this Agreement, all shares of Parent Common Stock in the Escrow Account shall be deemed to have a value equal to the Price Per Share, as adjusted for any stock splits, dividends, combinations or the like, irrespective of the actual value of such shares at the time they are distributed from the Escrow Account. The Company Equityholders (who hold Participating Escrow Stock) shall be treated as the owner of the Escrow Account for tax purposes. The Escrow Agreement shall provide that the Escrowed Shares will be entitled to receive all dividends declared and payable with respect to Parent Company Stock, any such dividends shall be added to the Escrowed Account and may be used to satisfy claims for indemnification. The Company Equityholders (who hold Participating Escrow Stock) shall be entitled to vote all Escrowed Shares on their own behalf.

1.12. Working Capital/Net Debt Adjustment.

(a) Pre-Closing Adjustment.

(i) <u>Pre-Closing Statement</u>. No later than five (5) Business Days before the Closing Date, the Company shall prepare and deliver to Parent an unaudited statement of estimated Working Capital as of the Closing Date (the "<u>Pre-Closing Statement</u>"), which shall be prepared by the Company in accordance with GAAP as applied in the preparation of the Financial Statements and shall fairly present in all material respects the estimated Working Capital (the "<u>Estimated Working Capital</u>") and estimated Net Debt (the "<u>Estimated Net Debt</u>") of the Company as of the Closing Date. The Company shall consult with Parent and its accountants with respect to the preparation of the Pre-Closing Statement, and the Pre-Closing Statement shall be in form and substance reasonably satisfactory to Parent. The Pre-Closing Statement shall be accompanied by reasonably detailed schedules indicating a calculation of the Estimated Working Capital and Estimated Net Debt as of the Closing Date and of the Adjustment Amount.

(ii) Adjustment Amount. As used herein, the "Adjustment Amount" shall mean an amount, which may be a positive or a negative number, equal to (A) the Estimated Working Capital as set forth in the Pre-Closing Statement minus the Working Capital Target, minus (B) the Estimated Net Debt as set forth in the Pre-Closing Statement. For clarity, no dollar amount shall be counted twice for purposes of determining the Adjustment Amount.

(b) Post-Closing Adjustment.

(i) <u>Closing Statement</u>. No later than 135 calendar days following the Closing Date, Parent shall prepare and deliver to the Stockholder Representative an unaudited statement of Working Capital as of the

Closing Date (the "Closing Statement"), which shall be prepared by Parent's personnel in accordance with GAAP as applied in the preparation of the Financial Statements and shall fairly present in all material respects the Working Capital (the "Closing Working Capital") and Net Debt ("Closing Net Debt") of the Company as of the Closing Date. The Closing Statement shall be accompanied by reasonably detailed schedules indicating a calculation of the Closing Working Capital and Closing Net Debt as of the Closing Date and of the Final Adjustment Amount.

(ii) <u>Final Adjustment Amount</u>. As used herein, the "Final Adjustment Amount" shall mean an amount, which may be a positive or a negative number, equal to (A)(I) the Closing Working Capital as set forth in the Closing Statement minus (II) the Estimated Working Capital as set forth in the Pre-Closing Statement, minus (B)(I) the Closing Net Debt as set forth in the Closing Statement minus (II) the Estimated Net Debt as set forth in the Pre-Closing Statement. For clarity, no dollar amount shall be counted twice for purposes of determining the Final Adjustment Amount.

(c) Distribution of the Final Adjustment Amount,

- (i) If the Final Adjustment Amount is a positive number, then Parent shall, within five Business Days of the determination of the Final Adjustment Amount, issue to each Company Equityholder, its Pro Rata Portion of such amount by check or by wire transfer of immediately available funds to the Exchange Agent for further distribution to the Company Equityholders.
- (ii) If the Final Adjustment Amount is a negative number, then Parent and the Stockholder Representative shall, within five Business Days of the determination of the Final Adjustment Amount, execute and deliver joint written instructions to the Escrow Agent, instructing the Escrow Agent to distribute to Parent from the Escrow Account a number of shares of Parent Common Stock equal to the Final Adjustment Amount divided by the Price Per Share; provided, that if the Final Adjustment Amount is a negative number and the absolute value of the Final Adjustment Amount exceeds the Stock Escrow Amount, Parent shall be entitled to recover, following distribution of the entire Stock Escrow Amount, from the Company Equityholder's Pro Rata Portion of such excess amount.
- (d) <u>Disputed Final Adjustment Amount</u>. If the Stockholder Representative disagrees with the Final Adjustment Amount as set forth in the Closing Statement, the Stockholder Representative shall notify Parent of such disagreement in writing specifying in reasonable detail the particulars of such disagreement within thirty (30) calendar days after the Stockholder Representative's receipt of the Closing Statement. If the Stockholder Representative fails to provide such notification within such period, then the Final Adjustment Amount as set forth in the Closing Statement shall be final and binding on the parties.
- (e) Resolution of Disputed Final Adjustment Amount. Parent and the Stockholder Representative shall use their reasonable efforts for a period of 30 calendar days after the Stockholder Representative's delivery of such notice (or such longer period as Parent and the Stockholder Representative shall mutually agree upon) to resolve any disagreements raised by the Stockholder Representative with respect to the calculation of the Final Adjustment Amount. If Parent and the Stockholder Representative resolve such disagreements, then the Final Adjustment Amount, as agreed between Parent and the Stockholder Representative, shall be the "Final Adjustment Amount" for purposes of this Agreement. If, at the end of such period, Parent and the Stockholder Representative are unable to resolve such disagreements, Parent and the Stockholder Representative shall jointly select and retain an independent auditor of recognized national standing to resolve any remaining disagreements; provided that Pricewaterhouse Coopers, L.L.P will be the independent auditor if Parent and the Stockholder Representative cannot agree on the selection of such independent auditor. The engagement agreement for the independent auditor must specify that the independent auditor will act as a neutral arbiter and not as a fiduciary to or advocate of either Parent or the Stockholder Representative. The independent auditor will consider only those items and amounts that Parent and the Stockholder Representative are unable to resolve. In making its determination, the independent auditor shall not assign any value with respect to a disputed amount that is greater

than the highest value for such amount claimed by either Parent or the Stockholder Representative or that is less than the lowest value for such amount claimed by either Parent or the Stockholder Representative. The determination by such independent auditor shall be final, binding and conclusive on the parties. If the independent auditor is retained pursuant to this Section 1.12.(e), the Final Adjustment Amount as finally determined by the independent auditor shall be the "Final Adjustment Amount" for purposes of this Agreement. Parent and the Stockholder Representative shall use their reasonable efforts to cause the independent auditor to make its determination within 30 calendar days of accepting its selection. The fees and expenses of such independent auditor shall be borne by Parent and the Stockholder Representative (solely on behalf of the Company Equityholders) in proportion to the aggregate amount of all disputed items as to which such party's claim was unsuccessful (i.e., if there are \$1,000,000 of disputed items and the independent auditor determines that the Stockholder Representative's claim prevails with respect to \$250,000 of such disputed items, then the Stockholder Representative (solely on behalf of the Company Equityholders) would be obligated to pay 75% of the fees and expenses and Parent would be obligated to pay 25% of the fees and expenses).

1.13. Company Designees. The parties shall cooperate to take all actions reasonably necessary so that as of immediately following the Effective Time (a) the number of directors that will comprise all of the members of the board of directors of Parent is increased by two (2) and (b) two (2) individuals currently serving on the Company Board and designated by the Company within five (5) calendar days after the date hereof (provided such designees are reasonably satisfactory to Parent's board of directors, including any applicable committee thereof)) (the "Company Designees") are appointed to Parent's board of directors. The Company Designees shall be appointed for a term of service until concluding at the next annual meeting of the stockholders of Parent to occur after the Effective Time and shall serve until their respective successors are duly elected. Each Company Designees shall provide the same type of information the Company customarily requests of any other director appointee, including the Company's standard Directors and Officer Questionnaire and (in the case of non-employee directors) shall be elgibible to receive compensation pursuant to the Company's non-employee director compensation plans, in effect, from time to time. It is understood that a condition of appointment of a Company Designee is that such Company Designees shall qualify to be a director, all in accordance with the organizational documents of Parent and at least one of the Company Designees shall qualify as "independent" under the rules of the Nasdaq Stock Market. The Company and Parent agree that in the event that any of the Company Designees does not meet the requirements of a Company Designee or is unable or otherwise fails to serve, for any reason, as a director of Parent at the Effective Time, the Company shall have the right to designate another individual to serve as a director of Parent's board of directors in place of such Company Designee; provided that such other individual to be appointed to Parent's board of directors immediately following the Effective Time shall be reasonably satisfactory to Parent's board of directors (including any applicable committee thereof) and shall meet the requirements for a Company Designee listed above.

Article II. CLOSING

2.1. The Closing. The closing of the transactions contemplated herein (the "Closing") shall take place at the offices of Latham & Watkins LLP, 355 South Grand Avenue, Los Angeles, California, at 9:00 a.m. on the fifth Business Day after all of the conditions set forth in Article VI (other than those conditions that by their terms are required to be satisfied or waived at the Closing, but subject to the satisfaction or waiver of such conditions) shall have been satisfied or waived by the party entitled to the benefit of the same or at such other time and date as shall be agreed upon by the parties. The day on which the Closing occurs shall be the "Closing Date".

2.2. Deliveries at Closing.

(a) <u>Deliveries by Parent and Merger Sub</u>. Upon the terms and subject to the conditions set forth in this Agreement, in reliance on the representations, warranties and agreements of the Company contained herein,

Parent and Merger Sub shall deliver (or cause to be delivered) to the Company, at the Closing on the Closing Date, the following (reasonably satisfactory in form and substance to the Company and its legal counsel):

- (i) a certificate executed by the Secretary or an Assistant Secretary of each of Parent and Merger Sub certifying as of the Closing Date (A) a true and complete copy of the certificate of incorporation of Parent and the certificate of incorporation of Merger Sub, each certified as of a recent date by the Secretary of State of the State of Delaware, (B) a true and complete copy of the bylaws of each of Parent and Merger Sub; (C) a true and complete copy of the resolutions of the board of directors of Parent and Merger Sub authorizing the execution, delivery and performance of this Agreement, the Ancillary Agreements and the consummation of the transactions contemplated hereby and thereby and (D) incumbency matters;
- (ii) a certificate of good standing and/or subsistence of Parent and Merger Sub, dated as of a recent date prior to the Closing, issued by the Secretary of State of the State of Delaware;
 - (iii) the Escrow Agreement, executed by Parent;
- (iv) The Company Designees have been appointed as directors of Parent and Parent has delivered evidence of such effective appointment;
- (v) a certificate executed by an executive officer of Parent and Merger Sub certifying as to the matters set forth in Section 6.1.(a) hereof as of the Closing Date;
- (vi) secured guarantees by Parent, in form and substance attached hereto as Exhibit D (with such changes as Parent may reasonably request) of the refinancing agreements referred to in clause (b)(viii) of this Section 2.2; and
- (vii) such other documents and instruments as in the opinion of counsel for the Company may be reasonably required to effectuate the terms of this Agreement and to comply with the terms hereof.
- (b) <u>Deliveries by the Company</u>. Upon the terms and subject to the conditions of this Agreement, in reliance upon the representations, warranties and agreements of Parent and Merger Sub contained herein and in consideration of the consideration to be paid to the Company Equityholders, the Company shall deliver (or cause to be delivered) to Parent and Merger Sub, at the Closing on the Closing Date, the following (reasonably satisfactory in form and substance to Parent and Merger Sub and their legal counsel):
- (i) a certificate executed by the Secretary or an Assistant Secretary of the Company certifying as of the Closing Date (A) a true and complete copy of the certificate of incorporation of the Company, as amended through the Closing Date, certified as of a recent date by the Secretary of State of the State of Delaware, (B) a true and complete copy of the bylaws of the Company, (C) a true and complete copy of the resolutions of the board of directors of the Company authorizing the execution, delivery and performance of this Agreement, the Ancillary Agreements and the consummation of the transactions contemplated hereby and thereby and (D) incumbency matters;
- (ii) a certificate of good standing and/or subsistence of the Company, dated as of a recent date prior to the Closing, issued by the Secretary of State of the State of Delaware and all states in which the Company is qualified to do business;
- (iii) a certificate executed by an executive officer of the Company certifying as to the matters set forth in Section 6.2.(a) hereof as of the Closing Date;
- (iv) a certificate by the Company that meets the requirements of United States Treasury Regulations Sections 1.897-2(h) and 1.1445-2(c)(3), establishing that the Company is not a "United States real property holding corporation" within the meaning of Section 897(c)(2) of the Code, and has not been such a

United States real property holding corporation within the 5-year period ending on the Closing Date, along with written authorization for Parent to deliver the notice form described in Treasury Regulations Section 1.897-2(h)(2) to the IRS on behalf of the Company upon the Closing of the Merger;

- (v) the Escrow Agreement, executed by the Stockholder Representative;
- (vi) resignations from the Persons holding the position of an officer or director of the Company, in office immediately prior to the Effective Time, resigning from such positions effective as of the Effective Time;
 - (vii) evidence of the termination of the Terminated Agreements;
- (viii) financing agreements, each in form and substance as attached hereto as Exhibit E (with such changes as Parent may agree to), from each Existing Lender, each of which will amend and restate the Existing Credit Arrangements;
- (ix) a cancellation agreement from each Series C Warrantholder with respect to the cancellation of such Series C Warrantholder's Series C Warrant Shares pursuant to <u>Section 1.7</u> hereof; and
- (x) such other documents and instruments as in the opinion of counsel for Parent and Merger Sub may be reasonably required to effectuate the terms of this Agreement and to comply with the terms hereof.

Article III. REPRESENTATIONS AND WARRANTIES OF THE COMPANY

The Company hereby represents and warrants to Parent and Merger Sub as follows, except as otherwise set forth on the Company's Disclosure Schedules, which representations and warranties are, as of the date hereof, and will be, as of the Closing Date, true and correct:

3.1. Organization of the Company. The Company is a corporation duly organized, validly existing and in good standing under the laws of the State of Delaware with full corporate power and authority to conduct its business as it is presently being conducted, to own and or use the properties and assets that it purports to own or use, and to perform all its obligations under its Contracts. The Company is duly qualified or licensed to do business as a foreign corporation and is in good standing in each jurisdiction listed on Schedule 3.1 of the Disclosure Schedules, being all the jurisdictions in which the character of its properties owned or leased or the nature of its activities make such qualification necessary, except where the failure to be so qualified or in good standing would not have a Material Adverse Effect. Copies of the Organizational Documents of the Company, and all amendments thereto, heretofore delivered or made available to Parent are true, accurate and complete as of the date hereof.

3.2. Subsidiaries.

(a) The Company does not have any direct or indirect Subsidiaries other than PocketGear Deutschland GmbH. ("<u>PocketGear</u>"), which is wholly-owned by the Company. PocketGear does not have any operational activities and does not have any liabilities. The Company has no other direct or indirect stock or other equity or ownership interest (whether controlling or not) in any corporation, association, partnership, limited liability company, joint venture or other entity. PocketGear is a German limited liability company duly organized, validly existing and in good standing under the laws of Germany, with full power and authority to conduct its business as it is presently being conducted, to own or use the properties and assets that it purports to own or use, and to perform all of its material obligations under its Material Contracts. PocketGear is duly qualified to do business as a foreign corporation and is in good standing in each jurisdiction listed on Schedule 3.2.(a) of the Disclosure Schedules, being all the jurisdictions in which the character of its properties owned or leased or the nature of its

activities make such qualification necessary, except where the failure to be so qualified or in good standing would not have a Material Adverse Effect. Copies of the Organizational Documents of PocketGear, and all amendments thereto, heretofore made available to Parent are true, accurate and complete as of the date hereof.

- (b) Schedule 3.2.(b) of the Disclosure Schedules sets forth a description of all the issued and outstanding equity interests of PocketGear. The Company owns of record and beneficially all of the issued and outstanding equity interests of PocketGear free and clear of all Encumbrances. All of the outstanding shares of capital stock of PocketGear are duly authorized, validly issued, fully paid and non-assessable, and none of such shares are subject to any preemptive right, right of first refusal, right of first offer or right of rescission. There are no accrued unpaid dividends on any shares of capital or other stock of PocketGear. No claim has been made in writing or otherwise or, to the Knowledge of Company, threatened in writing to PocketGear asserting that any Person is the holder or beneficial owner of, or has the right to acquire beneficial ownership of any securities (including options and warrants) of, or any other voting, equity or ownership interest in PocketGear.
- (c) Except as set forth in Schedule 3.2.(b) of the Disclosure Schedules, (i) there are no shares of capital or other stock of PocketGear authorized, issued or outstanding, (ii) there are no existing compensatory equity or equity-linked awards of any form, including, without limitation, any options, stock appreciation rights, stock units, warrants, calls, restricted shares, phantom shares, deferred shares, performance shares or similar rights, preemptive or similar rights, awards, convertible securities, bonds, debentures, notes or other indebtedness having general voting rights or debt convertible into securities having such rights ("Voting Debt") or subscriptions or other rights, agreements, arrangements or commitments of any character relating to the issued or unissued shares of capital or other stock of PocketGear obligating PocketGear to issue, transfer or sell or cause to be issued, transferred, sold or repurchased any shares of capital stock convertible into or exchangeable for such shares of capital or other stock, or obligating PocketGear to grant, extend or enter into any such option, warrant, call subscription or other right, agreement, arrangement or commitment and (iii) there are no outstanding contractual obligations of PocketGear to repurchase, redeem or otherwise acquire any share of capital or other stock of PocketGear or to provide funds to make any investment (in the form of a loan, capital contribution or otherwise) in any other Person.
- (d) PocketGear has not violated any Regulations in connection with the offer, sale or issuance of any of the capital stock or any other securities of PocketGear, and such capital stock and other securities have been issued and granted in compliance with all requirements set forth in PocketGear's Organizational Documents.
- (e) Schedule 3.2.(e) of the Disclosure Schedules sets forth all outstanding Long Term Liabilities of PocketGear, and for each Long Term Liability set forth thereon, identifies the debtor, the principal amount, the creditor, the maturity date, and the collateral, if any, securing such Long Term Liability. All letters of credit, fidelity bonds and surety bonds are in full force and effect and will continue in full force and effect immediately following the consummation of the transactions contemplated by this Agreement and the Ancillary Agreements. No Default exists with respect to the obligations of PocketGear under any such letters of credit, fidelity bonds or surety bonds and the Company has not received any notification of cancellation of any of such letters of credit, fidelity bonds or surety bonds.

3.3. <u>Capitalization</u>.

(a) The issued and outstanding capital stock of the Company, including the Company Options, the Series C Warrants, the shares of Company Restricted Stock and any securities convertible into capital stock of the Company, is held by the Persons and in the amounts set forth on Schedule 3.3.(a) of the Disclosure Schedules. The authorized capital stock of the Company consists of (i) 17,625,600 shares of Company Common Stock, (ii) 3,275,000 shares of Series A Preferred Stock, (iii) 2,569,169 shares of Series B Preferred Stock, (iv) 2,612,818 shares of Series C Preferred Stock, (v) 89,127 shares of Series D Preferred Stock, and (vi) 2,991,500 shares of Series FF Preferred Stock, and there are no other shares of other classes or series of capital stock of the Company authorized or outstanding. There are (A) (A) 3,750,205 shares of Company

Common Stock issued and outstanding (excluding any shares of Company Restricted Stock), (B) (B) 177,501 shares of Company Restricted Stock issued and outstanding, (C) 3,275,000 shares of Series A Preferred Stock issued and outstanding, (D) 2,569,169 shares of Series B Preferred Stock issued and outstanding, (E) 1,348,814 shares of Series C Preferred Stock issued and outstanding, (F) 89,127 shares of Series D Preferred Stock issued and outstanding, and (G) 2,991,500 shares of Series FF Preferred Stock issued and outstanding. There are 1,942,247 shares of Company Common Stock authorized for issuance under the Company Equity Plan (of which (1) 633,025 shares of Company Common Stock are subject to outstanding and unexercised Company Options (2) 177,501 shares of Company Common Stock are subject to outstanding Restricted Stock Awards (which are included in (B) above), (3) 175,429 shares of Company Common Stock have been issued pursuant to the exercise of stock options (which are included in (A) above), and (4) 14,625 shares of Company Common Stock have been issued pursuant to the exercise of stock options and exchanged for the same number of shares of Series FF Preferred Stock (which are included in (G) above)). As of the date hereof, the Company has issued Series C Warrants exercisable to purchase 601,600 shares of Series C Preferred Stock. All of the issued and outstanding shares of Company Capital Stock are, and all Company Capital Stock which may be issued pursuant to the exercise of outstanding Company Options, Series C Warrants or any other Company security, will be, when issued in accordance with the respective terms thereof, duly authorized, validly issued, fully paid and non-assessable. Except as set forth on Schedule 3.3.(a) of the Disclosure Schedules, none of such shares are subject to any preemptive right, right of first refusal, right of first offer, right of rescission or similar right. No claim has been made or threatened to the Company asserting that any Person other than a Person listed on Schedule 3.3.(a) or Schedule 3.3.(d) of the Disclosure Schedules is the holder or beneficial owner of, or has the right to acquire beneficial ownership of, any securities (including options and warrants) of, or any other voting, equity or ownership interest in the Company. Each grant of Company Options, Series C Warrants and Company Restricted Stock was duly authorized no later than the date on which the grant of such Company Option, Series C Warrant or Company Restricted Stock, as applicable, was by its terms to be effective by all necessary corporate action, including, as applicable, approval by the board of directors of the Company (or a duly authorized committee thereof) and any required Stockholder approval by the necessary number of votes or written consents, and all Company Options, Series C Warrants and Company Restricted Stock have been granted in compliance with applicable Regulations, including all applicable United States federal and state securities laws, and the Company Equity Plan. Each Company Option was granted with a per share exercise price that equaled or exceeded the fair market value of a share of Company Common Stock on the applicable date of grant, as determined by the Company in a manner consistent with the requirements of Section 409A of the Code.

(b) Except as set forth in Section 3.3.(a) hereof and Schedule 3.3.(a) or Schedule 3.3.(d) of the Disclosure Schedules, (i) there are no shares of capital or other stock or voting securities of the Company authorized, issued or outstanding, (ii) there is not Voting Debt or subscriptions or other rights, agreements, arrangements or commitments of any character, relating to the issued or unissued Company Capital Stock obligating the Company to issue, transfer or sell or cause to be issued, transferred, sold or repurchased any shares of capital stock of, or Voting Debt of, or other equity or voting interest in, the Company or securities convertible into or exchangeable for such shares or equity interests, or obligating the Company to grant, extend or enter into any such option or other compensatory equity or equity-linked award, warrant, call, subscription or other right, agreement, arrangement or commitment or pursuant to which any Person is or may be entitled to receive any payment or other value based on the value of the Company's equity interests, and (iii) there are no outstanding obligations of the Company to repurchase, redeem or otherwise acquire or retire for value any Company Capital Stock, or other capital or other stock of the Company or to provide funds to make any investment (in the form of a loan, capital contribution or otherwise) in any other Person.

(c) Except as set forth in Section 3.3.(a) or Schedule 3.3.(c) or Schedule 3.3.(d) of the Disclosure Schedules, there are no (i) securities of the Company reserved for issuance for any purpose, (ii) agreements pursuant to which registration rights in the securities of the Company have been granted, (iii) stockholder agreements between the Company and any current stockholders of the Company regarding the securities of the Company or (iv) agreements among stockholders of the Company with respect to the voting or transfer of the securities of the Company or with respect to the governance of or any other aspect of the Company's affairs

(including any rights of first refusal or similar restrictions on transfer). No distributions (including distributions for Taxes) are owed to the Company with respect to any Company Capital Stock or owed by the Company to any Company Stockholder.

- (d) <u>Schedule 3.3.(d)</u> of the Disclosure Schedules sets forth a true and complete list of each outstanding Company Option and each Restricted Stock Award and, as applicable, the name of the holder of each such Company Option or Restricted Stock Award, the state of residence of each such holder, the number of shares Company Common Stock originally granted under such award, the date of grant, the type of option (*i.e.*, tax-qualified incentive stock option or nonqualified stock option), whether such option permits early exercise, the exercise price per share of such Company Option, the vesting schedule (including any acceleration provisions), the number of shares vested/unvested, forfeited, and exercised with respect to each award and the applicable expiration date. An updated <u>Schedule 3.3.(d)</u> of the Disclosure Schedules accurate as of immediately prior to the Effective Time shall be delivered to Parent by the Company immediately prior to the Effective Time.
- (e) With respect to all outstanding Series C Warrants, <u>Schedule 3.3.(e)</u> of the Disclosure Schedules sets forth a true and complete list, as applicable, of the name of the holder of each such Series C Warrant, the number of shares of Series C Preferred Stock covered by such Series C Warrant, the date of grant, whether such warrant permits early exercise, the exercise price per share of such Series C Warrant, the vesting schedule and vested status of such Series C Warrant, if any, and the applicable expiration date.
- (f) The Company is in compliance in all material respects with all Regulations, including applicable federal or state securities laws, in connection with the offer, sale or issuance of any Company Capital Stock, Company Options, Series C Warrants or any other securities, and such Company Capital Stock, Company Options, Series C Warrants and other securities have been issued and granted in compliance with all requirements set forth in the Company's Organizational Documents.
- (g) The Closing Consideration Schedule delivered to Parent pursuant to Section 1.9.(a) hereof shall accurately reflect, as of immediately prior to the Effective Time, all Company Equityholders and their respective addresses and social security numbers or tax identification numbers, if applicable, the number and type of shares of Company Capital Stock, Series C Warrants and/or Company Options held by such Company Equityholders (including the respective certificate numbers and the class or series of such Company Capital Stock held by such Company Equityholders), the Pro Rata Allocation of the Escrowed Shares applicable to each Company Equityholder, the Aggregate Consideration to be paid to each Company Equityholders and such other information relevant thereto which Parent may reasonably request.
- (h) Schedule 3.3.(h) of the Disclosure Schedules sets forth all outstanding Long Term Liabilities of the Company, and for each Long Term Liability set forth thereon, identifies the debtor, the principal amount, the creditor, the maturity date, and the collateral, if any, securing such Long Term Liability. All letters of credit, fidelity bonds and surety bonds are in full force and effect and will continue in full force and effect immediately following the consummation of the transactions contemplated by this Agreement and the Ancillary Agreements. No Default exists with respect to the obligations of the Company under any such letters of credit, fidelity bonds or surety bonds and the Company has not received any notification of cancellation of any of such letters of credit, fidelity bonds or surety bonds.
- 3.4. <u>Authorization</u>. The Company has all requisite corporate power and authority to execute and deliver this Agreement and the Ancillary Agreements to be executed and delivered by the Company pursuant hereto, to consummate the transactions contemplated hereby and thereby and to perform its obligations hereunder and thereunder. Other than to the Company Stockholder Approval, the execution and delivery by the Company of this Agreement and the Ancillary Agreements to which the Company is to be a party, and the consummation by the Company of the transactions contemplated hereby and thereby have been duly authorized by all necessary corporate action and no other proceedings or actions on the part of the Company are necessary to authorize this Agreement

and the Ancillary Agreements to which the Company is to be a party and the transactions contemplated hereby and thereby. This Agreement has been duly executed and delivered by the Company and is, and upon execution and delivery of the Ancillary Agreements to which the Company is to be a party, each of such Ancillary Agreements will be, legal, valid and binding obligations of the Company enforceable against the Company in accordance with their terms, in each case, except as such enforceability may be limited by (a) bankruptcy, insolvency, moratorium, reorganization and other similar laws affecting creditors' rights generally and (b) the general principles of equity, regardless of whether asserted in a proceeding in equity or at law.

- 3.5. Consents and Approvals. Except as may be required by the HSR Act or as set forth on Schedule 3.5 of the Disclosure Schedules, (a) no notice to, declaration, filing or registration with, or authorization, consent or approval of, or Permit from, any Person (including under any Contract), and (b) no consent under any Contract from any other Person, is, in each case, required to be made or obtained by the Company, PocketGear, or any other Affiliate of the Company in connection with the execution, delivery and performance of this Agreement and the Ancillary Agreements to which the Company is a party and the consummation of the transactions contemplated hereby and thereby by the Company or PocketGear, except for (i) obtaining the Company Stockholder Approval, (ii) the filing of the Certificate of Merger with the Delaware Secretary of State and appropriate corresponding documents with the appropriate authorities of other states in which the Company is qualified as a foreign corporation to transact business, (iii) any filings required to be made with the SEC in connection with this Agreement and the transactions contemplated hereto, (iv) such consents, approvals, orders, authorizations, registrations, declarations, notices and filings as may be required under applicable state securities Laws, the rules and regulations of the NASDAQ Capital Market, and (v) consents and notices with respect to Contracts that generated less than \$100,000 in the twelve months prior to the Interim Balance Sheet Date or which are expected to generate less than \$100,000 in the twelve (12) months following the Interim Balance Sheet Date and where the failure to obtain the consent or give the notice under the Contract would not result in the creation of any material liability or materially adversely effect the operation of the Business.
- 3.6. <u>Absence of Certain Changes or Events</u>. Except as set forth on <u>Schedule 3.6</u> of the Disclosure Schedules, since December 31, 2013 (the "Balance Sheet Date"), there has not been any:
- (a) change in the Company's or PocketGear's authorized or issued capital stock; grant of any stock option or right to purchase shares of capital stock of the Company or PocketGear; issuance of, or commitment to issue, any shares of capital stock or any security convertible into or exchangeable for such capital stock; grant of any registration rights; purchase, redemption, retirement or other acquisition by the Company or PocketGear of any shares of any such capital stock; any repricing or other change in exercise price to any outstanding Company Options; or declaration or payment of any dividend or other distribution or payment in respect of shares of capital stock;
 - (b) amendment to any Organizational Documents of the Company or PocketGear;
- (c) actual or threatened adverse change in the financial condition, working capital, shareholders' equity, assets, Liabilities, reserves, revenues, income or earnings of the Business, the Company or PocketGear;
- (d) adoption of or change in any Tax or other accounting methods, principles or practices or change in any annual Tax accounting period; making or revocation of or change in any Tax election; settlement or compromise of any claim, notice, audit report or assessment in respect of Taxes; filing of any amended Tax Return; entry into any Tax allocation agreement, Tax sharing agreement, Tax indemnity agreement or closing agreement relating to any Tax; knowingly surrender any right to claim a Tax refund; or consent to any extension or waiver of the statute of limitations period applicable to any Tax claim or assessment; in each case by or with respect to the Company or PocketGear;
- (e) revaluation of any of the assets of the Company, PocketGear or relating to the Business, including, without limitation, writing off notes or accounts receivable, except in the ordinary course of business;

- (f) failure to take reasonable actions to prevent any damage, destruction, abandonment or loss (whether or not covered by insurance) adversely affecting any of the assets of the Company, PocketGear or the Business;
- (g) acquisition of any corporation, partnership, limited liability company, other business organization or division thereof or any material amount of assets, or entry into any joint venture, strategic alliance, exclusive dealing, noncompetition or similar contract or arrangement;
- (h) except for the Merger, any adoption of a plan of complete or partial liquidation, dissolution, merger, consolidation, restructuring, recapitalization or other reorganization of the Company or of PocketGear, or other alteration of the Company's or PocketGear's corporate structure;
- (i) cancellation of any Indebtedness or waiver or release of any material right or claim of the Company, PocketGear or relating to the Business;
- (j) (i) hiring or termination of any key employee, consultant or director, (ii) payment, announcement, promise or grant, whether oral or in writing, any material increase in or establishment of (as applicable) any wages, base pay, fees, salaries, compensation, bonuses, incentives, deferred compensation, pensions, severance, termination or change in control payments, retirement, profit sharing, fringe benefits, equity or equity-linked awards, employee benefit plans, or any other form of compensation or benefits payable by the Company, including without limitation, any material increase or material change pursuant to any Company Plan (except as required by any applicable Regulation) other than in the ordinary course of business, or (iii) entry into, adoption, termination or material amendment of any Company Plan other than the setting of quarterly bonus objectives in the ordinary course of business consistent with past practice;
- (k) adverse change in employee relations which has or is reasonably likely to have a material adverse effect on the productivity, the financial condition or, results of operations of the Company or PocketGear;
- (l) entry into, or amendment, cancellation or termination of, any Contract, Lease or Permit to which the Company or PocketGear is a party or otherwise relating to the Business, including, without limitation, any employment, consulting, distribution, dealer, sales representative, joint venture, credit or similar agreement or any Contract or transaction involving a total remaining commitment by or to the Company or PocketGear of at least \$250,000, except in the ordinary course of business;
 - (m) entry into any Contract with any Related Party of the Company or PocketGear;
- (n) mortgage, pledge or other Encumbrance of any assets of the Company, PocketGear or relating to the Business, except purchase money mortgages arising in the ordinary course of business;
- (o) sale, assignment or transfer of any assets of the Company, PocketGear or relating to the Business, other than in the ordinary course of business;
- (p) transfer of the registration of any Business Domain Names or failure to timely renew the registration of any Business Domain Names;
- (q) incurrence of Indebtedness by the Company or PocketGear for borrowed money or commitment to borrow money entered into by the Company or PocketGear, made or agreed to be made by the Company or PocketGear, or indebtedness for borrowed money guaranteed by the Company or PocketGear;
- (r) incurrence by the Company or PocketGear of Liabilities, except Liabilities incurred in the ordinary course of business or under this Agreement or the incurrence of Transaction Expenses, or increase or change in any assumptions underlying or methods of calculating, any doubtful account contingency or other reserves of the Company or PocketGear;

- (s) payment, discharge or satisfaction of any Liabilities of the Company or PocketGear other than in the ordinary course of business;
- (t) any single capital expenditure relating to the Business in excess of \$100,000 or capital expenditures that are, in the aggregate, in excess of \$100,000 or incurrence of any obligations to make any capital expenditures or execute any Lease;
- (u) failure to pay or satisfy when due any material Liability of the Company, PocketGear or related to the Business, except in the ordinary course of business;
- (v) except in the ordinary course of business consistent with past practice, any acceleration in the collection of or discounting of any accounts receivable, delays in the payment of accounts payable or deferrals of expenses, reduction of inventories or otherwise increases of cash in hand:
- (w) any amendment to or otherwise taking of actions that would permit or cause any Company Option or Restricted Stock Award to accelerate in contemplation of or as a consequence of the Merger or other transactions contemplated by this Agreement;
- (x) failure of the Company or PocketGear to carry on diligently the Business in the ordinary course so as to keep available to Parent the services of the Company's and PocketGear's employees, and to preserve for Parent the Business and the goodwill of the Company's and PocketGear's suppliers, customers and advertisers of the Business and others having business relations with the Company, PocketGear or the Business;
- (y) failure to maintain any Source Code, data or other information related to the Business (including any records of images, art, photographs or similar content posted on and/or removed from the Company's or PocketGear's website prior to the Closing Date) other than in a manner consistent with past practice;
- (z) disposition, sale, transfer, assignment, Encumbrance, pledge, license (except in the ordinary course of business), abandoning, dedicating to the public, failure to maintain or permitting to lapse any Intellectual Property of the Company, PocketGear or relating to the Business, or any disposition or disclosure to any Person of any Intellectual Property of the Company, PocketGear or relating to the Business not theretofore a matter of public knowledge;
- (aa) existence of any other event or condition which in any one case or in the aggregate has or might reasonably be expected to have a Material Adverse Effect; and
- (bb) agreement, whether oral or written, by the Company or PocketGear to do any of the things described in the preceding clauses (a) through (z) other than as expressly provided for herein.

3.7. Facilities.

- (a) Owned Real Property. Neither the Company nor PocketGear owns any Owned Real Property.
- (b) Actions. There are no pending or, to the Knowledge of the Company, threatened condemnation proceedings or other Actions relating to any Facility.
- (c) <u>Leases or Other Agreements</u>. Except for Facility Leases listed on <u>Schedule 3.7</u> of the Disclosure Schedules, neither the Company nor PocketGear are a party to any leases, subleases, licenses, occupancy agreements, options, rights, concessions or other agreements or arrangements, written or oral, granting to any Person the right to purchase, use or occupy any Facility, or any real property or any portion thereof or interest in any such Facility or real property.

- (d) Facility Leases and Leased Real Property. With respect to each Facility Lease, the Company or PocketGear has an unencumbered interest in the Leasehold Estates, other than Encumbrances which do not materially interfere with, or otherwise affect the present use and enjoyment of the Facility subject thereto or affected thereby. The Company and PocketGear enjoy peaceful and undisturbed possession of all the Leased Real Property. With respect to such Leased Real Property, (i) each lease, license or interest to which the Company or PocketGear is a party is in full force and effect and enforceable against the Company or PocketGear and the counterparty thereto in accordance with its terms, subject to bankruptcy and insolvency laws. Except as described in Schedule 3.7.(d) of the Disclosure Schedules: (A) there are no leases, subleases, licenses, concessions or any other Contracts to which the Company or PocketGear is a party granting to any Person other than the Company or PocketGear any right to possession, use occupancy or enjoyment of any of the Leased Real Property or any portion thereof and (B) the Company and PocketGear are not obligated under or bound by any option, right of first refusal, purchase Contract or other Contract to sell or otherwise dispose of any Leased Real Property or any other interest in any Leased Real Property.
- (e) <u>Certificate of Occupancy</u>. To the Knowledge of the Company, all Facilities have received all required approvals of governmental authorities (including, without limitation, Permits and a certificate of occupancy or other similar certificate permitting lawful occupancy of the Facilities) required in connection with the operation thereof and have been operated and maintained in accordance with applicable Regulations.
- (f) <u>Utilities</u>. All Facilities are supplied with utilities (including, without limitation, water, sewage, disposal, electricity, gas and telephone) and other services necessary for the operation of such Facilities as currently operated, and there is no condition which would reasonably be expected to result in the termination of the present access from any Facility to such utility services.
- (g) No Special Assessment. Neither the Company nor PocketGear has received notice of any special assessment relating to any Facility or any portion thereof and there is no pending or threatened special assessment.
- 3.8. Condition and Sufficiency of Assets. The buildings, structures and equipment of the Company and PocketGear are structurally sound and are in good operating condition and repair, normal wear and tear excepted. The buildings, structures and equipment of the Company and PocketGear are adequate for the uses to which they are being put, and none of such buildings, structures or equipment is in need of maintenance or repairs except for ordinary, routine maintenance and repairs that are not material in nature or cost. The buildings, structures and equipment of the Company and PocketGear are sufficient for the continued conduct of the Business of the Company and PocketGear after the Closing in substantially the same manner as conducted prior to the Closing. The improvements constructed on the buildings, structures and equipment of the Company and PocketGear, including, without limitation, all Leasehold Improvements, and all Fixtures and Equipment and other Tangible Property owned, leased or used by the Company and PocketGear at the Facilities are (a) insured to the extent and in a manner customary in the industry, (b) structurally sound with no known material defects, (c) in good operating condition and repair, subject to ordinary wear and tear, fit for use in the ordinary course of business, (d) not in need of maintenance, repair or correction except for ordinary routine maintenance and repair, (e) sufficient for the operation of the Company and PocketGear as presently conducted and (f) in conformity, in all material respects, with all applicable Regulations. The Company and PocketGear own, or have a valid leasehold or other interest in, and after the Closing Date, the Company and PocketGear will continue to own, or have a valid leasehold or other interest in, free and clear of all Encumbrances, all assets necessary for the conduct of the Business as presently conducted by the Company and PocketGear and to permit Parent (through the Surviving Corporation) to continue to conduct the Business in substantially the same manner as the Business has been conducted through the date hereof.

3.9. Contracts and Commitments.

- (a) <u>Contracts</u>. <u>Schedule 3.9.(a)</u> of the Disclosure Schedules sets forth a complete and accurate list of all Contracts to which the Company or PocketGear is a party or by which the Company or PocketGear or their assets are bound (including oral agreements) of the following categories:
- (i) Contracts not made in the ordinary course of business that are not otherwise required to be disclosed pursuant to clauses (ii)-(xx) below;
- (ii) all agreements, programs, policies, arrangements or other Contracts for the employment or engagement of any officer, employee or Contingent Worker on a full time, part time, consulting or other basis (A) providing annual compensation (whether cash and/or otherwise) that may exceed \$150,000, (B) providing for the payment of any cash or other compensation or benefits upon a termination of service and/or upon the consummation of the transactions contemplated by this Agreement or (C) otherwise restricting the Company's or PocketGear's ability to terminate the employment or engagement of any employee or consultant at any time for any lawful reason or for no reason without penalty or Liability;
- (iii) collective bargaining agreements, work council agreement, work force agreement or other Contracts with any labor union applicable to the employees or Contingent Workers of the Company or PocketGear, or any severance or change in control agreements, programs, policies or arrangements;
- (iv) Contracts containing covenants (including confidentiality provisions if applicable) limiting the freedom of the Company or PocketGear (or any of their subsidiaries) to engage in any line of business, expand the geographical scope of business anywhere in the world, compete with any Person or that otherwise have the effect of restricting in any material respect the Company or PocketGear (or any of their subsidiaries) from the development, manufacture, marketing or distribution of products and/or services, including without limitation, non-competition, non-solicitation and standstill obligations, or any contract, agreement or arrangement containing any exclusivity, noncompetition or most favored pricing terms to which the Company or PocketGear is subject;
- (v) Contracts containing restrictive covenant agreements (including confidentiality, non-competition, and non-solicitation) between the Company or PocketGear and any employee or Contingent Worker of the Company or PocketGear;
- (vi) Contracts with any Company Equityholder, any Affiliate of any Company Equityholder or any other Affiliate of the Company or PocketGear ("Affiliate Agreements");
- (vii) Contracts relating to the manufacture, distribution and shipment of the Products to the Company's or PocketGear's customers other than Contracts with Advertisers and Publishers);
- (viii) the form of agreement that the Company has entered into with each Person who is an advertiser (each, an "Advertiser") or publisher (each, a "Publisher"), together with a list of any such agreements that the Company or PocketGear has entered into that (A) deviate from such form and (B) have generated or are reasonably expected to generate more than \$250,000 in revenue or expense in any consecutive twelve (12) month period during the twenty four month period beginning twelve (12) months prior to the Interim Balance Sheet Date (the "Advertiser and Publisher Agreements");
 - (ix) Intentionally omitted;
- (x) options with respect to any property, real or personal, whether the Company or PocketGear shall be the grantor or grantee thereunder;

- (xi) Contracts involving future expenditures or Liabilities that have generated or are reasonably expected to generate more than \$250,000 in expense during the twenty-four month period beginning twelve (12) months prior to the Interim Balance Sheet Date and ending twelve (12) months after the Interim Balance Sheet Date;
 - (xii) Intentionally omitted;
- (xiii) Contracts that have generated or are reasonably expected to generate more than \$250,000 in revenue in any consecutive twelve (12) month period during the twenty four month period beginning twelve (12) months prior to the Interim Balance Sheet Date and ending twelve (12) months after the Interim Balance Sheet Date;
- (xiv) Contracts or commitments relating to commission arrangements with any officer, individual employee, consultant or other Person:
- (xv) promissory notes, loans, indentures, evidences of Indebtedness, letters of credit, guarantees, or other instruments or agreements relating to an obligation to pay money, whether the Company or PocketGear shall be the borrower, lender or guarantor thereunder or whereby any assets of the Company or PocketGear are pledged (excluding credit provided by the Company or PocketGear in the ordinary course of business to purchasers of the Company's or PocketGear's services or products);
 - (xvi) any Contract with the United States, state or local government or any agency or department thereof;
 - (xvii) Leases of real property;
 - (xviii) Leases of personal property not cancelable (without Liability) within 30 calendar days;
- (xix) written warranties, guaranties and or other similar undertakings with respect to contractual performance extended by the Company or PocketGear;
- (xx) joint venture, partnership and other Contracts (however named) involving a sharing of profits, losses costs or Liabilities by the Company or PocketGear with any other Person; and
- (xxi) material Contracts relating to the development, ownership, use or licensing of any patents, trademarks, copyrights, or other Intellectual Property, including agreements with current or former employees, consultants, or contractors regarding the disposition, development, appropriation or the nondisclosure of any of the Intellectual Property of the Company or PocketGear other than (A) the Company's standard form of Non-Disclosure, Non-Solicitation and Invention Assignment Agreement, and (B) Advertiser and Publisher Agreements entered into in the ordinary course of business.

The Company has delivered or made available to Parent true, correct and complete copies of all of the Contracts listed on <u>Schedule 3.9.(a)</u> of the Disclosure Schedules, including all amendments and supplements thereto. All Contracts and Leases listed (or required to be listed) on <u>Schedule 3.9.(a)</u>, together with Advertiser and Publisher Agreements are referred to herein as "<u>Material Contracts</u>."

(b) Absence of Defaults. All of the Material Contracts are valid, binding and enforceable obligations of the Company and PocketGear and, to the Knowledge of the Company, the other parties thereto, in accordance with their terms, except as such enforceability may be limited by (i) bankruptcy, insolvency, moratorium, reorganization and other similar laws affecting creditors' rights generally and (ii) the general principles of equity, regardless of whether asserted in a proceeding in equity or at law. Each of the Company and PocketGear has complied in all material respects with the provisions of each Material Contract and is not in material Default thereunder and, to the Knowledge of the Company, all other parties to the Material Contracts have complied in all material respects with the provisions thereof and no party is in material Default thereunder. No notice of any

claim of Default has been given to the Company or PocketGear. No event has occurred or circumstance exists that (with or without notice or lapse of time) may contravene, conflict with, or result in a violation or breach of, or give the Company or PocketGear or, to the Knowledge of the Company, any other Person the right to default or exercise any remedy under, or to accelerate the maturity or performance of, or to cancel, terminate, or modify, any Material Contract.

3.10. Permits.

- (a) Schedule 3.10 of the Disclosure Schedules sets forth a complete list of all Permits used in the operation of the Business or otherwise held by the Company or PocketGear. Each of the Company and PocketGear has, and at all times has had, all Permits required under any Regulation (including Environmental Laws) in the operation of its Business, and owns or possesses such Permits free and clear of all Encumbrances. Neither the Company nor PocketGear is in Default, nor has the Company or PocketGear received any written notice of any claim of Default, with respect to any such Permit. No present or former stockholder, director, officer, Representative or employee of the Company, PocketGear or any Affiliate thereof, or any other Person, owns or has any proprietary, financial or other interest (direct or indirect) in any Permit which the Company or PocketGear owns, possesses or uses.
- (b) The Permits set forth on <u>Schedule 3.10</u> of the Disclosure Schedules collectively constitute all of the Permits necessary to permit the Company and PocketGear to lawfully conduct and operate the Business in the manner that the Business is currently conducted and operated and to permit the Company and PocketGear to own and use their assets in the manner in which the Company and PocketGear currently own and use their assets.
- 3.11. No Conflict or Violation. Neither the execution, delivery or performance of this Agreement or the Ancillary Agreements, nor the consummation of the transactions contemplated hereby or thereby, nor compliance by the Company and PocketGear with any of the provisions hereof or thereof, will (a) violate or conflict with any provision of the Organizational Documents of the Company or PocketGear, (b) violate, conflict with, or result in or constitute a Default under, or result in the termination of, or accelerate the performance required by, or result in a right of termination or acceleration, or result in the creation of any Encumbrance upon or with respect to any of the assets of the Company or PocketGear under any of the terms, conditions or provisions of any material Contract, Lease or Permit, (i) to which the Company or PocketGear is a party or (ii) by which the material assets of the Company or PocketGear are bound, (c) violate any Regulation or Court Order, (d) impose any material Encumbrance on any of the assets of the Company or PocketGear, (e) cause Parent, the Company, PocketGear or any of their respective Affiliates to become subject to, or to become liable for the payment of, any Tax, or (f) contravene, conflict with, or result in a violation of any of the terms or requirements of, or give any governmental body the right to revoke, withdraw, suspend, cancel, terminate, or modify any material Permits that are held by the Company or PocketGear or that otherwise relate to the Business or any of the assets of the Company or PocketGear.
- 3.12. Financial Statements and Other Financial Information. The Company has heretofore delivered to Parent the Financial Statements, copies of which are attached hereto as Schedule 3.12 of the Disclosure Schedules. The Financial Statements (a) are in accordance with the Books and Records of the Company and PocketGear; (b) have been prepared in accordance with GAAP consistently applied throughout the periods covered thereby and in a manner consistent with the past practice of the Company; and (c) fairly and accurately present in all material respects the assets, Liabilities (including all reserves) and financial condition of the Company and PocketGear as of the respective dates thereof and the results of operations and changes in cash flows of the Company and PocketGear for the periods then ended (subject, in the case of the Interim Financial Statements, to normal year-end adjustments, which adjustments shall not be material in the aggregate), and the omission of footnotes required by GAAP. The Year End Financial Statements have been audited by Grant Thornton LLP, independent certified public accountants, whose report thereon is included with such Year End Financial Statements. At the respective dates of the Financial Statements, there were no Liabilities of the Company or PocketGear, which, in accordance with GAAP, should have been set forth or reserved for in the Financial Statements or the notes thereto, which are not set forth or reserved for in the Financial Statements or the notes thereto.

- 3.13. <u>Undisclosed Liabilities</u>. Neither the Company nor PocketGear has Liabilities or obligations of any nature (whether accrued, absolute, contingent or otherwise), except (a) as reflected or reserved against in the Interim Balance Sheet, (b) Current Liabilities incurred in the ordinary course of business since the Interim Balance Sheet Date, (c) Liabilities incurred in connection with the negotiation and execution of this Agreement, the Ancillary Documents or the consummation of the transactions contemplated hereby or thereby which constitute Company Transaction Expenses, (d) Liabilities not required under GAAP to be shown on the balance sheet of the Company and that would not exceed \$250,000 in the aggregate, or (e) liabilities disclosed on <u>Schedule 3.13</u>.
- 3.14. Books and Records. The Books and Records of the Company and PocketGear are complete and correct in all material respects. The Company has made and kept (and given Parent access to) Books and Records and accounts, which, in reasonable detail, accurately and fairly reflect the activities of the Company, PocketGear and the Business. The minute books of the Company and PocketGear previously delivered to Parent accurately and adequately reflect all action previously taken by the Company Stockholders, board of directors and committees of the board of directors of the Company and PocketGear. The copies of the stock book records of the Company previously delivered to Parent are true, correct and complete, and accurately reflect all transactions effected in the capital stock of the Company through and including the date hereof. Neither the Company nor PocketGear has engaged in any transaction, maintained any bank account or used any corporate funds except for transactions, bank accounts and funds which have been and are reflected in the normally maintained Books and Records of the Company and PocketGear.
- 3.15. Litigation. Except as set forth on Schedule 3.15 of the Disclosure Schedules, there are no Actions pending, or to the Knowledge of the Company, threatened in writing (a) that are against (i) the Company, PocketGear, the Business or the assets of the Company or PocketGear (including with respect to Environmental Laws), (ii) to the Knowledge of the Company, any officers or directors of the Company or PocketGear that are involved in the operations of the Business or (iii) to the Knowledge of the Company, any Company Stockholder in such Company Stockholder's capacity as a Company Stockholder, (b) seeking to delay, limit or enjoin the transactions contemplated by this Agreement or the Ancillary Agreements, (c) against the Company, PocketGear or, to the Knowledge of the Company, any officer or director of the Company or PocketGear that involve the risk of criminal liability or (d) in which the Company or PocketGear is a plaintiff. Neither the Company nor PocketGear is in Default with respect to or subject to any Court Order, and there are no unsatisfied judgments against the Company, PocketGear, the Business or the assets of the Company or PocketGear. There is not a reasonable likelihood of an adverse determination of any pending Actions set forth on Schedule 3.15 of the Disclosure Schedules or otherwise related to or affecting the Business. There are no Court Orders or agreements with, or liens by, any Governmental Authority or quasi-governmental entity relating to any Environmental Law which regulate, obligate, bind or in any way affect the Company, PocketGear, the Business or the assets of the Company or PocketGear.

3.16. Labor Matters.

- (a) <u>Schedule 3.16</u> of the Disclosure Schedules contains a true, correct and complete list of the names and current annual salary rates or current hourly wages, as applicable, bonus opportunity, hire date, accrued vacation and paid-time-off, principal work location and leave status of all present employees of the Company and PocketGear and each such employee's status as being exempt or nonexempt from the application of state and federal wage and hour laws applicable to employees who do not occupy a managerial, administrative, or professional position. No officer or Key Employee of the Company or PocketGear has informed the Company or PocketGear (whether orally or in writing) of any intent to terminate employment with or services for the Company or PocketGear, and, to the Knowledge of the Company, no such Person or Persons has any intent to terminate employment with or services for the Company or PocketGear.
- (b) In the three (3) years preceding the date of this Agreement, the Company and PocketGear have remained in material compliance with all applicable Regulations and Court Orders regarding the terms and

conditions of employment of employees, former employees or prospective employees and other labor or employment related matters, including without limitation any Regulation or Court Order relating to wrongful discharge, discrimination, personal rights, wages, hours, collective bargaining, fair labor standards or occupational health and safety and any other Employment Statute. In the three (3) years preceding the date of this Agreement, neither the Company nor PocketGear has relied upon Contingent Workers in violation of IRCA and no unauthorized workers performed services on behalf of any member of the Company Group, whether as employees or employees of subcontractors. To the Knowledge of the Company, the consummation of the transactions contemplated by this Agreement will not cause any current employee of the Company and PocketGear to become unauthorized to work in the United States.

- (c) The Company and PocketGear are not and have not at any time been a party to any collective bargaining or similar agreement, and there are no labor unions or other organizations representing, purporting to represent or, to the Knowledge of the Company, attempting to represent, any employee of the Company or PocketGear. There are no unfair labor practice charges pending before the National Labor Relations Board or any other Governmental Authority, any grievances, complaints, claims or judicial or administrative proceedings, in each case, which are pending or, to the Knowledge of the Company, threatened by or on behalf of any employees. Neither the Company nor PocketGear has experienced any strike, slowdown, work stoppage, picketing, lockouts or other organized work interruption with respect to any employees during the three (3) years preceding the date of this Agreement, nor, to the Knowledge of the Company, are any such strikes, slowdowns, work stoppages, picketings, lockouts or other organized work interruptions threatened.
- (d) (i) The Company and PocketGear have paid in full to all of their employees or adequately accrued for in accordance with GAAP all wages, salaries, commissions, bonuses and other cash compensation due to or on behalf of such employees; (ii) there is no claim with respect to payment of wages, salary or overtime pay that has been asserted or is now pending or, to the Knowledge of the Company, threatened before any Governmental Authority with respect to any persons currently or formerly employed by the Company or PocketGear; and (iii) neither the Company nor PocketGear is a party to, or otherwise bound by, any consent decree with, or citation by, any Governmental Authority relating to employees or employment practices.
- (e) There are no material Liabilities, whether contingent or absolute, of the Company or PocketGear relating to workers' compensation benefits that are not fully insured against by a bona fide third-party insurance carrier, subject to normal deductibles. With respect to each Company Plan and with respect to each state workers' compensation arrangement that is funded wholly or partially through an insurance policy or public or private fund, all premiums required to have been paid to date under such insurance policy or fund have been paid.
- (f) In the three (3) years prior to the date of this Agreement, neither the Company nor PocketGear has effectuated (i) a "plant closing" (as defined in the Worker Adjustment and Retraining Notification Act (the "WARN Act") or any similar state, local or foreign Regulation) affecting any site of employment or one or more facilities or operating units within any site of employment or Facility of the Company or (ii) a "mass layoff" (as defined in the WARN Act, or any similar state, local or foreign Regulation) affecting any site of employment or Facility of the Company or PocketGear.
- (g) <u>Schedule 3.16.(g)</u> contains a list of all of the Contingent Workers currently engaged by the Company or PocketGear, along with the position, date of retention and rate of remuneration for each such Person. Except as set forth on <u>Schedule 3.16.(g)</u>, neither the Company nor PocketGear engages or retains any Contingent Workers. The Company and PocketGear have properly classified all of their Contingent Workers as either employees or independent contractors and as exempt or non-exempt for all purposes and have made all appropriate filings in connection with services provided by, and compensation paid to, such Contingent Workers. Each such Contingent Worker has entered into customary covenants regarding confidentiality, non-competition and assignment of Intellectual Property in such Person's agreement with the Company or PocketGear, a copy of which has been previously delivered or made available to Parent.

3.17. Compliance with Law. Since January 1, 2011, neither the Company, PocketGear nor the conduct of the Business has violated any, and the Company, PocketGear and the Business are now in compliance in all material respects with, all, Regulations and Court Orders relating to the Company, PocketGear the Business or the assets of the Company or PocketGear taken as a whole. Neither the Company nor PocketGear has received any written notice to the effect that it is not in compliance with any such Regulations or Court Orders, and to the Knowledge of the Company, the Company and PocketGear have no reason to anticipate that any existing circumstances are likely to result in violations of any of the foregoing.

3.18. Intellectual Property.

- (a) <u>Disclosure of Intellectual Property Assets</u>. <u>Schedule 3.18.(a)</u> of the Disclosure Schedules sets forth a true and complete list of all of the Company Registered Intellectual Property, all of which is registered exclusively in the name of the Company or PocketGear. With respect to Company Registered Intellectual Property, Schedule 3.18.(a) of the Disclosure Schedules also sets forth: (i) for each patent, the number, normal expiration date and subject matter for each country in which such patent has been issued, or, if applicable, the application number, date of filing and subject matter for each country, (ii) for each trademark, service mark, logo, and trade name, the application/serial number or registration number, the class of goods covered and the expiration date for each country, (iii) for each copyright, the registration number and expiration date for each country, or, if applicable, the application number and date of filing for each country in which a copyright application has been filed, (iv) for each Internet domain name, the Internet domain name registration, the expiration date, the name of the registrant and the name of the registrar, and (v) all actions that must be taken by the Company or PocketGear within 60 days after the Closing Date, including the payment of any registration, maintenance or renewal fees or the filing of any documents for the purposes of maintaining or renewing any Company Registered Intellectual Property. Schedule 3.18.(a) of the Disclosure Schedules also sets forth all unregistered trademarks, service marks, logos and trade names owned by the Company or PocketGear that are material to the Business. True and correct copies of all patents and patent applications, trademark and service mark applications and registrations and copyright applications and registrations owned, controlled, filed or used by or on behalf of the Company or PocketGear or in which the Company or PocketGear has any interest whatsoever, or is otherwise material to the Business, have been provided or made available to Parent, Except for the Intellectual Property licensed under the Contracts set forth in Schedule 3.18.(b) of the Disclosure Schedules, the Customer and Publisher Agreements and the Commercial Software Licenses (each as defined in Section 3.18.(b), the Company or PocketGear exclusively own all Intellectual Property used in the operation of the Business. In the case of Intellectual Property licensed by the Company or PocketGear from other Persons under licenses that are disclosed on Schedule 3.18.(b) of the Disclosure Schedules the Company or PocketGear have the right to use the applicable Intellectual Property in accordance with the scope of rights licensed to the Company or PocketGear set forth in such licenses.
- (b) <u>Licensing Agreements</u>. <u>Schedule 3.18(b)</u> of the Disclosure Schedules sets forth a true and complete list of all Contracts containing either (i) a license of (or covenant not to sue, non-assertion, settlement or similar agreements or consents related to) Intellectual Property by the Company or PocketGear to a third party, other than customer and publisher agreements entered into in the ordinary course of business, substantially in the form of the Company's forms of customer and publisher agreements, copies of which have been provided or made available to Parent (altogether, the "<u>Customer and Publisher Agreements</u>"), or (ii) a license of (or covenant not to sue, non-assertion, settlement or similar agreements or consents related to) Intellectual Property by a third party to the Company or PocketGear, other than licenses or other agreements for commercially available software that is made available for a total cost of less than \$10,000 (altogether, the "<u>Commercial Software Licenses</u>) (and in each case that have not, prior to the date hereof, expired or been terminated pursuant to their terms or operation of law). Neither the Company nor PocketGear is a party to a license of (or covenant not to sue, non-assertion, settlement or similar agreements or consents related to) Intellectual Property from a third party under which the Company or PocketGear is an exclusive licensee. A true and correct copy of each such Contract has been provided or made available to Parent. Such Contracts, along with the Customer and Publisher Agreements and the Commercial Software Licenses, are legal, valid, binding, enforceable obligations of the Company and

PocketGear and, to the Knowledge of the Company, the other parties thereto, in accordance with their terms, except as such enforceability may be limited by (x) bankruptcy, insolvency, moratorium, reorganization and other similar laws affecting creditors' rights generally and (y) the general principles of equity, regardless of whether asserted in a proceeding in equity or at law. Such Contracts are in full force and effect. The underlying Company Intellectual Property for each such Contract under which the Company or PocketGear licenses Company Intellectual Property to a third party is not subject to any outstanding injunction, judgment, order, decree or ruling. To the Knowledge of the Company, the underlying Intellectual Property for each such Contract under which a third party licenses the Company or PocketGear is not subject to any outstanding injunction, judgment, order, decree or ruling. No action, suit, proceeding, hearing, investigation or complaint is pending or, to the Knowledge of the Company threatened, nor has any claim or demand been made, which challenges the legality, validity, enforceability or ownership of the underlying Company Intellectual Property for each such Contract under which the Company or PocketGear licenses Company Intellectual Property to a third party. Neither the Company nor PocketGear has granted any sublicense or similar right with respect to any such Contract. To the Knowledge of the Company, no action, suit, proceeding, hearing, investigation or complaint is pending or threatened, nor has any claim or demand been made, which challenges the legality, validity, enforceability or ownership of the underlying Intellectual Property for each such Contract under which a third party licenses the Company or PocketGear. Other than as contemplated by the Customer and Publisher Agreements, neither the Company nor PocketGear has granted any sublicense or similar right with respect to any such Contract under which a third party licenses the Company or PocketGear. The Company, PocketGear and, to the Knowledge of the Company, all other parties to such Contracts are in compliance with, and have not breached any term of, any such Contracts, and have not performed any act or omitted to perform any act which, with notice or lapse of time or both, will become a breach or default thereunder. No party to any such Contract has given notice of termination or repudiation of any provision thereof. Immediately following the Closing Date, the Surviving Corporation will continue to be permitted to exercise all of the Company's and PocketGear's rights under such Contracts on terms identical to those in effect as of the date of this Agreement and to the same extent the Company or PocketGear would have been able to had the transactions contemplated by this Agreement and the Ancillary Agreements not occurred and without the payment of any additional amounts or consideration other than ongoing fees, royalties or payments which the Company or PocketGear would otherwise be required to pay. Schedule 3.18.(b) of the Disclosure Schedules also sets forth a true and complete list of all royalties, fees, and similar payments that the Company or PocketGear is obligated to pay to any third party who owns or is a licensee of any Intellectual Property, and except as set forth therein, the Company and PocketGear are not obligated or under any liability whatsoever under such contracts to make any payments by way of royalties, fees or otherwise to any owner or licensee of, or other claimant to, any Intellectual Property.

(c) Ownership, Validity and Enforceability of Intellectual Property. The Company Intellectual Property is subsisting and, to the Knowledge of the Company, valid, enforceable and in full force and effect, and is not subject to any pending or, to the Knowledge of the Company, threatened action, suit, proceeding, hearing, investigation, charge, complaint, claim, or demand that challenges the validity, enforceability, registration, ownership or use of the item. Except as disclosed on Schedule 3.18(b) of the Disclosure Schedules, the Company Registered Intellectual Property has not expired or been cancelled or abandoned, and is not subject to any pending or, to the Knowledge of the Company, threatened opposition, cancellation, interference or similar proceeding. All necessary registration, maintenance and renewal fees currently due in connection with the Company Registered Intellectual Property have been made, and all necessary documents, recordations and certificates in connection with the Company Registered Intellectual Property have been timely filed with the relevant patent, copyright, trademark or other authorities in the United States or foreign jurisdictions, as the case may be, for the purposes of maintaining such Company Registered Intellectual Property have been made, and all necessary documents, recordations and certificates in connection with the Company Registered Intellectual Property have been timely filed with the relevant patent, copyright, trademark or other authorities in the United States or foreign jurisdictions, as the case may be, for the purposes of maintaining such Company Registered Intellectual Property. Except as set forth on Schedule 3.18.(b) of the Disclosure Schedules, the Company and PocketGear own all right, title, and interest (including the sole right to

enforce), free and clear of all Encumbrances, in and to all Company Intellectual Property, and with respect to Company Registered Intellectual Property are listed in the records of the appropriate United States, state and/or foreign authorities as the sole owner for each item thereof. There are not, and it is reasonably expected that that immediately after the Closing there will not be, any restrictions on the Company's, PocketGear's or their Affiliates' right to sell products owned by or offer services provided by the Company or PocketGear, in connection with the Business, or to transfer or license any Company Intellectual Property. Except with respect to licenses of Intellectual Property by a third party to the Company or PocketGear disclosed on Schedule 3.18.(b) of the Disclosure Schedules and licenses or other agreements for Commercial Software Licenses, (i) the Company owns all Intellectual Property rights in and all Software used in the operation of the Business and (ii) to the Knowledge of the Company, the use of all Software in the operation of the Business does not require rights under any third party patents that are not licensed to the Company or PocketGear. Except as disclosed on Schedule 3.18.(b) of the Disclosure Schedules, neither the Company nor PocketGear has (i) transferred ownership of, or granted any exclusive license of or exclusive right to use, or authorized the retention of any exclusive rights to use or joint ownership of, any Intellectual Property that is or was Company Intellectual Property, to any other Person or (ii) permitted the Company's or PocketGear's rights in such Company Intellectual Property to enter into the public domain.

(d) Protection of Intellectual Property. Each of the Company and PocketGear is taking and has taken reasonable steps to obtain, maintain, police and protect the Company Intellectual Property and to maintain and protect the confidentiality of any trade secrets and other material Confidential Information of the Business. Except as set forth on Schedule 3.18.(c)each of the Company and PocketGear has the right to use, or has ownership of, free and clear of all Encumbrances, all customer lists, customer account information, correspondence, submissions and licensing and purchasing histories relating to the current and former customers of the Business, and no Person other than the Company and PocketGear possesses any claims or rights with respect to use of such information. Without limiting the foregoing, all current and former employees, Contingent Workers of the Company and PocketGear or other Persons that the Company or PocketGear has engaged to participate in the creation or development of any Intellectual Property used or held for use by the Company or PocketGear have executed valid and enforceable agreements in which they have assigned or otherwise vested all of their rights in and to such Intellectual Property to the Company or PocketGear and have agreed to maintain the confidentiality of such Intellectual Property. Each of the Company and PocketGear is not, and to the Knowledge of the Company no other party to any such agreement is, in material breach thereof. No current or former employees, Contingent Workers of the Company or PocketGear own any Intellectual Property used or held for use by the Company or PocketGear. It is not necessary for the operation of the Business to utilize any Intellectual Property rights of any employee, Contingent Worker of the Company or PocketGear or of the Business developed, invented or made prior to such employee's employment or a Contingent Worker's retention by the Company or PocketGear or except for any such Intellectual Property rights that have previously been assigned to the Company or PocketGear. No Person has asserted, and to the Knowledge of the Company, no such Person has, any right, title, interest or other claim in, or the right to receive any royalties or other consideration with respect to any Intellectual Property of the Company or PocketGear. To the Knowledge of the Company, the employment or engagement of any current or former employees and/or Contingent Workers does not subject the Company or PocketGear to any Liability to any third party for improperly soliciting such Contingent Workers, or contractor to work for the Company or PocketGear, whether such Liability is based on contractual or other legal obligations to such third party.

(e) No Infringement. Each of the Company and PocketGear has not and does not, and the Company Intellectual Property and operation of the Business have not and do not infringe, dilute, conflict with, misappropriate, or otherwise violate any rights of any Person in or to any Intellectual Property. Except as set forth in Schedule 3.18.(e) of the Disclosure Schedules, neither the Company nor PocketGear has received notice from any Person claiming that the Company Intellectual Property or the operation of the Business infringes, dilutes, conflicts with, misappropriates or otherwise violates the rights of any Person in or to any Intellectual Property, violates any rights to privacy or publicity, or constitutes unfair competition or trade practices under the laws of any jurisdiction, including any claim that the Company or PocketGear must license or refrain from using

any Intellectual Property rights of any Person and no Person has, any right to make such a claim. The Company Intellectual Property and Intellectual Property licensed to the Company and PocketGear is all of the Intellectual Property necessary for the operation of the Business.

- (f) No Third Party Infringers. Each of the Company and PocketGear has taken reasonable steps to protect the Company Intellectual Property from infringement by any other Person. To the Knowledge of the Company, there has been no unauthorized use, unauthorized disclosure, infringement, dilution, violation or misappropriation by any Person of any Company Intellectual Property. No other Person (i) has the right to use the Company's or PocketGear's trademarks or service marks on the goods and/or services in connection with which they are now being used either in identical form or in such near resemblance thereto as to be likely, when applied to the goods or services of any such Person, to cause confusion with such trademarks or to cause a mistake or to deceive or (ii) has notified the Company or PocketGear in writing that such Person is claiming any ownership of or right to use any of the Company Intellectual Property.
- (g) <u>Liabilities</u>. Except as set forth on <u>Schedule 3.18.(g)</u> of the Disclosure Schedules, other than the Customer and Publisher Agreements neither the Company nor PocketGear has reasonable basis to believe that it has any obligation under any agreement to provide indemnification for infringement of any third-party rights or otherwise, or provide updates, enhancements, improvements, modifications, fixes, support or maintenance for any Intellectual Property, Software or product or service of the Company or PocketGear. No third party has asserted its right to indemnification (nor, to Company's Knowledge, do any circumstances exist that provide a basis for a third party to exercise its right to indemnification) for infringement of any third-party rights or otherwise under any Contract involving any Company Intellectual Property or Products.
- (h) No Order. There are no forbearances to sue, consents, settlement agreements, judgments or orders entered into in connection with or in settlement of litigation or similar litigation-related, inter partes or adversarial-related, or government-imposed obligations to which the Company or PocketGear is a party or is otherwise bound that (i) restrict the rights of the Company or PocketGear to use, transfer, license or enforce any Company Intellectual Property; (ii) restrict the current or planned conduct of the Business in order to accommodate a third party's Intellectual Property; or (iii) grant any third party any right with respect to any Company Intellectual Property that is owned by Company or PocketGear or exclusively licensed to the Company or PocketGear.
- (i) Privacy. Each of the Internet websites owned or operated by the Company and PocketGear in connection with the Business maintains a publicly posted privacy statement or policy that describes the Company's and PocketGear's practices with respect to the collection, use and disclosure of Personal Information and that complies in all material respects with all applicable legal requirements, including, without limitation, Data Protection Laws. The Company, PocketGear and the Business complies and has complied with all Regulations and its own published and internal privacy policies, terms of use and guidelines related to information privacy and security, including with respect to the collection, use, disposal, disclosure, maintenance and transmission of Personal Information, including without limitation, Data Protection Laws. Without limiting the foregoing, each of the Internet websites owned or operated by the Company and PocketGear comply with Section 22575 of California's Business and Professions Code. The Company has provided or made available to Parent true and accurate copies of all such privacy policies. The published privacy policies are accurate and completely implemented. Each of the Company and PocketGear takes and has taken reasonable measures to ensure that such information is protected against unauthorized access, loss, damage use, sharing, modification, or other misuse other than as expressly described in the published privacy policies, and there has been no unauthorized access, loss, damage use, sharing, modification, or other misuse of any such information by the Company or PocketGear. The execution, delivery and performance of this Agreement and the Ancillary Agreements complies and will comply with all Data Protection Laws and the Company's and PocketGear's applicable published privacy policies. The Company and PocketGear are not prohibited by any applicable Regulations or any Company or PocketGear privacy policy or agreement from providing Parent or any of its Affiliates with the Personal Information that has been, or will be, provided to Parent or its Affiliates, on the

Closing Date, in connection with the transactions contemplated by this Agreement and the Ancillary Agreements. None of the disclosures made by the Company or PocketGear to employees, users or customers regarding the collection, use and disclosure of Personal Information and required by applicable Regulations (if any) have been unfair, inaccurate, misleading (whether expressly or by omission), or deceptive. Neither the Company nor PocketGear has received any written inquiry or complaint from a regulatory authority in any jurisdiction from which it has processed Personal Information, or written inquiry or complaint giving rise to legal proceedings, regarding its collection, use, storage, or sharing of Personal Information.

(j) Information Technology Systems. All information technology systems, Software computers, workstations, routers, hubs, switches, communication lines and other technology equipment used or held for use in connection with the operation of the Business, including without limitation, all databases, websites, e-commerce platforms and associated documentation used in connection with the operation of the Business ("Information Technology Systems"), and Company Intellectual Property (and all parts thereof), are free of (i) any critical defects, including without limitation any critical error or critical omission in the processing of any transactions and (ii) disabling codes or instructions and any "back door," "time bomb," "Trojan horse," "worm," "drop dead device," "virus" or hardware components that would permit unauthorized access or the unauthorized disruption, impairment, disablement or erasure of such Information Technology Systems or Company Intellectual Property (or any parts thereof) or data or other Software of users ("Contaminants"). Each of the Company and PocketGear takes and has taken reasonable steps and implements and has implemented reasonable procedures, intended to ensure that the Information Technology Systems used in connection with the operation of the Business are free from Contaminants. Each of the Company and PocketGear has reasonable disaster recovery plans, procedures and facilities for the Business, and takes and has taken reasonable steps to safeguard and back-up at secure off-site locations the Information Technology Systems. There have been no unauthorized intrusions or breaches of the security of such Information Technology Systems. Each of the Company and PocketGear has implemented security patches or upgrades that are generally available for such Information Technology Systems where such patches or upgrades are reasonably required to maintain their security. To the extent that the Company or PocketGear receives, processes, transmits or stores any financial account numbers (such as credit cards, bank accounts, PayPal accounts, debit cards), passwords, CCV data, or other related data ("Cardholder Data"), each of the Company and PocketGear represents and warrants that information security procedures, processes and systems have at all times met or exceeded all applicable information security laws, legal or contractual standards, rules and requirements related to the collection, storage, processing and transmission of Cardholder Data, including those established by applicable governmental regulatory agencies, and the Payment Card Industry Standards Council (including the Payment Card Industry Data Security Standard). Each of the Company and PocketGear has implemented and maintained reasonable security measures which are in compliance with all applicable data security and privacy Regulations and designed to (i) secure Personal Information, protect against hazards or threats to the security or integrity of Personal Information, and (ii) prevent unauthorized access to Personal Information. Each of the Company and PocketGear has not experienced any material accidental loss, alteration, unauthorized disclosure or destruction of, misuse of, unauthorized third party access to, or damage to, data held or processed by the Company or PocketGear or on behalf of any of the Company's or PocketGear's customers or employees, including Personal Information and Cardholder Data and the Information Technology Systems have not otherwise materially malfunctioned or failed. As used in this Section 3.18.(j) and in Section 3.18.(i), "reasonable" shall mean reasonable for an established web based company whose principal business is conducted through the internet. Each of the Company and PocketGear has, and on the Closing Date Parent will have full access to, a catalog, that is to the Knowledge of the Company true and complete, of all images, art, photographs and similar content posted on the Company's and PocketGear's websites prior to the Closing Date (including any such content removed or taken down from the Company's or PocketGear's website). The Company has sufficient rights to use all Information Technology Systems all of which rights shall survive as of immediately following the Closing following the consummation of the transactions contemplated hereby. The Information Technology Systems are sufficient to conduct the business of the Company and PocketGear.

- (k) <u>Transaction</u>. Except as set forth on <u>Schedule 3.18.(k)</u> of the Disclosure Schedules, the transactions contemplated by this Agreement and the Ancillary Agreements will not, by virtue of any Contract to which the Company or PocketGear is a party or by which the Company, PocketGear or their assets are bound, result in Parent, the Company, PocketGear, or any of their respective Affiliates: (i) granting to any third party any incremental right to or with respect to any Intellectual Property owned by, or licensed to, any of them, (ii) being bound by, or subject to, any incremental non-compete or other incremental material restriction on the operation or scope of their respective businesses, or (iii) being obligated to pay any incremental royalties or other material amounts, or offer any incremental discounts, to any third party. As used in this paragraph, an "incremental" right, non-compete, restriction, royalty or discount refers to a right, non-compete, restriction, royalty or discount, as applicable, in excess of the rights, non-competes, restrictions, royalties or discounts payable that would have been required to be offered or granted, as applicable, had the parties not entered into this Agreement and the Ancillary Agreements or consummated the transactions contemplated hereby.
- (l) Open Source. Except as set forth on Schedule 3.18.(l) of the Disclosure Schedules, no open source, public source or freeware Intellectual Property, or any modification or derivative thereof, including any version of any Software licensed pursuant to any GNU general public license or GNU lesser general public license or other Software that is licensed pursuant to a license that purports to require the distribution of or access to Source Code or purports to restrict one's ability to charge for distribution of or to use Software for commercial purposes (collectively "Open Source"), has been used in, incorporated into, integrated or bundled with, or used in the development or compilation of, any Software that is Company Intellectual Property. No Open Source listed in Schedule 3.18.(l) of the Disclosure Schedules has been modified or distributed by or on behalf of Company or PocketGear in such a manner as would require the Company or PocketGear to publicly make available any Source Code for Software that is Company Intellectual Property.
- (m) <u>Source Code</u>. Neither the Company nor PocketGear has disclosed any Source Code for Software that is Company Intellectual Property ("<u>Company Source Code</u>") to any Person other than to employees and contractors performing services on the Company's or PocketGear's behalf who have executed confidentiality agreements. Except as set forth in <u>Schedule 3.18.(m)</u> of the Disclosure Schedules, neither the Company nor PocketGear has entered into any Contracts with any Person requiring, upon the absence or occurrence of an event or Default, the disclosure of any Company Source Code. The execution of this Agreement or any Ancillary Agreement will not result in the disclosure to a third Person of any Company Source Code (including without limitation any release from escrow of any such Source Code) or the grant of incremental rights to a Person with regard to such Source Code. No event has occurred, and no circumstance or condition exists, that (with or without notice or lapse of time, or both) will, or would reasonably be expected to, result in the disclosure or delivery by the Company, PocketGear, or any Person acting on behalf of the Company or PocketGear to any Person of any Company Source Code, and no portion of such Source Code has been disclosed, delivered or licensed to a third Person. As used in this paragraph, an "incremental" right refers to a right in excess of the rights that would have been required to be offered or granted had the parties not entered into this Agreement and the Ancillary Agreements or consummated the transactions contemplated hereby.

3.19. Employee Benefit Plans.

(a) Schedule 3.19.(a) of the Disclosure Schedules sets forth a complete and correct list of all Company Plans. The Company has made available to Parent a true and complete copy, as applicable, of (i) each Company Plan (including any amendments thereto) and descriptions of all material terms of any such plan that is not in writing, (ii) the three (3) most recent annual reports with accompanying schedules and attachments, filed with respect to each Company Plan required to make such a filing, (iii) the most recent summary plan description for each Company Plan for which a summary plan description is required by applicable law and any other notice or description provided to employees (as well as any modifications or amendments thereto), (iv) the most recently received IRS determination or opinion letter, if any, issued by the IRS and each currently pending application to the IRS for a determination or opinion letter with respect to any Company Plan that is intended to qualify under Section 401(a) of the Code, (v) the three (3) most recently prepared actuarial reports, financial statements and

trustee reports, if any, relating to the Company Plan, (vi) all material records, notices and filings concerning IRS or Department of Labor audits or investigations, and (vii) all non-routine, written communications relating to any Company Plan and any proposed Company Plans. Neither the Company nor PocketGear has any plan or commitment to adopt or enter into any additional Company Plan or to materially amend or terminate any existing Company Plan.

- (b) No Company Plan is, and neither the Company nor PocketGear nor any ERISA Affiliate of the foregoing contributes to, has ever contributed to or has any Liability or obligation, whether actual or contingent, with respect to any (A) "multiemployer plan" (within the meaning of Section 3(37) or Section 4001(a)(3) of ERISA), (B) "multiple employer plan" (within the meaning of Section 413(c) of the Code), (C) single employer plan or other pension plan subject to Title IV or Section 302 of ERISA or Section 412 of the Code, or (D) multiple employer welfare arrangement (within the meaning of Section 3(40) of ERISA).
- (c) Neither the Company nor PocketGear has any obligation to provide health, accident, disability, life insurance or death benefits with respect to any current or former employees, consultants or directors or any retirees of the Company or PocketGear, or the spouses, dependents or beneficiaries of any of the foregoing, beyond the termination of employment or service of any such employee, consultant, director or retiree, whether under a Company Plan or otherwise, other than as required under Section 4980B of the Code or other applicable Regulation.
- (d) Neither the execution and delivery of this Agreement, nor the consummation of the transactions contemplated hereby, either alone or in combination with another event (whether contingent or otherwise) will (i) entitle any current or former employee, consultant or director of the Company or PocketGear or any group of such employees, consultants or directors to any payment; (ii) increase the amount of compensation or benefits due to any such employee, consultant or director or any such group of employees, consultants or directors; or (iii) accelerate the vesting, funding or time of payment of any compensation, equity award or other benefit.
- (e) Each Company Plan has been operated in all material respects in accordance with its terms and the requirements of all applicable Regulations, including, without limitation, ERISA and the Code. Each of the Company and PocketGear has performed in all material respects all obligations required to be performed by it under, is not in any respect in default under or in violation of, and has no knowledge of any default or violation by any party to, any Company Plan. No Action is pending or, to the Knowledge of the Company, threatened against, by or on behalf of any Company Plan or the assets, fiduciaries or administrators thereof (other than claims for benefits in the ordinary course). With respect to each Company Plan, no lien has been imposed under the Code, ERISA or any other applicable Regulation.
- (f) No material excise tax would reasonably be expected to be imposed upon the Company or PocketGear under Chapter 43 of the Code. Neither the Company nor PocketGear has made any filing in respect of any Company Plan under the Employee Plans Compliance Resolution System or the Department of Labor Delinquent Filer Voluntary Compliance Program.
- (g) Each Company Plan that is intended to be qualified under Section 401(a) of the Code has timely received a favorable determination letter or is entitled to rely on a favorable opinion letter from the IRS, in either case, that has not been revoked and, to the Knowledge of the Company, no event or circumstance exists that has adversely affected or would reasonably be expected to adversely affect such qualification or exemption. Each trust established in connection with any Company Plan which is intended to be exempt from federal income taxation under Section 501(a) of the Code is so exempt, and no fact or event has occurred that would reasonably be expected to adversely affect the exempt status of any such trust. There has not been any prohibited transaction (within the meaning of Section 406 of ERISA or Section 4975 of the Code) with respect to any Company Plan.
- (h) All contributions (including all employer contributions and employee salary reduction contributions), premiums or payments required to be made with respect to any Company Plan have been timely

made, or, to the extent not yet due, have been adequately accrued on the Interim Balance Sheet to the extent required by GAAP. All such contributions have been fully deducted for income tax purposes and no such deduction has been challenged or disallowed by any Governmental Authority.

- (i) No Company Plan, and neither the Company nor PocketGear with respect to any Company Plan, in any case, is the subject of an audit or investigation by the IRS, the Department of Labor, the PBGC or any other Governmental Authority, nor is any such audit or investigation pending or, to the Knowledge of the Company, threatened.
- (j) <u>Schedule 3.19.(j)</u> of the Disclosure Schedules sets forth any and all Indebtedness owed to the Company or PocketGear by any current or former employee; consultant or director of the Company or PocketGear.
- (k) No Company Plan is subject to the laws of any jurisdiction outside of the United States or provides compensation or benefits to any employee or former employee of the Company or PocketGear (or any dependent thereof) who resides outside of the United States.
- (1) The Company and PocketGear and each of their ERISA Affiliates are in compliance in all material respects with (i) the applicable requirements of Section 4980B of the Code and any similar state law and (ii) the applicable requirements of the Patient Protection and Affordable Care Act of 2010, as amended. No Company Plan is a voluntary employee benefit association under Section 501(a)(9) of the Code. The obligations of all Company Plans that provide health, welfare or similar insurance are fully insured by bona fide third-party insurers. No Company Plan is maintained through a human resources and benefits outsourcing entity, professional employer organization, or other similar vendor or provider.
- 3.20. <u>Transactions with Certain Persons</u>. Except as set forth on <u>Schedule 3.20</u> of the Disclosure Schedules, no stockholder, officer, director, Representative or employee of the Company, PocketGear or any Affiliate of the Company or PocketGear nor to the Knowledge of the Company any member of any such Person's immediate family is presently, or within the past three years has (a) been a party to any transaction with the Company or PocketGear, including, without limitation, any contract, agreement or other arrangement (i) providing for the furnishing of services by, (ii) providing for the rental of real or personal property from, or (iii) otherwise requiring payments to (other than for services as officers, directors or employees of the Company or PocketGear) any such Person or corporation, partnership, trust or other entity in which any such Person has an interest as a stockholder, officer, director, trustee or partner; (b) been engaged in competition with the Business; or (c) has had business dealings or a material financial interest in any transaction with the Company, PocketGear or with respect to the Business (other than business dealings or transactions conducted in the ordinary course of the business with the Company or PocketGear at substantially prevailing market prices and on substantially prevailing market terms).
- 3.21. Certain Payments. To the Knowledge of the Company, neither the Company, PocketGear, nor any of its directors, officers, Representatives or employees or any other Person affiliated with or acting for or on behalf of the Company, PocketGear or the Business, has ever directly or indirectly made any contribution, gift, bribe, rebate, payoff, influence payment, kickback or similar other payment to any Person, private or public, regardless of form, whether in money, property or services (i) to obtain favorable treatment in securing business, (ii) to pay for favorable treatment for business secured, (iii) to obtain special concessions or for special concessions already obtained, for or in respect of the Company, PocketGear or any of their respective Affiliates or (iv) in violation of any Regulations (including, without limitation, the United States Foreign Corrupt Practices' Act). Neither the Company, PocketGear, nor any Affiliate of the Company or PocketGear has participated, directly or indirectly, in any boycotts or other similar practices affecting any of its actual or potential customers or advertisers.

3.22. Tax Matters.

(a) Each of the Company and PocketGear has duly and timely filed or caused to be timely filed with the appropriate Tax Authorities all income and other material Tax Returns required to be filed by, or with respect to,

such entity. All such Tax Returns are true, complete and accurate in all material respects. All income and other material Taxes due and owing by the Company or PocketGear (whether or not shown on any Tax Returns) have been timely paid. Neither the Company nor PocketGear is currently the beneficiary of any extension of time within which to file any Tax Return. No written claim has ever been made by a Tax Authority in a jurisdiction where the Company or PocketGear does not file Tax Returns that the Company or PocketGear is or may be subject to Tax by that jurisdiction.

- (b) The unpaid Taxes of the Company and PocketGear did not, as of the Interim Balance Sheet Date, exceed the reserve for Tax Liability (excluding any reserve for deferred Taxes established to reflect timing differences between book and Tax income) set forth on the face of the Interim Balance Sheet (rather than in any notes thereto), and since the Interim Balance Sheet Date, neither the Company nor PocketGear has incurred any Liability for Taxes outside the ordinary course of business or otherwise inconsistent with past custom and practice.
- (c) No deficiencies for Taxes with respect to the Company or PocketGear have been claimed, proposed or assessed by any Tax Authority or other Governmental Authority. There are no pending or threatened audits, assessments, investigations, disputes, claims or other actions for or relating to any Liability in respect of Taxes of the Company or PocketGear. There are no matters under discussion with any Tax Authority or other Governmental Authority, with respect to Taxes that are likely to result in an additional Liability for Taxes with respect to the Company or PocketGear. No issues relating to Taxes of the Company or PocketGear were raised by the relevant Tax Authority in any completed audit or examination that would reasonably be expected to result in a material amount of Taxes in a later taxable period.
- (d) The Company has delivered or made available to Parent complete and accurate copies of federal, state, local and foreign Tax Returns of the Company, PocketGear and their predecessors for all taxable years since the taxable year ended 2008, and complete and accurate copies of all audit or examination reports and statements of deficiencies assessed against or agreed to by the Company or PocketGear (or any predecessor of the Company or PocketGear) since the last day of the last taxable year, if any, for which the applicable statute of limitations is closed, with respect to Taxes of any type. The Company and PocketGear (or any predecessor of the Company and PocketGear) have not waived any statute of limitations in respect of Taxes or agreed to any extension of time with respect to a Tax assessment or deficiency, nor has any request been made in writing for any such extension or waiver. No power of attorney (other than powers of attorney authorizing employees of the Company and PocketGear to act on behalf of the Company and PocketGear) with respect to any Taxes is currently in force or has been executed or filed with any Tax Authority.
- (e) There are no Encumbrances for Taxes upon any property or asset of the Company or PocketGear (other than for current Taxes not yet due and payable).
- (f) All material Taxes required to have been withheld, collected, deposited or paid, as the case may be, in connection with amounts paid or owing to any employee, independent contractor, creditor, stockholder or other third party, by the Company and PocketGear have been timely withheld, collected, deposited or paid. The Company and PocketGear have properly classified all individuals providing services to the Company or PocketGear, as applicable, as employees or non-employees for all relevant purposes. No transaction contemplated herein is subject to the tax withholding provisions of Section 3406 of the Code, or of Subchapter A of Chapter 3 of the Code or of any other provision of law.
- (g) Neither the Company nor PocketGear will be required to include any item of income in, or exclude any item of deduction from, taxable income for any period (or any portion thereof) ending after the Closing Date as a result of any installment sale or other transaction on or prior to the Closing Date, any prepaid amount received on or prior to the Closing, any accounting method change or agreement with any Tax Authority filed or made on or prior to the Closing Date, or any election under Section 108(i) of the Code.

- (h) Neither the Company nor PocketGear is, or has ever been, a party to or bound by any Tax indemnity agreement, Tax sharing agreement, Tax allocation agreement or similar arrangements other than tax indemnification provisions contained in commercial agreements not primarily related to Taxes entered into in the ordinary course of business.
- (i) Neither the Company nor PocketGear has been a member of an affiliated group filing a consolidated federal income Tax Return (other than a group the common parent of which is the Company) or any similar group for federal, state, local or foreign Tax purposes. Neither the Company nor PocketGear has any Liability for the Taxes of any other Person under Treasury Regulations Section 1.1502-6 (or any similar provision of state, local or foreign law), as a transferee or successor, by Contract, (other than tax indemnification provisions contained in commercial agreements not primarily related to Taxes entered into in the ordinary course of business), or otherwise.
- (j) The Company (i) has not been a stockholder of a "controlled foreign corporation", other than PocketGear, as defined in Section 957 of the Code (or any similar provision of state, local or foreign law); (ii) has not been a stockholder of a "passive foreign investment company" within the meaning of Section 1297 of the Code; (iii) has not engaged in a trade or business, had a permanent establishment (within the meaning of an applicable Tax treaty) or otherwise become subject to Tax jurisdiction in a country other than the country of its formation; and (iv) has not been a "personal holding company" as defined in Section 542 of the Code (or any similar provision of state, local or foreign law).
- (k) Neither the Company nor PocketGear has been a party to a transaction that is a "reportable transaction," as such term is defined in Treasury Regulations Section 1.6011-4(b)(1), or any other transaction requiring disclosure under analogous provisions of state, local or foreign Tax law. If the Company or PocketGear has entered into any transaction such that, if the treatment claimed by it were to be disallowed, the transaction would constitute a substantial understatement of federal income Tax within the meaning of Section 6662 of the Code, then the Company or PocketGear believes that it has either (x) substantial authority for the tax treatment of such transaction or (y) disclosed on its Tax Return the relevant facts affecting the tax treatment of such transaction. Neither the Company nor PocketGear has participated, and does not plan to participate, in any Tax amnesty program.
- (1) The Company has not elected at any time to be treated as an S corporation within the meaning of Sections 1361 or 1362 of the Code (or any similar provision of state or local law).
- (m) Neither the Company nor PocketGear has constituted either a "distributing corporation" or a "controlled corporation" (within the meaning of Section 355(a)(1)(A) of the Code) in a distribution of stock qualifying for tax-free treatment under Section 355 of the Code. Neither the Company nor PocketGear has acquired any assets from any other corporation in a transaction in which the adjusted Tax basis in the acquired assets was determined by reference (in whole or in part) to the adjusted Tax basis of the acquired assets (or any other property) in the hands of the transferor.
- (n) The Company has made good faith efforts to comply with all transfer pricing rules. All documentation required by all relevant transfer pricing laws have been timely prepared.
- (o) PocketGear (i) has never been a "surrogate foreign corporation" within the meaning of Section 7874(b) of the Code, (ii) is not treated as a United States corporation under Section 7874(b) of the Code, (iii) was not created or organized in the United States such that PocketGear would be taxable in the United States as a domestic entity pursuant to the dual charter provisions of Treasury Regulations Section 30.7701-5(a), (iv) has no investment in United States property (within the meaning of Section 956 of the Code), (v) is not a passive foreign investment company (within the meaning of Section 1297 of the Code), and (vi) is not engaged in a trade or business in the United States for U.S. federal income tax purposes.

- (p) As of the Closing Date, PocketGear will not hold assets which constitute U.S. property within the meaning of Section 956 of the Code.
- (q) For the period commencing on the first day of any Straddle Period and ending at the close of business on the Closing Date, PocketGear has not had any item of income which could constitute subpart F income within the meaning of Section 952 of the Code.
- (r) No compensation has been or would reasonably be expected to be includable in the gross income of any "service provider" (within the meaning of Section 409A of the Code) of the Company or PocketGear as a result of the operation of Section 409A of the Code.
- (s) Neither the execution and delivery of this Agreement or the Ancillary Agreements, nor the consummation of the transactions contemplated hereby or thereby, either alone or in combination with another event (whether contingent or otherwise) will result in any "parachute payment" under Section 280G of the Code (or any corresponding provision of state, local, or foreign Tax law).
- (t) There is no Contract, agreement, plan or arrangement to which the Company or PocketGear is a party which requires the Company or PocketGear to pay a Tax gross-up or reimbursement payment to any Person, including without limitation, with respect to any Tax-related payments under Section 409A of the Code or Section 4999 of the Code.
- (u) This Agreement (including any exhibits hereto) sets forth the complete terms of the Merger. Neither the Company nor PocketGear has taken any action or knows of any fact relating to the Merger that would reasonably be expected to prevent the Merger from qualifying as a "reorganization" within the meaning of Section 368(a) of the Code.
- 3.23. Insurance. Schedule 3.23 of the Disclosure Schedules contains a complete and accurate list of all policies or binders of fire, liability, title, worker's compensation, product liability and other forms of insurance (showing as to each policy or binder the carrier, policy number, coverage limits, expiration dates, annual premiums and a general description of the type of coverage provided) maintained by the Company and PocketGear with respect to or that covers the Business. True and correct copies of all such policies or binders have been delivered or made available to Parent. All insurance coverage applicable to the Company, PocketGear or the Business is in full force and effect and insures the Company and PocketGear in reasonably sufficient amounts against all risks, that, to the Knowledge of the Company, are usually insured against by Persons operating similar businesses or properties of similar size in the localities where such businesses or properties are located, provides coverage as may be required by applicable Regulation and by any and all Material Contracts or Facility Leases to which the Company or PocketGear is a party and has been issued by insurers of recognized responsibility. There is no Default under any such coverage nor has there been any failure to give notice or present any claim under any such coverage in a due and timely fashion. There are no outstanding unpaid premiums except in the ordinary course of business and no notice of cancellation or nonrenewal of any such coverage has been received. All products liability, general liability and workers' compensation insurance policies maintained by the Company and PocketGear have been occurrence policies and not claims made policies. There are no provisions in such Insurance Policies for retroactive or retrospective premium adjustments. There are no outstanding performance bonds covering or issued for the benefit of the Company or PocketGear. To the Knowledge of the Company, there are no facts upon which an insurer might be justified in reducing coverage or increasing premiums on existing policies or binders. No insurer has advised the Company or PocketGear that it intends to reduce coverage, increase premiums or fail to renew existing policy or binder.
- 3.24. <u>Publishers</u>. <u>Schedule 3.24</u> of the Disclosure Schedules sets forth a complete and accurate list of the names and addresses of each of the ten largest publishers of the Business, based on the dollar amount of total orders or other transactions with the Company or PocketGear during the last fiscal year and during the nine-month period ended September 30, 2014, showing the approximate total payments to each such publisher during

such fiscal year and during the nine-month period ended September 30, 2014. Since December 31, 2013, there has been no adverse change in the business relationship of the Company or PocketGear with any publisher named on <u>Schedule 3.24</u> of the Disclosure Schedules. Neither the Company nor PocketGear has received any written communication from any publisher named on <u>Schedule 3.24</u> of the Disclosure Schedules of any intention to terminate materially reduce such publisher's business relationship with the Company or PocketGear.

- 3.25. Advertisers. Schedule 3.25 of the Disclosure Schedules sets forth a complete and accurate list of the names and addresses of each of the ten largest advertisers of the Business, based on the dollar amount of total orders or other transactions with the Company or PocketGear during the last fiscal year and during the nine-month period ended September 30, 2014, showing the approximate total payments to each such advertiser during such fiscal year and during the nine-month period ended September 30, 2014. Since December 31, 2013, there has been no adverse change in the business relationship of the Company or PocketGear with any advertiser named on Schedule 3.25 of the Disclosure Schedules. Neither the Company nor PocketGear has received any written communication from any advertiser named on Schedule 3.25 of the Disclosure Schedules of any intention to terminate materially reduce such advertiser's business relationship with the Company or PocketGear.
- 3.26. Accounts Receivable. The accounts receivable set forth on the Interim Balance Sheet, and all accounts receivable arising since September 30, 2014, represent *bona fide* claims of the Company or PocketGear against debtors for products provided or services performed or other charges arising on or before the date hereof, and all products provided and services performed which gave rise to said accounts were delivered or performed in accordance with the applicable orders, Contracts, customer or advertiser requirements.
- 3.27. Compliance with Environmental Laws. To the Knowledge of the Company, neither the Company nor PocketGear is in violation of any applicable statute, law, or regulation relating to the environment or occupational health and safety, including all Environmental Laws, and no material expenditures are required in order to comply with such existing statute, law or regulation, including Environmental Laws. Neither the Company nor PocketGear has received any written notice of alleged, actual or potential responsibility for, or any inquiry or investigation regarding an alleged violation of or non-compliance with the conditions of any Environmental Law and the to the Knowledge of the Company there is no basis for such claim, notice, inquiry or investigation.
- 3.28. <u>Banking Relationships</u>. <u>Schedule 3.28</u> of the Disclosure Schedules sets forth a complete and accurate description of all arrangements that the Company, PocketGear or the Business has with any banks, savings and loan associations or other financial institutions providing for checking accounts, safe deposit boxes, borrowing arrangements, and certificates of deposit or otherwise, indicating in each case account numbers, if applicable, and the person or persons authorized to act or sign on behalf of the Company or PocketGear in respect of any of the foregoing.
- 3.29. No Brokers. Except as set forth in <u>Schedule 3.29</u> of the Disclosure Schedules, neither the Company, PocketGear nor any of their officers, directors, employees, stockholders, Representatives or Affiliates has employed or made any agreement with any broker, finder or similar agent or any Person which will result in the obligation of the Company, PocketGear, Parent or any of their respective Affiliates to pay any finder's fee, brokerage fees or commission or similar payment in connection with the transactions contemplated hereby.

3.30. <u>Disclosure Documents</u>.

(a) None of the information supplied or to be supplied in writing by or on behalf of the Company or any Company Subsidiary for inclusion or incorporation by reference in (i) the Form S-4 will, at the time such document is filed with the U.S. Securities and Exchange Commission (the "SEC"), at any time such document is amended or supplemented or at the time such document is declared effective by the SEC, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary to make the statements therein not misleading, or (ii) the Joint Proxy Statement will, at the date it is first mailed to the

stockholders of the Company and of Parent, at the time of the Company Stockholder Meeting and the Parent Stockholder Meeting, at the time the Form S-4 is declared effective by the SEC or at the Effective Time, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances in which they were made, not misleading. All documents that the Company is responsible for filing with the SEC in connection with the Transactions, to the extent relating to the Company or other information supplied by or on behalf of the Company for inclusion therein, will comply as to form, in all material respects, with the provisions of the Securities Act or Exchange Act, as applicable, and the rules and regulations of the SEC thereunder and each such document required to be filed with any Governmental Authority (other than the SEC) will comply in all material respects with the provisions of any applicable Law as to the information required to be contained therein.

- (b) The representations and warranties contained in this <u>Section 3.30</u> will not apply to statements or omissions included in the Form S-4 (including the Joint Proxy Statement) to the extent based upon information supplied by or on behalf of Parent or Merger Sub.
- 3.31. No Other Agreements to Sell the Assets or Capital Stock of the Company. Neither the Company, PocketGear nor any of their respective officers, directors, stockholders, Representatives or Affiliates has any commitment or legal obligation, absolute or contingent, to any other Person other than Parent to sell, assign, transfer or effect a sale of any of the assets of the Company or PocketGear, to sell or effect a sale of any capital stock of the Company or PocketGear, to effect any merger, consolidation, liquidation, dissolution or other reorganization of the Company or PocketGear, or to enter into any agreement or cause the entering into of an agreement with respect to any of the foregoing. Neither the Company nor PocketGear is currently engaged in discussions or negotiations with any party other than Parent with respect to any of the foregoing.
- 3.32. <u>Vote Required</u>. The (i) affirmative vote, at a duly called and held meeting of stockholders of the Company or (ii) approval by written consent in lieu of a meeting (in each case to approve the Merger, the Charter Amendment and the other Transactions), of the holders of (y) at least a majority of the outstanding shares of Company Common Stock and Company Preferred Stock (voting on an as converted to Common Stock basis) and (z) 66% of the outstanding shares of Series A Preferred Stock, Series B Preferred Stock, Series C Preferred Stock and Series D Preferred Stock (voting on an as converted to Common Stock basis), in each case entitled to vote upon the Merger and the other Transactions (the "<u>Company Stockholder Approval</u>"), is the only vote of the holders of any class or series of shares of capital stock of the Company necessary to approve the Merger, the Charter Amendment and the other Transactions. The affirmative vote or consent of the stockholders of that Company that are party to the Support Agreement is sufficient to obtain the Company Stockholder Approval approving the Merger, the Charter Amendment and the other Transactions.

Article IV. REPRESENTATIONS AND WARRANTIES OF PARENT AND MERGER SUB

Except as (i) disclosed in any Parent SEC Documents filed or furnished to the SEC by Parent after January 1, 2014 or (ii) otherwise set forth on Parent's Disclosure Schedules, Parent and Merger Sub hereby, jointly and severally, represent and warrant to the Company as follows, which representations and warranties are, as of the date hereof, and will be, as of the Closing Date, true and correct:

- 4.1. <u>Organization of Parent and Merger Sub.</u> Parent and Merger Sub are corporations duly organized, validly existing and in good standing under the laws of the State of Delaware, with full corporate power and authority to own and lease their respective properties and assets and conduct their respective businesses as presently being conducted.
- 4.2. <u>Authorization</u>. Parent and Merger Sub have all requisite corporate power and authority to execute and deliver this Agreement and the Ancillary Agreements to be executed and delivered by Parent and Merger Sub pursuant hereto, subject to receipt of Parent Stockholder Approval, to consummate the transactions contemplated

hereby and thereby and to perform their obligations hereunder and thereunder. The execution and delivery by Parent and Merger Sub of this Agreement and the Ancillary Agreements to which they are to be parties and the consummation by Parent and Merger Sub of the transactions contemplated hereby and thereby have been duly approved by the board of directors of Parent and the board of directors and stockholder of Merger Sub. Subject to the receipt of the Parent Stockholder Approval, and except as otherwise set forth herein, no other corporate proceedings on the part of Parent or Merger Sub are necessary to authorize this Agreement and the Ancillary Agreements to which they are to be parties and the transactions contemplated hereby and thereby. This Agreement has been duly executed and delivered by each of Parent and Merger Sub and is, and upon execution and delivery of the Ancillary Agreements to which Parent and Merger Sub are or will be parties, such Ancillary Agreements will be, legal, valid and binding obligations of Parent and Merger Sub, enforceable against Parent and Merger Sub in accordance with their terms, in each case, except as such enforceability may be limited by (a) bankruptcy, insolvency, moratorium, reorganization or other similar laws affecting creditors' rights generally and (b) the general principles of equity, regardless of whether asserted in a proceeding in equity or at law. The affirmative vote of not less than a majority of the votes cast by the holders of Parent Common Stock at the Parent Stockholder Meeting to approve the issuance of Parent Common Stock in connection with the Merger (the "Parent Stockholder Approval") is the only vote of the holders of any class or series of shares of capital stock of Parent Common Stock in connection with the Merger.

4.3. No Conflict or Violation; Required Consents and Filings.

- (a) Neither the execution, delivery or performance of this Agreement or the Ancillary Agreements, nor the consummation of the transactions contemplated hereby or thereby, nor compliance by Parent and Merger Sub with any of the provisions hereof or thereof, will (a) violate or conflict with any provision of the Organizational Documents of Parent (subject to obtaining the Parent Stockholder Approval prior to the Closing) and Merger Sub, (b) violate, conflict with, or result in or constitute a Default under, or result in the termination of, or accelerate the performance required by, or result in a right of termination or acceleration under, or result in the creation of any Encumbrance upon any of Parent's or Merger Sub's assets under, any of the terms, conditions or provisions of any contract, indebtedness, note, bond, indenture, security or pledge agreement, commitment, license, lease, franchise, permit, agreement, authorization, concession, or other instrument or obligation to which Parent or Merger Sub is a party, except for any violation, conflict, Default, termination, acceleration or creation of Encumbrance which would not prevent or materially delay the ability of Parent to consummate the transactions contemplated by this Agreement or the Ancillary Agreements, or (c) violate any Regulation or Court Order.
- (b) The execution and delivery of this Agreement by each of Parent and Merger Sub does not, and the performance of this Agreement and the transactions contemplated hereby by each of Parent and Merger Sub will not, require any consent, approval, authorization or permit of, or filing with or notification to, any Governmental Authority, except (i) the filing with the SEC of (A) the Form S-4 and declaration of effectiveness of the Form S-4 and (B) such reports under, and other compliance with, the Exchange Act (and the rules and regulations promulgated thereunder) and the Securities Act (and the rules and regulations promulgated thereunder) as may be required in connection with this Agreement and NASDAQ, (iii) the filing of the Certificate of Merger and the acceptance for record by the Delaware Secretary or State, (v) such filings and approvals as may be required by any applicable state securities or "blue sky" Laws, (vi) such filings as may be required in connection with the HSR Act and (vii) where failure to obtain such consents, approvals, authorizations or permits, or to make such filings or notifications, individually or in the aggregate, has not had and would not reasonably be expected to be material adverse to Parent.
- 4.4. <u>Parent Common Stock</u>. All of the shares of Parent Common Stock to be issued as part of the Aggregate Stock Consideration shall, when issued and delivered in accordance with this Agreement, be duly authorized, validly issued, fully paid and non-assessable, and none of such shares of Parent Common Stock shall be subject to any preemptive rights.

4.5. No Brokers. Except for Needham & Company, LLC, neither Parent, Merger Sub, nor any of their respective Representatives or Affiliates has employed or made any agreement with any broker, finder or similar agent or any Person which will result in the obligation of Parent or Merger Sub or any of their respective Affiliates to pay any finder's fee, brokerage fees or commission or similar payment in connection with the transactions contemplated hereby.

4.6. SEC Reports and Financial Statements.

- (a) A true and complete copy of each annual, quarterly and other report, registration statement, and definitive proxy statement filed by Parent with the SEC since January 31, 2013 and prior to the date hereof (the "Parent SEC Documents") is available on the Web site maintained by the SEC at http://www.sec.gov, other than portions in respect of which confidential treatment was granted by the SEC. As of their respective filing dates, the Parent SEC Documents complied in all material respects with the requirements of the United States Securities Act and the Exchange Act, as the case may be, and the rules and regulations of the SEC promulgated thereunder applicable to such Parent SEC Documents, and none of the Parent SEC Documents contained on their filing dates any untrue statement of a material fact or omitted to state a material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances under which they were made, not misleading, except to the extent corrected by a Parent SEC Document filed prior to the date of this Agreement
- (b) Each of the consolidated financial statements (including, in each case, any notes thereto) contained in the Parent SEC Reports was prepared in accordance with GAAP applied on a consistent basis throughout the periods indicated (except as may be indicated in the notes thereto or, in the case of unaudited statements, as permitted by Form 10-Q of the SEC) and each fairly presents, in all material respects, the consolidated financial position, results of operations and cash flows of Parent and its consolidated subsidiaries as at the respective dates thereof and for the respective periods indicated therein, except as otherwise noted therein (subject, in the case of unaudited statements, to year-end adjustments).

4.7. Disclosure Documents.

- (a) None of the information supplied or to be supplied in writing by or on behalf of Parent or Merger Sub for inclusion or incorporation by reference in (i) the Form S-4 will, at the time such document is filed with the SEC, at any time such document is amended or supplemented or at the time such document is declared effective by the SEC, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary to make the statements therein not misleading, or (ii) the Joint Proxy Statement will, at the date it is first mailed to the stockholders of the Company and of Parent, at the time of the Company Stockholder Meeting and the Parent Stockholder Meeting, at the time the Form S-4 is declared effective by the SEC or at the Effective Time, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances in which they were made, not misleading. All documents that Parent is responsible for filing with the SEC in connection with the Transactions, to the extent relating to Parent or any Parent Subsidiary or other information supplied by or on behalf of Parent or any Parent Subsidiary for inclusion therein, will comply as to form, in all material respects, with the provisions of the Securities Act or Exchange Act, as applicable, and the rules and regulations of the SEC thereunder and each such document required to be filed with any Governmental Authority (other than the SEC) will comply in all material respects with the provisions of any applicable Law as to the information required to be contained therein.
- (b) The representations and warranties contained in this <u>Section 4.7</u> will not apply to statements or omissions included in the Form S-4 (including the Joint Proxy Statement) to the extent based upon information supplied by or on behalf of the Company.
- 4.8. <u>Parent Capitalization</u>. As of November 11, 2014, the authorized capital of Parent consists of 200,000,000 shares of Parent Common Stock, of which 38,551,821 are issued and 37,797,221 are outstanding, and 2,000,000 shares of Series A convertible preferred stock, \$0.0001 par value, of which 100,000 shares are issued and outstanding. All such shares of capital stock of Parent have been duly authorized and validly issued,

and are fully paid and nonassessable and have been issued and granted in all material respects in compliance with all applicable securities registration Laws. The rights, preferences, privileges and restrictions of the capital stock of Parent are as stated in Parent's certificate of incorporation in effect as of the date hereof.

- 4.9. <u>Litigation</u>. As of the date of this Agreement, there is no Action pending before any court of competent jurisdiction (or, to the Knowledge of Parent or Merger Sub, threatened in writing) against Parent or Merger Sub, any of Subsidiaries of Parent or Merger Sub or any of their assets or properties or any of their respective directors, officers or employees (in their capacities as such or relating to their employment, services or relationship with Parent or Merger Sub or any of their Subsidiaries) that would have a Parent Material Adverse Effect.
- 4.10. Merger Sub. Merger Sub is a direct wholly-owned subsidiary of Parent (a) was formed solely for the purpose of engaging in the transactions contemplated by this Agreement, (b) has engaged in no other business activities and (c) has conducted its operations only as contemplated by this Agreement. All of the issued and outstanding equity of Merger Sub is validly issued, fully paid and non-assessable and is owned, beneficially and of record, by Parent free and clear of all Encumbrances, options, rights of first refusal, stockholder agreements, limitations on Parent's voting rights and other encumbrances of any nature whatsoever.
- 4.11. <u>Absence of Certain Changes</u>. Except as expressly contemplated by this Agreement, between the date of Parent's most recent quarterly report on Form 10-Q required to be filed with the SEC and the date of this Agreement, there has not occurred any change, effect or event that has had or, with notice or lapse of time or both, would reasonably be expected to have a Parent Material Adverse Effect.
- 4.12. <u>Compliance with Laws</u>. As of the date of this Agreement. Parent is in material compliance with all applicable Regulations and Court Orders relating to Parent except for such violations that would not have a Parent Material Adverse Effect.
- 4.13. <u>Intellectual Property</u>. To Parent's knowledge, no Person has infringed, misappropriated, or otherwise violated, and no Person is currently infringing, misappropriating, or otherwise violating, any Intellectual Property owned (or purported to be owned) by, or exclusively licensed to, Parent or any of its Subsidiaries, except for such infringement, misappropriation or violation that would not have a Parent Material Adverse Effect. To Parent's knowledge, the operation of Parent's business does not infringe, violate or misappropriate any Intellectual Property of any other Person, except for such infringement, violation or misappropriation that would not have a Parent Material Adverse Effect.

Article V. COVENANTS

The Company, the Company Stockholders (by virtue of the execution and delivery of the Stockholder Transmittal Letters), Parent and Merger Sub each covenant and agree as follows:

5.1. Further Assurances.

(a) Upon the terms and subject to the conditions contained herein, the parties agree, both before and after the Closing, (i) to use their respective reasonable best efforts to take, or cause to be taken, all actions and to do, or cause to be done, all things necessary, proper or advisable to consummate and make effective the transactions contemplated by this Agreement and the Ancillary Agreements, (ii) to execute any documents, instruments or conveyances of any kind which may be reasonably necessary or advisable to carry out any of the transactions contemplated hereunder or thereunder, and (iii) to cooperate with each other in connection with the foregoing.

(b) Without limiting the foregoing, the parties agree to use their respective reasonable best efforts (A) to obtain all necessary waivers, consents and approvals necessary or desirable for the consummation of the transactions contemplated by this Agreement or the Ancillary Agreements from other parties to the Contracts and Leases, provided that neither Company, Parent, Merger Sub nor any of their respective Affiliates shall be required to make any payments, commence litigation or agree to modifications of the terms thereof in order to obtain any such waivers, consents or approvals; (B) to obtain all Permits as are required to be obtained under any Regulations that are necessary or advisable to consummate the transactions contemplated hereunder; (C) to defend all Actions challenging this Agreement or the consummation of the transactions contemplated hereby; (D) to seek to lift or rescind any injunction or restraining order or other Court Order adversely affecting the ability of the parties to consummate the transactions contemplated hereby; (E) to give all notices to, and make all registrations and filings with third parties, including, without limitation, submissions of information requested by Governmental Authorities that are necessary or advisable to consummate the transactions contemplated hereunder, (F) promptly make all necessary filings, and thereafter promptly make any other required submissions, with respect to this Agreement required under the HSR Act or any other applicable Law; and (G) to fulfill all conditions to this Agreement. Company and Parent shall keep each other apprised of the status of matters relating to the completion of the transactions contemplated hereby and provide each other party with a reasonable opportunity to review and comment on any waivers, consents, approvals, notices, orders, registrations and filings to be made, given or used by Parent or the Company and shall promptly deliver to the other party a copy of each such registration or filing made, each such notice given and each such waiver, consent, approval or order obtained by Parent or the Company prior to the Closing Date.

(c) If a Notification and Report Form pursuant to the HSR Act is required to be filed in order to consummate the transactions contemplated hereby, the Company and Parent agree (i) to file or submit a Notification and Report Form pursuant to the HSR Act and any notifications required under any other applicable antitrust, competition or merger control law, and the rules and regulations promulgated thereunder, in each case as promptly as reasonably practicable (but in any event, with respect to the Notification and Report Form pursuant to the HSR Act, within ten (10) days after the date hereof); (ii) to furnish to the other parties or their outside counsel such necessary information and reasonable assistance as the other parties may reasonably request in connection with their preparation of any filing or submission under the HSR Act and any other applicable antitrust, competition or merger control law; (iii) to provide to the other parties or their outside counsel, subject to applicable laws and reasonable advice of outside counsel, a draft of any filing or submission under the HSR Act or any other applicable antitrust, competition or merger control law and a reasonable opportunity to review such draft before making or causing to be made such filing or submission, and to consider in good faith the views of the other parties regarding such filing or submission; (iv) to respond as promptly as practicable to all inquiries or requests received from the Federal Trade Commission, the Antitrust Division of the Department of Justice or any other Governmental Authority for additional information or documentation; (v) to keep the other parties apprised of the status of any communications with, and any inquiries or requests from, any Governmental Authority; and (vi) to take any and all other commercially reasonable steps necessary or advisable to cause the expiration or termination of any waiting period (and any extension thereof) under the HSR Act, obtain any other waiting period expirations or terminations, waivers, consents and approvals under other applicable antitrust, competition and merger control laws, and remove any other legal or other impediment to consummation of the transactions contemplated by this Agreement and the Ancillary Agreements under such laws; provided, however, that notwithstanding anything in this Agreement to the contrary, Parent shall not be required to defend or otherwise contest any litigation or other proceeding challenging the transactions contemplated by this Agreement or the Ancillary Agreements under the FTC Act, the Clayton Act, or any other similar antitrust, competition or merger control law. The Parent, Company, PocketGear and their Affiliates shall not take any action, or enter into any agreement, to extend any applicable waiting or review period or to delay or not to consummate the transactions contemplated by this Agreement or the Ancillary Agreements to be consummated on the Closing Date, except with the prior written consent of the other persons.

(d) Notwithstanding anything in this Agreement to the contrary, in no event shall Parent or any of its Affiliates be required (i) to propose, or agree to accept, any undertaking or condition, (ii) to enter into any

consent decree, hold separate or license agreement, (iii) to make any sale, divestiture or other disposition, (iv) to accept any operational, financial or ownership restriction, (v) or to take or commit to take any action or nonaction in order to permit the consummation of the transactions contemplated by this Agreement and the Ancillary Agreements, and further, none of the Company, PocketGear or their Affiliates shall propose, agree to, or take any of the actions or commitments described in this subsection (d) without the prior written consent of Parent. If requested by Parent, the Company, PocketGear and their Affiliates shall agree to, and shall take, any action contemplated by this subsection (d), provided that such agreement or action is conditioned on the consummation of the transactions contemplated by this Agreement.

- 5.2. Audits. The Company shall use reasonable best efforts to cooperate with Parent and Parent's independent public accountants ("Parent's Accountants") in connection with the completion of any audit of the historical financial statements of the Company and/or the Business for the fiscal years ended December 31, 2011, 2012 and 2013 and the preparation of any periodic report required to be filed under the Securities Exchange Act of 1934, as amended (a "Periodic Report") and/or of any disclosure document of Parent or its Affiliates to be used by Parent or its Affiliates to obtain financing and/or to register securities of Parent or its Affiliates (in each case, a "Disclosure Document") including, without limitation, any registration statement under the Securities Act and any syndication memorandum for debt financing of Parent or its Affiliates. Such cooperation shall include, without limitation (a) assisting Parent and/or Parent's Accountants in the preparation, audit and review of financial statements for the Company and/or the Business for any fiscal years or portions thereof reasonably determined by Parent that may be necessary for inclusion in any Periodic Report and/or Disclosure Document or in any filing that Parent may desire to make with the SEC or other regulatory authority, (b) executing and providing any certifications, representation letters or similar documents and providing such other information as Parent or Parent's Accountants may reasonably request in conjunction with an audit of such financial statements and/or in conjunction with Parent's Accountants' preparation of any comfort letters with respect to any Periodic Report and/or Disclosure Document (and responding to any inquiries from Parent or Parent's Accountants in connection with any such certifications, representation letters or other information), (c) providing Parent with access to the Company's prior accountants' work papers relating to any such financial statements and information (if any), (d) assisting in the preparation of responses to comments received from the Staff of the SEC with respect to any such financial statements included in any SEC filings, (e) assisting in the preparation of any statutory or regulatory audits required to be filed with respect to the Company and/or the Business in any jurisdiction and (f) executing and providing appropriate managers' certifications of any such financial statements pursuant to the Sarbanes-Oxley Act of 2002 and responding to inquiries from Parent's management in connection with any such managers' certification. In addition, the Company shall cause its accountants to provide any necessary consents to the inclusion of the Company's audited financial statements in the Joint Proxy Statement and Form S-4.
- 5.3. No Solicitation. From the date hereof through the Closing or the earlier termination of this Agreement, neither the Company nor any Company Stockholder shall, and each of them shall cause each of their respective Representatives (including, without limitation, investment bankers, attorneys and accountants), not to, directly or indirectly, enter into, solicit, initiate or continue any discussions or negotiations with, or encourage or respond to any inquiries or proposals by, or participate in any negotiations with, or provide any information to, or otherwise cooperate in any other way with, any Person or group, other than Parent and its Affiliates and Representatives, concerning any sale of all or a portion of the Company's assets, or of any capital stock of the Company, or any merger, consolidation, liquidation, dissolution or similar transaction involving the Company (each such transaction being referred to herein as a "Takeover Proposal"). From the date hereof through the Closing or the earlier termination of this Agreement, neither the Company nor any Company Stockholder shall, directly or indirectly, through any member, officer, employee, Representative, agent or otherwise, solicit, initiate or encourage the submission of any proposal or offer from any Person relating to any Takeover Proposal or participate in any negotiations regarding, or furnish to any other Person any information with respect to the Company for the purposes of, or otherwise cooperate in any way with, or assist or participate in, facilitate or encourage, any effort or attempt by any other Person to seek or effect a Proposed Acquisition Transaction. The Company shall promptly advise such prospective purchaser or soliciting party, by written notice (with a copy to

Parent) of the terms of this Section 5.3 and will promptly notify Parent (orally and in writing) if any such offer, or any inquiry or contact with any Person with respect thereto, is made and shall provide Parent with a copy of such offer, the terms of any proposal, including, without limitation, the identity of the prospective purchaser or soliciting party, and shall keep Parent informed on the status of any negotiations regarding such offer. The Company agrees not to release any third party from, or waive any provision of, any confidentiality or standstill agreement to which the Company is a party.

5.4. Notification of Certain Matters.

- (a) From the date hereof through the Closing Date, the Company shall give prompt notice to Parent of (a) the occurrence, or failure to occur, of any event which occurrence or failure would be likely to cause any representation or warranty contained in this Agreement or in any exhibit or schedule hereto to be untrue or inaccurate in any respect and (b) any failure of the Company to comply with or satisfy any covenant, condition or agreement to be complied with or satisfied the Company under this Agreement or any exhibit or schedule hereto; provided, however, that such disclosure shall not be deemed to cure any breach of a representation, warranty, covenant or agreement or to satisfy any condition. The Company shall promptly notify Parent of any Default, the threat or commencement of any Action, or any development that occurs before the Closing that could in any way materially affect the Company or the Business.
- (b) From the date hereof through the Closing Date, Parent shall give prompt notice to the Company of (a) the occurrence, or failure to occur, of any event which occurrence or failure would be likely to cause any representation or warranty contained in this Agreement or in any exhibit or schedule hereto to be untrue or inaccurate in any respect and (b) any failure of Parent to comply with or satisfy any covenant, condition or agreement to be complied with or satisfied the Company under this Agreement or any exhibit or schedule hereto; <u>provided</u>, <u>however</u>, that such disclosure shall not be deemed to cure any breach of a representation, warranty, covenant or agreement or to satisfy any condition. Parent shall promptly notify the Company of any Default, the threat or commencement of any Action, or any development that occurs before the Closing that could in any way materially affect the Company or the Business.
- 5.5. <u>Investigation</u>. From the date hereof through the Closing Date, the Company shall, and shall cause its respective directors, officers, employees and agents to, use reasonable best efforts to afford the Representatives of Parent and its Affiliates access at all reasonable times and upon reasonable notice to the Company and the Business for the purpose of inspecting the same, and to the directors, officers, employees, agents, attorneys, accountants, properties, Company Plans, Books and Records and Contracts of the Company, and shall furnish Parent and its Representatives all financial, operating and other data and information as Parent or its Affiliates, through their respective Representatives, may reasonably request.
- 5.6. <u>Conduct of Business</u>. Between the date of this Agreement and the Closing Date, the Company shall conduct the Business in the ordinary course of business consistent with past practice and shall not take any action inconsistent with this Agreement or the consummation of the Closing. Without limiting the generality of the foregoing, the Company shall not, and the Company shall cause PocketGear not to, except as specifically contemplated by this Agreement or as consented to by Parent in writing:
- (a) issue, or commit to issue, any capital stock of the Company or PocketGear or any security convertible into or exchangeable for such capital stock, including, without limitation, any stock option or restricted stock; grant any registration rights; purchase, redeem, retire or otherwise acquire any capital stock; or declare or pay any dividend or other distribution or payment in respect of the capital stock, except for shares of Company Capital Stock issued upon the exercise of Company Options and Series C Warrants outstanding on the date hereof;
 - (b) amend any Organizational Documents of the Company or PocketGear;
- (c) adopt or change any material Tax or other accounting methods, principles or practices or change any annual Tax accounting period; make or change any material Tax election; settle or compromise any material

claim, notice, audit report or assessment in respect of Taxes; file any amended Tax Return; enter into any Tax allocation agreement, Tax sharing agreement, Tax indemnity agreement or closing agreement relating to any Tax; knowingly surrender any right to claim a Tax refund; or consent to any extension or waiver of the statute of limitations period applicable to any material Tax claim or assessment;

- (d) revalue any assets of the Company, PocketGear or relating to the Business, including, without limitation, writing off notes or accounts receivable, except in the ordinary course of business consistent with past practice;
- (e) fail to take reasonable actions to prevent any damage, destruction or loss (whether or not covered by insurance) adversely affecting any of the assets of the Company, PocketGear or the Business;
- (f) acquire any corporation, partnership, limited liability company, other business organization or division thereof or any material amount of assets, or enter into any joint venture, strategic alliance, exclusive dealing, noncompetition or similar contract or arrangement;
- (g) except for the Merger, adopt a plan of complete or partial liquidation, dissolution, merger, consolidation, restructuring, recapitalization or other reorganization of the Company or of PocketGear, or otherwise alter the Company's or PocketGear's corporate structure;
- (h) cancel any Indebtedness or waive or release any material right or claim of the Company or PocketGear or relating to the Business, except in the ordinary course of business consistent with past practice;
- (i) (i) hire or terminate any employee, Contingent Worker or director, (ii) pay, announce, promise or grant, whether orally or in writing, increase or establish (each, as applicable) any wages, base pay, fees, salaries, compensation, bonuses, incentives, deferred compensation, pensions, severance or termination or change in control payments, retirement, profit sharing, fringe benefits, equity or equity-linked awards, employee benefit plans, or any other form of compensation or benefits payable by the Company or PocketGear, including without limitation, any increase or change pursuant to any Company Plan (except as required by any Regulation), in each case except (A) pursuant to the terms of a Company Plan in effect as of the date of this Agreement, and (B) as required by applicable Regulation, or (iii) enter into, adopt, materially amend or terminate any Company Plan;
- (j) take any action which could reasonably be expect to cause, or fail to take any reasonable action to prevent, any change in employee relations which has or is reasonably likely to have a material adverse effect on the productivity, the financial condition or, results of operations of the Company or PocketGear;
 - (k) enter into any collective bargaining agreement or other Contracts with any labor union;
- (l) enter into, amend, cancel or terminate any Contract, Lease or Permit to which the Company or PocketGear is a party or otherwise relating to the Business, including, without limitation, any employment, consulting, distribution, dealer, sales representative, joint venture, credit or similar agreement or any Contract or transaction involving a total remaining commitment by or to the Company or PocketGear of at least \$50,000, except in the ordinary course of business;
 - (m) enter into any Contract with any Related Party of the Company or PocketGear;
- (n) mortgage, pledge or otherwise encumber any assets of the Company or PocketGear or relating to the Business, except purchase money mortgages arising in the ordinary course of business;
- (o) sell, assign or transfer any assets of the Company or PocketGear or relating to the Business, other than in the ordinary course of business;

- (p) incur Indebtedness of the Company or PocketGear for borrowed money, commit to borrow money, make or agree to make loans or guarantee Indebtedness;
- (q) incur Liabilities of the Company or PocketGear, except Liabilities incurred in the ordinary course of business, or increase or change any assumptions underlying or methods of calculating, any doubtful account contingency or other reserves of the Company or PocketGear;
 - (r) pay, discharge or satisfy any Liabilities of the Company or PocketGear other than in the ordinary course of business;
- (s) make any single capital expenditure relating to the Business in excess of \$10,000 or capital expenditures that are, in the aggregate, in excess of \$50,000 incur any obligations to make any capital expenditures or execute any Lease;
 - (t) fail to pay or satisfy when due any material Liability of the Company, PocketGear or related to the Business;
- (u) except in the ordinary course of business, accelerate the collection of or discount any accounts receivable, delay the payment of accounts payable or defer expenses, reduce inventories or otherwise increase cash in hand;
- (v) amend or otherwise take any action that would permit or cause any Company Option or Restricted Stock Award to accelerate in contemplation of or as a consequence of the Merger or other transactions contemplated by this Agreement;
- (w) fail to carry on diligently the Business in the ordinary course so as to preserve for Parent the Business and the goodwill of the customers, publishers and advertisers of the Business and others having business relations with the Company, PocketGear or the Business;
- (x) dispose, sell, transfer, assign, encumber, pledge, license, abandon, fail to maintain, or allow to lapse any Intellectual Property of the Company, PocketGear or relating to the Business or dispose or disclose (or agree to disclose upon the occurrence of any event, non-event or Default) to any Person any Intellectual Property of the Company, PocketGear or relating to the Business not theretofore a matter of public knowledge; or
- (y) agree, whether orally or in writing, to do any of the things described in the preceding clauses (a) through (x) other than as expressly provided for herein.

5.7. Tax Matters.

(a) The Stockholder Representative and Parent shall give prompt notice to each other of any inquiries, claims, assessments, audits, proposed adjustment, or similar events with respect to Taxes of the Company or PocketGear for any Pre-Closing Tax Period (any such inquiry, claim, assessment, audit, proposed adjustment or similar event, a "Tax Proceeding"). Any failure to so notify the other party of any Tax Proceeding shall not relieve such other party of any liability with respect to such Tax Proceeding except to the extent such party was actually and materially prejudiced as a result thereof. Parent shall have control of the conduct of all Tax Proceedings; provided, however, that (i) the Stockholder Representative shall be entitled to participate in such Tax Proceedings at the expense of the Company Equityholders and consult with Parent regarding such Tax Proceedings and (ii) Parent shall not settle any Tax Proceeding with respect to Taxes for which the Company Equityholders would be liable hereunder without the consent of the Stockholder Representative, not to be unreasonably withheld, conditioned or delayed. In the event of any conflict between this Section 5.7.(a) and Section 7.2.(e) with respect to any Tax Proceedings, this Section 5.7.(a) shall control.

- (b) The Company shall prepare or cause to be prepared and file or cause to be filed all Tax Returns of the Company and PocketGear that are required to be filed on or before the Closing Date, and the Company or PocketGear shall pay, or cause to be paid, all Taxes of the Company and PocketGear due on or before the Closing Date. All such Tax Returns shall be prepared by treating items on such Tax Returns in a manner consistent with the past practices of the Company or PocketGear, as applicable, with respect to such items, except as required by applicable law. At least ten (10) days prior to filing any such Tax Return, the Company shall submit a copy of such Tax Return to Parent's review and approval, which approval shall not be unreasonably withheld, conditioned or delayed.
- (c) Parent shall prepare or cause to be prepared and file or cause to be filed all Tax Returns of the Company and PocketGear for any taxable periods ending on or before the Closing Date which are filed after the Closing Date. Such Tax Returns shall be prepared by treating items on such Tax Returns in a manner consistent with the past practices of the Company and PocketGear, as applicable, with respect to such items, except as required by applicable law. Parent shall provide copies of such Tax Returns at least twenty (20) days before the due date (including extensions, if applicable) thereof, and shall permit the Stockholder Representative to review and comment on each such Tax Return described in the preceding sentence prior to filing and shall make such revisions to such Tax Returns as are reasonably requested by the Stockholder Representative within fifteen (15) days after receiving such Tax Returns from Parent. Parent shall pay, or cause the Company to pay to the applicable Tax Authority, all Taxes of the Company and PocketGear, as applicable, due with respect to such Tax Returns. Parent shall recover from the Escrow Account the amount of such Taxes of the Company and PocketGear, in each case except to the extent that such Taxes are included in the calculation of Closing Working Capital that is reflected on the final Closing Statement.
- (d) Parent shall prepare or cause to be prepared and file or cause to be filed all Tax Returns of the Company and PocketGear for any Straddle Periods. Parent shall provide copies of such Tax Returns at least twenty (20) days before the due date (including any extension, if applicable) thereof, and shall permit the Stockholder Representative to review and comment on each such Tax Return described in the preceding sentence prior to filing and shall consider in good faith such revisions to such Tax Returns which relate to the portion of such Straddle Period ending on the Closing Date as are reasonably requested by the Stockholder Representative within fifteen (15) days after receiving such Tax Returns from Parent. The Parent shall pay, or cause to be paid, all Taxes of the Company or PocketGear, as applicable, due with respect to such Tax Returns. Parent shall recover from the Escrow Account the amount of such Taxes of the Company and PocketGear, in each case except to the extent that such Taxes are included in the calculation of Closing Working Capital that is reflected on the final Closing Statement. For purposes of the preceding sentence, Taxes shall be allocated in the manner set forth in the last sentence of Section 5.7(a) hereof.
- (e) Parent and the Company, on the one hand, and the Stockholder Representative and the Company Equityholders, on the other hand, agree to use commercially reasonable efforts to furnish or cause to be furnished to the other, upon request, as promptly as practicable, such information and assistance relating to Taxes, including, without limitation, access to Books and Records, in such party's possession, as is reasonably necessary for the filing of all Tax Returns by such parties, the making of any election relating to Taxes, and the preparation for any audit by any Tax Authority and the prosecution or defense of any claim, suit or proceeding relating to any Tax. Parent, the Company, the Stockholder Representative and the Company Equityholders agree (i) to retain all Books and Records with respect to Tax matters pertinent to the Company and PocketGear relating to any taxable period beginning before the Closing Date until the expiration of the statute of limitations (and, to the extent notified by Parent, any extensions thereof) of the respective taxable periods, and to abide by all record retention agreements entered into with any Tax Authority, (ii) to deliver or make available to Parent, within 60 days after the Closing Date, copies of all such Books and Records in such party's possession.
- (f) All transfer, documentary, sales, use, stamp, registration, value-added and other similar Taxes (including all applicable real estate transfer Taxes) incurred in connection with this Agreement and the transactions contemplated hereby ("<u>Transfer Taxes</u>") shall be borne fifty percent (50%) by the Company

Equityholders and fifty percent (50%) by Parent. The Parties shall cooperate with each other in connection with the filing by the parties required by applicable law to do so of any Tax Returns relating to Transfer Taxes, including joining in the execution of any such Tax Return or other documentation where necessary. Parent and such filing parties shall, upon request of the other party, use their reasonable best efforts to obtain any certificate or other document from any Person as may be necessary to mitigate, reduce or eliminate any Transfer Taxes. The Company Equityholders agree to file in a timely manner all necessary documents (including, but not limited to, all Tax Returns) with respect to all such amounts for which the Company Equityholders are so liable. Parent agrees to file in a timely manner all necessary documents (including, but not limited to, all Tax Returns) with respect to all such amounts for which the Parent is so liable. The Company Equityholders shall provide Parent with evidence satisfactory to Parent that such Transfer Taxes have been paid by the Company Equityholders. Parent shall provide Company Equityholders with evidence satisfactory to Company Equityholders that such Transfer Taxes have been paid by the Parent. If Parent or any of its Affiliates pays any Transfer Taxes for which the Company Equityholders are responsible under this Section 5.7.(f), Parent shall be entitled to recover out of the Escrow Account an amount equal to such Transfer Taxes and, if necessary, the Company Equityholders shall indemnify Parent for such amount to the extent provided in Article VI. If the Company Equityholders for such amount to the extent provided in Article VII.

- (g) All Tax sharing agreements or similar agreements between the Company or PocketGear, on the one hand, and any of the Company Equityholders and their Affiliates, on the other hand, shall be terminated prior to the Closing Date, and, after the Closing Date, neither the Company nor PocketGear shall be bound thereby or have any liability thereunder.
- (h) Each of the Company and Parent shall use it reasonable best efforts to cause the Merger to qualify as a "reorganization" within the meaning of Section 368(a) of the Code, including by not taking any action that such party knows could prevent such qualification. Unless otherwise required by applicable law, the parties will report the Merger and other transactions contemplated by this Agreement in a manner consistent with the treatment of the Merger as a "reorganization" within the meaning of Section 368(a) of the Code.
- (i) Any payments made to a party pursuant to the indemnification provisions of <u>Article VII</u> shall constitute an adjustment of the consideration paid hereunder for Tax purposes and shall be treated as such by the parties on their Tax Returns to the extent permitted by Regulations.

5.8. Employment Matters.

- (a) Unless Parent directs the Company otherwise in writing no later than one business day prior to the Effective Time, the Company Board of Directors shall adopt resolutions terminating, effective at least one day prior to the Effective Time, any Company Plan qualified under Section 401(a) of the Code and containing a Code Section 401(k) cash or deferred arrangement (each, a "401(k) Plan"). Prior to the Effective Time, the Company shall provide Parent with executed resolutions of its Board of Directors authorizing such termination, in a form reasonably approved by Parent. The Company shall also take such other reasonable actions in furtherance of the termination of each 401(k) Plan as Parent may reasonably require.
- (b) The parties acknowledge that the transactions contemplated under this Agreement shall not, by themselves, constitute a termination of employment of any employee of the Company.
- (c) If any Person who is a "disqualified individual" (within the meaning of Section 280G of the Code and the Department of Treasury regulations promulgated thereunder) with respect to the Company or PocketGear may receive any payment(s) or benefit(s) that could constitute parachute payments under Section 280G of the Code in connection with the transactions contemplated by this Agreement, then:
 (a) the Company shall obtain and deliver to Parent a Parachute Payment Waiver from each such "disqualified individual"; and (b) as soon as practicable following the delivery of the Parachute Payment Waivers (if any) to Parent, the Company shall prepare and distribute to its stockholders a disclosure statement describing all potential parachute payments and

benefits that may be received by such disqualified individual(s) and shall submit such payments to its shareholders for approval, in each case, in accordance with the requirements of Section 280G(b)(5)(B) of the Code and the Department of Treasury regulations promulgated thereunder, such that, if approved by the requisite majority of the stockholders, such payments and benefits shall not be deemed to be "parachute payments" under Section 280G of the Code (the foregoing actions, a "280G Vote"). Prior to the Closing, if a 280G Vote is required, the Company shall deliver to Parent evidence reasonably satisfactory to Parent, (i) that a 280G Vote was solicited in conformance with Section 280G of the Code, and the requisite stockholder approval was obtained with respect to any payments and/or benefits that were subject to the Company stockholder vote (the "Section 280G Approval") or (ii) that the Section 280G Approval was not obtained and as a consequence, pursuant to the Parachute Payment Waiver, such "parachute payments" shall not be made or provided. The form of the Parachute Payment Waiver, the disclosure statement, any other materials to be submitted to the Company's shareholders in connection with the Section 280G Approval and the calculations related to the foregoing (the "Section 280G Soliciting Materials") shall be subject to advance review and approval by Parent, which approval shall not be unreasonably withheld.

(d) Nothing in this Agreement shall, or shall be construed so as to (a) prevent or restrict in any way the right of Parent to terminate, reassign, promote or demote any employee, independent contractor, manager or other service provider of the Company (or to cause any of the foregoing actions) at any time, or to change (or cause the change of) the title, powers, duties, responsibilities, functions, locations, salaries, other compensation or terms or conditions of employment or service of any such service providers; (b) create any third-party rights in any such current or former service provider of the Company (or any beneficiaries or dependents thereof); (c) constitute an amendment or modification of any Plan, or (d) obligate Parent or any of its Affiliates to adopt or maintain any Plan or other compensatory or benefits arrangement at any time or prevent Parent or any of its Affiliates from modifying or terminating any Plan or any other compensatory or benefits arrangement at any time.

5.9. Preparation of S-4 and Stockholder Meetings.

(a) As promptly as reasonably practicable following the date of this Agreement (and in no event later than 20 Business Days after the date of this Agreement), (i) Parent and the Company shall, as promptly as practicable, prepare and file a joint proxy statement in preliminary form to be sent to the Company Equityholders relating to the Company Stockholder Meeting and to the stockholders of Parent relating to the Parent Stockholder Meeting (together with any amendments or supplements thereto (the "Joint Proxy Statement") and (ii) Parent shall prepare and cause to be filed with the SEC the Form S-4, in which the Joint Proxy Statement will be included as a prospectus. The Company shall include the Company Board Recommendation in such Joint Proxy Statement. Each of the Company and Parent shall use its reasonable best efforts to (i) have the Form S-4 declared effective under the Securities Act as promptly as practicable after such filing, (ii) ensure that the Form S-4 complies in all material respects with the applicable provisions of the Exchange Act or Securities Act and (iii) keep the Form S-4 effective for so long as necessary to complete the Merger. Each of the Company and Parent shall furnish all information concerning itself, its Affiliates and the holders of its capital stock to the other and provide such other assistance as may be reasonably requested in connection with the preparation, filing and distribution of the Form S-4, including the proxy statement and the obtaining of the consent of their respective auditors as required by the Form S-4. Parent and the Company shall use their reasonable best efforts to respond as promptly as practicable to any comments or requests for additional information from the SEC with respect to the Form S-4 or the time of effectiveness of the Form S-4.

(b) If, at any time prior to the receipt of the Company Stockholder Approval or Parent Stockholder Approval, any information relating to the Company or Parent, or any of their respective Affiliates, should be discovered by the Company or Parent which, in the reasonable judgment of the Company or Parent, should be set forth in an amendment of, or a supplement to, the Form S-4 or the Joint Proxy Statement, so that any of such documents would not include any misstatement of a material fact or omit to state any material fact necessary to make the statements therein, in light of the circumstances under which they were made, not misleading, the party

which discovers such information shall promptly notify the other parties hereto, and the Company and Parent shall cooperate in the prompt filing with the SEC of any necessary amendment of, or supplement to, the Form S-4 or Joint Proxy Statement, as applicable, and, to the extent required by Law, in disseminating the information contained in such amendment or supplement to Company Equityholders and stockholders of Parent. For purposes of Section 3.30, Section 4.7 and this Section 5.9.(b), any information concerning or related to the Company, its Affiliates or the Company Stockholder Meeting will be deemed to have been provided by the Company, and any information concerning or related to Parent or its Affiliates will be deemed to have been provided by Parent.

(c) As promptly as practicable after the date of this Agreement, the Company shall, in accordance with applicable Law and the Company Charter and the Company Bylaws, (i) establish a record date for, duly call, give notice of, convene and hold a meeting of the stockholders of the Company to obtain the Company Stockholder Approval (the "Company Stockholder Meeting") or (ii) seek the Company Stockholder Approval by written consent, which written consent shall be solicited and delivered promptly after the effectiveness of the Form S-4. The Company shall, through the Company's Board of Directors, recommend to the holders of Company Equityholders that they give the Company Stockholder Approval (the "Company Board Recommendation") and use its reasonable best efforts to obtain the Company Stockholder Approval. The Company shall use reasonable best efforts to cause the Joint Proxy Statement to be mailed to the Company Equityholders as promptly as practicable after the Form S-4 is declared effective under the Securities Act. Notwithstanding the foregoing provisions of this Section 5.9, if, on a date for which the Company Stockholder Meeting is scheduled, the Company has not received proxies representing a sufficient number of shares of Company Common Stock to obtain the Company Stockholder Approval, whether or not a quorum is present, the Company shall have the right to make one or more successive postponements or adjournments of the Company Stockholder Meeting; provided, that, the Company Stockholder Meeting is not postponed or adjourned to a date on or after the Outside Date. Unless this Agreement is terminated in accordance with Section 8.1, nothing contained in this Agreement shall be deemed to relieve the Company of its obligation to submit the approval of the Merger Agreement to its stockholders for a vote on the approval thereof.

(d) As promptly as practicable after the date of this Agreement, Parent shall, in accordance with applicable Law and Parent's certificate of incorporation and bylaws, establish a record date for, duly call, give notice of, convene and hold a meeting of the stockholders of Parent to obtain the Parent Stockholder Approval (the "Parent Stockholder Meeting"), regardless of whether or not a Change of Recommendation shall have occurred. Parent shall, through Parent's Board, of Directors, recommend to the stockholders of Parent that they give the Parent Stockholder Approval (the "Parent Board Recommendation"). Parent shall use reasonable best efforts to cause the proxy statement to be mailed to Parent's stockholders as promptly as practicable after the Form S-4 is declared effective under the Securities Act. Notwithstanding the foregoing provisions of this Section 5.9, if, on a date for which the Parent Stockholder Meeting is scheduled, Parent has not received proxies representing a sufficient number of shares of Parent Common Stock to obtain the Parent Stockholder Approval, whether or not a quorum is present, Parent shall have the right to make one or more successive postponements or adjournments of the Parent Stockholder Meeting; provided, that, the Parent Stockholder Meeting is not postponed or adjourned to a date on or after the Outside Date. Notwithstanding anything contained herein to the contrary, prior to the time the Parent Stockholder Approval is obtained, but not after, Parent's Board of Directors may change, qualify, withhold, withdraw or modify, or publicly propose to change, qualify, withhold, withdraw or modify, the Parent Board Recommendation ("Change of Recommendation") if but only if (i) Parent's Board of Directors determines in good faith, after consultation with Parent's outside legal counsel, that the failure to make the Change of Recommendation would constitute a breach of the fiduciary duties of the Board of Directors of Parent to Parent's stockholders under applicable Law, (ii) the Company shall have received from Parent prior written notice of Parent's intention to make a Change of Recommendation at least three (3) Business Days prior to making any Change of Recommendation (a "Change of Recommendation Notice") and (iii) Parent shall have complied with clauses (x) through (z) as follows: (x) the Change of Recommendation Notice shall have described the reasons for the Change of Recommendation in reasonable detail, (y) Parent shall have given the Company a three (3) Business Day period following the Company's receipt of the Change of Recommendation Notice to propose revisions to the terms of this Agreement or make other proposals and shall have negotiated in good faith

with the Company (and caused its Representatives to negotiate with the Company) with respect to such proposed revisions or other proposals, if any, and (z) after considering the results of negotiations with the Company and taking into account the proposals made by the Company, if any, after consultation with its outside legal counsel, Parent's Board of Directors shall have determined, in good faith, that the failure to make the Change of Recommendation would constitute a breach of the fiduciary duties of the Board of Directors of Parent to Parent's stockholders under applicable Law.

- (e) The Company and Parent will use their respective reasonable best efforts to hold the Company Stockholder Meeting and the Parent Stockholder Meeting on the same date and as soon as reasonably practicable after the date of this Agreement (unless the Company Stockholder Approval has been previously obtained by written consent).
- 5.10. <u>Listing</u>. Parent shall use its reasonable best efforts to cause the shares of Parent Common Stock to be issued in the Merger to be approved for listing on the NASDAQ, subject to official notice of issuance, prior to the Effective Time.
- 5.11. <u>Public Announcements</u>. On and after the date hereof and through the Closing Date (including in connection with any termination of this Agreement), the Company and PocketGear shall not issue any press release or otherwise make any public statements with respect to this Agreement or the transactions contemplated hereby without the prior written consent of parent, unless required by applicable Law upon the advice of legal counsel. The initial press release of Parent regarding the transactions contemplated hereby, to be released as promptly as practicable following the execution of this Agreement shall be substantially in the form agreed to by the Company and Parent prior to the execution of this Agreement.
- 5.12. <u>Consents; Notices</u>. Prior to the Closing. the Company shall use reasonable best efforts to obtain all necessary consents, waivers and approvals of any third parties to any Material Contract, Facility Lease or Permit as are required thereunder in connection with the Merger or for any such Material Contracts, Facility Leases or Permits as set forth in <u>Schedule 3.5</u> of the Disclosure Schedules. Such consents, waivers and approvals shall be in a form acceptable to Parent. The Company shall send each of the notices described on <u>Schedule 3.5</u> of the Disclosure Schedules promptly following the execution of this Agreement.
- 5.13. <u>Termination of Certain Agreements</u>. Prior to the Effective Time, the Company shall take all actions necessary to terminate as of the Effective Time, and shall cause to be terminated as of the Effective Time, the Contracts listed on <u>Schedule 5.13</u> of the Disclosure Schedules (collectively, the "<u>Terminated Agreements</u>"), in each case without any further liability to or obligation of the Company or the Surviving Corporation.

5.14. Director and Officers Indemnification.

(a) For a period of six (6) years following the Effective Time, the Surviving Corporation shall honor in all respects the obligations of the Company with respect to all rights to indemnification (including advancement of expenses) or exculpation existing in favor of, and all limitations on the personal liability of, any Person who is now, or has been at any time prior to the date hereof, or who becomes prior to the Effective Time, a director or officer of the Company or of Pocketgear under the Organizational Documents or director or officer of the Company in any indemnification agreements in effect as of the date hereof and set forth in Schedule 5.14 of the Disclosure Schedule (each, a "D&O Indemnified Party" and collectively, the "D&O Indemnified Parties"). Notwithstanding the foregoing, the obligations of the Surviving Corporation (i) shall be subject to any limitation imposed by applicable Regulation and (ii) shall be deemed not to release any D&O Indemnified Party from his or her obligations pursuant to this Agreement or any Ancillary Agreement, nor shall such D&O Indemnified Party have any right of contribution, indemnification or right of advancement from Parent, the Surviving Corporation or any of their Affiliates with respect with any Damages claimed by any of the Parent Indemnitees against such D&O Indemnified Party solely in his or her capacity as a Company Equityholder pursuant to this Agreement or any Ancillary Agreement. The Company hereby represents to Parent that no claim for indemnification has been made as of the date hereof by any director or officer of the Company.

- (b) Prior to the Effective Time, the Company shall purchase (and pay in full all premiums on) an extended reporting period endorsement under the Company's existing directors' and officers' liability insurance coverage for the Company's directors and officers that shall provide such directors and officers with coverage for six (6) years following the Effective Time; provided, however, that to the extent any premiums with respect to such policy have not been paid in full by the Company prior to the Effective Time, then such payments shall be included in the calculation of Company Transaction Expenses.
- (c) The current and former directors, officers, employees and agents of the Company entitled to the indemnification, liability limitation, exculpation and insurance set forth in this <u>Section 5.14</u> are intended to be third party beneficiaries of this <u>Section 5.14</u>. This <u>Section 5.14</u> shall survive the consummation of the transactions contemplated by this Agreement and shall be binding on all successors and assigns of Parent.
- 5.15. Financing. Prior to the Closing. the Company shall use reasonable best efforts to, and shall cause PocketGear to use reasonable best efforts to, cooperate with Parent (and use reasonable best efforts to cause the independent accounting firm and other advisers retained by the Company to cooperate with Parent) in connection any equity and/or debt financing by Parent (whether or not directly related to the Transactions contemplated by this Agreement) (collectively, the "Financing"), including, without limitation, (i) facilitating customary due diligence, (ii) arranging for members of senior management of the Company and PocketGear to participate in meetings, presentations, road shows, due diligence sessions (or other sessions with prospective lenders, investors and rating agencies), drafting sessions, sessions with rating agencies or other syndication activities reasonably required in order to satisfy the conditions to consummation of the Financing, (iii) assisting Parent and its financing sources with the preparation of customary materials for rating agency presentations, bank information memoranda (including customary authorization and representation letters) and other written offering materials used to syndicate such Financing, to the extent information contained therein relates to the Company or PocketGear, (iv)(A) assisting with the preparation of any pledge and security documents, any loan agreement, currency or interest hedging agreement, and other definitive financing documents and using reasonable best efforts to assist in the negotiation and execution of collateral related documentation involving third parties and (B) providing documents requested by Parent or its financing sources relating to the existing indebtedness of the Company and the release of related Liens, including customary payoff letters, (v) facilitating the pledging of collateral and the granting of security interests in collateral in connection with the Financing, provided that no pledge or grant of security interests shall be effective until the Closing, and (vi) furnishing to Parent and its financing sources as promptly as practicable such financial and other pertinent information regarding the Company and its subsidiaries as may be reasonably requested by Parent to satisfy the conditions to consummation of the Financing, including, without limitation, applicable "know your customer" information. Without limiting the foregoing, the Company and Parent shall each use reasonable best efforts to cooperate with each other to cause the deliverables described in Section 2.2(a)(vi) and Section 2.2(b)(viii) to be finalized and delivered.

5.16. <u>Tax Representation Letters</u>.

- (a) The Company shall use its reasonable best efforts to (i) obtain the opinions to be attached as exhibits to Parent's Joint Proxy Statement and Form S-4 and the opinions of counsel referred to in Article VI, and (ii) deliver to Goodwin Procter LLP, counsel to the Company, and Latham & Watkins LLP, counsel to Parent, tax representation letters, dated as of the effective date of the Form S-4 and the Closing Date, respectively, and signed by an officer of the Company, in form and substance as set forth in Schedule 5.14(a) with such changes as are mutually agreeable to the Company and Parent, containing representations of the Company as shall be reasonably necessary or appropriate to enable Goodwin Procter LLP, to render an opinion on the effective date of the Form S-4 and on the Closing Date, as described in Article VI, respectively, and Latham & Watkins LLP to render an opinion on the effective date of the Form S-4 and on the Closing Date, as described in Section 6.2.(n).
- (b) Parent shall use its reasonable best efforts to (i) obtain the opinions to be attached as exhibits to Parent's Joint Proxy Statement and Form S-4 and the opinions of counsel referred to in <u>Article VI</u>, and

(iii) deliver to Goodwin Procter LLP, counsel to the Company, and Latham & Watkins LLP, counsel to Parent, tax representation letters, dated as of the effective date of the Form S-4 and the Closing Date, respectively, and signed by an officer of Parent, in form and substance as set forth in Schedule 5.14(b) with such changes as are mutually agreeable to the Company and Parent, containing representations of Parent as shall be reasonably necessary or appropriate to enable Latham & Watkins LLP to render an opinion on the effective date of the Form S-4 and on the Closing Date, as described in Article VI, respectively, and Goodwin Procter LLP, to render an opinion on the effective date of the Form S-4 and on the Closing Date, as described in Section 6.1.(e).

5.17. Transfer Restrictions with respect to Parent Common Stock.

- (a) All shares of Parent Common Stock to be issued in connection with the Merger shall be subject to restrictions on the transfer of such shares during the period specified herein (the "Lock-Up Period"). Each Company Equityholder agrees (whether by signing the Support Agreement or any Letter of Transmittal or by accepting such Parent Common Stock), that such Company Equityholder will not offer, sell, contract to sell, pledge, grant any option to purchase, make any short sale or otherwise dispose of any shares of Parent Common Stock, or any options or warrants to purchase any shares of Parent Common Stock, or any securities convertible into, exchangeable for or that represent the right to receive shares of Parent Common Stock, whether now owned or hereinafter acquired, owned directly by a Company Equityholder (including holding as a custodian) or with respect to which any Company Equityholder has beneficial ownership within the rules and regulations of the SEC (collectively the "Subject Shares"). The foregoing restriction is expressly agreed to preclude the Company Equityholders from engaging in any hedging or other transaction which is designed to or which reasonably could be expected to lead to or result in a sale or disposition of a Company Equityholder's Subject Shares even if such Subject Shares would be disposed of by someone other than the applicable Company Equityholder. Such prohibited hedging or other transactions would include without limitation any short sale or any purchase, sale or grant of any right (including without limitation any put or call option) with respect to any of a Company Equityholder's Subject Shares or with respect to any security that includes, relates to, or derives any significant part of its value from such Subject Shares. Prior to the date hereof or substantially concurrently herewith, certain Company Equityholders have entered into or are each entering into the Support Agreement containing substantially the same or more restrictive than the restrictions during the Lock-Up Period as the restrictions set forth herein (it being acknowledge, in the event of any conflict between the provisions of this Section 5.17 and the Support Agreement, the terms of the Support Agreement shall control).
- (b) Except as otherwise provided in the Support Agreement with respect to specific Company Equityholders, the Lock-Up Period will commence on the Effective Date (the "Commencement Date") and expire (i) 180 days following the Commencement Date with respect to 33 and 1/3% of each Company Equityholder's Subject Shares, (ii) 270 days following the Commencement Date with respect to 33 and 1/3% of each Company Equityholder's Subject Shares and (iii) 365 days following the Commencement Date with respect to 33 and 1/3% of each Company Equityholder's Subject Shares.
- (c) Notwithstanding the foregoing, each Company Equityholder may transfer the Company Equityholder's Subject Shares (i) as a bona fide gift or gifts, provided that the done or donees thereof agree to be bound in writing by the restrictions set forth herein, (ii) to any trust for the direct or indirect benefit of the applicable Company Equityholder or the immediate family of such Company Equityholder, provided that the trustee of the trust agrees to be bound in writing by the restrictions set forth herein, and provided further that any such transfer shall not involve a disposition for value or (iii) to Parent in connection with the settlement of indemnification claims pursuant to Article VII. For purposes of this Agreement, "immediate family" shall mean any relationship by blood, marriage or adoption, not more remote than first cousin. In addition, notwithstanding the foregoing, if a Company Equityholder is a corporation, limited liability company, partnership or trust it may transfer the Subject Share to any wholly-owned subsidiary of such corporation, limited liability company or partnership or to the stockholders, partners, members or beneficiaries of such Company Equityholder, provided, however, that in any such case, it shall be a condition to the transfer that (i) the transferee execute an agreement stating that the transferee is receiving and holding such

Subject Shares subject to the provisions of this Section 5.17 and there shall be no further transfer of such Subject Shares except in accordance with this Section 5.17 (and, for the avoidance of doubt, the provisions of this Section 5.17, including the exception to transfer restrictions shall apply to any such transferee as if the transferee were the original recipient of the Subject Shares hereto), (ii) that any such transfer shall not involve a disposition for value, and (iii) that no public announcement or filing under the Exchange Act shall be required of or voluntarily made by or on behalf of the applicable Company Equityholder or Parent in connection with any such transfer. In addition, notwithstanding anything herein to the contrary, nothing herein shall prevent a Company Equityholder from establishing a written plan meeting the requirements of Rule 10b5-1 under the Exchange Act (a "10b5-1 plan"), relating to the sale of such Company Equityholder's Subject Shares, provided that (i) such shares may not be sold until after 180 days following the Commencement Date, (ii) the aggregate number of shares sold by such Company Equityholder under a 10b5-1 plan or otherwise shall not exceed the limitations set forth in Section 5.17(b) and (iii) the establishment of such plan will not result in any public filing or other public announcement of such Company Equityholder during the 180 days following the Commencement Date.

- (d) Parent shall instruct the Exchange Agent to include restrictions on the transfer of the Subject Shares consistent with those set forth in this <u>Section 5.17</u> and any certification evidencing the Subject Shares shall include legends reflecting that transfers of the Subject Shares are subject to the restrictions set forth in this <u>Section 5.17</u> and the Support Agreement.
- 5.18. <u>Documents and Information</u>. After the Closing Date, Parent shall, and shall cause the Company to, until the fifth anniversary of the Closing Date, retain all books, records and other documents pertaining to the business of the Company in existence on the Closing Date and make the same available for inspection and copying by the Stockholder Representative during normal business hours of the applicable Company, upon reasonable request and upon reasonable notice.
- 5.19. Other Agreements of Parent. Parent agrees to pay the Company Transaction Expenses described in clause (a) of the definition of Company Transaction Expenses that remain unpaid as of the Effective Time within one (1) business day following the Closing Date so long as the Company has delivered to Parent invoices and payment instructions for such Company Transaction Expenses (and reasonable documentation underlying such Company Transaction Expenses) with the delivery of the Pre-Closing Statement. In addition, for the avoidance of doubt, Parent will honor the indemnification obligations under the indemnification agreements set forth on Schedule 5.14 of the Disclosure Schedule to any Person other than the officer or director to the extent such other Person is entitled to indemnification thereunder (for the avoidance of doubt, the indemnification of directors and officers is covered by Section 5.14).

Article VI. CONDITIONS TO CLOSING

- 6.1. <u>Conditions to Obligations of the Company</u>. The obligations of the Company to consummate the transactions provided for hereby are subject, in the discretion of the Company, to the satisfaction, on or prior to the Closing Date, of each of the following conditions, any of which may be waived by the Company:
- (a) Representations, Warranties and Covenants. (i) All representations and warranties of Parent and Merger Sub contained in this Agreement (A) that are qualified by materiality or words of similar import shall be true and correct in all respects, as of the date hereof and at and as of the Closing, except that such representations and warranties that are made as of a specific date need only be true and correct as of such date, and (B) that are not qualified by materiality shall be true and correct in all material respects at and as of the date of this Agreement and at and as of the Closing Date, except that such representations and warranties that are made as of a specific date need only be true and correct as of such date and (B) Parent and Merger Sub shall have performed and satisfied in all material respects all agreements and covenants required hereby to be performed by Parent and Merger Sub prior to or on the Closing Date.

- (b) No Actions or Court Orders. No suit, Action, investigation, inquiry or other proceeding by any Governmental Authority or other Person or legal or administrative proceeding shall have been instituted or threatened which seeks to restrain, enjoin, prevent the consummation of or otherwise affect the transactions contemplated by this Agreement or the Ancillary Agreements or which questions the validity or legality of the transactions contemplated hereby or thereby
- (c) <u>Registration Statement</u>. The Form S-4 shall have become effective in accordance with the provisions of the Securities Act. No stop order suspending the effectiveness of the Form S-4 shall have been issued by the SEC and remain in effect and no proceeding to that effect shall have been commenced or threatened.
- (d) <u>Listing.</u> The shares of Parent Common Stock to be issued in the Merger shall have been approved for listing on the NASDAQ, subject to official notice of issuance.
- (e) Section 368 Opinion. The Company shall have received a written opinion of its counsel, Goodwin Procter LLP dated as of the Closing Date and in form and substance as set forth in Exhibit G, and with such changes as are mutually agreeable to the Company and Parent, to the effect that, on the basis of facts, representations and assumptions set forth in such opinion, the Merger will qualify as a reorganization within the meaning of Section 368(a) of the Code; provided, however, that the condition set forth in this Section 6.1(e) shall not be waivable by the Company after adoption of this Agreement by the holders of Company Common Stock and Company Preferred Stock, unless further stockholder approval is obtained with appropriate disclosure. In rendering such opinion, counsel may rely upon the Company Tax Representation Letter and the Parent Tax Representation Letter.
- (f) <u>Antitrust and Competition Laws</u>. Any waiting period expirations or terminations, consents, clearances, waivers and approvals under any applicable antitrust, competition or merger control laws shall have been obtained, and there is no order or injunction prohibiting, limiting or preventing the consummation of the transactions contemplated herein pursuant to any antitrust, competition or merger control law.
- (g) <u>Ancillary Agreements</u>. Parent and Merger Sub shall have executed and delivered the Ancillary Agreements to which Parent and Merger Sub are parties.
 - (h) Company Stockholder Approval. The Company Stockholder Approval shall have been obtained.
- (i) <u>Stockholder Consent and Approval</u>. Parent shall have delivered to the Company evidence of (i) the adoption and approval by Parent, as the sole stockholder of Merger Sub, of this Agreement, the Merger and the other transactions contemplated hereby in accordance with the DGCL and Parent's Organizational Documents and (ii) the Parent Stockholder Approval.
 - (j) Parent Stockholder Approval. The Parent Stockholder Approval shall have been obtained.
- (k) $\underline{\text{Other Deliveries}}$. Parent and Merger Sub shall have delivered to the Company, each of the items set forth in $\underline{\text{Section 2.2.(a)}}$ hereof.
- (1) <u>Compliance with Laws</u>. The consummation of the transactions contemplated by this Agreement will not be prohibited by any applicable Regulations.
- 6.2. <u>Conditions to Obligations of Parent and Merger Sub</u>. The obligations of Parent and Merger Sub to consummate the transactions provided for hereby are subject, in the discretion of Parent and Merger Sub, to the satisfaction, on or prior to the Closing Date, of each of the following conditions, any of which may be waived by Parent and Merger Sub:
- (a) <u>Representations, Warranties and Covenants</u>. All representations and warranties of the Company contained in this Agreement (A) that are qualified by materiality or Material Adverse Effect or words of similar

import shall be true and correct in all respects, as of the date hereof and at and as of the Closing, except that such representations and warranties that are made as of a specific date need only be true and correct as of such date, and (B) that are not qualified by materiality or Material Adverse Effect shall be true and correct in all material respects at and as of the date of this Agreement and at and as of the Closing Date, except that such representations and warranties that are made as of a specific date need only be true and correct as of such date and (B) the Company shall have performed and satisfied in all material respects all agreements and covenants required hereby to be performed by the Company prior to or on the Closing Date.

- (b) <u>Consents; Regulatory Compliance and Approval</u>. All third-party consents set forth on <u>Schedule 6.2(b)</u> and all consents, approvals or authorizations from Governmental Authorities necessary for the valid consummation of the transactions contemplated hereby shall have been obtained.
- (c) No Actions or Court Orders. No suit, Action, investigation, inquiry or other proceeding by any governmental body or other Person or legal or administrative proceeding shall have been instituted or threatened which seeks to restrain, enjoin, prevent the consummation of or otherwise affect the transactions contemplated by this Agreement or the Ancillary Agreements or which questions the validity or legality of the transactions contemplated hereby or thereby.
- (d) <u>Registration Statement</u>. The Form S-4 shall have become effective in accordance with the provisions of the Securities Act. No stop order suspending the effectiveness of the Form S-4 shall have been issued by the SEC and remain in effect and no proceeding to that effect shall have been commenced or threatened.
- (e) <u>Antitrust and Competition Laws</u>. Any waiting period expirations or terminations, consents, clearances, waivers and approvals under any applicable antitrust, competition or merger control laws shall have been obtained, and there is no order or injunction prohibiting, limiting or preventing the consummation of the transactions contemplated herein pursuant to any antitrust, competition or merger control law.
- (f) <u>Compliance with Laws</u>. The consummation of the transactions contemplated by this Agreement will not be prohibited by any applicable Regulations.
- (g) <u>Company Stockholder Approval.</u> The Company shall have delivered to Parent evidence of the Company Stockholder Approval (which approval shall include the approval or written consent of the stockholders that are party to the Support Agreement).
 - (h) Parent Stockholder Approval. The Parent Stockholder Approval shall have been obtained.
- (i) <u>Dissenting Stockholders</u>. Stockholders holding not more than 5% of the Company Capital Stock outstanding immediately prior to the Effective Time (or on an as-converted to Company Common Stock basis) shall have exercised, and shall not have any continued right to exercise, appraisal, dissenters' or similar rights under any applicable Law with respect to shares of capital stock of the Company held by them.
- (j) <u>Non-competition Agreements</u>. The Non-competition Agreement shall be in full force and effect and shall not have been repudiated by any party thereto.
- (k) <u>401(k) Plan Termination</u>. The Company shall have provided Parent with executed resolutions of its Board of Directors authorizing such termination, in a form reasonably approved by Parent.
- (1) <u>Section 280G Matters</u>. If a 280G Vote is required under <u>Section 5.8.(c)</u> hereof, (i) the Company shall have received and delivered to Parent a Parachute Payment Waiver from each Person that is eligible to receive a payment that may constitute a "parachute payment" under Section 280G of the Code prior to soliciting

the Section 280G Approval and (ii) the Company's stockholders shall have (A) approved, pursuant to the method provided for in the regulations promulgated under Section 280G of the Code, any such "parachute payments" or (B) shall have voted upon and disapproved such "parachute payments," and, as a consequence, such "parachute payments" shall not be paid or provided for in any manner and Parent and its Affiliates shall not have any Liabilities with respect to such "parachute payments."

- (m) <u>Listing.</u> The shares of Parent Common Stock to be issued in the Merger shall have been approved for listing on the Nasdaq, subject to official notice of issuance.
- (n) Section 368 Opinion. Parent shall have received the written opinion of its counsel, Latham & Watkins LLP, dated as of the Closing Date and in form and substance as set forth in Exhibit H, and with such changes as are mutually agreeable to Parent and the Company, to the effect that, on the basis of facts, representations and assumptions set forth in such opinion, the Merger will qualify as a reorganization within the meaning of Section 368(a) of the Code; provided, however, that the condition set forth in this Section 6.2(n) shall not be waivable by Parent or Merger Sub after adoption of this Agreement by the holders of Company Common Stock and Company Preferred Stock, unless further stockholder approval is obtained with appropriate disclosure. In rendering such opinion, Latham & Watkins LLP may rely upon the Company Tax Representation Letter and the Parent Tax Representation Letter.
 - (o) Material Changes. Since the Balance Sheet Date, there shall not have been any Material Adverse Change.
- (p) <u>Ancillary Agreements</u>. Each Company Stockholder shall have executed and delivered each Ancillary Agreement to which such Company Stockholder is to be a party.
- (q) <u>Financing Condition</u>. Each of the agreements instruments and other deliverables referenced in <u>Section 2.2(a)(vi)</u> (and Exhibit D) and <u>Section 2.2(b)(viii)</u> (and Exhibit E) shall have been executed and delivered by the applicable lenders and other parties thereto (other than Parent) and such agreements, instruments and other deliverables shall be in full force and effect, and no event of default or similar event shall exist under the Existing Credit Arrangements or will exist upon entry into the financing agreements that will amend and restate the Existing Credit Arrangements.
- (r) Net Debt. If the Closing Date is on or before January 31, 2015, Net Debt as of the Closing shall not be greater than \$13,750,000. If the Closing Date is after January 31, 2015 but prior to the February 28, 2015 then Net Debt as of the Closing shall not be greater than \$14,500,000. If the Closing Date is after February 28, 2015, but prior to the Outside Date then Net Debt as of the Closing shall not be greater than \$15,000,000.
 - (p) $\underline{Other\ Deliveries}$. Parent shall have received from the Company each of the items set forth in $\underline{Section\ 2.2.(b)}$ hereof.

ARTICLE VII. INDEMNIFICATION; REMEDIES

7.1. <u>Survival of Representations, Etc.</u> All of the representations and warranties made by any party in this Agreement or in any attachment, Exhibit, Schedule, the Disclosure Schedules, certificate, document or list delivered by any such party pursuant hereto shall survive the Closing for a period of twelve months following the Closing Date; provided, however, that (i) (a) the representations and warranties set forth in <u>Sections 3.1</u> (Organization), <u>3.2</u> (Subsidiaries), <u>3.3</u> (Capitalization), <u>3.4</u> (Authorization), <u>3.29</u> (No Brokers), <u>4.1</u> (Organization), <u>4.2</u> (Authorization) and <u>4.5</u> (No Brokers) shall survive for six (6) years following the Closing Date, and (b) the representations and warranties set forth in <u>Section 3.22</u> (Tax Matters) shall survive until thirty (30) days following the expiration of the applicable statute of limitations under applicable Tax law with respect

to the matters addressed in such sections (such representations and warranties set forth in clauses (a) and (b), collectively, the "<u>Fundamental Representations</u>") and (ii) the representations and warranties set forth in <u>Section 3.18</u> (Intellectual Property) shall survive until thirty (30) months following the Closing Date ("<u>IP Representations</u>"). Claims based upon or arising out of any such representations and warranties may be asserted at any time before the expiration date of the applicable representations and warranties. The Company Stockholders shall be entitled to rely upon the representations and warranties of Parent and Merger Sub set forth in this Agreement. Parent and Merger Sub shall be entitled to rely upon the representations and warranties of the Company set forth in this Agreement. The expiration of the representations and warranties provided herein shall not affect the rights of a Parent Indemnitee in respect of any Claim made by such Parent Indemnitee that is submitted prior to the expiration of the applicable survival period provided herein, with respect to the matters described in the Claim Notice for such Claim. All of the covenants, agreements and obligations of the parties under this Agreement to be performed after the Closing will survive the Closing in accordance with their respective terms, subject to any applicable statute of limitations.

7.2. Indemnification.

(a) Company Equityholders' Agreement to Indemnify. Each of the Company Equityholders, severally and not jointly, shall indemnify, save and hold harmless Parent, Merger Sub, the Company and each of Parent's direct and indirect stockholders, Affiliates and Subsidiaries (including, following the Closing, the Surviving Corporation), and each of their respective Representatives (collectively, the "Parent Indemnitees"), from and against any and all costs, losses, Taxes, Liabilities, obligations, damages, lawsuits, deficiencies, claims, demands, expenses (whether or not arising out of third-party claims), reasonable attorneys' fees and all amounts paid in investigation, defense or settlement of any of the foregoing (herein, "Damages"), incurred in connection with, arising out of, resulting from or incident to (i) any breach of any representation or warranty or the inaccuracy of any representation made by the Company in or pursuant to Article III of this Agreement (in each case without giving effect to materiality qualifications or Material Adverse Effect qualifications with respect to the calculation of Damages incurred by a Parent Indemnitee); (ii) any breach of any covenant or agreement made by the Company in or pursuant to this Agreement; (iii) any claim by any Person for brokerage or finder's fees or commissions or similar payments based upon any agreement or understanding alleged to have been made by any such Person with the Company or any Company Stockholder (or any Person acting on their behalf) in connection with any transactions contemplated by this Agreement; (iv) any Long Term Liabilities that are unpaid as of the Closing and not taken into account in the final calculation of Closing Net Debt on the Closing Statement; (v) any Company Transaction Expenses that are unpaid as of the Closing and not taken into account in the final calculation of Net Debt on the Closing Statement; (vi) any Excess Dissenting Share Payments; (vii) any Claim or allegation that any Company Equityholder or other Person is entitled to any amount in connection with the Merger (other than, with respect to a Company Stockholder, the consideration allocable to such Company Stockholder in accordance with Section 1.6 hereof, with respect to a Series C Warrantholder, the consideration allocable to such Series C Warrantholder in accordance with Section 1.7, and, with respect to a Company Optionholder, the consideration allocable to such Company Optionholder in accordance with Section 1.8, in each case as set forth in the Closing Consideration Schedule); or (viii) (I) any Taxes of the Company or PocketGear with respect to any Pre-Closing Tax Period except to the extent such Taxes are included in the calculation of Working Capital that is reflected on the final Closing Statement, (II) any Transfer Taxes allocated to the Company Equityholders pursuant to Section 5.7.(f); (III) any Taxes of any Person (other than the Company) for which the Company or PocketGear is held liable under Treasury Regulations Section 1.1502-6 (or any similar provision of state, local or foreign law), by reason of such entity being included in any consolidated, affiliated, combined or unitary group at any time on or before the Closing Date, or as a transferee or successor, by Contract, or otherwise; (IV) Taxes imposed on or payable by third parties with respect to which the Company or PocketGear has an obligation to indemnify such third party pursuant to a transaction consummated prior to the Closing; and (V) all Taxes attributable to any inclusion under Section 951(a) of the Code by the Parent Indemnitees at the end of the taxable year of a non-U.S. Subsidiary that includes the Closing Date allocable to the Pre-Closing Tax Period as determined pursuant to the following sentence of this Section 7.2 or (ix) the matter set forth on Schedule 7.2(a)(ix) of the Disclosure Schedule. With respect to any Straddle Period, in the case of any

Taxes that are imposed on a periodic basis, the portion of such Tax that relates to the portion of such Straddle Period that is deemed to end on the Closing Date shall (A) in the case of Property Taxes, deemed to be the amount of such Taxes for the entire Straddle Period multiplied by a fraction (1) the numerator of which is the number of days of such Straddle Period in the Pre-Closing Tax Period and (2) the denominator of which is the number of days in the entire Straddle Period, and (B) in the of all other Taxes, determined as though the taxable year terminated at the close of business on the Closing Date. Notwithstanding anything to the contrary herein, the Company makes no representations or warranties with respect to the ability of Parent or any of its Affiliates to utilize net operating losses of the Company after the Closing.

For purposes of clarifying the meaning of "several" indemnification by each Company Equityholder under this Section 7.2(a), (x) any portion of the Escrowed Account that is distributed to Parent pursuant to this Article VII shall be deemed to have been "severally" recovered from all of the Company Equityholders, and (y) with respect to Claims for indemnification under this Section 7.2(a) that are not recovered from the Escrow Account, (i) if any indemnification payment is owed, each Company Equityholder's indemnification obligation pursuant to this Article VII shall be equal to (and limited to) such Company Equityholder's Pro Rata Portion of the applicable Damages with respect to which the indemnification payment is made (except in the case of fraud (which shall be interpreted to require the element of scienter) committed by a Company Equityholder and (ii) such Claim shall be subject to Section 7.2.(h).

- (b) By Parent. Parent shall indemnify, save and hold harmless the Company Equityholders (collectively, the "Company Equityholder Indemnitees"), from and against any and all Damages incurred in connection with, arising out of, resulting from or incident to (i) any breach of any representation or warranty or the inaccuracy of any representation made by Parent or Merger Sub in or pursuant to Article III of this Agreement (in each case without giving effect to materiality qualifications or Material Adverse Effect qualifications with respect to the calculation of Damages incurred by a Company Equityholder Indemnitee); or (ii) any breach of any covenant or agreement made by Parent or Merger Sub in or pursuant to this Agreement. For the avoidance of doubt, the Surviving Corporation shall not be considered a Company Equityholder Indemnitee for purposes of this Article VII.
- (c) <u>Cooperation</u>. The indemnified party and the indemnifying party shall cooperate with each other and their attorneys in the investigation, trial and defense of any lawsuit or action arising from third party claims and any appeal arising therefrom; <u>provided</u>, <u>however</u>, that the party not controlling the defense of such lawsuit or action may, at its own cost, participate in (but not control) the investigation, trial and defense of such lawsuit or action and any appeal arising therefrom. The parties shall cooperate with each other in any notifications to insurers.
- (d) Indemnifying Parties. References in this Article VII to "indemnifying party" when a Parent Indemnitee is an indemnified party shall be deemed to be references to the Company Equityholders collectively from and after the Closing (and to the Company Equityholders and the Company prior to the Closing) and, for the avoidance of doubt, the Stockholder Representative shall represent the Company Equityholders with respect to all such matters as provided in Section 7.2.(g) except for any right or obligation to receive or make payment (and any notice requirement with respect to any notice required to be provided under this Article VII by an indemnified party shall be deemed satisfied if such notice is delivered to the Stockholder Representative).

(e) Procedure for Indemnification; Defense of Third-Party Claims.

(i) If an indemnification claim for Damages pursuant to this <u>Article VII</u> (a "<u>Claim</u>") is to be made by an indemnified party hereunder for any matter not involving a third party claim, such claim shall be asserted by the indemnified party by delivery of a written notice describing the claim with reasonable specificity, a good faith estimate of the amount thereof (if known) and the basis thereof (a "<u>Claim Notice</u>") to the indemnifying party as soon as reasonably practicable after the indemnified party becomes aware of any fact, condition or event which is reasonably likely to give rise to Damages for which indemnification may be sought under this <u>Article VII</u>, <u>provided</u>, that the failure of the indemnified party to give prompt notice hereunder shall not affect rights to

indemnification hereunder, except to the extent that the indemnifying party demonstrates actual damage caused by such failure. Following the delivery of a Claim Notice, the indemnifying party and its Representatives shall be given access (including electronic access, to the extent available) at reasonable times as they may reasonably require to the Books and Records of the Company and Pocket Gear and access to such personnel or representatives of the Company and its Representatives, including but not limited to the individuals responsible for the matters that are subject of the Claim Notice, at reasonable times as they may reasonably require for the purposes of investigating or resolving any disputes or responding to any matters or inquiries raised in the Claim Notice; <u>provided</u> that the indemnified party shall be entitled to withhold information from the indemnifying party if its provision to the indemnifying party would cause the attorney-client privilege thereof to be waived and there is no method of providing such information to the indemnifying party in a manner which would not result in such a waiver.

(ii) If a Claim is to be made by an indemnified party as a result of a third-party claim (a "Third Party Claim"), the indemnified party shall, subject to this Article VII, give a Claim Notice to the indemnifying party with respect to such Claim as soon as practicable after the indemnified party becomes aware of any fact, condition or event which may give rise to Damages for which indemnification may be sought under this Article VII, provided, that the failure of the indemnified party to give prompt notice hereunder shall not affect the rights to indemnification hereunder, except to the extent that the indemnifying party demonstrates actual damage caused by such failure. If any lawsuit or enforcement action is filed by a third party against an indemnified party, written notice thereof shall be given to the indemnifying party as promptly as practicable, provided that the failure of the indemnified party to give prompt notice hereunder shall not affect rights to indemnification hereunder, except to the extent that the indemnifying party demonstrates actual damage caused by such failure. The Stockholder Representative shall be entitled, if it so elects at the cost and expense of the Company Equityholders, to participate in the defense of such claim and consult with Parent in any defense of such claim and Parent shall consider in good faith any reasonable comments or recommendations of the Stockholder Representative with respect to the defense of such claim, it being understood that Parent shall have the sole right to control such defense (including the right to settle any such claim, subject to Section 7.2(e)(iii), if applicable); provided, further, that the parties shall cooperate in good faith to implement reasonable arrangements designed to preserve any existing attorney-client privilege; provided, further, that the indemnified party shall be entitled to withhold information from the indemnifying party if its provision to the indemnifying party would cause the attorney-client privilege thereof to be waived and there is no method of providing such information to the indemnifying party in a manner which would not result in such a waiver.

(iii) Except for third-party claims (A) seeking non-monetary remedy, (B) seeking claims for equitable relief, or (C) involving potential criminal liability, if the Stockholder Representative provides written notice to Parent within 30 days after receipt of a Claim Notice in respect of a third-party claim for which a Parent Indemnitee is seeking indemnification hereunder, which notice by the Stockholder Representative irrevocably acknowledges the Company Equityholders' obligation to indemnify such claim fully, a Parent Indemnitee shall not enter into a settlement of such third-party claim without the prior written consent of the Stockholder Representative (such consent not to be unreasonably withheld, conditioned or delayed).

(iv) After the giving of any Claim Notice pursuant hereto, the amount of indemnification to which an indemnified party shall be entitled under this Article VII shall be determined: (a) by the written agreement between Parent and the Stockholder Representative; (b) by a decision of the Arbitrator as contemplated by Section 9.16; or (c) by any other means to which Parent and the Stockholder Representative shall agree. Within thirty days after delivery of a Claim Notice, the indemnifying party shall deliver to the indemnified party a written response. If any dispute with respect to the matters set forth in the Claim Notice is not resolved within thirty days following the delivery by the indemnifying party of such response, Parent and the Stockholder Representative shall each have the right to submit such dispute to dispute resolution in accordance with the provisions of Section 9.16. If it shall be determined upon resolution of a Claim as provided herein that a Parent Indemnitee is entitled to indemnification hereunder, such Parent Indemnitee shall be entitled to recover Damages as provided in Section 7.2(a) and Section 7.2(b) hereof. If it shall be determined upon resolution of a Claim as

provided herein that a Company Equityholder Indemnitee is entitled to recover any amount from Parent pursuant to this Article VII, Parent shall satisfy its indemnification obligations hereunder by, at Parent's sole election, either (A) delivering to the Company Equityholder Indemnitees shares of Common Stock having a value equal to such amount (which shares shall be valued at the Price Per Share for this purpose) or (B) paying to the Company Equityholders such amount by check or by wire transfer of immediately available funds.

(f) Provisions Relating to Damages.

- (i) The term "<u>Damages</u>" as used in this <u>Article VII</u> is not limited to matters asserted by third parties against a Parent Indemnitee, but includes Damages incurred or sustained by a Parent Indemnitee in the absence of third-party claims. Payments by a party of amounts for which such party is indemnified hereunder shall not be a condition precedent to recovery. No Company Equityholder shall have any right to reimbursement or contribution from, subrogation to or indemnification from the Surviving Corporation (or any Affiliate thereof) with respect to any indemnification claim of a Parent Indemnitee against any Company Equityholder hereunder or otherwise in connection with this Agreement.
- (ii) Notwithstanding the foregoing, the term "<u>Damages</u>" as used in this <u>Article VII</u> (A) shall be calculated net of any insurance proceeds actually received by the indemnified party with respect to the applicable Claim (but Damages shall include a reasonable estimate of Damages resulting from increased future premiums (if any) resulting from such Claim) and (B) the fees and expenses of only one firm of attorneys (and, where applicable, separate local counsel in the relevant jurisdictions) representing the indemnified parties shall be included in Damages unless said firm of attorneys determines that a conflict of interest would make joint representation of all indemnified parties inadvisable.
- (iii) The right to indemnification, payment of Damages or other remedy based on any representations, warranties, covenants and obligations will not be affected by any investigation conducted with respect to, or any knowledge acquired (or capable of being acquired) at any time, whether before or after the execution and delivery of this Agreement or the Closing, with respect to the accuracy or inaccuracy of or compliance with, any such representation, warranty, covenant or obligation. The waiver of any condition based on the accuracy of any representation or warranty, or on the performance of or compliance with any covenant or obligation, will not affect the right to indemnification, payment of Damages, or other remedies based on such representations, warranties, covenants and obligations.
- (iv) Notwithstanding anything to the contrary in this Agreement, in no event shall any party be liable for, and Damages shall not include, any punitive damages (except for punitive damages that are the subject of third party claims) and no "multiple of profits" or "multiple of cash flow" or similar valuation methodologies based on multiples shall be used in calculating the amount of any Damages. For the avoidance of doubt, any recovery of consequential damages by an indemnified party hereunder will be limited to consequential damages that an indemnified party would have the right to recover under applicable general principles of contract law under the circumstances (e.g. reasonably foreseeable damages); provided that nothing in this sentence will limit the ability of an indemnified party to recover any consequential damages that are the subject of a third party claim.
- (v) The amount of any indemnifiable losses payable under this Article VII shall be net of any Tax Benefits actually realized by any Person that is entitled to indemnification under this Agreement directly as a result of such indemnifiable loss. "Tax Benefit" means any Tax refund, Tax credit or other reduction in otherwise required Tax payments actually realized for the taxable year in which the indemnified losses were incurred (as such actually realized amount is determined in good faith by such Person that is entitled to indemnification).

(g) Appointment of the Stockholder Representative.

(i) By executing this Agreement, the Company (and, upon execution of the Written Consent or a Stockholder Transmittal Letter, each Company Equityholder) shall be deemed to have constituted and appointed,

effective from and after the date of this Agreement, Shareholder Representative Services LLC as the agent and attorney-in-fact for and on behalf of each Company Equityholder to act as the Stockholder Representative under this Agreement in accordance with the terms of this Section 7.2.(g). In the event of the resignation, removal, dissolution, liquidation, bankruptcy, death or incapacity of the Stockholder Representative, the Company Equityholders shall promptly appoint a successor Stockholder Representative (and in any event within ten Business Days of such vacancy), and such appointment shall become effective as to any such successor when a copy of such instrument shall have been delivered to Parent. The Stockholder Representative may resign at any time. The Company Equityholders holding a majority of the Pro Rata Portion of all of the Company Equityholders shall have the right to remove a Stockholder Representative and to appoint a successor Stockholder Representative. The Stockholder Representative shall have full power and authority to represent all of the Company Equityholders and their successors with respect to all matters arising under this Agreement, and all actions taken by the Stockholder Representative hereunder shall be binding upon all such Company Equityholders as if expressly confirmed and ratified in writing by each of them, and no Company Equityholders shall have the right to object, dissent, protest or otherwise contest the same. The Stockholder Representative shall take any and all actions that it believes are necessary or appropriate under this Agreement and the Escrow Agreement for and on behalf of the Company Equityholders as if the Company Equityholders were acting on their own behalf, including giving and receiving any notice or instruction permitted or required under this Agreement or the Escrow Agreement by the Stockholder Representative or any Company Equityholders, interpreting all of the terms and provisions of this Agreement and the Escrow Agreement, authorizing payments to be made with respect hereto, defending all Claims against the Company Equityholders pursuant to Section 7.2(a) of this Agreement, consenting to, compromising or settling all indemnification claims, conducting negotiations with Parent and its agents regarding such claims, dealing with Parent under this Agreement and the Escrow Agreement with respect to all matters arising under this Agreement or the Escrow Agreement, taking any and all other actions specified in or contemplated by this Agreement and the Escrow Agreement and engaging counsel, accountants or other agents in connection with the foregoing matters. Without limiting the generality of the foregoing, the Stockholder Representative shall have full power and authority to interpret all the terms and provisions of this Agreement and the Escrow Agreement and to consent to any amendment hereof or thereof on behalf of all of the Company Equityholders.

- (ii) The Stockholder Representative has been duly authorized by the Company Equityholders for or on behalf of each Company Equityholder, following the Effective Time, to:
 - (A) take all actions required by, and exercise all rights granted to, the Stockholder Representative in this Agreement and the Escrow Agreement;
 - (B) receive all notices or other documents given or to be given to the Company Equityholders by Parent or Merger Sub pursuant to this Agreement or the Escrow Agreement;
 - (C) receive and accept service of legal process in connection with any claim or other proceeding against the Company Equityholders arising under this Agreement or the Escrow Agreement;
 - (D) undertake, compromise, defend and settle any such suit or proceeding on behalf of the Company Equityholders arising under this Agreement (including, without limitation, under Article VII hereof) or the Escrow Agreement;
 - (E) execute and deliver all agreements, certificates and documents required or deemed appropriate by the Stockholder Representative in connection with any of the transactions contemplated by this Agreement, including the Escrow Agreement;
 - (F) engage special counsel, accountants and other advisors and incur such other expenses in connection with any of the transactions contemplated by this Agreement or the Escrow Agreement;
 - (G) work with Parent to resolve any issues relating to the determination and final accounting of Working Capital and the Final Adjustment Amount; and

- (H) take such other action as the Stockholder Representative may deem appropriate, including, without limitation:
- (I) agreeing to any modification or amendment of or waiver with respect to this Agreement or the Escrow Agreement and executing and delivering an agreement of such modification or amendment or waiver;
 - (II) agreeing to any termination of this Agreement pursuant to Article VIII hereof; and
- (III) all such other matters as the Stockholder Representative may deem necessary or appropriate to carry out the intents and purposes of this Agreement.
- (iii) The Stockholder Representative shall receive no compensation for services as the Stockholder Representative other than pursuant to the terms of that of that certain engagement letter to be entered into by and among the Stockholder Representative, the Company and certain of the Company Equityholders. The Stockholder Representative will incur no liability of any kind with respect to any action or omission by the Stockholder Representative in connection with the Stockholder Representative's services pursuant to this Agreement and the Escrow Agreement, except in the event of liability directly resulting from the Stockholder Representative's gross negligence or willful misconduct. The Company Equityholder's will indemnify (based on each such Company Equityholder's Pro Rata Portion), defend and hold harmless the Stockholder Representative from and against any and all losses, liabilities, damages, claims, penalties, fines, forfeitures, actions, fees, costs and expenses (including the fees and expenses of counsel and experts and their staffs and all expense of document location, duplication and shipment) (collectively, "Representative Losses") arising out of or in connection with the Stockholder Representative's execution and performance of this Agreement and the Escrow Agreement, in each case as such Representative Loss is suffered or incurred; provided, that in the event that any such Representative Loss is finally adjudicated to have been directly caused by the gross negligence or willful misconduct of the Stockholder Representative, the Stockholder Representative will reimburse the Company Equityholders the amount of such indemnified Representative Loss to the extent attributable to such gross negligence or willful misconduct. If not paid directly to the Stockholder Representative by the Company Equityholders, any such Representative Losses may be recovered by the Stockholder Representative from (i) the funds in the Expense Fund and (ii) the amounts in the Escrow Account at such time as remaining amounts would otherwise be distributable to the Company Equityholders; provided, that while this section allows the Stockholder Representative to be paid from the Expense Fund and the Escrow Account, this does not relieve the Company Equityholders from their obligation to promptly pay such Representative Losses as they are suffered or incurred, nor does it prevent the Stockholder Representative from seeking any remedies available to it at law or otherwise. In no event will the Stockholder Representative be required to advance its own funds on behalf of the Company Equityholders or otherwise. The Company Equityholders acknowledge and agree that the foregoing indemnities will survive the resignation or removal of the Stockholder Representative or the termination of this Agreement. None of Parent, the Company or Merger Sub shall have any obligation to reimburse the Stockholder Representative for any such expenses.
- (iv) Without limiting the generality of this Section 7.2.(g), following the Effective Time, the Company Equityholders agree that the Stockholder Representative shall act as representative of the Company Equityholders for all purposes under this Article VII. Parent agrees that any claim for indemnification made by any of the Parent Indemnitees pursuant to this Agreement shall be made exclusively through the Stockholder Representative acting on behalf of the Company Equityholders (and the Company Equityholders agree that any such claim against the Company Equityholders by a Parent Indemnitee may be made by providing notice of such claim to the Stockholder Representative).
- (v) The Stockholder Representative is hereby appointed and constituted the true and lawful attorney-in-fact of each Company Equityholder, with full power in his, her or its name and on his, her or its behalf to act according to the terms of this Agreement and the Escrow Agreement in the absolute discretion of the Stockholder Representative and in general to do all things and to perform all acts. This power of attorney and

all authority hereby conferred is granted and shall be irrevocable and shall not be terminated by any act of any Company Equityholder, by operation of law (whether by such Company Equityholder's death, disability protective supervision) or any other event. Without limitation to the foregoing, this power of attorney is to ensure the performance of a special obligation, and, accordingly, each Company Equityholder hereby renounces its, his or her right to renounce this power of attorney unilaterally. Each Company Equityholder hereby waives any and all defenses that may be available to contest, negate or disaffirm the action of the Stockholder Representative taken in good faith under this Agreement or the Escrow Agreement. Notwithstanding the power of attorney granted in this Section 7.2.(g), no agreement, instrument, acknowledgement or other act or document shall be ineffective by reason only of the Company Equityholders having signed or given such directly instead of the Stockholder Representative.

(vi) Any action taken by the Stockholder Representative pursuant to the authority granted in this <u>Section 7.2.(g)</u> shall be effective and absolutely binding on each Company Equityholder notwithstanding any contrary action of, or direction from, any Company Equityholder.

(vii) Upon the Closing, the Company will wire to the Stockholder Representative US\$25,000(the "Expense Fund"), which will be used for the purposes of paying directly, or reimbursing the Stockholder Representative for, any third party expenses pursuant to this Agreement and the Escrow Agreement. The Company Equityholders will not receive any interest or earnings on the Expense Fund and irrevocably transfer and assign to the Stockholder Representative any ownership right that they may otherwise have had in any such interest or earnings. The Stockholder Representative will not be liable for any loss of principal of the Expense Fund other than as a result of its gross negligence or willful misconduct. The Stockholder Representative will hold these funds separate from its corporate funds, will not use these funds for its operating expenses or any other corporate purposes and will not voluntarily make these funds available to its creditors in the event of bankruptcy. Contemporaneous with or as soon as practicable following the completion of the Stockholder Representative's responsibilities hereunder, the Stockholder Representative shall disburse the balance of the Expense Fund to the Company Equityholders based on such Company Equityholders' Pro Rata Portion of such balance. For tax purposes, the Expense Fund shall be treated as having been received and voluntarily set aside by the Company Equityholders at the time of Closing. The parties agree that the Stockholder Representative is not acting as a withholding agent or in any similar capacity in connection with the Expense Fund. If any tax reporting is required with respect to the ultimate distribution of any balance of the Expense Fund, then the Stockholder Representative will provide to Parent or its designated agent, upon request, information regarding the amounts distributed to each Company Equityholders, to be used by Parent or its agent in completing any required tax reporting. Any portion of the Expense Fund that remains undeliverable or unclaimed after six months of the initial delivery attempt shall promptly be paid to Parent and handled in the same manner as other unclaimed funds as provided in the Escrow Agreement.

(h) Limitations on Indemnity.

(i) Neither Parent, on the one hand, nor any of the Company Equityholders, on the other hand, shall be liable under this Article VII for any Damages until the aggregate amount otherwise due to the Company Equityholder Indemnitees or Parent Indemnitees, as the case may be, exceeds an accumulated total of \$350,000 (the "Basket"); provided, however, that this limitation shall apply only to the extent a party is entitled to indemnification exclusively under Section 7.2 (a)(i) (in the case of a claim for Damages by the Parent Indemnitees) or Section 7.2.(b)(i) hereof (in the case of a claim for Damages by Stockholder Representative on behalf of the Company Equityholder Indemnitees) and shall not apply to breaches of the Fundamental Representations or in the case of fraud (which shall be interpreted to require the element of scienter). Once the aggregate amount of Damages exceeds the Basket, then the Company Equityholder Indemnitees or Parent Indemnitees, as the case may be, shall have the right to recover all Damages from and including the first dollar of damages without regard to the Basket, subject to the other limitations set forth in this Article VII.

(ii) The parties agree and acknowledge that in no event shall (A) the aggregate liability of the Company Equityholders in connection with indemnification claims pursuant to Section 7.2(a)(i) exceed

\$4,500,000 (including any amounts recovered from the Escrow) except for breaches of Fundamental Representations, breaches of IP Representations or in the case of fraud (which shall be interpreted to require the element of scienter) or (B) Parent's aggregate liability in connection with indemnification claims pursuant to Section 7.2.(b)(i) exceed \$4,500,000 except for breaches of the Fundamental Representations or in the case of fraud (which shall be interpreted to require the element of scienter). Notwithstanding the foregoing, the parties agree and acknowledge that in no event shall the aggregate liability of the Company Equityholders in connection with indemnification claims for breaches of the IP Representations exceed \$20,000,000 in the aggregate (including any amounts recovered from the Escrow). Notwithstanding the foregoing, the parties agree and acknowledge that in no event shall (1) the aggregate liability of the Company Equityholders in connection with indemnification claims pursuant to Section 7.2(a)(ix) exceed fifty percent (50%) of the Aggregate Consideration in the aggregate (including any amounts recovered from the Escrow) and (2) the Company Equityholders have any liability in connection with indemnification claims pursuant to Section 7.2(a)(ix) after the earlier of (X) the full settlement and release of claims related to the matter set forth on Schedule 7.2(a)(ix) or (Y) the third anniversary of the Closing Date; provided, however, that if Parent submits a Claim Notice with respect to any of the matters set forth on Schedule 7.2(a)(ix) as a result of a Third Party Claim that is made prior to the third anniversary of the Closing Date, then the Company Equityholders will be continue to be responsible any Damages of Parent Indemnitees incurred in connection with, arising out of, resulting from or incident to the indemnification claims set forth in such Claim Notice irrespective of whether such Damages are incurred prior to, on or after the third anniversary of the Closing Date.

(iii) The parties agree and acknowledge that, notwithstanding anything contained in this Agreement to the contrary, (A) subject to Section 7.2.(h)(iii)(B), the aggregate liability of the Company Equityholders in connection with indemnification claims pursuant to Section 7.2(a) (including with respect to breaches Fundamental Representation) shall not exceed the total value of the Aggregate Consideration (valued at the Price Per Share) payable to the Company Equityholders hereunder (including the Stock Escrow Amount), (B) no Company Equityholder shall be liable to any Parent Indemnitee pursuant to this Article VII for any Damages in excess, in the aggregate, of the total value of the Aggregate Consideration (valued at the Price Per Share) payable to such Company Equityholder hereunder (including any such Company Equityholder's Pro Rata Allocation of the Stock Escrow Amount), except in the case of fraud (which shall be interpreted to require the element of scienter) committed by a Company Equityholder (and in the event of such fraud only with respect to the Company Equityholder who committed such fraud) and (C) Parent shall not be liable to the Company Equityholder Indemnitees pursuant to this Article VII for any Damages in excess, in the aggregate, of the total value of the Aggregate Consideration (valued at the Price Per Share) payable to the Company Equityholders hereunder (including the Stock Escrow Amount).

(iv) the Parent Indemnitees and the Company Equityholder Indemnitees shall not be entitled to indemnification pursuant to Section 7.2(a) for any Damages underlying any such indemnification claim to the extent that such Damage was specifically taken into account in the calculation of Final Adjustment Amount or in the determination of the Aggregate Consideration (including any adjustment thereto contemplated by Section 1.12) (it being understood that nothing in this Section 7.2.(h)(iv) shall limit the right of any indemnified party to recover any such Damages to the extent such Damages exceed the amount specifically taken into account in the calculation of the Final Adjustment Amount or the Aggregate Consideration).

7.3. Escrow Account.

(a) In the event of any claim for indemnification under Section 7.2(a) the Parent Indemnitees shall first recover any Damages from the Escrow Account. If the then remaining Escrowed Shares in the Escrow Account are insufficient to cover Damages arising from such Claim, then, on the terms and subject to the conditions set forth herein, the Parent Indemnitees may seek recovery for the remaining amount of such Damages directly from each Company Equityholder in an amount not to exceed such Company Equityholder's Pro Rata Portion of such remaining amount of Damages; provided, however, that the Company Equityholders may elect to satisfy their

indemnification obligations hereunder by (in lieu of cash payment), delivering to Parent shares of Parent Common Stock in book-entry form equal to such amount (which shares shall be valued at the Price Per Share, rounded to the nearest whole share of Parent Common Stock).

- (b) Within five (5) Business Days after the resolution pursuant to Section 7.2.(e)(iii) hereof of a Claim for which a Parent Indemnitee is entitled to recover Damages from the Escrow Account, Parent and the Stockholder Representative shall jointly instruct the Escrow Agent to deliver from the Escrow Account to Parent a number of shares of Parent Common Stock in book-entry form equal to the amount of Damages divided by the Price Per Share (which shares shall be valued at the Price Per Share, rounded to the nearest whole share of Parent Common Stock) (or if there are insufficient shares in the Escrow Account to cover such Damages, all shares of Parent Common Stock in the Escrow Account).
- (c) Within five (5) Business Days after the one year anniversary of the Closing Date, Parent and the Stockholder Representative shall jointly instruct the Escrow Agent to deliver from the Escrow Account to each Company Equityholder such Company Equityholder's Pro Rata Allocation of (i) the number of shares of Parent Common Stock remaining in the Escrow Account minus (ii) (A) the Pending Claims Amount determined as of the one year anniversary of the Closing divided by (B) the Price Per Share.
- 7.4. Exclusive Remedy. Except in the case of fraud (but without limiting any limitations on remedies as a result of fraud that are expressly set forth in this Article VII), this Article VII shall be the exclusive remedy of the Parent Indemnitees and the Company Equityholder Indemnitees from and after the Closing Date for any and all claims arising under this Agreement, including, without limitation, claims of inaccuracy in or breach of any representation, warranty, covenant, agreement or obligation in this Agreement; provided, however, that the foregoing shall not be deemed a waiver by any party of any right to specific performance or injunctive relief.

ARTICLE VIII. TERMINATION

- 8.1. <u>Termination</u>. This Agreement may be terminated at any time prior to Closing, whether before or after receipt of the Parent Stockholder Approval:
 - (a) By mutual written consent of Parent and the Company;
- (b) By Parent, on the one hand, or the Company, on the other hand, (i) if the Closing shall not have occurred on or before April 30, 2015 (the "Outside Date"), or (ii) if, at the Parent Stockholder Meeting (including any adjournment or postponement thereof), the Parent Stockholder Approval shall not have been obtained; provided, however, that this provision shall not be available to Parent if the Company has the right to terminate this Agreement under clause (e) of this Section 8.1 and this provision shall not be available to the Company if Parent has the right to terminate this Agreement under clause (d) of this Section 8.1;
- (c) By Parent if (i) there is a material breach of any representation or warranty set forth in Article III hereof or in the Support Agreement or any covenant or agreement to be complied with or performed by the Company or the Company Stockholders pursuant to the terms of this Agreement or the failure of a condition set forth in Section 6.2 hereof to be satisfied (and such condition is not waived in writing by Parent) on or prior to the Outside Date, or the occurrence of any event which results or would result in the failure of a condition set forth in Section 6.2 hereof to be satisfied on or prior to the Outside Date or (ii) the Company Stockholder Approval shall not have been obtained at the Company Stockholder Meeting (including any adjournment or postponement thereof); provided that Parent may not terminate this Agreement prior to the Closing if the Company and the Company Stockholders have not had a reasonably adequate opportunity to cure such failure (not to exceed five (5) Business Days);

- (d) By the Company if there is a material breach of any representation or warranty set forth in <u>Article IV</u> hereof or of any covenant or agreement to be complied with or performed by Parent or Merger Sub pursuant to the terms of this Agreement or the failure of a condition set forth in <u>Section 6.1</u> hereof to be satisfied (and such condition is not waived in writing by the Company) on or prior to the Outside Date, or the occurrence of any event which results or would result in the failure of a condition set forth in <u>Section 6.1</u> hereof to be satisfied on or prior to the Outside Date; <u>provided</u> the Company may not terminate this Agreement prior to the Closing Date if Parent and Merger Sub have not had a reasonably adequate opportunity to cure such failure (not to exceed five (5) Business Days); or
- (e) By the Company, if (i) all of the conditions set forth in Article VI have been satisfied or waived in accordance with this Agreement as of the date the Closing should have been consummated pursuant to the terms of this Agreement (other than (A) those conditions that by their nature are to be satisfied (including the delivery of documents) at the Closing, which conditions would have been satisfied assuming the Closing had occurred at such time and (B) the condition set forth in Section 6.2.(g)), (ii) the Parent Stockholder Approval shall not have been obtained at the Parent Stockholder Meeting (or any adjournment or postponement thereof), (iii) the Company has certified in writing to Parent on such date that any conditions set forth in Article VI have been satisfied or waived in accordance with this Agreement (or that conditions that by their nature are to be satisfied (including by delivery of documents) at the Closing, which conditions would have been satisfied assuming the Closing had occurred at such time) and that it is ready, willing and able to consummate the transactions contemplated hereby on such date and throughout the three (3) Business Days following delivery of such notice (and such confirmation, a "Closing Failure Notice."), and (iv) Parent fails to consummate the Closing within three (3) Business Days following the delivery of the Closing Failure Notice.
 - 8.2. Effect of Termination. In the event of termination of this Agreement pursuant to Section 8.1 hereof:
- (a) Each party will redeliver all documents, work papers and other material of any other party relating to the transactions contemplated hereby, and all copies thereof, whether so obtained before or after the execution hereof, to the party furnishing the same;
- (b) The provisions under the caption "Confidentiality" in that certain amended and restated non-binding summary of terms, dated August 18, 2014, by and between the Company and Parent shall continue in full force and effect (the "Summary of Terms");
- (c) No party hereto shall have any Liability to any other party to this Agreement, except as stated in Section 7.2(g)(iii) and subsections (a) and (b) of this Section 8.2 and except for any breach of this Agreement occurring prior to the proper termination of this Agreement (subject to the provisions of Section 8.3); and
- (d) The foregoing provisions shall not limit or restrict the availability of specific performance or other injunctive relief to the extent that specific performance or such other relief would otherwise be available to a party hereunder.

8.3. Termination Amount.

(a) If this Agreement is terminated by the Company pursuant to Section 8.1(d) or Section 8.1(e) then Parent shall pay the Company an amount equal to the sum of (i) a nonrefundable fee in the amount in cash equal to \$1,000,000 and (ii) expense reimbursement up to \$500,000 of Company Transaction Expenses (collectively, the "Termination Amount"). Parent shall pay the Termination Amount to the Company, by wire transfer of immediately available funds to an account designated by the Company, within two (2) Business Days after the date of termination of this Agreement pursuant to Section 8.1.(d) or Section 8.1.(d) unless, solely in the case of a termination of this Agreement pursuant to Section 8.1.(d) as a result of willful and intentional breach by Parent or Merger Sub, the Company notifies Parent in writing at the time of termination that it will pursue a remedy other than the Termination Amount (in which case Parent will have no obligation to pay the Termination Amount).

- (b) The parties acknowledge and agree that the payment of the Termination Amount shall constitute liquidated damages, and not a penalty, that the Company would not have entered into this Agreement without the agreements contained in this <u>Section 8.3</u> and that the payment of the Termination Amount in the circumstances specified herein is supported by due and sufficient consideration;
- (c) Notwithstanding anything to the contrary in this Agreement, the Company's right to terminate this Agreement pursuant to Section 8.1.(d) (other than as a result of a willful and intentional breach by Parent or Merger Sub) or Section 8.1(e) and receive the Termination Amount pursuant to Section 8.3.(a) shall constitute the sole and exclusive remedy (whether in respect of any other document or theory of law or equity, in contract in tort or otherwise) of the Company, any of its Subsidiaries and any of their respective former, current or future direct or indirect stockholders, controlling persons, general or limited partners, members, managers, directors, officers, employees, agents, assignees, representatives or Affiliates or any former, current, or future direct or indirect stockholder, controlling person, general or limited partner, member, manager, director, officer, employee, agent, assignee, representative or Affiliate of any of the foregoing (collectively the "Company Related Parties") against Parent, Merger Sub, any potential debt financing source or any of their respective former, current or future direct or indirect stockholders, controlling persons, general or limited partners, members, managers, directors, officers, employees, agents, assignees, representatives or Affiliates or any former, current or future direct or indirect stockholder, controlling person, general or limited partner, member, manager, director, officer, employee, agent, assignee, representative or Affiliate of any of the foregoing (collectively the "Parent Related Parties") for all Damages suffered as a result of the failure of the Merger to be consummated or in respect of this Agreement, the Ancillary Agreements or the transactions contemplated hereby and thereby (including in respect of any breach, whether or not willful and intentional, of any representation, warranty, covenant or agreement except with respect to any obligations that survive such termination of this Agreement in accordance with Section 8.2) and/or in connection with any oral representations made or alleged to be made in connection therewith or in respect of any other document or theory of law or equity, in contract, in tort or otherwise, and, upon receipt of the Termination Amount in accordance with this Section 8.3), none of the Parent Related Parties shall have any further liability or obligation to any of the Company Related Parties arising out of or relating to this Agreement, the Ancillary Agreements or the transactions contemplated hereby and thereby (including in respect of any breach, whether or not willful and intentional, of any representation, warranty, covenant or agreement) and/or in connection with any oral representations made or alleged to be made in connection therewith or in respect of any other document or theory of law or equity, in contract, in tort or otherwise, and none of the Company Related Parties shall seek to recover any other monetary damages or seek any other remedy based on a claim at law or in equity with respect thereto, except with respect to any obligations that survive such termination of this Agreement in accordance with Section 8.2; provided, however, that (i) this Section 8.3 and Section 9 shall survive the termination of this Agreement and shall remain in full force and effect, and (ii) the termination of this Agreement shall not relieve any Party from any liability for any fraud (which shall be interpreted to include the element of scienter). Notwithstanding anything contained herein to the contrary, this Section 8.3 does not limit (or apply to) any remedy of the Company Related Parties in the event that the Company exercises its right to terminate this agreement pursuant to Section 8.1(d) as a result of a willful and intentional breach by Parent or Merger Sub and pursue a remedy other than the Termination Amount.

ARTICLE IX. MISCELLANEOUS

9.1. <u>Defined Terms</u>. As used herein, the terms below shall have the following meanings. Any such term, unless the context otherwise requires, may be used in the singular or plural, depending upon the reference.

"Action" shall mean any action, complaint, claim, suit, litigation, proceeding, labor dispute, arbitral action, governmental audit, formal inquiry, criminal prosecution, civil or criminal investigation or unfair labor practice charge or complaint.

"Adjusted Aggregate Consideration" shall mean an amount equal to (a) the Parent Common Stock Closing Consideration multiplied by (b) the Closing Price Per Share.

"Adjusted Fully Diluted Number" shall mean without duplication, (i) the total number of shares of Company Common Stock (including without limitation, any shares of Company Restricted Stock) outstanding immediately prior to the Effective Time, <u>plus</u> (ii) the total number of shares of Company Common Stock issuable upon exercise of all Company Options (whether or not vested or exercisable) outstanding immediately prior to the Effective Time; <u>plus</u> (iii) the total number of shares of Company Common Stock issuable upon conversion of all shares of Series A Preferred Stock (assuming for this purpose that all shares of Series A Preferred Stock are converted) outstanding immediately prior to the Effective Time, <u>plus</u> (iv) the total number of shares of Company Common Stock issuable upon conversion of all shares of Series FF Preferred Stock (assuming for this purpose that all shares of Series FF Preferred Stock are converted) outstanding immediately prior to the Effective Time.

"Affiliate" shall mean, when used with reference to any specified Person, any other Person directly or indirectly controlling, controlled by, or under direct or indirect common control with, such specified Person. For purposes of this definition, "control," when used with respect to any specified Person, means the power to direct or cause the direction of management or policies of such Person, directly or indirectly, whether through the ownership of voting securities, by contract or otherwise. Any stockholder of the Company that has a designee on the Company's Board of Directors (or has the right to designate a member of the Company's Board of Directors) shall be deemed an Affiliate of the Company for this purpose.

"Aggregate Company Optionholder Consideration" means (a) the number of shares of Company Common Stock issuable upon exercise of all Company Options (whether or not vested or exercisable) outstanding immediately prior to the Effective Time multiplied by (b) the Per Share Common Consideration Equivalent Amount.

"Aggregate Consideration" shall mean an amount equal to the sum of the Aggregate Stock Consideration and the Stock Escrow Amount.

"Aggregate Series B Preferred Liquidation Preference" shall mean an amount equal to (a) \$5.838464591 multiplied by the number of shares of Series B Preferred Stock outstanding as of immediately prior to the Effective Time, plus (b) the aggregate amount of accrued dividends on the Series B Preferred Stock as of immediately prior to the Effective Time as determined pursuant to the Restated Charter, plus (without duplication) (c) the aggregate amount of any declared and unpaid dividends (if any) on the Series B Preferred Stock as of immediately prior to the Effective Time as determined pursuant to the Restated Charter.

"Aggregate Series B Preferred Stock Consideration" shall mean a number of shares of Parent Common Stock (rounded down to the nearest share) equal to (a) the Aggregate Series B Preferred Liquidation Preference divided by (b) the Closing Price Per Share.

"Aggregate Series C Preferred Liquidation Preference" shall mean an amount equal to (a) \$7.4139234 multiplied by the number of shares of Series C Preferred Stock outstanding as of immediately prior to the Effective Time (excluding, for the avoidance of doubt, the Series C Warrants), plus (b) the aggregate amount of accrued dividends on the Series C Preferred Stock (excluding, for the avoidance of doubt, the Series C Warrants) as of immediately prior to the Effective Time as determined pursuant to the Restated Charter, plus (without duplication) (c) the aggregate amount of any declared and unpaid dividends (if any) on the Series C Preferred Stock (excluding, for the avoidance of doubt, the Series C Warrants) as of immediately prior to the Effective Time as determined pursuant to the Restated Charter.

"Aggregate Series C Preferred Stock Consideration" shall mean a number of shares of Parent Common Stock (rounded down to the nearest share) equal to (a) the Aggregate Series C Preferred Liquidation Preference divided by (b) the Closing Price Per Share.

- "Aggregate Series C Other Warrants Exercise Price" shall mean the aggregate exercise price payable to the Company upon exercise of all Series C Other Warrants that have not been exercised immediately prior to the Effective Time.
- <u>"Aggregate Series C Other Warrants Liquidation Preference"</u> shall mean an amount equal to (a) \$7.4139234 <u>multiplied by</u> the number of Series C Warrants underlying the Series C Other Warrants immediately prior to the Effective Time plus (b) the aggregate amount of dividends deemed to have been accrued on the Series C Warrants (as provided in Section 2(e)(ii) of the Series C Other Warrants) underlying the Series C Other Warrants as of immediately prior to the Effective Time <u>minus</u> (c) the Aggregate Series C Other Warrants Exercise Price.
- "Aggregate Series C Other Warrants Stock Consideration" shall mean a number of shares of Parent Common Stock (rounded up to the nearest share) equal to (a) the Aggregate Series C Other Warrants Liquidation Preference divided by (b) the Closing Price Per Share.
- "Aggregate Series C Penny Warrants Exercise Price" shall mean the aggregate exercise price payable to the Company upon exercise of all Series C Penny Warrants that have not been exercised immediately prior to the Effective Time.
- "Aggregate Series C Penny Warrants Liquidation Preference" shall mean an amount equal to (a) \$7.4139234 multiplied by the number of Series C Warrants underlying the Series C Penny Warrants immediately prior to the Effective Time plus (b) the aggregate amount of dividends deemed to have been accrued on the Series C Warrants (as provided in Section 2(e)(ii) of the Series C Penny Warrants) underlying the Series C Penny Warrants as of immediately prior to the Effective Time minus (c) the Aggregate Series C Penny Warrants Exercise Price.
- "Aggregate Series C Penny Warrants Stock Consideration" shall mean a number of shares of Parent Common Stock (rounded down to the nearest share) equal to (a) the Aggregate Series C Penny Warrants Liquidation Preference divided by (b) the Closing Price Per Share.
- "Aggregate Series D Preferred Liquidation Preference" shall mean an amount equal to (a) \$16.83 multiplied by the number of shares of Series D Preferred Stock outstanding as of immediately prior to the Effective Time, plus (b) the aggregate amount of accrued dividends on the Series D Preferred Stock as of immediately prior to the Effective Time as determined pursuant to the Restated Charter, plus (without duplication) (c) the aggregate amount of any declared and unpaid dividends (if any) on the Series D Preferred Stock as of immediately prior to the Effective Time as determined pursuant to the Restated Charter.
- "Aggregate Series D Preferred Stock Consideration" shall mean a number of shares of Parent Common Stock (rounded down to the nearest share) equal to (a) the Aggregate Series D Preferred Liquidation Preference divided by (b) the Closing Price Per Share.
- "Aggregate Stock Consideration" shall mean an amount equal to (A) 100,000,000 minus (B) the Stock Escrow Amount, plus (C) the aggregate exercise price of all Company Options, plus (D) the Adjustment Amount (which may be a negative number).
- "Ancillary Agreements" shall mean the Notes, the Employment Arrangements and all other agreements, instruments, documents and certificates executed, filed or otherwise prepared, exchanged or delivered in accordance with this Agreement.
- "Books and Records" shall mean all business records, tangible data, documents, management information systems (including related computer software), files, customer lists, supplier lists, blueprints, specifications, designs, drawings, plans, operation or maintenance manuals, bids, personnel records, invoices, sales literature, all Tax Returns and all worksheets, notes, files or documents related thereto, and all other books and records maintained by (a) Parent or (b) the Company or PocketGear with respect to the Business, as applicable.

- "Business" shall mean any historical, current or planned business or business activities of the Company or PocketGear.
- "Business Day" shall mean a day other than Saturday, Sunday or any day on which banks located in the State of California are authorized or obligated to close.
- "Business Domain Names" shall mean the Internet domain names registered in the name of the Company or PocketGear, or any way used in connection with the Business.
 - "CERCLA" shall mean the Comprehensive Environmental Response, Compensation and Liability Act.
 - "Charter Amendment" shall mean an amendment to the Restated Charter in substantially the form attached hereto as Exhibit L.
- "Closing Price Per Share" shall mean an amount equal to (a) (i) if the average of the closing prices of the Parent Common Stock on the NASDAQ Capital Market over the thirty (30) day period ending three (3) calendar days prior to the Closing (the "VWAP") is greater than \$4.50, \$4.50, (ii) if the VWAP is less than \$3.50, \$3.50 or (iii) the VWAP, in any case, multiplied by (b) eighty percent (80%).
 - "Code" shall mean the Internal Revenue Code of 1986, as amended.
- "Company Capital Stock" shall mean the Company Common Stock, the Company Preferred Stock, the Company Restricted Stock and any other shares of capital stock, if any, of the Company, taken together.
 - "Company Common Stock" shall mean shares of common stock, par value \$0.001, of the Company.
 - "Company Equity Plan" shall mean the 2008 Stock Incentive Plan of the Company, as amended.
- "Company Equityholders" shall mean, collectively, all of the Company Stockholders, including any holders of Company Restricted Stock, and each Series C Warrantholder entitled to receive consideration in connection with the Merger pursuant to Section 1.7.
- "Company Equityholder Remaining Consideration" means the Remaining Consideration minus the Aggregate Company Optionholder Consideration.
- "Company Information" shall mean all information about the Company and PocketGear, supplied or to be supplied by the Company for inclusion or incorporation by reference, including such information as may be reasonably requested by Parent, reasonably sufficient to permit Parent to prepare and file with the SEC: (i) the Proxy Statement and (ii) any other required filings under applicable Regulations.
- "Company Intellectual Property" shall mean any and all Intellectual Property that is owned by, purported to be owned by, the Company or PocketGear.
 - "Company Optionholders" shall mean holders of Company Options.
- "Company Options" shall mean all options to purchase Company Common Stock that are outstanding and unexpired as of the date hereof.
- "Company Plan" shall mean each Plan under which the Company or PocketGear or any other of the Company's subsidiaries has any obligation or liability, whether actual or contingent, direct or indirect, to provide compensation or benefits to or for the benefit of any of its current or former employees, consultants, managers or directors, or the spouses, beneficiaries or other dependents thereof.

"Company Preferred Stockholder" shall mean any holder of Company Preferred Stock immediately prior to the Effective Time.

"Company Registered Intellectual Property" shall mean the applications, registrations and filings for Company Intellectual Property that have been registered, filed, certified or otherwise perfected or recorded or are the subject of a pending application for such, with or by any Governmental Authority or the Internet domain name registrar, by or on behalf of or in the name of the Company or PocketGear (including all Internet domain names).

"Company Restricted Stock" shall mean each share of Company Common Stock then outstanding that remains subject to vesting or other lapse restrictions.

"Company Stockholder" shall mean any holder of Company Capital Stock immediately prior to the Effective Time.

"Company Transaction Expenses" shall mean the aggregate amount of (a) all fees and expenses incurred by the Company or PocketGear in connection with the negotiation, preparation, execution and performance of this Agreement and the transactions contemplated hereby, including, without limitation, all legal, financial advisory, accounting, consulting and other fees and expenses and any broker's or finder's fees, (b) all amounts (plus any associated Taxes required to be paid by the Company or PocketGear with respect thereto) payable by the Company or PocketGear, whether actually paid immediately or in the future, under any "change of control," retention, transaction, termination, severance or similar arrangements as a result of the consummation of the transactions contemplated hereby and/or a termination of service that occurs on or prior to the Closing (including, without limitation, any such amounts payable to any employee of the Company or PocketGear at the election of such employee pursuant to any such arrangements), (c) any associated Taxes required to be paid by the Company or PocketGear with respect to the Assumed Option Consideration, Deemed Exercise Option Consideration or the Restricted Stock Consideration (other than as a result of any exercise or vesting of such Assumed Option Consideration or Restricted Stock Consideration that occurs after the Closing), (d) any costs and expenses incurred by Parent, the Company or PocketGear in connection with the refinancing of the Existing Credit Agreements, and (e) the amount of the Expense Fund.

"Confidential Information" shall mean any and all trade secrets, confidential business or technical information, and proprietary information and materials, whether or not stored in any medium, relating to the Company, PocketGear or the Business, including, but not limited to, business information, technology, technical documentation, product or service specifications or strategies, marketing plans, research and development, designs, formulae, computer programs, pricing information, financial information, information relating to existing, previous and potential suppliers, customers, contracts and other know-how.

"Continuing Service Provider" shall mean (i) all employees of the Company and PocketGear who (A) at the Effective Time, continue their employment with the Company and PocketGear or Parent or one of Parent's subsidiaries, or (B) remain or become, at the Effective Time, employees of the Surviving Corporation, and (ii) all Contingent Workers who, at the Effective Time, continue their service with the Company and PocketGear or Parent, other than any such service providers who are ineligible to be included on a registration statement filed by Parent on Form S-8.

"Contingent Workers" means any Person that is an independent contractor or consultant of the Company or PocketGear.

"Contract" shall mean any agreement, contract, note, loan, evidence of Indebtedness, purchase order, letter of credit, indenture, security or pledge agreement, franchise agreement, undertaking, practice, covenant not to compete, license, instrument, obligation, promise or commitment (in each case, whether written, electronic or oral and whether expressed or implied), but excluding all Leases.

"Court Order" shall mean any judgment, decision, consent decree, injunction, ruling or order of any federal, state or local court or governmental agency, department or authority that is binding on any Person or its property under applicable law.

"Current Assets" shall mean all current assets of the Company and its subsidiaries on a consolidated basis determined in accordance with GAAP applied in a manner reasonably agreed to by the parties; provided, however, that Current Assets shall include any unbilled accounts receivable (to the extent provided by GAAP) but shall not include cash and cash equivalents. Notwithstanding the foregoing, (i) Current Assets shall exclude any accounts receivable from shareholders or related parties and (ii) any accounts receivable or unbilled accounts receivable set forth on the Pre-Closing Statement which have not been collected within one-hundred and twenty (120) calendar days following the Closing Date will be deemed to not constitute Current Assets to the extent such accounts receivable or unbilled accounts receivable exceed the related reserve or allowance (if any) for bad debt included as a Current Liability on the Pre-Closing Statement notwithstanding the accounting treatment of any such accounts receivable required under GAAP and shall not be reflected as Current Assets on the Closing Statement.

"<u>Current Liabilities</u>" shall mean all current liabilities of the Company and its subsidiaries on a consolidated basis determined in accordance with GAAP applied in a manner reasonably agreed to by the parties, other than the current portion of Indebtedness to the extent such current portion is included in Long Term Liabilities. Notwithstanding the foregoing, all unpaid Company Transaction Expenses incurred by the Company prior to the Effective Time shall for the purposes of hereof be considered part of Net Debt.

"<u>Data Protection Laws</u>" shall mean the data protection and privacy laws of each country where the Company or PocketGear is established and those of each country where any such data is collected, transmitted, secured, stored, shared or otherwise processed, including without limitation such laws and Regulations that implement the European Data Protection Directive and the Privacy and Electronic Communications Directive (2002/58/EC) in the European Economic Area ("<u>EEA</u>") and any applicable guidelines and codes issued by a competent data protection authority, or other competent governmental body or agency, in respect of such laws, and shall also mean the Massachusetts data protection law, 201 CMR 17.00, and Section 5 of the (U.S.) Federal Trade Commission Act and any and all laws and Regulations governing privacy, cybercrime, use of electronic data, or unfair or deceptive trade practices.

"<u>Default</u>" shall mean (a) a breach of or default under any Contract, Lease or Permit, (b) the occurrence of an event that with the passage of time or the giving of notice or both would constitute a breach of or default under any Contract, Lease or Permit, or (c) the occurrence of an event that with or without the passage of time or the giving of notice or both would give rise to a right of termination, renegotiation or acceleration under any Contract, Lease or Permit.

"Disclosure Schedules" shall mean a schedule executed and delivered by the Company to Parent and Merger Sub as of the date hereof which sets forth the exceptions to the representations and warranties contained in Article III hereof and certain other information called for by this Agreement. Unless otherwise specified, each reference in this Agreement to any numbered schedule is a reference to that numbered schedule which is included in the Disclosure Schedules.

"Employment Statute" shall mean any federal, state or municipal employment, labor or employment discrimination law, including without limitation, the National Labor Relations Act, Title VII of the Civil Rights Act of 1964, the Civil Rights Act of 1991, the Age Discrimination in Employment Act of 1967, the Americans with Disabilities Act of 1990, the Fair Labor Standards Act, ERISA, the WARN, the Family and Medical Leave Act, the Immigration Reform and Control Act of 1986 (the "IRCA"), the California Fair Employment and Housing Act, the California Family Rights Act, and the California Labor Code, and all amendments to each such Act as well as the Regulations issued thereunder.

"Encumbrance" shall mean any claim, lien, pledge, option, charge, community property interest, equitable interest, right of first refusal or restriction of any kind, easement, security interest, deed of trust, mortgage, pledge, hypothecation, right-of-way, encroachment, building or use restriction, conditional sales agreement, encumbrance or other right of third parties, whether voluntarily incurred or arising by operation of law, and includes, without limitation, any agreement to give any of the foregoing in the future, and any contingent sale or other title retention agreement or lease in the nature thereof.

"Environmental Laws" shall mean all Regulations which regulate or relate to the protection or clean-up of the environment, the use, treatment, storage, transportation, generation, manufacture, processing, distribution, handling or disposal of, or emission, discharge or other release or threatened release of, Hazardous Substances or otherwise dangerous substances, wastes, pollution or materials (whether, gas, liquid or solid), the preservation or protection of waterways, groundwater, drinking water, air, wildlife, plants or other natural resources, or the health and safety of persons or property, including, without limitation, protection of the health and safety of employees. Environmental Laws shall include, without limitation, the Federal Insecticide, Fungicide, Rodenticide Act, Resource Conservation & Recovery Act, Clean Water Act, Safe Drinking Water Act, Atomic Energy Act, Occupational Safety and Health Act, Toxic Substances Control Act, Clean Air Act, CERCLA, Emergency Planning and Community Right-to-Know Act, Hazardous Materials Transportation Act and all analogous or related federal, state or local law, each as amended.

"ERISA" shall mean the Employee Retirement Income Security Act of 1974, as amended.

"ERISA Affiliate" shall mean any entity (whether or not incorporated) other than the Company that, together with the Company, is required to be treated as a single employer under Section 414(b), (c), (m) or (o) of the Code.

"Escrow Agent" shall mean American Stock Transfer & Trust Company, LLC.

"Escrow Agreement" shall mean that certain Escrow Agreement, dated as of the Closing Date, by and among Parent, the Stockholder Representative and the Escrow Agent, in substantially the form attached hereto as Exhibit F.

"Exchange Act" shall mean the United States Securities Exchange Act of 1934, as amended, including all rules and regulations issued thereunder.

"Existing Credit Arrangements" shall mean the Purchase Agreement dated April 14, 2013 by and between the Company and North Atlantic SBIC IV, L.P., the Purchase Agreement dated October 31, 2013 by and between the Company and North Atlantic SBIC IV, L.P. and the Amended and Restated Loan Security Agreement dated August 8, 2012 between Silicon Valley Bank and the Company, as amended.

"Existing Lenders" shall mean Silicon Valley Bank and North Atlantic SBIC IV, L.P.

"Facilities" shall mean any real property, leaseholds or other real property interest currently owned, held, occupied or operated by the Company or PocketGear and any buildings, structures or equipment (including motor vehicles) currently owned or operated by the Company or PocketGear.

"Facility Leases" shall mean all of the leases of Facilities listed on Schedule 3.7 of the Disclosure Schedules.

"Financial Statements" shall mean the Year End Financial Statements and the Interim Financial Statements.

"Fixtures and Equipment" shall mean all of the furniture, fixtures, furnishings, automobiles, trucks, spare parts, supplies, equipment and other tangible personal property owned by the Company or PocketGear and used in connection with the Business, wherever located and including any such fixtures and equipment in the possession of any of the Company's or PocketGear's suppliers, including all warranty rights with respect thereto.

"Form S-4" shall mean the registration statement on Form S-4, and all amendments thereto, filed with the SEC in connection with the registration of Parent Common Stock under the Securities Act.

"GAAP" shall mean United States generally accepted accounting principles and practices, consistently applied.

"Governmental Authority" shall mean any United States, foreign, supra-national, federal, state, provincial, local or self-regulatory governmental, regulatory or administrative authority, agency, division, body, organization or commission or any judicial or arbitral body.

"Hazardous Substance" shall mean any pollutant, contaminant, chemical, waste and any toxic, infectious, carcinogenic, reactive, corrosive, ignitable or flammable chemical or chemical compound or hazardous substance, material or waste, whether solid, liquid or gas, including, without limitation, any quantity of asbestos in any form, urea formaldehyde, PCB's, radon gas, crude oil or any fraction thereof, all forms of natural gas, petroleum products or by-products or derivatives, radioactive substance or material, pesticide waste waters, sludges, slag and any other substance, material or waste that is subject to regulation, control or remediation under any Environmental Laws.

"HSR Act" shall mean the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended.

"Indebtedness" shall mean, without duplication, (i) all obligations for borrowed money (including all sums due on early termination and repayment or redemption calculated to the Closing Date) or extensions of credit (including under credit cards, bank overdrafts, and advances) or extensions of credit (including under credit cards, bank overdrafts, and advances), (ii) all obligations evidenced by bonds, debentures, notes, or other similar instruments (including all sums due on early termination and repayment or redemption calculated to the Closing Date), (iii) all obligations to pay the deferred purchase price of property or services, except trade accounts payable arising in the ordinary course of business, (iv) all obligations of others secured by an Encumbrance on any asset of the Company or PocketGear, (v) all obligations, contingent or otherwise, directly or indirectly guaranteeing any obligations of any other Person, (vi) all obligations to reimburse the issuer in respect of letters of credit or under performance or surety bonds, or other similar obligations, and (vii) all obligations in respect of bankers' acceptances and under reverse repurchase agreements.

"Insurance Policies" shall mean the insurance policies of the Company or PocketGear listed on Schedule 3.23 of the Disclosure Schedules.

"Intellectual Property" shall mean any and all rights in or affecting intellectual or industrial property, existing now or in the future in the United States or anywhere in the universe. Intellectual Property includes, without limitation, any and all such rights in, to, or subsisting in the following:

- (a) all issued patents, reissued or reexamined patents, revivals of patents, divisions, continuations and continuations-in-part of patents, all renewals and extensions thereof utility models, and certificates of invention, regardless of country or formal name;
- (b) all published or unpublished nonprovisional and provisional patent applications, including the right to file other or further applications, reexamination proceedings, invention disclosures and records of invention;
- (c) all copyrights, copyrightable works, semiconductor topography and mask work rights, including, without limitation, all rights of authorship, use, publication, reproduction, distribution, performance, transformation, moral rights and ownership of copyrightable works, semiconductor topography works and mask

works, the right to create derivative works, and all applications for registration, registrations, renewals and extensions of registrations, together with all other interests accruing by reason of international copyright, semiconductor topography and mask work conventions;

- (d) all trademarks, service marks, symbols, logos, trade names, domain names, d/b/a's, certification marks, brand names, corporate names, fictitious names, 1-800, 1-888, 1-877 and other "vanity" telephone numbers, together with the goodwill of the business associated therewith, all applications for registration and registrations thereof, renewals thereof, the right to bring opposition and cancellation proceedings and any and all rights under the laws of trade dress;
- (e) all proprietary information and materials, whether or not patentable or copyrightable, and whether or not reduced to practice, including without limitation all technology, ideas, research and development, inventions, designs, manufacturing and operating specifications and processes, schematics, know-how, formulae, customer and supplier lists, shop rights, designs, drawings, patterns, trade secrets, confidential information, technical data, databases, data compilations and collections, web addresses and sites, Software, architecture, and documentation
 - (f) all other intangible assets, properties and rights; and
- (g) all claims, causes of action and rights to sue for past, present and future infringement or unconsented use of any of the foregoing intellectual and industrial property rights set forth in the foregoing paragraphs (a) through (f), the right to file applications and obtain registrations, and all rights arising therefrom and pertaining thereto and all products, proceeds and revenues arising from or relating to any and all of the foregoing.
 - "Interim Balance Sheet" shall mean the balance sheet of the Company dated the Interim Balance Sheet Date.
 - "Interim Balance Sheet Date" shall mean September 30, 2014.
- "Interim Financial Statements" shall mean the Interim Balance Sheet and the unaudited statements of income and cash flow of the Company for the period ended on the Interim Balance Sheet Date.
 - "IRS" shall mean the United States Internal Revenue Service or any successor agency.
- "Knowledge" or "Knowledge of the Company" shall mean the actual knowledge of any of Jud Bowman, Tim Oakley, Ken Hayes, Jeff Henderson (with respect to Section 3.18 only), Jim Harvey and Val Stiebris, in each case following reasonable due inquiry.
- "Law" shall mean any federal, state, local or foreign statute, code, ordinance, rule, treaty, regulation, order, judgment, writ, stipulation, award, injunction, and any enforceable judicial or administrative interpretation thereof, including any judicial or administrative order, consent decree, judgment, stipulation, injunction, permit, authorization, policy, opinion, or agency requirement, in each case having the force and effect of Law.
 - "Leased Real Property" shall mean all leased property described in the Facility Leases.
 - "Leasehold Estates" shall mean all of the Company's or PocketGear's rights and obligations as lessee under the Leases.
- "<u>Leasehold Improvements</u>" shall mean all leasehold improvements situated in or on the Leased Real Property and owned by the Company or PocketGear.

"<u>Leases</u>" shall mean all of the existing leases with respect to the personal or real property of the Company or PocketGear listed on <u>Schedule 3.7</u> of the Disclosure Schedules, and leases with respect to the personal and real property of the Company or PocketGear which are not required to be listed on <u>Schedule 3.7</u> of the Disclosure Schedules.

"<u>Liabilities</u>" shall mean any direct or indirect liability, indebtedness, obligation, commitment, expense, claim, deficiency, deferred income, guaranty or endorsement of or by any Person of any type, whether accrued, absolute, contingent, matured, unmatured or other, and shall include all reserves which shall include a reserve for accrued but unpaid Taxes (but not for any deferred Tax liabilities.

"<u>Liquidation Preference</u>" shall mean, with respect to any share of Company Preferred Stock, the Original Issue Price (as defined in the Restated Charter) of such share of Company Preferred Stock plus all accrued dividends and all declared and unpaid dividends on such share of Company Preferred Stock as determined pursuant to the Restated Charter.

"Long Term Liabilities" shall mean all Indebtedness (including the current portion thereof) and other long term liabilities of the Company or any of its subsidiaries on a consolidated basis determined in accordance with GAAP (including any such liabilities arising as a result of the consummation of the Merger and the transactions contemplated hereby). Any pre-payment or other penalties required to be paid in connection for Indebtedness to be repaid in full on the Closing Date shall be included in the calculation of Long Term Liabilities for this purpose.

"Material Adverse Effect" or "Material Adverse Change shall mean with respect to the Company, PocketGear or the Business any material adverse effect or change in the condition (financial or other), business, results of operations, assets, Liabilities or operations of the Company, PocketGear or the Business taken as a whole or on the ability of the Company to consummate the transactions contemplated hereby, or any event or condition which would or would reasonably be expected to, with the passage of time, constitute a "Material Adverse Effect" or "Material Adverse Change"; provided, however, that none of the following shall be taken into account in determining whether there has been or would reasonably be expected to be a "Material Adverse Effect" or "Material Adverse Change": (a) any change in general economic, credit, capital or financial markets or general political conditions; (b) any outbreak of hostilities, acts of war or terrorism; (c) any hurricane, tornado, flood, earthquake or other natural disaster occurring after the date hereof; (d) any change in applicable law or GAAP enacted after the date hereof; (e) any adverse change in general conditions in the industries in which such Person primarily operates; (f) except with respect to Section 3.5, the negotiation, execution or announcement of the transactions contemplated by, or the performance of obligations under, this Agreement or the Ancillary Agreements, including effects related to the identity of Parent; (g) any failure to meet internal or published projections, estimates or forecasts of revenues, earnings, or other measures of financial or operating performance for any period (provided that the underlying causes of such failure may be considered in determining whether there was or is a Material Adverse Effect); or (h) actions specifically required to be taken by the Company pursuant to this Agreement or that are taken following receipt of Parent's written consent to such action; except in the cases of clauses (a) – (e), to the extent that such Person is disproportionately adversely affected thereby as compared with other participants in the industries in which such Person primarily operates.

"Net Debt" shall mean (i) all Long Term Liabilities minus (ii) all cash and cash equivalents of the Company determined in accordance with GAAP plus (iii) all unpaid Company Transactions Expenses incurred by the Company prior to the Effective Time (including Company Transaction Expenses that have not been billed prior to the Effective Time but that will be billed or become payable in the future).

"Non-Competition Agreements" shall mean the non-competition agreement in the form attached as <u>Exhibit I</u> between Parent and the Company Stockholder identified therein and dated as of the date hereof, which shall become effective as of the Closing.

"Option Exchange Ratio" shall mean a ratio, the numerator of which is the Per Share Common Consideration Equivalent Amount and the denominator of which is the Closing Price Per Share.

- "ordinary course of business" or "ordinary course" or any similar phrase shall describe any action taken by a Person if:
- (a) such action is consistent with the past practices of such Person and is taken in the ordinary course of the normal day-to-day operations of such Person;
- (b) such action is not required to be authorized by the board of directors of such Person and is not required to be authorized by the parent company (if any) of such Person; and
- (c) such action is similar in nature and magnitude to actions customarily taken, without any authorization by the board of directors in the ordinary course of the normal day-to-day operations of other Persons that are in the same line of business as such Person.
- "Organizational Documents" shall mean (a) the articles or certificate of incorporation, all certificates of determination and designation, and the bylaws of a corporation; (b) the partnership agreement and any statement of partnership of a general partnership; (c) the limited partnership agreement and the certificate or articles of limited partnership of a limited partnership; (d) the operating agreement, limited liability company agreement and the certificate or articles of organization or formation of a limited liability company; (e) any charter or similar document adopted or filed in connection with the creation, formation or organization of any other Person; and (f) any amendment to any of the foregoing.
- "Owned Real Property" shall mean all real property owned in fee by the Company or PocketGear, including, without limitation, all rights, easements and privileges appertaining or relating thereto, all buildings, fixtures, and improvements located thereon and all Facilities thereon, if any.
- "Parachute Payment Waiver" means, with respect to any Person, a written agreement waiving such Person's right to receive any "parachute payments" (within the meaning of Section 280G of the Code and the Department of Treasury regulations promulgated thereunder) solely to the extent required to avoid the imposition of a tax by virtue of the operation of Section 280G of the Code and to accept in substitution therefor the right to receive such payments only if approved by the stockholders of the Company in a manner that complies with Section 280G(b)(5)(B) of the Code and the regulations promulgated thereunder.
- "<u>Parent Common Stock</u>" shall mean the common stock, par value \$0.0001 per share, of Parent (it being understood in the event all Parent Common Stock is converted or exchanged for other securities following the Effective Time, Parent Common Stock shall thereafter refer to the resulting securities).
- "<u>Parent Closing Stock Consideration</u>" means the number of shares of Parent Common Stock equal to the quotient of (i) the Aggregate Consideration divided by (ii) the Price Per Share.
- "Parent Material Adverse Effect" shall mean with respect to Parent any material adverse effect or change in the condition (financial or other), business, results of operations, assets, Liabilities or operations of Parent, Merger Sub or their Subsidiaries, taken as a whole, or on the ability of Parent and Merger Sub to consummate the transactions contemplated hereby, or any event or condition which would or would reasonably be expected to, with the passage of time, constitute a "Material Adverse Effect"; provided, however, that none of the following shall be taken into account in determining whether there has been or would reasonably be expected to be a "Material Adverse Effect" or "Material Adverse Change": (a) any change in general economic, credit, capital or financial markets or general political conditions; (b) any outbreak of hostilities, acts of war or terrorism; (c) any hurricane, tornado, flood, earthquake or other natural disaster occurring after the date hereof; (d) any change in applicable law or GAAP enacted after the date hereof; (e) any adverse change in general

conditions in the industries in which such Person primarily operates; or (f) any failure to meet internal or published projections, estimates or forecasts of revenues, earnings, or other measures of financial or operating performance for any period (provided that the underlying causes of such failure may be considered in determining whether there was or is a Material Adverse Effect); (g) except with respect to Section 4.3(b), the negotiation, execution or announcement of the transactions contemplated by, or the performance of obligations under, this Agreement or the Ancillary Agreements, including effects related to the identity of the Company; or (h) actions specifically required to be taken by Parent pursuant to this Agreement or that are taken following receipt of the Company's written consent to such action; except in the cases of clauses (a) - (e), to the extent that such Person is disproportionately adversely affected thereby as compared with other participants in the industries in which such Person primarily operates.

"Parent Options" shall mean options to purchase Parent Common Stock.

"Participating Escrow Stock" shall mean Company Common Stock, Company Restricted Stock, Series A Preferred Stock and Series FF Preferred Stock,

"PBGC" shall mean the Pension Benefit Guaranty Corporation.

"Pending Claims Amount" shall mean, as of a specified date, the amount that would be necessary in Parent's reasonable good faith judgment to satisfy any then pending and unsatisfied or unresolved Claims specified in any Claim Notice delivered to the Stockholder Representative prior to such date if such Claims were resolved in full in favor of the Parent Indemnitees.

"Per Share Common Consideration" shall mean a number of shares of Parent Common Stock equal to (a)(i) the Company Equityholder Remaining Consideration minus the number obtained by multiplying (A) the total number of Escrowed Shares (i.e., 1,000,000) by (B) the Closing Price Per Share divided by (ii) the Adjusted Fully Diluted Number minus the number of shares of Common Stock issuable upon exercise of all Company Options (whether or not vested or exercisable) outstanding immediately prior to the Effective Time divided by (b) the Closing Price Per Share.

"<u>Per Share Common Consideration Equivalent Amount</u>" shall mean an amount equal to (a) the Remaining Consideration <u>divided</u> <u>by</u> (b) the Adjusted Fully Diluted Number.

"<u>Per Share Series B Preferred Consideration</u>" shall mean, with respect to a share of Series B Preferred Stock, a number of shares of Parent Common Stock equal to the Aggregate Series B Preferred Stock Consideration <u>multiplied by</u> a fraction, the numerator of which is the Liquidation Preference of such share of Series B Preferred Stock and the denominator of which is the Aggregate Series B Preferred Liquidation Preference, in each case immediately prior to the Effective Time.

"<u>Per Share Series C Preferred Consideration</u>" shall mean, with respect to a share of Series C Preferred Stock, a number of shares of Parent Common Stock equal to the Aggregate Series C Preferred Stock Consideration <u>multiplied by</u> a fraction, the numerator of which is the Liquidation Preference of such share of Series C Preferred Stock and the denominator of which is the Aggregate Series C Preferred Liquidation Preference, in each case immediately prior to the Effective Time.

"<u>Per Share Series D Preferred Consideration</u>" shall mean, with respect to a share of Series D Preferred Stock, a number of shares of Parent Common Stock equal to the Aggregate Series D Preferred Stock Consideration <u>multiplied by</u> a fraction, the numerator of which is the Liquidation Preference of such share of Series D Preferred Stock and the denominator of which is the Aggregate Series D Preferred Liquidation Preference, in each case immediately prior to the Effective Time.

"<u>Permits</u>" shall mean all licenses, permits, franchises, approvals, authorizations, consents or orders of, or filings with, any Governmental Authority, whether foreign, federal, state or local, or any other Person, necessary or desirable for the past, present or anticipated conduct of, or relating to the operation of the Business.

"Person" shall mean any individual, corporation (including any non-profit corporation), general or limited partnership, limited liability company, joint venture, estate, trust, association, organization, labor union, or other entity or governmental body.

"Personal Information" shall mean any information relating to an identified or identifiable natural person; an "identifiable person" is one who can be identified, including without limitation unique device or browser identifiers, names, addresses, telephone numbers, email addresses, social security numbers, and/or account information; and shall also mean "personal information" as defined by applicable laws relating to the collection, use, sharing, storage, and/or disclosure of information about an identifiable individual.

"Plan" shall mean (i) each employment, consulting, severance, termination, pension, retirement, supplemental retirement, excess benefit, profit sharing, bonus, incentive, deferred compensation, retention, transaction, change in control and similar plan, program, arrangement, agreement, policy or commitment, (ii) each compensatory stock option, restricted stock, stock unit, deferred stock, performance stock, stock appreciation, deferred stock or other equity or equity-based plan, program, arrangement, agreement, policy or commitment, (iii) each savings, life, health, disability, accident, medical, dental, vision, cafeteria, flex spending, adoption/dependent/employee assistance, tuition, vacation, paid time off, other welfare fringe benefit and other employee compensation plan, program, arrangement, agreement, policy or commitment, including in each case, each "employee benefit plan" as defined in Section 3(3) of ERISA and any trust, escrow, funding, insurance or other agreement related to any of the foregoing.

"<u>Pre-Closing Tax Period</u>" shall mean (a) any Tax period ending on or before the Closing Date and (b) with respect to any Straddle Period, the portion of such period ending on (and including) the Closing Date.

"<u>Price Per Share</u>" shall mean \$4.50 as adjusted appropriately in the case of any stock split, reverse stock split, stock dividend or similar recapitalization event with respect to Parent Common Shares.

"Pro Rata Allocation" shall mean, with respect to any Company Equityholder, the number of shares of Parent Common Stock issuable to such Company Equityholder pursuant to Article II hereof solely in respect of their Participating Escrow Stock (excluding, for the avoidance of doubt, shares of Parent Common Stock issuable upon exercise of Company Options assumed pursuant to Article II hereof) divided by the aggregate number of shares of Parent Common Stock issuable to all Company Equityholders pursuant to Article II hereof solely in respect of their Participating Escrow Stock (excluding, for the avoidance of doubt any shares of Parent Common Stock issuable upon exercise of Company Options assumed pursuant to Section 1.8.(b)).

"Pro Rata Portion" shall mean, with respect to any Company Equityholder, the number of shares of Parent Common Stock issuable to such Company Equityholder pursuant to Article II hereof (excluding, for the avoidance of doubt, shares of Parent Common Stock issuable upon exercise of Company Options assumed pursuant to Article II hereof) divided by the aggregate number of shares of Parent Common Stock issuable to all Company Equityholders pursuant to Article II hereof (excluding, for the avoidance of doubt any shares of Parent Common Stock issuable upon exercise of Company Options assumed pursuant to Section 1.8.(b)).

"Products" shall mean all products, technology and services of the Company or PocketGear developed (including products and services under development), owned, provided, distributed, marketed, imported for resale, sold or licensed out by or on behalf of the Company or PocketGear since inception or which the Company or PocketGear intends to sell, distribute, market or license out in the future.

"Property Tax" shall mean all real property Taxes, personal property Taxes and similar ad valorem Taxes.

"Regulations" shall mean any laws, statutes, ordinances, regulations, rules, notice requirements, court decisions, agency guidelines, principles of law and orders of any foreign, federal, state or local government and any other governmental department or agency, including, without limitation, Environmental Laws, energy, motor vehicle safety, public utility, zoning, building and health codes, occupational safety and health, privacy and security of personally identifiable information, the sending of commercial emails, and laws respecting employment practices, employee documentation, terms and conditions of employment and wages and hours.

"Related Party" shall mean with respect to any specified Person, (a) any Affiliate of such specified Person or any director, officer, general partner or managing member of such Affiliate; (b) any Person who serves as a director, officer, general partner, managing member, manager or in a similar capacity of such specified Person; (c) any immediate family member of a Person described in cause (b); or (d) any other Person who holds, individually or together with any Affiliate of such other Person and any members of such Person's immediate family, more than 5% of the outstanding equity or ownership interests of such specified Person (and any Affiliate, director, officer, general partner, managing member of manager of any such other Person described in this clause (d)).

"Remaining Consideration" shall mean (a) the Adjusted Aggregate Consideration minus (b) the Aggregate Series B Preferred Liquidation Preference minus (c) the Aggregate Series C Preferred Liquidation Preference minus (c) the Aggregate Series C Other Warrants Liquidation Preference minus (d) the Aggregate Series C Penny Warrants Liquidation Preference minus (e) the Aggregate Series D Preferred Liquidation Preference.

"Representative" shall mean any officer, director, principal, attorney, agent, employee or other representative.

"Restated Charter" shall mean the Company's certificate of incorporation filed with the Secretary of State of the State of Delaware on October 31, 2013, as the same may be amended and/or restated from time to time.

"Securities Act" shall mean the Securities Act of 1933, as amended, and the rules and regulations promulgated thereunder.

"Selling Expenses" shall mean all discounts, selling commissions, broker fees and stock transfer Taxes applicable to the sale of Registrable Securities and fees and disbursements of counsel for the Company Stockholders.

"Series A Preferred Stock" shall mean the Series A Convertible Preferred Stock, par value \$0.001 per share, of the Company.

"Series B Preferred Stock" shall mean the Series B Convertible Preferred Stock, par value \$0.001 per share, of the Company.

"Series C Other Warrants" shall mean the Preferred Stock Purchase Warrant dated April 4, 2013 initially exercisable to purchase 112,800 shares of Series C Preferred Stock and the Preferred Stock Purchase Warrant dated October 31, 2013 initially exercisable to purchase 67,680 shares of Series C Preferred Stock, in each case for an exercise price equal to \$7.4139234 per share.

"Series C Other Warrants Consideration" shall mean with respect to a Series C Other Warrants number of shares of Parent Common Stock equal to the Aggregate Series C Other Warrants Stock Consideration multiplied by a fraction, the numerator of which is the number of Series C Warrant Shares underlying such Series C Other Warrant and the denominator of which is the number of Series C Warrants underlying all Series C Other Warrants that are outstanding and unexercised immediately prior to the Effective Time.

"Series C Penny Warrants" shall mean the Preferred Stock Purchase Warrant dated April 4, 2013 initially exercisable to purchase 263,200 shares of Series C Preferred Stock and the Preferred Stock Purchase Warrant dated October 31, 2013 initially exercisable to purchase 157,920 shares of Series C Preferred Stock, in each case for an exercise price equal to \$0.001 per share.

"Series C Penny Warrants Consideration" shall mean with respect to a Series C Penny Warrant a number of shares of Parent Common Stock equal to the Aggregate Series C Penny Warrants Stock Consideration <u>multiplied by</u> a fraction, the numerator of which is the number of Series C Warrants underlying such Series C Penny Warrant and the denominator of which is the number of Series C Warrant Shares underlying all Series C Penny Warrants that are outstanding and unexercised immediately prior to the Effective Time.

"Series C Preferred Stock" shall mean the Series C Convertible Preferred Stock, par value \$0.001 per share, of the Company.

"Series C Warrants" shall mean the Series C Penny Warrants and the Series C Other Warrants.

"Series C Warrant Shares" shall mean the shares of Series C Preferred Stock underlying the Series C Warrants.

"Series C Warrantholders" shall mean holders of the Series C Warrants.

"Series D Preferred Stock" shall mean the Series D Convertible Preferred Stock, par value \$0.001 per share, of the Company.

"Series FF Preferred Stock" shall mean the Series FF Convertible Preferred Stock, par value \$0.001 per share, of the Company

"Software" shall mean any and all (i) computer programs, including any and all software implementations of algorithms, models and methodologies, whether in Source Code or object code, (ii) databases, compilations, data aggregation programs and search engine technologies, including any and all data and collections of data, whether machine readable or otherwise, (iii) descriptions, flow-charts and other work product used to design, plan, organize and develop any of the foregoing and (iv) all user documentation, including user manuals and training materials, relating to any of the foregoing.

"Source Code" shall mean computer software in a form that is readily suitable for review and edit by trained programmers, including related programmer comments embedded therein. For avoidance of doubt, Source Code excludes any software that is wholly or substantially in binary form and any documentation or other information provided or used with such software.

"Stock Escrow Amount" equals \$4,500,000.

"Straddle Period" shall mean any Tax period beginning before or on the Closing Date and ending after the Closing Date.

"Subsidiary" shall mean, with respect to a Person, any corporation, partnership, limited liability company, joint venture or other entity in which such Person, directly or indirectly, owns a majority of the capital stock or other equity interests or which such Person otherwise, directly or indirectly, controls through an investment or participation in the equity of such entity.

"Tangible Property" shall mean all computer equipment and other furniture, equipment and other tangible personal property used in the Business, including any such furniture, equipment or other tangible personal property used by the Company or PocketGear pursuant to a license, lease or similar right.

"Tax" shall mean any and all federal, state, local or foreign taxes, including, without limitation, any net income, alternative or addon minimum, gross income, gross receipts, branch profits, capital stock, sales, use, ad valorem, value added, transfer, franchise, profits, license, registration, recording, documentary, conveyancing, gains, withholding, payroll, employment, social security, unemployment, disability, excise, severance, stamp, occupation, premium, property, environmental or windfall profit, escheat, custom duty, estimated tax or other tax, or other like assessment or charge of any kind whatsoever, together with any interest, penalty, addition to tax or additional amount imposed by any Tax Authority.

"<u>Tax Authority</u>" shall mean any governmental agency, board, bureau, body, department or authority of any United States federal, state or local jurisdiction or any foreign jurisdiction, having or purporting to exercise jurisdiction with respect to any <u>Tax</u>.

"<u>Tax Return</u>" shall mean any return, declaration, report, claim for refund, information return or statement of any kind relating to Taxes, including any schedule or attachment thereto and any amendment thereof, filed or required to be filed with any Tax Authority.

"Treasury Regulations" shall mean the United States Treasury regulations promulgated under the Code.

"Working Capital" shall mean the Current Assets minus the Current Liabilities of the Company.

"Working Capital Target" shall mean \$1,350,000.

"Year End Financial Statements" shall mean the audited Balance Sheets and related audited statements of income of the Company as of and for the fiscal years ended as of December 31, 2011, 2012 and 2013.

The following terms shall have the meanings defined for such terms in the Sections set forth below:

Defined Term	Section
10b5-1 plan	5.17(c)
280(G) Vote	5.8.(c)
401(k) Plan	5.8.(a)
Adjustment Amount	1.12.(a)(ii)
Advertiser	3.9.(a)(viii)
Advertiser and Publisher Agreements	3.9.(a)(viii)
Affiliate Agreements	3.9.(a)(v)
Agreement	Preamble
Arbitrator	9.16.(a)
Assumed Option Consideration	1.10.(a)(iii)
Balance Sheet Date	3.6
Basket	7.2.(h)
Cardholder Data	3.18.(j)
Certificate	1.4
Certificate of Merger	1.2
Change of Recommendation	5.9(d)
Change of Recommendation Notice	5.9(d)
Claim	7.2.(e)(i)
Claim Notice	7.2.(e)(i)
Closing	2.1
Closing Consideration Schedule	1.10
Closing Date	2.1
Closing Failure Notice	8.1.(d)
Closing Net Debt	1.12.(b)(i)

D.C. LT	g .:
Defined Term Classing Statement	Section 1 12 (b)(i)
Closing Statement	1.12.(b)(i)
Closing Working Capital	1.12.(b)(i)
Commercial Software Licenses	3.18.(b)
Company	Preamble
Company Equityholder Indemnitees	7.2.(b)
Company Options	1.8.(a)
Company Preferred Stock	1.6.(a)
Company Related Parties	8.3.(c)
Company Tax Representation Letter	5.16
Contaminants	3.18.(j)
Customer and Publisher Agreements	3.18.(b)
Damages	7.2
Deemed Exercise Option	1.8(c)
Deemed Exercise Option Consideration	1.10(a)(iv)
DGCL	Recitals
Disclosure Document	5.2
Dispute Notice	9.16.(a)
Disputes	9.16
Dissenting Shares	1.9.(a)
Dissenting Stockholder	1.9.(d)
D&O Indemnified Party	5.14
Effective Time	1.2
Escrow Account	1.11
Escrowed Shares	1.11
Estimated Net Debt	1.12.(a)(i)
Estimated Working Capital	1.12.(a)(i)
Excess Dissenting Share Payments	1.9.(c)
Exchange Agent	1.4.(a)
Exchange Fund	1.4.(a)
Expense Fund	7.2(g)(vii)
Final Adjustment Amount	1.12.(c)(i)
Fundamental Representations	7.1
Information Technology Systems	3.18.(j)
IP Representations	7.1
JAMS	9.16.(b)
Joint Proxy Statement	5.9
Letter of Transmittal	1.4.(c)(i)
Material Contracts	3.9.(a)
Merger	Recitals
Merger Sub	Preamble
Open Source	3.18.(1)
Outside Date	8.1.(b)
Parent	Preamble
Parent Closing Stock Consideration	1.6(h)
Parent Indemnitees	7.2
Parent Related Parties	8.3.(c)
Parent SEC Documents	4.6
Parent Stockholder Approval	4.2
Parent Tax Representation Letter	5.16
Parent's Accountants	5.2
Periodic Report	5.2
•	

<u>Defined Term</u>	Section
PocketGear	3.2.(a)
Pre-Closing Statement	1.12.(a)(i)
Proposed Acquisition Transaction	5.3
Publisher	3.9.(a)(viii)
Restricted Stock Award	1.8(d)
Restricted Stock Consideration	1.10(a)(v)
SEC	9.10
Section 280G Approval	5.8.(c)
Section 280G Soliciting Materials	5.8.(c)
Stockholder Notice	5.10
Stockholder Representative	Preamble
Summary of Terms	8.2.(b)
Surviving Corporation	1.1
Tax Benefit	(f)
Tax Proceeding	5.7.(a)
Terminated Agreements	5.13
Termination Amount	8.3.(a)
Transfer Taxes	5.7.(f)
Treasury Shares	1.6.(e)
Voting Debt	3.2.(c)
WARN Act	3.16.(f)
Warrant Consideration	1.10.(a)(ii)
	.,.,

9.2. Notices. All notices, requests, demands, Claims and other communications which are required or may be given under this Agreement shall be in writing and shall be deemed to have been duly given when received if personally delivered; when transmitted if transmitted by electronic mail or confirmed facsimile with a copy sent by another means specified herein; the Business Day after it is sent if sent for next day delivery to a domestic address by recognized overnight delivery service (e.g. Federal Express); and five Business Days after the date mailed by certified or registered mail, postage prepaid, if sent by certified or registered mail, return receipt requested. In each case notice shall be sent to:

If to the Company (prior to the Closing), addressed to:

Appia, Inc. 320 Blackwell Street 3rd Floor Durham, NC 27701 Attn: Judson S. Bowman Telephone: (919) 251-6144

Telephone: (919) 251-614 E-Mail: jud@appia.com

With a copy to (which shall not constitute notice):

Goodwin Procter LLP Exchange Place Boston, Massachusetts 02109

Fax: (617) 523-1231

Attention: Joseph C. Theis Jr., Esq. Email: jtheis@goodwinprocter.com

If to the Stockholder Representative, addressed to:

Shareholder Representative Services LLC 1614 15th Street, Suite 200 Denver, CO 80202 Attn: Managing Director

Telephone: (303) 648-4085 Fax: (303) 623-0294

E-Mail: deals@srsacquiom.com

With a copy to (which shall not constitute notice):

Goodwin Procter LLP Exchange Place Boston, Massachusetts 02109

Fax: (617) 523-1231

Attention: Joseph C. Theis Jr., Esq. Email: jtheis@goodwinprocter.com

If to Parent, Merger Sub or the Surviving Corporation, addressed to:

Mandalay Digial Group, Inc. 1300 Guadelupe Austin, TX 78701 Attn: Bill Stone

Telephone: 512-365-9991

E-Mail: Bill@digitalturbine.com

With a copy to (which shall not constitute notice):

Latham & Watkins LLP 355 South Grand Avenue Los Angeles, California 90071 Attn: W. Alex Voxman, Esq. and David M. Wheeler, Esq.

Telephone: (213) 485-1234 Fax: (213) 891-8763

E-Mail: alex.voxman@lw.com and david.wheeler@lw.com

or to such other place and with such other copies as either party may designate as to itself by written notice to the others.

- 9.3. <u>Rules of Construction</u>. The parties agree that they have been represented by counsel during the negotiation and execution of this Agreement and, therefore, waive the application of any law, regulation, holding or rule of construction providing that ambiguities in any agreement or other document will be construed against the party drafting such agreement or document.
- 9.4. <u>Titles</u>. The table of contents, titles, captions or headings of the Articles and Sections herein are inserted for convenience of reference only and are not intended to be a part of or to affect the meaning or interpretation of this Agreement.
- 9.5. Entire Agreement. This Agreement, including the Exhibits and Schedules hereto, the Disclosure Schedules and the other agreements, documents and written understandings referred to herein or otherwise entered into or delivered by the parties hereto on the date of this Agreement (including without limitation the

Ancillary Agreements), constitute the entire agreement and understanding and supersede all other prior covenants, agreements (including any summaries of terms between the parties), undertakings, obligations, promises, arrangements, communications, representations and warranties, whether oral or written, by any party hereto or by any director, officer, employee, agent, Affiliate or Representative of any party hereto; provided, that the provisions under the caption "Confidentiality" in the Summary of Terms shall survive the termination of this Agreement and remain in effect if the Closing does not occur. There are no covenants, agreements, undertakings or obligations with respect to the subject matter of this Agreement other than those expressly set forth or referred to herein or in other agreements, documents and written understandings entered into or delivered by the parties hereto on the date of this Agreement, and no representations or warranties of any kind or nature whatsoever, express or implied, including any implied warranties of merchantability or fitness for a particular purpose, are made or shall be deemed to be made herein by the parties hereto except those expressly made herein.

- 9.6. <u>Assignment</u>. Neither this Agreement nor any of the rights or obligations hereunder may be assigned by the Company without the prior written consent of Parent. Parent and Merger Sub may, without the consent of the Company assign all or any portion of their rights and obligations hereunder; <u>provided</u> that, if such assignment by Parent and Merger Sub is made prior to the Closing, then such assignment shall require the prior written consent of the Stockholder Representative unless the assignee(s) is(are) an Affiliate(s) of Parent or Merger Sub; <u>provided</u>, <u>further</u>, that such assignee(s) execute(s) a joinder to and agree(s) to be bound by this Agreement. No such assignment shall release the assignor from any of its obligations hereunder. In addition, Parent and Merger Sub may assign their rights hereunder to a lender as collateral security.
- 9.7. <u>Amendment or Modification</u>. This Agreement may not be amended or modified (before or after receipt of the Parent Stockholder Approval) except in an instrument in writing signed by the Stockholder Representative and Parent. No amendment, supplement, modification or waiver of this Agreement shall be binding unless executed in writing by the party to be bound thereby. For the purposes of this <u>Section 9.7</u>, any amendment, modification or waiver of this Agreement signed by the Stockholder Representative shall be binding upon and effective against the Company Equityholders whether or not they have signed such amendment, modification or waiver.
- 9.8. Waiver. Except where a specific period for action or inaction is provided herein, neither the failure nor any delay on the part of any party in exercising any right, power or privilege under this Agreement or the documents referred to in this Agreement shall operate as a waiver thereof, nor shall any waiver on the part of any party of any such right, power or privilege, nor any single or partial exercise of any such right, power or privilege, preclude any other or further exercise thereof or the exercise of any other such right, power or privilege. The failure of a party to exercise any right conferred herein within the time required shall cause such right to terminate with respect to the transaction or circumstances giving rise to such right, but not to any such right arising as a result of any other transactions or circumstances. No party shall be deemed to have waived any claim arising out of this Agreement, or any power, right, privilege or remedy under this Agreement, unless the waiver of such claim, power, right, privilege or remedy is expressly set forth in a written instrument duly executed and delivered on behalf of such Party; and any such waiver shall not be applicable or have any effect except in the specific instance in which it is given.
- 9.9. Severability. If any term or other provision of this Agreement is invalid, illegal or incapable of being enforced as a result of any rule of law or public policy, all other terms and other provisions of this Agreement shall nevertheless remain in full force and effect so long as the economic or legal substance of the transactions contemplated by this Agreement is not affected in any manner materially adverse to any party. Upon such determination that any term or other provision is invalid, illegal or incapable of being enforced, the parties hereto shall negotiate in good faith to modify this Agreement so as to effect the original intent of the parties as closely as possible in an acceptable manner to the end that the transactions contemplated by this Agreement are fulfilled to the greatest extent possible.
- 9.10. <u>References</u>. All references to "days" or "months" shall be deemed references to calendar days or months. All references to "\$" or "dollars" shall be deemed references to United States dollars. Unless the context

otherwise requires, any reference to an "Article", "Section," "Exhibit," or "Schedule" shall be deemed to refer to an article of this Agreement, section of this Agreement, exhibit to this Agreement or a schedule to this Agreement, as applicable. Any reference to any federal, state, county, local or foreign statute or law shall be deemed also to refer to all rules and regulations promulgated thereunder, unless the context requires otherwise. For all purposes of and under this Agreement, (a) the words "includer", "includes" and "including" shall be deemed to be immediately followed by the words "without limitation"; (b) words (including defined terms) in the singular shall be deemed to include the plural and vice versa; (c) words of one gender shall be deemed to include the other genders as the context requires; (d) "or" is not exclusive; (e) the word "will" shall be construed to have the same meaning and effect as the word "shall", (f) unless otherwise stated, any reference herein to any Person shall be construed to include such Person's successors and assigns, (g) the words "made available to Parent" or other words of similar import refer to the Company having made the applicable documents available to Parent by posting them in the Company's online data room located at https://services.intralinks.com (as reflected on the data room index dated as of the date hereof set forth on Schedule 9.10) and granting Parent access to such data room to review such documents and (h) the terms "hereof," "herein," "hereto," "herewith", "hereunder" and any other words of similar import shall, unless otherwise stated, be construed to refer to this Agreement as a whole (including the exhibits hereto and the Disclosure Schedules) and not to any particular term or provision of this Agreement, unless otherwise specified. Subject to Section 9.2, any reference to any notice or other communication received "in writing" shall be deemed to include any written notice or communication, including any notice or communication by mail or other means of physical delivery, facsimile, electronic mail or any other means of written communication.

- 9.11. <u>Burden and Benefit</u>. This Agreement shall be binding upon and shall inure to the benefit of, the parties hereto and their respective successors and permitted assigns. This Agreement and all of its conditions and provisions are for the sole and exclusive benefit of the parties hereto and their respective successors and permitted assigns, and nothing in this Agreement, except as set expressly stated in Section 5.14.(a), is intended to confer upon any Person other than the parties hereto any rights or remedies of any nature whatsoever under or by reason of this Agreement or any provision hereof; <u>provided</u>, <u>however</u>, that any Person that is not a party to this Agreement but, by the terms of <u>Article VII</u>, is entitled to indemnification, shall be considered a third-party beneficiary of this Agreement, with full rights of enforcement as though such Person was a signatory to this Agreement.
- 9.12. Governing Law. This Agreement (and any claim or controversy arising out of or relating to this Agreement) shall be governed by the law of the State of Delaware without regard to conflict of law principles that would result in the application of any law other than the law of the State of Delaware.
- 9.13. Consent to Jurisdiction. Subject to Section 9.16 hereof, each party to this Agreement hereby irrevocably and unconditionally submits, for itself and its property, to the exclusive jurisdiction of the Chancery Court of the State of Delaware and any state appellate court therefrom within the State of Delaware (or, if the Chancery Court of the State of Delaware declines to accept jurisdiction over a particular matter, any state or federal court within the State of Delaware), in any action or proceeding arising out of or relating to this Agreement or the agreements delivered in connection herewith or the transactions contemplated hereby or thereby or for recognition or enforcement of any judgment relating thereto, and each of the parties hereby irrevocably and unconditionally (a) agrees not to commence any such action or proceeding except in such courts, (b) agrees that any claim in respect of any such action or proceeding may be heard and determined in such courts, (c) waives, to the fullest extent it may legally and effectively do so, any objection which it may now or hereafter have to the laying of venue of any such action or proceeding in any such courts, and (d) waives, to the fullest extent permitted by law, the defense of an inconvenient forum to the maintenance of such action or proceeding in any such courts. Each of the parties hereto agrees that a final judgment in any such action or proceeding shall be conclusive and may be enforced in other jurisdictions by suit on the judgment or in any other manner provided by law. Each party to this Agreement irrevocably consents to service of process in the manner provided for notices in Section 9.2 hereof. Nothing in this Agreement will affect the right of any party to this Agreement to serve process in any other manner permitted by law.

- 9.14. Waiver of Trial by Jury. EACH PARTY TO THIS AGREEMENT ACKNOWLEDGES AND AGREES THAT ANY CONTROVERSY WHICH MAY ARISE UNDER THIS AGREEMENT IS LIKELY TO INVOLVE COMPLICATED AND DIFFICULT ISSUES, AND THEREFORE IT HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVES ANY RIGHT IT MAY HAVE TO A TRIAL BY JURY IN RESPECT OF ANY LITIGATION DIRECTLY OR INDIRECTLY ARISING OUT OF OR RELATING TO THIS AGREEMENT AND ANY OF THE AGREEMENTS DELIVERED IN CONNECTION HEREWITH OR THE TRANSACTIONS CONTEMPLATED HEREBY OR THEREBY. EACH PARTY CERTIFIES AND ACKNOWLEDGES THAT (A) NO REPRESENTATIVE, AGENT OR ATTORNEY OF ANY OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE EITHER OF SUCH WAIVERS, (B) IT UNDERSTANDS AND HAS CONSIDERED THE IMPLICATIONS OF SUCH WAIVERS, (C) IT MAKES SUCH WAIVERS VOLUNTARILY, AND (D) IT HAS BEEN INDUCED TO ENTER INTO THIS AGREEMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS SECTION 9.14.
- 9.15. <u>Legal Fees</u>. If any party to this Agreement brings an action to enforce its rights under this Agreement, the prevailing party shall be entitled to recover its costs and expenses, including without limitation reasonable legal fees, incurred in connection with such action, including any appeal of such action.
- 9.16. <u>Arbitration</u>. It is understood and agreed between the parties hereto that if the transactions contemplated by this Agreement are consummated, from and after the Closing Date, any and all claims, grievances, demands, controversies, causes of action or disputes of any nature whatsoever (including, but not limited to, tort and contract claims, and claims upon any law, statute, order, or regulation) (hereinafter "<u>Disputes</u>"), arising out of, in connection with, or in relation to (a) this Agreement, or (b) questions of arbitrability under this Agreement, shall be resolved by final, binding, nonjudicial arbitration in accordance with the Federal Arbitration Act, 9 U.S.C. Section 1, et seq. pursuant to the following procedures:
- (a) Any party may send another party or parties written notice identifying the matter in dispute and invoking the procedures of this Section (the "<u>Dispute Notice</u>"). Within thirty (30) days from delivery of the Dispute Notice, each party involved in the dispute shall meet at a mutually agreed location in Los Angeles, California, for the purpose of determining whether they can resolve the dispute themselves by written agreement, and, if not, whether they can agree upon an impartial third-party arbitrator (the "<u>Arbitrator</u>") to whom to submit the matter in dispute for final and binding arbitration.
- (b) If such parties fail to resolve the dispute by written agreement or agree on the Arbitrator within the later of thirty (30) days from any such initial meeting or within sixty (60) days from the delivery of the Dispute Notice, any such party may make written application to the Judicial Arbitration and Mediation Services ("JAMS"), in Los Angeles, California for the appointment of a single Arbitrator to resolve the dispute by arbitration. At the request of JAMS the parties involved in the dispute shall meet with JAMS at its offices within thirty (30) days of such request to discuss the dispute and the qualifications and experience which each party respectively believes the Arbitrator should have; provided, however, that the selection of the Arbitrator shall be the exclusive decision of JAMS and shall be made within thirty (30) days of the written application to JAMS. The Arbitrator shall be a disinterested party.
- (c) Within sixty (60) days of the selection of the Arbitrator, the parties involved in the dispute shall meet in Los Angeles, California with such Arbitrator at a place and time designated by such Arbitrator after consultation with such parties and present their respective positions on the dispute. Each party shall have no longer than one day to present its position, the entire proceedings before the Arbitrator shall be no more than three consecutive days, and the decision of the Arbitrator shall be made in writing no more than thirty (30) days following the end of the proceeding. Such an award shall be a final and binding determination of the dispute and shall be fully enforceable as an arbitration decision in any court having jurisdiction and venue over such parties. The prevailing party or parties (as determined by the Arbitrator) shall in addition be awarded by the Arbitrator such party's or parties' own legal fees and expenses in connection with such proceeding. The non-prevailing party or parties (as determined by the Arbitrator) shall pay the Arbitrator's fees and expenses.

- 9.17. Specific Performance. Each of the parties hereto acknowledges and agrees that the other parties would be damaged irreparably, and in a manner for which monetary damages would not be an adequate remedy, in the event any of the provisions of this Agreement are not performed in accordance with its specific terms or otherwise are breached. Accordingly, each of the parties hereto agrees that the other parties shall be entitled to an injunction or injunctions to prevent breaches of the provisions of this Agreement and to enforce specifically this Agreement and the terms and provisions hereof in any action instituted in any court of the United States or any state thereof having jurisdiction over the parties and the matter, in addition to any other remedy to which they may be entitled, at law or in equity.
- 9.18. <u>Cumulative Remedies</u>. All rights and remedies of either party hereto are cumulative of each other and of every other right or remedy such party may otherwise have at law or in equity, and the exercise of one or more rights or remedies shall not prejudice or impair the concurrent or subsequent exercise of other rights or remedies.
- 9.19. Expenses. Except as otherwise expressly provided herein, whether or not the transactions contemplated hereby are consummated, all costs and expenses incurred in connection with this Agreement and the transactions contemplated hereby shall be paid by the party incurring such expenses (it being understood that any unpaid Company Transaction Expenses shall be paid by the Surviving Corporation following the Closing and shall be taken into account in the calculation of Working Capital of the Company as of the Closing Date as Current Liabilities).
- 9.20. Representation by Counsel. Each party hereto represents and agrees with each other that it has been represented by or had the opportunity to be represented by, independent counsel of its own choosing, and that it has had the full right and opportunity to consult with its respective attorney(s), that to the extent, if any, that it desired, it availed itself of this right and opportunity, that it or its authorized officers (as the case may be) have carefully read and fully understand this Agreement in its entirety and have had it fully explained to them by such party's respective counsel, that each is fully aware of the contents thereof and its meaning, intent and legal effect, and that it or its authorized officer (as the case may be) is competent to execute this Agreement and has executed this Agreement free from coercion, duress or undue influence.
- 9.21. Execution and Counterparts. This Agreement may be executed in one or more counterparts, each of which when executed shall be deemed an original and all of which together shall constitute one and the same instrument. The parties agree that this Agreement shall be legally binding upon the electronic transmission, including by facsimile or email, by each party of a signed signature page to this Agreement to the other party.

9.22. Acknowledgement of Representation.

- (a) It is acknowledged by each of the Parties that the Company has retained Goodwin Procter LLP ("Goodwin") to act as its counsel in connection with this Agreement, the Ancillary Documents or any transaction contemplated hereby or thereby (the "Current Representation"), and that no other Party to this Agreement has the status of a client of Goodwin for conflict of interest or any other purposes as a result thereof.
- (b) <u>Affirmation of Representation</u>. Parent and the Company hereby agree that Goodwin may represent the Stockholders, the Seller Representative or any officer, director, manager, employee, shareholder, partner or member of the Company (any such Person, a "<u>Designated Person</u>") in any matter involving or arising from the Current Representation, including any interpretation or application of this Agreement or any Ancillary Document, and including for the avoidance of doubt any litigation, arbitration, dispute or mediation between or among Parent, the Company or any of their respective Affiliates, and any Designated Person, even though the interests of such Designated Person may be directly adverse to Parent, the Company or any of their respective Affiliates, and even though Goodwin may have represented the Company in a substantially related matter, or may be representing the Company in ongoing matters.

- (c) <u>Waiver of Conflict</u>. Parent and the Company hereby waive and agree not to assert (i) any claim that Goodwin has a conflict of interest in any representation described in <u>Section 6.1.(b)</u> above, and (ii) any confidentiality obligation with respect to any communication between Goodwin and any Designated Person occurring during the Current Representation.
- (d) <u>Retention of Privilege</u>. Parent and the Company hereby agree that as to all communications (whether before, at or after the Closing) between Goodwin and any Designated Person that relate in any way to the Current Representation, the attorney-client privilege and all rights to any other evidentiary privilege, and the protections afforded to information relating to representation of a client under applicable rules of professional conduct, belong to such Designated Person and may be controlled by such Designated Person and shall not pass to or be claimed by Parent or the Company.
- (e) <u>Further Assurances</u>. Parent and the Company agree to take, and to cause their respective Affiliates to take, all steps necessary to implement the intent of this <u>Section 9.21</u>. Parent and the Company further agree that Goodwin and its partners and employees are third party beneficiaries of this <u>Section 9.21</u>.

[Signature Page Follows]

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be duly executed on their respective behalf, by their respective officers thereunto duly authorized, all as of the day and year first set forth above.

MANDALAY DIGITAL GROUP, INC.

By: /s/ Bill Stone

Name: Bill Stone

Title: Chief Executive Officer

DTM MERGER SUB, INC.

By: /s/ Bill Stone

Name: Bill Stone

Title: Chief Executive Officer

Signature Page to Merger Agreement

APPIA, INC.

By: /s/ Judson S. Bowman

Name: Judson S. Bowman

Title: CEO

Signature Page to Merger Agreement

SHAREHOLDER REPRESENTATIVE SERVICES LLC, solely in its capacity as the Stockholder Representative

Ву:	/s/ illegible		
	Name:		
	Title:		

Signature Page to Merger Agreement

Appendix B – Form of Support Agreement

Appendix B

SUPPORT AGREEMENT

This **SUPPORT AGREEMENT**, dated as of November 13, 2014 (this "<u>Agreement</u>"), is by and among Mandalay Digital Group, Inc., a Delaware corporation ("<u>Parent</u>"), on the one hand, and the Person listed on the signature page hereto (the "<u>Stockholder</u>") who is a (i) holder of shares of the Company's Series A Preferred Stock (the <u>Series A Preferred Stockholders</u>"), (ii) holder of shares of the Company's Series B Preferred Stock (the "<u>Series B Preferred Stockholders</u>"), (iii) holder of shares of the Company's Series C Preferred Stock (the "<u>Series C Preferred Stockholders</u>"), (iv) holder of shares of the Company's Series D Preferred Stock (the "<u>Series D Preferred Stockholders</u>" and, together with the Series A Preferred Stockholders, the Series B Preferred Stockholders, the Series C Preferred Stockholders"), and/or (v) holder of shares of the Company's Common Stock (the "<u>Common Stockholders</u>" and, together with the Preferred Stockholders, the "<u>Stockholders</u>"), on the other hand.

WHEREAS, concurrently with the execution of this Agreement, Parent, DTM Merger Sub, Inc. ("Merger Sub"), Appia, Inc. (the "Company") and Shareholder Representative Services LLC (the "Stockholder Representative") have entered into an Agreement and Plan of Merger (as such agreement may be amended from time to time and whether or not such agreement has been terminated, the "Merger Agreement") pursuant to which, among other things, Merger Sub will merge with and into the Company, in accordance with the General Corporation Law of the State of Delaware (the "DGCL") and subject to the conditions set forth herein, which merger will result in, among other things, the Company becoming a wholly owned subsidiary of Parent (the "Merger") (any term not otherwise defined herein shall have the meaning given such term in the Merger Agreement);

WHEREAS, the Stockholder owns beneficially and of record shares of Company Capital Stock (together with any shares of Company Capital Stock of which record or beneficial ownership is acquired after the date hereof by the Stockholder (including pursuant to the exercise of Company Options or Series C Warrants), the "Shares"); and

WHEREAS, as a condition to the willingness of, and as an inducement to, Parent and Merger Sub to enter into the Merger Agreement, the Stockholder has agreed to enter into this Agreement and certain other Stockholders have agreed to enter into agreements identical to this Agreement pursuant to which, among other things, the Stockholders have agreed to execute and deliver, in accordance with the requirements of the DGCL, a written consent to, and to vote all Shares owned by them in favor of, the approval and adoption of the Merger Agreement, the Merger and the other transactions contemplated thereby.

NOW, THEREFORE, in consideration of the foregoing and the mutual representations, warranties, covenants and agreements herein contained, and intending to be legally bound hereby, the parties hereto hereby agree as follows:

ARTICLE I CERTAIN DEFINITIONS

1.1 <u>Capitalized Terms</u>. Capitalized terms used in this Agreement and not defined herein shall have the meanings ascribed to them in the Merger Agreement.

ARTICLE II CONSENT, VOTING MATTERS AND PROXY

2.1 Written Consent. In furtherance of the foregoing, the Stockholder shall execute and deliver to the Company within 48 hours after the Form S-4 is declared effective by the SEC, an Action by Written Consent of Stockholders in the form attached hereto as Exhibit A to consent to, approve of and adopt (i) the amendment to the Restated Charter attached hereto as Exhibit B (the "Charter Amendment") and (ii) the Merger Agreement, the Merger and the other transactions contemplated thereby.

2.2 Agreement to Vote.

- (a) From the date hereof until the earlier to occur of (x) the Effective Time and (y) the termination of the Merger Agreement in accordance with its terms (such earlier event, the "Termination Event"), the Stockholder shall, from time to time, at the request of Parent, at any meeting (whether annual or special and whether or not an adjourned or postponed meeting) of all or any class of the Company's stockholders, however called, or in connection with any written consent of the Company's stockholders, as applicable, (I) waive the invalidity or absence of notice of any such meeting, (II) appear at such meeting or otherwise cause the Stockholder's shares of Company Capital Stock to be counted as present thereat for purposes of establishing a quorum and (III) if and to the extent the following matters are put to a vote of the Company's stockholders, vote or consent (or cause to be voted or consented), in person or by proxy (and including by Section 2.1), all of the Stockholder's shares of Company Capital Stock as to which the Stockholder has, directly or indirectly, the right to vote or direct the voting:
 - (i) in favor of the adoption and approval of the Charter Amendment, the Merger Agreement, the terms thereof, the Merger, the other transactions contemplated thereby and any actions required in furtherance thereof and hereof and in favor of any adjournment of any meeting of the Company's stockholders for the purpose of soliciting additional votes in favor of any of the foregoing;
 - (ii) against any action or agreement that would be reasonably likely to result in a material breach of any representation, warranty, covenant or agreement of the Company under the Merger Agreement; and
 - (iii) against any of the actions prohibited by Section 5.3 of the Merger Agreement.
- (b) From the date hereof until the occurrence of a Termination Event, the Stockholder agrees that it shall not (i) enter into any Contract or understanding with any Person to vote or give voting instructions in a manner inconsistent with this Agreement (including this Section 2.2) or (ii) grant any proxy or power of attorney for another Person to (A) vote, (B) execute any consent with respect to, (C) deposit into a voting trust or (D) enter into any other Contract or understanding to vote, any shares of Company Capital Stock in any manner.
- (c) From the date hereof until the occurrence of a Termination Event, the Stockholder hereby constitutes and appoints as the proxies of the Stockholder and hereby grants a power of attorney to Bill Stone and Andrew Schleimer (or such other Persons designated by Parent), and each of them, with full power of substitution, to vote all the Stockholder's shares of Company Capital Stock (whether by proxy, in person or by written consent) in the manner provided in this Agreement upon the failure of the Stockholder to vote his, her or its shares of Company Capital Stock in accordance with the terms of this Agreement (or upon any attempt to vote his, her or its shares of Company Capital Stock in a manner which is inconsistent with the terms of this Agreement). Each of the proxy and power of attorney granted pursuant to the immediately preceding sentence is given in consideration of the agreements and covenants of Parent and the parties in connection with the transactions contemplated by this Agreement and, as such, each is coupled with an interest and shall be irrevocable until the occurrence of a Termination Event. The Stockholder hereby revokes any and all previous proxies or powers of attorney with respect to the Stockholder's shares of Company Capital Stock. Notwithstanding anything to the contrary provided in this Agreement, Parent shall provide the Stockholder with prior written notice of Parent's intent to exercise the proxy set forth in this Section 2.2(c).

ARTICLE III REPRESENTATIONS AND WARRANTIES OF THE STOCKHOLDER

The Stockholder hereby represents and warrants to Parent as follows:

- **3.1** <u>Organization</u>. The Stockholder, if not a natural person, is duly organized, validly existing and in good standing under the laws of its jurisdiction of incorporation or organization.
- 3.2 Authorization. The Stockholder has all requisite power and authority, and has taken all action necessary, to execute and deliver this Agreement, to consummate the transactions contemplated hereby and thereby and to perform the Stockholder's obligations hereunder and thereunder. The execution and delivery by the Stockholder of this Agreement to which the Stockholder is to be a party, and the consummation by the Stockholder of the transactions contemplated hereby and thereby have been duly authorized by all necessary corporate or other action and no other proceedings or actions on the part of the Stockholder are necessary to authorize this Agreement to which the Stockholder is to be a party and the transactions contemplated hereby and thereby. This Agreement has been duly executed and delivered by the Stockholder and is legal, valid and binding obligations of the Stockholder enforceable against the Stockholder in accordance with their terms, in each case, except as such enforceability may be limited by (a) bankruptcy, insolvency, moratorium, reorganization and other similar laws affecting creditors' rights generally and (b) the general principles of equity, regardless of whether asserted in a proceeding in equity or at law.
- 3.3 No Conflict. The execution and delivery by the Stockholder of this Agreement does not (a) if the Stockholder is an entity, conflict with or violate any provision of the Organizational Documents of the Stockholder; (b) conflict with or violate any law applicable to the Stockholder or by which any property or asset of the Stockholder is bound; or (c) result in any breach or default under any Contract to which the Stockholder is a party or to which any of the Stockholder's property or assets are subject, except in the case of clauses (b) and (c), as would not reasonably be expected to adversely affect the ability of the Stockholder to consummate the transactions contemplated by this Agreement, the Merger Agreement or the Ancillary Agreements to which the Stockholder is to be a party.
- **3.4** <u>Absence of Litigation, Claims and Orders</u>. There are no Actions pending or, to the Stockholder's knowledge, currently threatened against the Stockholder or its Affiliates, which would reasonably be expected to adversely affect the ability of the Stockholder to consummate the transactions contemplated by this Agreement, the Merger Agreement.
- 3.5 Ownership of Shares or Options. The Stockholder is the sole record and beneficial owner of the number and type of shares of Company Capital Stock or holds the number of Series C Warrants or Company Options, as the case may be, set forth on the Stockholder's or signature page hereto. Such shares of Company Capital Stock or Company Options set forth on the Stockholder's signature page hereto constitute all of the shares of Company Capital Stock owned of record or beneficially by the Stockholder or with respect to which the Stockholder has the power to vote or direct the vote or all of the Series C Warrants or Company Options held by the Stockholder, as the case may be. The Stockholder has full power and authority to sell, transfer, assign and deliver such shares of Company Capital Stock to Parent as provided in the Merger Agreement. The Stockholder has, and will have as of the time the Stockholder is required to perform its obligations hereunder, voting power and power to agree to all of the matters set forth in this Agreement, with no restrictions on the Stockholder's voting power pertaining thereto. In the case of a Stockholder, each of the shares of Company Capital Stock owned by the Stockholder is owned free and clear of, and none of such shares of Company Capital Stock is subject to, any Encumbrances other than Restated Voting Agreement, Amended and Restated Investor Rights Agreement, Amended and Restated Right of First Refusal and Co-Sale, each by and among the Company and certain of its stockholders.
- 3.6 No Reliance. The Stockholder acknowledges that it has conducted such investigation, if any, with respect to the Merger, the adequacy of the Merger Consideration and the terms of this Agreement and the Merger Agreement as the Stockholder believes appropriate or desirable.

3.7 No Brokers. Neither the Stockholder nor any of its Representatives or Affiliates has employed or made any agreement with any broker, finder or similar agent or any Person which will result in the obligation of the Stockholder, the Company, Parent, or any of their respective Affiliates to pay any finder's fee, brokerage fees or commission or similar payment in connection with the transactions contemplated hereby.

ARTICLE IV COVENANTS OF THE STOCKHOLDER

The Stockholder hereby covenants and agrees as follows:

- 4.1 Proprietary Information. The Stockholder hereby agrees that (i) he, she or it will hold, and will direct its Affiliates, directors, officers, employees, agents, representatives and advisors (including attorneys, accountants, consultants, bankers and financial advisors) to hold, in confidence all data and information in its possession relating to the businesses and operations of the Company that is confidential or proprietary in nature (the "Proprietary Information") and (ii) will not disclose such Proprietary Information to any Person (other than to the Stockholder's Affiliates, directors, officers, employees, agents, representatives and advisors (including attorneys, accountants, consultants, bankers and financial advisors), but in such event only on a need-to-know basis); provided, however, that this Section 4.1(b) shall be inoperative as to such portions of the Proprietary Information that (A) become generally available to the public other than as a result of disclosure which, to the knowledge of the Stockholder, is in violation of a confidentiality obligation, including information disclosed by Parent via an official statement, securities filing or press release, (B) are obtained from a third party which, to the knowledge of the Stockholder is not bound by a confidentiality obligation, (C) the Stockholder is required by applicable law, rule or regulation, including stock exchange rules and regulations, to divulge or disclose such information, in which case the Stockholder shall use commercially reasonable efforts to cooperate with the Company to limit such disclosure to the extent permitted under applicable law and to the extent practicable, cooperate with Parent's efforts to seek to obtain a protective order over, or confidential treatment of such information or (D) is disclosed by the Stockholder with the written consent of Parent and, prior to the Effective Time, the Company.
- 4.2 Public Announcements. The Stockholder shall consult with and obtain the written approval of Parent and, prior to the Effective Time, the Company before issuing any press release or making any other public disclosure with respect to this Agreement, the Merger Agreement or the transactions contemplated hereby or thereby and shall not issue any such press release or make any such public disclosure prior to such consultation and approval; provided that if the Stockholder is a venture capital fund, the Stockholder may disclose to its existing and prospective limited partners or members, subject to a duty of confidentiality subject to a duty of confidentiality as set forth in the governing documentation of the Stockholder, only (a) the existence of the Merger and the Merger Agreement, (b) the return on the Stockholder's investment in the Company and, (c) solely with respect to the Stockholder's existing limited partners or members, such other information as is required to be disclosed by the Stockholder to such limited partners or members pursuant to the applicable partnership or limited liability company agreement or otherwise permitted to be disclosed in the good faith exercise of the Stockholder's fiduciary duties as the general partner or manager of such venture capital fund.
- **4.3** <u>Additional Shares</u>. The Stockholder hereby agrees, until the occurrence of a Termination Event, to promptly notify Parent of the number of any shares of Company Capital Stock acquired by the Stockholder after the date hereof (including by way of the exercise of Series C Warrants or Company Options). The Stockholder shall vote in accordance with this Agreement any shares of Company Capital Stock that the Stockholder acquires after the date hereof.
- **4.4 <u>Further Assurances</u>**. From time to time until the occurrence of a Termination Event, at Parent's request, the Stockholder shall execute and deliver such additional documents and instruments and take all such further action as may be reasonably necessary to consummate and make effective, in the most expeditious

manner possible, the transactions contemplated by this Agreement and the Merger Agreement, including granting any waiver required under any Contract to which the Stockholder is a party which is inconsistent with any obligations of the Stockholder hereunder.

4.5 No Solicitation of Other Proposals. The Stockholder acknowledges and agrees that it has read and understands Section 5.3 of the Merger Agreement and hereby agrees that from the date hereof until the occurrence of a Termination Event, it shall be bound by the provisions of Section 5.3 of the Merger Agreement to the same extent as if the Stockholder were a party to the Merger Agreement.

ARTICLE V TRANSFER RESTRICTIONS

- **5.1** Restrictions on Transfer. Prior to the occurrence of a Termination Event, in addition to (and not in limitation of) any existing transfer restrictions applicable to the Company Capital Stock, the Stockholder shall not, directly or indirectly, sell, transfer, hedge, pledge, encumber, assign or otherwise dispose of (including by gift, bequest, appointment or otherwise) (collectively, "Transfer"), or consent to or permit any Transfer of, the Stockholder's shares of Company Capital Stock or any interest therein. If any ownership (whether legal or beneficial) of any of the Stockholder's shares of Company Capital Stock shall pass to any Person (including the Stockholder's administrators, successors or receivers), whether by operation of law or otherwise, then such shares of Company Capital Stock shall continue to be subject to this Agreement and the obligations of the Stockholder hereunder.
- **5.2** Stop Transfer; Legend. The Stockholder agrees that it will not request that the Company register the Transfer (by book-entry or otherwise) of any certificate or uncertificated interest representing any of the Stockholder's shares of Company Capital Stock.
- 5.3 Lock-Up. All shares of Parent Common Stock to be issued in connection with the Merger shall be subject to restrictions on the transfer of such shares during the period specified herein (the "Lock-Up Period"). During the Lock-Up Period, the Stockholder agrees, that the Stockholder will not offer, sell, contract to sell, pledge, grant any option to purchase, make any short sale or otherwise dispose of any shares of Parent Common Stock, or any options or warrants to purchase any shares of Parent Common Stock, or any securities convertible into, exchangeable for or that represent the right to receive shares of Parent Common Stock, whether now owned or hereinafter acquired, owned directly by the Stockholder (including holding as a custodian) or with respect to which the Stockholder has beneficial ownership within the rules and regulations of the SEC (collectively the "Subject Shares"). The foregoing restriction is expressly agreed to preclude the Stockholder from engaging in any hedging or other transaction which is designed to or which reasonably could be expected to lead to or result in a sale or disposition of the Stockholder's Subject Shares even if such Subject Shares would be disposed of by someone other than the Stockholder. Such prohibited hedging or other transactions would include without limitation any short sale or any purchase, sale or grant of any right (including without limitation any put or call option) with respect to any of the Stockholder's Subject Shares or with respect to any security that includes, relates to, or derives any significant part of its value from such Subject Shares. The Lock-Up Period will commence on the Effective Date (the "Commencement Date") and expire (i) 180 days following the Commencement Date with respect to 33 and 1/3%% of the Stockholder's Subject Shares, (ii) 270 days following the Commencement Date with respect to 33 and 1/3%% of the Stockholder's Subject Shares and (iii) 365 days following the Commencement Date with respect to 33 and 1/3%% of the Stockholder's Subject Shares. Notwithstanding the foregoing, the Stockholder may transfer the Stockholder's Subject Shares (i) as a bona fide gift or gifts, provided that the donee or donees thereof agree to be bound in writing by the restrictions set forth herein, (ii) to any trust for the direct or indirect benefit of the Stockholder or the immediate family of the Stockholder, provided that the trustee of the trust agrees to be bound in writing by the restrictions set forth herein, and provided further that any such transfer shall not involve a disposition for value or (iii) to Parent in connection with the settlement of indemnification claims pursuant to Article VII. For purposes of this Agreement, "immediate family" shall mean any relationship by blood, marriage or adoption, not more remote

than first cousin. In addition, notwithstanding the foregoing, if a Stockholder is a corporation, limited liability company, partnership or trust, the corporation, limited liability company, partnership or trust it may transfer the Subject Share to any wholly-owned subsidiary of such corporation, limited liability company or partnership or to the stockholders, partners, members or beneficiaries of the Stockholder, provided, however, that in any such case, it shall be a condition to the transfer that (i) the transferee execute an agreement stating that the transferee is receiving and holding such Subject Shares subject to the provisions of this Section 5.3 and there shall be no further transfer of such Subject Shares except in accordance with this Section 5.3 (and, for the avoidance of doubt, the provisions of this Section 5.3, including the exception to transfer restrictions shall apply to any such transferee as if the transferee were the original recipient of the Subject Shares hereto), (ii) that any such transfer shall not involve a disposition for value, and (iii) that no public announcement or filing under the Exchange Act shall be required of or voluntarily made by or on behalf of the Stockholder or Parent in connection with any such transfer. In addition, notwithstanding anything herein to the contrary, nothing herein shall prevent the Stockholder from establishing a written plan meeting the requirements of Rule 10b5-1 under the Exchange Act (a "10b5-1 plan"), relating to the sale of the Stockholder's Subject Shares, provided that (i) such shares may not be sold until after 180 days following the Commencement Date, (ii) the aggregate number of shares sold by the Stockholder under a 10b5-1 plan or otherwise shall not exceed the limitations set forth in this Section 5.3 and (iii) the establishment of such plan will not result in any public filing or other public announcement of the Stockholder during the 180 days following the Commencement Date.

ARTICLE VI WAIVERS AND RELEASE

6.1 No Dissenters' Rights. the Stockholder hereby waives and agrees not to exercise any dissenters' or appraisal rights pursuant to Section 262 of the DGCL or any other right to object to the Merger, the Merger Agreement or the transactions contemplated thereby.

6.2 Release. In consideration of Parent's covenants and agreements contained herein and in the Merger Agreement, effective as of the Effective Time, the Stockholder (on its own behalf and on behalf of its heirs, executors, administrators, agents, successors and permitted assigns) hereby releases, waives and discharges (i) Parent, Merger Sub, the Company and their respective Affiliates, Subsidiaries, officers, directors, stockholders, partners, members, agents, successors and permitted assigns (collectively, the "Parent Released Persons") and (ii) the Company and its Affiliates, Subsidiaries, officers, directors, stockholders, partners, members, agents, successors and permitted assigns (collectively, the "Company Released Persons") from any and all actions, causes of action, suits, debts, dues, sums of money, accounts, bonds, bills, covenants, controversies, agreements, promises, variances, trespasses, losses, judgments, executions, claims and demands whatsoever (including those sounding in contract or tort, loss of profits, interference with business contracts, interference with contractual relations, damage to business reputation, increased cost of doing business, interference with economic or business relationship or any prospect thereof, interference with expectancy of business advantage, in each case whether current or prospective), obligations, Contracts, covenants, fees, costs and Damages of any kind whatsoever (whether direct, indirect, consequential, incidental or otherwise, including legal fees incurred in connection herewith or in connection with any costs associated with appearing as a third party witness, with the enforcement of this Agreement and with the posting of any bond in connection with any appeal process), known or unknown, in its own right or derivatively, in law or equity (collectively, the "Claims"), that (1) with respect to the Company Released Persons, in any way arises from or out of, is based upon, or relates to any omissions, acts, events or facts that have occurred or occur at any time prior to or as of the Effective Time and (2) with respect to the Parent Released Persons, in any way arises from or out of, is based upon, or relates to any omissions, acts, events or facts that have occurred or occur at any time prior to or as of the Effective Time solely with respect to the Company's conduct of its business, including without limitation in each case (a) the preparation, negotiation, execution or consummation of the Merger Agreement, the Ancillary Agreements or any other agreement, document, certificate or instrument delivered in connection with the transactions contemplated by the Merger Agreement, and (b) Claims in respect of a breach by the Company's

board of directors or its individual directors and officers of their obligations, including in connection with the negotiation and execution of the Merger Agreement and the consummation of the transactions contemplated thereby; provided, that the foregoing release of Parent Release Persons and Company Release Persons shall not extend to (i) any Claims resulting from or arising under the Merger Agreement, this Agreement or any agreement, document or instrument contemplated by or entered into in connection with the Merger Agreement to which the Stockholder is a party, except to the extent such Claims seek consideration for any Company Capital Stock in excess of that contemplated by the Merger Agreement and only with respect to such excess consideration, (ii) any Claims in connection with the Stockholder's status as a current or former employee or consultant of the Company for (A) reimbursement of expenses or (B) compensation or benefits payable or due to the Stockholder, other than compensation under or relating to any stock option or other compensatory equity or equity-linked compensation plan or award, (iii) Claims arising under the Company's organizational documents, any indemnification agreement with the Company or any directors' and officers' liability insurance policy of the Company for or right to indemnity or advancement of expenses or (iv) any rights of the Stockholder necessary to enable the Stockholder Representative or the Stockholder to defend any indemnification claim asserted pursuant to the Merger Agreement. This Section 6.2 is for the benefit of the Parent Released Persons and Company Released Persons and shall be enforceable by any of them directly against the Stockholder. With respect to such Claims, the Stockholder hereby expressly waives any and all rights conferred upon him, her or it by any statute or rule of law which provides that a release does not extend to claims which the claimant does not know or suspect to exist in his or her favor at the time of executing the release, which if known by him, her or it must have materially affected his, her or its settlement with the released party, including, without limitation, the following provisions of California Civil Code Section 1542 or similar provisions of other applicable Law: "A GENERAL RELEASE DOES NOT EXTEND TO CLAIMS WHICH THE CREDITOR DOES NOT KNOW OR SUSPECT TO EXIST IN HIS OR HER FAVOR AT THE TIME OF EXECUTING THE RELEASE, WHICH IF KNOWN BY HIM OR HER MUST HAVE MATERIALLY AFFECTED HIS OR HER SETTLEMENT WITH THE DEBTOR."

ARTICLE VII AGREEMENT TO BE BOUND BY CERTAIN PROVISIONS OF THE MERGER AGREEMENT

- 7.1 <u>Certain Provisions Applicable to the Stockholder</u>. the Stockholder acknowledges, approves of and agrees to be bound by the provisions of Article I (The Merger), Section 5.7 (Tax Matters), and Article VII (Indemnification; Remedies) of the Merger Agreement, subject to the limitations set forth therein. This <u>Section 7.1</u> is for the benefit of the Company, the Surviving Corporation, Parent and any other Parent Indemnitee and shall be directly enforceable by them.
- **7.2** Stockholder Representative. The Stockholder acknowledges and accepts the appointment of the Stockholder Representative and the other provisions relating thereto as set forth in Section 7.2(g) of the Merger Agreement and agrees to be bound by Section 7.2(g) of the Merger Agreement. This Section 7.2 is for the benefit of the Stockholder Representative, the Company, the Surviving Corporation, Parent and any other Parent Indemnitee and shall be directly enforceable by them.

ARTICLE VIII MISCELLANEOUS

8.1 <u>Termination</u>. This Agreement shall terminate and be of no further force and effect upon the termination of the Merger Agreement in accordance with its terms; <u>provided</u>, <u>however</u>, that (i) this <u>Article VIII</u> shall survive such termination and remain in full force and effect in accordance with their terms, and (ii) no such termination will relieve any party from liability for breach of this Agreement by such party that occurred prior to such termination.

- **8.2** Notices. All notices, requests, demands, Claims and other communications which are required or may be given under this Agreement shall be in writing and shall be deemed to have been duly given when received if personally delivered; when transmitted if transmitted by electronic mail or confirmed facsimile with a copy sent by another means specified herein; the Business Day after it is sent if sent for next day delivery to a domestic address by recognized overnight delivery service (e.g., Federal Express); and five (5) Business Days after the date mailed by certified or registered mail, postage prepaid, if sent by certified or registered mail, return receipt requested, as follows:
 - (a) If to Parent, then as provided for in Section 9.2 of the Merger Agreement;
 - (b) If to the Stockholder, then to the address set forth on the Stockholder's signature page hereto;

or to such other address as the party to whom notice is to be given may have furnished to the other party in writing in accordance with this Section 8.2.

- **8.3** Entire Agreement. This Agreement, the Merger Agreement (including all Exhibits hereto and thereto), and the other agreements, documents and written understandings referred to herein or in the Merger Agreement or otherwise entered into or delivered by the parties hereto on the date of this Agreement (including without limitation the Ancillary Agreements), constitute the entire agreement and understanding and supersede all other prior covenants, agreements (including any letters of intent between the parties), undertakings, obligations, promises, arrangements, communications, representations and warranties, whether oral or written, by any party hereto or by any director, officer, employee, agent, Affiliate or Representative of any party hereto.
- **8.4** <u>Assignment</u>. Neither this Agreement nor any of the rights or obligations hereunder may be assigned without the prior written consent of each of the parties hereto; <u>provided</u>, <u>however</u>, that Parent may assign this Agreement or all or any of its rights or obligations hereunder to one or more of its Affiliates without the consent of any other party. No assignment shall release the assignor from any of its obligations hereunder.
- **8.5** <u>Amendment</u>. This Agreement may be amended with the written consent of (i) Parent and (ii) the Stockholder at any time by execution of an instrument in writing signed on behalf of the party(ies) against whom enforcement is sought (it being understood that any such amendment shall only be enforced against the Stockholder that has consented to such amendment).

8.6 Waiver.

- (a) Any party hereto may extend the time for the performance of any of the obligations or other acts required hereunder, waive any inaccuracies in the representations and warranties contained herein or in any document delivered pursuant hereto and waive compliance with any of the agreements or conditions contained herein. Any such extension or waiver shall be valid only if set forth in an instrument in writing signed by the party or parties to be bound thereby.
- (b) Except where a specific period for action or inaction is provided herein, neither the failure nor any delay on the part of any party in exercising any right, power or privilege under this Agreement or the documents referred to in this Agreement shall operate as a waiver thereof, nor shall any waiver on the part of any party of any such right, power or privilege, nor any single or partial exercise of any such right, power or privilege, preclude any other or further exercise thereof or the exercise of any other such right, power or privilege. The failure of a party to exercise any right conferred herein within the time required shall cause such right to terminate with respect to the transaction or circumstances giving rise to such right, but not to any such right arising as a result of any other transactions or circumstances.
- **8.7** <u>Severability</u>. If any term or other provision of this Agreement is invalid, illegal or incapable of being enforced as a result of any rule of law or public policy, all other terms and other provisions of this Agreement shall nevertheless remain in full force and effect so long as the economic or legal substance of the

transactions contemplated by this Agreement is not affected in any manner materially adverse to any party. Upon such determination that any term or other provision is invalid, illegal or incapable of being enforced, the parties hereto shall negotiate in good faith to modify this Agreement so as to effect the original intent of the parties as closely as possible in an acceptable manner to the end that the transactions contemplated by this Agreement are fulfilled to the greatest extent possible.

8.8 <u>Burden and Benefit</u>. This Agreement shall be binding upon and shall inure to the benefit of, the parties hereto and their respective successors and permitted assigns. This Agreement and all of its conditions and provisions are for the sole and exclusive benefit of the parties hereto and their respective successors and permitted assigns, and nothing in this Agreement, express or implied, is intended to confer upon any Person other than the parties hereto any rights or remedies of any nature whatsoever under or by reason of this Agreement or any provision hereof.

8.9 Survival of Representations, Etc. All of the representations and warranties made by the Stockholder in this Agreement shall survive the execution of this Agreement and the Closing for a period of 12 months following the Closing, except that the representations and warranties set forth in Sections 3.1 (Organization), 3.2 (Authorization), 3.5 (Ownership of Shares or Options) and 3.7 (No Brokers) hereof shall survive for a period of five (5) years following the Closing Date. The covenants and agreements of the Stockholders in this Agreement shall survive the execution of this Agreement and the Closing, except as expressly limited by their terms and except that the covenant in Section 4.1, shall survive the Closing for a period of 24 months following the Closing. Notwithstanding anything to the contrary herein, the Stockholders' aggregate liability for Claims of indemnity under or related to the Merger Agreement or any attachment, Exhibit, Schedule, the Disclosure Schedules, certificate, document or list delivered by the Company, its Representatives or the Stockholders pursuant to the Merger Agreement to the Parent Indemnitees (the foregoing capitalized terms as defined in the Merger Agreement), the Letter of Transmittal and hereunder shall not exceed the total consideration (including any portion of the Escrow Amount and valuing Parent Common Stock at the Price Per Share) actually paid to the Stockholder pursuant to the Merger Agreement, except in the case of fraud (which shall include the element of scienter and in which case recourse shall be sought only against the Stockholder who committed such fraud).

8.10 Governing Law This Agreement (and any claim or controversy arising out of or relating to this Agreement) shall be governed by the law of the State of Delaware without regard to conflict of law principles that would result in the application of any law other than the law of the State of Delaware.

8.11 Consent to Jurisdiction Each party to this Agreement hereby irrevocably and unconditionally submits, for itself and its property, to the exclusive jurisdiction of the Chancery Court of the State of Delaware and any state appellate court therefrom within the State of Delaware (or, if the Chancery Court of the State of Delaware declines to accept jurisdiction over a particular matter, any state or federal court within the State of Delaware), in any action or proceeding arising out of or relating to this Agreement or the agreements delivered in connection herewith or the transactions contemplated hereby or thereby or for recognition or enforcement of any judgment relating thereto, and each of the parties hereby irrevocably and unconditionally (a) agrees not to commence any such action or proceeding except in such courts, (b) agrees that any claim in respect of any such action or proceeding may be heard and determined in such courts, (c) waives, to the fullest extent it may legally and effectively do so, any objection which it may now or hereafter have to the laying of venue of any such action or proceeding in any such courts, and (d) waives, to the fullest extent permitted by law, the defense of an inconvenient forum to the maintenance of such action or proceeding in any such courts. Each of the parties hereto agrees that a final judgment in any such action or proceeding shall be conclusive and may be enforced in other jurisdictions by suit on the judgment or in any other manner provided by law. Each party to this Agreement irrevocably consents to service of process in the manner provided for notices in Section 8.2 hereof. Nothing in this Agreement will affect the right of any party to this Agreement to serve process in any other manner permitted by law.

- 8.12 Waiver of Trial by Jury. EACH PARTY TO THIS AGREEMENT ACKNOWLEDGES AND AGREES THAT ANY CONTROVERSY WHICH MAY ARISE UNDER THIS AGREEMENT IS LIKELY TO INVOLVE COMPLICATED AND DIFFICULT ISSUES, AND THEREFORE IT HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVES ANY RIGHT IT MAY HAVE TO A TRIAL BY JURY IN RESPECT OF ANY LITIGATION DIRECTLY OR INDIRECTLY ARISING OUT OF OR RELATING TO THIS AGREEMENT AND ANY OF THE AGREEMENTS DELIVERED IN CONNECTION HEREWITH OR THE TRANSACTIONS CONTEMPLATED HEREBY OR THEREBY. EACH PARTY CERTIFIES AND ACKNOWLEDGES THAT (A) NO REPRESENTATIVE, AGENT OR ATTORNEY OF ANY OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE EITHER OF SUCH WAIVERS, (B) IT UNDERSTANDS AND HAS CONSIDERED THE IMPLICATIONS OF SUCH WAIVERS, (C) IT MAKES SUCH WAIVERS VOLUNTARILY, AND (D) IT HAS BEEN INDUCED TO ENTER INTO THIS AGREEMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS SECTION 8.12.
- **8.13** Specific Performance. Each party hereto acknowledges and agrees that the other party would be damaged irreparably, and in a manner for which monetary damages would not be an adequate remedy, in the event any of the provisions of this Agreement are not performed in accordance with its specific terms or otherwise are breached. Accordingly, each party agrees that the other party shall be entitled to seek an injunction or injunctions to prevent breaches of the provisions of this Agreement and to enforce specifically this Agreement and the terms and provisions hereof in any action instituted in any court of the United States or any state thereof having jurisdiction over the parties and the matter, in addition to any other remedy to which such party may be entitled, at law or in equity.
- **8.14** <u>Cumulative Remedies</u>. All rights and remedies of either party hereto are cumulative of each other and of every other right or remedy such party may otherwise have at law or in equity, and the exercise of one or more rights or remedies shall not prejudice or impair the concurrent or subsequent exercise of other rights or remedies.
- **8.15** Expenses and Legal Fees. All costs and expenses incurred in connection with this Agreement and the transactions contemplated hereby shall be paid by the party incurring such expenses. If any party to this Agreement brings an action to enforce its rights under this Agreement, the prevailing party shall be entitled to recover its costs and expenses, including without limitation reasonable legal fees, incurred in connection with such action, including any appeal of such action.
- **8.16** Additional Parties; Counterparts. Notwithstanding Section 8.5, no consent shall be necessary to add additional Stockholders as signatories to this Agreement after the date hereof. This Agreement may be executed and delivered by facsimile or electronic signature and in two (2) or more counterparts, each of which shall be deemed an original, but all of which together shall constitute one (1) and the same instrument.
- **8.17** Stockholder Capacity. The Stockholder is entering into this Agreement solely in the Stockholder's capacity as the record holder or beneficial owner of Shares and nothing herein shall limit or affect any actions taken (or any failures to act) by the Stockholder in the Stockholder's capacity as a director or officer of the Company (if applicable). The taking of any actions (or any failures to act) by the Stockholder in the Stockholder's capacity as a director or officer of the Company (if applicable) shall not be deemed to constitute a breach of this Agreement, regardless of the circumstances related thereto.

[Remainder of Page Intentionally Left Blank.]

Appendix C - Opinion of Needham & Company, LLC



Needham & Company, LLC 445 Park Avenue, New York, NY 10022-4406 (212) 371-8300

November 12, 2014

Board of Directors Mandalay Digital Group, Inc. 2811 Cahuenga Boulevard West Los Angeles, CA 90068

Gentlemen:

We understand that Mandalay Digital Group, Inc. (the "Company), DTM Merger Sub, Inc., a wholly-owned subsidiary of the Company ("Merger Sub"), Appia, Inc. ("Appia"), and Shareholder Representative Services LLC, as representative of the Appia equityholders, propose to enter into an Agreement and Plan of Merger (the "Merger Agreement") whereby, upon the terms and subject to the conditions set forth in the Merger Agreement, Merger Sub will be merged with and into Appia and Appia will become a wholly-owned subsidiary of the Company (the "Merger"). The terms and conditions of the Merger will be set forth more fully in the Merger Agreement.

Pursuant to the proposed Merger Agreement, we understand that, at the Effective Time (as defined in the Merger Agreement), the outstanding shares of capital stock of Appia, other than shares owned by Appia, the Company or Merger Sub and other than dissenting shares, together with the outstanding options and warrants to purchase capital stock of Appia, will be converted into the right to receive shares of common stock, par value \$0.0001 per share, of the Company ("Company Common Stock") with an aggregate value, based on a per share value of \$4.50 per share, equal to the sum of (i) \$100,000,000, (ii) the aggregate exercise price of all Company Options (as defined in the Merger Agreement) and (iii) the Adjustment Amount (as defined in the Merger Agreement, and which may be a negative number) (the "Consideration"), and with such shares of Company Common Stock to be allocated among the former equityholders of Appia as specified in the Merger Agreement.

You have asked us to advise you as to the fairness, from a financial point of view, to the Company of the Consideration to be paid by the Company in the Merger pursuant to the Merger Agreement.

For purposes of this opinion we have, among other things: (i) reviewed a draft of the Merger Agreement dated November 10, 2014; (ii) reviewed certain publicly available information concerning the Company and Appia and certain other financial and operating data of the Company and Appia furnished to us by the Company; (iii) reviewed the historical stock prices and trading volumes of the Company Common Stock; (iv) held discussions with members of management of the Company concerning the current operations of and future business prospects for the Company and Appia and joint prospects for the combined companies, including the potential cost savings and other synergies that may be achieved by the combined companies; (v) reviewed certain financial forecasts with respect to the Company and Appia prepared by the respective managements of the Company and Appia and provided to us by the Company and held discussions with members of management of the Company concerning those forecasts; (vi) compared certain publicly available financial data of companies whose securities are traded in the public markets and that we deemed generally relevant to similar data for the Company and Appia; (vii) reviewed the financial terms of certain other business combinations that we deemed generally relevant; and (viii) reviewed such other financial studies and analyses and considered such other matters as we have deemed appropriate.

Boston Office: One Federal Street, Boston, MA 02110 (617) 457-0910

California Offices: 3000 Sand Hill Road, Building 1, Menlo Park, CA 94025 (650) 854-9111

100 Spear Street, San Francisco, CA 94105 (415) 262-4860

Chicago Office: 180 North LaSalle, Suite 3700, Chicago, IL 60601 (312) 981-0412

Board of Directors Mandalay Digital Group, Inc. November 12, 2014 Page 2

Needham & Company, LLC

In connection with our review and in arriving at our opinion, we have assumed and relied on the accuracy and completeness of all of the financial, accounting, legal, tax and other information discussed with or reviewed by us for purposes of this opinion and have neither attempted to verify independently nor assumed responsibility for verifying any of such information. We have assumed the accuracy in all material respects of the representations and warranties contained in the Merger Agreement and all agreements related thereto. In addition, we have assumed, with your consent, that the Merger will qualify as a reorganization within the meaning of Section 368(a) of the Internal Revenue Code of 1986 and will be consummated upon the terms and subject to the conditions set forth in the draft Merger Agreement dated November 10, 2014 without waiver, modification or amendment of any material term, condition or agreement thereof and that, in the course of obtaining the necessary regulatory or third party approvals, consents and releases for the Merger, no material delay, limitation, restriction or condition will be imposed that would have an adverse effect on the Company, Appia or the contemplated benefits of the Merger. We have also assumed, with your consent, that any adjustments pursuant to the terms of the Merger Agreement to the aggregate exercise price of all Company Options (as provided to us by the Company and used by us in our analyses) or the Adjustment Amount calculated as of September 30, 2014, will not be material to our analyses. With respect to the financial forecasts for the Company and Appia (including adjustments thereto expected by the Company to result from the Merger) prepared by the respective managements of the Company and Appia and provided to us by the Company, we have assumed, with your consent and based upon discussions with management of the Company, that such forecasts have been reasonably prepared on bases reflecting the best currently available estimates and judgments of such managements, at the time of preparation, of the future operating and financial performance of the Company, Appia and the combined companies. We have relied, without independent verification, upon the estimates of management of the Company of the potential cost savings and other synergies, including the amount and timing thereof, that may be achieved as a result of the proposed Merger. We express no opinion with respect to any of such forecasts (including such cost savings and other synergies) or the assumptions on which they were based.

We have not assumed any responsibility for or made or obtained any independent evaluation, appraisal or physical inspection of the assets or liabilities of the Company, Appia or any of their respective subsidiaries nor have we evaluated the solvency or fair value of the Company, Appia or any of their respective subsidiaries under any state or federal laws relating to bankruptcy, insolvency or similar matters. Further, our opinion is based on economic, monetary and market conditions as they exist and can be evaluated as of the date hereof and we assume no responsibility to update or revise our opinion based upon circumstances and events occurring after the date hereof. Our opinion as expressed herein is limited to the fairness, from a financial point of view, to the Company of the Consideration to be paid by the Company in the Merger pursuant to the Merger Agreement and we express no opinion as to the fairness of the Merger to the holders of any class of securities, creditors or other constituencies of the Company, or as to the Company's underlying business decision to engage in the Merger or the relative merits of the Merger as compared to other business strategies that might be available to the Company. We were not requested to, and did not, provide independent financial advice to you or the Company during the course of negotiations between the Company and Appia or participate in the negotiation or structuring of the Merger. In addition, we express no opinion with respect to the amount or nature or any other aspect of any compensation payable to or to be received by any officers, directors or employees of any party to the Merger, or any class of such persons, relative to the consideration to be paid by the Company pursuant to the Merger Agreement or with respect to the fairness of any such compensation. Our opinion does not constitute a recommendation to any stockholder of the Company as to how such stockholder should vote with respect to the Merger.

Board of Directors Mandalay Digital Group, Inc. November 12, 2014 Page 3

Needham & Company, LLC

We are not expressing any opinion as to the value of the Company Common Stock when issued pursuant to the Merger or the prices at which the Company Common Stock will actually trade at any time.

We have been engaged by the Company as financial advisor to render this opinion and will receive a fee for our services that is not contingent on the consummation of the Merger. In addition, the Company has agreed to indemnify us for certain liabilities arising out of our role as financial advisor and out of the rendering of this opinion and to reimburse us for certain of our out-of-pocket expenses. We have not in the past two years provided investment banking or financial advisory services to the Company or Appia for which we received compensation. We may in the future provide investment banking and financial advisory services to the Company, Appia and their respective affiliates unrelated to the proposed Merger, for which services we would expect to receive compensation. In the ordinary course of our business, we may actively trade the equity securities of the Company for our own account or for the accounts of customers or affiliates and, accordingly, may at any time hold a long or short position in such securities.

This letter and the opinion expressed herein are provided at the request and for the information of the Board of Directors of the Company and may not be disclosed publicly, quoted or referred to, or used for any other purpose without our prior written consent, except that this letter may be disclosed in connection with any registration statement or proxy statement used in connection with the Merger provided that this letter is quoted in full in such registration statement or proxy statement. This opinion has been approved by a fairness committee of Needham & Company, LLC.

Based upon and subject to the foregoing, it is our opinion that, as of the date hereof, the Consideration to be paid by the Company in the Merger pursuant to the Merger Agreement is fair to the Company from a financial point of view.

Very truly yours,

NEEDHAM & COMPANY, LLC

needham & Company, LLC

PART II

INFORMATION NOT REQUIRED IN PROSPECTUS

Item 20. Indemnification of Directors and Officers

Section 145 of the Delaware General Corporation Law, or the Delaware Law, provides that a corporation may indemnify directors and officers as well as other employees and individuals against expenses (including attorneys' fees), judgments, fines and amounts paid in settlement in connection with specified actions, suits or proceedings, whether civil, criminal, administrative or investigative (other than an action by or in the right of the corporation — a "derivative action"), if they acted in good faith and in a manner they reasonably believed to be in or not opposed to the best interests of the corporation, and, with respect to any criminal action or proceeding, had no reasonable cause to believe their conduct was unlawful. A similar standard is applicable in the case of derivative actions, except that indemnification only extends to expenses (including attorneys' fees) incurred in connection with defense or settlement of such action, and the statute requires court approval before there can be any indemnification where the person seeking indemnification has been found liable to the corporation. Under Section 145, a corporation shall indemnify an agent of the corporation for expenses actually and reasonably incurred if and to the extent such person was successful on the merits in a proceeding or in defense of any claim, issue or matter therein.

Section 145 of the Delaware Law provides that they are not exclusive of other indemnification that may be granted by a corporation's charter, bylaws, disinterested director vote, stockholders vote, agreement or otherwise. The limitation of liability contained in our certificate of incorporation, as amended, and the indemnification provision included in our bylaws, as amended, are consistent with Delaware Law Sections 102(b)(7) and 145. We have purchased directors and officers liability insurance.

Section 145 of the Delaware Law authorizes court to award, or a corporation's board of directors to grant, indemnity to directors and officers in terms sufficiently broad to permit such indemnification under certain circumstances for liabilities (including reimbursement for expenses incurred) arising under the Securities Act of 1933. Our certificate of incorporation, as amended, and our bylaws, as amended, provide for indemnification of our directors, officers, employees and other agents to the maximum extent permitted by the Delaware Law. In addition, we enter into indemnification agreements with our officers and directors in the ordinary course.

Item 21. Exhibits and Financial Statement Schedules

(a) Exhibits

See the Exhibit Index, which is incorporated by reference in this item.

(b) Financial Statement Schedules

Not applicable.

(c) Reports, Opinions or Appraisals.

Not applicable.

Item 22. Undertakings

The undersigned registrant hereby undertakes:

- (a) To file, during any period in which offers or sales are being made, a post-effective amendment to this Registration Statement:
 - (1) To include any prospectus required by Section 10(a)(3) of the Securities Act of 1933.
- To reflect in the prospectus any facts or events arising after the effective date of the registration statement (or the most recent post-effective amendment thereof) that, individually or in the aggregate, represent a fundamental change in the information set forth in the registration statement. Notwithstanding the foregoing, any increase or decrease in volume of securities offered (if the total dollar value of securities offered would not exceed that which was registered) and any deviation from the low or high end of the estimated maximum offering range may be reflected in the form of prospectus filed with the SEC pursuant to Rule 424(b) if, in the aggregate, the changes in volume and price represent no more than a 20 percent change in the maximum aggregate offering price set forth in the "Calculation of Registration Fee" table in the effective registration statement.
- (3) To include any material information with respect to the plan of distribution not previously disclosed in the registration statement or any material change to such information in the registration statement.
- (b) That, for the purpose of determining any liability under the Securities Act of 1933, each such post-effective amendment shall be deemed to be a new registration statement relating to the securities offered therein, and the offering of such securities at that time shall be deemed to be the initial bona fide offering thereof.
- (c) To remove from registration by means of a post-effective amendment any of the securities being registered that remain unsold at the termination of the offering.
- (d) For purposes of determining any liability under the Securities Act of 1933, each filing of the registrant's annual report pursuant to Section 13(a) or Section 15(d) of the Securities Exchange Act of 1934 (and, where applicable, each filing of an employee benefit plan's annual report pursuant to Section 15(d) of the Exchange Act) that is incorporated by reference in the registration statement shall be deemed to be a new registration statement relating to the securities offered therein, and the offering of such securities at that time shall be deemed to be the initial bona fide offering thereof.
- (e) That prior to any public reoffering of the securities registered hereunder through use of a prospectus that is a part of this Registration Statement, by any person or party who is deemed to be an underwriter within the meaning of Rule 145(c), the issuer undertakes that such reoffering prospectus will contain the information called for by the applicable registration form with respect to reofferings by persons who may be deemed underwriters, in addition to the information called for by the other items of the applicable form.
- (f) That every prospectus (1) that is filed pursuant to paragraph (e) immediately preceding, or (2) that purports to meet the requirements of Section 10(a)(3) of the Securities Act of 1933 and is used in connection with an offering of securities subject to Rule 415, will be filed as a part of an amendment to the registration statement and will not be used until such amendment is effective, and that, for purposes of determining any liability under the Securities Act of 1933, each such post-effective amendment shall be deemed to be a new registration statement relating to the securities offered therein, and the offering of such securities at that time shall be deemed to be the initial bona fide offering thereof.
- (g) Insofar as indemnification for liabilities arising under the Securities Act of 1933 may be permitted to directors, officers and controlling persons of the registrant pursuant to the provisions described in

Item 20 above, or otherwise, the registrant has been advised that in the opinion of the SEC such indemnification is against public policy as expressed in the Securities Act of 1933 and is, therefore, unenforceable. In the event that a claim for indemnification against such liabilities (other than the payment by the registrant of expenses incurred or paid by a director, officer or controlling person of the registrant in the successful defense of any action, suit or proceeding) is asserted by such director, officer or controlling person in connection with the securities being registered, the registrant will, unless in the opinion of its counsel the matter has been settled by controlling precedent, submit to a court of appropriate jurisdiction the question of whether such indemnification by it is against public policy as expressed in the Securities Act of 1933 and will be governed by the final adjudication of such issue.

- (h) To respond to requests for information that is incorporated by reference into the prospectus pursuant to Item 4, 10(b), 11, or 13 of this Form, within one business day of receipt of such request, and to send the incorporated documents by first-class mail or other equally prompt means. This includes information contained in documents filed subsequent to the effective date of the registration statement through the date of responding to the request.
- (i) To supply by means of a post-effective amendment all information concerning a transaction, and the company being acquired involved therein, that was not the subject of and included in the registration statement when it became effective.

SIGNATURES

Pursuant to the requirements of the Securities Act of 1933, as amended, the Registrant certifies that it has reasonable grounds to believe that the Registrant meets all of the requirements for filing on Form S-4 and has duly caused this registration statement to be signed on its behalf by the undersigned, thereunto duly authorized, in the City of Los Angeles, State of California, on January 16, 2015.

MANDALAY DIGITAL GROUP, INC.

By: /s/ William Stone

William Stone Chief Executive Officer

Pursuant to the requirements of the Securities Act of 1933, as amended, this Registration Statement has been signed by the following persons in the capacities and on the dates indicated.

/s/ William Stone	Chief Executive Officer	January 16, 2015
William Stone	(Principal Executive Officer)	·
/s/ Andrew Schleimer	Chief Financial Officer	January 16, 2015
Andrew Schleimer	(Principal Financial Officer)	
* Kirstie Brown	Principal Financial Officer	January 16, 2015
Kirstie Brown	(Principal Accounting Officer)	
*	Chairman of the Board	January 16, 2015
Peter Guber		
*	Director	January 16, 2015
Paul Schaeffer		
*	Director	January 16, 2015
Christopher Rogers		
*	Director	January 16, 2015
Rob Deutschman		
*	Director	January 16, 2015
Jeff Karish		
*	Director	January 16, 2015
Peter Adderton		
* By: /s/ William Stone		
William Stone		
Attorney-in-fact		

EXHIBIT INDEX

Exhibit No.	Description
2.1	Agreement and Plan of Merger, dated November 13, 2014, by and among Mandalay Digital Group, Inc., DTM Merger Sub, Inc., Appia, Inc. and Shareholder Representative Services LLC (included as Appendix A to the proxy statement/prospectus contained in this Registration Statement).
3.1	Certificate of Incorporation (incorporated by reference to the Registrant's Current Report on Form 8-K (File No. 000-10039), filed with the Commission on November 14, 2007).
3.2	Certificate of Amendment of Certificate of Incorporation, dated August 14, 2012 (incorporated by reference to Appendix B of the Registrant's Definitive Information Statement on Form 14-C (File No. 000-10039), filed with the Commission on July 10, 2012).
3.3	Certificate of Amendment of Certificate of Incorporation, dated March 28, 2013 (incorporated by reference to the Registrant's Current Report on Form 8-K (File No. 000-10039), filed with the Commission on April 18, 2013).
3.4	Certificate of Correction of Certificate of Amendment, dated April 9, 2013 (incorporated by reference to the Registrant's Current Report on Form 8-K (File No. 000-10039), filed with the Commission on April 18, 2013).
3.5	Bylaws (incorporated by reference to the Registrant's Current Report on Form 8-K (File No. 000-10039), filed with the Commission on November 14, 2007).
3.6	Certificate of Amendment of the Bylaws of NeuMedia, Inc., dated February 2, 2012 (incorporated by reference to our Current Report on Form 8-K (File No. 000-10039), filed with the Commission on February 7, 2012).
4.1	Form of Common Stock Purchase Warrant to be issued to North Atlantic SBIC IV, L.P.
5.1	Form of Opinion of Manatt, Phelps & Phillips, LLP.
8.1	Form of Opinion of Latham & Watkins, LLP regarding certain tax matters.
8.2	Form of Opinion of Goodwin Procter, LLP regarding certain tax matters.
10.1	Form of Securities Purchase Agreement between the Registrant and North Atlantic SBIC IV, L.P.
10.2	Form of Unconditional Secured Guaranty and Pledge Agreement between the Registrant and North Atlantic SBIC IV, L.P.
10.3	Form of Unconditional Secured Guaranty and Pledge Agreement between the Registrant and Silicon Valley Bank
10.4	Support Agreement dated November 13, 2014, among Mandalay Digital Group, Inc. and certain stockholders of Appia, Inc. listed therein (included as Appendix B to the proxy statement/prospectus contained in this Registration Statement)
23.1	Consent of SingerLewak LLP (Mandalay Digital).
23.2	Consent of Grant Thornton LLP (Appia).
23.3*	Consent of Manatt, Phelps & Phillips, LLP (included in Exhibit 5.1 hereto).
23.4*	Consent of Latham & Watkins, LLP (included in Exhibit 8.1 hereto).
23.5*	Consent of Goodwin Procter, LLP (included in Exhibit 8.2 hereto).
24.1†	Power of Attorney (included on the signature page to the Registration Statement).
99.1†	Consent of Needham & Company, LLC
99.2†	Consent of Judson Bowman to be named a director of the Registrant upon completion of the merger.

Exhibit No.	Description
99.3†	Consent of Craig Forman to be named a director of the Registrant upon completion of the merger.
99.4	Form of Proxy Card

^{*}To be filed by amendment. †Previously filed.

THIS WARRANT HAS NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AND HAS BEEN ACQUIRED FOR INVESTMENT AND NOT WITH A VIEW TO, OR IN CONNECTION WITH, THE SALE OR DISTRIBUTION HEREOF. NO SALE OR DISTRIBUTION OF THIS WARRANT MAY BE EFFECTED WITHOUT AN EFFECTIVE REGISTRATION STATEMENT RELATED THERETO OR AN OPINION OF COUNSEL IN A FORM SATISFACTORY TO THE ISSUER THAT SUCH REGISTRATION IS NOT REQUIRED UNDER THE SECURITIES ACT OF 1933.

MANDALAY DIGITAL GROUP, INC.

COMMON STOCK PURCHASE WARRANT

Date of Issuance:	[, 2	20]	ı
--------------------------	------	-----	---

FOR VALUE RECEIVED, MANDALAY DIGITAL GROUP, INC., a Delaware corporation ("<u>Mandalay</u>"), hereby certifies that NORTH ATLANTIC SBIC IV, L.P., a Delaware limited partnership (the "<u>Registered Holder</u>"), is entitled, subject to the terms set forth below, to purchase from Mandalay, at any time on or after the Vesting Date (as defined in <u>Section 1</u> below) and on or before the Expiration Date (as defined in <u>Section 6</u> below) shares of Mandalay's Common Stock, par value \$0.001 per share ("<u>Mandalay Common Stock</u>"), at an exercise price per share (the "<u>Exercise Price</u>") equal to \$0.001 (subject to adjustment as provided herein). The shares of stock issuable upon exercise of this Common Stock Purchase Warrant ("<u>Warrant</u>") are referred to hereinafter as the "<u>Warrant Shares</u>".

This Warrant is issued pursuant to, and is subject to the terms and conditions of, that certain Securities Purchase Agreement dated of even date herewith (as the same may be amended, modified, supplemented, extended or restated, from time to time, the "Purchase Agreement") by and among Mandalay, the Registered Holder, and Appia, Inc. ("Appia"). All capitalized terms used though not defined herein but defined in the Purchase Agreement shall have the meanings given to such terms in the Purchase Agreement.

1. <u>Vesting; Number of Shares.</u> This Warrant shall become fully vested on the date with is twelve (12) months from the date hereof (the "<u>Vesting Date</u>") if, and only if, all interest, principal and other amounts owed to the Registered Holder pursuant to that certain Debenture issued by Appia on or about the date hereof (the "<u>Obligations</u>") pursuant to the Purchase Agreement have not been paid in full on or before such Vesting Date. In the event the Obligations have been paid in full on or before the Vesting Date, then effective upon payment of the Obligations, this Warrant shall expire and be of no further force or effect. If this Warrant shall not have vested on the Vesting Date, then this Warrant shall automatically expire and be of no further force or effect. If and when this Warrant is vested, subject to the other terms and conditions hereinafter set forth, the Registered Holder shall then be entitled, upon surrender of this Warrant, to purchase from Mandalay up to Four Hundred Thousand (400,000) shares of Mandalay Common Stock (subject to adjustment as provided herein).

2. Exercise.

- (a) Manner of Exercise. This Warrant may be exercised by the Registered Holder, in whole or in part, by surrendering this Warrant, with the purchase/exercise form appended hereto as Exhibit A duly executed by such Registered Holder or by such Registered Holder's duly authorized attorney, at the principal office of Mandalay, or at such other office or agency as Mandalay may designate, accompanied by payment in full of the aggregate Exercise Price payable in respect of the number of Warrant Shares purchased upon such exercise (the "Purchase Price"). The Purchase Price may be paid by cash, check, wire transfer, or by the surrender of promissory notes or other instruments representing indebtedness of Mandalay to the Registered Holder.
- (b) Effective Time of Exercise. Each exercise of this Warrant shall be deemed to have been effected immediately prior to the close of business on the day on which this Warrant shall have been surrendered to Mandalay as provided in Section 2(a) hereinabove. At such time, the person or persons in whose name or names any certificates for Warrant Shares shall be issuable upon such exercise as provided in Section 2(d) hereinabove shall be deemed to have become the holder or holders of record of the Warrant Shares to be represented by such certificates.

(c) Net Issue Exercise.

(i) In lieu of exercising this Warrant in the manner provided in <u>Section 2(a)</u> hereinabove, the Registered Holder may elect to receive a number of Warrant Shares equal to the value of this Warrant (or the portion thereof being canceled) by surrender of this Warrant at the principal office of Mandalay together with notice of such election on the purchase/exercise form appended hereto as <u>Exhibit A</u> duly executed by such Registered Holder or such Registered Holder's duly authorized attorney, in which event Mandalay shall issue to such Registered Holder a number of Warrant Shares computed using the following formula:

$$X = \frac{Y (A - B)}{A}$$

Where X = The number of Warrant Shares to be issued to the Registered Holder.

- Y = The number of shares of Warrant Shares purchasable under this Warrant or, if only a portion of the Warrant is being exercised, the portion of the Warrant being cancelled (at the date of such calculation).
- A = The fair market value of one Warrant Share (at the date of such calculation).
- B = The Exercise Price (as adjusted to the date of such calculation).
 - (ii) For purposes of this Section 2(c), the fair market value of one Warrant Share on the date of calculation shall mean:

(A) as of any particular date: (1) the volume weighted average of the closing sales prices of Mandalay Common Stock for such day on all domestic securities exchanges on which Mandalay Common Stock may at the time be listed; (2) if there have been

no sales of Mandalay Common Stock on any such exchange on any such day, the average of the highest bid and lowest asked prices for Mandalay Common Stock on all such exchanges at the end of such day; (3) if on any such day Mandalay Common Stock is not listed on a domestic securities exchange, the closing sales price of Mandalay Common Stock as quoted on the OTC Bulletin Board, the Pink OTC Markets, or similar quotation system or association for such day; or (4) if there have been no sales of Mandalay Common Stock on the OTC Bulletin Board, the Pink OTC Markets, or similar quotation system or association on such day, the average of the highest bid and lowest asked prices for Mandalay Common Stock quoted on the OTC Bulletin Board, the Pink OTC Markets or similar quotation system or association at the end of such day; in each case, averaged over twenty (20) consecutive Business Days ending on the Business Day immediately prior to the day as of which fair market value is being determined; provided, that if Mandalay Common Stock is listed on any domestic securities exchange, the term "Business Day" as used in this sentence means Business Days on which such exchange is open for trading.

- (B) if item (A) hereinabove is not applicable, then the fair market value of a Warrant Share shall be the highest price per share which Mandalay could obtain on the date of calculation from a hypothetical willing, able, sophisticated and experienced buyer (but not a current employee or director), acting at arms length in an open and unrestricted market, when neither is under compulsion to buy or sell and when both have reasonable knowledge of all relevant facts, with such price to be determined in good faith by the Mandalay Board and the Registered Holder, unless Mandalay is at such time subject to a Liquidity Event (as described in Section 7(d) below) in which case the fair market value of a Warrant Share shall be deemed to be the value received by the holders of such stock pursuant to such event. If Mandalay, on the one hand, and the Registered Holder, on the other hand, cannot agree upon the fair market value of a Warrant Share under this paragraph, then the dispute shall be settled by the appointment of an independent third-party arbitrator in accordance with the rules of the American Arbitration Association.
- (d) <u>Delivery to Holder</u>. As soon as practicable after the exercise of this Warrant in whole or in part, and in any event within ten (10) days thereafter, Mandalay at its expense shall cause to be issued in the name of, and delivered to, the Registered Holder:
 - (i) a certificate or certificates for the number of Warrant Shares to which such Registered Holder shall be entitled, and
- (ii) in case such exercise is in part only, a new warrant or warrants (dated the date hereof) of like tenor and calling in the aggregate on the face or faces thereof for the number of Warrant Shares equal (without giving effect to any adjustment therein) to the number of such shares called for on the face of this Warrant minus the number of such shares purchased by the Registered Holder upon such exercise as provided in <u>Section 2(a) or 2(c)</u> hereinabove (without giving effect to any adjustment thereto).
 - (e) Payment of Dividends. If, between the issue and exercise dates of this Warrant:
- (i) Mandalay shall have declared and paid dividends on Mandalay Common Stock, then upon exercise of this Warrant, the Registered Holder shall be entitled to

receive payment from Mandalay of an amount equal to any dividend payment to which such Registered Holder would have been entitled if such Registered Holder had exercised this Warrant prior to the declaration and payment of said dividends; and

(ii) any dividends shall have accrued on Mandalay Common Stock, or shall have been declared on Mandalay Common Stock but be unpaid thereon, then, upon exercise of this Warrant, the Warrant Shares shall automatically and without further action on the part of Mandalay or the Registered Holder (unless the Registered Holder otherwise agrees) be deemed to have accrued an amount of dividends that would have accrued on such Warrant Shares if such Warrant Shares had been issued on the date hereof and be entitled to receive payment from Mandalay of an amount equal to any declared, but unpaid, dividend at the same time, and on the same terms and conditions, as the other holders of Mandalay Common Stock.

3. Adjustments.

- (a) Stock Splits and Dividends. If outstanding shares of Mandalay Common Stock shall be subdivided into a greater number of shares of Mandalay Common Stock or other Capital Stock of Mandalay, or a dividend of other or additional Capital Stock or other securities or property (other than cash) of Mandalay shall be paid in respect of such Mandalay Common Stock, then the Exercise Price in effect immediately prior to such subdivision or at the record date of such dividend shall simultaneously with the effectiveness of such subdivision or immediately after the record date of such dividend be proportionately reduced. If outstanding shares of Mandalay Common Stock shall be combined into a smaller number of shares, then the Exercise Price in effect immediately prior to such combination shall, simultaneously with the effectiveness of such combination, be proportionately increased. When any adjustment is required to be made in the Exercise Price, the number of Warrant Shares purchasable upon the exercise of this Warrant shall be changed to the number determined by dividing (i) an amount equal to the number of shares issuable upon the exercise of this Warrant immediately prior to such adjustment, multiplied by the Exercise Price in effect immediately after such adjustment. If an adjustment in the Exercise Price made hereunder would reduce the Exercise Price to an amount below par value of Mandalay Common Stock, then such adjustment in Exercise Price made hereunder shall reduce the Exercise Price to the par value of Mandalay Common Stock.
- (b) <u>Reclassification</u>, <u>Etc.</u> In case there occurs any reclassification or change of the outstanding securities of Mandalay (or any other corporation the stock or securities of which are at the time receivable upon the exercise of this Warrant) or any similar corporate reorganization on or after the date hereof, then and in each such case the Registered Holder, upon the exercise hereof at any time after the consummation of such reclassification, change or reorganization shall be entitled to receive, in lieu of the stock or other securities and property receivable upon the exercise hereof prior to such consummation, the stock or other securities or property to which such Registered Holder would have been entitled upon such consummation if such Registered Holder had exercised this Warrant immediately prior thereto, all subject to further adjustment pursuant to the provisions of this <u>Section 3</u>.
- (c) <u>Adjustment Certificate</u>. When any adjustment is required to be made to the Warrant Shares or the Exercise Price pursuant to this <u>Section 3</u>, Mandalay shall promptly

mail to the Registered Holder a certificate setting forth (i) a brief statement of the facts requiring such adjustment, (ii) the Exercise Price after such adjustment, and (iii) the kind and amount of stock or other securities or property into which this Warrant shall be exercisable after such adjustment.

(d) <u>Acknowledgement</u>. For the avoidance of doubt, it is acknowledged that the holder of this Warrant shall be entitled to the benefit of all adjustments pursuant to this <u>Section 3</u> which occur after the date hereof and prior to the exercise of this Warrant.

4. Assignment.

- (a) The holder of this Warrant acknowledges that this Warrant has not been registered under the Securities Act of 1933, as amended (the "Securities Act"), and agrees not to sell, pledge, distribute, offer for sale, transfer or otherwise dispose of this Warrant in the absence of (i) an effective registration statement under the Securities Act as to this Warrant and registration or qualification of this Warrant under any applicable U.S. federal or state securities law then in effect, or (ii) an opinion of counsel, reasonably satisfactory to Mandalay, that such registration and qualification are not required.
- (b) This Warrant is not assignable by the Registered Holder without Mandalay's prior written consent, except that assignments by the Registered Holder to an Affiliate (as defined below) of the Registered Holder shall be permitted provided that the assignee agrees in writing to be bound by all of the terms of this Warrant. For purposes of the foregoing, an "Affiliate" of the Registered Holder shall mean any other person or entity who directly or indirectly, controls, is controlled by, or is under common control with the Registered Holder, including, without limitation, any general partner, managing member, officer, or director of such Registered Holder, or any venture capital fund now or hereafter existing which is controlled by one or more general partners or managing members of, or shares the same management company with, the Registered Holder.
- **5.** <u>No Impairment</u>. Without the prior written consent of the Registered Holder, Mandalay shall not avoid or seek to avoid the observance or performance of any of the terms of this Warrant.

6. Termination.

(a) Provided that the Registered Holder receives prior notices in accordance with Section 7 below, and subject to
Section 6(b) and Section 6(c) hereinbelow, this Warrant (and the right to purchase securities upon exercise hereof) shall terminate on the
earliest to occur of the following (the "Expiration Date"):

(i) 5:00 P.M. Boston, Massachusetts time on [, 20]
--

- (ii) the occurrence of any Liquidity Event following the Vesting Date.
- (b) If on the Expiration Date this Warrant has not been exercised in full and the fair market value of one Warrant Share (or other security issuable upon the exercise hereof)

as determined in accordance with <u>Section 2(c)</u> above is greater than the Exercise Price in effect on such date, then this Warrant shall automatically be deemed on and as of such date to be exercised and converted pursuant to <u>Section 2(c)</u> into Warrant Shares, and Mandalay shall promptly deliver a certificate representing such Warrant Shares.

(c) If a Liquidity Event occurs prior to the Vesting Date, then (i) Mandalay shall reserve and set aside an amount of the consideration payable in respect of such Liquidity Event, which would otherwise be distributed to Mandalay's Stockholders, sufficient to pay to the Registered Holder the full amount of payment such holder would have been entitled to had this Warrant been fully vested at the time of the Liquidity Event (the "Liquidity Payment"), and (ii) if this Warrant subsequently vests, then this Warrant shall automatically be deemed on and as of the Vesting Date to be exercised and converted pursuant to Section 2(c) into Warrant Shares (subject to any adjustments resulting from such Liquidity Event or other events under Section 3), and Mandalay shall, at the option of the Registered Holder, promptly deliver the a certificate representing such Warrant Shares or the Liquidity Payment. In lieu of the foregoing, Mandalay may elect to accelerate vesting of this Warrant.

7. Notices of Certain Transactions. In case:

- (a) Mandalay shall set a record date for the holders of its Capital Stock (or other stock or securities at the time deliverable upon the exercise of this Warrant) for the purpose of entitling or enabling them to receive any dividend or other distribution, or to receive any right to subscribe for or purchase any shares of stock of any class or any other securities, or to receive any other right, to subscribe for or purchase any shares of stock of any class or any other securities, or to receive any other right, or
- (b) Mandalay shall set an effective date of any capital reorganization of Mandalay, any reclassification of the Capital Stock of Mandalay, any consolidation or merger of Mandalay, any consolidation or merger of Mandalay with or into another corporation (other than a consolidation or merger in which Mandalay is the surviving entity), or any transfer of all or substantially all of the assets of Mandalay, or
- (c) Mandalay shall set an effective date of any redemption of any Capital Stock of Mandalay (or other stock or securities at the time deliverable upon the exercise of this Warrant), or
 - (d) Mandalay shall set an effective date of any Liquidity Event,

then, and in each such case, Mandalay will mail or cause to be mailed to the Registered Holder of this Warrant a notice specifying, as the case may be, (i) the record date for the purpose of such dividend, distribution or right, and stating the amount and character of such dividend, distribution or right, or (ii) the effective date on which such Liquidity Event, reorganization, reclassification, consolidation, merger, transfer, dissolution, liquidation, winding-up, or redemption is to take place, and the time, if any is to be fixed, as of which the holders of record of Capital Stock (or such other stock or securities at the time deliverable upon such reorganization, reclassification, consolidation, merger, transfer, dissolution, liquidation, winding-up, redemption or conversion) are to be determined. Such notice shall be given at least ten (10) Business Days prior to the record date or effective date for the event specified in such notice.

Such notice shall specify all material terms of the event specified in such notice. If any material term changes after such notice is given, or such notice fails to include any such material term, such notice shall be void and Mandalay shall be obligated to provide a new notice to the Registered Holder pursuant to the terms of this Section 7.

8. Sale of Company. If following the Vesting Date, Mandalay shall at any time prior to the Expiration Date undergo a Liquidity Event, it shall provide the Registered Holder the ten (10) Business Days prior written notice of the Liquidity Event as required by Section 7 above, then the Registered Holder shall have until the effective date of such Liquidity Event (the "Sale Exercise Date") to exercise this Warrant in accordance with the provisions of Section 2 above.

9. Intentionally Deleted.

- 10. Replacement of Warrant. Upon receipt of evidence reasonably satisfactory to Mandalay of the loss, theft, destruction or mutilation of this Warrant and (in the case of loss, theft or destruction) upon delivery of an indemnity agreement (with surety if reasonably required) in a form reasonably satisfactory to Mandalay, or (in the case of mutilation) upon surrender and cancellation of this Warrant, Mandalay will issue, in lieu thereof, a new Warrant of like tenor.
- 11. <u>Mailing of Notices</u>. Any notice required or permitted pursuant to this Warrant shall be in writing and shall be deemed sufficient upon delivery pursuant to Section 11.5 of the Purchase Agreement.
- 12. No Rights as Stockholder. Except as may be otherwise provided in (i) Mandalay's Certificate of Incorporation, or (ii) the Purchase Agreement, until the exercise of this Warrant, the Registered Holder of this Warrant shall not solely by virtue of holding this Warrant have or exercise any rights as a Stockholder of Mandalay.
- 13. No Fractional Shares. No fractional shares of Mandalay Common Stock (or other stock or securities at the time deliverable upon the exercise of this Warrant) shall be issued in connection with any exercise hereunder. In lieu of any fractional shares which would otherwise be issuable, Mandalay shall pay cash equal to the product of such fraction multiplied by the fair market value of one share of Mandalay Common Stock (or other stock or securities at the time deliverable upon the exercise of this Warrant) on the date of exercise, as determined in good faith by the Mandalay Board.
- **14.** <u>Amendment or Waiver</u>. Any term of this Warrant may be amended or waived upon written consent of Mandalay and the Registered Holder.
- 15. <u>Headings.</u> The headings in this Warrant are for purposes of reference only and shall not limit or otherwise affect the meaning of any provision of this Warrant.

16. <u>Governing Law</u>. This Warrant shall be governed, construed and interpreted in accordance with the laws of the State of Delaware without giving effect to principles of conflicts of law.

[THE REMAINDER OF THIS PAGE IS LEFT BLANK INTENTIONALLY]

IN WITNESS WHEREOF, Mandalay Digital Group, Inc., through its duly authorized officer undersigned hereto, has executed this Common Stock Purchase Warrant under its seal as of the date first written above.

Mandalay:	
MANDALA	Y DIGITAL GROUP, INC.
By:	
Name:	
Title:	
Witnessed:	
By:	
Name:	
	MANDALA By: Name: Title: Witnessed: By:

[Signature Page to Common Stock Purchase Warrant]

$\underline{EXHIBIT\ A}$

PURCHASE/EXERCISE FORM

To: MANDALAY DIGITAL GROUP, INC.	Dated:, 20_
The undersigned Registered Holder, pursuant to the provisions set forth in the attached Warrant, hereby irre	evocably elects to (choose one):
(a) purchase shares of Mandalay Common Stock covered by such Warrant and herew representing the full purchase price for such shares at the price per share provided for in such Warrant, or	ith makes payment of \$,
Section 2(d) of such Warrant. shares purchasable under the Warrant pursuant to the No.	et Issue Exercise provisions of
Registered Holder (if an entity):	
Entity Name:	
By (signature):	
Signatory Name:	
Signatory Title:	
Registered Holder (if an individual):	
Signature:	
Signatory Name:	

FORM OF OPINION

Mandalay Digital Group, Inc. 2811 Cahuenga Boulevard West Los Angeles, California 90068

Re: Mandalay Digital Group, Inc.

Registration Statement on Form S-4 (File No. 333-200695)

Ladies and Gentlemen:

We have acted as counsel to Mandalay Digital Group, Inc., a Delaware corporation (the "Company"), in connection with the preparation of the Registration Statement on Form S-4 (File No. 333-200695) (the "Registration Statement") filed with the Securities and Exchange Commission pursuant to the Securities Act of 1933, as amended (the "Securities Act") in connection with the issuance by the Company of up to 19,500,000 shares of the Company's common stock, \$0.0001 par value per share (the "Shares"), pursuant to that certain Agreement and Plan of Merger, dated November 13, 2014, by and among the Company, DTM Merger Sub, Inc., a wholly-owned subsidiary of the Company ("Merger Sub"), Appia, Inc. ("Appia"), and Shareholder Representative Services LLC, as stockholder representative (the "Merger Agreement"), pursuant to which Merger Sub will merge with and into Appia so that Appia, as the surviving corporation, will become a wholly-owned subsidiary of the Company (the "Merger"). This opinion is delivered in accordance with the requirements of Item 601(b)(5) of Regulation S-K under the Securities Act.

In rendering this opinion, we have examined the Registration Statement, the Merger Agreement, the Certificate of Incorporation and bylaws, as amended, of the Company, and resolutions adopted by the Company's board of directors. We have also examined originals, or copies certified to our satisfaction, of such records, agreements, instruments and documents as we have deemed necessary as a basis for the opinion hereinafter expressed. As to various matters of fact relevant to the opinion hereinafter expressed, we have relied upon, and have assumed the accuracy of, certificates and oral or written statements and other information of or from public officials, officers and representatives of the Company.

In all such examinations we have assumed the genuineness of all signatures, the authenticity of all documents, certificates and records submitted to us as originals, the conformity to original documents, certificates and records of all documents, certificates and records submitted to us as copies, and the truthfulness of all statements of fact contained therein. We also have assumed, with respect to all parties to agreements or instruments relevant hereto, other than the Company, with respect to the Merger Agreement, that such parties had the requisite power and authority (corporate or other) to execute, deliver and perform such agreements or instruments, that such agreements or instruments have been duly authorized by all requisite action (corporate or otherwise), executed and delivered by such parties and that such agreements and instruments are the valid, binding and enforceable obligations of such parties. For purposes of this opinion, we have also assumed that, prior to the issuance of any Shares, (i) the Registration Statement, including all amendments, will have become effective under the Securities Act and (ii) the certificate of merger relating to the transactions contemplated by the Merger Agreement will have been filed with and accepted by the Delaware Secretary of State.

Based upon and subject to the foregoing, and subject to the limitations, assumptions and qualifications set forth herein, we are of the opinion that, the Shares have been duly authorized and, when the Shares are issued and paid for in accordance with the terms of the Merger Agreement, such Shares will be validly issued, fully paid and nonassessable.

The foregoing opinion is based on, and is limited to, the Delaware General Corporation Law as in effect on the date hereof, and we render no opinion with respect to the laws of any other jurisdiction.

We hereby consent to the filing of this opinion with the Securities and Exchange Commission as an exhibit to the Registration Statement and to the use of our name under the caption "Legal Matters" in the proxy statement/prospectus forming a part of the Registration Statement. In giving this consent, we do not admit that we are in the category of persons whose consent is required under Section 7 of the Securities Act or the rules and regulations thereunder.

Sincerely,

Manatt, Phelps & Phillips, LLP

LATHAM&WATKINS LLP

 $[\square]$

Mandalay Digital Group, Inc. 2811 Cahuenga Bouelvard West Los Angeles, California 90068

Re: Agreement and Plan of Merger, dated as November 13, 2014

Ladies and Gentlemen:

We have acted as special counsel to Mandalay Digital Group, Inc., a Delaware corporation ("Parent"), in connection with the proposed merger of DTM Merger Sub, Inc., a Delaware corporation ("Merger Sub") and a direct wholly-owned subsidiary of Parent, with and into Appia, Inc., a Delaware corporation ("Company"), with the Company surviving as a direct wholly-owned subsidiary of Parent (the "Merger") pursuant to the Agreement and Plan of Merger dated as of November 13, 2014, by and among Parent, Merger Sub, Company, and Shareholder Representative Services LLC (the "Merger Agreement"). This opinion is being delivered in connection with[, as of the declaration of effectiveness by the Securities and Exchange Commission of,] the Registration Statement on Form S-4 (File No. 333-[I]) (as amended through the date hereof, the "Registration Statement") of Parent, including the proxy statement/prospectus (the "Proxy Statement/Prospectus") forming a part thereof, relating to the transactions contemplated by the Merger Agreement. Capitalized terms not defined herein have the meanings specified in the Merger Agreement unless otherwise indicated.

In rendering our opinion, we have examined and, with your consent, are expressly relying upon (without any independent investigation or review thereof) the truth and accuracy of the factual statements, representations and warranties contained in (i) the Merger Agreement (including any Exhibits and Schedules thereto), (ii) the Registration Statement and the Proxy Statement/Prospectus, (iii) the respective tax representation letters of Parent and Merger Sub and of Company, each dated as of [I] and delivered to us for purposes of this opinion (the "Officer's Certificates"), and (iv) such other documents and corporate records as we have deemed necessary or appropriate for purposes of our opinion.

355 South Grand Avenue Los Angeles, California 90071-1560 Tel: +1.213.485.1234 Fax: +1.213.891.8763 www.lw.com

FIRM / AFFILIATE OFFICES Abu Dhabi Milan Barcelona Moscow Beijing Munich Boston New Jersey Brussels New York Century City Orange County Chicago Paris Doha Rivadh Dubai Rome

Düsseldorf San Diego Frankfurt San Francisco Hamburg Shanghai Silicon Valley Hong Kong Houston Singapore London Tokvo Los Angeles Washington, D.C.

Madrid

[[]] Page 2

LATHAM&WATKINS LLP

In addition, we have assumed, with your consent, that:

- 1. Original documents (including signatures) are authentic, and documents submitted to us as copies conform to the original documents, and there has been (or will be by the Effective Time of the Merger) execution and delivery of all documents where execution and delivery are prerequisites to the effectiveness thereof;
- 2. The Merger will be consummated in the manner contemplated by, and in accordance with the provisions of, the Merger Agreement, the Registration Statement and the Proxy Statement/Prospectus, and the Merger will be effective under the laws of the State of Delaware:
- 3. All factual statements, descriptions and representations contained in any of the documents referred to herein or otherwise made to us are true, complete and correct in all respects and will remain true, complete and correct in all respects up to and including the Effective Time, and no actions have been taken or will be taken which are inconsistent with such factual statements, descriptions or representations or which make any such factual statements, descriptions or representations untrue, incomplete or incorrect at the Effective Time;
- 4. Any statements made in any of the documents referred to herein "to the knowledge of" or similarly qualified are true, complete and correct in all respects and will continue to be true, complete and correct in all respects at all times up to and including the Effective Time, in each case without such qualification;
- 5. The parties have complied with and, if applicable, will continue to comply with, the covenants contained in the Merger Agreement, the Registration Statement and the Proxy Statement/Prospectus; and
- 6. There will be no change in applicable U.S. federal income tax law from the date hereof through the Effective Time.

Based upon and subject to the foregoing, and subject to the qualifications and limitations stated in the Registration Statement, the Proxy Statement/Prospectus and the Officer's Certificates, the statements of law and legal conclusions contained under the caption "Material U.S. Federal Income Tax Consequences," regarding the material U.S. federal income tax consequences of the Merger constitute the opinion of Latham & Watkins LLP.

In addition to the matters set forth above, this opinion is subject to the exceptions, limitations and qualifications set forth below.

1. This opinion represents our best judgment regarding the application of U.S. federal income tax laws arising under the Code, existing judicial decisions, administrative regulations and published rulings and procedures, but does not address all of the U.S. federal income tax consequences of the Merger. We express no opinion as to U.S. federal, state, local, foreign, or other tax

[0] Page 3

LATHAM&WATKINS LLP

consequences, other than as set forth herein. Our opinion is not binding upon the Internal Revenue Service or the courts, and there is no assurance that the Internal Revenue Service will not assert a contrary position. Furthermore, no assurance can be given that future legislative, judicial or administrative changes, on either a prospective or retroactive basis, would not adversely affect the accuracy of the conclusions stated herein. Nevertheless, we undertake no responsibility to advise you of any new developments in the application or interpretation of the U.S. federal income tax laws.

2. No opinion is expressed as to any transaction other than the Merger as described in the Merger Agreement, the Registration Statement and the Proxy Statement/Prospectus, or to any transaction whatsoever, including the Merger, if, to the extent relevant to our opinion, either all the transactions described in the Merger Agreement are not consummated in accordance with the terms of the Merger Agreement and without waiver or breach of any provisions thereof or all of the factual statements, representations, warranties and assumptions upon which we have relied are not true and accurate at all relevant times.

This opinion is rendered only to you, and is solely for your use in connection with the Parent's filing of the Registration Statement upon the understanding that we are not hereby assuming professional responsibility to any other person whatsoever. This opinion may not be relied upon by you for any other purpose, or furnished to, quoted to, or relied upon by any other person, firm or corporation for any purpose, without our prior written consent, except that this opinion may be furnished or quoted to your legal counsel and to judicial regulatory authorities having jurisdiction over you, and provided, however, that this opinion may be relied upon by persons entitled to rely on it pursuant to applicable provisions of federal securities law. We consent to the filing of this opinion as an exhibit to the Registration Statement and to the reference to our firm name therein under the captions "Material U.S. Federal Income Tax Consequences" and "Legal Matters." In giving this consent, we do not admit that we are within the category of persons whose consent is required under Section 7 of the Securities Act of 1933, as amended, or the rules or regulations of the Securities and Exchange Commission promulgated thereunder

Very truly yours,

DRAFT

[], 201[]

Appia, Inc. [Address] [Address]

Ladies and Gentlemen:

We have acted as counsel to Appia, Inc., a Delaware corporation (the "Company"), in connection with the proposed merger (the "Merger") of DTM Merger Sub, Inc., a Delaware corporation ("Merger Sub") and a wholly owned subsidiary of the Mandalay Digital Group, Inc. ("Parent") with and into Company with Company surviving the Merger as a wholly-owned subsidiary of Parent. The Merger will be consummated pursuant to the Agreement and Plan of Merger dated as of November 13, 2014 (the "Merger Agreement"). This opinion is being delivered in connection with[, as of the declaration of effectiveness by the Securities and Exchange Commission of,] the Registration Statement on Form S-4 (File No. 333-[]) (as amended through the date hereof, the "Registration Statement") of Parent, including the proxy statement/prospectus (the "Proxy Statement/Prospectus") forming a part thereof, relating to the transactions contemplated in the Merger Agreement. Capitalized terms not defined herein have the meanings specified in the Merger Agreement unless otherwise indicated.

For purposes of the opinion set forth below, we have reviewed and relied upon, without independent investigation thereof, the Merger Agreement, the Registration Statement, the Proxy Statement/Prospectus and such other documents, records and instruments as we have deemed necessary or appropriate as a basis for our opinion. In addition, in rendering our opinion we have relied upon certain statements, representations, warranties and covenants made by Parent and the Company in the Merger Agreement and in representation letters as of the date hereof and provided to us on the date hereof in connection with our preparation of this opinion, which statements, representations, warranties and covenants we have neither independently investigated nor verified. We have assumed that such statements, representations and warranties are, and always have been, true, correct and complete, that such statements, representations and warranties will be true, correct and complete as of the Effective Time of Merger, that no actions that are inconsistent with such statements, representations, warranties and covenants will be taken, and that all representations, statements and warranties qualified as to knowledge or belief or otherwise are and will be true, correct and complete as if made without such qualification. We also have assumed the genuineness of all signatures, the authenticity of all documents

submitted to us as originals, the conformity to the original documents of all documents submitted to us as copies, the authority and capacity of the individual or individuals who executed any such documents on behalf of any person, the conformity to the final documents of all documents submitted to us as drafts and the accuracy and completeness of all records made available to us. In addition, this opinion is based on the assumption that (i) the Merger will be consummated in accordance with the Merger Agreement, the Registration Statement, and the Proxy Statement/Prospectus, (ii) the Merger will qualify as a merger under the applicable laws of Delaware, (iii) each of the parties to the Merger Agreement will comply with all reporting obligations with respect to the Merger required under the Code and the Treasury Regulations thereunder, and (iv) the Merger Agreement is valid and binding in accordance with its terms.

Any inaccuracy in, or breach of, any of the aforementioned statements, representations, warranties, covenants, agreements or assumptions, or any change after the date hereof in applicable law, could adversely affect our opinion. In particular, the failure of any of the aforementioned statements, representations, warranties, covenants, agreements or assumptions to be true as of the Effective Time of Merger could cause the opinion set forth below to be invalid. Similarly, any change in applicable law between the date hereof and the Effective Time of Merger could also cause the opinion set forth below to be invalid. No ruling has been or will be sought from the Internal Revenue Service by any party to the Merger Agreement as to the United States federal income tax consequences of any aspect of the Merger.

Based upon and subject to the assumptions and qualifications set forth herein, the discussion contained under the caption "Material U.S. Federal Income Tax Consequences," represents our opinion with respect to the material U.S. federal income tax consequences of the Merger.

* * * * *

We express no opinion herein other than the opinion expressly set forth above. No opinion is expressed as to the tax consequences of any transaction under foreign, state or local tax law. Our opinion is not binding on the IRS, and the IRS, or a court of law, may disagree with the opinion contained herein. No ruling has been or will be sought from the IRS by any party to the Merger Agreement as to the United States federal income tax consequences of any aspect of the Merger. The opinion set forth in this letter is based on relevant current provisions of the Code, and the Treasury Regulations thereunder (including proposed and temporary Treasury Regulations), and interpretations of the foregoing as expressed in court decisions, applicable legislative history, and the administrative rulings and practices of the Internal Revenue Service (the "IRS"), all as of the date hereof and all of which are subject to change (possibly with retroactive effect). Changes in applicable law could adversely affect our opinion. We do not undertake to advise you as to any changes in applicable law after the date hereof that may affect our opinion.

This opinion is rendered only to you in connection with the filing of the Registration Statement, and is not to be used, circulated, quoted or otherwise referred to or relied upon for any other purpose without our express written permission. In addition, this opinion letter may not be relied upon by or furnished to any other person, firm, corporation or entity without our prior written consent. We hereby consent to the filing of this opinion as Exhibit 8.1 to the Registration Statement. In giving this consent, we do not admit that we are within the category of persons whose consent is required under Section 7 of the Securities Act or the rules and regulations thereunder.

Very truly yours,

Goodwin Procter LLP

SECURITIES PURCHASE AGREEMENT¹

THIS SECURITIES PURCHASE AGREEMENT is made as of [_____], 201[__], by and among (i) Appia, Inc., a Delaware corporation ("Appia"), (ii) Mandalay Digital Group, Inc., a Delaware corporation ("Mandalay" and together with Appia, the "Companies" with each, a "Company"), and (iii) North Atlantic SBIC IV, L.P., a Delaware limited partnership (the "Purchaser" and collectively with the Companies, the "Parties" with each, a "Party").

WHEREAS, on April 4, 2013, Appia and the Purchaser entered into a certain Purchase Agreement, pursuant to which, among other things, Appia issued to the Purchaser a Subordinated Debenture in the amount of \$5,000,000 (the "**First Debenture**");

WHEREAS, on October 31, 2013, Appia and the Purchaser entered into another Purchase Agreement, pursuant to which, among other things, Appia issued to the Purchaser another Subordinated Debenture in the amount of \$3,000,000 (the "Second Debenture");

WHEREAS, on or about November [13], 2014, a certain Agreement and Plan of Merger was entered into among Appia, Mandalay and the other parties thereto (the "Merger Agreement"), pursuant to which, among other things, Mandalay is directly or indirectly acquiring substantially all of the voting control of Appia (the "Acquisition"); and

WHEREAS, in connection and simultaneously with consummation of the Acquisition, the Parties are entering into this Agreement in order to, among other things, provide for the contribution and exchange of the First Debenture and the Second Debenture for a Subordinated Secured Debenture newly issued by Appia to the Purchaser and guaranteed by Mandalay.

NOW, THEREFORE, in consideration of the mutual covenants contained herein, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties, each intending to be legally bound, hereby agree as follows:

SECTION 1. DEFINITIONS.

All capitalized terms not defined herein but defined in <u>Appendix A</u> attached hereto shall have the meanings given to such terms in <u>Appendix A</u>. All terms defined in this Agreement shall also have such defined meanings when used in the other Financing Documents or any certificate or other document made or delivered pursuant hereto or thereto, unless otherwise specified herein or therein.

All references in this Agreement to Sections, Subsections, Exhibits, Schedules and Appendices refer to the Sections, Subsections, Exhibits, Schedules and Appendices of this Agreement unless otherwise indicated. All Exhibits, Schedules and Appendices attached to this Agreement are incorporated herein and made a part hereof.

SECTION 2. THE NEW DEBENTURE.

2.1 <u>Sale and Purchase</u>. Subject to the terms and conditions hereof, and in exchange for the First Debenture and the Second Debenture, Appia shall sell, issue and deliver to the Purchaser, and the Purchaser shall receive from Appia, a Subordinated Secured Debenture (the "<u>New Debenture</u>") in the aggregate principal amount of **Eight Million and 00/100 Dollars (\$8,000,000.00)** (the "<u>Commitment</u>"). The New Debenture shall be substantially in the form of Exhibit A attached hereto, shall be duly executed

Note to draft: Table of contents to be added.

by Appia, shall be dated as of the date of the consummation of the Acquisition pursuant to the Merger Agreement (the "Closing Date"), and shall be made payable to the order of the Purchaser in the amount of the Commitment. Contemporaneously with the issuance of the New Debenture, Appia shall pay in full all amounts of accrued but theretofore unpaid interest then outstanding on the First Debenture and the Second Debenture and all applicable prepayment premiums in respect of the First Debenture and the Second Debenture. The closing of the contribution and exchange of the First Debenture and the Second Debenture and the issuance of the New Debenture, the Common Shares, and the Warrant under this Agreement shall take place remotely via the mutual exchange of documents and signatures if and when all conditions precedent set forth herein have been satisfied or waived and simultaneously with the closing of the Acquisition pursuant to the Merger Agreement on the Closing Date, or at such other time and place as the Parties may mutually agree (the "Closing").

2.2 Subordination. Appia hereby represents that Appia and its Subsidiaries, if any exist, currently owe the Indebtedness set forth on Schedule 2.2 (which does not and shall not exceed Five Million and 00/100 Dollars (\$5,000,000.00) of Senior Debt, and does not and shall not exceed One Million and 00/100 Dollars (\$1,000,000.00) in Purchase Money Indebtedness) ("Existing Senior Debt") which is secured by the assets set forth on Schedule 2.2. As of the Closing, Appia and its Subsidiaries, if any exist, shall have no Senior Debt other than the Existing Senior Debt set forth on Schedule 2.2 (which does not and shall not exceed Five Million and 00/100 Dollars (\$5,000,000.00) of Senior Debt, and does not and shall not exceed One Million and 00/100 Dollars (\$1,000,000.00) in Purchase Money Indebtedness). At such time as the Senior Debt is less than Five Million and 00/100 Dollars (\$5,000,000.00) in the aggregate, if Appia desires to incur Senior Debt not set forth on Schedule 2.2 ("New Senior Debt"), the Purchaser shall agree to subordinate the payment of the principal of and interest on the New Debenture in a commercially reasonable manner, substantially pursuant to the terms of a subordination agreement described in the following sentence, to the payment of such New Senior Debt incurred by Appia in an amount which, together with all existing Senior Debt, shall not exceed Five Million and 00/100 Dollars (\$5,000,000.00). The Purchaser agrees, upon request of any future holder of such New Senior Debt or Replacement Senior Debt, to enter into a commercially reasonable subordination agreement for the benefit of such future holder of such New Senior Debt or Replacement Senior Debt; provided, however, that such subordination agreement shall be on terms which are commercially reasonable and reasonably satisfactory to the Purchaser, as determined by the Purchaser in its good faith discretion, and shall, among other provisions, permit payment of principal and interest under the New Debenture according to its terms unless the Existing, New, or Replacement Senior Debt (as applicable) is in default under the terms thereof.

2.3 Use of Proceeds. The New Debenture shall be issued in exchange for the First Debenture and the Second Debenture.

2.4 Interest Rate. Subject to the provisions of Section 2.5, the unpaid principal balance outstanding under the New Debenture, from time to time during the first twelve (12) months following the Closing Date, shall accrue interest at the rate of Ten and 00/100 Percent (10.00%) per annum. Subject to the provisions of Section 2.5, the unpaid principal balance outstanding under the New Debenture, from time to time after the first twelve (12) months following the Closing Date, shall accrue interest at the rate of Fourteen and 00/100 Percent (14.00%) per annum. Interest per annum shall be calculated on the basis of actual number of days elapsed and an assumed 365-day year. Therefore, each dollar of principal outstanding hereunder for all or any part of a day shall accrue interest equal to 1/365th of the per annum interest accruing hereunder on each such dollar. Interest shall accrue on each day or part thereof that any principal is outstanding, including Sundays, legal holidays, and other non-business days.

2.5 Default Rate.

- (a) Notwithstanding the provisions of <u>Section 2.4</u> to the contrary, upon the occurrence of an Event of Default or if Appia shall fail to pay any interest on the New Debenture within fifteen (15) days after any such interest becomes due in accordance with the terms of this Agreement and the other Financing Documents (an "<u>Interest Default</u>"), the unpaid principal balance outstanding under the New Debenture and any accrued but unpaid interest thereon shall bear interest at a rate per annum (hereinafter referred to as the "<u>Default</u> <u>Rate</u>") equal to Sixteen and 00/100 Percent (16.00%), in each case from the date of the occurrence of the Event of Default or Interest Default until such Event of Default or Interest Default is waived or cured.
- (b) Notwithstanding any provision contained in this Agreement or any other Financing Document to the contrary, in no event shall the amount paid or agreed to be paid by a Company (or any other Person) as interest or as a premium on the New Debenture or any other Obligations exceed the highest lawful rate permissible under any law applicable thereto.
- **2.6 Repayment of the New Debenture.** Appia hereby agrees to pay interest on the unpaid principal amount of the New Debenture, in arrears, on a monthly basis, commencing on the first day of the month immediately following the Closing Date and continuing on the first day of each consecutive month thereafter until the Maturity Date; provided, however, that any such interest accruing at the Default Rate shall be due and payable on demand. On the Maturity Date (or such earlier date on which the New Debenture become due and payable pursuant to Section 9.1), the entire remaining outstanding balance of the New Debenture (including, without limitation, all unpaid principal, all accrued but unpaid interest, and all unpaid fees, charges, costs and expenses) shall be immediately due and payable in full. Any amounts due on a day that is not a Business Day shall be due and payable on the next Business Day.

2.7 Method of Payment.

- (a) Payments to be made by Appia in respect of the New Debenture on account of regularly scheduled monthly interest payments shall be made without set off or counterclaim and shall be made prior to 3:00 p.m., Boston, Massachusetts time, on the due date thereof to the Purchaser by electronic transfer to an account designated by the Purchaser (the "Payment Account"), or such other place or account as the Purchaser may specify in writing from time to time, in Dollars and in immediately available funds. Payments received by the Purchaser after such time shall be deemed to have been received on the next Business Day.
- (b) Unless otherwise agreed to by the Parties in writing, all other payments (including prepayments) to be made by Appia in respect of the New Debenture, whether on account of principal, interest, fees or otherwise, shall be made without set off or counterclaim and shall be made prior to 3:00 p.m., Boston, Massachusetts time, on the due date thereof to the Purchaser, by electronic transfer to the Payment Account (or such other place or account as the Purchaser may specify in writing from time to time), in Dollars and in immediately available funds. Payments received by the Purchaser after such time shall be deemed to have been received on the next Business Day.

2.8 Prepayments.

(a) Optional Prepayment. Appia may at any time and from time to time prepay the New Debenture, in whole or in part. Appia shall give written notice to the Purchaser prior to 3:00 p.m., Boston, Massachusetts time, at least one (1) Business Day prior to such prepayment, specifying the date and respective amounts of prepayment.

- (b) <u>Mandatory Prepayments</u>. Appia shall be obligated to, and shall, prepay the New Debenture in full immediately prior to the consummation of any Liquidity Event.
- (c) <u>Application of Prepayments</u>. All amounts received for the prepayment of the New Debenture shall be applied to the Obligations as follows: *first*, to any unpaid fees, charges, costs and expenses then due and payable under this Agreement or any of the other Financing Documents to the Purchaser; *second*, to any accrued and unpaid interest then due and owing under the New Debenture to the Purchaser; and then *third*, to the unpaid principal balance under the New Debenture to the Purchaser. Any amounts prepaid on account of the New Debenture may not be reborrowed.
- 2.9 Purchase Price Allocation. The Parties, having adverse interests as a result of arm's length bargaining, agree that the Common Shares and the Warrant are part of an investment unit within the meaning of Section 1273(c)(2) of the Code, which includes the New Debenture. Within ninety (90) days following the Closing Date, the Companies shall together prepare and deliver to the Purchaser an allocation of the issue price of the investment among the Closing Securities, calculated in accordance with GAAP, which will also state the original issue discount, if any, on the New Debenture, and which will be mutually agreed upon by the Parties. The Parties agree to prepare their federal income tax returns in a manner consistent with the foregoing and, pursuant to Treas. Reg. § 1.1273-1; provided, however, if the Parties do not reach such an agreement with respect to the allocation, the Parties shall then refer the matter to an independent national or regionally recognized accounting firm reasonably acceptable to the Purchaser for resolution within forty-five (45) days following the Closing Date, and the Parties shall file their respective tax returns in a manner consistent with such resolution. Such accounting firm's fees shall be borne equally among the Parties.
- 2.10 Board Observer. If any interest, principal or other amount due pursuant to the New Debenture remains outstanding on the date which is twelve (12) months from the date hereof, then, for so long as any amount remains outstanding pursuant to the New Debenture, the Purchaser shall be entitled to designate one (1) individual (the "NAC Observer") who shall have the right to attend, in a non-voting observer capacity, all meetings (whether in person, by conference telephone, or otherwise) of the Mandalay Board (including any subcommittees thereof) and to participate in discussions of matters brought to the members of the Mandalay Board for vote or discussion. The NAC Observer shall receive advance notice in substantially the same form and by substantially the same means as the members of the Mandalay Board of any such meetings. In this respect, Mandalay shall give the NAC Observer copies of all notices (in substantially the same form and by substantially the same means as such notices are provided to the members of the Mandalay Board) as well as other materials that it provides to the members of the Mandalay Board for meetings. The NAC Observer shall be entitled to be reimbursed for reasonable out-of-pocket costs and expenses incurred by him or her in connection with attendance at meetings of the Mandalay Board. Notwithstanding the foregoing, (i) the obligations of Mandalay pursuant to this Section 2.10 shall be subject to the NAC Observer agreeing in writing to a customary confidentiality agreement with respect to information so provided, and (ii) Mandalay and the Mandalay Board may withhold any information from and exclude the NAC Observer from any meeting or portion thereof to the extent access to such information or attendance at such meeting could adversely affect the attorney-client privilege between Mandalay and its counsel or result in disclosure of Mandalay's trade secrets or if such information or meeting involves a material conflict of interest with the Purchaser or the NAC Observer.

SECTION 3. COMMON SHARES AND WARRANT.

3.1 <u>Common Shares Issued on the Closing Date</u>. Subject to the terms and conditions hereof, Mandalay shall issue and deliver to the Purchaser on the Closing Date, and the Purchaser shall receive from Mandalay, Two Hundred Thousand (200,000) authorized and theretofore unissued shares (the "<u>Common Shares</u>") of Mandalay Common Stock.

- 3.2 Warrant Issued on the Closing Date. Subject to the terms and conditions hereof, Mandalay shall issue and deliver to the Purchaser on the Closing Date, and the Purchaser shall receive from Mandalay, a Common Stock Purchase Warrant to purchase Four Hundred Thousand (400,000) authorized and theretofore unissued shares of Mandalay Common Stock at an exercise price of \$0.001 per share in substantially the form attached hereto as Exhibit B (the "Warrant"). The Warrant shall be shall be duly executed by Mandalay, shall be dated as of the Closing Date, and shall be exercisable in whole or in part at any time and from time to time during the period commencing on the date such warrant vests and ending on the tenth (10th) anniversary of the Closing Date, unless such exercise period is shortened pursuant to the terms of the Warrant. So long as the Purchaser holds the Warrant or any Warrant Shares, Mandalay shall not amend or restate the Warrant without the prior written consent of the Purchaser.
- **3.3** <u>Registration.</u> Following the Closing, Mandalay shall be obligated to register the Common Shares and the Warrant Shares as set forth on <u>Appendix B</u> attached hereto.
- **SECTION 4.** <u>PURCHASER CLOSING CONDITIONS.</u> In addition to all other conditions set forth in this Agreement, the Purchaser shall not be obligated to consummate the transactions anticipated to occur at Closing unless and until all of the following provisions have been satisfied (or waived by the Purchaser) as of the Closing Date:
- 4.1 Financing Documents, Warrant, Etc. The Purchaser shall have received the following Financing Documents, each as duly executed by the parties thereto, with their signatures properly witnessed and notarized thereon where indicated: (i) this Agreement; (ii) the New Debenture; (iii) the Appia Security Agreement; and (iv) the Mandalay Guaranty. The Purchaser shall also have received the Warrant duly executed and witnessed by Mandalay. The Purchaser shall also have received all disclosure schedules contemplated pursuant to this Agreement and the other Financing Documents, which shall be true, accurate and complete in all material respects.
- **4.2** Actions to Perfect Liens. The Purchaser shall have received evidence in form and substance reasonably satisfactory to it that all filings, recordings and registrations, including, without limitation, the filing of duly executed financing statements on form UCC-1, necessary or, in the opinion of the Purchaser, reasonably desirable to perfect the Liens created by the Security Documents shall have been completed (or, to the extent that any such filings, recordings, registrations and other actions shall not have been completed, arrangements satisfactory to the Purchaser for the completion thereof shall have been made).
- **4.3** <u>Lien Searches.</u> The Purchaser shall have received the results of a recent search by a Person reasonably satisfactory to the Purchaser, of the UCC, judgment and tax lien filings which may have been filed with respect to personal property of Appia and Mandalay and any Subsidiary of Appia and Mandalay. The results of such search shall be reasonably satisfactory to the Purchaser.
- **4.4** <u>UCC-3 Termination Statements.</u> The Purchaser shall have received UCC-3 termination statements and any other instrument necessary to terminate or discharge the Liens granted by Appia and Mandalay and any Subsidiary of Appia and Mandalay to any Person (other than Permitted Liens) (or, to the extent that any such UCC-3 termination statements or any other instrument shall not have been obtained and filed, arrangements satisfactory to the Purchaser for the obtaining and filing thereof shall have been made).

4.5 Corporate Proceedings of the Companies.

- (a) The Purchaser shall have received a copy of the resolutions, in form and substance reasonably satisfactory to the Purchaser, of the board of directors or governing body of Appia (the "Appia Board") and, if applicable, any Subsidiary of Appia authorizing (i) the execution, delivery and performance by Appia (and all Subsidiaries of Appia) of this Agreement and the other Financing Documents to which it is a party, (ii) the issuance and sale by Appia of the New Debenture contemplated hereunder, and (iii) the granting by them of the Liens created pursuant to the Security Documents to which Appia (and any Subsidiary of Appia) is a party, all as certified by the Secretary or an Assistant Secretary of Appia as of the Closing Date, which certificate shall be in form and substance reasonably satisfactory to the Purchaser and shall state that the resolutions thereby certified have not been amended, modified, revoked or rescinded.
- (b) The Purchaser shall have received a copy of the resolutions, in form and substance reasonably satisfactory to the Purchaser, of the board of directors or governing body of Mandalay (the "Mandalay Board") and, if applicable, any Subsidiary of Mandalay authorizing (i) the execution, delivery and performance by Mandalay (and all Subsidiaries of Mandalay) of this Agreement and the other Financing Documents to which it is a party, and the Warrant, (ii) the issuance and sale by Mandalay of the Common Shares, the Warrant, and the Warrant Shares contemplated hereunder, and (iii) the granting of the Liens created pursuant to the Security Documents to which Mandalay (and any Subsidiary of Mandalay) is a party, all as certified by the Secretary or an Assistant Secretary of Mandalay as of the Closing Date, which certificate shall be in form and substance reasonably satisfactory to the Purchaser and shall state that the resolutions thereby certified have not been amended, modified, revoked or rescinded.

4.6 Incumbency Certificates.

- (a) The Purchaser shall have received a certificate of Appia (and all Subsidiaries, of Appia) dated as of the Closing Date, as to the incumbency and signature of the officers of Appia executing any Financing Document reasonably satisfactory in form and substance to the Purchaser, executed by the Chief Executive Officer, President or any Vice President and the Secretary or an Assistant Secretary of Appia.
- (b) The Purchaser shall have received a certificate of Mandalay (and all Subsidiaries of Mandalay), dated as of the Closing Date, as to the incumbency and signature of the officers of Mandalay executing any Financing Document and the Warrant reasonably satisfactory in form and substance to the Purchaser, executed by the Chief Executive Officer, President or any Vice President and the Secretary or an Assistant Secretary of Mandalay.
- **4.7** Charter Documents. The Purchaser shall have received true and complete copies of the Certificate of Incorporation and Bylaws of each of Appia and Mandalay, as certified by the Secretary or an Assistant Secretary of such Company as of the Closing Date.
- **4.8** <u>Legal Existence, Good Standing, Tax Good Standing, and Foreign Qualification Certificates.</u> The Purchaser shall have received (as applicable) certificates of incorporation or legal existence, good standing, tax good standing, and foreign qualification or registration as a foreign company for both Companies and each of their Subsidiaries, each dated no earlier than ten (10) days prior to the Closing Date and issued by the appropriate Governmental Authorities.
- **4.9** <u>Insurance</u>. The Purchaser shall have received evidence in form and substance reasonably satisfactory to it that those Sections of the Security Documents requiring the maintenance of insurance shall have been satisfied.

- **4.10 <u>Legal Opinions.</u>** The Purchaser shall have received executed legal opinions of Goodwin Proctor LLP, outside counsel to Appia, and Latham & Watkins LLP, outside counsel to Mandalay, covering this Agreement, the other Financing Documents, and the Warrant in form and substance reasonably satisfactory to the Purchaser.
 - 4.11 Fees and Expenses. The Purchaser shall have received payment of the Closing Expenses.
- **4.12 Representations and Warranties.** Each of the representations and warranties made by Appia, Mandalay, and each of their Subsidiaries (if any) in or pursuant to the Financing Documents shall be true and correct in all material respects (except where such representations and warranties are qualified by materiality or other similar qualifier in which case they shall be true and correct in all respects as so qualified) on and as of the Closing Date as if made on and as of such respective dates, except to the extent such representations and warranties expressly relate to an earlier date in which case such representations and warranties shall be true and correct in all respects (except where such representations and warranties are qualified by materiality or other similar qualifier in which case they shall be true and correct as so qualified) as of such earlier date; provided, however, that any breach of the representations made or deemed made under Section 5.1 shall not, in and of itself, relieve the Purchaser of its obligation to consummate the Closing unless the Purchaser shall have the right not to consummate the Closing based upon another misrepresentation or other matter, notwithstanding, that the misrepresentation or other matter giving rise thereto may also constitute a breach of Section 5.1 and, provided, further, that nothing in the aforesaid proviso shall in any way constitute a waiver of an Event of Default or in any way limit the Purchaser's rights and remedies under this Agreement, including, without limitation, the provisions of Section 11.7.
- **4.13** No Default. No Default or Event of Default under this Agreement (as if in effect) shall have occurred and be continuing on the Closing Date or after giving effect to the issuance and sale of the Closing Securities.
- **4.14 SBIC Matters.** Appia shall have executed and delivered to the Purchaser a Size Status Declaration on SBA Form 480, an Assurance of Compliance on SBA Form 652, and an SBIC Side Letter in the form set forth on Exhibit C attached hereto, and shall have provided to the Purchaser information necessary for the preparation of a Portfolio Financing Report on SBA Form 1031.
- **4.15** <u>Compliance Certificates</u>. The Purchaser shall have received fully executed Compliance Certificates in the forms of <u>Exhibit D-1</u> and <u>Exhibit D-2</u> dated as of the Closing Date.
- **4.16** Waiver of Anti-Dilution and Other Provisions. The Purchaser shall have received evidence reasonably satisfactory to it of the waiver of any applicable anti-dilution, preemptive, participation and similar rights with respect to the transactions envisioned hereby.
- **4.17** <u>Subordination Amendment</u>. The Purchaser shall have received the Subordination Amendment duly executed by Silicon Valley Bank, with signature of a duly authorized officer of it properly witnessed where indicated thereon.
- **4.18** First Debenture; Second Debenture. The Purchaser shall have received from Appia payment in full for all amounts of accrued but theretofore unpaid interest then outstanding on the First Debenture and the Second Debenture and all applicable prepayment premiums in respect of the First Debenture and the Second Debenture.

- **4.19** Merger Agreement. All documents to be delivered pursuant to the Merger Agreement shall be fully executed, and the Acquisition shall have consummated.
- **4.20 SVB Loan Agreement.** The SVB Loan Agreement and all documents to be delivered pursuant thereto shall be fully executed, and the transactions contemplated thereby shall have consummated simultaneously with the Closing.
- **SECTION 4A. COMPANY CONDITIONS PRECEDENT.** The effectiveness of this Agreement and the agreement of Appia and Mandalay to sell and issue the Closing Securities are subject to the satisfaction or waiver of the following conditions precedent as of the Closing Date:
- **4A.1** <u>Financing Documents</u>. The Companies shall have received the Financing Documents, each as duly executed by the parties thereto, with their signatures properly witnessed and notarized where indicated thereon.
- **4A.2** <u>Subordination Amendment</u>. Appia shall have received the Subordination Amendment, as duly executed by the Purchaser, with signature of a duly authorized officer of it properly witnessed where indicated thereon.
- **4A.3** Representations and Warranties. Each of the representations and warranties made by the Purchaser in or pursuant to the Financing Documents shall be true and correct in all material respects (except where such representations and warranties are qualified by materiality or other similar qualifier in which case they shall be true and correct in all respects as so qualified) on and as of the Closing Date as if made on and as of such date.

SECTION 5. REPRESENTATION AND WARRANTIES OF APPIA.

To induce the Purchaser to enter into this Agreement and to purchase the Closing Securities, Appia hereby makes the following representations and warranties to the Purchaser as of the Closing Date, as qualified in the Master Disclosure Schedule delivered by Appia to the Purchaser on the Closing Date, which shall be arranged by section numbers that correspond to the section numbers in this Agreement and the disclosures in any Section of the Disclosure Schedules shall provide information regarding, and shall qualify only, the corresponding numbered and lettered Section of this Section 5; provided, however, that any disclosure made in one Section which is referred to in one or more other Sections shall be deemed to qualify such other Sections as if such disclosure had been fully set forth in such other Section(s). No information in any Disclosure Schedule shall be deemed to qualify a statement contained in this Section 5 unless the Disclosure Schedule identifies the qualification with reasonable particularity so that it is readily apparent from a reading of such information that such information is applicable to another section of the Master Disclosure Schedule.

For purposes of these representations and warranties (other than those in <u>Sections 5.2, 5.3, and 5.5)</u>, including, without limitation, for purposes of the terms defined in <u>Appendix A</u> hereto, the term "Company" shall include any Subsidiaries of such Company, unless otherwise noted herein.

- **5.1** <u>SVB Loan Agreement Representations</u>. Appia hereby makes to the Purchaser each and every representation and warranty it has made and is making in the SVB Loan Agreement, such representations and warranties being deemed restated and made here for the benefit of and reliance by the Purchaser as if made directly to the Purchaser.
- **5.2** <u>Authorization</u>. Appia has all requisite corporate power and has taken all necessary corporate action required for the due authorization, execution, delivery and performance by Appia of the

Financing Documents and the consummation of the transactions contemplated hereby and thereby. Appia has taken all actions under its Certificate of Incorporation or its Bylaws as are necessary to provide the Purchaser with the rights hereby contemplated. The Financing Documents, when executed and delivered by Appia, shall constitute valid and legally binding obligations of Appia, enforceable against Appia in accordance with their respective terms except (i) as limited by applicable bankruptcy, insolvency, reorganization, moratorium, fraudulent conveyance, or other laws of general application relating to or affecting the enforcement of creditors' rights generally, or (ii) as limited by laws relating to the availability of specific performance, injunctive relief, or other equitable remedies.

- **5.3** Governmental Consents. Other than filings required to be made under applicable federal and state securities laws, neither the execution, delivery or performance of the Financing Documents by Appia nor the consummation by it of the obligations and transactions contemplated hereby or thereby (including, without limitation, the issuance and delivery of the New Debenture) requires any consent of, authorization by, exemption from, filing with or notice to any Governmental Authority or any other Person.
- **5.4** Solvency. After giving effect to the transactions contemplated to occur on the Closing Date (including the Acquisition), Appia is Solvent.

5.5 Federal Regulations.

- (a) SBIC Regulations.
- (i) Appia acknowledges and understands that the Purchaser is an SBIC and that the SBIC Regulations prohibit certain uses of proceeds of loans or investments made by a SBIC.
 - (ii) No part of the proceeds of the New Debenture will be used in violation of the SBIC Regulations.
- (iii) As of November 1, 2014, the Consolidated Tangible Net Worth for Appia was not in excess of Eighteen Million and 00/100 Dollars (\$18,000,000.00). For each of Appia's fiscal years ended December 31, 2012 and December 31, 2013, the average Consolidated Net Income after payment of federal taxes for each such fiscal year (excluding any carryover losses for each such fiscal year) for Appia was not in excess of Six Million and 00/100 Dollars (\$6,000,000.00).
- (b) <u>Margin Regulations</u>. No part of the proceeds of the New Debenture will be used for "purchasing" or "carrying" any "margin stock" within the respective meanings of each of the quoted terms under Regulations G, U or X of the Board of Governors of the Federal Reserve System as now and from time to time hereafter in effect. If requested by the Purchaser, Appia will furnish to the Purchaser, a statement to the foregoing effect in conformity with the requirements of FR Form U-1 referred to in said Regulation U.
- **5.6** <u>Investment Company Act</u>. Appia is not an "investment company" within the meaning of the Investment Company Act of 1940, as amended. Appia is not subject to regulation under any Federal or State statute or regulation which limits its ability to incur the Indebtedness contemplated herein.
- **5.7** <u>Disclosure</u>. No representation or warranty by Appia contained in any Financing Document, in the <u>Master Disclosure Schedule</u>, or in any certificate furnished at the Closing Date to the

Purchaser pursuant to any Financing Document contains any untrue statement of a material fact or omits to state a material fact necessary to make the statements herein or therein, in light of the circumstances in which it was made, not misleading. It is understood that this representation is qualified by the fact that Appia has not delivered to the Purchaser, and has not been requested to deliver, a private placement or similar memorandum or any written disclosure of the types of information customarily furnished to purchasers of securities.

SECTION 5A. REPRESENTATION AND WARRANTIES OF MANDALAY.

To induce the Purchaser to enter into this Agreement and to purchase the Closing Securities, Mandalay hereby makes the following representations and warranties to the Purchaser as of the Closing Date, except as otherwise described in the Master Disclosure Schedule delivered by the Companies to the Purchaser on the Closing Date, which shall be arranged by section numbers that correspond to the section numbers in this Agreement and the disclosures in any Section of the Disclosure Schedules shall provide information regarding, and shall qualify only, the corresponding numbered and lettered Section of this Section 5A; provided, however, that any disclosure made in one Section which is referred to in one or more other Sections shall be deemed to qualify such other Sections as if such disclosure had been fully set forth in such other Section(s). No information in any Disclosure Schedule shall be deemed to qualify a statement contained in this Section 5A unless the Disclosure Schedule identifies the qualification with reasonable particularity so that it is readily apparent from a reading of such information that such information is applicable to another section of the Master Disclosure Schedule.

- **5A.1** Organization of the Purchaser. Mandalay is a corporation duly organized, validly existing, and in good standing under the laws of the State of Delaware, with full corporate power and authority to own and lease their respective properties and assets and conduct their respective businesses as presently being conducted.
- **5A.2** <u>Authorization</u>. Mandalay has all requisite corporate power and authority to execute and deliver this Agreement and the other Financing Documents and the Warrant to be executed and delivered by Mandalay pursuant hereto and to consummate the transactions contemplated hereby and thereby and to perform their obligations hereunder and thereunder. The execution and delivery by Mandalay of this Agreement and the Warrant and the other Financing Documents to which it is a party and the consummation by Mandalay of the transactions contemplated hereby and thereby have been duly approved by the Mandalay Board. No other corporate proceedings on the part of Mandalay is necessary to authorize this Agreement and the other Financing Documents to which it is a party and the Warrant and the transactions contemplated hereby and thereby. This Agreement and the Warrant and the other Financing Documents to which it is a party have been duly executed and delivered by Mandalay and are legal, valid and binding obligations of Mandalay, enforceable against it in accordance with their terms, in each case, except as such enforceability may be limited by (i) bankruptcy, insolvency, moratorium, reorganization or other similar laws affecting creditors' rights generally, and (ii) the general principles of equity, regardless of whether asserted in a proceeding in equity or at law.
- **5A.3** No Conflict or Violation; Required Consents and Filings. Neither the execution, delivery or performance of this Agreement or the Warrant or the other Financing Documents to which Mandalay is a party, nor the consummation of the transactions contemplated hereby or thereby, nor compliance by Mandalay with any of the provisions hereof or thereof, will (i) violate or conflict with any provision of the Organizational Documents of Mandalay, (ii) violate, conflict with, or result in or constitute a Default under, or result in the termination of, or accelerate the performance required by, or result in a right of termination or acceleration under, or result in the creation of any Encumbrance upon any of Mandalay's assets under, any of the material terms, conditions or provisions of any material contract, indebtedness, note, bond, indenture, security or pledge agreement, commitment, license, lease,

franchise, permit, agreement, authorization, concession, or other instrument or obligation to which Mandalay is a party, except for any violation, conflict, Default, termination, acceleration or creation of Encumbrance which would not prevent or materially delay the ability of Mandalay to consummate the transactions contemplated by this Agreement or the other Financing Documents, or (iii) violate any material Regulation or Court Order. The execution and delivery of this Agreement by Mandalay does not, and the performance of this Agreement and the transactions contemplated hereby by Mandalay will not, require any consent, approval, authorization or permit of, or filing with or notification to, any Governmental Authority, except (i) the filing with the SEC of (A) the Form S-4 and declaration of effectiveness of the Form S-4 and (B) such reports under, and other compliance with, the Exchange Act (and the rules and regulations promulgated thereunder) and the Securities Act (and the rules and regulations promulgated thereunder) as may be required in connection with this Agreement and NASDAQ, (iii) the filing of the Certificate of Merger and the acceptance for record by the Delaware Secretary or State, (v) such filings and approvals as may be required by any applicable state securities or "blue sky" Laws, (vi) such filings as may be required in connection with the HSR Act, and (vii) where failure to obtain such consents, approvals, authorizations or permits, or to make such filings or notifications, individually or in the aggregate, has not had and would not reasonably be expected to be material adverse to Mandalay.

- **5A.4** Reservation of Warrant Shares. A sufficient number of authorized but unissued shares of Mandalay Common Stock have been reserved for issuance upon exercise of the Warrant (the "Warrant Shares").
- **5A.5** <u>Mandalay Common Stock.</u> All of the Common Shares and Warrant Shares, when issued and delivered in accordance with this Agreement or the Warrant, shall be duly authorized, validly issued, fully paid, non-assessable, and free from all taxes imposed on Mandalay and Liens and will be free of restrictions on transfer other than restrictions on transfer under applicable state and federal securities laws. None of such shares of Mandalay Common Stock shall be subject to any preemptive rights.
- **5A.6** No Indebtedness. Neither Mandalay nor any of its Subsidiaries has any Indebtedness, contingent or otherwise, which rank pari passu or senior to the Obligations in (i) a principal amount greater than \$5,000,000.00, or (ii) Purchase Money Indebtedness of more than \$1,000,000.00 in aggregate principal amount.
- **5A.7** No Brokers. Neither Mandalay nor any of its respective Representatives or Affiliates has employed or made any agreement with any broker, finder or similar agent or any Person which will result in the obligation of Mandalay or any of its Affiliates to pay any finder's fee, brokerage fees or commission, or similar payment in connection with the transactions contemplated hereby.
- **5A.8** SEC Reports and Financial Statements. A true and complete copy of each annual, quarterly and other report, registration statement, and definitive proxy statement filed by Mandalay with the SEC since January 1, 2013 and prior to the Closing Date (the "Mandalay SEC Documents") is available on the Web site maintained by the SEC at http://www.sec.gov, other than portions in respect of which confidential treatment was granted by the SEC. As of their respective filing dates, the Mandalay SEC Documents complied in all material respects with the requirements of the United States Securities Act and the Exchange Act, as the case may be, and the rules and regulations of the SEC promulgated thereunder applicable to such Mandalay SEC Documents, and none of the Mandalay SEC Documents contained on their filing dates any untrue statement of a material fact or omitted to state a material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances under which they were made, not misleading, except to the extent corrected by a Mandalay SEC Document filed prior to the date of this Agreement.

5A.9 Disclosure Documents. None of the information supplied or to be supplied in writing by or on behalf of Mandalay for inclusion or incorporation by reference in (i) the Form S-4 will, at the time such document is filed with the SEC, at any time such document is amended or supplemented or at the time such document is declared effective by the SEC, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary to make the statements therein not misleading, or (ii) the Joint Proxy Statement (as defined in the Merger Agreement) will, at the date it is first mailed to the Stockholders of Appia and Mandalay, at the time of the Company Stockholder Meeting (as defined in the Merger Agreement) and the Parent Stockholder Meeting (as defined in the Merger Agreement), at the time the Form S-4 is declared effective by the SEC or as of the Closing, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances in which they were made, not misleading. All documents that Mandalay is responsible for filing with the SEC in connection with the transactions contemplated by this Agreement, to the extent relating to Mandalay or any Subsidiary or other information supplied by or on behalf of Mandalay or any Subsidiary for inclusion therein, will comply as to form, in all material respects, with the provisions of the Securities Act or Exchange Act, as applicable, and the rules and regulations of the SEC thereunder and each such document required to be filed with any Governmental Authority (other than the SEC) will comply in all material respects with the provisions of any applicable Law as to the information required to be contained therein.

SECTION 6. REPRESENTATIONS AND WARRANTIES OF PURCHASER. To induce the Companies to enter into this Agreement and to issue the Closing Securities, the Purchaser hereby represents and warrants to the Companies as follows:

- 6.1 Legal Existence. The Purchaser is duly organized, validly existing, and in good standing under the laws of the State of Delaware.
- **6.2** <u>Power; Authorization.</u> The Purchaser has the power and authority, and the legal right, to make, deliver and perform the Financing Documents to which it is a party. All action on the part of the Purchaser and, as applicable, its officers and partners necessary for the authorization, execution, delivery and performance of all obligations of the Purchaser under the Financing Documents to which the Purchaser is a party has been taken.
- **6.3** Enforceable Obligations. Each of the Financing Documents to which the Purchaser is a party, when executed and delivered by the Purchaser, will constitute a legal, valid and binding obligation of the Purchaser, enforceable against the Purchaser, except as such enforceability may be limited by applicable bankruptcy, insolvency, reorganization, moratorium or similar laws affecting the enforcement of creditors' rights generally and by general equitable principles (whether enforcement is sought by proceedings in equity or at law).
- **6.4** Accredited Investor. The Purchaser is an "Accredited Investor" (as such term is defined in Rule 501 of Regulation D of the Securities Act). The financial situation of the Purchaser is such that it can afford to bear the economic risk of holding the unregistered Securities for an indefinite period of time. The Purchaser can afford to suffer the complete loss of its investment in the Securities. The knowledge and experience of the Purchaser in financial and business matters is such that it is capable of evaluating the risk of the investment in the Securities. The Purchaser acknowledges that it has had access to such financial and other information, and has been afforded the opportunity to ask such questions of representatives of Mandalay and Appia and receive answers thereto, as the Purchaser has deemed necessary in connection with its decision to purchase the Securities. The Purchaser is acquiring the Securities if and when it exercise the Warrant for its own account for investment purposes and not with a view to, or for sale in connection with, any distribution thereof, nor with any present intention of distributing or selling the same. By executing this Agreement, the Purchaser further represents that the Purchaser does not presently have any contract, undertaking, agreement or arrangement with any Person to sell, transfer or grant participation to such Person with respect to any of the Securities.

- **6.5** Brokers or Finders. No Person has or will have, as a result of the transaction contemplated by this Agreement or any of the other Financing Documents, any right, interest or claim against or upon the Purchaser, Appia or Mandalay for the commission, fee or other compensation as a finder or broker because of any act or omission by the Purchaser. The Purchaser agrees to defend, indemnify, and hold harmless Appia and Mandalay from and against any such right, interest or claim against or upon a Company for any commission, fee or other compensation due or claimed to be due to any finder or broker arising out of any act or omission by the Purchaser.
 - 6.6 No Public Market. The Purchaser understands that no public market now exists for the New Debenture or the Warrant.
- **6.7** No General Solicitation. Neither the Purchaser, nor any of its officers, employees, agents or partners has either directly or indirectly, including through a broker or finder (i) engaged in any general solicitation, or (ii) published any advertisement in connection with the offer and sale of the New Debenture.
- **6.8** <u>Domicile</u>. The office or offices of the Purchaser in which its principal place of business is identified in the address or addresses of the Purchaser set forth in Section 11.5(b).
- **6.9** <u>Disclosure of Information</u>. The Purchaser has received a copy of the Master Disclosure Schedule and a copy of any document referenced therein which the Purchaser has requested. The Purchaser believes it has received all the information it considers necessary or appropriate for deciding whether to purchase the Securities. The Purchaser has had an opportunity to discuss the business, management and financial affairs of Appia and Mandalay with their respective management, as well as the terms and conditions of the offering of the Securities.

SECTION 7. CERTAIN COVENANTS RELATED TO THE SECURITIES AND OUTSTANDING OBLIGATIONS.

For so long as either (i) any amount of the Obligations (other than inchoate indemnity obligations) remains outstanding, or (ii) the Purchaser continues to hold any of the Securities, Appia hereby agrees, subject to the limitations set forth in the following sections of this Section 7, with the Purchaser as follows:

- **7.1** Financial Statements. If the Obligations have not been paid in full on or before the first anniversary of the Closing Date, Appia shall furnish to the Purchaser:
 - (a) As soon as available, but in any event within one hundred eighty (180) days after the end of each fiscal year of Appia, a copy of the audited consolidated balance sheet and statement of stockholders' equity of Appia and its Subsidiaries as at the end of such year and the related audited statements of income and retained earnings and of cash flows for such year, setting forth in each case in comparative form the figures for the previous year, from a nationally or regionally recognized accounting firm reasonably acceptable to Purchaser; and
 - (b) As soon as available, but in any event not later than thirty (30) days after the end of each of the first eleven (11) calendar months of each fiscal year of Appia, the unaudited consolidated balance sheet of Appia and its Subsidiaries as at the end of such month and the related unaudited consolidated statements of income and retained earnings and of cash flows of

Appia and its Subsidiaries for such month and the portion of the fiscal year through the end of such month, setting forth in each case in comparative form the figures for the month compared to the prior month and to budget and a management discussion and analysis in reasonable detail of the prior month's results, certified by a Responsible Officer of Appia as being fairly stated in all material respects (subject to normal year end audit adjustments).

All such financial statements shall be prepared in accordance with GAAP (subject in the case of the unaudited financials to the absence of footnotes and normal year end audit adjustments) applied consistently throughout the periods reflected therein and with prior periods.

- **7.2** <u>Budgets.</u> If the Obligations have not been paid in full on or before the first anniversary of the Closing Date, Appia shall furnish to the Purchaser as soon as available, but in any event prior to the beginning of the fiscal year, Appia's annual budget and business plan, including Capital Expenditures, approved by the Appia Board.
- **7.3** Other Information. Appia shall, and shall cause each of its Subsidiaries to, furnish to the Purchaser promptly, such other information relating to the financial condition, business or corporate affairs of Appia as the Purchaser may reasonably request in writing, including, without limitation, any such information to the extent required by the SBIC Regulations.
- **7.4** Media Releases and Governmental Filings. Appia shall furnish to the Purchaser promptly, copies of all filings by Appia or any of its Subsidiaries with any Governmental Authority and of all press or media releases by Appia or any of its Subsidiaries.
- **7.5** Notice of Liquidity Event. Appia shall, and shall cause each of its Subsidiaries to, give prompt notice to the Purchaser of any expected Liquidity Event. Any notice under this Section 7.5 shall be accompanied by a statement of a Responsible Officer setting forth details of the occurrence referred to therein and stating what action Appia proposes to take with respect thereto but shall not be required to include any privileged information. In the event that Appia becomes subject to the reporting requirements under the Exchange Act, then notices required hereby to be delivered may be deemed delivered via the prompt public filing of copies of all periodic and other reports, proxy statements and other materials filed by Borrower with the SEC, any Governmental Authority succeeding to any or all of the functions of the SEC or with any national securities exchange, or distributed to its shareholders, as the case may be. Documents required hereby to be delivered (to the extent any such documents are included in materials otherwise filed with the SEC) may be delivered electronically and if so delivered, shall be deemed to have been delivered on the date on which Appia posts such documents, or provides a link thereto, on Appia's website on the Internet at Appia's website address.
- **7.6** Inspection of Property; Books and Records; Discussions. Appia shall, and shall cause each of its Subsidiaries to: (i) keep proper financial records in conformity with GAAP and all Requirements of Law; and (ii) (A) permit representatives of the Purchaser, at the Purchaser's own cost and expense, (including their respective officers, employees, consultants, agents and other designees (including without limitation, the SBA)) to visit and inspect any of its properties and examine and make abstracts from any of its books and records at any reasonable time during normal business hours, upon reasonable prior notice, and as often as may reasonably be desired; and (B) permit, upon reasonable notice, which shall not be less than three (3) Business Days, during normal business hours, representatives of the Purchaser (including their respective officers, employees, consultants, agents and other designees (including without limitation, the SBA)) to discuss the business, operations, properties and financial and other condition of Appia and its Subsidiaries with officers and senior management employees of Appia and its Subsidiaries and with its independent certified public accountants. Notwithstanding anything to the contrary in this Section 7.6, if there shall have occurred and be continuing an Event of Default, all costs associated with the rights of the Purchaser described in Section 7.6(ii)(A) above shall be borne by Appia.

7.7 Confidentiality. The Purchaser agrees to use the same degree of care as the Purchaser uses to protect its own confidential information to keep confidential any information furnished to the Purchaser pursuant to this Agreement that Appia identifies as being confidential or proprietary (so long as such information is not in the public domain), except that the Purchaser may disclose such proprietary or confidential information: (i) to any partner, subsidiary or parent of the Purchaser for the purpose of evaluating its investment in the Companies as long as such partner, subsidiary or parent is advised of and agrees or has agreed to be bound by the confidentiality provisions of this Section 7.7 or comparable restrictions; (ii) at such time as it enters the public domain through no fault of the Purchaser; (iii) that is communicated to it free of any obligation of confidentiality; (iv) that is developed by the Purchaser or its agents independently of and without reference to any confidential information communicated by Appia or Mandalay; (v) as required by applicable law; or (vi) to its attorneys, accountants and tax advisors who are subject to obligations of confidentiality in connection with the Purchaser's investment in the Companies.

SECTION 8. <u>ADDITIONAL COVENANTS</u>. For so long as any amount of the Obligations remains outstanding, Appia hereby covenants and agrees with the Purchaser as follows:

- **8.1** <u>Limitation on Indebtedness.</u> Without the consent of the Purchaser, Appia shall not, and shall not permit any of its Subsidiaries to, create, incur, assume or suffer to exist any Indebtedness, except:
 - (a) Senior Debt which shall not exceed Five Million and 00/100 Dollars (\$5,000,000.00), and Purchase Money Indebtedness which shall not exceed One Million and 00/100 Dollars (\$1,000,000.00). At such time as the aggregate amount outstanding of Senior Debt is less than Five Million and 00/100 Dollars (\$5,000,000.00), Appia may incur additional Indebtedness other than Purchase Money Indebtedness, but only to the extent that Senior Debt is less than Five Million and 00/100 Dollars (\$5,000,000.00). At such time as the aggregate amount outstanding of Purchase Money Indebtedness is less than One Million and 00/100 Dollars (\$1,000,000.00), Appia may incur additional Purchase Money Indebtedness, but only to the extent that Purchase Money Indebtedness is less than One Million and 00/100 Dollars (\$1,000,000.00);
 - (b) the Obligations;
 - (c) current liabilities which are incurred in the ordinary course of business and which are not incurred through (i) the borrowing of money or (ii) the obtaining of credit except for credit on an open account basis customarily extended and in fact extended in connection with normal purchases of goods and services, including credit incurred in the ordinary course of business with corporate credit cards;
 - (d) Indebtedness with respect to taxes or assessments which are not yet due or which are being contested in good faith by appropriate proceedings, provided that adequate reserves with respect thereto are maintained on the books of Appia or its Subsidiaries, as the case may be, in conformity with GAAP;
 - (e) Indebtedness of Appia or any Subsidiary, including Indebtedness of Appia to any Subsidiary and of any Subsidiary to Appia or any other Subsidiary, so long as such Indebtedness is subordinated in right of payment to all Obligations on such terms and conditions as the Purchaser may reasonably require; and

- (f) extensions, refinancings and renewals of Indebtedness set forth above in <u>Section 8.1(a)</u>, provided that the principal amount is not increased or the terms modified to impose materially more burdensome terms upon a Company or any Subsidiary.
- **8.2** <u>Limitation on Contingent Liabilities.</u> Without the consent of the Purchaser, Appia shall not, and Appia shall not permit any of its Subsidiaries to, agree to or become liable for any Contingent Indebtedness, except for: (i) the Obligations; (ii) guarantees made in the ordinary course of the business by Appia of obligations of any Subsidiary, which obligations are otherwise permitted under this Agreement; and (iii) endorsements for collection or deposit in the ordinary course of business.
- **8.3** <u>Limitation on Liens</u>. Without the consent of the Purchaser, Appia shall not, and Appia shall not permit any of its Subsidiaries to, create, incur, assume or suffer to exist any Lien upon any of its property, assets or revenues, whether now owned or hereafter acquired, except for the following (hereinafter referred to collectively as "<u>Permitted Liens</u>"):
 - (a) Liens to secure the Senior Debt or Replacement Senior Debt;
 - (b) Liens created pursuant to the Security Documents;
 - (c) Liens for taxes not yet due or which are being contested in good faith by appropriate proceedings, provided that adequate reserves with respect thereto are maintained on the books of Appia or its Subsidiaries, as the case may be, in conformity with GAAP;
 - (d) statutory landlords' liens and carriers', warehousemen's, mechanics', materialmen's, repairmen's or other like Liens arising in the ordinary course of business and imposed without action of such parties for sums which are not overdue;
 - (e) judgment Liens created by or resulting from any litigation or legal proceeding provided that such Liens do not also constitute an Event of Default;
 - (f) pledges or deposits made in the ordinary course of business in connection with workers' compensation, unemployment insurance, and other social security legislation and deposits securing liability to insurance carriers under insurance or self-insurance arrangements;
 - (g) deposits to secure the performance of bids, trade contracts (other than for borrowed money), leases, statutory obligations (other than liens arising under ERISA or environmental liens), surety and appeal bonds, indemnity and performance bonds, and other obligations of a like nature incurred in the ordinary course of business;
 - (h) Liens consisting of easements, zoning restrictions, flowage rights, rights-of-way, covenants, conditions, restrictions, reservations, licenses, agreements and other similar matters, which, in the aggregate, are not substantial in amount and which do not in any case materially detract from the use of the property subject thereto or materially interfere with the ordinary conduct of the business of a Company or its Subsidiaries;
 - (i) Liens to secure Indebtedness for Purchase Money Indebtedness to the extent that such Indebtedness is permitted under Section 8.1(a); provided, however, that (A) each such Lien is given only to secure the purchase price of the property which is the subject of such Purchase Money Indebtedness, does not extend to any other property, and is given at the time of acquisition of the property; and (B) the Purchase Money Indebtedness secured thereby does not exceed the lesser of the cost of such property or its fair market value at the time of acquisition;

- (j) Liens in favor of lessors under Capitalized Leases to the extent that the Capitalized Lease Obligations thereunder is Indebtedness permitted under <u>Section 8.1(a)</u>; <u>provided, however</u>, that each such Lien extends only to the property which is subject of such Capitalized Lease, is given only to secure the Capitalized Lease Obligations under such Capitalized Lease, and is given at the commencement date of such Capitalized Lease; and
- (k) Liens incurred in connection with the extension, renewal or refinancing of the Indebtedness secured by Liens of the type described in <u>Section 8.3(a)</u>; <u>provided</u> that any extension, renewal or replacement Lien shall be limited to the property encumbered by the existing Lien and the principal amount of the Indebtedness being extended, renewed or refinanced (as may have been reduced by any payment thereon) does not increase;
- **8.4** <u>Limitation on Fundamental Changes</u>. Other than in connection with a Liquidity Event in respect of which the New Debenture is repaid in full in accordance with <u>Section 2.8(b)</u>. Appia shall not, and Appia shall not permit any of its Subsidiaries to, experience a Liquidity Event or make any material change in its present method of conducting business (which consists of advertising, marketing and sales within the mobile industry and other reasonably related industries) or its currently proposed business, except:
 - (a) any Subsidiary of Appia may be merged or consolidated with Appia (provided that Appia shall be the continuing or surviving corporation) or with or into any one or more wholly-owned Subsidiaries of Appia (provided that such wholly-owned Subsidiary or Subsidiaries shall be the continuing or surviving corporation);
 - (b) any wholly-owned Subsidiary may sell, lease, transfer or otherwise dispose of any or all of its assets (upon voluntary liquidation or otherwise) to Appia or any other wholly- owned Subsidiary of Appia;
 - (c) pursuant to any sale of assets expressly permitted by Section 8.5; and
 - (d) Appia may be merged into or consolidated with Mandalay provided that Mandalay shall assume all of Appia's obligations and responsibilities pursuant to this Agreement.
- **8.5** <u>Limitation on Sale of Assets</u>. Without the consent of the Purchaser, Appia shall not, and Appia shall not permit any of its Subsidiaries to, convey, sell, lease, assign, transfer or otherwise dispose of any of its property, business or assets (including, without limitation, receivables and leasehold interests), whether now owned or hereafter acquired, or, in the case of any Subsidiary, issue or sell any shares of such Subsidiary's Capital Stock to any Person except:
 - (a) the conveyance, sale, lease, assignment, transfer or other disposition of Obsolete Property or surplus property;
 - (b) the sale of inventory in the ordinary course of business;
 - (c) the sale or discount for fair value, without recourse and consistent with sound business practices of accounts receivable arising in the ordinary course of business in connection with the compromise or collection thereof;
 - (d) the license of Intellectual Property Assets in the ordinary course of business;

- (e) leases or subleases of property not materially interfering with the ordinary course of conduct of the business of a Company or its Subsidiaries;
 - (f) the sale or transfer of property and assets to the extent and as permitted by <u>Section 8.4(b)</u> or <u>Section 8.4(d)</u>;
- (g) the sale or transfer of "equipment" (as defined in the UCC) so long as such equipment is promptly replaced by equipment of equal or greater value;
 - (h) the sale or transfer of property or assets in connection with Permitted Liens; and
- (i) the sale or transfer of property or assets with a fair market value not exceeding One Hundred Thousand and 00/100 Dollars (\$100,000.00) in the aggregate per annum.
- **8.6** <u>Limitation on Sales and Leasebacks</u>. Without the consent of the Purchaser, Appia shall not, and Appia shall not permit any of its Subsidiaries to, enter into any arrangement with any Person providing for the leasing by Appia or any Subsidiary of real or personal property which has been or is to be sold or transferred by Appia or such Subsidiary to such Person or to any other Person to whom funds have been or are to be advanced by such Person on the security of such property or rental obligations of Appia or such Subsidiary.
- **8.7** <u>Payment of Obligations.</u> Appia shall, and shall cause each of its Subsidiaries to, pay, discharge or otherwise satisfy at or before maturity or before they become delinquent, as the case may be, all of the material obligations of Appia or such Subsidiary, whichever is applicable, (except obligations for which the amount or validity thereof is currently being contested in good faith by appropriate proceedings and reserves in conformity with GAAP with respect thereto have been provided on the books of Appia or any of its Subsidiaries, as the case may be).
- **8.8** Conduct of Business and Maintenance of Existence. Appia shall, and shall cause each of its Subsidiaries to: (i) continue to engage in business of the same general type as now conducted by Appia (which consists of advertising, marketing and sales within the mobile industry) or proposed to be conducted by Appia or a business reasonably related thereto and as reasonably following from the Acquisition; (ii) preserve, renew and keep in full force and effect its existence and take all reasonable action to maintain all rights, privileges and franchises necessary or desirable in the normal conduct of its business except as otherwise permitted pursuant to this Section 8.8; and (iii) comply in all material respects with all Contractual Obligations and Requirements of Law, except where (A) any such Contractual Obligation is being contested in good faith, a bona fide dispute exists with respect to any such Contractual Obligation, or failure to comply therewith would not, in the aggregate, reasonably be expected to have a Material Adverse Effect, or (B) any such Requirement of Law is being contested in good faith or the failure to comply therewith would not, in the aggregate, reasonably be expected to have a Material Adverse Effect.

A remediable failure by Appia to comply with the covenants set forth in this <u>Section 8.8</u> shall not constitute a Default or Event of Default hereunder if such failure or failures, in the aggregate, would not reasonably be expected to have a Material Adverse Effect, <u>unless</u> such failure shall continue unremedied for a period of twenty (20) days after the earlier of (i) the date on which a Responsible Officer of Appia first learns of such failure or (ii) the date on which written notice thereof shall have been given to Appia by the Purchaser.

8.9 <u>Maintenance of Property; Insurance</u>. Appia shall, and shall cause each of its Subsidiaries to (i) keep all property material to the conduct of its business in good working order and

condition, normal wear and tear excepted; (ii) maintain insurance with financially sound and reputable insurance companies on such of its property and in at least such amounts and against at least such risks as are usually insured against in the same general area by companies engaged in the same or a similar business; and (iii) furnish to the Purchaser, upon written request, full information as to the insurance carried.

A remediable failure by Appia to comply with the covenants set forth in clause (i) or clause (ii) hereinabove shall not constitute a Default or Event of Default hereunder unless such failure shall continue unremedied for a period of twenty (20) days after the earlier of (A) the date on which a Responsible Officer of Appia first learns of such failure or (B) the date on which written notice thereof shall have been given to Appia by the Purchaser.

- **8.10** Maintenance of Liens of the Security Documents. Appia shall, and shall cause each of its Subsidiaries to, promptly upon the reasonable request of the Purchaser, at the sole cost and expense of Appia and its Subsidiaries, execute, acknowledge and deliver, or cause the execution, acknowledgement and delivery of, and thereafter register, file or record, or cause to be registered, filed or recorded, in an appropriate governmental office, any document or instrument supplemental to or confirmatory of the Security Documents or otherwise reasonably deemed by the Purchaser necessary or desirable for the continued validity, perfection and priority of the Liens on the Collateral covered thereby.
- **8.11** Pledge of After Acquired Property. If at any time following the Closing, Appia or any of its Subsidiaries acquires property of any nature whatsoever having a value in excess of Fifty Thousand Dollars (\$50,000) which is intended by the terms of the applicable Security Document to be, but is not, subject to the Liens created by the Security Documents, Appia shall, or shall cause its relevant Subsidiaries to, as soon as possible and in no event later than thirty (30) days after the relevant acquisition date and, to the extent permitted by applicable law, grant to the Purchaser a first priority (subject to Permitted Liens) Lien on such property as collateral security for the Obligations pursuant to documentation reasonably satisfactory in form and substance to the Purchaser. Appia, at its own expense, shall execute, acknowledge and deliver, or cause the execution, acknowledgement and delivery of, and thereafter register, file or record in an appropriate governmental office, any document or instrument (including legal opinions, title insurance, consents and corporate documents) and take all such actions reasonably deemed by the Purchaser to be necessary or desirable to ensure the creation, priority and perfection of such Lien.
- **8.12** Prohibition of Liens. Without the consent of the Purchaser, Appia shall not, and Appia shall not permit any of its Subsidiaries to, enter into with any Person any agreement (other than this Agreement and the other Financing Documents) which prohibits or limits the ability of Appia or any of its Subsidiaries to create, incur, assume or suffer to exist any Lien (other than Permitted Liens) upon the property or assets of Appia, whether now existing or hereafter arising, wherever such property or assets may be located.
- **8.13** New Subsidiaries. Except as permitted by Section 8.15(i), Appia shall not create or acquire any direct or indirect Subsidiary without the express written consent of the Purchaser. With regard to any Subsidiary which the Purchaser permits Appia to create or acquire, Appia shall cause, at its sole cost and expense, each such Subsidiary that is organized (unless Purchaser otherwise consents in writing), immediately upon such creation or acquisition to execute and deliver to the Purchaser the following agreements and documents, which agreements and documents shall be in form and substance reasonably satisfactory to the Purchaser: (i) an instrument pursuant to which such Subsidiary shall become a guarantor of the Obligations, (ii) a security agreement (it being acknowledged and agreed that such security agreement shall be substantially in the same form as the Appia Security Agreement); (iii) any and all UCC financing statements which the Purchaser deems necessary and appropriate in order to

implement such security agreement; and (iv) such other agreements, documents, financing statements, instruments, opinions and certificates and completion of such other matters, as the Purchaser may reasonably deem necessary or appropriate to assure that the Purchaser shall have the same rights and protections with respect to such Subsidiary as they have with respect to the Companies and that such Subsidiary shall be subject to the same obligations under the Financing Documents as the Companies.

- **8.14** Restricted Payments. Without the consent of the Purchaser, Appia shall not, and Appia shall not permit any of its Subsidiaries to, declare or pay any dividend on, or make any payment on account of, or set apart assets for a sinking or other analogous fund for, the purchase, redemption, defeasance, retirement or other acquisition of, any shares of any class of Capital Stock of Appia or any or its Subsidiaries or any warrants or options to purchase any such shares of Capital Stock, whether now or hereafter outstanding, or make any other distribution in respect thereof, either directly or indirectly, whether in cash or property or in obligations of either Company or any of its Subsidiaries, in each case other than (i) repurchases of stock from former employees or consultants pursuant to the terms of a previously existing written agreement with such employees and consultants who performed services for Appia or any Subsidiary in connection with the cessation of such employment or service at no greater than the original purchase price, provided that no such repurchase can occur without the Purchaser's consent if an Event of Default has occurred, is continuing or would exist after giving effect to the repurchase, (ii) wholly-owned Subsidiaries declaring and paying dividends to Appia or other wholly- owned Subsidiaries, and (iii) Appia declaring and paying dividends to Mandalay.
- **8.15** <u>Limitation on Investments, Loans and Advances</u>. Without the consent of the Purchaser, Appia shall not, and Appia shall not permit any of its Subsidiaries to, make any advance, loan, extension of credit, or capital contribution to, or purchase any stock, bonds, notes, debentures or other securities of, or any assets constituting a business unit of, or make any other investment in, any Person (an "Investment"), except:
 - (a) investments in cash, federally insured deposit accounts and Cash Equivalents;
 - (b) securities held by Appia or any of its Subsidiaries prior to the Closing Date and listed on Schedule 8.15;
 - (c) Investments by Appia in any wholly-owned Subsidiary and Investments by any such wholly-owned Subsidiary in Appia or in any other wholly-owned Subsidiary;
 - (d) extensions of trade credit and endorsements of negotiable instruments and other negotiable documents in the ordinary course of business;
 - (e) loans or advances to employees, officers or directors of Appia or any of its Subsidiaries, or guarantee the payment of any such loan granted by a third party, in an aggregate amount for Appia and its Subsidiaries not to exceed Fifty Thousand Dollars (\$50,000) at any time outstanding;
 - (f) Investments (including debt obligations) received in connection with the bankruptcy or reorganization of customers or suppliers and in settlement of delinquent obligations of, and other disputes with, customers or suppliers arising in the ordinary course of a Appia's business;
 - (g) Investments consisting of notes receivable of, or prepaid royalties and other credit extensions, to customers and suppliers who are not Affiliates, in the ordinary course of business, provided that this <u>Section 8.15(g)</u> shall not apply to Investments of Appia in any Subsidiary;

- (h) Additional Investments in an aggregate amount not to exceed One Hundred Thousand and 00/100 Dollars (\$100,000.00) at any time outstanding; and
- (i) Up to One Million and 00/100 Dollars (\$1,000,000) of Investments in the aggregate in one or more Persons who, in connection with such Investments, become direct or indirect wholly-owned Subsidiaries of Appia; provided that no more than Four Hundred Thousand and 00/100 Dollars (\$400,000) of such Investment shall be funded with cash or Cash Equivalents.
- **8.16** <u>Limitation on Transactions with Affiliates</u>. Appia shall not, and Appia shall not permit any of its Subsidiaries to, enter into any transaction (including, without limitation, any purchase, sale, lease or exchange of property or the rendering of any service) with any Affiliate unless such transaction (i) is otherwise permitted under this Agreement; or (ii)(A) is in the ordinary course of Appia's or such Subsidiary's business or approved by a majority of the Appia Board who are disinterested in such transaction; and (B) is upon fair and reasonable terms no less favorable to Appia or such Subsidiary, as the case may be, than it would obtain in a comparable arm's length transaction with a Person which is not an Affiliate; <u>provided, however</u>, that the foregoing restriction shall not prohibit any employment agreement entered into by Appia or any of its Subsidiaries in the ordinary course of business; any issuance of securities in connection with employment arrangements, stock options and stock ownership plans of Appia entered into in the ordinary course of business or participation in future equity financings of Appia. Nothing in this Section 8.16 prohibits any transaction otherwise permitted in this Agreement.
- **8.17** Senior Debt. Without the consent of the Purchaser, Appia shall not increase the aggregate amount of any debt permitted by Section 8.1(a) above to an amount which at any time and in the aggregate exceeds the sum of Six Million and 00/100 Dollars (\$6,000,000.00), and of which Senior Debt shall not exceed Five Million Dollars (\$5,000,000) and Purchase Money Indebtedness shall not exceed One Million Dollars (\$1,000,000).
- **8.18** Redemption Rights. Without the consent of the Purchaser, other than with respect to Mandalay, Appia shall not grant Redemption Rights to any Stockholder or permit the exercise of any Redemption Rights by any stockholder which would permit the exercise of such Redemption Rights prior to the payment in full of the New Debenture.
 - 8.19 Notices. Appia shall, and shall cause each of its Subsidiaries to, give prompt notice to the Purchaser of:
 - (a) the following events, as soon as practicable and in any event within thirty (30) days after Appia has knowledge thereof: (i) the occurrence or expected occurrence of any Reportable Event with respect to any Plan, a failure to make any required contribution to a Plan, the creation of any Lien in favor of the PBGC or a Plan or (ii) the institution of proceedings or the taking of any other action by the PBGC or Appia or any Commonly Controlled Entity with respect to the withdrawal from, or the termination, reorganization or insolvency of, any Plan;
 - (b) the occurrence of any Default or Event of Default of which Appia has knowledge; or
 - (c) any development or event which would reasonably be expected to have a Material Adverse Effect.

Any notice under this <u>Section 8.19</u> shall be accompanied by a statement of a Responsible Officer setting forth details of the occurrence referred to therein and stating what action Appia proposes to take with respect thereto but shall not be required to include any privileged information.

8.20 <u>Limitation on Changes in Fiscal Year</u>. Appia shall not, and Appia shall not permit any of its Subsidiaries to, change the fiscal year of Appia or any of its Subsidiaries other than one time to conform their respective fiscal years with that of Mandalay.

SECTION 9. EVENTS OF DEFAULT.

- 9.1 Events of Default; Acceleration. If any of the following events ("Events of Default") shall occur:
- (a) Appia shall fail to pay any principal amount of the New Debenture or fees, charges, costs or expenses (other than an Interest Default permitted by Section 2.5(a)) due and payable under the New Debenture or any of the Financing Documents when such payment comes due in accordance with the terms of this Agreement and the other Financing Documents; or Appia shall fail to pay any other amount payable hereunder when due (other than an Interest Default permitted by Section 2.5(a)); or
- (b) any representation or warranty made by a Company or any other Obligor herein or in any other Financing Document or which is contained in any certificate, document, financial or other statement furnished by it at any time under or in connection with this Agreement or any such other Financing Document shall prove, when taken as a whole, to have been incorrect in any material respect on or as of the date made; or
- (c) the failure by Appia or any other Obligor to punctually perform, observe, comply with or satisfy (i) any covenant, agreement or condition contained in <u>Sections 7 or 8</u> of this Agreement, or (ii) any covenant, agreement or condition contained in any Financing Document, subject to any applicable cure period set forth therein; or
- (d) Appia or any Subsidiary shall be in default in the observance or performance of any other covenant contained in this Agreement or any other Financing Document (other than as provided in paragraphs (a) through (c) of this Section 9.1), and such default (if remediable) shall continue unremedied for a period of twenty (20) days after the earlier of (i) the date on which a Responsible Officer of such Company first learns of such default or (ii) the date on which written notice thereof shall have been given to such Company by the Purchaser; or
- (e) Appia or any Subsidiary shall fail to pay when due or shall fail to observe or perform any term, covenant or agreement evidencing or securing any Indebtedness of such Company which, together with all such other due but unpaid Indebtedness, exceeds the sum of Two Hundred Thousand and 00/100 Dollars (\$200,000.00), which results in the proper acceleration of such Indebtedness or the proper declaration of an event of default under any agreement relating to such Indebtedness; or
- (f) a Company or any Subsidiary, (i) shall make an assignment for the benefit of creditors; or (ii) shall admit in writing its inability to pay its debts as they become due, or its inability to pay or perform under the Financing Documents; or (iii) shall file a voluntary petition in bankruptcy; or (iv) shall file any petition, answer, or document seeking for itself any reorganization, arrangement, composition, readjustment, liquidation, dissolution or similar relief under any present or future statute, law or regulation pertinent to such circumstances; or (v) shall

seek or consent to or acquiesce in the appointment of any trustee, receiver, or liquidator of such Company or of all or any substantial part (i.e., 33-1/3% or more) of the assets or property of such Company; or (vi) other than as permitted herein, shall cease operations of its business as its business has normally been conducted (which consists of advertising, marketing and sales within the mobile industry) or is currently proposed to be conducted, or terminate substantially all of its employees; or (vii) such Company's directors or majority shareholders shall take any action initiating any of the foregoing actions described in clauses (i) through (vi); or

- (g) either (i) forty-five (45) days shall have expired after the commencement of an involuntary action against a Company or any Subsidiary seeking reorganization, arrangement, composition, readjustment, liquidation, dissolution or similar relief under any present or future statute, law or regulation, without such action being dismissed or all orders or proceedings thereunder affecting the operations or the business of such Company or any Subsidiary being stayed; or (ii) a stay of any such order or proceedings shall thereafter be set aside and the action setting it aside shall not be timely appealed; or (iii) such Company or any Subsidiary shall file any answer admitting or not contesting the material allegations of a petition filed against such Company or any Subsidiary in any such proceedings; or (iv) the court in which such proceedings are pending shall enter a decree or order granting the relief sought in any such proceedings; or (v) twenty (20) days shall have expired after the appointment, without the consent or acquiescence of a Company or any Subsidiary, of any trustee, receiver or liquidator of such Company or any Subsidiary or of all or any substantial part of the properties of such Company or any Subsidiary without such appointment being vacated; or
- (h) (i) Appia or any Commonly Controlled Entity shall fail to pay when due any amount that it shall have become liable to pay to the PBGC or to a Plan under Title IV of ERISA, unless (A) such liability is being contested in good faith by appropriate proceedings, such Company or such Commonly Controlled Entity, as the case may be, has established and is maintaining adequate reserves in accordance with GAAP and no lien shall have been filed to secure such liability or (B) which would not have a Material Adverse Effect; or (ii) the PBGC shall institute proceedings under Title IV of ERISA to terminate or to cause a trustee to be appointed to administer any such Plan or Plans; or (iii) a condition shall exist by reason of which the PBGC would be entitled to obtain a decree adjudicating that any such Plan or Plans must be terminated; or
- (i) one or more final, nonappealable judgments or decrees shall be entered against Appia or any of its Subsidiaries involving individually monetary damages of Two Hundred Fifty and 00/100 Dollars (\$250,000.00) (in excess of what is paid or covered by insurance) or in the aggregate, monetary damages of Two Hundred Fifty Thousand and 00/100 Dollars (\$250,000.00) (in excess of what is paid or covered by insurance) or more, and all such judgments or decrees shall not have been vacated, discharged, stayed or bonded pending appeal within thirty (30) days from the entry thereof; or
- (j) if any of the Financing Documents or the Warrant (or any provision contained therein) shall be cancelled, terminated, revoked, curtailed or rescinded otherwise than in accordance with the terms thereof or with the express prior written agreement, consent or approval of the Purchaser, or any action at law, suit or in equity or other legal proceeding be commenced by or on behalf of a Company or any of its officers of members of its Board of Directors and results in, or is reasonably likely to result in, a finding, order, decree or judgment which does or would cancel, revoke, curtail or rescind any of the Financing Documents or the Warrant, or any Governmental Authority of competent jurisdiction shall make a determination that, or issue a judgment, order, decree or ruling to the effect that, any of the Financing Documents or the Warrant (or any provision contained therein) is illegal, invalid or unenforceable in accordance with the terms thereof; or

- (k) any Lien created by any of the Security Documents shall, by reason of any breach by any Obligor thereto of any of its covenants or other obligations contained in such Security Documents, cease to be enforceable and of the same effect and priority purported to be created thereby; or
- (l) a material portion of the property of a Company or any of its Subsidiaries is damaged by fire or other casualty, or otherwise lost or stolen, the restoration or replacement cost of which property exceeds the amount of insurance proceeds readily available for such restoration or replacement; or
- (m) any default shall exist and remains unwaived, unforborn or uncured with respect to any of the Senior Debt or Replacement Senior Debt if, as a result of such default, any holder of the Senior Debt or Replacement Senior Debt, is entitled and elects to cause any such Senior Debt or Replacement Senior Debt to become due prior to its stated date of maturity; or
- (n) any payment which a Company knew or should have known was be made in violation of any subordination agreement entered into between the Purchaser and another holder of Company Indebtedness or Company Subsidiary Indebtedness; or
- (o) (i) any guaranty of any Obligations terminates or ceases for any reason to be in full force and effect; (ii) Mandalay does not perform any obligation or covenant under the Mandalay Guaranty; (iii) any circumstance described in Sections 9.1(b), (f), (g), and (i) hereinabove with respect to (i.e., as and as-if applied to) Mandalay (provided that with respect to Mandalay, the applicable threshold under Section 9.1(i) shall be Five Hundred Thousand Dollars (\$500,000)); or (iv) the liquidation, winding up, or termination of existence of Mandalay.

then, and in any such event, so long as the same may be continuing, the Purchaser may, by notice to the Companies, declare all amounts owing with respect to this Agreement, the New Debenture, and the other Financing Documents to be, and they shall thereupon forthwith become, immediately due and payable without presentment, demand, protest or other notice of any kind, all of which are hereby expressly waived by the Companies; provided, however, that in the event of any Event of Default specified in Section 9.1(f), Section 9.1(g) or Section 9.1(h), all such amounts shall become immediately due and payable automatically and without any requirement of notice from the Purchaser.

9.2 Effect of Schedules and Loan Agreement Representations. For the avoidance of doubt and notwithstanding the consummation of the Closing by the Purchaser or anything else contained herein or in the Financing Documents: (i) no circumstance, condition, subject, item, occurrence or other matter which constitutes an Event of Default under this Agreement or the other Financing Documents shall be deemed waived by virtue of it having been set forth on the Master Disclosure Schedule or any other disclosure schedule delivered in connection with this Agreement or the other Financing Documents; and (ii) whether an Event of Default has occurred shall be determined without regard to any disclosure set forth on the Master Disclosure Schedule or any other disclosure schedule delivered in connection with this Agreement or the other Financing Documents (but any circumstance, condition, subject, item, occurrence or other matter disclosed therein will be acknowledged by Borrower to have occurred for purposes of identifying an Event of Default).

SECTION 10. RIGHTS AND REMEDIES.

- **10.1** Rights of Purchaser. Upon the occurrence and during the continuance of any one or more of the Events of Default, the Purchaser may proceed in accordance with applicable law to enforce the legal or equitable rights or remedies as it may have under this Agreement or any other Financing Documents or otherwise by virtue of applicable law.
- 10.2 Setoff and Adjustments. In addition to any rights and remedies of the Purchaser provided by law, the Purchaser shall have the right, without prior notice to either Company, any such prior notice being expressly waived by each Company to the extent permitted by applicable law, upon any amount becoming due and payable by a Company hereunder (whether at the stated maturity, by acceleration or otherwise) to set-off and appropriate and apply against such amount then due and payable any and all deposits (general or special, time or demand, provisional or final), in any currency, and any other credits, indebtedness or claims, in any currency, in each case whether direct or indirect, absolute or contingent, at any time held or owing by the Purchaser or any branch or agency thereof to or for the credit or the account of a Company. The Purchaser agrees to promptly notify the Companies after any such set-off and application made by the Purchaser, provided that the failure to give such notice shall not affect the validity of such set-off and application.
- 10.3 No Waiver; Cumulative Remedies. No failure to exercise and no delay in exercising on the part of the Purchaser, any right, remedy, power or privilege hereunder or under the other Financing Documents shall operate as a waiver thereof; nor shall any single or partial exercise of any right, remedy, power or privilege hereunder preclude any other or further exercise thereof or the exercise of any other right, remedy, power or privilege. The rights, remedies, powers and privileges herein provided are cumulative and not exclusive of any rights, remedies, powers and privileges provided by law.
- **10.4** <u>Distribution of Collateral Proceeds</u>. In the event that, following the occurrence and during the continuance of any Event of Default, the Purchaser receives any monies in connection with the enforcement of any of the Security Documents, or otherwise with respect to the realization upon any of the Collateral, such monies shall be distributed for application as follows: (i) *first*, to the Obligations in such order or preference as the Purchaser may determine; (ii) *second*, upon payment and satisfaction in full or other provisions for payment in full satisfaction of all of the Obligations, to the payment of any obligations required to be paid pursuant to §9-615 of the UCC; and then (c) *third*, the excess, if any, shall be returned to the applicable Company or to such other Persons as are entitled thereto.

SECTION 11. MISCELLANEOUS.

- 11.1 Amendments and Waivers. Neither this Agreement nor any other Financing Document or the Warrant, nor any terms hereof or thereof may be amended, supplemented or modified except in a writing signed by the Parties. No provision of this Agreement or of any other Financing Document or of the Warrant, or any terms hereof or thereof, may be waived except in a writing signed by the party waiving its rights. In the case of any waiver, the Parties shall be restored to their former positions and rights hereunder and under the other Financing Documents, and any Default or Event of Default waived shall be deemed to be cured and not continuing; no such waiver shall extend to any subsequent or other Default or Event of Default or impair any right consequent thereon.
- 11.2 <u>Survival of Covenants</u>. Except for those which by their terms survive termination of the Financing Documents or for another period of time, all agreements, representations, covenants and warranties made by a Company in the Financing Documents shall remain in full force and effect until all Obligations (other than inchoate indemnity obligations) have been paid in full and satisfied, notwithstanding the fact that the New Debenture may, from time to time, be in a zero or credit position.

- 11.3 Prior Discussions; Counterparts. The Financing Documents and the Warrant incorporate all discussions and negotiations between the Parties and either express or implied, concerning the Obligations, notwithstanding any custom, usage or oral agreement or understanding to the contrary. This Agreement may be executed in two or more counterparts, each of which shall constitute an original, but such counterparts together shall constitute one and the same instrument. Any proof of this Agreement shall require production of only one such counterpart.
- 11.4 <u>Severability.</u> Any provision of this Agreement which is prohibited or unenforceable in any jurisdiction shall, as to such jurisdiction, be ineffective to the extent of such prohibition or unenforceability without invalidating the remaining provisions hereof, and any such prohibition or unenforceability in any jurisdiction shall not invalidate or render unenforceable such provision in any other jurisdiction.
- 11.5 Notices. All notices, requests and demands to or upon the Parties to be effective shall be in writing (including by facsimile transmission), and, unless otherwise expressly provided herein, shall be deemed to have been duly given when delivered by hand, or when sent during normal business hours (or on the following Business Day, if sent after normal business hours) by facsimile transmission or by telex, answer back received, or on the first Business Day after delivery to any overnight delivery service, freight prepaid, or three (3) Business Days after being sent by certified mail, return receipt requested, postage prepaid, and addressed in the case of the Parties as follows or to such other address as may be hereafter notified by the respective parties hereto:

(a)	If to Mandalay,		
	then:	[] [] [] Attention: [] Facsimile: [] Telephone: []	
	with a copy to:	[] [] [] [] Facsimile: [] Telephone: []	
(b)	If to Appia,		
	then:	[Appia, Inc. 320 Blackwell Street, 4th Floo Durham, NC 27701 Attention: Judson S. Bowman, Facsimile: [] Telephone: 919.251.6144]	
	with a copy to:	[] [] [] Attention: [] Facsimile: [] Telephone: []	
		-26	

(c) If to the Purchaser,

then: North Atlantic SBIC IV, L.P.

c/o North Atlantic Capital Corporation

Two City Center Portland, Maine 04101

Attention: David M. Coit, Managing Director

Facsimile: 207.772.3257 Telephone: 207.772.4470

with a copy to: Nixon Peabody LLP

100 Summer Street

Boston, Massachusetts 02110 Attention: David A. Martland, Esq.

Facsimile: 866.368.6632 Telephone: 617.345.6145

11.6 Expenses.

(a) Mandalay agrees to pay all reasonable and documented actual out-of-pocket expenses, including: (i) the reasonable costs of producing and reproducing this Agreement, the other Financing Documents, the Warrant, any subordination agreement entered into pursuant hereto, and the other agreements and instruments mentioned herein, (ii) the reasonable fees, expenses and disbursements of counsel to the Purchaser incurred in connection with the preparation and negotiation of the Financing Documents, the Warrant, any subordination agreement entered into pursuant hereto, and other instruments mentioned herein, (iii) all reasonable fees, expenses and disbursements of the Purchaser incurred in connection with UCC searches, in a cumulative amount not to exceed Seventy-Five Thousand and 00/100 Dollars (\$75,000.00) in the aggregate without written notice to Mandalay (the "Closing Expenses"). At Closing, Mandalay shall pay the Closing Expenses. The Purchaser may deduct the Closing Expenses directly against the purchase price paid at the Closing, and Appia shall treat the full amount of such payment as a payment for the New Debenture at Closing.

(b) In addition, the Companies agree to pay all reasonable out-of-pocket expenses (including without limitation reasonable attorneys' fees and costs and reasonable accounting and similar professional fees and charges) incurred by the Purchaser in connection with the enforcement of or preservation of rights under any of the Financing Documents against a Company or any Subsidiary (including without limitation any expenses incurred relating to entering into a subordination agreement for the benefit of any future holder of the Senior Debt or Replacement Senior Debt discussed in Section 2.2) or the administration thereof after the occurrence of a Default or Event of Default.

11.7 Indemnification.

(a) Third-Party Claims. Each Company agrees to defend, indemnify and hold harmless the Purchaser from and against any and all third-party claims, actions and suits whether groundless or otherwise, and from and against any and all actual liabilities, losses, damages and expenses of every nature and character arising out of this Agreement, any of the other Financing Documents, or the Warrant or the transactions contemplated hereby including, without

limitation, (i) any actual or proposed use by either Company of the proceeds of the New Debenture, (ii) any actual or alleged infringement of any patent, copyright, trademark, service mark, or similar right by either Company, or (iii) with respect to a Company or any other Obligor and their respective properties and assets, the violation of any Environmental Law, the presence, disposal, escape, seepage, leakage, spillage, discharge, emission, release or threatened release of any Hazardous Substances or any action, suit, proceeding or investigation brought or threatened with respect to any Hazardous Substances (including, but not limited to, claims with respect to wrongful death, personal injury, or damage to property), in each case including, without limitation, the reasonable fees and disbursements of counsel incurred in connection with any such investigation, litigation or other proceeding; provided, however, such Company shall not be obligated to indemnify the Purchaser for such third-party claims which have been determined by a court of competent jurisdiction to have arisen out of the Purchaser's actual bad faith, willful misconduct, or gross negligence or solely from the Purchaser's breach of any of its obligations or duties under this Agreement, any of the other Financing Documents, or the Warrant. In litigation, or the preparation therefore, the Purchaser shall be entitled to select one counsel to serve as its own counsel and, in addition to the foregoing indemnity, the Companies agree to pay promptly the reasonable fees and expenses of such counsel.

- (b) Inter-Party Disputes. Each Company agrees to defend, indemnify and hold harmless the Purchaser from and against any and all actual liabilities, obligations, claims, damages, costs, losses and expenses (including court costs and attorney's reasonable fees and expenses), as determined by a court of competent jurisdiction, that the Purchaser sustains or incurs arising out of this Agreement, or the preparation of this Agreement, or any of the Financing Documents or the Warrant, or in collecting or enforcing the Obligations, or in enforcing any of the Purchaser's rights or remedies, or in the prosecution or defense of any action or proceeding concerning any matter arising out of this Agreement, any of the other Financing Documents or the Warrant, or the Obligations, or on account of the Purchaser's relationship with a Company or any other Obligor (except for such claims which have been determined by a court of competent jurisdiction to have arisen out of the Purchaser's actual bad faith, willful misconduct, or gross negligence or solely from Purchaser's breach of any of its obligations or duties under this Agreement, any of the other Financing Documents, or the Warrant).
- (c) <u>General</u>. If, and to the extent that the obligations of a Company under this <u>Section 11.7</u> are unenforceable for any reason, the Companies hereby agree to make the maximum contribution to the payment in satisfaction of such obligations which is permissible under applicable law. The covenants contained in this Section 11.7 shall survive payment or satisfaction in full of all other Obligations.
- 11.8 Acknowledgements. Each Company hereby acknowledges that (i) such Company has been advised by counsel in the negotiation, drafting, execution and delivery of this Agreement, the other Financing Documents, and the Warrant; (ii) the Purchaser has no fiduciary relationship with or fiduciary duty to either Company arising out of or in connection with this Agreement or any of the other Financing Documents or the Warrant; and (iii) no joint venture is created hereby or by the other Financing Documents or the Warrant or otherwise exists by virtue of the transactions contemplated hereby between the Parties.
- 11.9 <u>Successors and Assigns</u>. This Agreement shall be binding upon and inure to the benefit of the Parties and their respective successors and permitted assigns, except that neither Company may assign or transfer any of its rights or obligations under this Agreement without the consent of the Purchaser. The rights of the Purchaser under this Agreement and the other Financing Documents are not

assignable without the written consent of Mandalay, except that assignments by the Purchaser of its rights under this Agreement and the other Financing Documents to its Affiliates shall be permitted without the consent of either Company so long as the Purchaser's assignee agrees in writing to be bound by all of the provisions set forth in this Agreement and the other Financing Documents that were applicable to the Purchaser.

11.10 Loss, Theft, Destruction or Mutilation of the New Debenture. Upon receipt of an affidavit of an officer of the Purchaser, as to the loss, theft, destruction or mutilation of the New Debenture or any other Financing Document or the Warrant which is not of public record, and, in the case of any such loss, theft, destruction or mutilation, upon cancellation of the New Debenture or other Financing Document or the Warrant, the applicable Company shall issue, in lieu thereof, a replacement New Debenture or other Financing Document or the Warrant in the same principal amount thereof and otherwise of like tenor; provided, however, that the Purchaser agrees to defend, indemnify and hold harmless the Companies and their Stockholders, directors and officers from any and all claims, loss or damage whatsoever arising out of the issuance of a replacement New Debenture or other Financing Document or the Warrant pursuant to this Section 11.10.

11.11 Waiver of Jury Trial. TO THE EXTENT PERMITTED BY APPLICABLE LAW, EACH PARTY HEREBY WAIVES ITS RESPECTIVE RIGHTS TO A TRIAL BY JURY OF ANY CLAIM OR CAUSE OF ACTION BASED UPON OR ARISING OUT OF THIS AGREEMENT, ANY OF THE OTHER FINANCING DOCUMENTS OR ANY OF THE TRANSACTIONS CONTEMPLATED HEREIN OR THEREIN, INCLUDING CONTRACT CLAIMS, TORT CLAIMS, BREACH OF DUTY CLAIMS, AND ALL OTHER COMMON LAW OR STATUTORY CLAIMS.

11.12 Governing Law; Consent to Jurisdiction. This Agreement and the other Financing Documents and the Warrant shall be construed in accordance with and governed by, the laws of the State of New York, without giving effect to the conflict of laws principles thereof. Each Party agrees that any suit, action or proceeding against the other arising out of or based upon this Agreement may be instituted in a United States Federal or state court located in the State of Delaware, and any court properly appealable therefrom, and each Party irrevocably submits to the non-exclusive jurisdiction of such courts in any suit, action or proceeding. Each Party irrevocably waives, to the fullest extent permitted by law, any objection to any suit, action or proceeding that may be brought in connection with this Agreement in such courts whether on the grounds of venue, residence or domicile or on the ground that any such suit, action or proceeding has been brought in an inconvenient forum. Each Party agrees that final judgment in any such suit, action or proceeding brought in such court shall be conclusive and binding upon the Parties and may be enforced in any court to the jurisdiction of which they are subject, by a suit upon judgment.

[THE REMAINDER OF THIS PAGE IS LEFT INTENTIONALLY BLANK]

IN WITNESS WHEREOF, the Parties have caused this Securities Purchase Agreement to be duly executed and delivered under seal by their proper and duly authorized officers as of the day and year first above written.

<u>Appia</u> :	<u> </u>
APPIA	, INC.
	Judson S. Bowman Chief Executive Officer
Manda	alay:
MANI	DALAY DIGITAL GROUP, INC.
	[] []
The Pu	ırchaser:
NORT	H ATLANTIC SBIC IV, L.P.
	North Atlantic Investors SBIC IV, LLC General Partner
- 1	David M. Coit Managing Director

[Signature Page to Securities Purchase Agreement]

APPENDIX A

- 1. <u>Definitions</u>. As used in the Agreement, the following terms shall have the following meanings:
 - "Acquisition": as defined in the recitals to this Agreement.
- "Affiliate": means, with respect to any specified Person, any other Person who, directly or indirectly, controls, is controlled by, or is under common control with such Person, including without limitation, any general partner, managing member, officer or director of such Person, or any venture capital fund now or hereafter existing which is controlled by one or more general partners or managing members of, or shares the same management company with, such Person.
 - "Agreement": this Securities Purchase Agreement, as duly amended, supplemented or otherwise modified from time to time.
 - "Appia": as defined in the preamble to this Agreement.
- "Appia Security Agreement": that certain Security Agreement All Assets to be executed by Appia and the Purchaser substantially in the form attached as Exhibit E.
 - "Audited Statements": as defined in Section 5.21(a).
 - "Balance Sheet": as defined in Section 5.21(a).
 - "Business": the business of a Company and its Subsidiary as currently conducted and currently proposed to be conducted.
- "Business Day": any day excluding Saturday, Sunday and any day which is a legal holiday under the laws of the State of Delaware, or is a day on which banking institutions located in the State of Delaware are required or authorized by any Requirement of Law to be closed.
- "Capital Expenditures": as to any Person for any period, the aggregate amount paid or accrued by such Person for the rental, lease, purchase (including by way of the acquisition of securities of a Person), construction or use of any property during such period, the value or cost of which, in accordance with GAAP, would appear on such Person's consolidated balance sheet in the category of property, plant or equipment at the end of such period, excluding (i) any such expenditure in respect of any Replacement Asset and (ii) any such expenditure made to restore, replace or rebuild property to the condition of such property immediately prior to any damage, loss, destruction or condemnation of such property, to the extent such expenditure is made with insurance proceeds or condemnation awards relating to any such damage, loss, destruction or condemnation.
- "Capital Stock": any and all shares, interests, participations or other equivalents (however designated) of capital stock of a corporation, any and all equivalent ownership interests in a Person (other than a corporation) and any and all warrants or options to purchase any of the foregoing.
- "Capitalized Lease": any lease of property, real or personal, the obligations of the lessee in respect of which are required in accordance with GAAP to be capitalized on a balance sheet of the lessee.

"Capitalized Lease Obligations": as to any Person, the obligations of such Person to pay rent or other amounts under any Capitalized Leases; the amount of such obligations at any time shall be the capitalized amount thereof at such time determined in accordance with GAAP.

"Cash Equivalents": (i) securities issued or directly and fully guaranteed or insured by the United States Government, or any agency or instrumentality thereof, having maturities of not more than one year from the date of acquisition, (ii) marketable general obligations issued by any state of the United States of America or any political subdivision of any such state or any public instrumentality thereof maturing within one year from the date of acquisition thereof and, at the time of acquisition thereof, having a credit rating of "A" or better from either Standard & Poor's Ratings Group or Moody's Investors Service, Inc.; (iii) certificates of deposit, time deposits, Eurodollar time deposits, money market accounts, overnight bank deposits or bankers' acceptances having maturities of not more than one year from the date of acquisition thereof of any domestic commercial bank the long-term debt of which is rated at the time of acquisition thereof at least A or the equivalent thereof by Standard & Poor's Ratings Group, or A or the equivalent thereof by Moody's Investors Service, Inc., and having capital and surplus in excess of Five Hundred Million and 00/100 Dollars (\$500,000,000.00), (iv) repurchase obligations with a term of not more than seven days for underlying securities of the types described in clauses (i), (ii) and (iii) entered into with any bank meeting the qualifications specified in clause (iii) above, (v) commercial paper rated at the time of acquisition thereof at least A-2 or the equivalent thereof by Standard & Poor's Ratings Group or P-2 or the equivalent thereof by Moody's Investors Service, Inc., or carrying an equivalent rating by a nationally recognized rating agency, if both of the two named rating agencies cease publishing ratings of investments, and in either case maturing within one year after the date of acquisition thereof and (vi) other investment instruments approved in writing by the Purchaser and offered by any financial institution which has a combined capital and surplus of not less than One Hundred Million and 00/100 Dollars (\$100,000,000.00).

"Closing": as defined in Section 2.1.

"Closing Date": as defined in Section 2.1.

"Closing Expenses": as defined in Section 11.6(a).

"Closing Securities": the New Debenture, the Common Shares, and the Warrant.

"Code": as defined in Section 5.24.

"Collateral": all right, title and interest of the Obligors in and to their assets, now owned or hereinafter acquired, upon which a Lien is purported to be created by any Security Document.

"Commitment": as defined in Section 2.1.

"Common Shares": as defined in Section 3.1.

"Common Stock": as defined in Section 5.2(b).

"Commonly Controlled Entity": an entity, whether or not incorporated, which is under common control with the Company within the meaning of Section 4001 of ERISA or is part of a group which includes the Company and which is treated as a single employer under Section 414(b) or (c) of the Code or, solely for purposes of determining liability under Section 412 of the Code, which is treated as a single employer under Section 414(b), (c), (m) or (o) of the Code.

"Company" and "Companies": as defined in the preamble to this Agreement.

"Consolidated Net Income": for any period, the net income of a Company and its Subsidiaries for such period as determined on a consolidated basis in accordance with GAAP, but excluding from the determination of Consolidated Net Income (without duplication) (i) any extraordinary or non-recurring gains or losses or gains or losses from Asset Sales, (ii) effects of discontinued operations, (iii) the income (or loss) of any Person in which any other Person (other than the Company or any of the Subsidiaries) has a joint interest, except to the extent of the amount of dividends or other distributions actually paid in cash to such Company or any of its Subsidiaries by such Person during such period and (iv) the income (or loss) of any Person accrued prior to the date it becomes a Subsidiary of such Company or is merged into or consolidated with such Company or any of its Subsidiaries or the date such Person's assets are acquired by such Company or any of its Subsidiaries.

"Consolidated Tangible Net Worth": at any date of determination, the sum of the aggregate tangible assets of a Company and its Subsidiaries after having excluded (i) the book value of all Intangible Assets of such Company and its Subsidiaries and (ii) all liabilities of such Company and its Subsidiaries (including all deferred income taxes), all as determined on a consolidated basis in accordance with GAAP consistently applied.

"Contingent Indebtedness": means any assumption, guarantee, endorsement or otherwise becoming directly or contingently liable (including, without limitation, liable by way of agreement, contingent or otherwise to purchase or provide funds for payment, to supply funds to, or otherwise invest in any debtor or otherwise to assure any creditor against any loss) in connection with any Indebtedness of any other Person.

"Contractual Obligation": as to any Person, any provision of any security issued by such Person or of any agreement, instrument or other undertaking to which such Person is a party or by which it or any of its property is bound.

"Default Rate": as defined in Section 2.5(a).

"Defaults": any event which, upon the giving of notice or the lapse of time or both, would constitute an Event of Default.

"Dollars" and "\$": dollars in lawful currency of the United States of America.

"Domain Names": as defined in Section 5.10(a).

"Environmental Law": as defined in Section 5.15.

"ERISA": as defined in Section 5.23.

"Events of Default": as defined in Section 9.1.

"Exchange Act": the Securities Exchange Act of 1934, as amended, or any similar successor federal statute, and the rules and regulations of the Commission thereunder, all as the same shall be in effect at the time.

"Existing Senior Debt": as defined in Section 2.2.

"Financial Statements": as defined in Section 5.21(a).

"Financing Documents": this Agreement, the New Debenture, the Security Documents, and any and all other agreements, guaranties, instruments, documents, certificates, financing statements, powers of attorney, consents and filings, whether heretofore, now, or hereafter executed by or on behalf of a Company, any of its Subsidiaries, or any other Person and delivered to the Purchaser in connection with the New Debenture, all as may be amended, modified, supplemented, restated or extended from time to time.

"First Debenture": as defined in the recitals to this Agreement.

"GAAP": generally accepted accounting principles in the United States of America in effect from time to time.

"Governmental Authority": any nation or government, any state or other political subdivision thereof and any entity exercising executive, legislative, judicial, regulatory or administrative functions of or pertaining to government.

"Indebtedness": of any Person at any date, without duplication, (i) all indebtedness of such Person for borrowed money or for the deferred purchase price of property or services (other than current trade liabilities incurred in the ordinary course of business and payable in accordance with customary practices), including any Purchase Money Indebtedness, (ii) any other indebtedness of such Person which is evidenced by a note, bond, debenture or similar instrument, (iii) all obligations of such Person under Capitalized Leases, (iv) all obligations of such Person in respect of acceptances issued or created for the account of such Person (v) any Contingent Indebtedness and (f) all indebtedness of others of the types described in (i) through (iv) above secured by any Lien on any property owned by such Person even though such Person has not assumed or otherwise become liable for the payment thereof (the amount of such indebtedness with respect to such Person being deemed to be the lesser of the value of such property or the amount of indebtedness of others so secured).

"Interest Default": as defined in Section 2.5(a).

"Intangible Assets": as to any Person, any and all goodwill, organizational expense, licenses, patents, trademarks, trademarks, copyrights, capitalized research and development expenses, deferred charges and all other intangible assets of such Person.

"Intellectual Property Assets": any and all of the following, as they exist throughout the world: (i) patents, patent applications of any kind, patent rights, inventions, discoveries and invention disclosures (whether or not patented) (collectively, "Patents"); (ii) rights in registered and unregistered trademarks, service marks, trade names, trade dress, logos, packaging design, slogans and Internet domain names, and registrations and applications for registration of any of the foregoing (collectively, "Marks"); (iii) copyrights in both published and unpublished works, including without limitation all compilations, databases and computer programs, manuals and other documentation and all copyright registrations and applications, and all derivatives, translations, adaptations and combinations of the above (collectively, "Copyrights"); (iv) rights in know-how, trade secrets, confidential or proprietary information, research in progress, algorithms, data, designs, processes, formulae, drawings, schematics, blueprints, flow charts, models, strategies, prototypes, techniques, Beta testing procedures and Beta testing results (collectively, "Trade Secrets"); (v) Domain Names; (vi) any and all other intellectual property

rights and/or proprietary rights relating to any of the foregoing; and (vii) goodwill, franchises, licenses, permits, consents, approvals, and claims of infringement and misappropriation against third parties.

"Investment": as defined in Section 8.15.

"knowledge": including the phrase "to the knowledge of [a Company]," shall mean the actual knowledge after reasonable investigation of any Senior Executive.

"Licenses In": as defined in Section 5.10(a).

"Licenses Out": as defined in Section 5.10(a).

"Lien": any mortgage, pledge, hypothecation, assignment, deposit arrangement, encumbrance, lien (statutory or other), charge or other security interest or any preference, priority or other security agreement or preferential arrangement of any kind or nature whatsoever (including, without limitation, any conditional sale or other title retention agreement and any Capitalized Lease having substantially the same economic effect as any of the foregoing) but excluding any license, permit or other right to use.

"Liquidity Event": any one or more of the following events: (i) any liquidation, dissolution or winding-up of Appia or Mandalay or of any Subsidiary, whether voluntary or involuntary; (ii) a merger or consolidation in which (A) Appia or Mandalay is a constituent party or (B) a Subsidiary of Appia or Mandalay is a constituent party and Appia or Mandalay issues shares of Capital Stock pursuant to such merger or consolidation, except any such merger or consolidation involving Appia or Mandalay or a Subsidiary of Appia or Mandalay in which the shares of capital stock of Appia or Mandalay outstanding immediately prior to such merger or consolidation continue to represent, or are converted or exchanged for equity interests which represent, immediately following such merger or consolidation, in substantially the same relative proportions more than Fifty Percent (50%) by voting power of the equity interests of (1) the surviving or resulting Person or (2) if the surviving or resulting Person is a wholly owned Subsidiary of another Person immediately following such merger or consolidation, the Person that is the parent entity of such surviving or resulting Person; (iii) the sale of shares of Capital Stock in one or more related transactions as a result of which those Persons who held One Hundred and 00/100 Percent (100.00%) of the shares of voting Capital Stock of Appia or Mandalay immediately prior to such transaction or transactions do not hold more than Fifty and 00/100 Percent (50.00%) of the shares of voting Capital Stock of Appia or Mandalay after giving effect to such transaction or transactions; or (iv) the sale or disposition in one or a series of transactions by Appia or Mandalay of all or substantially all of the assets of Appia or Mandalay or any of their Subsidiaries; or (v) a Liquidity Event within the meaning of the Certificate of Incorporation of Appia or Mandalay.

"Mandalay": as defined in the preamble to this Agreement.

"Mandalay Common Stock": the Common Stock, par value \$0.001 per share, of Mandalay.

"Mandalay Guaranty": means that certain Subordinated Guaranty to be executed by Mandalay in favor of the Purchaser in the form attached as Exhibit F.

"Material Adverse Effect": a material adverse effect or change on (i) the business, operations, property, prospects, or condition (financial or otherwise) of a Company and its Subsidiaries taken as a whole or (ii) the validity or enforceability of this Agreement or any of the other Financing Documents or the Warrant or the rights or remedies of the Purchaser hereunder or thereunder; provided, however, that none of the following shall constitute, or shall be considered in determining whether there has occurred, and no event, circumstance, change or effect resulting from or arising out of any of the following shall constitute, a Material Adverse Effect: (A) changes in the national or world economy or financial markets as a whole or changes in general economic conditions that affect the industries in which the Companies and their Subsidiaries conduct their business, so long as such changes or conditions do not adversely affect the Companies and their Subsidiaries, taken as a whole, in a materially disproportionate manner relative to other similarly situated participants in the industries or markets in which they operate; and (B) any change in applicable Law, rule or regulation or GAAP or interpretation thereof after the Closing Date, so long as such changes do not adversely affect the Companies and their Subsidiaries, taken as a whole, in a materially disproportionate manner relative to other similarly situated participants in the industries or markets in which they operate.

"Material Contract": all written and oral contracts, agreements, deeds, mortgages, leases, subleases, licenses, instruments, notes, commitments, commissions, undertakings, arrangements and understandings to which Appia is a party or under which any of such Appia's properties or assets are subject or bound (i) which by their terms involve, or would reasonably be expected to involve, obligations (contingent or otherwise) of, or aggregate payments by or to Appia of, in excess of Two Hundred Fifty Thousand and 00/100 Dollars (\$250,000.00) per annum, (ii) which are not terminable by Appia upon thirty (30) days' or less notice without penalty or premium, (iii) the breach or termination of which would reasonably be expected to have a Material Adverse Effect on Appia, or (iv) which involve indemnification by Appia with respect to infringements of proprietary rights (other than standard agreements with its customers and channel partners entered into in the ordinary course of business).

"Material Customers": as defined in Section 5.27.

"Material Suppliers": as defined in Section 5.27.

"Maturity Date": the date which is two (2) years after the Closing Date.

"Merger Agreement": as defined in the recitals to this Agreement.

"NAC Observer": as defined in Section 2.10.

"New Debenture": as defined in Section 2.1.

"New Senior Debt": as defined in Section 2.2.

"Obligations": all Indebtedness, obligations and liabilities of a Company or any of its Subsidiaries to the Purchaser, individually or collectively, now existing or hereafter arising, direct or indirect, joint or several, absolute or contingent, matured or unmatured, liquidated or unliquidated, secured or unsecured, arising by contract, operation of law or otherwise, arising or incurred under any of the Financing Documents.

"Obligors": collectively, each Company and any Subsidiary which is now or hereafter becomes a party to any Financing Document.

"Obsolete Property": any property of a Company or any of its Subsidiaries which is obsolete, outdated or worn out or the useful life of which has substantially ended, in each case in the good faith determination of such Company or any applicable Subsidiary.

"Option Plan": as defined in Section 5.2(e).

"Party" and "Parties": as defined in the preamble to this Agreement.

"Payment Account": as defined in Section 2.7(a).

"PBGC": the Pension Benefit Guaranty Corporation established pursuant to Subtitle A of Title IV of ERISA and any Governmental Authority which succeeds to the powers and functions thereof.

"Permitted Liens": as defined in Section 8.3.

"Person": an individual, partnership, corporation, business trust, joint stock company, limited liability company, trust, unincorporated association, joint venture, Governmental Authority, or other entity of whatever nature.

"<u>Plan</u>": at a particular time, any employee benefit plan which is covered by ERISA and in respect of which a Company or a Commonly Controlled Entity is (or, if such plan were terminated at such time, would under Section 4069 of ERISA be deemed to be) an "employer" as defined in Section 3(5) of ERISA.

"Preferred Stock": as defined in Section 5.2(a).

"Prepayment Premium": as defined in Section 2.8(c).

"Products": any and all products and/or services currently researched, designed, developed, manufactured, performed, licensed, sold, distributed and/or otherwise made commercially available by Appia.

"Purchase Money Indebtedness": any Indebtedness incurred by Appia or any of its Subsidiaries, whichever is applicable, in connection with the acquisition by Appia or any of its Subsidiaries, whichever is applicable, of any real or personal property.

"Purchaser": as defined in the preamble to this Agreement.

"Redemption Right": any right to cause Appia to redeem or repurchase any of its shares of Capital Stock.

"Related Party": as defined in Section 5.13.

"Replacement Asset": any property acquired by a Company or any of its Subsidiaries subsequent to the Closing Date which replaces Obsolete Property of the same type and utility as the property acquired.

"Replacement Senior Debt": any senior indebtedness of Appia and of its Subsidiaries, including any replacement or refinancing of the Senior Debt, which, (i) together with all outstanding amounts of the Senior Debt, which, together with the Indebtedness set forth in Section 8.1(a) above, shall not exceed Six Million and 00/100 Dollars (\$6,000,000.00) in the

aggregate and of which such amounts and Senior Debt shall not exceed Five Million Dollars (\$5,000,000) and Purchase Money Indebtedness shall not exceed One Million Dollars (\$1,000,000); and (ii) is on terms, including, without limitation, interest rate, payment terms, representations and warranties, covenants and events of default and remedies, which, taken as a whole, are no more onerous to the Companies or the Purchaser than the Senior Debt as reasonably determined by the Purchaser.

"Reportable Event": any of the events set forth in Section 4043(b) of ERISA, other than those events as to which the thirty day notice period is waived under PBGC Reg. 29 CFR 4043.

"Requirement of Law": as to any Person, the Certificate (or Articles) of Incorporation (or Organization) and Bylaws or other organizational or governing documents of such Person, and any law, treaty, rule or regulation or determination of an arbitrator or a court or other Governmental Authority, in each case applicable to or binding upon such Person or any of its property or to which such Person or any of its property is subject.

"<u>Responsible Officer</u>": as to any Person, the chief executive officer and/or the president of such Person, or with respect to financial matters, the chief financial officer of such Person or, in either case, such other executive officers as may be designated from time to time by such Person in writing to the Purchaser.

"SBA": Small Business Administration of the United States of America, together with any successor thereto.

"SBIC": a Small Business Investment Company, which is a company licensed and subsidized by the SBA to provide equity capital and long-term loans to small businesses.

"SBIC Act": Small Business Investment Company Act, as amended from time to time.

"SBIC Regulations": the SBIC Act and the regulations issued by the SBA from time to time thereunder, codified as Title 13 of the Code of Federal Regulations, including 13 C.F.R.§107.220.

"SEC": Securities and Exchange Commission of the United States of America, together with any successor thereto.

"Second Debenture": as defined in the recitals to this Agreement.

"Securities": collectively, the Closing Securities and the Warrant Shares.

"Securities Act": the Securities Act of 1933, as amended from time to time.

"Security Documents": collectively, the Appia Security Agreement, the Mandalay Guaranty, and all other security agreements, pledge agreements, financing statements, assignments, mortgages, agreements, documents and instruments now or hereafter delivered to the Purchaser granting a Lien on any asset or assets of any Person to secure the Obligations or to secure any guarantee of any such Obligations.

"Senior Debt": all Indebtedness and other obligations of Appia and of its Subsidiaries, contingent or otherwise, which rank pari passu or senior to the Obligations, including the Indebtedness set forth on Schedule 2.2.

"Solvent": when used with respect to any Person, means that, as of any date of determination, (i) the amount of the "present fair saleable value" of the assets of such Person will, as of such date, exceed the amount that will be required to pay all "liabilities of such Person, contingent or otherwise", as of such date (as such quoted terms are determined in accordance with applicable federal and state laws governing determinations of the insolvency of debtors) as such debts become absolute and matured, (ii) such Person will not have, as of such date, an unreasonably small amount of capital with which to conduct its business, and (iii) such Person will be able to pay its debts as they mature, taking into account the timing of and amounts of cash to be received by such Person and the timing of and amounts of cash to be payable on or in respect of indebtedness of such Person; in each case after giving effect to (A) as of the Closing Date the making of the extensions of credit to be made on the Closing Date and to the application of the proceeds of such extensions of credit and (B) on any date after the Closing Date, the making of any extension of credit to be made on such date, and to the application of the proceeds of such extension of credit. For purposes of this definition, (i) "debt" means liability on a "claim", and (ii) "claim" means any (x) right to payment, whether or not such a right is reduced to judgment, liquidated, unliquidated, fixed, contingent, matured, disputed, undisputed, legal equitable, secured or unsecured or (y) right to an equitable remedy for breach of performance if such breach gives rise to a right to payment, whether or not such right to an equitable remedy is reduced to judgment, fixed, contingent, matured or unmatured, disputed, undisputed, secured or unsecured.

"Stockholder" or "Stockholders": a holder or, collectively, the holders of Capital Stock of a Company.

"<u>Subordination Amendment</u>": the Amendment to Subordination Agreement to be executed by the Purchaser and Silicon Valley Bank substantially in the form of <u>Exhibit G</u>, as the same may be amended, supplemented or otherwise modified from time to time.

"Subsidiary": as to any Person, a corporation, partnership or other entity, both foreign and domestic, of which shares of stock or other ownership interests having ordinary voting power (other than stock or such other ownership interests having such power only by reason of the happening of a contingency) to elect a majority of the board of directors or other managers of such corporation, partnership or other entity are at the time owned, or the management of which is otherwise controlled, directly or indirectly through one or more intermediaries, or both, by such Person. Unless otherwise qualified, all references to a "Subsidiary" or to "Subsidiaries" in this Agreement shall refer to a Subsidiary or Subsidiaries of Appia or Mandalay.

"SVB Loan Agreement": that certain Second Amended and Restated Loan and Security Agreement, dated as of the Closing Date, by and between Silicon Valley Bank and Appia.

"<u>Total Senior Debt</u>": Senior Debt or Replacement Senior Debt, Indebtedness with respect to Capitalized Lease Obligations of a Company or any of its Subsidiaries secured by a lien described in <u>Section 8.3(j)</u>, and Purchase Money Indebtedness of a Company or any of its Subsidiaries secured by a lien described in <u>Section 8.3(j)</u>.

"UCC": the Uniform Commercial Code as from time to time in effect in the State of Delaware.

"Warrant": as defined in Section 3.2.

"Warrant Shares": as defined in Section 5A.3.

2. <u>Use of Terms</u>. The use of the singular of terms which are defined in the plural shall mean and refer to any one of them; and pronouns used herein shall be deemed to include the singular and the plural and all genders. The use of the connective "or" is not intended to be exclusive; the term "may not" is intended to be prohibitive and not permissive; use of "includes" and "including" is intended to be interpreted as expansive and amplifying and not as limiting in any way.

The words "hereof", "herein" and "hereunder" and words of similar import when used in this Agreement shall refer to this Agreement as a whole and not to any particular provision of this Agreement.

* * * * *

A-10

APPENDIX B

1.1 Shelf Registration.

- (a) As promptly as reasonably possible, and in any event on or prior to the later of (i) the 15th Business Day after the Closing Date or (ii) the 5th Business Day after the date on which all audited financial statements required to be included in or incorporated by reference into the Registration Statement (as such term is hereinafter defined) are available (in either case, the "Filing Date"), Mandalay shall prepare and file with the SEC a "shelf" Registration Statement covering the resale of all Common Shares and Warrant Shares (collectively, the "Registrable Securities") for an offering to be made on a continuous basis pursuant to Rule 415 (the "Registration Statement"). If for any reason the SEC does not permit all of the Registrable Securities to be included in such Registration Statement, then Mandalay shall prepare and file with the SEC a separate Registration Statement with respect to any such Registrable Securities not included with the initial Registration Statements, as expeditiously as possible, but in no event later than the date which is 30 days after the date on which the SEC shall indicate as being the first date such filing may be made. The Registration Statement shall be on Form S-3; in the event Form S-3 is not available for the registration of the resale of Registrable Securities hereunder, Mandalay shall (i) register the resale of the Registrable Securities on another appropriate form in accordance herewith as the Purchaser may consent and (ii) attempt to register the Registrable Securities on Form S-3 as soon as such form is available, provided that Mandalay shall maintain the effectiveness of the Registration Statements then in effect until such time as a Registration Statement on Form S-3 covering the Registrable Securities has been declared effective by the SEC.
- (b) Mandalay shall use its best efforts to cause the Registration Statement to be declared effective by the SEC as promptly as reasonably possible after the filing thereof, and shall use its best efforts to keep the Registration Statement continuously effective under the Securities Act until the earlier of (i) the third anniversary of the date that the Registration Statement is first declared effective by the SEC (the "Effective Date"), (ii) the date when all Registrable Securities covered by such Registration Statement have been sold publicly, or (iii) the date on which the Registrable Securities are eligible for sale without registration and without volume limitation pursuant to Rule 144 (the "Effectiveness Period"); provided, however, that if the Warrant expires on or before the Vesting Date (as such term is defined in the Warrant) because the Obligations (as such term is defined in the Warrant) have been paid in full, then the Effectiveness Period shall expire no later than the second anniversary of the Effective Date. Mandalay shall notify the Purchaser in writing promptly (and in any event within one Business Day) after receiving notification from the SEC that the Registration Statement has been declared effective.
- (c) As promptly as reasonably possible, and in any event no later than the 7th day after the Registration Statement ceases to be effective pursuant to applicable securities laws due to the passage of time or the occurrence of an event requiring Mandalay to file a post-effective amendment to the Registration Statement, Mandalay shall prepare and file with the SEC a post-effective amendment to the Registration Statement (a "Post-Effective Amendment"). Mandalay shall use its best efforts to cause the Post-Effective Amendment to be declared effective by the SEC as promptly as possible after the filing thereof. Mandalay shall notify the Purchaser in writing promptly (and in any event within one Business Day) after receiving notification from the SEC that the Post-Effective Amendment has been declared effective.
- 1.2 Registration Procedures. In connection with Mandalay's registration obligations hereunder, Mandalay shall:
 - (a) Not less than three days prior to the filing of a Registration Statement or any related

Prospectus or any amendment or supplement thereto (including any document that would be incorporated or deemed to be incorporated therein by reference), Mandalay shall (i) furnish to the Purchaser and Nixon Peabody LLP ("Purchaser Counsel") copies of all such documents proposed to be filed, which documents (other than those incorporated or deemed to be incorporated by reference) will be subject to the review of the Purchaser and Purchaser Counsel, and (ii) cause its officers and directors, counsel and independent certified public accountants to respond to such inquiries as shall be necessary, in the reasonable opinion of Purchaser Counsel, to conduct a reasonable investigation within the meaning of the Securities Act. Mandalay shall not file a Registration Statement or any such Prospectus or any amendments or supplements thereto to which Purchaser shall reasonably object.

- (b) (i) Prepare and file with the SEC such amendments, including Post-Effective Amendments, to each Registration Statement and the Prospectus used in connection therewith as may be necessary to keep the Registration Statement continuously effective for the Effectiveness Period and prepare and file with the SEC such additional Registration Statements in order to register for resale under the Securities Act all of the Registrable Securities; (ii) cause the related Prospectus to be amended or supplemented by any required Prospectus supplement, and as so supplemented or amended to be filed pursuant to Rule 424; (iii) respond as promptly as reasonably possible, and in any event within ten days, to any comments received from the SEC with respect to the Registration Statement or any amendment thereto and as promptly as reasonably possible provide the Purchaser true and complete copies of all correspondence from and to the SEC relating to the Registration Statement; and (iv) comply in all material respects with the provisions of the Securities Act and the Exchange Act with respect to the disposition of all Registrable Securities covered by the Registration Statement during the applicable period in accordance with the intended methods of disposition by the Purchaser thereof set forth in the Registration Statement as so amended or in such Prospectus as so supplemented.
- (c) Notify the Purchaser and Purchaser Counsel as promptly as reasonably possible, and confirm such notice in writing no later than one day thereafter, of any of the following events: (i) the SEC notifies Mandalay whether there will be a "review" of any Registration Statement; (ii) the SEC comments in writing on any Registration Statement (in which case Mandalay shall deliver to the Purchaser and Purchaser Counsel a copy of such comments and of all written responses thereto); (iii) any Registration Statement or any Post-Effective Amendment is declared effective; (iv) the SEC or any other federal or state governmental authority requests any amendment or supplement to any Registration Statement or Prospectus or requests additional information related thereto; (v) the SEC issues any stop order suspending the effectiveness of any Registration Statement or initiates any proceedings for that purpose; (vi) Mandalay receives notice of any suspension of the qualification or exemption from qualification of any Registrable Securities for sale in any jurisdiction, or the initiation or threat of any proceeding for such purpose; or (vii) the financial statements included or incorporated by reference in any Registration Statement become ineligible for inclusion or incorporation therein or any statement made in any Registration Statement or Prospectus or any document incorporated or deemed to be incorporated therein by reference is untrue in any material respect or any revision to a Registration Statement, Prospectus or other document is required so that it will not contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary to make the statements therein, in the light of the circumstances under which they were made, not misleading.
- (d) Use its best efforts to avoid the issuance of or, if issued, obtain the withdrawal of (i) any order suspending the effectiveness of any Registration Statement, or (ii) any suspension of the qualification (or exemption from qualification) of any of the Registrable Securities for sale in any jurisdiction, as soon as possible.
- (e) Furnish to the Purchaser and Purchaser Counsel, without charge, at least one conformed copy of each Registration Statement and each amendment thereto, including financial statements and

schedules, all documents incorporated or deemed to be incorporated therein by reference, and all exhibits (including those previously furnished or incorporated by reference) promptly after the filing of such documents with the SEC.

- (f) Promptly deliver to the Purchaser and Purchaser Counsel, without charge, as many copies of the Prospectus or Prospectuses (including each form of prospectus) and each amendment or supplement thereto as they may reasonably request. Mandalay hereby consents to the use of such Prospectus and each amendment or supplement thereto by each of the Purchaser in connection with the offering and sale of the Registrable Securities covered by such Prospectus and any amendment or supplement thereto.
- (g) (i) In the time and manner required by each Trading Market, prepare and file with such Trading Market an additional shares listing application covering all of the Registrable Securities; (ii) take all steps necessary to cause such Registrable Securities to be approved for listing on each Trading Market as soon as possible thereafter; (iii) provide to the Purchaser evidence of such listing; and (iv) maintain the listing of such Registrable Securities on each such Trading Market or another Eligible Market.
- (h) Prior to any public offering of Registrable Securities, use its best efforts to register or qualify or cooperate with the Purchaser and Purchaser Counsel in connection with the registration or qualification (or exemption from such registration or qualification) of such Registrable Securities for offer and sale under the securities or Blue Sky laws of such jurisdictions within the United States as any Purchaser requests in writing, to keep each such registration or qualification (or exemption therefrom) effective during the Effectiveness Period and to do any and all other acts or things necessary or advisable to enable the disposition in such jurisdictions of the Registrable Securities covered by a Registration Statement.
- (i) Cooperate with the Purchaser to facilitate the timely preparation and delivery of certificates representing Registrable Securities to be delivered to a transferee pursuant to a Registration Statement, which certificates shall be free, to the extent permitted by this Agreement, of all restrictive legends, and to enable such Registrable Securities to be in such denominations and registered in such names as any the Purchaser may request.
- (j) Upon the occurrence of any event described in Section 1.2(c)(vii), as promptly as reasonably possible, prepare a supplement or amendment, including a post-effective amendment, to the Registration Statement or a supplement to the related Prospectus or any document incorporated or deemed to be incorporated therein by reference, and file any other required document so that, as thereafter delivered, neither the Registration Statement nor such Prospectus will contain an untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary to make the statements therein, in the light of the circumstances under which they were made, not misleading.
- (k) Cooperate with any due diligence investigation undertaken by the Purchaser in connection with the sale of Registrable Securities, including, without limitation, by making available any documents and information; provided that Mandalay will not deliver or make available to any Purchaser material, nonpublic information unless the Purchaser specifically requests in advance to receive material, nonpublic information in writing.
 - (1) Comply with all applicable rules and regulations of the SEC.

1.3 Registration Expenses.

Mandalay shall pay (or reimburse the Purchaser for) all fees and expenses incident to the performance of or compliance with this Agreement by Mandalay, including without limitation (a) all registration and filing fees and expenses, including without limitation those related to filings with the

SEC, any Trading Market and in connection with applicable state securities or Blue Sky laws, (b) printing expenses (including without limitation expenses of printing certificates for Registrable Securities and of printing prospectuses requested by the Purchaser), (c) messenger, telephone and delivery expenses, (d) fees and disbursements of counsel for Mandalay and up to \$15,000 for the Purchaser Counsel, (e) fees and expenses of all other Persons retained by Mandalay in connection with the consummation of the transactions contemplated by this Agreement, and (f) all listing fees to be paid by Mandalay to the Trading Market.

1.4 Indemnification.

(a) Indemnification by Mandalay. Mandalay shall, notwithstanding any termination of this Agreement, indemnify and hold harmless the Purchaser, the officers, directors, partners, members, agents, brokers (including brokers who offer and sell Registrable Securities as principal as a result of a pledge or any failure to perform under a margin call of Common Stock), investment advisors and employees of each of them, each Person who controls the Purchaser (within the meaning of Section 15 of the Securities Act or Section 20 of the Exchange Act) and the officers, directors, partners, members, agents and employees of each such controlling Person, to the fullest extent permitted by applicable law, from and against any and all Losses, as incurred, arising out of or relating to any untrue or alleged untrue statement of a material fact contained in the Registration Statement, any Prospectus or any form of prospectus or in any amendment or supplement thereto or in any preliminary prospectus, or arising out of or relating to any omission or alleged omission of a material fact required to be stated therein or necessary to make the statements therein (in the case of any Prospectus or form of prospectus or supplement thereto, in the light of the circumstances under which they were made) not misleading, except to the extent, but only to the extent, that (i) such untrue statements, alleged untrue statements, omissions or alleged omissions are based solely upon information regarding the Purchaser furnished in writing to Mandalay by the Purchaser expressly for use therein, or to the extent that such information relates to the Purchaser or the Purchaser's proposed method of distribution of Registrable Securities and was reviewed and expressly approved in writing by the Purchaser expressly for use in the Registration Statement, such Prospectus or such form of Prospectus or in any amendment or supplement thereto or (ii) in the case of an occurrence of an event of the type specified in Section 1.2(c)(v)-(vii), the use by the Purchaser of an outdated or defective Prospectus after Mandalay has notified the Purchaser in writing that the Prospectus is outdated or defective and prior to the receipt by the Purchaser of the Advice contemplated in Section 1.5.

(b) *Indemnification by Purchaser*. The Purchaser shall indemnify and hold harmless Mandalay, its directors, officers, agents and employees, each Person who controls Mandalay (within the meaning of Section 15 of the Securities Act and Section 20 of the Exchange Act), and the directors, officers, agents or employees of such controlling Persons, to the fullest extent permitted by applicable law, from and against all Losses (as determined by a court of competent jurisdiction in a final judgment not subject to appeal or review) arising solely out of any untrue statement of a material fact contained in the Registration Statement, any Prospectus, or any form of prospectus, or in any amendment or supplement thereto, or arising solely out of any omission of a material fact required to be stated therein or necessary to make the statements therein (in the case of any Prospectus or form of prospectus or supplement thereto, in the light of the circumstances under which they were made) not misleading to the extent, but only to the extent, that such untrue statement or omission is contained in any information so furnished in writing by the Purchaser to Mandalay specifically for inclusion in such Registration Statement or such Prospectus or to the extent that (i) such untrue statements or omissions are based solely upon information regarding the Purchaser furnished in writing to Mandalay by the Purchaser expressly for use therein, or to the extent that such information relates to the Purchaser or the Purchaser's proposed method of distribution of Registrable Securities and was reviewed and expressly approved in writing by the Purchaser expressly for use in the Registration Statement, such Prospectus or such form of Prospectus or in any amendment or supplement thereto or (ii) in the case of an occurrence of an event of the type specified in Section 1.2(c)(v)-(vii),

the use by the Purchaser of an outdated or defective Prospectus after Mandalay has notified the Purchaser in writing that the Prospectus is outdated or defective and prior to the receipt by the Purchaser of the Advice contemplated in Section 1.5. In no event shall the liability of any selling Purchaser hereunder be greater in amount than the dollar amount of the net proceeds received by the Purchaser upon the sale of the Registrable Securities giving rise to such indemnification obligation.

(c) Conduct of Indemnification Proceedings. If any Proceeding shall be brought or asserted against any Person entitled to indemnity hereunder (an "Indemnified Party"), such Indemnified Party shall promptly notify the Person from whom indemnity is sought (the "Indemnifying Party") in writing, and the Indemnifying Party shall assume the defense thereof, including the employment of counsel reasonably satisfactory to the Indemnified Party and the payment of all fees and expenses incurred in connection with defense thereof; provided, that the failure of any Indemnified Party to give such notice shall not relieve the Indemnifying Party of its obligations or liabilities pursuant to this Agreement, except (and only) to the extent that it shall be finally determined by a court of competent jurisdiction (which determination is not subject to appeal or further review) that such failure shall have proximately and materially adversely prejudiced the Indemnifying Party.

An Indemnified Party shall have the right to employ separate counsel in any such Proceeding and to participate in the defense thereof, but the fees and expenses of such counsel shall be at the expense of such Indemnified Party or Parties unless: (i) the Indemnifying Party has agreed in writing to pay such fees and expenses; or (ii) the Indemnifying Party shall have failed promptly to assume the defense of such Proceeding and to employ counsel reasonably satisfactory to such Indemnified Party in any such Proceeding; or (iii) the named parties to any such Proceeding (including any impleaded parties) include both such Indemnified Party and the Indemnifying Party, and such Indemnified Party shall have been advised by counsel that a conflict of interest is likely to exist if the same counsel were to represent such Indemnified Party and the Indemnifying Party (in which case, if such Indemnified Party notifies the Indemnifying Party in writing that it elects to employ separate counsel at the expense of the Indemnifying Party, the Indemnifying Party shall not have the right to assume the defense thereof and such counsel shall be at the expense of the Indemnifying Party). The Indemnifying Party shall not be liable for any settlement of any such Proceeding effected without its written consent, which consent shall not be unreasonably withheld. No Indemnifying Party shall, without the prior written consent of the Indemnified Party, effect any settlement of any pending Proceeding in respect of which any Indemnified Party is a party, unless such settlement includes an unconditional release of such Indemnified Party from all liability on claims that are the subject matter of such Proceeding.

All fees and expenses of the Indemnified Party (including reasonable fees and expenses to the extent incurred in connection with investigating or preparing to defend such Proceeding in a manner not inconsistent with this Section 1.4) shall be paid to the Indemnified Party, as incurred, within ten Trading Days of written notice thereof to the Indemnifying Party (regardless of whether it is ultimately determined that an Indemnified Party is not entitled to indemnification hereunder; provided, that the Indemnifying Party may require such Indemnified Party to undertake to reimburse all such fees and expenses to the extent it is finally judicially determined that such Indemnified Party is not entitled to indemnification hereunder).

(d) *Contribution*. If a claim for indemnification under Section 1.4(a) or (b) is unavailable to an Indemnified Party (by reason of public policy or otherwise), then each Indemnifying Party, in lieu of indemnifying such Indemnified Party, shall contribute to the amount paid or payable by such Indemnified Party as a result of such Losses, in such proportion as is appropriate to reflect the relative fault of the Indemnifying Party and Indemnified Party in connection with the actions, statements or omissions that resulted in such Losses as well as any other relevant equitable considerations. The relative fault of such Indemnifying Party and Indemnified Party shall be determined by reference to, among other things,

whether any action in question, including any untrue or alleged untrue statement of a material fact or omission or alleged omission of a material fact, has been taken or made by, or relates to information supplied by, such Indemnifying Party or Indemnified Party, and the parties' relative intent, knowledge, access to information and opportunity to correct or prevent such action, statement or omission. The amount paid or payable by a party as a result of any Losses shall be deemed to include, subject to the limitations set forth in Section 1.4(c), any reasonable attorneys' or other reasonable fees or expenses incurred by such party in connection with any Proceeding to the extent such party would have been indemnified for such fees or expenses if the indemnification provided for in this Section 1.4 was available to such party in accordance with its terms

The parties hereto agree that it would not be just and equitable if contribution pursuant to this Section 1.4(d) were determined by pro rata allocation or by any other method of allocation that does not take into account the equitable considerations referred to in the immediately preceding paragraph. Notwithstanding the provisions of this Section 1.4(d), no Purchaser shall be required to contribute, in the aggregate, any amount in excess of the amount by which the net proceeds actually received by the Purchaser from the sale of the Registrable Securities subject to the Proceeding exceeds the amount of any damages that the Purchaser has otherwise been required to pay by reason of such untrue or alleged untrue statement or omission or alleged omission. No Person guilty of fraudulent misrepresentation (within the meaning of Section 11(f) of the Securities Act) shall be entitled to contribution from any Person who was not guilty of such fraudulent misrepresentation.

The indemnity and contribution agreements contained in this Section 1.4 are in addition to any liability that the Indemnifying Parties may have to the Indemnified Parties.

1.5 Dispositions.

Each Purchaser agrees that it will comply with the prospectus delivery requirements of the Securities Act as applicable to it in connection with sales of Registrable Securities pursuant to the Registration Statement. Each Purchaser further agrees that, upon receipt of a notice from Mandalay of the occurrence of any event of the kind described in Sections 1.2(c)(v), (vi) or (vii), the Purchaser will discontinue disposition of such Registrable Securities under the Registration Statement until the Purchaser's receipt of the copies of the supplemented Prospectus and/or amended Registration Statement contemplated by Section 1.2(j), or until it is advised in writing (the "Advice") by Mandalay that the use of the applicable Prospectus may be resumed, and, in either case, has received copies of any additional or supplemental fillings that are incorporated or deemed to be incorporated by reference in such Prospectus or Registration Statement. Mandalay may provide appropriate stop orders to enforce the provisions of this paragraph.

1.6 Piggy-Back Registrations.

If at any time during the Effectiveness Period there is not an effective Registration Statement covering all of the Registrable Securities and Mandalay shall determine to prepare and file with the SEC a registration statement relating to an offering for its own account or the account of others under the Securities Act of any of its equity securities, other than on Form S-4 or Form S-8 (each as promulgated under the Securities Act) or their then equivalents relating to equity securities to be issued solely in connection with any acquisition of any entity or business or equity securities issuable in connection with stock option or other employee benefit plans, then Mandalay shall send to the Purchaser written notice of such determination and if, within fifteen days after receipt of such notice, the Purchaser shall so request in writing, Mandalay shall include in such registration statement all or any part of the Registrable Securities the Purchaser requests to be registered.

* * * * *

UNCONDITIONAL SECURED GUARANTY AND PLEDGE AGREEMENT

This UNCONDITIONAL SECURED GUARANTY AND PLEDGE AGREEMENT (this "Agreement") is entered into as of ______, 2014, by MANDALAY DIGITAL GROUP, INC., a Delaware corporation ("Guarantor"), in favor of NORTH ATLANTIC SBIC IV, L.P., a Delaware limited partnership ("Purchaser").

For and in consideration of all extensions of credit, loans and other financial accommodations provided by Purchaser to Appia, Inc. ("Borrower"), which loans were and/or will be made pursuant to a Securities Purchase Agreement among Borrower, Guarantor and Purchaser, dated of even date herewith, as amended from time to time, and any and all duly made modifications, extensions or renewals thereof (the "Purchase Agreement"), Guarantor hereby unconditionally and irrevocably guarantees the prompt and complete payment of all Obligations and Borrower's performance of the Purchase Agreement and the other Financing Documents according to their terms. Capitalized terms used but not otherwise defined herein shall have the meanings given them under the Purchase Agreement. This is the "Mandalay Guaranty" contemplated by the Purchase Agreement and is subject to the provisions of a certain Amended and Restated Subordination Agreement, dated of even date herewith, between Silicon Valley Bank ("SVB") and Purchaser.

SECTION 1 – GUARANTEE

- 1.1 If Borrower does not pay or perform when due any Obligations, Guarantor shall upon demand by Purchaser immediately pay to Purchaser or perform such Obligations ("Guarantor Obligations").
- 1.2 The obligations hereunder are independent of the obligations of Borrower, and a separate action or actions may be brought and prosecuted against Guarantor whether action is brought against Borrower or whether Borrower be joined in any such action or actions. Guarantor waives the benefit of any statute of limitations affecting its liability hereunder or the enforcement thereof. Guarantor's liability under this Agreement is not conditioned or contingent upon the genuineness, validity, regularity or enforceability of the Financing Documents.
- 1.3 Purchaser's right to (a) renew, extend or otherwise change the terms of the Financing Documents or any part thereof, (b) take security for the payment due under this Agreement or the other Financing Documents, (c) exchange, enforce, waive or release any such security, and (d) apply any security and direct its sale as Purchaser, in its discretion, chooses, shall not affect the Guarantor's obligations or liability hereunder.
- 1.4 Guarantor waives any right to require Purchaser to (a) proceed against Borrower or any other Person; (b) proceed against or exhaust any security held from Borrower or any other Person; or (c) pursue any other remedy in Purchaser's power whatsoever. Purchaser may, at its election, exercise, decline or fail to exercise, any right or remedy it may have against Borrower or any security held by Purchaser, including without limitation the right to foreclose

upon any such security by judicial or nonjudicial sale, without affecting or impairing in any way the liability of Guarantor hereunder. Guarantor waives any defense arising by reason of any disability or other defense of Borrower or any other guarantor, or by reason of the cessation from any cause whatsoever of the liability of Borrower or any other guarantor. Guarantor waives any setoff, defense or counterclaim that Borrower or Guarantor may have against Purchaser, other than payment in full of the Obligations (other than inchoate indemnity obligations). Guarantor waives any defense arising out of the absence, impairment or loss of any right of reimbursement or subrogation or any other rights against Borrower. Until all of the Obligations (other than contingent indemnity obligations for which no claim has been asserted) have been paid in full, (1) Guarantor shall have no right of subrogation or reimbursement for claims arising out of or in connection with this Agreement, (2) Guarantor shall have no right of contribution or other rights against Borrower in connection with the Obligations, (3) Guarantor waives any right to enforce any remedy that Purchaser now has or may hereafter have against Borrower, and (4) Guarantor waives all rights to participate in any security now or hereafter held by Purchaser. Guarantor waives all presentments, demands for performance, notices of nonperformance, protests, notices of protest, notices of dishonor and notices of acceptance of this Agreement and of the existence, creation or incurrence of new or additional Indebtedness. Guarantor assumes the responsibility for being and keeping itself informed of the financial condition of Borrower and of all other circumstances bearing upon the risk of nonpayment of any Indebtedness or nonperformance of any obligation of Borrower, warrants to Purchaser that it will keep so informed, and agrees that absent a request for particular information by Guarantor, Purchaser shall have no duty to advise Guarantor of information known to Purchaser regarding such condition or any such circumstances. Guarantor waives the benefits of California Civil Code sections 2809, 2810, 2819, 2845, 2847, 2848, 2849, 2850, 2899 and 3433.

1.5 If Borrower becomes insolvent, is adjudicated bankrupt, or files a petition for reorganization, arrangement, composition or similar relief under any present or future provision of the United States Bankruptcy Code or reorganization or insolvency laws of any applicable jurisdiction, or if such a petition is filed against Borrower, and in any such proceeding some or all of any Indebtedness or obligations under the Purchase Agreement are terminated or rejected or any obligation of Borrower is modified or abrogated, or if Borrower's obligations are otherwise avoided for any reason, Guarantor agrees that Guarantor's liability hereunder shall not thereby be affected or modified and such liability shall continue in full force and effect as if no such action or proceeding had occurred until payment in full of the Obligations (other than inchoate indemnity obligations). This Agreement shall continue to be effective or be reinstated, as the case may be, if any payment must be returned by Purchaser upon the insolvency, bankruptcy or reorganization of a Borrower or any other guarantor or otherwise, as though such payment had not been made until payment in full of the Obligations (other than inchoate indemnity obligations).

1.6 Any Indebtedness of Borrower now or hereafter held by Guarantor is hereby subordinated to any Indebtedness of Borrower to Purchaser; and such Indebtedness of Borrower to Guarantor shall be collected, enforced and received by Guarantor as trustee for Purchaser and be paid over to Purchaser in the form received, with any necessary endorsements thereon, on account of the Indebtedness of Borrower to Purchaser but without reducing or affecting in any manner the liability of Guarantor under the other provisions of this Agreement.

SECTION 2 – GRANT OF SECURITY INTEREST

- 2.1 To secure the payment and performance of all of the Guarantor Obligations when due, Guarantor hereby grants Purchaser, a continuing security interest in, and pledges to Purchaser, the Collateral (as defined below), wherever located, whether now owned or existing or hereafter acquired or arising, and all proceeds and products thereof. Guarantor represents, warrants, and covenants that the security interest granted herein is and shall at all times continue to be a second priority perfected security interest in the Collateral (subject only to Permitted Liens that are permitted to have superior priority to Purchaser's Lien under this Agreement). If Guarantor shall acquire a commercial tort claim in an amount greater than Fifty Thousand Dollars (\$50,000), Guarantor shall promptly notify Purchaser in a writing signed by Guarantor of the general details thereof and grant to Purchaser in such writing a security interest therein and in the proceeds thereof, all upon the terms of this Agreement, with such writing to be in form and substance reasonably satisfactory to Purchaser. "Collateral" means all of Guarantor's right, title and interest in and to the following:
- (a) All goods, Accounts (including health-care receivables), Equipment, Inventory, contract rights or rights to payment of money, leases, license agreements, franchise agreements, General Intangibles, commercial tort claims, documents, instruments (including any promissory notes), chattel paper (whether tangible or electronic), cash, deposit accounts, fixtures, letter-of-credit rights (whether or not the letter of credit is evidenced by a writing), securities, and all other investment property, supporting obligations, and financial assets, whether now owned or hereafter acquired, wherever located; and
- (b) all Guarantor's Books relating to the foregoing, and any and all claims, rights and interests in any of the above and all substitutions for, additions, attachments, accessories, accessions and improvements to and replacements, products, proceeds and insurance proceeds of any or all of the foregoing. For the purposes of this Agreement, "Guarantor's Books" are all Guarantor's books and records including ledgers, federal and state tax returns, records regarding Guarantor's assets or liabilities, the Collateral, business operations or financial condition, and all computer programs or storage or any equipment containing such information.
- (c) Notwithstanding anything to the contrary in this Agreement, the Collateral does not include any of the following:
 (i) more than 65% of the presently existing and hereafter arising issued and outstanding shares of capital stock owned by Guarantor of any Foreign Subsidiary which shares entitle the holder thereof to vote for directors or any other matter, (ii) any intent-to-use trademarks at all times prior to the first use thereof, whether by the actual use thereof in commerce, the recording of a statement of use with the United States Patent and Trademark Office or otherwise; (iii) any interest of Guarantor as a lessee or sublessee under a real property lease; (iv) rights held under a license that are not assignable by their terms without the consent of the licensor thereof (but only to the extent such restriction on assignment is enforceable under applicable law); or (v) any interest of Guarantor as a lessee under an Equipment lease if Guarantor is prohibited by the terms of such lease from granting a security interest in such lease or under which such an assignment or Lien would cause a default to occur under such lease; provided, however, that upon termination of such prohibition, such interest shall immediately become Collateral without any action by Guarantor or Purchaser.

2.2 Guarantor hereby authorizes Purchaser to file financing statements, without notice to Guarantor, with all appropriate jurisdictions to perfect or protect Purchaser's interest or rights hereunder. Such financing statements may indicate the Collateral as "all assets of the Debtor" or words of similar effect, or as being of an equal or lesser scope, or with greater detail, all in Purchaser's discretion.

SECTION 3 - PLEDGE

- 3.1 As security for the full, prompt and complete payment and performance when due (whether by stated maturity, by acceleration or otherwise) of all the Guarantor Obligations, Guarantor hereby pledges to Purchaser, and grants to Purchaser, a second priority security interest in all of the following (collectively, the "**Pledged Collateral**"):
- (a) the shares of capital stock or other equity securities of the entities listed on Exhibit A attached hereto, now owned or hereafter acquired (whether in connection with any recapitalization, reclassification, or reorganization of the capital of such entities or any successors in interest thereto) by Guarantor, subject to the limitation set forth in Section 2.1(c)(i) (the "Pledged Shares"), together with all proceeds and substitutions thereof, all cash, stock and other monies and property paid thereon, all rights to subscribe for securities declared or granted in connection therewith, and all other cash and noncash proceeds of the foregoing. On the date hereof, (i) the certificate or certificates representing the Pledged Shares will be delivered to SVB to be held for itself and as bailee for Purchaser, and (ii) an instrument of assignment duly executed in blank by Guarantor will be delivered to Purchaser. To the extent required by the terms and conditions governing the Pledged Shares, Guarantor shall cause the books of each entity whose Pledged Shares are part of the Pledged Collateral and any transfer agent to reflect the pledge of the Pledged Shares. Upon the occurrence and during the continuation of an Event of Default, Purchaser may effect the transfer of any securities included in the Pledged Collateral (including but not limited to the Pledged Shares) into the name of Purchaser and cause new certificates representing such securities to be issued in the name of Purchaser or its transferee;
- (b) all voting trust certificates held by Guarantor evidencing the right to vote any Pledged Shares subject to any voting trust; and
- (c) all additional shares and voting trust certificates of the entities listed on $\underline{\text{Exhibit A}}$ from time to time acquired by Guarantor in any manner, subject to the limitation set forth in Section 2.1(c)(i) (which additional shares shall be deemed to be part of the Pledged Shares), and the certificates representing such additional shares, and all dividends, cash, instruments and other property or proceeds from time to time received, receivable or otherwise distributed in respect of or in exchange for any or all of such Pledged Shares.
- 3.2 Guarantor agrees to pay prior to delinquency all taxes, charges, Liens and assessments, in each case imposed by any Governmental Authority, against the Pledged Collateral, except those with respect to which the amount or validity is being contested in good faith by appropriate proceedings and with respect to which reserves in conformity with GAAP have been provided on the books of Guarantor, and upon the failure of Guarantor to do so, contemporaneous with written notice thereof from Purchaser to Guarantor, Purchaser at its option may pay any of them.

- 3.3 In the event that during the term of this Agreement, any reclassification, readjustment, or other change is declared or made in the capital structure of the issuer of the Pledged Shares, all new, substituted and additional shares, options, or other securities, issued or issuable to Guarantor by reason of any such change or exercise shall be delivered to and held by SVB, if any Senior Debt to SVB is then outstanding, to hold for itself and as bailee for Purchaser, or, if no Senior Debt to SVB is then outstanding, to Purchaser under the terms of this Agreement in the same manner as the Pledged Collateral originally pledged hereunder.
- 3.4 Notwithstanding anything herein to the contrary, Guarantor may exercise any rights under the Pledged Shares to vote such Pledged Shares and receive dividends in respect of such Pledged Shares while no Event of Default has occurred and is continuing.
- 3.5 So long as no Event of Default has occurred and is continuing, Guarantor shall be entitled to exercise any and all voting and other consensual rights pertaining to the Pledged Collateral or any part thereof and Guarantor shall be entitled to receive and to retain and use free and clear of the Lien created hereby any and all non-cash profits, non-cash dividends, or distributions in the form of capital stock, instruments or other property received, receivable or otherwise distributed in respect of, or in exchange for, any Pledged Collateral; provided, however, that any and all non-cash profits, non-cash dividends, or distributions in the form of capital stock, instruments or other property received, receivable or otherwise distributed in respect of, or in exchange for, any Pledged Collateral, shall promptly be delivered to Purchaser to be held as Collateral and shall, if received by Guarantor, be received in trust for the benefit of Purchaser, be segregated from the other property of Guarantor, and promptly be delivered to Purchaser as Pledged Collateral in the same form as so received (with any necessary endorsement).

SECTION 4 – REPRESENTATIONS AND WARRANTIES

- 4.1 Guarantor hereby represents and warrants to Purchaser that:
- (a) the execution, delivery and performance by Guarantor of this Agreement (i) does not contravene any material Requirement of Law or any material contractual restriction binding on or affecting Guarantor or by which Guarantor's property may be affected; (ii) does not require any authorization or approval or other action by, or any notice to or filing with, any Governmental Authority or any other Person under any material indenture, mortgage, deed of trust, lease, agreement or other instrument to which Guarantor is a party or by which Guarantor or any of its property is bound, except such as have been obtained or made; and (iii) does not result in the imposition or creation of any Lien upon any property of Guarantor except the Lien provided to Purchaser pursuant to Section 2 hereof;
- (b) Guarantor has the corporate, limited liability or limited partnership, as applicable, power to execute, deliver and perform this Agreement and the execution, delivery and performance of this Agreement has been duly authorized by all requisite action;
- (c) this Agreement is a valid and binding obligation of Guarantor, enforceable against Guarantor in accordance with its terms, except as the enforceability thereof may be subject to or limited by bankruptcy, insolvency, reorganization, arrangement, moratorium or other similar laws relating to or affecting the rights of creditors generally;

- (d) there is no action, suit or proceeding affecting Guarantor pending or threatened before any court, arbitrator, or Governmental Authority, domestic or foreign, which may have a material adverse effect on the ability of Guarantor to perform its obligations under this Agreement;
 - (e) Guarantor's obligations hereunder are not subject to any offset or defense against Purchaser or Borrower of any kind;
- (f) in connection with this Agreement, Guarantor has delivered to Purchaser a completed certificate signed by Guarantor, entitled "Perfection Certificate," and Guarantor represents and warrants to Purchaser that, except as Purchaser may be notified pursuant to the terms of this Agreement, (a) Guarantor's exact legal name is that indicated on the Perfection Certificate and on the signature page hereof; (b) Guarantor is an organization of the type and is organized in the jurisdiction set forth in the Perfection Certificate; (c) the Perfection Certificate accurately sets forth Guarantor's organizational identification number or accurately states that Guarantor has none; (d) the Perfection Certificate accurately sets forth Guarantor's place of business, or, if more than one, its chief executive office as well as Guarantor's mailing address (if different than its chief executive office); (e) except as set forth on the Perfection Certificate, Guarantor (and each of its predecessors) has not, in the past five (5) years, changed its jurisdiction of formation, organizational structure or type, or any organizational number assigned by its jurisdiction; and (f) all other information set forth on the Perfection Certificate pertaining to Guarantor and each of its Subsidiaries is accurate and complete (it being understood and agreed that Guarantor may from time to time update certain information in the Perfection Certificate after the date hereof to the extent permitted by one or more specific provisions in this Agreement);
- (g) the financial statements of Guarantor, copies of which have been and shall be furnished to Purchaser, fairly present in all material respects the financial position of Guarantor for the dates and periods purported to be covered thereby, and no Event of Default has occurred, is occurring, or would occur upon the effectiveness of this Agreement.
- (h) after the incurrence of Guarantor's obligations under this Agreement, the fair salable value of Guarantor's assets (including goodwill minus disposition costs) exceeds the fair value of its liabilities; Guarantor is not left with unreasonably small capital after the transactions in this Agreement or the other Financing Documents; and Guarantor is able to pay its debts (including trade debts) as they mature;
- (i) all representations and warranties contained in this Agreement are true at the time of Guarantor's execution of this Agreement and, with respect to the representations and warranties contained in clauses (j) through (o) of this Section 4.1, shall be true upon the issuance and receipt by Guarantor of any additional Pledged Shares and/or Pledged Collateral after the date hereof;

- (j) Guarantor is, at the time of delivery of the Pledged Shares to Purchaser hereunder, the sole holder of record and the sole beneficial owner of the Pledged Collateral, free and clear of any Lien thereon or affecting title thereto, except for the Lien created by this Agreement and Permitted Liens;
- (k) none of the Pledged Shares have been transferred in violation of applicable federal or state securities, or similar laws, which such transfer may be subject to in the United States of America or any applicable jurisdiction;
 - (1) all of the Pledged Shares have been duly authorized, validly issued, fully paid and are non-assessable;
- (m) to Guarantor's knowledge, (i) there are no subscriptions, warrants, rights of first refusal or other restrictions on transfer relative to, or options exercisable with respect to the Pledged Shares; and (ii) the Pledged Shares are not the subject of any present or threatened suit, action, arbitration, administrative or other proceeding, and Guarantor knows of no reasonable grounds for the institution of any such proceedings;
- (n) no consent, approval, authorization or other order of any Person and no consent or authorization of any Governmental Authority is required to be made or obtained which has not yet been made or obtained by Guarantor either (i) for the pledge by Guarantor of the Pledged Collateral pursuant to this Agreement or for the execution, delivery, or performance of this Agreement by Guarantor; or (ii) for the exercise by Purchaser of the voting or other rights provided for in this Agreement or the remedies with respect to the Pledged Collateral pursuant to this Agreement, except, in the case of clause (i) and (ii), as (x) may be required in connection with such disposition by laws affecting the offer and sale of securities generally and (y) by the laws of the United States and any applicable foreign jurisdiction; and
- (o) the pledge, grant of a security interest in, and delivery of the Pledged Collateral pursuant to this Agreement, will create a valid first priority Lien on and in the Pledged Collateral, and the proceeds thereof, securing the payment of the Guarantor Obligations.

SECTION 5 – COVENANTS

- 5.1 Guarantor covenants and agrees to the following:
- (a) Deliver to Purchaser, (i) within five (5) days of filing, copies of all periodic and other reports, proxy statements and other materials filed by Guarantor with the SEC, any Governmental Authority succeeding to any or all of the functions of the SEC or with any national securities exchange, or distributed to its shareholders, as the case may be. Documents required to be delivered pursuant to the terms hereof (to the extent any such documents are included in materials otherwise filed with the SEC) may be delivered electronically and if so delivered, shall be deemed to have been delivered on the date on which Guarantor posts such documents, or provides a link thereto, on Guarantor's website on the Internet at Guarantor's website address, (ii) a prompt report of any legal actions pending or threatened in writing against Guarantor or any of its Subsidiaries that could result in damages or costs to Guarantor or any of its Subsidiaries of, individually or in the aggregate, Two Hundred Fifty Thousand Dollars (\$250,000) or more, and (iii) financial information reasonably requested by Purchaser;

- (b) Timely file, and require each of its Subsidiaries to timely file, all required tax returns and reports and timely pay, and require each of its Subsidiaries to timely pay, all foreign, federal, state and material local taxes, assessments, deposits and contributions owed by Guarantor and each of its Subsidiaries, except for deferred payment of any contested taxes, provided that Guarantor (i) in good faith contests its obligation to pay the taxes by appropriate proceedings promptly and diligently instituted and conducted, (ii) notifies Purchaser in writing of the commencement of, and any material development in, the proceedings, (iii) posts bonds or takes any other steps required to prevent the governmental authority levying such contested taxes from obtaining a Lien upon any of the Collateral that is other than a "Permitted Lien," and shall deliver to Purchaser, on reasonable demand, appropriate certificates attesting to such payments, and pay all amounts necessary to fund all present pension, profit sharing and deferred compensation plans in accordance with their terms;
- (c) Keep its business and the Collateral insured for risks and in amounts standard for companies in Guarantor's industry and location. Insurance policies shall be in a form, with companies, and in amounts standard for companies in Guarantor's industry and location. All property policies shall have a lender's loss payable endorsement showing Purchaser as a lender loss payee and waive subrogation against Purchaser. All liability policies shall show, or have endorsements showing, Purchaser as an additional insured. All policies (or their respective endorsements) shall provide that the insurer shall give Purchaser at least twenty (20) days notice before canceling, amending, or declining to renew its policy. At Purchaser's reasonable request, Guarantor shall deliver certified copies of policies and evidence of all premium payments. If an Event of Default exists, proceeds payable under any policy shall, at Purchaser's option, be payable to Purchaser on account of the Guarantor Obligations unless SVB has prior rights thereto. After the occurrence and during the continuance of an Event of Default, all proceeds payable under such casualty policy shall, at the option of Purchaser, be payable to Purchaser on account of the Guarantor Obligations unless SVB has prior rights thereto. If Guarantor fails to obtain insurance as required under this Section 5.1(c) or to pay any amount or furnish any required proof of payment to third persons and Purchaser, Purchaser may make all or part of such payment or obtain such insurance policies required in this Section 5.1(c), and take any action under the policies Purchaser deems prudent;
- (d) Provide Purchaser five (5) days prior written notice before establishing any Collateral Account at or with any bank or financial institution. For each Collateral Account that Guarantor at any time maintains in the United States, Guarantor shall cause the applicable bank or financial institution at or with which any Collateral Account is maintained to execute and deliver a Control Agreement or other appropriate instrument with respect to such Collateral Account to perfect Purchaser's Lien in such Collateral Account in accordance with the terms hereunder which Control Agreement may not be terminated without the prior written consent of Purchaser unless this Agreement and Purchaser's Liens hereunder are terminated. The provisions of the previous sentence shall not apply to (i) deposit accounts exclusively used for payroll, payroll taxes and other employee wage and benefit payments to or for the benefit of Guarantor's employees and identified to Purchaser by Guarantor as such, or (ii) other Collateral Accounts holding balances not in excess of Seven Hundred Fifty Thousand Dollars (\$750,000) in the aggregate at any time;

- (e) (i) Protect, defend and maintain the validity and enforceability of its Intellectual Property material to its business; (ii) promptly advise Purchaser in writing of material infringements of its Intellectual Property material to its business; (iii) not allow any Intellectual Property material to Guarantor's business to be abandoned, forfeited or dedicated to the public without Purchaser's written consent, and (iv) provide written notice to Purchaser (which may be delivered electronically and if so delivered, shall be deemed to have been delivered on the date on which Guarantor posts such documents, or provides a link thereto, on Guarantor's website on the Internet at Guarantor's website address) within ten (10) days of entering or becoming bound by any Restricted License (other than over-the-counter software that is commercially available to the public). Guarantor shall take such commercially reasonable steps as Purchaser requests to obtain the consent of, or waiver by, any person whose consent or waiver is necessary for (A) any Restricted License to be deemed "Collateral" and for Purchaser to have a security interest in it that might otherwise be restricted or prohibited by law or by the terms of any such Restricted License, whether now existing or entered into in the future, and (B) Purchaser to have the ability in the event of a liquidation of any Collateral to dispose of such Collateral in accordance with Purchaser's rights and remedies under this Agreement and the other Financing Documents;
- (f) Allow Purchaser, or its agents, at reasonable times, on three (3) Business Days' notice (provided no notice is required if an Event of Default has occurred and is continuing), to inspect the Collateral and audit and copy Guarantor's Books. Such inspections or audits shall be conducted no more often than once every twelve (12) months unless an Event of Default has occurred and is continuing. The foregoing inspections and audits shall be at Guarantor's expense, and the charge therefor shall be \$850 per person per day (or such higher amount as shall be equal to SVB's then-current standard charge for the same), plus reasonable out-of-pocket expenses. In the event Guarantor and Purchaser schedule an audit more than ten (10) days in advance, and Guarantor cancels or seeks to reschedule the audit with less than ten (10) days written notice to Purchaser, then (without limiting any of Purchaser's rights or remedies), Guarantor shall pay Purchaser a fee of \$1,000 plus any out-of-pocket expenses incurred by Purchaser to compensate Purchaser for the anticipated costs and expenses of the cancellation or rescheduling;
- (g) Guarantor shall, at any time and from time to time, execute and deliver such further instruments and take such further action as may reasonably be requested by Purchaser to effect the purposes of this Agreement;
- (h) Guarantor shall make any payments and do any act reasonably necessary or convenient to protect the Collateral and the value of Purchaser's interest therein. If Guarantor fails to pay any amount or furnish any required proof of payment to third Persons as required hereunder, Purchaser may make all or part of the payment or obtain insurance policies, and take any action under the policies as Purchaser deems prudent. Any amounts paid by Purchaser are immediately due from and payable by Guarantor, bearing interest at the Default Rate and secured by the Collateral. No payments by Purchaser are an agreement to make similar payments in the future or Purchaser's waiver of any Event of Default; and
- (i) (i) Maintain its and all its Subsidiaries' legal existence and good standing in their respective jurisdictions of formation and maintain qualification in each

jurisdiction in which the failure to so qualify would reasonably be expected to have a material adverse effect on Guarantor's business or operations; (ii) comply, and have each Subsidiary comply, with all laws, ordinances and regulations to which it is subject, noncompliance with which could have a material adverse effect on Guarantor's business; (iii) obtain all of the Governmental Approvals necessary for the performance by Guarantor of its obligations under the Financing Documents to which it is a party and the grant of a security interest to Purchaser in all of its property; and (iv) promptly provide copies of any such obtained Governmental Approvals to Purchaser.

- 5.2 Guarantor shall not do any of the following without Purchaser's prior written consent:
- (a) Convey, sell, lease, transfer, assign, or otherwise dispose of (collectively, "Transfer"), or permit any of its Subsidiaries to Transfer, all or any part of its business or property, except for Transfers (a) of Inventory in the ordinary course of business; (b) of worn-out or obsolete Equipment or Equipment that is no longer useful for Guarantor's business; (c) in connection with Permitted Liens and Investments that are not otherwise prohibited under this Agreement or the other Financing Documents; (d) of non-exclusive licenses for the use of the property of Guarantor or its Subsidiaries in the ordinary course of business and licenses that could not result in a legal transfer of title of the licensed property but that may be exclusive in respects other than territory and that may be exclusive as to territory only as to discreet geographical areas outside of the United States; and (e) consisting of Guarantor's use or transfer of money or Cash Equivalents in the ordinary course of business in a manner that is not prohibited by the terms of this Agreement or the other Financing Documents;
- (b) (i) Liquidate or dissolve; (ii) permit or suffer a Change in Control, or (iii) permit Borrower to cease being a wholly-owned Subsidiary of Guarantor;
- (c) Guarantor shall not, without at least thirty (30) days prior written notice to Purchaser: (1) add any new offices or business locations, including warehouses (unless such new offices or business locations contain less than One Hundred Thousand Dollars (\$100,000) in Guarantor's assets or property) or deliver any portion of the Collateral valued, individually or in the aggregate, in excess of One Hundred Thousand Dollars (\$100,000) to a bailee at a location other than to a bailee and at a location already disclosed in the Perfection Certificate, (2) change its jurisdiction of organization, (3) change its organizational structure or type, (4) change its legal name, or (5) change any organizational number (if any) assigned by its jurisdiction of organization. If Guarantor intends to deliver any portion of the Collateral valued, individually or in the aggregate, in excess of One Hundred Thousand Dollars (\$100,000) to a bailee, and Purchaser and such bailee are not already parties to a bailee agreement governing both the Collateral and the location to which Guarantor intends to deliver the Collateral, then Guarantor will first receive the written consent of Purchaser, and use commercially reasonable efforts to cause such bailee to execute and deliver a bailee agreement in form and substance satisfactory to Purchaser in its reasonable discretion;
- (d) Merge or consolidate, or permit any of its Subsidiaries to merge or consolidate, with any other Person. A Subsidiary may merge or consolidate into another Subsidiary or into Guarantor;

(e) Create, incur, assume, or be liable for any Indebtedness, or permit any Subsidiary to do so, other than Permitted Indebtedness;

- (f) Create, incur, allow, or suffer any Lien on any of its property, or assign or convey any right to receive income, including the sale of any Accounts, or permit any of its Subsidiaries to do so, except for Permitted Liens, permit any Collateral not to be subject to the security interest granted herein, or enter into any agreement, document, instrument or other arrangement (except with or in favor of Purchaser or in connection with Permitted Liens) with any Person which directly or indirectly prohibits or has the effect of prohibiting Guarantor from assigning, mortgaging, pledging, granting a security interest in or upon, or encumbering any of Guarantor's Intellectual Property, except, in each case, as is otherwise permitted in Section 5.2(a) hereof and the definition of "Permitted Liens" herein;
 - (g) Maintain any Collateral Account except pursuant to the terms of Section 5.1(d) hereof;
- (h) Pay (other than in the form of Guarantor's capital stock) any dividends or make any distribution or payment or redeem, retire or purchase any capital stock; provided, however, that Guarantor may repurchase capital stock from former directors, officers, employees or consultants at the original purchase price thereof, but not to exceed in the aggregate of Five Hundred Thousand Dollars (\$500,000) per fiscal year, as permitted by Guarantor's equity incentive plans, restricted stock purchase agreements, stock option agreements, stock grant agreements or similar agreements, so long as an Event of Default does not exist at the time of such repurchase and would not exist after giving effect to such repurchase;
- (i) (i) Make or permit any payment on any Subordinated Debt, except under the terms of the subordination, intercreditor, or other similar agreement to which such Subordinated Debt is subject, or (ii) amend any provision in any document relating to the Subordinated Debt which would increase the amount thereof or adversely affect the subordination thereof to Guarantor Obligations owed to Purchaser, except as may be permitted under the terms of the subordination, intercreditor, or other similar agreement to which such Subordinated Debt is subject; and
- (j) Become an "investment company" or a company controlled by an "investment company", under the Investment Company Act of 1940, as amended, or undertake as one of its important activities extending credit to purchase or carry margin stock (as defined in Regulation U of the Board of Governors of the Federal Reserve System), or use the proceeds of any credit extended pursuant to the Purchase Agreement for that purpose; fail to meet the minimum funding requirements of ERISA, permit a Reportable Event or Prohibited Transaction, as defined in ERISA, to occur; fail to comply with the Federal Fair Labor Standards Act or violate any other law or regulation, if the violation could reasonably be expected to have a material adverse effect on Guarantor's business, or permit any of its Subsidiaries to do so; withdraw or permit any Subsidiary to withdraw from participation in, permit partial or complete termination of, or permit the occurrence of any other event with respect to, any present pension, profit sharing and deferred compensation plan which could reasonably be expected to result in any material liability of Guarantor, including any liability to the Pension Benefit Guaranty Corporation or its successors or any other governmental agency.

- (k) Form any direct or indirect Subsidiary or acquire any direct or indirect Subsidiary after the date hereof, unless Guarantor (a) causes any such new Domestic Subsidiary to provide to Purchaser a joinder to this Agreement, together with such appropriate financing statements and/or Control Agreements, all in form and substance reasonably satisfactory to Purchaser (including being sufficient to grant Purchaser a second priority Lien (subject only to Permitted Liens) in and to the assets of such newly formed or acquired Domestic Subsidiary), (b) provides to Purchaser appropriate certificates and powers and financing statements, pledging all of the direct or beneficial ownership interest in any such new Subsidiary (except to the extent shares of any such new Foreign Subsidiary are excluded from the Collateral pursuant to Section 2.1(c)(i)), in form and substance reasonably satisfactory to Purchaser, and (c) provides to Purchaser all other documentation in form and substance reasonably satisfactory to Purchaser which in its opinion is appropriate with respect to the execution and delivery of the applicable documentation referred to above. Any document, agreement, or instrument executed or issued pursuant to this Section 5.2(k) shall be a Financing Document.
- (l) After the repayment in full of all Senior Debt owed to SBV, Guarantor shall promptly provide written notice thereof to Purchaser.
- 5.3 So long as Purchaser has any commitment to extend credit to Borrower or Borrower has any Obligations (other than contingent indemnity obligations for which no claim has been asserted), Guarantor agrees that Guarantor:
- (a) will not (i) except as otherwise permitted by the Financing Documents, sell, transfer or otherwise dispose of, or grant any option or warrant with respect to, any of the Pledged Collateral (or any part thereof or interest therein) except with the prior written consent of Purchaser, or (ii) create or permit to exist any Lien upon or with respect to any of the Pledged Collateral, except for Permitted Liens. If any Pledged Collateral, or any part thereof, is sold, transferred or otherwise disposed of in violation of this Section 5.3, the security interest of Purchaser shall continue in the Pledged Collateral, to the extent permitted under applicable law, notwithstanding such sale, transfer or other disposition, and Guarantor will deliver any proceeds thereof to Purchaser to be held as Pledged Collateral hereunder;
- (b) shall, at Guarantor's own expense, promptly execute, acknowledge, and deliver all such instruments and take all such actions as Purchaser from time to time may reasonably request to ensure to Purchaser the benefits of the Lien in and to the Pledged Collateral intended to be created by this Agreement;
- (c) shall maintain, preserve and, at Purchaser's request, use commercially reasonable efforts to defend the title to the Pledged Collateral and the Lien of Purchaser thereon against the claim of any other Person; and
- (d) upon obtaining any shares of capital stock or other equity securities that should be pledged pursuant to Section 3.1 of this Agreement, shall promptly deliver to Purchaser a duly executed Pledge Supplement in substantially the form of <u>Exhibit B</u> attached

hereto (a "**Pledge Supplement**") identifying such additional shares of capital stock or other equity securities. Guarantor hereby authorizes Purchaser to attach each Pledge Supplement to this Agreement and agrees that all shares of capital stock or other equity securities listed thereon shall for all purposes hereunder constitute Pledged Collateral.

SECTION 6 – EVENTS OF DEFAULT

- 6.1 Upon the occurrence and during the continuation of an Event of Default, Purchaser shall have all of the rights of a secured party under the Code with respect to the Collateral. Guarantor's obligations hereunder are not limited to the Collateral or any exercise by Purchaser of rights and remedies against the same, and Purchaser may pursue any other available rights and remedies against Guarantor, whether hereunder, at law or otherwise, without resort to the Collateral if Purchaser deems it in its best interests to do so.
 - 6.2 During the existence and continuation of an Event of Default,
- (a) Purchaser may, to the extent permitted by applicable law, at its election, apply, set off, collect or sell in one or more sales, or take such steps as may be necessary to liquidate and reduce to cash in the hands of Purchaser in whole or in part, with or without any previous demands or demand of performance or notice or advertisement, the whole or any part of the Pledged Collateral in such order as Purchaser may elect, and any such sale may be made either at public or private sale at its place of business or elsewhere, or at any broker's board or securities exchange, either for cash or upon credit or for future delivery; provided, however, that if such disposition is at private sale, then the purchase price of the Pledged Collateral shall be equal to the public market price then in effect, or, if at the time of sale no public market for the Pledged Collateral exists, then, in recognition of the fact that the sale of the Pledged Collateral would have to be registered under the Securities Act of 1933, as amended (the "Act"), and that the expenses of such registration are commercially unreasonable for the type and amount of collateral pledged hereunder, Purchaser and Guarantor hereby agree that such private sale shall be at a purchase price mutually agreed to by Purchaser and Guarantor or, if the parties cannot agree upon a purchase price, then at a purchase price established by Purchaser in the exercise of its reasonable discretion. Purchaser shall be under no obligation to delay the sale of any of the Pledged Shares for the period of time necessary to permit Guarantor to register such securities for public sale under the Act, or under applicable state securities laws, even if Guarantor would agree to do so. Purchaser may be the purchaser of any or all Pledged Collateral so sold and hold the same thereafter in its own right free from any claim of Guarantor or right of redemption. To the extent permitted by applicable law, demands of performance, notices of sale, advertisements and presence of property at sale are hereby waived. Any sale hereunder may be conducted by any officer or agent of Purchaser; and

(b) Purchaser shall have all of the rights of a secured party under the Code with respect to the Pledged Collateral. Guarantor's obligations hereunder are not limited to the Pledged Collateral or any exercise by Purchaser of rights and remedies against the same, and Purchaser may pursue any other available rights and remedies against any Guarantor, whether hereunder, at law or otherwise, without resort to the Pledged Collateral if Purchaser deems it in its best interests to do so.

- 6.3 During the existence of an Event of Default, the proceeds of the sale of any of the Pledged Collateral and all sums received or collected by Purchaser from or on account of such Pledged Collateral under Section 6.3 shall be applied by Purchaser to the payment of expenses incurred or paid by Purchaser in connection with any sale, transfer or delivery of the Pledged Collateral, to the payment of any other costs, charges, attorneys' fees or expenses mentioned herein, and to the payment of the Obligations or any part hereof, all in such order and manner as Purchaser in its discretion may determine.
- 6.4 Upon the transfer by Purchaser of all or any part of the Obligations pursuant to the terms of the Purchase Agreement, Purchaser may transfer all or any part of the Pledged Collateral to the transferee of the Obligations and shall be fully discharged thereafter from all liability and responsibility with respect to such Pledged Collateral so transferred, and the transferee shall be vested with all the rights and powers of Purchaser hereunder with respect to such Pledged Collateral so transferred; but with respect to any Pledged Collateral not so transferred, Purchaser shall retain all rights and powers hereby given.

SECTION 7 – **DEFINITIONS**

- 7.1 "Change in Control" means any event, transaction, or occurrence as a result of which (a) any "person" (as such term is defined in Sections 3(a)(9) and 13(d)(3) of the Exchange Act), other than a trustee or other fiduciary holding securities under an employee benefit plan of Guarantor, is or becomes a beneficial owner (within the meaning Rule 13d-3 promulgated under the Exchange Act), directly or indirectly, of securities of Guarantor, representing fifty percent (50%) or more of the combined voting power of Guarantor's then outstanding securities; or (b) during any period of twelve (12) consecutive calendar months, individuals who at the beginning of such period constituted the Board of Directors of Guarantor (together with any new directors whose election by the Board of Directors of Guarantor was approved by a vote of not less than two-thirds of the directors then still in office who either were directors at the beginning of such period or whose election or nomination for election was previously so approved) cease for any reason other than death or disability to constitute a majority of the directors then in office.
 - 7.2 "Collateral Account" is any Deposit Account, Securities Account, or Commodity Account.
- 7.3 "Commodity Account" is any "commodity account" as defined in the Code with such additions to such term as may hereafter be made.
- 7.4 "Control Agreement" is any control agreement entered into among the depository institution at which Guarantor maintains a Deposit Account or the securities intermediary or commodity intermediary at which Guarantor maintains a Securities Account or a Commodity Account, Guarantor, and Purchaser pursuant to which Purchaser obtains control (within the meaning of the Code) over such Deposit Account, Securities Account, or Commodity Account.
- 7.5 "**Deposit Account**" is any "deposit account" as defined in the Code with such additions to such term as may hereafter be made.

- 7.6 "Foreign Subsidiary" means any Subsidiary which is not organized under the laws of the United States or any state or territory thereof or the District of Columbia.
 - 7.7 "Intellectual Property" means all of Guarantor's right, title, and interest in and to the following:
 - (a) its Copyrights, Trademarks and Patents;
- (b) any and all trade secrets and trade secret rights, including, without limitation, any rights to unpatented inventions, know-how, operating manuals;
 - (c) any and all source code;
 - (d) any and all design rights which may be available to Guarantor;
- (e) any and all claims for damages by way of past, present and future infringement of any of the foregoing, with the right, but not the obligation, to sue for and collect such damages for said use or infringement of the Intellectual Property rights identified above; and
 - (f) all amendments, renewals and extensions of any of the Copyrights, Trademarks or Patents.

7.8 "Permitted Indebtedness" is:

- (a) Guarantor's Indebtedness to Purchaser under this Agreement and the other Financing Documents;
- (b) Subordinated Debt;
- (c) unsecured Indebtedness to trade creditors and corporate credit cards incurred in the ordinary course of business;
- (d) Indebtedness incurred as a result of endorsing negotiable instruments received in the ordinary course of business;
- (e) Indebtedness secured by Liens permitted under clauses (a) and (c) of the definition of "Permitted Liens" hereunder;
- (f) extensions, refinancings, modifications, amendments and restatements of any items of Permitted Indebtedness (a) through (e) above, provided that the principal amount thereof is not increased or the terms thereof are not modified to impose more burdensome terms upon Guarantor or its Subsidiary, as the case may be;
- (g) Indebtedness (including, without limitation, Contingent Indebtedness) to SVB not to exceed Five Million Dollars (\$5,000,000) plus accrued interest and fees; and

(h) Indebtedness of Foreign Subsidiaries of Guarantor so long as Guarantor is not an obligor or credit party with respect to any such Indebtedness and no recourse to Guarantor or any of its property may be had by the creditors with respect to such Indebtedness.

7.9 "Permitted Liens" are:

- (a) Liens arising under this Agreement or the other Financing Documents;
- (b) Liens for taxes, fees, assessments or other government charges or levies, either (i) not due and payable or (ii) being contested in good faith and for which Guarantor maintains adequate reserves on its books, provided that no notice of any such Lien has been filed or recorded under the Internal Revenue Code of 1986, as amended, and the Treasury Regulations adopted thereunder;
- (c) purchase money Liens (i) on Equipment acquired or held by Guarantor incurred for financing the acquisition of the Equipment securing no more than One Million Dollars (\$1,000,000) in the aggregate amount outstanding at any time, or (ii) existing on Equipment when acquired, if the Lien is confined to the property and improvements and the proceeds of the Equipment;
- (d) Liens of carriers, warehousemen, suppliers, or other Persons that are possessory in nature arising in the ordinary course of business so long as such Liens attach only to Inventory and Equipment and which are not delinquent or remain payable without penalty or which are being contested in good faith and by appropriate proceedings which proceedings have the effect of preventing the forfeiture or sale of the property subject thereto;
- (e) Liens to secure payment of workers' compensation, employment insurance, old-age pensions, social security and other like obligations incurred in the ordinary course of business (other than Liens imposed by ERISA);
- (f) Liens incurred in the extension, renewal or refinancing of the indebtedness secured by Liens described in (a) through (c), but any extension, renewal or replacement Lien must be limited to the property encumbered by the existing Lien and the principal amount of the indebtedness may not increase;
- (g) leases or subleases of real property granted in the ordinary course of Guarantor's business (or, if referring to another Person, in the ordinary course of such Person's business), and leases, subleases, non-exclusive licenses or sublicenses of personal property (other than Intellectual Property) granted in the ordinary course of Guarantor's business (or, if referring to another Person, in the ordinary course of such Person's business), if the leases, subleases, licenses and sublicenses do not prohibit granting Purchaser a security interest therein;
 - (h) non-exclusive licenses of Intellectual Property granted to third parties in the ordinary course of business;

- (i) licenses of Intellectual Property that could not result in a legal transfer of title of the licensed property that may be exclusive in respects other than territory;
 - (j) Liens arising from attachments or judgments, orders, or decrees in circumstances not constituting an Event of Default;
- (k) Liens in favor of other financial institutions arising in connection with Guarantor's deposit and/or securities accounts held at such institutions, provided that Purchaser has a perfected security interest in the amounts held in such deposit and/or securities accounts; and
- (I) Liens in favor of SVB in connection with Indebtedness permitted under clause (g) of the definition of "Permitted Indebtedness" hereunder.
- 7.10 "**Restricted License**" is any material license or other agreement with respect to which Guarantor is the licensee (a) that prohibits or otherwise restricts Guarantor from granting a security interest in Guarantor's interest in such license or agreement or any other property, or (b) for which a default under or termination of could interfere with the Purchaser's right to sell any Collateral.
- 7.11 "Securities Account" is any "securities account" as defined in the Code with such additions to such term as may hereafter be made.
- 7.12 "**Subordinated Debt**" is indebtedness incurred by Guarantor subordinated to all of Guarantor's now or hereafter indebtedness to Purchaser (pursuant to a subordination, intercreditor, or other similar agreement in form and substance reasonably satisfactory to Purchaser entered into between Purchaser and the other creditor), on terms acceptable to Purchaser.

SECTION 8 - MISCELLANEOUS

8.1 Guarantor agrees to pay reasonable attorneys' fees and all other costs and expenses which may be incurred by Purchaser in the enforcement of this Agreement. No terms or provisions of this Agreement may be changed, waived, revoked or amended without Purchaser's prior written consent. Should any provision of this Agreement be determined by a court of competent jurisdiction to be unenforceable, all of the other provisions shall remain effective. This Agreement embodies the entire agreement between the parties hereto with respect to the matters set forth herein, and supersedes all prior agreements among the parties with respect to the matters set forth herein. No course of prior dealing among the parties, no usage of trade, and no parole or extrinsic evidence of any nature shall be used to supplement, modify or vary any of the terms hereof. Purchaser may assign this Agreement without in any way affecting Guarantor's liability under it. This Agreement shall inure to the benefit of Purchaser, Guarantor and their respective successors and assigns. This Agreement is in addition to the guaranties of any other guarantors of the Obligations. Notwithstanding the foregoing, so long as no Event of Default shall have occurred and be continuing, Purchaser shall not assign its interest in this Agreement to any Person who in the reasonable estimation of Purchaser is (a) a direct competitor of Guarantor, whether as an operating company or direct or indirect parent with voting control over such operating company, or (b) a vulture fund or distressed debt fund.

8.2 New York law governs this Agreement without regard to principles of conflicts of law. Guarantor and Purchaser each submit to the exclusive jurisdiction of the State and Federal courts in Delaware, provided, however, that if for any reason Purchaser cannot avail itself of such courts in the State of Delaware, Guarantor accepts jurisdiction of the courts and venue in Maine; provided, further, that nothing in this Agreement shall be deemed to operate to preclude Purchaser from bringing suit or taking other legal action in any other jurisdiction to realize on the Collateral or any other security for the Obligations, or to enforce a judgment or other court order in favor of Purchaser. Guarantor expressly submits and consents in advance to such jurisdiction in any action or suit commenced in any such court, and Guarantor hereby waives any objection that it may have based upon lack of personal jurisdiction, improper venue, or forum non conveniens and hereby consents to the granting of such legal or equitable relief as is deemed appropriate by such court. Guarantor hereby waives personal service of the summons, complaints, and other process issued in such action or suit and agrees that service of such summons, complaints, and other process may be made by registered or certified mail addressed to Guarantor at the address set forth on the signature page hereto and that service so made shall be deemed completed upon the earlier to occur of Guarantor's actual receipt thereof or three (3) days after deposit in the U.S. mails, proper postage prepaid.

TO THE EXTENT PERMITTED BY APPLICABLE LAW, GUARANTOR AND PURCHASER EACH WAIVE THEIR RIGHT TO A JURY TRIAL OF ANY CLAIM OR CAUSE OF ACTION ARISING OUT OF OR BASED UPON THIS AGREEMENT, THE OTHER FINANCING DOCUMENTS OR ANY CONTEMPLATED TRANSACTION, INCLUDING CONTRACT, TORT, BREACH OF DUTY AND ALL OTHER CLAIMS. THIS WAIVER IS A MATERIAL INDUCEMENT FOR BOTH PARTIES TO ENTER INTO THIS AGREEMENT. EACH PARTY HAS REVIEWED THIS WAIVER WITH ITS COUNSEL.

8.3 This Agreement may be executed in counterpart signature pages, all of which taken together shall be deemed to be one original of this instrument. Delivery of an executed counterpart to this Agreement by facsimile or electronic mail shall be effective as a manually executed counterpart to this Agreement.

8.4 All covenants, representations and warranties made in this Agreement continue in full force until this Agreement has terminated pursuant to its terms and all Obligations and/or Guarantor Obligations have been satisfied. So long as Borrower and Guarantor have satisfied the Obligations and Guarantor Obligations, respectively (other than inchoate indemnity obligations, any other obligations which, by their terms, are to survive the termination of this Agreement, this Agreement shall be deemed terminated, subject to Section 1.5. Those obligations that are expressly specified in this Agreement as surviving this Agreement's termination shall continue to survive notwithstanding this Agreement's termination.

[Signature begins on next page.]

IN WITNESS WHEREOF, the undersigned Guarantor has executed this Unconditional Secured Guaranty and Pledge Agreement as of this __ day of November, 2014.

	Guarantor:
	MANDALAY DIGITAL GROUP, INC., a Delaware corporation
	By:
	Name:
	Title:
	Address:
Acknowledged and Accepted by:	
<u>Purchaser</u> :	
NORTH ATLANTIC SBIC IV, L.P., A Delaware limited partnership	
By:	
Name:	
Title:	
Address: []	

EXHIBIT A – PLEDGED SHARES

						% of	
Name of		Number of	Number of	Number of	% of Shares	Outstanding	
Pledged Share	Jurisdiction of	Shares	Shares	Shares Owned	Owned by	Shares	Certificate
Issuer	Organization	Authorized	Outstanding	by Guarantor	Guarantor	Pledged	Number

Appia, Inc.

EXHIBIT B – FORM OF PLEDGE SUPPLEMENT

PLEDGE SUPPLEMENT

This Pledge Supplement referred to below. The underst					Guaranty and Pledge Agreement ditional Secured Guaranty and
Pledge Agreement, dated as of	f, 2014 (as	amended, restated	, modified, renewed	d, supplemented of	or extended from time to time, the
"Guaranty and Pledge Agre	ement"; the terms defin	ed therein and not	otherwise defined h	nerein being used	as therein defined), made by the
undersigned, as Guarantor, in	favor of North Atlantic	SBIC IV, L.P., as	Purchaser.		
The shares of capital sto pledged by the undersigned ar	1 2		0 11		me part of the Pledged Collateral arantor Obligations.
The undersigned hereby are true and correct in all mate					uaranty and Pledge Agreement te hereof.
				DALAY DIGIT	AL GROUP, INC.,
			By:		
			Name	: <u> </u>	_
			Title:		
Name of	Number of	Number of	Number of	Number of	% of Outstanding

Shares

Outstanding

Shares Owned

by Guarantor

Shares

Issued

Certificate

Number

Shares

Pledged

Pledged Share

Issuer

Jurisdiction of

Organization

Shares

Authorized

UNCONDITIONAL SECURED GUARANTY AND PLEDGE AGREEMENT

This UNCONDITIONAL SECURED GUARANTY AND PLEDGE AGREEMENT (this "Agreement") is entered into as of 2014, by MANDALAY DIGITAL GROUP, INC., a Delaware corporation ("Guarantor") in favor of SILICON VALLEY BANK ("Bank").

For and in consideration of all extensions of credit, loans and other financial accommodations provided by Bank to Appia, Inc. ("Borrower"), which loans were and will be made pursuant to an Second Amended and Restated Loan and Security Agreement among Borrower and Bank, dated as of , 2014, as amended from time to time, and any and all modifications, extensions or renewals thereof (the "Loan Agreement"), Guarantor hereby unconditionally and irrevocably guarantees the prompt and complete payment of all Obligations and Borrower's performance of the Loan Agreement and the other Loan Documents (other than any warrant or other equity instruments) according to their terms. Capitalized terms used but not otherwise defined herein shall have the meanings given them under the Loan Agreement.

SECTION 1 - GUARANTEE

- 1.1 If Borrower does not pay when due any Obligations, Guarantor shall upon demand by Bank immediately pay to Bank such Obligations ("Guarantor Obligations").
- 1.2 The obligations hereunder are independent of the obligations of Borrower, and a separate action or actions may be brought and prosecuted against Guarantor whether action is brought against Borrower or whether Borrower be joined in any such action or actions. Guarantor waives the benefit of any statute of limitations affecting its liability hereunder or the enforcement thereof. Guarantor's liability under this Agreement is not conditioned or contingent upon the genuineness, validity, regularity or enforceability of the Loan Documents.
- 1.3 Bank's right to (a) renew, extend or otherwise change the terms of the Loan Documents or any part thereof, (b) take security for the payment due under this Agreement or the Loan Documents, (c) exchange, enforce, waive or release any such security, and (d) apply any security and direct its sale as Bank, in its discretion, chooses, shall not affect the Guarantor's obligations or liability hereunder.
- 1.4 Guarantor waives any right to require Bank to (a) proceed against Borrower or any other Person; (b) proceed against or exhaust any security held from Borrower or any other Person; or (c) pursue any other remedy in Bank's power whatsoever. Bank may, at its election, exercise, decline or fail to exercise, any right or remedy it may have against Borrower or any security held by Bank, including without limitation the right to foreclose upon any such security by judicial or nonjudicial sale, without affecting or impairing in any way the liability of Guarantor hereunder. Guarantor waives any defense arising by reason of any disability or other defense of Borrower or any other guarantor, or by reason of the cessation from any cause whatsoever of the liability of Borrower or any other guarantor. Guarantor waives any setoff, defense or counterclaim that Borrower may have against Bank, other than payment in full of the Obligations (other than inchoate indemnity obligations). Guarantor waives any defense arising

out of the absence, impairment or loss of any right of reimbursement or subrogation or any other rights against Borrower. Until all of the Obligations (other than contingent indemnity obligations for which no claim has been asserted) have been paid in full, (d) Guarantor shall have no right of subrogation or reimbursement for claims arising out of or in connection with this Agreement, (e) Guarantor shall have no right of contribution or other rights against Borrower in connection with the Obligations, (f) Guarantor waives any right to enforce any remedy that Bank now has or may hereafter have against Borrower, and (g) Guarantor waives all rights to participate in any security now or hereafter held by Bank. Guarantor waives all presentments, demands for performance, notices of nonperformance, protests, notices of protest, notices of dishonor and notices of acceptance of this Agreement and of the existence, creation or incurrence of new or additional Indebtedness. Guarantor assumes the responsibility for being and keeping itself informed of the financial condition of Borrower and of all other circumstances bearing upon the risk of nonpayment of any Indebtedness or nonperformance of any obligation of Borrower, warrants to Bank that it will keep so informed, and agrees that absent a request for particular information by Guarantor, Bank shall have no duty to advise Guarantor of information known to Bank regarding such condition or any such circumstances. Guarantor waives the benefits of California Civil Code sections 2809, 2810, 2819, 2845, 2847, 2848, 2849, 2850, 2899 and 3433.

1.5 If Borrower becomes insolvent, is adjudicated bankrupt or files a petition for reorganization, arrangement, composition or similar relief under any present or future provision of the United States Bankruptcy Code or reorganization or insolvency laws of any applicable jurisdiction, or if such a petition is filed against Borrower, and in any such proceeding some or all of any Indebtedness or obligations under the Loan Agreement are terminated or rejected or any obligation of Borrower is modified or abrogated, or if Borrower's obligations are otherwise avoided for any reason, Guarantor agrees that Guarantor's liability hereunder shall not thereby be affected or modified and such liability shall continue in full force and effect as if no such action or proceeding had occurred until payment in full of the Obligations (other than inchoate indemnity obligations). This Agreement shall continue to be effective or be reinstated, as the case may be, if any payment must be returned by Bank upon the insolvency, bankruptcy or reorganization of a Borrower or any other guarantor or otherwise, as though such payment had not been made until payment in full of the Obligations (other than inchoate indemnity obligations).

1.6 Any Indebtedness of Borrower now or hereafter held by Guarantor is hereby subordinated to any Indebtedness of Borrower to Bank; and such Indebtedness of Borrower to Guarantor shall be collected, enforced and received by Guarantor as trustee for Bank and be paid over to Bank on account of the Indebtedness of Borrower to Bank but without reducing or affecting in any manner the liability of Guarantor under the other provisions of this Agreement.

SECTION 2 – GRANT OF SECURITY INTEREST

2.1 To secure the payment and performance of all of the Guarantor Obligations when due, Guarantor hereby grants Bank, a continuing security interest in, and pledges to Bank, the Collateral (as defined below), wherever located, whether now owned or hereafter acquired or arising, and all proceeds and products thereof. Guarantor represents,

warrants, and covenants that the security interest granted herein is and shall at all times continue to be a first priority perfected security interest in the Collateral (subject only to Permitted Liens that are permitted to have superior priority to Bank's Lien under this Agreement). If Guarantor shall acquire a commercial tort claim in an amount greater than Fifty Thousand Dollars (\$50,000), Guarantor shall promptly notify Bank in a writing signed by Guarantor of the general details thereof and grant to Bank in such writing a security interest therein and in the proceeds thereof, all upon the terms of this Agreement, with such writing to be in form and substance reasonably satisfactory to Bank. "Collateral" means all of Guarantor's right, title and interest in and to the following:

- (a) All goods, Accounts (including health-care receivables), Equipment, Inventory, contract rights or rights to payment of money, leases, license agreements, franchise agreements, General Intangibles, commercial tort claims, documents, instruments (including any promissory notes), chattel paper (whether tangible or electronic), cash, deposit accounts, fixtures, letter-of-credit rights (whether or not the letter of credit is evidenced by a writing), securities, and all other investment property, supporting obligations, and financial assets, whether now owned or hereafter acquired, wherever located; and
- (b) all Guarantor's Books relating to the foregoing, and any and all claims, rights and interests in any of the above and all substitutions for, additions, attachments, accessories, accessions and improvements to and replacements, products, proceeds and insurance proceeds of any or all of the foregoing. For the purposes of this Agreement, "Guarantor's Books" are all Guarantor's books and records including ledgers, federal and state tax returns, records regarding Guarantor's assets or liabilities, the Collateral, business operations or financial condition, and all computer programs or storage or any equipment containing such information.
- (c) Notwithstanding anything to the contrary in this Agreement, the Collateral does not include any of the following:
 (i) more than 65% of the presently existing and hereafter arising issued and outstanding shares of capital stock owned by Guarantor of any Foreign Subsidiary which shares entitle the holder thereof to vote for directors or any other matter, (ii) any intent-to-use trademarks at all times prior to the first use thereof, whether by the actual use thereof in commerce, the recording of a statement of use with the United States Patent and Trademark Office or otherwise; (iii) any interest of Guarantor as a lessee or sublessee under a real property lease; (iv) rights held under a license that are not assignable by their terms without the consent of the licensor thereof (but only to the extent such restriction on assignment is enforceable under applicable law); or (v) any interest of Guarantor as a lessee under an Equipment lease if Guarantor is prohibited by the terms of such lease from granting a security interest in such lease or under which such an assignment or Lien would cause a default to occur under such lease; provided, however, that upon termination of such prohibition, such interest shall immediately become Collateral without any action by Guarantor or Bank.
- 2.2 Guarantor hereby authorizes Bank to file financing statements, without notice to Guarantor, with all appropriate jurisdictions to perfect or protect Bank's interest or rights hereunder. Such financing statements may indicate the Collateral as "all assets of the Debtor" or words of similar effect, or as being of an equal or lesser scope, or with greater detail, all in Bank's discretion.

SECTION 3 - PLEDGE

- 3.1 As security for the full, prompt and complete payment and performance when due (whether by stated maturity, by acceleration or otherwise) of all the Guarantor Obligations, Guarantor hereby pledges to Bank, and grants to Bank, a first priority security interest in all of the following (collectively, the "**Pledged Collateral**"):
- (a) the shares of capital stock or other equity securities of the entities listed on Exhibit A attached hereto, now owned or hereafter acquired (whether in connection with any recapitalization, reclassification, or reorganization of the capital of such entities or any successors in interest thereto) by Guarantor, subject to the limitation set forth in Section 2.1(c)(i) (the "Pledged Shares"), together with all proceeds and substitutions thereof, all cash, stock and other monies and property paid thereon, all rights to subscribe for securities declared or granted in connection therewith, and all other cash and noncash proceeds of the foregoing. On the date hereof, the certificate or certificates representing the Pledged Shares will be delivered to Bank, accompanied by an instrument of assignment duly executed in blank by Guarantor. To the extent required by the terms and conditions governing the Pledged Shares, Guarantor shall cause the books of each entity whose Pledged Shares are part of the Pledged Collateral and any transfer agent to reflect the pledge of the Pledged Shares. Upon the occurrence and during the continuation of an Event of Default, Bank may effect the transfer of any securities included in the Pledged Collateral (including but not limited to the Pledged Shares) into the name of Bank and cause new certificates representing such securities to be issued in the name of Bank or its transferee:
- (b) all voting trust certificates held by Guarantor evidencing the right to vote any Pledged Shares subject to any voting trust; and
- (c) all additional shares and voting trust certificates of the entities listed on Exhibit A from time to time acquired by Guarantor in any manner, subject to the limitation set forth in Section 2.1(c)(i) (which additional shares shall be deemed to be part of the Pledged Shares), and the certificates representing such additional shares, and all dividends, cash, instruments and other property or proceeds from time to time received, receivable or otherwise distributed in respect of or in exchange for any or all of such Pledged Shares.
- 3.2 Guarantor agrees to pay prior to delinquency all taxes, charges, Liens and assessments, in each case imposed by any Governmental Authority, against the Pledged Collateral, except those with respect to which the amount or validity is being contested in good faith by appropriate proceedings and with respect to which reserves in conformity with GAAP have been provided on the books of Guarantor, and upon the failure of Guarantor to do so, contemporaneous with written notice thereof from Bank to Guarantor, Bank at its option may pay any of them.
- 3.3 In the event that during the term of this Agreement, any reclassification, readjustment, or other change is declared or made in the capital structure of the issuer of the Pledged Shares, all new, substituted and additional shares, options, or other securities, issued or issuable to Guarantor by reason of any such change or exercise shall be delivered to and held by Bank under the terms of this Agreement in the same manner as the Pledged Collateral originally pledged hereunder.

3.4 Notwithstanding anything herein to the contrary, Guarantor may exercise any rights under the Pledged Shares to vote such Pledged Shares and receive dividends in respect of such Pledged Shares while no Event of Default has occurred and is continuing.

SECTION 4 – REPRESENTATIONS AND WARRANTIES

- 4.1 Guarantor hereby represents and warrants to Bank that:
- (a) the execution, delivery and performance by Guarantor of this Agreement (i) does not contravene any material Requirement of Law or any material contractual restriction binding on or affecting Guarantor or by which Guarantor's property may be affected; (ii) does not require any authorization or approval or other action by, or any notice to or filing with, any Governmental Authority or any other Person under any material indenture, mortgage, deed of trust, lease, agreement or other instrument to which Guarantor is a party or by which Guarantor or any of its property is bound, except such as have been obtained or made; and (iii) does not result in the imposition or creation of any Lien upon any property of Guarantor except the Lien provided to Bank pursuant to Section 2 hereof;
- (b) Guarantor has the corporate, limited liability or limited partnership, as applicable, power to execute, deliver and perform this Agreement and the execution, delivery and performance of this Agreement has been duly authorized by all requisite action;
- (c) this Agreement is a valid and binding obligation of Guarantor, enforceable against Guarantor in accordance with its terms, except as the enforceability thereof may be subject to or limited by bankruptcy, insolvency, reorganization, arrangement, moratorium or other similar laws relating to or affecting the rights of creditors generally;
- (d) there is no action, suit or proceeding affecting Guarantor pending or threatened before any court, arbitrator, or Governmental Authority, domestic or foreign, which may have a material adverse effect on the ability of Guarantor to perform its obligations under this Agreement;
 - (e) Guarantor's obligations hereunder are not subject to any offset or defense against Bank or Borrower of any kind;
- (f) in connection with this Agreement, Guarantor has delivered to Bank a completed certificate signed by Guarantor, entitled "**Perfection Certificate**," and Guarantor represents and warrants to Bank that, except as Bank may be notified pursuant to the terms of this Agreement, (a) Guarantor's exact legal name is that indicated on the Perfection Certificate and on the signature page hereof; (b) Guarantor is an organization of the type and is organized in the jurisdiction set forth in the Perfection Certificate; (c) the Perfection Certificate accurately sets forth Guarantor's organizational identification number or accurately states that Guarantor has none; (d) the Perfection Certificate accurately sets forth Guarantor's place of business, or, if more than one, its chief executive office as well as Guarantor's mailing address (if different than its chief executive office); (e) except as set forth in the Perfection Certificate,

Guarantor (and each of its predecessors) has not, in the past five (5) years, changed its jurisdiction of formation, organizational structure or type, or any organizational number assigned by its jurisdiction; and (f) all other information set forth on the Perfection Certificate pertaining to Guarantor and each of its Subsidiaries is accurate and complete (it being understood and agreed that Guarantor may from time to time update certain information in the Perfection Certificate after the date hereof to the extent permitted by one or more specific provisions in this Agreement);

- (g) the financial statements of Guarantor, copies of which have been and shall be furnished to Bank, fairly present in all material respects the financial position of Guarantor for the dates and periods purported to be covered thereby, and no Event of Default under section 8.11 of the Loan Agreement has occurred, is continuing or would occur upon the effectiveness of this Agreement;
- (h) after the incurrence of Guarantor's obligations under this Agreement, the fair salable value of Guarantor's assets (including goodwill minus disposition costs) exceeds the fair value of its liabilities; Guarantor is not left with unreasonably small capital after the transactions in this Agreement or the other Loan Documents; and Guarantor is able to pay its debts (including trade debts) as they mature;
- (i) all representations and warranties contained in this Agreement are true at the time of Guarantor's execution of this Agreement and, with respect to the representations and warranties contained in clauses (j) through (o) of this Section 4.1, shall be true upon the issuance and receipt by Guarantor of any additional Pledged Shares and/or Pledged Collateral after the date hereof;
- (j) Guarantor is, at the time of delivery of the Pledged Shares to Bank hereunder, the sole holder of record and the sole beneficial owner of the Pledged Collateral, free and clear of any Lien thereon or affecting title thereto, except for the Lien created by this Agreement and Permitted Liens;
- (k) none of the Pledged Shares have been transferred in violation of applicable federal or state securities, or similar laws, which such transfer may be subject to in the United States of America or any applicable jurisdiction;
 - (l) all of the Pledged Shares have been duly authorized, validly issued, fully paid and are non-assessable;
- (m) to Guarantor's knowledge, (i) there are no subscriptions, warrants, rights of first refusal or other restrictions on transfer relative to, or options exercisable with respect to the Pledged Shares; and (ii) the Pledged Shares are not the subject of any present or threatened suit, action, arbitration, administrative or other proceeding, and Guarantor knows of no reasonable grounds for the institution of any such proceedings;
- (n) no consent, approval, authorization or other order of any Person and no consent or authorization of any Governmental Authority is required to be made or obtained which has not yet been made or obtained by Guarantor either (i) for the pledge by Guarantor of the Pledged Collateral pursuant to this Agreement or for the execution, delivery, or

performance of this Agreement by Guarantor; or (ii) for the exercise by Bank of the voting or other rights provided for in this Agreement or the remedies with respect to the Pledged Collateral pursuant to this Agreement, except, in the case of clause (i) and (ii), as (x) may be required in connection with such disposition by laws affecting the offer and sale of securities generally and (y) by the laws of the United States and any applicable foreign jurisdiction; and

(o) the pledge, grant of a security interest in, and delivery of the Pledged Collateral pursuant to this Agreement, will create a valid first priority Lien on and in the Pledged Collateral, and the proceeds thereof, securing the payment of the Guarantor Obligations.

SECTION 5 - COVENANTS

- 5.1 Guarantor covenants and agrees to the following:
- (a) Deliver to Bank, (i) within five (5) days of filing, copies of all periodic and other reports, proxy statements and other materials filed by Guarantor with the SEC, any Governmental Authority succeeding to any or all of the functions of the SEC or with any national securities exchange, or distributed to its shareholders, as the case may be. Documents required to be delivered pursuant to the terms hereof (to the extent any such documents are included in materials otherwise filed with the SEC) may be delivered electronically and if so delivered, shall be deemed to have been delivered on the date on which Guarantor posts such documents, or provides a link thereto, on Guarantor's website on the Internet at Guarantor's website address, (ii) a prompt report of any legal actions pending or threatened in writing against Guarantor or any of its Subsidiaries that could result in damages or costs to Guarantor or any of its Subsidiaries of, individually or in the aggregate, Two Hundred Fifty Thousand Dollars (\$250,000) or more, and (iii) financial information reasonably requested by Bank;
- (b) Timely file, and require each of its Subsidiaries to timely file, all required tax returns and reports and timely pay, and require each of its Subsidiaries to timely pay, all foreign, federal, state and material local taxes, assessments, deposits and contributions owed by Guarantor and each of its Subsidiaries, except for deferred payment of any contested taxes, provided that Guarantor (i) in good faith contests its obligation to pay the taxes by appropriate proceedings promptly and diligently instituted and conducted, (ii) notifies Bank in writing of the commencement of, and any material development in, the proceedings, (iii) posts bonds or takes any other steps required to prevent the governmental authority levying such contested taxes from obtaining a Lien upon any of the Collateral that is other than a "Permitted Lien," and shall deliver to Bank, on reasonable demand, appropriate certificates attesting to such payments, and pay all amounts necessary to fund all present pension, profit sharing and deferred compensation plans in accordance with their terms;
- (c) Keep its business and the Collateral insured for risks and in amounts standard for companies in Guarantor's industry and location. Insurance policies shall be in a form, with companies, and in amounts standard for companies in Guarantor's industry and location. All property policies shall have a lender's loss payable endorsement showing Bank as a lender loss payee and waive subrogation against Bank. All liability policies shall show, or have endorsements showing, Bank as an additional insured. [All policies (or their respective endorsements) shall provide that the insurer shall give Bank at least twenty (20) days' notice

before canceling, amending, or declining to renew its policy.][NTD: Guarantor to confirm with its insurer that such endorsements can be obtained] At Bank's reasonable request, Guarantor shall deliver certified copies of policies and evidence of all premium payments. If an Event of Default exists, proceeds payable under any policy shall, at Bank's option, be payable to Bank on account of the Obligations. After the occurrence and during the continuance of an Event of Default, all proceeds payable under such casualty policy shall, at the option of Bank, be payable to Bank on account of the Guarantor Obligations. If Guarantor fails to obtain insurance as required under this Section 5.1(c) or to pay any amount or furnish any required proof of payment to third persons and Bank, Bank may make all or part of such payment or obtain such insurance policies required in this Section 5.1(c), and take any action under the policies Bank deems prudent;

- (d) Provide Bank five (5) days prior written notice before establishing any Collateral Account at or with any bank or financial institution other than Bank or Bank's Affiliates. For each Collateral Account that Guarantor at any time maintains in the United States, Guarantor shall cause the applicable bank or financial institution (other than Bank) at or with which any Collateral Account is maintained to execute and deliver a Control Agreement or other appropriate instrument with respect to such Collateral Account to perfect Bank's Lien in such Collateral Account in accordance with the terms hereunder which Control Agreement may not be terminated without the prior written consent of Bank unless this Agreement and Bank's Liens hereunder are terminated. The provisions of the previous sentence shall not apply to (i) deposit accounts exclusively used for payroll, payroll taxes and other employee wage and benefit payments to or for the benefit of Guarantor's employees and identified to Bank by Guarantor as such, or (ii) other Collateral Accounts holding balances not in excess of Seven Hundred Fifty Thousand Dollars (\$750,000) in the aggregate at any time;
- (e) (i) Protect, defend and maintain the validity and enforceability of its Intellectual Property material to its business; (ii) promptly advise Bank in writing of material infringements of its Intellectual Property material to its business; (iii) not allow any Intellectual Property material to Guarantor's business to be abandoned, forfeited or dedicated to the public without Bank's written consent, and (iv) provide written notice to Bank (which may be delivered electronically and if so delivered, shall be deemed to have been delivered on the date on which Guarantor posts such documents, or provides a link thereto, on Guarantor's website on the Internet at Guarantor's website address) within ten (10) days of entering or becoming bound by any Restricted License (other than over-the-counter software that is commercially available to the public). Guarantor shall take such commercially reasonable steps as Bank requests to obtain the consent of, or waiver by, any person whose consent or waiver is necessary for (A) any Restricted License to be deemed "Collateral" and for Bank to have a security interest in it that might otherwise be restricted or prohibited by law or by the terms of any such Restricted License, whether now existing or entered into in the future, and (B) Bank to have the ability in the event of a liquidation of any Collateral to dispose of such Collateral in accordance with Bank's rights and remedies under this Agreement and the other Loan Documents;
- (f) Allow Bank, or its agents, at reasonable times, on three (3) Business Days' notice (provided no notice is required if an Event of Default has occurred and is continuing), to inspect the Collateral and audit and copy Guarantor's Books. Such inspections or audits shall be conducted no more often than once every twelve (12) months unless an Event of

Default has occurred and is continuing. The foregoing inspections and audits shall be at Guarantor's expense, and the charge therefor shall be \$850 per person per day (or such higher amount as shall represent Bank's then-current standard charge for the same), plus reasonable out-of-pocket expenses. In the event Guarantor and Bank schedule an audit more than ten (10) days in advance, and Guarantor cancels or seeks to reschedule the audit with less than ten (10) days written notice to Bank, then (without limiting any of Bank's rights or remedies), Guarantor shall pay Bank a fee of \$1,000 plus any out-of-pocket expenses incurred by Bank to compensate Bank for the anticipated costs and expenses of the cancellation or rescheduling;

- (g) Guarantor shall, at any time and from time to time, execute and deliver such further instruments and take such further action as may reasonably be requested by Bank to effect the purposes of this Agreement;
- (h) Guarantor shall make any payments and do any act reasonably necessary or convenient to protect the Collateral and the value of Bank's interest therein. If Guarantor fails to pay any amount or furnish any required proof of payment to third Persons as required hereunder, Bank may make all or part of the payment or obtain insurance policies, and take any action under the policies as Bank deems prudent. Any amounts paid by Bank are Bank Expenses and immediately due and payable, bearing interest at the Default Rate and secured by the Collateral. No payments by Bank are an agreement to make similar payments in the future or Bank's waiver of any Event of Default; and
- (i) (i) Maintain its and all its Subsidiaries' legal existence and good standing in their respective jurisdictions of formation and maintain qualification in each jurisdiction in which the failure to so qualify would reasonably be expected to have a material adverse effect on Guarantor's business or operations; (ii) comply, and have each Subsidiary comply, with all laws, ordinances and regulations to which it is subject, noncompliance with which could have a material adverse effect on Guarantor's business; (iii) obtain all of the Governmental Approvals necessary for the performance by Guarantor of its obligations under the Loan Documents to which it is a party and the grant of a security interest to Bank in all of its property; and (iv) promptly provide copies of any such obtained Governmental Approvals to Bank.
 - 5.2 Guarantor shall not do any of the following without Bank's prior written consent:
- (a) Convey, sell, lease, transfer, assign, or otherwise dispose of (collectively, "Transfer"), or permit any of its Subsidiaries to Transfer, all or any part of its business or property, except for Transfers (a) of Inventory in the ordinary course of business; (b) of worn-out or obsolete Equipment or Equipment that is no longer useful for Guarantor's business; (c) in connection with Permitted Liens and Investments that are not otherwise prohibited under this Agreement or the other Loan Documents; (d) of non-exclusive licenses for the use of the property of Guarantor or its Subsidiaries in the ordinary course of business and licenses that could not result in a legal transfer of title of the licensed property but that may be exclusive in respects other than territory and that may be exclusive as to territory only as to discreet geographical areas outside of the United States; and (e) consisting of Guarantor's use or transfer of money or Cash Equivalents in the ordinary course of business in a manner that is not prohibited by the terms of this Agreement or the other Loan Documents;

- (b) (i) Liquidate or dissolve; (ii) permit or suffer a Change in Control, or (iii) permit Borrower to cease being a wholly-owned Subsidiary of Guarantor;
- (c) Guarantor shall not, without at least thirty (30) days prior written notice to Bank: (1) add any new offices or business locations, including warehouses (unless such new offices or business locations contain less than One Hundred Thousand Dollars (\$100,000) in Guarantor's assets or property) or deliver any portion of the Collateral valued, individually or in the aggregate, in excess of One Hundred Thousand Dollars (\$100,000) to a bailee at a location other than to a bailee and at a location already disclosed in the Perfection Certificate, (2) change its jurisdiction of organization, (3) change its organizational structure or type, (4) change its legal name, or (5) change any organizational number (if any) assigned by its jurisdiction of organization. If Guarantor intends to deliver any portion of the Collateral valued, individually or in the aggregate, in excess of One Hundred Thousand Dollars (\$100,000) to a bailee, and Bank and such bailee are not already parties to a bailee agreement governing both the Collateral and the location to which Guarantor intends to deliver the Collateral, then Guarantor will first receive the written consent of Bank, and use commercially reasonable efforts to cause such bailee to execute and deliver a bailee agreement in form and substance satisfactory to Bank in its reasonable discretion;
- (d) Merge or consolidate, or permit any of its Subsidiaries to merge or consolidate, with any other Person. A Subsidiary may merge or consolidate into another Subsidiary or into Guarantor;
- (e) Create, incur, assume, or be liable for any Indebtedness, or permit any Subsidiary to do so, other than Permitted Indebtedness;
- (f) Create, incur, allow, or suffer any Lien on any of its property, or assign or convey any right to receive income, including the sale of any Accounts, or permit any of its Subsidiaries to do so, except for Permitted Liens, permit any Collateral not to be subject to the first priority security interest granted herein, or enter into any agreement, document, instrument or other arrangement (except with or in favor of Bank or in connection with Permitted Liens) with any Person which directly or indirectly prohibits or has the effect of prohibiting Guarantor from assigning, mortgaging, pledging, granting a security interest in or upon, or encumbering any of Guarantor's Intellectual Property, except, in each case, as is otherwise permitted in Section 5.2(a) hereof and the definition of "Permitted Liens" herein;
 - (g) Maintain any Collateral Account except pursuant to the terms of Section 5.1(d) hereof;
- (h) Pay (other than in the form of Guarantor's capital stock) any dividends or make any distribution or payment or redeem, retire or purchase any capital stock; provided, however, that Guarantor may repurchase capital stock from former directors, officers, employees or consultants at the original purchase price thereof, but not to exceed in the aggregate of Five Hundred Thousand Dollars (\$500,000) per fiscal year, as permitted by

Guarantor's equity incentive plans, restricted stock purchase agreements, stock option agreements, stock grant agreements or similar agreements, so long as an Event of Default does not exist at the time of such repurchase and would not exist after giving effect to such repurchase;

- (i) (i) Make or permit any payment on any Subordinated Debt, except under the terms of the subordination, intercreditor, or other similar agreement to which such Subordinated Debt is subject, or (ii) amend any provision in any document relating to the Subordinated Debt which would increase the amount thereof or adversely affect the subordination thereof to Guarantor Obligations owed to Bank, except as may be permitted under the terms of the subordination, intercreditor, or other similar agreement to which such Subordinated Debt is subject;
- (j) Become an "investment company" or a company controlled by an "investment company", under the Investment Company Act of 1940, as amended, or undertake as one of its important activities extending credit to purchase or carry margin stock (as defined in Regulation U of the Board of Governors of the Federal Reserve System), or use the proceeds of any Credit Extension for that purpose; fail to meet the minimum funding requirements of ERISA, permit a Reportable Event or Prohibited Transaction, as defined in ERISA, to occur; fail to comply with the Federal Fair Labor Standards Act or violate any other law or regulation, if the violation could reasonably be expected to have a material adverse effect on Guarantor's business, or permit any of its Subsidiaries to do so; withdraw or permit any Subsidiary to withdraw from participation in, permit partial or complete termination of, or permit the occurrence of any other event with respect to, any present pension, profit sharing and deferred compensation plan which could reasonably be expected to result in any material liability of Guarantor, including any liability to the Pension Benefit Guaranty Corporation or its successors or any other governmental agency; or
- (k) Form any direct or indirect Subsidiary or acquire any direct or indirect Subsidiary after the date hereof, unless Guarantor (a) causes any such new Domestic Subsidiary to provide to Bank a joinder to this Agreement, together with such appropriate financing statements and/or Control Agreements, all in form and substance reasonably satisfactory to Bank (including being sufficient to grant Bank a first priority Lien (subject to Permitted Liens) in and to the assets of such newly formed or acquired Domestic Subsidiary), (b) provides to Bank appropriate certificates and powers and financing statements, pledging all of the direct or beneficial ownership interest in any such new Subsidiary (except to the extent shares of any such new Foreign Subsidiary are excluded from the Collateral pursuant to Section 2.1(c)(i)), in form and substance reasonably satisfactory to Bank, and (c) provides to Bank all other documentation in form and substance reasonably satisfactory to Bank which in its opinion is appropriate with respect to the execution and delivery of the applicable documentation referred to above. Any document, agreement, or instrument executed or issued pursuant to this Section 5.2(k) shall be a Loan Document.
- 5.3 So long as Bank has any commitment to make Credit Extensions to any Borrower under the Loan Agreement or any Borrower has any Obligations (other than contingent indemnity obligations for which no claim has been asserted) outstanding under the Loan Documents, Guarantor agrees that Guarantor:

- (a) will not (i) except as otherwise permitted by the Loan Documents, sell, transfer or otherwise dispose of, or grant any option or warrant with respect to, any of the Pledged Collateral (or any part thereof or interest therein) except with the prior written consent of Bank, or (ii) create or permit to exist any Lien upon or with respect to any of the Pledged Collateral, except for Permitted Liens. If any Pledged Collateral, or any part thereof, is sold, transferred or otherwise disposed of in violation of this Section 5.3, the security interest of Bank shall continue in the Pledged Collateral, to the extent permitted under applicable law, notwithstanding such sale, transfer or other disposition, and Guarantor will deliver any proceeds thereof to Bank to be held as Pledged Collateral hereunder;
- (b) shall, at Guarantor's own expense, promptly execute, acknowledge, and deliver all such instruments and take all such actions as Bank from time to time may reasonably request to ensure to Bank the benefits of the Lien in and to the Pledged Collateral intended to be created by this Agreement;
- (c) shall maintain, preserve and, at Bank's request, use commercially reasonable efforts to defend the title to the Pledged Collateral and the Lien of Bank thereon against the claim of any other Person; and
- (d) upon obtaining any shares of capital stock or other equity securities that should be pledged pursuant to Section 3.1 of this Agreement, shall promptly deliver to Bank a duly executed Pledge Supplement in substantially the form of Exhibit B attached hereto (a "Pledge Supplement") identifying such additional shares of capital stock or other equity securities. Guarantor hereby authorizes Bank to attach each Pledge Supplement to this Agreement and agrees that all shares of capital stock or other equity securities listed thereon shall for all purposes hereunder constitute Pledged Collateral.

SECTION 6 - EVENTS OF DEFAULT

- 6.1 Upon the occurrence and during the continuation of an Event of Default, Bank shall have all of the rights of a secured party under the Code with respect to the Collateral. Guarantor's obligations hereunder are not limited to the Collateral or any exercise by Bank of rights and remedies against the same, and Bank may pursue any other available rights and remedies against Guarantor, whether hereunder, at law or otherwise, without resort to the Collateral if Bank deems it in its best interests to do so.
 - 6.2 So long as no Event of Default has occurred and is continuing:
- 6.2.1 Guarantor shall be entitled to exercise any and all voting and other consensual rights pertaining to the Pledged Collateral or any part thereof; and
- 6.2.2 Guarantor shall be entitled to receive and to retain and use free and clear of the Lien created hereby any and all payments, including, but not limited to, profits, dividends or distributions paid in respect of its Pledged Collateral; *provided*, *however*, that any and all:

(a) non-cash profits, non-cash dividends or distributions in the form of capital stock, instruments or other property received, received, receivable or otherwise distributed in respect of, or in exchange for, any Pledged Collateral, and

(b) profits, dividends and other distributions paid or payable in cash in respect of any Pledged Collateral in connection with total liquidation or dissolution,

shall, promptly be delivered to Bank, in the case of (a) above, to be held as Collateral and shall, if received by Guarantor, be received in trust for the benefit of Bank, be segregated from the other property of Guarantor, and promptly be delivered to Bank as Pledged Collateral in the same form as so received (with any necessary endorsement), and in the case of (b) above, to be applied to the Obligations to the extent permitted by the Loan Agreement or otherwise to be held as Pledged Collateral.

6.3 During the existence and continuation of an Event of Default,

(a) Bank may, to the extent permitted by applicable law, at its election, apply, set off, collect or sell in one or more sales, or take such steps as may be necessary to liquidate and reduce to cash in the hands of Bank in whole or in part, with or without any previous demands or demand of performance or notice or advertisement, the whole or any part of the Pledged Collateral in such order as Bank may elect, and any such sale may be made either at public or private sale at its place of business or elsewhere, or at any broker's board or securities exchange, either for cash or upon credit or for future delivery; provided, however, that if such disposition is at private sale, then the purchase price of the Pledged Collateral shall be equal to the public market price then in effect, or, if at the time of sale no public market for the Pledged Collateral exists, then, in recognition of the fact that the sale of the Pledged Collateral would have to be registered under the Securities Act of 1933, as amended (the "Act"), and that the expenses of such registration are commercially unreasonable for the type and amount of collateral pledged hereunder, Bank and Guarantor hereby agree that such private sale shall be at a purchase price mutually agreed to by Bank and Guarantor or, if the parties cannot agree upon a purchase price, then at a purchase price established by Bank in the exercise of its reasonable discretion. Bank shall be under no obligation to delay the sale of any of the Pledged Shares for the period of time necessary to permit Guarantor to register such securities for public sale under the Act, or under applicable state securities laws, even if Guarantor would agree to do so. Bank may be the purchaser of any or all Pledged Collateral so sold and hold the same thereafter in its own right free from any claim of Guarantor or right of redemption. To the extent permitted by applicable law, demands of performance, notices of sale, advertisements and presence of property at sale are hereby waived. Any sale hereunder m

(b) Bank shall have all of the rights of a secured party under the Code with respect to the Pledged Collateral. Guarantor's obligations hereunder are not limited to the Pledged Collateral or any exercise by Bank of rights and remedies against the same, and Bank may pursue any other available rights and remedies against any Guarantor, whether hereunder, at law or otherwise, without resort to the Pledged Collateral if Bank deems it in its best interests to do so.

- 6.4 During the existence of an Event of Default, the proceeds of the sale of any of the Pledged Collateral and all sums received or collected by Bank from or on account of such Pledged Collateral under Section 6.3 shall be applied by Bank to the payment of expenses incurred or paid by Bank in connection with any sale, transfer or delivery of the Pledged Collateral, to the payment of any other costs, charges, attorneys' fees or expenses mentioned herein, and to the payment of the Obligations or any part hereof, all in such order and manner as Bank in its discretion may determine.
- 6.5 Upon the transfer by Bank of all or any part of the Obligations pursuant to the terms of the Loan Agreement, Bank may transfer all or any part of the Pledged Collateral to the transferee of the Obligations and shall be fully discharged thereafter from all liability and responsibility with respect to such Pledged Collateral so transferred, and the transferee shall be vested with all the rights and powers of Bank hereunder with respect to such Pledged Collateral so transferred; but with respect to any Pledged Collateral not so transferred, Bank shall retain all rights and powers hereby given.

SECTION 7 - DEFINITIONS

- 7.1 "Change in Control" means any event, transaction, or occurrence as a result of which (a) any "person" (as such term is defined in Sections 3(a)(9) and 13(d)(3) of the Exchange Act), other than a trustee or other fiduciary holding securities under an employee benefit plan of Guarantor, is or becomes a beneficial owner (within the meaning Rule 13d-3 promulgated under the Exchange Act), directly or indirectly, of securities of Guarantor, representing fifty percent (50%) or more of the combined voting power of Guarantor's then outstanding securities; or (b) during any period of twelve (12) consecutive calendar months, individuals who at the beginning of such period constituted the Board of Directors of Guarantor (together with any new directors whose election by the Board of Directors of Guarantor was approved by a vote of not less than two-thirds of the directors then still in office who either were directors at the beginning of such period or whose election or nomination for election was previously so approved) cease for any reason other than death or disability to constitute a majority of the directors then in office.
- 7.2 "Control Agreement" is any control agreement entered into among the depository institution at which Guarantor maintains a Deposit Account or the securities intermediary or commodity intermediary at which Guarantor maintains a Securities Account or a Commodity Account, Guarantor, and Bank pursuant to which Bank obtains control (within the meaning of the Code) over such Deposit Account, Securities Account, or Commodity Account.
- 7.3 "**Domestic Subsidiary**" means any Subsidiary which is organized under the laws of the United States or any state or territory thereof or the District of Columbia
 - 7.4 "Foreign Subsidiary" means any Subsidiary which is not a Domestic Subsidiary.
 - 7.5 "Intellectual Property" means all of Guarantor's right, title, and interest in and to the following:
 - (a) its Copyrights, Trademarks and Patents;

- (b) any and all trade secrets and trade secret rights, including, without limitation, any rights to unpatented inventions, know-how, operating manuals;
 - (c) any and all source code;
 - (d) any and all design rights which may be available to Guarantor;
- (e) any and all claims for damages by way of past, present and future infringement of any of the foregoing, with the right, but not the obligation, to sue for and collect such damages for said use or infringement of the Intellectual Property rights identified above; and
 - (f) all amendments, renewals and extensions of any of the Copyrights, Trademarks or Patents.

7.6 "Permitted Indebtedness" is:

- (a) Guarantor's Indebtedness to Bank under this Agreement and the other Loan Documents;
- (b) Subordinated Debt;
- (c) unsecured Indebtedness to trade creditors and corporate credit cards incurred in the ordinary course of business;
- (d) Indebtedness incurred as a result of endorsing negotiable instruments received in the ordinary course of business;
- (e) Indebtedness secured by Liens permitted under clauses (a) and (c) of the definition of "Permitted Liens" hereunder;
- (f) extensions, refinancings, modifications, amendments and restatements of any items of Permitted Indebtedness (a) through (e) above, provided that the principal amount thereof is not increased or the terms thereof are not modified to impose more burdensome terms upon Guarantor or its Subsidiary, as the case may be;
- (g) Indebtedness (including without limitation, Contingent Obligations) to North Atlantic not to exceed Eight Million Dollars (\$8,000,000) in principal amount, plus accrued interest and fees; and
- (h) Indebtedness of Foreign Subsidiaries of Guarantor so long as Guarantor is not an obligor or credit party with respect to any such Indebtedness and no recourse to Guarantor or any of its property may be had by the creditors with respect to such Indebtedness.

7.7 "Permitted Liens" are:

(a) Liens arising under this Agreement and the other Loan Documents;

- (b) Liens for taxes, fees, assessments or other government charges or levies, either (i) not due and payable or (ii) being contested in good faith and for which Guarantor maintains adequate reserves on its books, provided that no notice of any such Lien has been filed or recorded under the Internal Revenue Code of 1986, as amended, and the Treasury Regulations adopted thereunder;
- (c) purchase money Liens (i) on Equipment acquired or held by Guarantor incurred for financing the acquisition of the Equipment securing no more than One Million Dollars (\$1,000,000) in the aggregate amount outstanding, or (ii) existing on Equipment when acquired, if the Lien is confined to the property and improvements and the proceeds of the Equipment;
- (d) Liens of carriers, warehousemen, suppliers, or other Persons that are possessory in nature arising in the ordinary course of business so long as such Liens attach only to Inventory and Equipment and which are not delinquent or remain payable without penalty or which are being contested in good faith and by appropriate proceedings which proceedings have the effect of preventing the forfeiture or sale of the property subject thereto;
- (e) Liens to secure payment of workers' compensation, employment insurance, old-age pensions, social security and other like obligations incurred in the ordinary course of business (other than Liens imposed by ERISA);
- (f) Liens incurred in the extension, renewal or refinancing of the indebtedness secured by Liens described in (a) and (c), but any extension, renewal or replacement Lien must be limited to the property encumbered by the existing Lien and the principal amount of the indebtedness may not increase;
- (g) leases or subleases of real property granted in the ordinary course of Guarantor's business (or, if referring to another Person, in the ordinary course of such Person's business), and leases, subleases, non-exclusive licenses or sublicenses of personal property (other than Intellectual Property) granted in the ordinary course of Guarantor's business (or, if referring to another Person, in the ordinary course of such Person's business), if the leases, subleases, licenses and sublicenses do not prohibit granting Bank a security interest therein;
 - (h) non-exclusive licenses of Intellectual Property granted to third parties in the ordinary course of business;
- (i) licenses of Intellectual Property that could not result in a legal transfer of title of the licensed property that may be exclusive in respects other than territory;
 - (j) Liens arising from attachments or judgments, orders, or decrees in circumstances not constituting an Event of Default;
- (k) Liens in favor of other financial institutions arising in connection with Guarantor's deposit and/or securities accounts held at such institutions, provided that Bank has a perfected security interest in the amounts held in such deposit and/or securities accounts; and

- (l) Liens in favor of North Atlantic in connection with Indebtedness permitted under clause (g) of the definition of "Permitted Indebtedness" hereunder.
- 7.8 "**Restricted License**" is any material license or other agreement with respect to which Guarantor is the licensee (a) that prohibits or otherwise restricts Guarantor from granting a security interest in Guarantor's interest in such license or agreement or any other property, or (b) for which a default under or termination of could interfere with the Bank's right to sell any Collateral.
- 7.9 "**Subordinated Debt**" is indebtedness incurred by Guarantor subordinated to all of Guarantor's now or hereafter indebtedness to Bank (pursuant to a subordination, intercreditor, or other similar agreement in form and substance reasonably satisfactory to Bank entered into between Bank and the other creditor), on terms acceptable to Bank.

SECTION 8 - MISCELLANEOUS

- 8.1 Guarantor agrees to pay reasonable attorneys' fees and all other costs and expenses which may be incurred by Bank in the enforcement of this Agreement. No terms or provisions of this Agreement may be changed, waived, revoked or amended without Bank's prior written consent. Should any provision of this Agreement be determined by a court of competent jurisdiction to be unenforceable, all of the other provisions shall remain effective. This Agreement embodies the entire agreement between the parties hereto with respect to the matters set forth herein, and supersedes all prior agreements among the parties with respect to the matters set forth herein. No course of prior dealing among the parties, no usage of trade, and no parole or extrinsic evidence of any nature shall be used to supplement, modify or vary any of the terms hereof. Bank may assign this Agreement without in any way affecting Guarantor's liability under it. This Agreement shall inure to the benefit of Bank, Guarantor and their respective successors and assigns. This Agreement is in addition to the guaranties of any other guarantors of the Obligations. Notwithstanding the foregoing, so long as no Event of Default shall have occurred and be continuing, Bank shall not assign its interest in this Agreement to any Person who in the reasonable estimation of Bank is (a) a direct competitor of Guarantor, whether as an operating company or direct or indirect parent with voting control over such operating company, or (b) a vulture fund or distressed debt fund.
- 8.2 North Carolina law governs this Agreement without regard to principles of conflicts of law. Guarantor and Bank each submit to the exclusive jurisdiction of the State and Federal courts in Wake County, North Carolina, provided, however, that if for any reason Bank cannot avail itself of such courts in the State of North Carolina, Guarantor accepts jurisdiction of the courts and venue in Santa Clara County, California; provided, further, that nothing in this Agreement shall be deemed to operate to preclude Bank from bringing suit or taking other legal action in any other jurisdiction to realize on the Collateral or any other security for the Obligations, or to enforce a judgment or other court order in favor of Bank. Guarantor expressly submits and consents in advance to such jurisdiction in any action or suit commenced in any such court, and Guarantor hereby waives any objection that it may have based upon lack of personal jurisdiction, improper venue, or forum non conveniens and hereby consents to the granting of such legal or equitable relief as is deemed appropriate by such court. Guarantor hereby waives personal service of the summons, complaints, and other process issued in such

action or suit and agrees that service of such summons, complaints, and other process may be made by registered or certified mail addressed to Guarantor at the address set forth on the signature page hereto and that service so made shall be deemed completed upon the earlier to occur of Guarantor's actual receipt thereof or three (3) days after deposit in the U.S. mails, proper postage prepaid.

TO THE EXTENT PERMITTED BY APPLICABLE LAW, GUARANTOR AND BANK EACH WAIVE THEIR RIGHT TO A JURY TRIAL OF ANY CLAIM OR CAUSE OF ACTION ARISING OUT OF OR BASED UPON THIS AGREEMENT, THE OTHER LOAN DOCUMENTS OR ANY CONTEMPLATED TRANSACTION, INCLUDING CONTRACT, TORT, BREACH OF DUTY AND ALL OTHER CLAIMS. THIS WAIVER IS A MATERIAL INDUCEMENT FOR BOTH PARTIES TO ENTER INTO THIS AGREEMENT. EACH PARTY HAS REVIEWED THIS WAIVER WITH ITS COUNSEL.

8.3 This Agreement may be executed in counterpart signature pages, all of which taken together shall be deemed to be one original of this instrument. Delivery of an executed counterpart to this Agreement by facsimile or electronic mail shall be effective as a manually executed counterpart to this Agreement.

8.4 All covenants, representations and warranties made in this Agreement continue in full force until this Agreement has terminated pursuant to its terms and all Obligations and/or Guarantor Obligations have been satisfied. So long as Borrower and Guarantor have satisfied the Obligations and Guarantor Obligations, respectively (other than inchoate indemnity obligations, any other obligations which, by their terms, are to survive the termination of this Agreement, and any Obligations under Bank Services Agreements that are cash collateralized in accordance with Section 4.1 of the Loan Agreement), this Agreement shall be deemed terminated, subject to Section 1.5. Those obligations that are expressly specified in this Agreement as surviving this Agreement's termination shall continue to survive notwithstanding this Agreement's termination.

[Signature begins on next page.]

IN WITNESS WHEREOF, the undersigned Guarantor has executed this Unconditional Secured Guaranty and Pledge Agreement as of this day of November, 2014.

MANDALAY DIGITAL GROUP, INC., a Delaware corporation				
Ву:				
Name:				
Title:				
Address:				
SILICON VALLEY BANK				
Ву:				
Name:				
Title:				
Address: 3005 Carrington Mill Blvd., Suite 530				

Attn: Chris Stoecker

EXHIBIT A – PLEDGED SHARES

					% of	% of	
Name of		Number of	Number of	Number of	Shares	Outstanding	
Pledged Share	Jurisdiction of	Shares	Shares	Shares Owned	Owned by	Shares	Certificate
Issuer	Organization	Authorized	Outstanding	by Guarantor	Guarantor	Pledged	Number

Appia, Inc.

EXHIBIT B – FORM OF PLEDGE SUPPLEMENT

PLEDGE SUPPLEMENT

This Pledge Supplement, dated as of, 20, is delivered pursuant to Se referred to below. The undersigned hereby agrees that this Pledge Supplement may be Pledge Agreement, dated as of, 2014 (as amended, restated, modified, renew "Guaranty and Pledge Agreement"; the terms defined therein and not otherwise defundersigned, as Guarantor, in favor of Silicon Valley Bank, as Bank.	be attached to the Unconditional Secured Guaranty and red, supplemented or extended from time to time, the			
The shares of capital stock or other equity securities listed on this Pledge Suppledged by the undersigned and referred to in the Guaranty and Pledge Agreement at				
The undersigned hereby certifies that the representation and warranties set forth in Section 4.1 of the Guaranty and Pledge Agreement are true and correct in all material respects with respect to the Pledged Shares listed below on and as of the date hereof.				
	MANDALAY DIGITAL GROUP, INC., a Delaware corporation			
	Ву:			
	Name:			
	Title:			

Number of

Shares

Outstanding

Number of

Shares

Issued

Number of

Shares

Authorized

Name of

Pledged Share

Issuer

Jurisdiction of

Organization

% of

Outstanding

Shares

Pledged

Certificate

Number

Number of

Shares

Owned by

Guarantor

CONSENT OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM

We consent to the incorporation by reference in Amendment No. 1 to Registration Statement (No. 333-200695) on Form S-4 of Mandalay Digital Group, Inc. (the "Company") of our report dated June 30, 2014, relating to our audit of the consolidated financial statements of the Company and Subsidiaries, which appears in the Annual Report on Form 10-K of the Company for the years ended March 31, 2014 and 2013.

We also consent to the reference to our Firm under the caption "Experts" in the Prospectus, which is part of this Registration Statement.

/s/ SingerLewak LLP

Los Angeles, California January 16, 2015

CONSENT OF INDEPENDENT CERTIFIED PUBLIC ACCOUNTANTS

We have issued our report dated June 26, 2014, with respect to the consolidated financial statements of Appia, Inc. as of December 31, 2013 and 2012 and for the years then ended, contained in the Registration Statement and Prospectus of Mandalay Digital Group, Inc. We consent to the use of the aforementioned report in the Registration Statement and Prospectus, and to the use of our name as it appears under the caption "Experts."

/s/ Grant Thornton LLP

Raleigh, North Carolina January 16, 2015

MANDALAY DIGITAL GROUP, INC.

Annual Meeting of Stockholders
[_____], 2015 [____] a.m. Pacific Time

THIS PROXY IS SOLICITED BY THE BOARD OF DIRECTORS

The undersigned stockholder of Mandalay Digital Group, Inc., a Delaware corporation (the "Company"), hereby acknowledges receipt
of the Notice of Annual Meeting of Stockholders and Proxy Statement, each dated [], 2015, and hereby appoints William G. Stone and
Andrew Schleimer, and each of them, as proxy and attorney-in-fact of the undersigned, each with full power of substitution, and hereby
authorizes each of them to represent the undersigned and vote all of the shares of common stock of the Company held of record by the
undersigned on [], 2015, at the Annual Meeting of Stockholders (the "Annual Meeting") to be held at the offices of Manatt, Phelps &
Phillips, LLP, 11355 West Olympic Boulevard, Los Angeles, California 90064, on [], 2015 at [] a.m. Pacific Time, and at any and
all postponements or adjournments thereof, as fully as the undersigned might do if personal present thereat, on the matters set forth in the
Notice of Annual Meeting of Stockholders and Proxy Statement.

THIS PROXY, WHEN PROPERLY EXECUTED AND RETURNED IN A TIMELY MANNER, WILL BE VOTED AT THE ANNUAL MEETING, AND AT ANY ADJOURNMENTS OR POSTPONEMENTS THEREOF, IN THE MANNER HEREIN DIRECTED. IF NO DIRECTION IS INDICATED, THE PROXY WILL BE VOTED <u>FOR</u> THE AUTHORIZATION TO ISSUE SHARES OF THE COMPANY'S COMMON STOCK PURSUANT TO THE MERGER AGREEMENT WITH APPIA, INC. AND THE LOAN TRANSACTION, <u>FOR</u> THE ELECTION OF ALL NOMINEES AS DIRECTORS, <u>FOR</u> THE ADVISORY VOTE APPROVING EXECUTIVE COMPENSATION, <u>FOR</u> HOLDING AN ADVISORY VOTE ON EXECUTIVE COMPENSATION EVERY ONE YEAR, AND <u>FOR</u> RATIFICATION OF THE APPOINTMENT OF SINGERLEWAK LLP AS THE COMPANY'S INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM FOR FISCAL YEAR ENDING MARCH 31, 2015, AND IN ACCORDANCE WITH THE JUDGMENT OF THE PERSONS NAMED AS PROXIES HEREIN ON ANY OTHER MATTERS THAT MAY PROPERLY COME BEFORE THE ANNUAL MEETING.

THIS PROXY CARD IS VALID ONLY WHEN SIGNED AND DATED.

(Continued, and to be marked, dated and signed, on the other side)

IMPORTANT NOTICE REGARDING THE AVAILABILITY OF PROXY MATERIALS FOR THE 2015 ANNUAL MEETING OF STOCKHOLDERS TO BE HELD ON [____], 2015

The Proxy Statement and our 2014 Annual Report to Stockholders are available on the internet at: www.proxyvote.com

MANDALAY DIGITAL GROUP, INC. ATTN: ANDREW SCHLEIMER 2811 CAHUENGA BOULEVARD WEST LOS ANGELES, CA 90068

TO VOTE, MARK BLOCKS BELOW IN BLUE OR BLACK INK. KEEP THIS PORTION FOR YOUR RECORDS

DETACH AND RETURN THIS PORTION ONLY PLEASE SIGN AND DATE AND RETURN PROMPTLY

THIS PROXY CARD IS VALID ONLY WHEN SIGNED AND DATED

1. Approval of the issuance and sale of (a) up to 19,500,000 shares of the Company's common stock pursuant to a merger agreement with Appia, Inc. (including shares of common stock issuable upon	4. Non-binding advisory vote on the frequency of future advisory votes on executive compensation.				
exercise of stock options to be assumed by the Company pursuant to the merger agreement), and (b) an additional 600,000 shares of the Company's common stock (including a warrant to purchase 400,000 shares) to be issued to a lender in connection with a	ONE TWO THREE □YEAR □YEARS □YEARS □ABSTAIN				
separate loan transaction, all of which aggregates to more than 20% of the Company's common stock outstanding. □FOR □AGAINST □ABSTAIN	5. Ratification of the appointment of SingerLewak LLP as the Company's independent registered public accounting firm for fiscal year ending March 31, 2015.				
	□FOR □AGAINST □ABSTAIN				
2. Election of six directors to a one-year term ending as of the Company's 2016 annual meeting of stockholders. NOMINEES:	Note : In their discretion, the proxies are authorized to act upon such other matters as may properly come before the Annual Meeting of				
01 William G. Stone 04 Peter Guber	Stockholders or any postponement(s) or adjournment(s) thereof.				
02 Robert Deutschman05 Jeffrey Karish03 Christopher Rogers06 Paul Schaeffer	Please check here ☐ if you plan to attend the meeting.				
[] FOR ALL NOMINEES [] FOR ALL EXCEPT					
To withhold authority to vote for any individual nominee(s), mark "For All Except" and write the number(s) of the nominee(s) on the line below.	For address change, mark here □ and indicate new address at right:				
3. Non-binding advisory vote approving the compensation of the Company's named executive officers.					
□FOR □AGAINST □ABSTAIN					
Date					
Please sign exactly as your name(s) appear(s) hereon. When shares ar administrator, attorney trustee or guardian, please give full title as such duly authorized officer, giving full title as such. If signer is a partnersh	h. When signing for a corporation, please sign full corporate name by				
Signature	Signature (Joint Owners)				

PLEASE MAIL IN THE ENVELOPE PROVIDED.